

Journal of Singularities Volume 11

www.journalofsing.org

Journal

of

Singularities

Volume 11 2015

ISSN: # 1949-2006

Journal of Singularities

Editor-in-Chief: David B. Massey

Managing Editors:

Paolo Aluffi Victor Goryunov Lê Dũng Tráng

Associate Editors:

Lev Birbrair Jean-Paul Brasselet Felipe Cano Torres Alexandru Dimca Terence Gaffney Sabir Gusein-Zade Helmut Hamm Kevin Houston Ilia Itenberg Françoise Michel András Némethi Mutsuo Oka Anne Pichon Maria Ruas José Seade Joseph Steenbrink Duco van Straten Alexandru Suciu Kiyoshi Takeuchi

©2015, Worldwide Center of Mathematics, LLC

Table of Contents

On real anti-bicanonical curves with one double point on the 4-th real Hirzebruch surface
Sachiko Saito
Resonant bands, Aomoto complex, and real 4-nets
On the Lojasiewicz exponents of quasi-homogeneous functions
L ² -Riemann-Roch for singular complex curves
Symmetries and stabilization for sheaves of vanishing cycles with an Appendix by Jörg Schürmann
The punctual Hilbert schemes for the curve singularities of type A_{2d}
Families of distributions and Pfaff systems under duality164Federico Quallbrunn
Free divisors in a pencil of curves

ON REAL ANTI-BICANONICAL CURVES WITH ONE DOUBLE POINT ON THE 4-TH REAL HIRZEBRUCH SURFACE

SACHIKO SAITO

ABSTRACT. We list up all the candidates for the real isotopy types of real anti-bicanonical curves with one real nondegenerate double point on the 4-th real Hirzebruch surface \mathbb{RF}_4 by enumerating the connected components of the moduli space of real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$. We also list up all the candidates for the non-increasing simplest degenerations of real nonsingular anti-bicanonical curves on \mathbb{RF}_4 . We find an interesting correspondence between the real isotopy types of real anti-bicanonical curves with one real nondegenerate double point on \mathbb{RF}_4 and the non-increasing simplest degenerations of real nonsingular anti-bicanonical curves on \mathbb{RF}_4 . This correspondence is very similar to the one provided by the rigid isotopic classification of real sextic curves on \mathbb{RP}^2 with one real nondegenerate double point by I. Itenberg.

1. INTRODUCTION

This paper is a continuation of [10] and [11]. In [10] the moduli spaces of " $(\mathcal{D}\mathbb{R})$ -nondegenerate" real K3 surfaces with non-symplectic holomorphic involutions (namely, real 2-elementary K3 surfaces) are formulated and it is shown that the connected components of such a moduli space are in **one to one correspondence** with the isometry classes of integral involutions of the K3 lattice of certain type (see Theorem 2.11 below and [10] for more precise statements). As its applications, we obtain the real isotopic classifications (more precisely, deformation classifications) of real nonsingular sextic curves on the real projective plane \mathbb{RP}^2 , real nonsingular curves of bidegree (4, 4) on the real $\mathbb{P}^1 \times \mathbb{P}^1$ (hyperboloid, ellipsoid), and real nonsingular anti-bicanonical curves on the real Hirzebruch surfaces \mathbb{RF}_m (m = 1, 4). Here the *m*-th Hirzebruch surface \mathbb{F}_m ($m \ge 0$) means the ruled surface over \mathbb{P}^1 having an exceptional section *s* with $s^2 = -m$. \mathbb{F}_4 has 2 real structures (anti-holomorphic involutions). The real part of \mathbb{F}_4 is homeomorphic to the 2-dimensional torus or the empty set. On the other hand, \mathbb{F}_3 has a unique real structure and its real part is homeomorphic to the Klein's bottle (see [2]).

We say a complex curve A on a real Hirzebruch surface is **real** if its anti-holomorphic involution can be restricted to the curve A. We say two real curves $\mathbb{R}A$, $\mathbb{R}A'$ on a real nonsingular surface $\mathbb{R}B$ are **real isotopic** if there exists a continuous map $\Phi : \mathbb{R}B \times [0,1] \to \mathbb{R}B$ (a "real isotopy from $\mathbb{R}A$ to $\mathbb{R}A'$ ") such that $\Phi_t := \Phi(\ , t) : \mathbb{R}B \to \mathbb{R}B$ is a homeomorphism for any $t \in [0,1]$, $\Phi_0 = \mathrm{id}_{\mathbb{R}B}$, and $\Phi_1(\mathbb{R}A) = \mathbb{R}A'$. Moreover, two real curves $\mathbb{R}A$, $\mathbb{R}A'$ in a fixed class on a real nonsingular surface $\mathbb{R}B$ are **rigidly isotopic** if there exists a real isotopy $\Phi : \mathbb{R}B \times [0,1] \to \mathbb{R}B$ from $\mathbb{R}A$ to $\mathbb{R}A'$ such that $\Phi_t(\mathbb{R}A)$ is contained in the same class for any $t \in [0,1]$.

Using the same method as above, we obtain the real isotopic classifications of real nonsingular anti-bicanonical curves on the real Hirzebruch surfaces \mathbb{RF}_2 and \mathbb{RF}_3 (see also Theorem 2.13 and Remark 2.26 of this paper) in [11]. Especially, all the connected components of the moduli space of real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$, which are defined below,

¹⁹⁹¹ Mathematics Subject Classification. 14J28, 14P25, 14J10.

partially supported by JSPS Grant-in-Aid for Challenging Exploratory Research 25610001.

are enumerated in [11]. Here we should remark that any real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$ are $(\mathcal{D}\mathbb{R})$ -nondegenerate in the sense of [10].

Any 2-elementary K3 surface (X, τ) of type $S \cong (3, 1, 1)$ has an elliptic fibration and a unique reducible fiber E + F, where the curves E and F are defined in [11] (see also Subsection 2.2 of this paper). There are two types of F, which are reducible and irreducible. Let A be the fixed point set (curve) of τ on X, and $\pi : X \to Y := X/\tau$ be the quotient map. A intersects $\pi(E)$ at a single point. If we contract the curve $\pi(E)$ on Y to a point, then we get the 3-th Hirzebruch surface \mathbb{F}_3 , and the image of the curve A is a real **nonsingular** anti-bicanonical curve on \mathbb{F}_3 . This enables us to enumerate up (see [11]) all the real isotopy types of real nonsingular anti-bicanonical curves on \mathbb{RF}_3 .

However, on the other hand, if we contract the curve $\pi(F)$ on Y to a point, then we get the 4-th Hirzebruch surface \mathbb{F}_4 , and the image of the curve A is real anti-bicanonical and has **one real double point**. Even though a real 2-elementary K3 surface of type $(S, \theta) \cong ((3, 1, 1), -id)$ is always $(\mathcal{D}\mathbb{R})$ -nondegenerate, the double point is possibly degenerate, namely, a real cusp point. Thus, the main difficulty of the real isotopic classification of real anti-bicanonical curves on \mathbb{F}_4 with one real double point is that the connected components (equivalently, the isometry classes of integral involutions of the K3 lattice) of the moduli space ([10]) of real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$ cannot distinguish degenerate and nondegenerate double points, equivalently, reducible F and irreducible F. Hence, the connected components cannot distinguish the topological types (node, cusp, or isolated point) of the real double points (Remark 2.25). This problem was left to the readers in [11] (see Remark 8).

Thus, the aim of this paper is: Classify the real isotopy types of real anti-bicanonical curves with one real nondegenerate double point on \mathbb{RF}_4 .

First we list up all the candidates for the real isotopy types of real anti-bicanonical curves with one real nondegenerate double point on \mathbb{RF}_4 (Theorem 2.24) using some well-known topological interpretations ([10], [11]) of some arithmetic invariants of integral involutions of the K3 lattice.

Unfortunately, the realizability of each real isotopy type listed in Theorem 2.24 has not been resolved in this paper. We only know that at least one of real isotopy types with nondegenerate double points can be realized for **each** isometry class (see Remark 2.25).

In order to distinguish the real isotopy types, we should remove real 2-elementary K3 surfaces which yield anti-bicanonical curves with degenerate double points on \mathbb{RF}_4 from the moduli space (period domain) in the sense of [10]. We follow Itenberg's argument ([4],[5],[6]) for the rigid isotopic classification of real sextic curves on \mathbb{RP}^2 with one nondegenerate double point. Lemma 4.6 provides a sufficient condition for the double point to be non-degenerate.

Moreover, according to his papers [4] and [5], real curves of degree 6 on \mathbb{RP}^2 with one nondegenerate double point are obtained by "non-increasing simplest degenerations" of real nonsingular curves of degree 6 on \mathbb{RP}^2 . Hence, we next list up the candidates for the non-increasing simplest degenerations of real nonsingular anti-bicanonical curves on \mathbb{RF}_4 (Theorem 3.4).

Then, we find an obvious interesting correspondence between the real isotopy types of curves with one real nondegenerate double point on \mathbb{RF}_4 (Theorem 2.24) and the non-increasing simplest degenerations of nonsingular curves on \mathbb{RF}_4 (Theorem 3.4). See Remark 3.5. We also get some similar properties (Lemma 3.7) to the degenerations of real nonsingular curves of degree 6 on \mathbb{RP}^2 .

In the final section 4, we review and confirm the periods of marked real 2-elementary K3 surfaces $((X, \tau, \varphi), \alpha)$ of type (S, θ) satisfying $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$ for a fixed integral involution ψ , and give some further problems which are inspired by the argument in [4] and [5].

Acknowledgments. The author would like to thank the referee for his careful reading and valuable advice, Professor Ilia Itenberg for his inspiring papers, and Professor Viacheslav Nikulin for his constant encouragement.

2. Real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$

2.1. Real 2-elementary K3 surfaces.

Definition 2.1 (Real 2-elementary K3 surface). A triple (X, τ, φ) is called a **real K3 surface with non-symplectic (holomorphic) involution** (or **real 2-elementary K3 surface**) if (1) (X, τ) is a K3 surface X with a non-symplectic holomorphic involution τ , i.e., "2-elementary K3 surface" ([8]).

(2) φ is an anti-holomorphic involution on X.

(3) $\varphi \circ \tau = \tau \circ \varphi$

Note that any K3 surface with a non-symplectic holomorphic involution is algebraic.

For a real K3 surface with non-symplectic involution (X, τ, φ) , we call $\tilde{\varphi} := \tau \circ \varphi = \varphi \circ \tau$ the **related (anti-holomorphic) involution** of φ ([10]). The triple $(X, \tau, \tau \circ \varphi)$ is also a real K3 surface with non-symplectic involution.

For a 2-elementary K3 surface (X, τ) , we denote by $H_{2+}(X, \mathbb{Z})$ the fixed part of

$$\tau_*: H_2(X, \mathbb{Z}) \to H_2(X, \mathbb{Z})$$

Note that $H_{2+}(X,\mathbb{Z}) \subset N(X)$, where N(X) is the Picard lattice of X.

We fix an even unimodular lattice \mathbb{L}_{K3} of signature (3, 19). The isometry class of such lattices is unique (the K3 lattice).

Let $\alpha: H_2(X, \mathbb{Z}) \to \mathbb{L}_{K3}$ be an isometry (marking). If we temporary set $S := \alpha(H_{2+}(X, \mathbb{Z}))$, then S is a primitive (,i.e., \mathbb{L}_{K3}/S is free,) hyperbolic (i.e., S is of signature $(1, \operatorname{rank} S - 1))$ 2-elementary (i.e., $S^*/S \cong (\mathbb{Z}/2\mathbb{Z})^a$ for some nonnegative integer a) sublattice of \mathbb{L}_{K3} .

Now let $S \quad (\subset \mathbb{L}_{K3})$ be a primitive hyperbolic 2-elementary sublattice of the K3 lattice \mathbb{L}_{K3} .

Definition 2.2. We set $r(S) := \operatorname{rank} S$. The non-negative integer a(S) is defined by the equality

$$S^*/S \cong (\mathbb{Z}/2\mathbb{Z})^{a(S)}.$$

We define

$$\delta(S) := \begin{cases} 0 & \text{if } z \cdot \sigma(z) \equiv 0 \mod 2 \quad (\forall z \in \mathbb{L}_{K3}) \\ 1 & \text{otherwise,} \end{cases}$$

where we define $\sigma : \mathbb{L}_{K3} \to \mathbb{L}_{K3}$ to be the unique integral involution whose fixed part is S.

It is known that the triplet

$$(r(S), a(S), \delta(S))$$

determines the isometry class of the lattice S([8]).

If S and S' are primitive hyperbolic 2-elementary sublattices of the K3 lattice \mathbb{L}_{K3} , and S is isometric to S'; then there exists an automorphism f of \mathbb{L}_{K3} such that f(S') = S([1], [7]). Hence, if (X, τ) and (X', τ') are two 2-elementary K3 surfaces, $\alpha : H_2(X, \mathbb{Z}) \to \mathbb{L}_{K3}$ is an isometry, and $H_{2+}(X, \mathbb{Z})$ is isometric to $H_{2+}(X', \mathbb{Z})$; then there exists an isometry (marking) $\alpha' : H_2(X', \mathbb{Z}) \to \mathbb{L}_{K3}$ such that $\alpha'(H_{2+}(X', \mathbb{Z})) = \alpha(H_{2+}(X, \mathbb{Z}))$. Thus, only the isometry class of $H_{2+}(X, \mathbb{Z})$ is essential for 2-elementary K3 surfaces (X, τ) . Henceforth, we often fix an isometry class, equivalently, the invariants $(r(S), a(S), \delta(S))$ instead of fixing a particular sublattice.

We quote the following formulation from [10]. We additionally fix a half-cone $V^+(S)$ of the cone $V(S) := \{x \in S \otimes \mathbb{R} \mid x^2 > 0\}$. We also fix a fundamental chamber $\mathcal{M} \subset V^+(S)$ for the group $W^{(-2)}(S)$ generated by reflections in all elements with square (-2) in S. This is equivalent to fixing a fundamental subdivision

$$\Delta(S) = \Delta(S)_+ \cup -\Delta(S)_+$$

of all elements with square -2 in S. \mathcal{M} and $\Delta(S)_+$ define each other by the condition

 $\mathcal{M} \cdot \Delta(S)_+ \ge 0.$

Let (X, τ) be a 2-elementary K3 surface, and $\alpha : H_2(X, \mathbb{Z}) \to \mathbb{L}_{K3}$ be a marking such that $\alpha(H_{2+}(X,\mathbb{Z})) = S$. We always assume that $\alpha_{\mathbb{R}}^{-1}(V^+(S))$ contains a hyperplane section of X and the set $\alpha^{-1}(\Delta(S)_+)$ contains only classes of effective curves of X.

Now let θ be an integral involution of S.

Definition 2.3 (the action of φ on $H_{2+}(X,\mathbb{Z})$). A real 2-elementary K3 surface (X, τ, φ) is called that of **type** (S, θ) if there exists an isometry (marking)

$$\alpha: H_2(X, \mathbb{Z}) \cong \mathbb{L}_{K3}$$

such that $\alpha(H_{2+}(X,\mathbb{Z})) = S$ and the following diagram commutes:

$$\begin{array}{ccc} H_{2+}(X,\mathbb{Z}) & \stackrel{\alpha}{\longrightarrow} & S \\ & \varphi_* & & & \downarrow \theta \\ & H_{2+}(X,\mathbb{Z}) & \stackrel{\alpha}{\longrightarrow} & S. \end{array}$$

Definition 2.4 (marked real 2-elementary K3 surfaces). A pair $((X, \tau, \varphi), \alpha)$ of a real 2elementary K3 surface (X, τ, φ) of type (S, θ) and an isometry (marking) $\alpha : H_2(X, \mathbb{Z}) \cong \mathbb{L}_{K3}$ such that

- $\alpha(H_{2+}(X,\mathbb{Z})) = S,$
- $\alpha \circ \varphi_* = \theta \circ \alpha$ on $H_{2+}(X, \mathbb{Z})$,
- $\alpha_{\mathbb{R}}^{-1}(V^+(S))$ contains a hyperplane section of X and
- the set $\alpha^{-1}(\Delta(S)_+)$ contains only classes of effective curves of X

is called a marked real 2-elementary K3 surface of type (S, θ) .

Note that we have $\theta(V^+(S)) = -V^+(S)$ and $\theta(\Delta(S)_+) = -\Delta(S)_+$ for a marked real 2elementary K3 surface of type (S, θ) .

Definition 2.5 (Integral involution ψ of \mathbb{L}_{K3} of type (S, θ)). Let S be a hyperbolic 2-elementary sublattice of \mathbb{L}_{K3} , $\theta : S \to S$ be an integral involution of the lattice S, and $\psi : \mathbb{L}_{K3} \to \mathbb{L}_{K3}$ be an integral involution of the lattice \mathbb{L}_{K3} such that the following diagram commutes:

$$\begin{array}{cccc} S & \subset & \mathbb{L}_{K3} \\ \theta \downarrow & & \downarrow \psi \\ S & \subset & \mathbb{L}_{K3}. \end{array}$$

We say such a pair (\mathbb{L}_{K3}, ψ) (or ψ) an integral involution of \mathbb{L}_{K3} of type (S, θ) .

Let $((X, \tau, \varphi), \alpha)$ be a marked real 2-elementary K3 surface of type (S, θ) . If we set

$$\psi := \alpha \circ \varphi_* \circ \alpha^{-1} : \mathbb{L}_{K3} \to \mathbb{L}_{K3},$$

then we have $\psi(S) = S$, and $\psi(x) = \theta(x)$ for every $x \in S$ because $\alpha^{-1}(x) \in H_{2+}(X, \mathbb{Z})$. Hence, (\mathbb{L}_{K3}, ψ) is an integral involution of \mathbb{L}_{K3} of type (S, θ) .

Definition 2.6 (the associated integral involution). We call the integral involution ψ of \mathbb{L}_{K3} of type (S, θ) the associated integral involution of \mathbb{L}_{K3} with a marked real 2-elementary K3 surface $((X, \tau, \varphi), \alpha)$ of type (S, θ) if the following diagram commutes:

$$\begin{array}{ccc} H_2(X,\mathbb{Z}) & \stackrel{\alpha}{\longrightarrow} & \mathbb{L}_{K3} \\ \varphi_* & & & \downarrow \psi \\ H_2(X,\mathbb{Z}) & \stackrel{\alpha}{\longrightarrow} & \mathbb{L}_{K3}. \end{array}$$

Note that the fixed part \mathbb{L}_{K3}^{ψ} of ψ is hyperbolic for any associated integral involution of \mathbb{L}_{K3} .

Definition 2.7 (($\mathcal{D}\mathbb{R}$)-nondegenerate, [10]). $x \in N(X) \otimes \mathbb{R}$ is **nef** if $x \neq 0$ and $x \cdot C \geq 0$ for any effective curve on X. We say that a 2-elementary K3 surface (X, τ) of type S is (\mathcal{D})-degenerate if there exists $h \in \mathcal{M}$ such that h is not nef. This is equivalent to the existence of an exceptional curve (i.e., irreducible and having negative self-intersection) with square -2 on the quotient surface $Y := X/\tau$. This is also equivalent to have an element $\delta \in N(X)$ with $\delta^2 = -2$ such that $\delta = (\delta_1 + \delta_2)/2$ where $\delta_1 \in S$, $\delta_2 \in S_{N(X)}^{\perp}$ and $\delta_1^2 = \delta_2^2 = -4$.

A real 2-elementary K3 surface (X, τ, φ) of type (S, θ) is $(\mathcal{D}\mathbb{R})$ -degenerate if there exists a real element $h \in S_{-} \cap \mathcal{M}$ which is not nef, where we set $S_{\pm} := \{x \in S \mid \theta(x) = \pm x\}$. This is equivalent to have an element $\delta \in N(X)$ with $\delta^2 = -2$ such that $\delta = (\delta_1 + \delta_2)/2$ where $\delta_1 \in S$, $\delta_2 \in S_{N(X)}^{\perp}$ and $\delta_1^2 = \delta_2^2 = -4$, and δ_1 must be orthogonal to an element $h \in S_{-} \cap \operatorname{int}(\mathcal{M})$ with $h^2 > 0$. Here $\operatorname{int}(\mathcal{M})$ denote the interior part of \mathcal{M} , i. e., the polyhedron \mathcal{M} without its faces.

Let $\Delta(S, L)^{(-4)}$ be the set of all elements δ_1 in S such that $\delta_1^2 = -4$ and there exists $\delta_2 \in S_L^{\perp}$ such that $(\delta_2)^2 = -4$ and $\delta = (\delta_1 + \delta_2)/2 \in L$. Let $W^{(-4)}(S, L) \subset O(S)$ be the group generated by reflections in all roots from $\Delta(S, L)^{(-4)}$, and $W^{(-4)}(S, L)_{\mathcal{M}}$ be the stabilizer subgroup of \mathcal{M} in $W^{(-4)}(S, L)$. Let G be the subgroup generated by reflections s_{δ_1} in all elements $\delta_1 \in \Delta(S, L)^{(-4)}$ which are contained either in S_+ or in S_- and satisfy $s_{\delta_1}(\mathcal{M}) = \mathcal{M}$. (Then G is a subgroup of $W^{(-4)}(S, L)_{\mathcal{M}}$.)

Definition 2.8 (Isometries with respect to the group G). Let $(\mathbb{L}_{K3}, \psi_1)$ and $(\mathbb{L}_{K3}, \psi_2)$ be two integral involutions of \mathbb{L}_{K3} of type (S, θ) . An **isometry with respect to the group** G from $(\mathbb{L}_{K3}, \psi_1)$ to $(\mathbb{L}_{K3}, \psi_2)$ means an isometry $f : \mathbb{L}_{K3} \to \mathbb{L}_{K3}$ such that $f(S) = S, f|_S \in G$, and the following diagram commutes:

$$\mathbb{L}_{K3} \xrightarrow{f} \mathbb{L}_{K3}$$

$$\downarrow \psi_1 \qquad \psi_2 \downarrow$$

$$\mathbb{L}_{K3} \xrightarrow{f} \mathbb{L}_{K3}.$$

In the above definition, remark that $\theta \circ f|_S = f|_S \circ \theta$ on S, hence, $f|_S$ is at least an automorphism of (S, θ) . However, we require the condition " $f|_S \in G$ ".

Definition 2.9. We say two integral involutions $(\mathbb{L}_{K3}, \psi_1)$ and $(\mathbb{L}_{K3}, \psi_2)$ of type (S, θ) are **isometric with respect to the group** G if there exists an isometry with respect to the group G from $(\mathbb{L}_{K3}, \psi_1)$ to $(\mathbb{L}_{K3}, \psi_2)$.

By an **automorphism** of an integral involution (\mathbb{L}_{K3}, ψ) of type (S, θ) with respect to the **group** G we mean an isometry with respect to the group G from (\mathbb{L}_{K3}, ψ) to itself. Namely, an isometry $f : \mathbb{L}_{K3} \to \mathbb{L}_{K3}$ which satisfies that

$$\psi \circ f = f \circ \psi, \ f(S) = S \text{ and } f|_S \in G$$

Definition 2.10 (analytically isomorphic with respect to G). Two marked real 2-elementary K3 surfaces $((X, \tau, \varphi), \alpha)$ and $((X', \tau', \varphi'), \alpha')$ of type (S, θ) are **analytically isomorphic** with respect to the group G if there exists an analytic isomorphism $f : X \to X'$ such that $f \circ \tau = \tau' \circ f$, $f \circ \varphi = \varphi' \circ f$ and $\alpha' \circ f_* \circ \alpha^{-1} | S \in G$.

By considering their associated integral involutions, we get a natural map from the moduli space of marked real 2-elementary K3 surfaces of type (S, θ) to the set of isometry classes with respect to G of integral involutions of \mathbb{L}_{K3} of type (S, θ) such that the fixed part \mathbb{L}_{K3}^{ψ} of ψ is hyperbolic.

Theorem 2.11 ([10]). The natural map above gives a bijective correspondence between the connected components of the moduli space of $(\mathcal{D}\mathbb{R})$ -nondegenerate marked real 2-elementary K3 surfaces of type (S, θ) and the set of isometry classes with respect to G of integral involutions of \mathbb{L}_{K3} of type (S, θ) such that the fixed part \mathbb{L}_{K3}^{ψ} of ψ is hyperbolic.

2.2. Elliptic fibrations of 2-elementary K3 surfaces of type $S \cong (3, 1, 1)$. We quote some basic facts from [1]. Let (X, τ) be a 2-elementary K3 surface of type S with invariants

$$(r(S), a(S), \delta(S)) = (3, 1, 1),$$

and $A := X^{\tau}$ be the fixed point set (curve) of τ . Then we have

 $A = A_0 \cup A_1$ (disjoint union),

where A_0 is a nonsingular rational curve $(\cong \mathbb{P}^1)$ with $A_0^2 = -2$ and A_1 is a nonsingular curve of genus 9. (X, τ) has a structure of an elliptic pencil |E + F| with its section A_0 and the unique reducible fiber E + F having the following properties below:

- (i): E is an irreducible nonsingular rational curve with $E^2 = -2$ and $E \cdot A_0 = 1$.
- (ii): $E \cdot F = 2$, $F^2 = -2$, $F \cdot A_0 = 0$, and F is either an irreducible nonsingular rational curve (type IIa), or the union of two irreducible nonsingular rational curves F' and F'' which are conjugate by τ and $F' \cdot F'' = 1$ (type IIb) (that is, the reducible fiber E + F corresponds to the extended root system $\widetilde{\mathbb{A}}_1$ or $\widetilde{\mathbb{A}}_2$). We have $(F')^2 = (F'')^2 = -2$. See also §2.4 of [1].
- (iii): The classes $[A_0]$, [E] and [F] generate the lattice $H_{2+}(X,\mathbb{Z}) \cong S$. Moreover, $A_1 \cdot E = 1, A_1 \cdot F = 2$. The Gram matrix of the lattice $H_{2+}(X,\mathbb{Z})$ with respect to the basis [E], [F] and $[A_0]$ is as follows:

$$\begin{array}{cccc} [E] & [F] & [A_0] \\ E] & -2 \\ F] & 2 & -2 \\ A_0] & 1 & 0 & -2 \end{array}$$

We next consider the quotient complex surface $Y := X/\tau$ and let $\pi : X \to Y$ be the quotient map. Then, A, considered in Y, is contained in $|-2K_Y|$ (an anti-bicanonical curve on Y).

We define the curves:

$$e := \pi(E)$$
 and $f := \pi(F)$

If F is the union of two nonsingular rational curves F' and F'' which are conjugate by τ and $F' \cdot F'' = 1$, then

$$f = \pi(F) = \pi(F' \cup F'') = \pi(F') = \pi(F'').$$

We use the same symbols A_0 and A_1 for their images in Y by π . Then, the Picard group Pic(Y) of Y is generated by the classes [e], [f] and $[A_0]$. The Gram matrix of Pic(Y) with respect to the basis [e], [f] and $[A_0]$ is as follows:

$$\begin{array}{cccc} [e] & [f] & [A_0] \\ [e] & -1 \\ [f] & 1 & -1 \\ A_0] & 1 & 0 & -4 \end{array}$$

For A_1 , we have

$$A_1 \cdot e = 1$$
 and $A_1 \cdot f = 2$.

We have:

(1) F is a nonsingular rational curve if and only if A_1 intersects f in two distinct points.

(2) F is a union of two nonsingular rational curves if and only if A_1 touches f.

[

Remark 2.12 ([11]). For any real 2-elementary K3 surface (X, τ, φ) of type (S, θ) with $S \cong (3, 1, 1)$, we have $\theta = -\text{id}$ and $G = \{\text{id}\}$, where id stands for the identity map on S.

2.3. Enumeration of the connected components of the moduli space. For any 2-elementary K3 surface (X, τ) of type $S \cong (3, 1, 1)$, all exceptional curves on Y are exactly the curves e, f and A_0 . Hence, (X, τ) is (\mathcal{D}) -nondegenerate and any real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong ((3, 1, 1), -\text{id})$ is $(\mathcal{D}\mathbb{R})$ -nondegenerate (see Definition 2.7). By Theorem 2.11, we have:

Theorem 2.13 ([11]CTheorem 1). The connected components of the moduli space (in the sense of [10]) of marked real 2-elementary K3 surfaces $((X, \tau, \varphi), \alpha)$ of type $(S, \theta) \cong ((3, 1, 1), -id)$ are in bijective correspondence with the isometry classes of integral involutions of \mathbb{L}_{K3} of type $(S, \theta) \cong ((3, 1, 1), -id)$ with respect to $G = \{id\}$.

The complete isometry invariants of integral involutions ψ of \mathbb{L}_{K3} of type $(S, \theta) \cong ((3, 1, 1), -id)$ are the data

(2.1)
$$(r(\psi), a(\psi), \delta_{\psi S}, H(\psi)).$$

See [10], Subsection 2.3 (after the equation (2.20)) for the definition of the invariant $\delta_{\psi S}$ and $H(\psi)$. Remark that $H(\psi)$ is a subgroup of the discriminant group $S_{-}/2S_{-} = S/2S = \mathbb{Z}/2\mathbb{Z}(\alpha([F]))$. Hence, $H(\psi) = 0$ or isomorphic to $\mathbb{Z}/2\mathbb{Z}$.

All realizable data (2.1) are enumerated in [11]. We have 12 data of "Type 0" ($\Rightarrow H(\psi) = 0$), 12 data of "Type Ia", 39 data of "Type Ib" with $H(\psi) = 0$, and 39 data of "Type Ib" with $H(\psi) \cong \mathbb{Z}/2\mathbb{Z}$. See also TABLE 1 for $H(\psi) = 0$ case and TABLE 2 for $H(\psi) \cong \mathbb{Z}/2\mathbb{Z}$ case, where 0 or 1 in each cell stands for the value of $\delta_{\psi S}$.

Thus we have exactly **102** connected components of the moduli of real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$. By [10], we find how invariants of an involution ψ and those of its "related integral involution" $\sigma \circ \psi$ are calculated from each other. Here $\sigma : \mathbb{L}_{K3} \to \mathbb{L}_{K3}$ is defined to be the integral involution of \mathbb{L}_{K3} whose fixed part is S. Identifying related pairs of anti-holomorphic involutions on each K3 surface, we have exactly **51** connected components.

															r			-
9									1									
8								1		0,1								
7							1		1		1							
6						1		1		0,1		1						
5					1		1		1		1		1					
4				1		0, 1		1		0, 1		1		0,1				
3			1		1		1		1		1		1		1			
2 0,1 1 0 1 0,1 1 0 1																		
0		0								0								0
$a(\psi) \ / \ r(\psi)$	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18
		m	_ 1		11 /	1 1		(()	>	()	c)	c	TT ()					

TABLE 1. All the data $(r(\psi), a(\psi), \delta_{\psi S})$ for $H(\psi) = 0$.

10										1								
9									0, 1		1							
8								1		1		1						
7							1		0, 1		1		1					
6						1		1		1		1		1				
5					0, 1		1		0, 1		1		0, 1		1			
4				1		1		1		1		1		1		1		
3			1		0		1		0, 1		1		0		1		0, 1	
2		1						1		1						1		1
1	0								0								0	
$a(\psi) \ / \ r(\psi)$	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18

TABLE 2. All the data $(r(\psi), a(\psi), \delta_{\psi S})$ for $H(\psi) \cong \mathbb{Z}/2\mathbb{Z}$.

2.4. The regions A_+ and A_- in the real part $Y(\mathbb{R})$ of Y. Let (X, τ, φ) be a real 2-elementary K3 surface of type $(S, \theta) \cong ((3, 1, 1), -\text{id})$. We use the notation in Subsection 2.2. Let $X_{\varphi}(\mathbb{R})$ denote the real part of (X, φ) , i.e., the fixed point set of φ , and let $Y(\mathbb{R})$ be the real part of the quotient surface Y with the real structure $\varphi_{\text{mod }\tau}$.

The real part

$$\mathbb{R}A = \mathbb{R}A_0 \cup \mathbb{R}A_1$$

of the branch curve A divides $Y(\mathbb{R})$ into two regions ${}^1A_+$ and A_- .

Either A_+ or A_- is doubly covered by the real part $X_{\varphi}(\mathbb{R})$, and the other by the real part $X_{\widetilde{\varphi}}(\mathbb{R})$, where we set $\widetilde{\varphi} := \tau \circ \varphi$. Since $X_{\varphi}(\mathbb{R})$ is always non-empty ([11], p.27), $Y(\mathbb{R})$, A_+ , and A_- are also non-empty. Regions A_{\pm} could have several connected components.

We distinguish two regions A_+ and A_- as follows.

Definition 2.14 (The regions A_+ and A_-). For a real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong ((3, 1, 1), -id)$, we define the regions A_+ and $A_- (\subset Y(\mathbb{R}))$ as follows:

$$\begin{cases} A_+ := \pi(X_{\varphi}(\mathbb{R})) & \text{if } H(\psi) \cong \mathbb{Z}/2\mathbb{Z}, \\ A_- := \pi(X_{\varphi}(\mathbb{R})) & \text{if } H(\psi) = 0. \end{cases}$$

¹These two regions are called "positive curves" in [10].

Note that if $H(\psi) \cong \mathbb{Z}/2\mathbb{Z}$, then $H(\sigma \circ \psi) = 0$.

An important topological characterization of the invariant $H(\psi)$ is as follows.

Lemma 2.15 ([10]). Suppose that the curve F is irreducible. Then we have $H(\psi) \cong \mathbb{Z}/2\mathbb{Z}$ if and only if $[F_{\varphi}(\mathbb{R})] = 0$ in $H_1(X_{\varphi}(\mathbb{R}); \mathbb{Z}/2\mathbb{Z})$, where we set $F_{\varphi}(\mathbb{R}) := X_{\varphi}(\mathbb{R}) \cap F$.

2.5. Real anti-bicanonical curves with one real double point on \mathbb{RF}_4 . We now contract the exceptional curve $f = \pi(F)$ to a point. We get a blow up bl : $Y \to \mathbb{F}_4$, where \mathbb{F}_4 is the 4-th Hirzebruch surface.

We set

$$s := bl(A_0), \quad A'_1 := bl(A_1), \quad c := bl(e),$$

and we have

$$bl(A) = bl(A_0) + bl(A_1) = s + A'_1 \in |-2K_{\mathbb{F}_4}|,$$

i.e., bl(A) is an anti-bicanonical curve, where s is the exceptional section of \mathbb{F}_4 with $s^2 = -4$ and c is a fiber of the fibration $\mathbb{F}_4 \to s$ with $c^2 = 0$. Since $s \cdot A'_1 = 0$, A'_1 does not intersect the section s.

We have $-2K_{\mathbb{F}_4} \sim 12c + 4s$. It follows that $A'_1 \in |12c + 3s|$.

Let

$$\mathbb{R}\mathrm{bl}(A) = \mathbb{R}s \cup \mathbb{R}A_1'$$

be the real part of the curve bl(A), where $\mathbb{R}s$ and $\mathbb{R}A'_1$ are the real parts of s and A'_1 respectively.

We set

$$P_0 := \operatorname{bl}(f) \ (\in \mathbb{R}c).$$

Since $A_1 \cdot f = 2$ in Y, A'_1 has one real double point P_0 . If A_1 intersects with f at two distinct points in Y, then they are real points or non-real conjugate points. In the former case P_0 is a real node, and in the latter case it is a real isolated point. Anyway it is a nondegenerate double point. If A_1 touches to f in Y, then P_0 is a real cusp. (a degenerate double point)

Thus, there are three topological types of the curve $\mathbb{R}A'_1$ near the double point P_0 .

Node case: P_0 is a real node of $\mathbb{R}A'_1$.

Cusp case: P_0 is a real cusp of $\mathbb{R}A'_1$. (degenerate double point)

Isolated point case: P_0 is a real isolated point of $\mathbb{R}A'_1$.

See Figure 1 below.

We have $A'_1 \cdot s = 0$ and $A'_1 \cdot c = 3$ (A'_1 is a trigonal curve). Since $f \cdot A_0 = 0$ in Y, the section s does not meet the double point P_0 . We may assume that e does not pass through any intersection point of A_1 and f in Y. (See Figure 1.) Since $A_1 \cdot e = 1$ in Y, the intersection point of A_1 with e is **real** and does not meet f. Via the map bl, this point goes to a real intersection point of A'_1 with c with multiplicity 1. Thus we set

 $P_1 := \operatorname{bl}(A_1 \cap e).$ (the intersection point)

Since $A'_1 \cdot c = 3$, A'_1 intersects with c at P_0 with multiplicity 2 and at P_1 with multiplicity 1.

Since $A'_1 \cap s = \emptyset$, any **non-contractible (possibly real singular)** components of $\mathbb{R}A'_1$ are "parallel" to $\mathbb{R}s$. See Figure 2, 3 and 4. But two types of non-contractible singular components in Figure 2 are real isotopic, and three types of non-contractible components in Figure 4 are real isotopic.

Definition 2.16. We call a connected component of $\mathbb{R}A'_1$ an **oval** if it has no real singular points and contractible in $\mathbb{R}\mathbb{F}_4$ (a torus), namely, realizes 0 in $H_1(\mathbb{R}\mathbb{F}_4;\mathbb{Z})$.



FIGURE 1. The real double point P_0 of the curve A'_1



FIGURE 2. A non-contractible component with a node of type Node (1)



FIGURE 3. A non-contractible component with a node of type Node (*)

Remark 2.17. Since $A'_1 \cdot c = 3$, A'_1 is a **trigonal** curve on \mathbb{F}_4 . See [3], [12] for related results. We see that the interior of each oval of $\mathbb{R}A'_1$ does not contain any other ovals. Moreover, the ovals are canonically ordered according to their projections to the base $\mathbb{R}s$.

We now get the following possibilities.



FIGURE 4. A non-contractible component with a real cusp or without singular points

Node case:

Node (1) case: We consider the case when both the node P_0 and the intersection point P_1 are contained in the same connected singular component of $\mathbb{R}A'_1$. Then $\mathbb{R}A'_1$ meets $\mathbb{R}c$ at only P_0 and P_1 , and the singular component is not contractible. See Figure 2.

Note that $\mathbb{R}A'_1$ might have some ovals. Since $A'_1 \cdot c = 3$, the interior of any oval of $\mathbb{R}A_1$ does not contain other ovals. The interior of the node also does not contain ovals.

The non-contractible singular component containing P_0 and P_1 and the section $\mathbb{R}s$ divide $\mathbb{R}\mathbb{F}_4$ (a torus) into three parts, which are the interior of the node and two non-contractible regions.

Definition 2.18 (The regions R_1 and R_2 in Node (1) case). Let R_1 denote the non-contractible region which is connected with the interior of the node in the blow up of \mathbb{RF}_4 , and let R_2 denote the other non-contractible region. We define the integers α and β as follows:

(2.2)
$$\begin{aligned} \alpha &:= \# \{ \text{ovals contained in } R_1 \}, \\ \beta &:= \# \{ \text{ovals contained in } R_2 \}. \end{aligned}$$

See the left figure of Figure 5.



FIGURE 5. The regions R_1 and R_2 in Node (1) or Node (2) case

Node (2) case and Node (*) case: When the node P_0 and the intersection point P_1 respectively are contained in **different** connected components of $\mathbb{R}A'_1$, the component containing P_1 is nonsingular and not contractible like the rightmost figure of Figure 4 above. The component containing P_0 can be either **contractible** (the left figure of Figure 6) or **non-contractible** (Figure 3).



FIGURE 6. Contractible components with a real node or a real cusp

Node (2) case: When the component containing the node P_0 is contractible (the left figure of Figure 6), $\mathbb{R}A'_1$ might have some ovals. The component containing P_1 and the section $\mathbb{R}s$ divide $\mathbb{R}\mathbb{F}_4$ into two regions.

Definition 2.19 (The regions R_1 and R_2 in Node (2) case). Let R_1 denote the region which does not contain the contractible component containing the node P_0 , and let R_2 denote the other region. (Since $A'_1 \cdot c = 3$, the interior of the contractible component containing P_0 and the interior of any oval of $\mathbb{R}A_1$ cannot contain any other ovals.)

We define the integers α and β by (2.2). See **the right** figure of Figure 5.

Node (*) case: If the component containing the node P_0 is non-contractible (see Figure 3), then $\mathbb{R}A'_1$ has no ovals (see Figure 7).



FIGURE 7. Node (*)

Cusp case:

Cusp (1): When both the cusp P_0 and the point P_1 are contained in the same connected component of $\mathbb{R}A'_1$, $\mathbb{R}A'_1$ meets $\mathbb{R}c$ at only P_0 and P_1 , and the component containing P_0 and P_1 is not contractible (the leftmost figure of Figure 4). $\mathbb{R}A'_1$ might have some ovals. The non-contractible component containing P_0 and P_1 and the section $\mathbb{R}s$ divide $\mathbb{R}\mathbb{F}_4$ into two regions (the left figure of Figure 8). One of these regions goes to a non-orientable region via the blow up of $\mathbb{R}\mathbb{F}_4$.

Definition 2.20 (The regions R_1 and R_2 in Cusp (1) case). Let R_2 denote the region which goes to a **non-orientable region** via the blow up of \mathbb{RF}_4 . and let R_1 denote the other region. (Since $A'_1 \cdot c = 3$, the interior of any oval of $\mathbb{R}A_1$ does not contain any other ovals.)

We define the integers α and β by (2.2). See the left figure of Figure 8.



FIGURE 8. The regions R_1 and R_2 in Cusp (1) or Cusp (2) case

Cusp (2) When the cusp P_0 and P_1 respectively are contained in **different** connected components of $\mathbb{R}A'_1$, the component containing P_1 is **not contractible** (the rightmost figure of Figure 4). The component containing the cusp P_0 should be **contractible** (the right figure of Figure 6). (If not, then this component would be like the middle figure of Figure 4 and the number of the intersection points of $\mathbb{R}A'_1$ with $\mathbb{R}c$ would be even. This contradicts with $A'_1 \cdot c = 3$.) $\mathbb{R}A'_1$ might have some ovals. **The component containing** P_1 and **the section** $\mathbb{R}s$ divide $\mathbb{R}\mathbb{F}_4$ into two regions.

Definition 2.21 (The regions R_1 and R_2 in Cusp (2) case). Let R_1 denote the region which does not contain the contractible component which contains the cusp P_0 , and let R_2 denote the other region. (the right figure of Figure 8) (Since $A'_1 \cdot c = 3$, the interior of any oval of $\mathbb{R}A_1$ does not contain any ovals.)

We define the integers α and β by (2.2). See the right figure of Figure 8 above.

Isolated point case:

In this case the connected component containing P_1 is nonsingular and non-contractible like the right figure of Figure 4. $\mathbb{R}A'_1$ might have some ovals. The component containing P_1 and the section $\mathbb{R}s$ divide $\mathbb{R}F_4$ into two regions.

Definition 2.22 (The regions R_1 and R_2 in Isolated point case). Let R_1 denote the region which **does not contain the isolated point**, and let R_2 denote the other region. (Since $A'_1 \cdot c = 3$, the interior of any oval of $\mathbb{R}A_1$ does not contain any other ovals.)

We define the integers α and β by (2.2). See Figure 9.



FIGURE 9. The regions R_1 and R_2 in **Isolated point** case

From the above argument, we have:

Proposition 2.23. The real isotopy type of the singular connected component, which has one double point, of the curve $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$ is one of the following 6 types:

- Node (1), Node (2), Node (*),
- Cusp (1), Cusp (2), or
- Isolated point.

See TABLE 3. \Box



TABLE 3. Real isotopy types of the singular component of the curve $\mathbb{R}A'_1$.

2.6. Topology of the real parts of K3 surfaces X viewed via the blow up $Y \to \mathbb{F}_4$. We determine the topology of the real parts $X_{\varphi}(\mathbb{R})$ and $X_{\widetilde{\varphi}}(\mathbb{R})$ of a real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong ((3, 1, 1), -\mathrm{id})$, and the real part $Y(\mathbb{R})$ of the quotient surface Y with the real structure $\varphi_{\mathrm{mod} \tau}$. Recall Proposition 2.23.

I. Node (1), Cusp (1) and Isolated point cases.

In these cases, by the definitions of the regions A_{\pm} and R_1 , R_2 , we see that:

- A_+ is homeomorphic to the disjoint union of (an annulus with α holes) and (β disks), and
- A_{-} is homeomorphic to the disjoint union of (((an annulus $\backslash D^2) \cup$ **Möbius band**) with β holes) and (α disks).

14

For example, see Figure 10 for Node (1) case.



FIGURE 10. The regions A_+ and A_- in Node (1) case

Suppose that the invariant $H(\psi) = 0$ for the involution φ .

If F is **irreducible**, then we have $[X_{\varphi}(\mathbb{R}) \cap F] \neq 0$ and $\pi(X_{\varphi}(\mathbb{R})) = A_{-}$. On the other hand, for $\widetilde{\varphi}$, we have $[X_{\widetilde{\varphi}}(\mathbb{R}) \cap F] = 0$ and $\pi(X_{\widetilde{\varphi}}(\mathbb{R})) = A_+$.

We can say that

Thus we have

$$X_{\varphi}(\mathbb{R}) \sim \Sigma_{2+\beta} \cup \alpha S^2.$$

Moreover, we have $(r(\psi), a(\psi), \delta_{\psi S}) \neq (10, 10, 0), (10, 8, 0),$

$$r(\psi) = 9 + \alpha - \beta, \quad a(\psi) = 9 - \alpha - \beta.$$

On the other hand, we have

$$X_{\widetilde{\varphi}}(\mathbb{R}) \sim \Sigma_{1+\alpha} \cup \beta S^2.$$

Hence, we have $H(\sigma \circ \psi) \cong \mathbb{Z}/2\mathbb{Z}$ and

$$r(\sigma \circ \psi) = 10 - \alpha + \beta, \quad a(\sigma \circ \psi) = 10 - \alpha - \beta.$$

We omit the cusp and isolated point cases.

II. Node (2) and Cusp (2) cases.

In these cases, by the definitions of the regions A_{\pm} and R_1 , R_2 , we see that

- A_+ is homeomorphic to the disjoint union of (an annulus with α holes) and ((β + 1) disks), and
- A_{-} is homeomorphic to the disjoint union of (((an annulus $\backslash D^2) \cup$ **Möbius band**) with $(\beta + 1)$ holes) and $(\alpha$ disks).

For example, See Figure 11 for Node (2) case.



FIGURE 11. The regions A_+ and A_- in Node (2) case

Suppose that $A_{-} = \pi(X_{\varphi}(\mathbb{R}))$, namely, the invariant $H(\psi) = 0$ for the involution φ . Then we have

$$X_{\varphi}(\mathbb{R}) \sim \Sigma_{2+(\beta+1)} \cup \alpha S^2$$

Moreover, we have $(r(\psi), a(\psi), \delta_{\psi S}) \neq (10, 10, 0), (10, 8, 0),$

$$r(\psi) = 8 + \alpha - \beta, \quad a(\psi) = 8 - \alpha - \beta.$$

On the other hand, we have $A_+ = \pi(X_{\widetilde{\varphi}}(\mathbb{R}))$ and

$$X_{\widetilde{\varphi}}(\mathbb{R}) \sim \Sigma_{1+\alpha} \cup (\beta+1)S^2,$$

Moreover, we have $H(\sigma \circ \psi) \cong \mathbb{Z}/2\mathbb{Z}$ and

$$r(\sigma \circ \psi) = 11 - \alpha + \beta, \quad a(\sigma \circ \psi) = 9 - \alpha - \beta.$$

We omit the cusp cases.

16

III. Node (*) case.

In this case, we see that

- A₊ is homeomorphic to D² \ 2D² and
 A_− is the disjoint union of an Möbius band and an annulus.

See Figure 12.



FIGURE 12. The regions A_+ and A_- in **Node (*)** case

Suppose that, for φ , $A_{-} = \pi(X_{\varphi}(\mathbb{R}))$. Then we see that

$$X_{\varphi}(\mathbb{R}) \sim T^2 \cup T^2,$$

and

$$A_+ = \pi(X_{\widetilde{\varphi}}(\mathbb{R})), \ \ X_{\widetilde{\varphi}}(\mathbb{R}) \sim \Sigma_2.$$

Moreover, we have

$$H(\psi) = 0$$
 and $(r(\psi), a(\psi), \delta_{\psi S}) = (10, 8, 0).$

On the other hand, we have

$$H(\sigma \circ \psi) \cong \mathbb{Z}/2\mathbb{Z}$$

and

$$(r(\sigma \circ \psi), a(\sigma \circ \psi), \delta_{\sigma \circ \psi S}) = (9, 9, 0).$$

2.7. Real isotopy types of real anti-bicanonical curves $\mathbb{R}bl(A)$ with one real double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$. In Node and Cusp cases, the number of connected components of $\mathbb{R}A'_1$ equals to that of connected components of $\mathbb{R}A_1$. Hence, we have

 $1 \leq \#\{\text{Connected components of } \mathbb{R}A_1'\} \leq 10.$

In Node (1) and Cusp (1) cases, we have

$$0 \le \alpha + \beta \le 9.$$

If for φ , $A_{-} = \pi(X_{\varphi}(\mathbb{R}))$, namely, the invariant $H(\psi) = 0$, then $\alpha + \beta = 9 - a(\psi)$.

In Node (2) and Cusp (2) cases, we have

$$0 \le \alpha + \beta \le 8.$$

If for φ , $A_{-} = \pi(X_{\varphi}(\mathbb{R}))$, namely, the invariant $H(\psi) = 0$, then $\alpha + \beta = 8 - a(\psi)$.

In Isolated point case, the number of connected components of $\mathbb{R}A'_1$ equals to that of connected components of $\mathbb{R}A_1$ plus 1.

Hence, we have

 $2 \leq \#$ {Connected components of $\mathbb{R}A'_1$ } ≤ 11 .

Hence, we have

$$0 \le \alpha + \beta \le 9.$$

If for φ , $A_{-} = \pi(X_{\varphi}(\mathbb{R}))$, namely, the invariant $H(\psi) = 0$, then $\alpha + \beta = 8 - a(\psi)$.

We already have all the isometry classes. Recall TABLE 1 AND TABLE 2 in Subsection 2.3.

Theorem 2.24. We have the following.

- For each isometry class with H(ψ) = 0, the real isotopy type of a real anti-bicanonical curve Rbl(A) = Rs ∪ RA'₁ on RF₄ with one real nondegenerate double point on RA'₁ on RF₄ is one of the data listed up in Table 4.
- For each isometry class with H(ψ) ≃ Z/2Z, the real isotopy type of a real anti-bicanonical curve Rbl(A) = Rs ∪ RA'₁ on RF₄ with one real nondegenerate double point on RA'₁ on RF₄ is one of the data listed up in Table 5.

Note that the isometry class No.k and the isometry class No.k' are related integral involutions for each k = 1, ..., 50. The isometry class $(10, 8, 0, H(\psi) = 0)$ and $(9, 9, 0, H(\psi) \cong \mathbb{Z}/2\mathbb{Z})$ are also related integral involutions. \Box

Isom	etry cla	ss				Noc	de (1)	Isola	ated point	Noc	de (2)	Node (*)
of typ	pe ((3,1	1, 1), -io	d)									
No.	$r(\psi)$	$a(\psi)$	$\delta_{\varphi S}$	g	k	α	β	α	β	α	β	
1	1	1	1	10	0	0	8	0	8	0	7	
2	2	0	0	10	1	1	8	1	8	1	7	
3	2	2	0	9	0	0		0	7	0	6	
4	2	2	1	9	0	0	7	0	7	0	6	
5	3	1	1	9	1	1	7	1	7	1	6	
6	3	3	1	8	0	0	6	0	6	0	5	
7	4	2	1	8	1	1	6	1	6	1	5	
8	4	4	1	1	0	1	5	1	5	1	4	
9	5	3	1	6		1	0 4	1	3 4	1	4	
10	5		1	7	0	0	4	0	4	0	3	
11	6	<u> </u>	0	6		1	<u>J</u>	2		4	4 9	
12	6	4	1	6	1	1	4	1	4	1	3 3	
14	6		1	5	1	0	3	0	3	0	2	
15	7	3	1	6	$\frac{0}{2}$	2	4	2	4	2	3	
16	7	5	1	5	1	1	3	1	3	1	2	
17	7	7	1	4	0	0	2	0	2	0	1	
18	8	2	1	6	3	3	4	3	4	3	3	
19	8	4	1	5	$\frac{0}{2}$	2	3	$\frac{1}{2}$	3	$\frac{0}{2}$	2	
20	8	6	1	4	-	-	2	1	2	-	1	
21	8	8	1	3	0	0	- 1	0	1	0	0	
22	9	1	1	6	4	4	4	4	4	4	3	
23	9	3	1	5	3	3	3	3	3	3	2	
24	9	5	1	4	2	2	2	2	2	2	1	
25	9	7	1	3	1	1	1	1	1	1	0	
26	9	9	1	2	0	0	0	0	0			
27	10	0	0	6	5	5	4	5	4	5	3	
28	10	2	0	5	4	4	3	4	3	4	2	
29	10	2	1	5	4	4	3	4	3	4	2	
30	10	4	0	4	3	3	2	3	2	3	1	
31	10	4	1	4	3	3	2	3	2	3	1	
32	10	6	0	3	2	2	1	2	1	2	0	
33	10	6	1	3	2	2	1	2	1	2	0	
	10	8	0	2	1							$T^2 \cup T^2$
34	10	8	1	2	1	1	0	1	0			
35	11	1	1	5	5	5	3	5	3	5	2	
36	11	3	1	4	4	4	2	4	2	4	1	
37	11	5	1	3	3	3	1	3	1	3	0	
38	11	7	1	2	2	2	0	2	0			
39	12	2	1	4	5	5	2	5	2	5	1	
40	12	4	1	3	4	4	1	4	1	4	0	
41	12	6	1	2	3	3	0	3	0	-		
42	13	3			5	5	1	5	1	5	0	
43	13	5		2	4	4	0	4	0	0	6	
44	14	2			6	6	1	6	1	6	0	
45	14	4	0	$\frac{2}{2}$	5	5	0	5	0			
46	14	4		2	5	5	0	5 C	0			
47	15	3	1	2	6	6	0	6	0			
48	16	2	1	2	1	1	0	1	0			
49	17	1		2	8	8	0	8	0			
50	18	0	0	2	9	9	U	9	0			

TABLE 4. Candidates for real isotopy types of real anti-bicanonical curves $\mathbb{R}bl(A)$ with one real **nondegenerate** double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$ for each isometry class of type $(S, \theta) \cong ((3, 1, 1), -\mathrm{id})$ with $H(\psi) = 0$

Isom	etry cla	ss				Noc	le (1)	Isola	ted point	Nod	le (2)	Node (*)
of typ	pe((3, 1))	(, 1), -ic	d) (b									
No.	$r(\psi)$	$a(\psi)$	$\delta_{\varphi S}$	g	k	α	β	α	β	α	β	
1'	18	2	1	1	8	0	8	0	8	0	7	
2'	17	1	0	2	8	1	8	1	8	1	7	
3'	17	3	0	1	7	0	7	0	7	0	6	
4'	17	3	1	1	7	0	7	0	7	0	6	
5'	16	2	1	2	7	1	7	1	7	1	6	
6'	16	4	1	1	6	0	6	0	6	0	5	
7'	15	3	1	2	6	1	6	1	6	1	5	
8'	15	5	1	1	5	0	5	0	5	0	4	
9'	14	4	1	2	5	1	5	1	5	1	4	
10'	14	6	1	1	4	0	4	0	4	0	3	
11'	13	3	0	3	5	2	5	2	5	2	4	
12'	13	5	0	2	4	1	4	1	4	1	3	
13'	13	5	1	2	4	1	4	1	4	1	3	
14'	13	7	1	1	3	0	3	0	3	0	2	
15'	12	4	1	3	4	2	4	2	4	2	3	
16'	12	6	1	2	3	1	3	1	3	1	2	
17'	12	8	1	1	2	0	2	0	2	0	1	
18'	11	3	1	4	4	3	4	3	4	3	3	
19'	11	5	1	3	3	2	3	2	3	2	2	
20'	11	7	1	2	2	1	2	1	2	1	1	
21'	11	9	1	1	1	0	1	0	1	0	0	
22'	10	2	1	5	4	4	4	4	4	4	3	
23'	10	4	1	4	3	3	3	3	3	3	2	
24'	10	6	1	3	2	2	2	2	2	2	1	
25'	10	8	1	2	1	1	1	1	1	1	0	
26'	10	10	1	1	0	0	0	0	0			
27'	9	1	0	6	4	5	4	5	4	5	3	
28'	9	3	0	5	3	4	3	4	3	4	2	
29'	9	3	1	5	3	4	3	4	3	4	2	
30'	9	5	0	4	2	3	2	3	2	3	1	
31'	9	5	1	4	2	3	2	3	2	3	1	
32'	9	7	0	3	1	2	1	2	1	2	0	
33'	9	7	1	3	1	2	1	2	1	2	0	
	9	9	0	2	0	1	0	1	0			Σ_2
34'	9	9	1	2	0	1	0	1	0			_
35'	8	2	1	6	3	5	3	5	3	5	2	
36'	8	4	1	5	2	4	2	4	2	4	1	
37'	8	6	1	4	1	3	1	3	1	3	0	
38'	8	8	1	3	0	2	0	2	0	-	-	
39'	7	3	1	6	2	5	2	5	2	5	1	
40'	7	5	1	5	1	4	1	4	1	4	0	
41'	7	7	1	4	0	3	0	3	0	-		
42'	6	4	1	6	1	5	1	5	1	5	0	
43'	6	6	1	5	0	4	0	4	0	~	~	
44'	5	3	0	7	1	6	1	6	1	6	0	
45'	5	5		6	0	5	0	5	0	~	~	
46'	5	5	1	6	0	5	0	5	0			
47'	4	4	1	7	0	6	0	6	0			
48'	3	3	1	8	0	7	0	7	0			
49'	2	2	1	q	0	8	0	8	0			
50'	1	1	0	10	0	9	0	9	0			
00	1	1		1 10			~		~			1

TABLE 5. Candidates for real isotopy types of real anti-bicanonical curves \mathbb{R} bl(A) with one real **nondegenerate** double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$ for each isometry class of type $(S, \theta) \cong ((3, 1, 1), -\mathrm{id})$ with $H(\psi) \cong \mathbb{Z}/2\mathbb{Z}$

Remark 2.25. By Theorem 2.24, we find that each isometry class of integral involutions of the K3 lattice \mathbb{L}_{K3} of type $(S, \theta) \cong ((3, 1, 1), -id)$ may contain several real isotopy types (for example, Node (1), Isolated point, and Node (2)) of real anti-bicanonical curves $\mathbb{R}bl(A)$ with one real nondegenerate double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$. Hence, the realizability of all the real isotopy types listed in Table 4,5 has not been solved yet. However, we can take a real 2-elementary K3 surface for which F is irreducible in the same connected component of the moduli (see [11], notes after Theorem 2). Hence, for each isometry class, at least one of real isotopy types is realizable. Especially, Node (*) with the isometry class (10, 8, 0, $H(\psi) = 0$) is realizable. It is conjectured that Node (1) with $(\alpha, \beta) = (1, 0)$ and Isolated point with $(\alpha, \beta) = (1, 0)$ do not exist for the isometry class (9, 9, 0, $H(\psi) \cong \mathbb{Z}/2\mathbb{Z})$. See also Remark 3.5 below.

Remark 2.26 ([11]). On the other hand, if we contract the exceptional curve $e = \pi(E)$ in Y, then we get a map onto the 3-th Hirzebruch surface \mathbb{F}_3 : $\mathrm{bl}_1: Y \to \mathbb{F}_3$. Then $s := \mathrm{bl}_1(A_0)$ is the exceptional section of \mathbb{F}_3 with $s^2 = -3$ and $c := \mathrm{bl}_1(f)$ is a fiber. We have

$$bl_1(A) = s + bl_1(A_1) \in |-2K_{\mathbb{F}_3}|.$$

 $A_1 := bl_1(A_1)$ is a real **nonsingular** curve of genus 9. The real isotopic classification of $\mathbb{R}A_1$ on $\mathbb{R}F_3$ was already done in [11], Theorem 1.

In order to **distinguish** the real isotopy types of real anti-bicanonical curves with one real double point on \mathbb{RF}_4 , we expect that Itenberg's argument of the rigid isotopic classification of real curves of degree 6 on \mathbb{RP}^2 with one nondegenerate double point are helpful. See [4],[5] and [6], and also the last section 4 of this paper. This classification corresponds to that of real **nonsingular** curves A in $|-2K_{\mathbb{F}_1}|$ on the first real Hirzebruch surface \mathbb{RF}_1 when we blow up \mathbb{P}^2 at the nondegenerate double point to \mathbb{F}_1 . The double coverings X of \mathbb{F}_1 ramified along the nonsingular curves A are real 2-elementary K3 surfaces of type $(S,\theta) \cong (\langle 2 \rangle \oplus \langle -2 \rangle, -id)$. Moreover, the classification of real curves of degree 6 on \mathbb{RP}^2 with one nondegenerate double point is related to that of "non-increasing simplest degenerations" (conjunctions and contractions) of real **nonsingular** curves of degree 6 on \mathbb{RP}^2 . See [4] and [5].

Thus we next pay attention to the degenerations of real nonsingular anti-bicanonical curves on \mathbb{RF}_4 .

3. Degenerations of nonsingular real anti-bicanonical curves on \mathbb{RF}_4

3.1. Review of nonsingular real anti-bicanonical curves on \mathbb{RF}_4 . The contents of this subsection are quoted from the last section of [10].

Let \mathbb{U} be the even unimodular lattice of signature (1,1) (the hyperbolic plane). Consider real 2-elementary K3 surfaces (X, τ, φ) of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$. All these real 2-elementary K3 surfaces are (\mathcal{D}) -nondegenerate.

Let A be the fixed point set of τ . Then A is a real nonsingular curve. We have $Y := X/\tau = \mathbb{F}_4$. Let $\pi : X \to \mathbb{F}_4$ be the quotient map. We use the same symbol A for its image in \mathbb{F}_4 by π . Then $A \in |-2K_{\mathbb{F}_4}|$. Let s be the exceptional section with $s^2 = -4$ of \mathbb{F}_4 , and c the fiber of the fibration $f : \mathbb{F}_4 \to s$, where $c^2 = 0$. We have $-2K_{\mathbb{F}_4} \sim 12c + 4s$. The nonsingular curve A has two irreducible components s and A_1 ;

$$A = s \cup A_1$$
 (disjoint union).

Conversely, any nonsingular curve A_1 in |12c + 3s| gives a nonsingular curve $A = s + A_1$ in $|-2K_{\mathbb{F}_4}|$.

We set $C := \pi^*(c)$ and $E := \pi^*(s)/2$ in $H_2(X, \mathbb{Z})$. Then $C^2 = 0$, $E^2 = -2$ and $C \cdot E = 1$. Cand E generate the fixed part of $\tau_* : H_2(X, \mathbb{Z}) \to H_2(X, \mathbb{Z})$. Hence, $S \cong \mathbb{Z}C + \mathbb{Z}E \cong \mathbb{U}$.

An isometry class of an integral involution (\mathbb{L}_{K3}, ψ) of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ is determined (see [10]) by the data

(3.1)
$$(r(\psi), a(\psi), \delta_{\psi} = \delta_{\psi S}).$$

The complete list of the data (3.1) is given in Section 7 (Figure 30) of [10].

There are 14 isometry classes with $\delta_{\psi} = 0$ and 49 isometry classes with $\delta_{\psi} = 1$. Thus we have 63 classes.

If we identify related integral involutions, there are 10 isometry classes with $\delta_{\psi} = 0$ and 27 isometry classes with $\delta_{\psi} = 1$. Thus we have 37 classes.

If $(r(\psi), a(\psi), \delta_{\psi}) = (10, 10, 0)$, then $\mathbb{RF}_4 = \emptyset$. Hence, $\mathbb{R}A = \emptyset$.

If $(r(\psi), a(\psi), \delta_{\psi}) \neq (10, 10, 0)$, then \mathbb{RF}_4 is not empty and homeomorphic to a 2-torus. We have $\mathbb{R}A \supset \mathbb{R}s \neq \emptyset$. The real curve $\mathbb{R}A_1$ is contained in the open cylinder $\mathbb{RF}_4 \setminus \mathbb{R}s$. The region $A_- = \pi(X_{\varphi}(\mathbb{R}))$ with the invariants (3.1) has the real isotopy type given in Figure 13.

When $(r(\psi), a(\psi), \delta_{\psi}) \neq (10, 10, 0)$ and $\neq (10, 8, 0)$, we set

$$g := (22 - r(\psi) - a(\psi))/2$$
 and $k := (r(\psi) - a(\psi))/2.$



FIGURE 13. A region A_{-} with $(r(\psi), a(\psi), \delta_{\psi}) \neq (10, 8, 0), (10, 10, 0)$ and a region A_{-} with $(r(\psi), a(\psi), \delta_{\psi}) = (10, 8, 0)$

Since all real 2-elementary K3 surfaces of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ are (\mathcal{D}) -nondegenerate, by Theorem 2.11, we have:

Theorem 3.1 ([10], Theorem 27). A connected component of the moduli of the regions

$$A_{-} := \pi(X_{\varphi}(\mathbb{R})),$$

curves $A \in [-2K_{\mathbb{F}_4}]$, up to the action of the automorphisms group of \mathbb{F}_4 over \mathbb{R} is determined by the data (3.1), equivalently, the real isotopy types of A_- and the invariants $\delta_{\psi} = \delta_{\psi S}$. All the data are given in Section 7 (Figure 30) of [10]. See also Figure 13. 3.2. Degenerations of nonsingular real anti-bicanonical curves on \mathbb{RF}_4 . We next introduce the notions of "non-increasing simplest degenerations" (conjunctions, contractions) of nonsingular real anti-bicanonical curves on \mathbb{RF}_4 as analogies of Section 3 (pp.284–285) of [4].

As stated in the previous subsection, a nonsingular curve in $|-2K_{\mathbb{F}_4}|$ has two irreducible components s and A_1 , where A_1 is a nonsingular curve in |12c + 3s|.

Let C_0 be a real curve in |12c+3s| on \mathbb{F}_4 with one nondegenerate double point, and

$$C_t \quad (-\varepsilon < t < \varepsilon)$$

be a smoothing (, where every C_t $(t \neq 0)$ is a real nonsingular curve in |12c + 3s| on \mathbb{F}_4 ,) such that

 \sharp {ovals of a nonsingular curve $C_{t_{-1}}$ } $\geq \sharp$ {ovals of a nonsingular curve C_{t_1} }

for any $t_{-1} < 0$ and any $t_1 > 0$.

We call such a family C_t $(t_{-1} \le t \le 0)$ a non-increasing simplest degeneration of $C_{t_{-1}}$ to C_0 .

We do not know whether non-increasing simplest degenerations of C_{t-1} to C_0 are realizable for any pair C_{t-1} and C_0 .

We define 8 kinds of non-increasing simplest degenerations.

Definition 3.2 (Conjunctions 1), 2), 1'), 2') and Contractions 3), 3')). First we fix **an isometry class** of integral involutions of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ with $(r(\psi), a(\psi), \delta_{\psi}) \neq (10, 8, 0)$ and $\neq (10, 10, 0)$.

Take a corresponding real 2-elementary K3 surfaces (X, τ, φ) of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$. Let $\pi : X \to X/\tau = \mathbb{F}_4$ be the quotient map. Then we get a real nonsingular curve $A = s + A_1$ in $|-2K_{\mathbb{F}_4}|$ where A_1 is a real nonsingular curve in |12c+3s| on \mathbb{RF}_4 . Suppose that the fixed point set $X_{\varphi}(\mathbb{R})$ is homeomorphic to $\Sigma_g \cup kS^2$. Then the region $\pi(X_{\varphi}(\mathbb{R})) (\subset \mathbb{RF}_4)$ is the disjoint union of an annulus with (g-1) holes and k disks. The boundary of the annulus with (g-1) holes consists of (see Figure 13) the non-contractible component of $\mathbb{R}A_1$, $\mathbb{R}s$, and the (g-1) empty ovals.

- Conjunction 1) The conjunction of the non-contractible component and one of the (g-1) empty ovals. (Recall that the annulus with (g-1) holes is covered by the fixed point set of the anti-holomorphic involution φ .)
- Conjunction 1') The conjunction of the non-contractible component and one of the k empty ovals. (Remark that the region (annulus) surrounded by the non-contractible component, $\mathbb{R}s$ and the k empty ovals is covered by the fixed point set of the related involution $\tilde{\varphi}$ of the anti-holomorphic involution φ .)
- Conjunction 2) The conjunction of two of the (g-1) empty ovals.
- Conjunction 2') The conjunction of two of the k empty ovals.
- Contraction 3) The contraction of one of the (g-1) empty ovals.
- Contraction 3') The contraction of one of the k empty ovals.

Definition 3.3 (Conjunctions 4), 4')). Consider the isometry class of integral involutions of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ with $(r(\psi), a(\psi), \delta_{\psi}) = (9, 9, 1)$ or (11, 9, 1). Remark that these two involutions are related.

If $(r(\psi), a(\psi), \delta_{\psi}) = (9, 9, 1)$, then we have g = 2, k = 0. Then the region $\pi(X_{\varphi}(\mathbb{R})) (\subset \mathbb{RF}_4)$ is an annulus with one hole. The boundary of the annulus consists of the non-contractible component of $\mathbb{R}A_1$, $\mathbb{R}s$, and the empty oval.

• Conjunction 4) The empty oval conjuncts with itself and becomes the union of two real lines (Node (*)) on \mathbb{RF}_4 .

If $(r(\psi), a(\psi), \delta_{\psi}) = (11, 9, 1)$, then we have g = 1, k = 1. Then the region $\pi(X_{\varphi}(\mathbb{R})) (\subset \mathbb{RF}_4)$ is the disjoint union of an annulus and one disk. The boundary of the annulus consists of the non-contractible component of $\mathbb{R}A_1$ and $\mathbb{R}s$.

• Conjunction 4') The empty oval conjuncts with itself and becomes the union of two real lines (Node (*)) on \mathbb{RF}_4 .

See Figure 14. Compare with Figure 13.



FIGURE 14. Non-increasing simplest degenerations of nonsingular real antibicanonical curves on \mathbb{RF}_4 .

 \leq

conjunction 4)

24

isomet	try class					conjunction 1)	conjunction 2)	contraction 3)]
of ty	ype (U, -	-id)						.	
(nonsi	ngular o	curve)				\rightarrow Node (1)	\rightarrow Node (2)	\rightarrow Isolated point	
$r(\psi)$	$a(\psi)$	δ_{ψ}	9	k	g - 1	α, β	α, β	α, β	No.
1	1	1	10	0	9	0, 8	0, 7	0, 8	1
2	0	0	10	1	9	1, 8	1, 7	1, 8	2
2	2	0	9	0	8	0, 7	0, 6	0, 7	3
2	2	1	9	0	8	0, 7	0, 6	0, 7	4
3	1	1	9	1	8	1, 7	1, 6	1, 7	5
3	3	1	8	0	7	0, 6	0, 5	0, 6	6
4	2	1	8	1	7	1, 6	1, 5	1, 6	7
4	4	1	7	0	6	0, 5	0, 4	0, 5	8
5	3	1	7	1	6	1, 5	1, 4	1, 5	9
5	5	1	6	0	5	0, 4	0, 3	0, 4	10
6	2	0	7	2	6	2, 5	2, 4	2, 5	11
6	4	0	6	1	5	1, 4	1, 3	1, 4	12
6	4	1	6	1	5	1, 4	1, 3	1, 4	13
6	6	1	5	0	4	0, 3	0, 2	0, 3	14
7	3	1	6	2	5	2, 4	2, 3	2, 4	15
7	5	1	5	1	4	1, 3	1, 2	1, 3	16
7	7	1	4	0	3	0, 2	0, 1	0, 2	17
8	2	1	6	3	5	3, 4	3, 3	3, 4	18
8	4	1	5	2	4	2, 3	2, 2	2, 3	19
8	6	1	4	1	3	1, 2	1, 1	1, 2	20
8	8	1	3	0	2	0, 1	0, 0	0, 1	21
9	1	1	6	4	5	4, 4	4, 3	4, 4	22
9	3	1	5	3	4	3, 3	3, 2	3, 3	23
9	5	1	4	2	3	2, 2	2, 1	2, 2	24
9	7	1	3	1	2	1, 1	1, 0	1, 1	25
9	9	1	2	0	1	0, 0	impossible	0, 0	26
10	0	0	6	5	5	5, 4	5, 3	5, 4	27
10	2	0	5	4	4	4, 3	4, 2	4, 3	28
10	2	1	5	4	4	4, 3	4, 2	4, 3	29
10	4	0	4	3	3	3, 2	3, 1	3, 2	30
10	4	1	4	3	3	3, 2	3, 1	3, 2	31
10	6	0	3	2	2	2, 1	2, 0	2, 1	32
10	6	1	3	2	2	2, 1	2, 0	2, 1	33
10	8	1	2	1	1	1, 0	impossible	1, 0	34
11	1	1	5	5	4	5, 3	5, 2	5, 3	35
11	3	1	4	4	3	4, 2	4, 1	4, 2	36
11	5	1	3	3	2	3, 1	3, 0	3, 1	37
11	7	1	2	2	1	2, 0	impossible	2, 0	38
12	2	1	4	5	3	5, 2	5, 1	5, 2	39
12	4	1	3	4	2	4, 1	4, 0	4, 1	40
12	6	1	2	3	1	3, 0	impossible	3, 0	41
13	3	1	3	5	2	5, 1	5, 0	5, 1	42
13	5	1	2	4	1	4,0	impossible	4, 0	43
14	2	0	3	6	2	6, 1	6, 0	6, 1	44
14	4	0	2	5	1	5, 0	impossible	5, 0	45
14	4	1	2	5	1	5, 0	impossible	5, 0	46
15	3	1	2	6	1	6, 0	impossible	6, 0	47
16	2	1	2	7	1	7, 0	impossible	7, 0	48
17	1	1	2	8	1	8,0	impossible	8, 0	49
18	0	0	2	9	1	9.0	impossible	9, 0	50
		÷		÷	-	.,.		~, ~	1 33

TABLE 6. Conjunction 1), Conjunction 2) and Contraction 3)

We list up candidates for possible non-increasing simplest degenerations for each of 63 isometry classes of integral involutions of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ with the invariant $(r(\psi), a(\psi), \delta_{\psi})$.

Theorem 3.4. We can enumerate up the following candidates:

- For Conjunction 1) Conjunction 2) Contraction 3), see TABLE 6.
- For Conjunction 1') Conjunction 2') Contraction 3'), see TABLE 7.
- For Conjunction 4) and 4'), see TABLE 8. \Box

isome	try class	3				conjunction 1')	conjunction 2')	contraction 3')	Π			
of t	ype (U, -	-id)										
(nons	ingular o	curve)				\rightarrow Node (1)	\rightarrow Node (2)	\rightarrow Isolated point				
$r(\psi)$	$a(\psi)$	δ_{ψ}	<i>g</i>	k	g - 1	α, eta	α, β	α, β	No.			
19	1	1	1	9	0	0, 8	0, 7	0, 8	1'			
18	0	0	2	9	1	1, 8	1, 7	1, 8	2'			
18	2	0	1	8	0	0, 7	0, 6	0, 7	3'			
18	2	1	1	8	0	0, 7	0, 6	0, 7	4'			
17	1	1	2	8	1	1, 7	1, 6	1, 7	5'			
17	3	1	1	7	0	0, 6	0, 5	0, 6	6'			
16	2	1	2	7	1	1, 6	1, 5	1, 6	7'			
16	4	1	1	6	0	0, 5	0, 4	0, 5	8'			
15	3	1	2	6	1	1, 5	1, 4	1, 5	9'			
15	5	1	1	5	0	0, 4	0, 3	0, 4	10'			
14	2	0	3	6	2	2, 5	2, 4	2, 5	11'			
14	4	0	2	5	1	1, 4	1, 3	1, 4	12'			
14	4	1	2	5	1	1, 4	1, 3	1, 4	13'			
14	6	1	1	4	0	0, 3	0, 2	0, 3	14'			
13	3	1	3	5	2	2, 4	2, 3	2, 4	15'			
13	5	1	2	4	1	1, 3	1, 2	1, 2	16'			
13	7	1	1	3	0	0, 2	0, 1	0, 2	17'			
12	2	1	4	5	3	3, 4	3, 3	3, 4	18'			
12	4	1	3	4	2	2, 3	2, 2	2, 3	19'			
12	6	1	2	3	1	1, 2	1, 1	1, 2	20'			
12	8	1	1	2	0	0, 1	0, 0	0, 1	21'			
11	1	1	5	5	4	4, 4	4, 3	4, 4	22'			
11	3	1	4	4	3	3, 3	3, 2	3, 3	23'			
11	5	1	3	3	2	2, 2	2, 1	2, 2	24'			
11	7	1	2	2	1	1, 1	1, 0	1, 1	25'			
11	9	1	1	1	0	0, 0	impossible	0, 0	26'			
10	0	0	6	5	5	5, 4	5, 3	5, 4	27'			
10	2	0	5	4	4	4, 3	4, 2	4, 3	28'			
10	2	1	5	4	4	4, 3	4, 2	4, 3	29'			
10	4	0	4	3	3	3, 2	3, 1	3, 2	30'			
10	4	1	4	3	3	3, 2	3, 1	3, 2	31'			
10	6	0	3	2	2	2, 1	2, 0	2, 1	32'			
10	6	1	3	2	2	2, 1	2, 0	2, 1	33'			
10	8	1	2	1	1	1, 0	impossible	1, 0	34'			
9	1	1	6	4	5	5, 3	5, 2	5, 3	35'			
9	3	1	5	3	4	4, 2	4, 1	4, 2	36'			
9	5	1	4	2	3	3, 1	3, 0	3, 1	37'			
9	7	1	3	1	2	2, 0	impossible	2, 0	38'			
8	2	1	6	3	5	5, 2	5, 1	5, 2	39'			
8	4	1	5	2	4	4, 1	4, 0	4, 1	40'			
8	6	1	4	1	3	3, 0	impossible	3, 0	41'			
7	3	1	6	2	5	5, 1	5, 0	5, 1	42'			
7	5	1	5	1	4	4, 0	impossible	4, 0	43'			
6	2	0	7	2	6	6, 1	6, 0	6, 1	44'			
6	4	0	6	1	5	5, 0	impossible	5, 0	45'			
6	4	1	6	1	5	5, 0	impossible	5, 0	46'			
5	3	1	7	1	6	6, 0	impossible	6, 0	47'			
4	2	1	8	1	7	7, 0	impossible	7, 0	48'			
3	1	1	9	1	8	8, 0	impossible	8, 0	49'			
2	0 0 10 1 9			9	9, 0	9,0 impossible 9,0						

TABLE 7. Conjunction 1'), Conjunction 2') and Contraction 3')

	isomet of ty (nonsi	try class pe (U, - ngular o	-id) curve)				conjunction 4)	isomet of ty (nonsi	try class ype (U, - ingular o	-id) curve)				conjunction 4')
	$r(\psi)$	$a(\psi)$	δ_{ψ}	g	k	g - 1		$r(\psi)$	$a(\psi)$	δ_{ψ}	<i>g</i>	k	g - 1	
[9	9	1	2	0	1	Node (*)	11	9	1	1	1	0	Node (*)

TABLE 8. Conjunctions 4) and 4')

Remark 3.5. We observe an obvious interesting correspondence between the data of Theorem 2.24 (TABLES 4 - 5) and those of Theorem 3.4 (TABLES 6 - 8).

If we prove the existence of all the non-increasing simplest degenerations listed in TABLES 6, 7, and 8, then we can say that every real isotopy type with an anti-holomorphic involution in Theorem 2.24 except Node (1) with $(\alpha, \beta) = (1, 0)$ and Isolated point with $(\alpha, \beta) = (1, 0)$ in the isometry class $(9, 9, 0, H(\psi) \cong \mathbb{Z}/2\mathbb{Z})$ can be obtain by certain non-increasing simplest degeneration of a real nonsingular curve in |12c + 3s| on \mathbb{F}_4 . Recall Remark 2.25.

To state the following lemma 3.7, we distinguish the two anti-holomorphic involutions φ_{\pm} on (X, τ) of type $(S, \theta) \cong ((3, 1, 1), -id)$ as follows.

Definition 3.6 (Definition of φ_{\pm}). We define the anti-holomorphic involutions φ_{\pm} on a 2elementary K3 surface (X, τ) of type $(S, \theta) \cong ((3, 1, 1), -id)$ such that

$$\pi(X_{\varphi_+}(\mathbb{R})) = A_{\pm}$$

respectively. Here recall Definition 2.14 of the two regions A_{\pm} .

Thus, if $H(\psi) = 0$ for a real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong ((3, 1, 1), -id)$, then we set $A_{-} = \pi(X_{\varphi}(\mathbb{R}))$, and

$$\varphi_{-} := \varphi \text{ and } \varphi_{+} := \widetilde{\varphi} = \tau \circ \varphi.$$

Comparing the data in Theorem 2.24 and Theorem 3.4, we have:

Lemma 3.7. Fix an isometry class of integral involutions of \mathbb{L}_{K3} of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ with $(r(\psi), a(\psi), \delta_{\psi}) \neq (10, 8, 0), (10, 10, 0),$ and take a corresponding real 2-elementary K3 surface of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ and a real nonsingular curve $A = s + A_1$ in $|-2K_{\mathbb{F}_4}|$ where A_1 is a real nonsingular curve in |12c + 3s| on \mathbb{RF}_4 .

Then we have the following:

• Take real curves A'_1 with one nondegenerate double point on \mathbb{F}_4 from the degenerations of types 1)—3) of the real nonsingular curve A_1 . Choose real 2-elementary K3 surfaces (X, τ, φ_-) of type $(S, \theta) \cong ((3, 1, 1), -id)$. (See Definition 3.6 and also Remark 4.7 below.)

Then, for all such marked real 2-elementary K3 surfaces $((X, \tau, \varphi_{-}), \alpha)$ obtained from A'_1 , their associated integral involutions $\alpha \circ (\varphi_{-})_* \circ \alpha^{-1}$ of \mathbb{L}_{K3} are **isometric** with respect to $G = \{id\}$.

Take real curves A'₁ with one nondegenerate double point on F₄ from the degenerations of types 1')—3') of the real nonsingular curve A₁. Choose real 2-elementary K3 surfaces (X, τ, φ₋) of type (S, θ) ≅ ((3, 1, 1), -id). (See Definition 3.6 and also Remark 4.7 below.)

Then, for all such marked real 2-elementary K3 surfaces $((X, \tau, \varphi_{-}), \alpha)$ obtained from A'_1 , their associated integral involutions $\alpha \circ (\varphi_{-})_* \circ \alpha^{-1}$ of \mathbb{L}_{K3} are **isometric** with respect to $G = \{id\}$. \Box

Lemma 3.7 is an analogy of Proposition 3.3 of [4].

4. Appendix: Period domains and further problems

We use the terminology defined in Subsection 2.1, and review the formulations of period domains of marked real 2-elementary K3 surfaces in [10], [4].

Lemma 4.1. Let $((X, \tau, \varphi), \alpha)$ and $((X', \tau', \varphi'), \alpha')$ be two marked real 2-elementary K3 surfaces of type (S, θ) . Let (\mathbb{L}_{K3}, ψ) and (\mathbb{L}_{K3}, ψ') be their associated integral involutions respectively. Suppose that f is an **isometry** with respect to the group G from (\mathbb{L}_{K3}, ψ) to (\mathbb{L}_{K3}, ψ') . Then $((X', \tau', \varphi'), f^{-1} \circ \alpha')$ is also a marked real 2-elementary K3 surface of type (S, θ) , and its associated integral involution is also (\mathbb{L}_{K3}, ψ) .

Especially, if f is an **automorphism** of (\mathbb{L}_{K3}, ψ) with respect to the group G, then $((X, \tau, \varphi), f \circ \alpha)$ is also a marked real 2-elementary K3 surface of type (S, θ) , and its associated integral involution is also (\mathbb{L}_{K3}, ψ) . (Recall Definition 2.9.)

Proof. We have $\psi' \circ f = f \circ \psi$, f(S) = S, and $f|_S \in G$. We have $\alpha' \circ \varphi'_* \circ {\alpha'}^{-1} \circ f = f \circ \alpha \circ \varphi_* \circ {\alpha}^{-1}$, and hence, $f^{-1} \circ \alpha' \circ {\varphi'}_* \circ {\alpha'}^{-1} \circ f = \alpha \circ \varphi_* \circ {\alpha}^{-1}$. We have

$$(f^{-1} \circ \alpha') \circ \varphi'_* \circ (f^{-1} \circ \alpha')^{-1} = \alpha \circ \varphi_* \circ \alpha^{-1} = \psi.$$

Here $(f^{-1} \circ \alpha') : H_2(X', \mathbb{Z}) \to \mathbb{L}_{K3}$ is an isometry with

$$(f^{-1} \circ \alpha')(H_{2+}(X', \mathbb{Z})) = f^{-1}(S) = S$$

and $(f^{-1} \circ \alpha') \circ \varphi'_* = f^{-1} \circ (\alpha' \circ \varphi'_*) = f^{-1} \circ (\psi' \circ \alpha') = \psi \circ f^{-1} \circ \alpha'$. Since $f^{-1} \circ \alpha'(H_{2+}(X', \mathbb{Z})) = S$ and $\psi|_S = \theta$, we have

$$(f^{-1} \circ \alpha') \circ \varphi'_* = \theta \circ f^{-1} \circ \alpha' \text{on } H_{2+}(X', \mathbb{Z}).$$

Hence,

$$(f^{-1} \circ \alpha') : H_2(X', \mathbb{Z}) \to \mathbb{L}_{K3}$$

is another marking of (X', τ', φ') . If we take this new marking of (X', τ', φ') , then its associated integral involution is ψ , which is the same as $((X, \tau, \varphi), \alpha)$. Moreover, since $f|_S \in G$, we have $f(\mathcal{M}) = \mathcal{M}$. Hence, if we set $\beta := (f^{-1} \circ \alpha')$, then $\beta_{\mathbb{R}}^{-1}(V^+(S))$ contains a hyperplane section of X' and the set $\beta^{-1}(\Delta(S)_+)$ contains only classes of effective curves of X'. \Box

Let us fix an integral involution (\mathbb{L}_{K3}, ψ) of type (S, θ) through this subsection.

We set

$$\Omega_{\psi} := \{ v \ (\neq 0) \in \mathbb{L}_{K3} \otimes \mathbb{C} \mid v \cdot v = 0, \ v \cdot \overline{v} > 0, \ v \cdot S = 0, \ \psi_{\mathbb{C}}(v) = \overline{v} \} / \mathbb{R}^{\times}.$$

Let $((X, \tau, \varphi), \alpha)$ be a marked real 2-elementary K3 surface of type (S, θ) satisfying

$$\alpha \circ \varphi_* \circ \alpha^{-1} = \psi,$$

namely, ψ is the associated integral involution with $((X, \tau, \varphi,) \alpha)$. We denote by

$$H (\subset H_2(X,\mathbb{C}))$$

the Poincare dual of the complex 1-dimensional space $H^{2,0}(X)$. Then we have the complex 1-dimensional subspace $\alpha_{\mathbb{C}}(H)$ in $\mathbb{L}_{K3} \otimes \mathbb{C}$. Then we have

$$\alpha_{\mathbb{C}}(H) \in \Omega_{\psi}.$$

Definition 4.2 (Periods). We say $\alpha_{\mathbb{C}}(H)$ the **period** of a marked real 2-elementary K3 surface $((X, \tau, \varphi), \alpha)$ of type (S, θ) satisfying $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$.

By Lemma 4.1, all marked real 2-elementary K3 surfaces whose associated integral involutions are isometric to (\mathbb{L}_{K3}, ψ) with respect to G can be found in Ω_{ψ} if we change their markings appropriately.

A point in Ω_{ψ} is not necessarily the period of some marked real 2-elementary K3 surface of type (S, θ) satisfying $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$.

Definition 4.3 (Equivalence). We say a point $[v] (\in \Omega_{\psi})$ is **equivalent** to a point $[v'] (\in \Omega_{\psi})$ if $[v'] = f_{\mathbb{C}}([v])$ for an automorphism f of (\mathbb{L}_{K3}, ψ) of type (S, θ) with respect to the group G.

Lemma 4.4. If a point $[v] (\in \Omega_{\psi})$ is equivalent to $[v'] (\in \Omega_{\psi})$ and [v] is the period of some marked real 2-elementary K3 surface $((X, \tau, \varphi), \alpha)$ of type (S, θ) satisfying $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$, then [v'] is also the period of a marked real 2-elementary K3 surface $((X, \tau, \varphi), \alpha')$ of type (S, θ) satisfying $(\alpha') \circ \varphi_* \circ (\alpha')^{-1} = \psi$ where α' is some another marking of (X, τ, φ) .

Proof. Since [v] is equivalent to [v'], we have $[v'] = f_{\mathbb{C}}([v])$ for an automorphism f of (\mathbb{L}_{K3}, ψ) of type (S, θ) with respect to the group G. By Lemma 4.1, $((X, \tau, \varphi), f \circ \alpha)$ is also a marked real 2-elementary K3 surface of type (S, θ) satisfying $(f \circ \alpha) \circ \varphi_* \circ (f \circ \alpha)^{-1} = \psi$. Moreover, the period of $((X, \tau, \varphi), f \circ \alpha)$ is

$$(f \circ \alpha)_{\mathbb{C}}(H) = f_{\mathbb{C}}(\alpha_{\mathbb{C}}(H)) = f_{\mathbb{C}}([v]) = [v']$$

where $H \ (\subset H_2(X,\mathbb{C}))$ is the Poincare dual of $H^{2,0}(X)$. It is sufficient that we set $\alpha' = f \circ \alpha$. \Box

Remark 4.5 ([10]). By the global Torelli theorem, if two periods are equivalent, then corresponding marked real 2-elementary K3 surfaces are analytic isomorphic (see Definition 2.10). The converse is also true.

The domain Ω_{ψ} has two connected components which are interchanged by $-\psi$. We see $-\psi$ is an **automorphism** of (\mathbb{L}_{K3}, ψ) with respect to the group G. Hence, by Lemma 4.4 and Remark 4.5, it is sufficient that we investigate the quotient space $\Omega_{\psi}/-\psi$. We set

$$\mathbb{L}_{\pm} := \{ x \in \mathbb{L}_{K3} \mid \psi(x) = \pm x \}$$

Note that the lattices \mathbb{L}_{\pm} depend on the integral involution ψ .

For $[v] \in \Omega_{\psi}$ $(v \in \mathbb{L}_{K3} \otimes \mathbb{C})$, we have the decomposition $v = v_{+} + v_{-}$, where $v_{\pm} \in \mathbb{L}_{\pm} \otimes \mathbb{R}$.

We restrict ourselves the case when $S \subset \mathbb{L}_{-}$, namely, $\theta = -id$, and set

$$\mathbb{L}_{-,S} := \mathbb{L}_{-} \cap S^{\perp}.$$

 $\mathbb{L}_{-,S}$ also depends on the integral involution ψ .

Since $v_{-} \in \mathbb{L}_{-,S} \otimes \mathbb{R}$ and $v_{+}^{2} = v_{-}^{2} > 0$, we see that \mathbb{L}_{+} , $\mathbb{L}_{-,S}$ is a hyperbolic lattice.

Let \mathcal{L}_+ , $\mathcal{L}_{-,S}$ be the hyperbolic spaces obtained from $\mathbb{L}_+ \otimes \mathbb{R}$, $\mathbb{L}_{-,S} \otimes \mathbb{R}$ respectively. Then we have

$$\Omega_{\psi}/-\psi = \mathcal{L}_{+} \times \mathcal{L}_{-,S}$$
 (a direct product).

We now fix a primitive hyperbolic 2-elementary sublattice S with $S \cong (3,1,1)$ of the K3 lattice \mathbb{L}_{K3} and set $\theta := -id$. Remark that $G = \{id\}$ for this case (Remark 2.12).

We use the terminology and facts stated in Subsection 2.2. For a real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong ((3, 1, 1), -id)$, the fixed point curve A is a disjoint union of A_0 and A_1 . X has an elliptic fibration with the exceptional section A_0 . Its unique reducible fiber is E + F. There are two kinds of the curve F, which is irreducible or not. The classes $[A_0]$, [E] and [F] generate $H_{2+}(X,\mathbb{Z})$. We have an orthogonal decomposition

$$\mathbb{Z}([A_0], [E] + [F]) \oplus \mathbb{Z}([F])$$

of $H_{2+}(X,\mathbb{Z})$. The subgroups $\mathbb{Z}([A_0], [E] + [F]), \mathbb{Z}([F])$ are isometric to the hyperbolic plane and the lattice $\langle -2 \rangle$ respectively.

We fix an integral involution (\mathbb{L}_{K3}, ψ) of type $(S, \theta) \cong ((3, 1, 1), -\mathrm{id})$. Moreover, we fix a decomposition $S = \mathbb{U} \oplus \langle -2 \rangle$, where \mathbb{U} is a sublattice of S isometric to the hyperbolic plane. Let \mathcal{F} be the generator of $\langle -2 \rangle$ with $\mathcal{M} \cdot \mathcal{F} \ge 0$.

We consider marked real 2-elementary K3 surfaces $((X, \tau, \varphi), \alpha)$ of type $(S, \theta) \cong ((3, 1, 1), -id)$ (see Definition 2.4) such that

$$\alpha(\mathbb{Z}([A_0], [E] + [F])) = \mathbb{U}, \quad \alpha([F]) = \mathcal{F},$$

and

$$\alpha \circ \varphi_* \circ \alpha^{-1} = \psi.$$

Any real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong ((3, 1, 1), -id)$ has such a marking α .

We want to know a criterion for the double point of a real anti-bicanonical curve $\mathbb{R}bl(A)$ with one real double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$ to be **nondegenerate**. At present we can prove the following lemma.

Lemma 4.6. For $[\omega] \in \Omega_{\psi}/-\psi$, $[\omega]$ is the period of a marked real 2-elementary K3 surface of type $(S, \theta) \cong ((3, 1, 1), -\text{id})$ (see Definition 2.4) such that $\alpha(\mathbb{Z}([A_0], [E]+[F])) = \mathbb{U}$, $\alpha([F]) = \mathcal{F}$, and $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$ obtained from a real anti-bicanonical curve $\mathbb{R}bl(A)$ with one real **nondegenerate** double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$ if there are no $\mathbf{v} \ (\neq \pm \mathcal{F})$ in \mathbb{L}_{K3} satisfying that $\mathbf{v} \cdot \omega = 0$, $\mathbf{v} \cdot \mathbb{U} = 0$, and $\mathbf{v}^2 = -2$.

Proof. For a real anti-bicanonical curve with one real degenerate double point on \mathbb{RF}_4 , the corresponding marked real 2-elementary K3 surface $(X, \tau, \varphi, \alpha)$ has a unique reducible fiber E + F where $F^2 = -2$, F is the union of two nonsingular rational curves F' and F''. Here F' and F'' are conjugate by τ and $F' \cdot F'' = 1$. Hence, $(F')^2 = (F'')^2 = -2$. (see Subsection 2.2) Both [F'] and [F''] are orthogonal to $[A_0]$ and [E] + [F]. We set $\mathbf{v} := \alpha[F']$ or $\alpha[F'']$. Since $\mathbb{U} = \alpha(\mathbb{Z}([A_0], [E] + [F]))$, we have $\mathbf{v} \ (\neq \pm \mathcal{F})$, $\mathbf{v} \cdot \omega = 0$, $\mathbf{v} \cdot \mathbb{U} = 0$, and $\mathbf{v}^2 = -2$ for the period $[\omega]$ of $(X, \tau, \varphi, \alpha)$.

Problem 1 (cf. [4], the top of p.281). Is the converse of Lemma 4.6 also true?

If the converse of Lemma 4.6 is true, then we can get the precise image $(\subset \mathcal{L}_+ \times \mathcal{L}_{-,S})$ of the period map on the set of all marked real 2-elementary K3 surfaces of type $(S, \theta) \cong ((3, 1, 1), -id)$ such that $\alpha(\mathbb{Z}([A_0], [E] + [F])) = \mathbb{U}, \alpha([F]) = \mathcal{F}$, and $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$ obtained from real anti-bicanonical curves $\mathbb{R}bl(A)$ with one real **nondegenerate** double point on $\mathbb{R}A'_1$ on $\mathbb{R}\mathbb{F}_4$.

Let \mathbf{v} be an element of \mathbb{L}_+ with square -2. The reflection on \mathcal{L}_+ with respect to the real hyperplane \mathbf{v}^{\perp} is well-defined and it sends a point in $\Omega_{\psi}/-\psi$ to its equivalent point. Also, the reflection on $\mathcal{L}_{-,S}$ with respect to the real hyperplane \mathbf{v}^{\perp} where \mathbf{v} is an element of $\mathbb{L}_{-,S}$ with square -2 is well-defined and it sends a point in $\Omega_{\psi}/-\psi$ to its equivalent point. Let

$$\Omega_+$$
 (respectively, $\Omega_{-,S}$)

be the open fundamental domains with respect to the groups generated by the reflections with respect to the real hyperplanes \mathbf{v}^{\perp} satisfying $\mathbf{v}^2 = -2$ and $\mathbf{v} \in \mathbb{L}_+$ (respectively, $\mathbb{L}_{-,S}$) and we consider the direct product $\Omega_+ \times \Omega_{-,S}$.

If the converse of Lemma 4.6 is true, then the periods of marked real 2-elementary K3 surfaces $(X, \tau, \varphi, \alpha)$ of type ((3, 1, 1), -id) obtained from some real anti-bicanonical curves with one real **nondegenerate** double point on \mathbb{RF}_4 and satisfying $\alpha \circ \varphi_* \circ \alpha^{-1} = \psi$ are contained in $\Omega_+ \times \Omega_{-,S}$ up to equivalence. Moreover, we should remove more $\omega = \omega_+ + \omega_-$ orthogonal to some $\mathbf{v} \in \mathbb{L}_{K3}$ such that $\mathbf{v} \ (\neq \pm \mathcal{F}), \mathbf{v} \cdot \mathbb{U} = 0, \mathbf{v}^2 = -2, \mathbf{v} \notin \mathbb{L}_+$ and $\mathbf{v} \notin \mathbb{L}_{-,S}$. Hence, as the argument before Theorem 2.1 of Itenberg's paper [4], we should remove some (-6)-orthogonal real hyperplanes (extra "walls") from $\Omega_{-,S}$ or Ω_+ . This seems to be the reason why the connected components of the moduli space in the sense of [10] (equivalently, the isometry classes of integral involutions

30
of the K3 lattice) cannot distinguish the topological types (node or isolated point) of the real double points (Remark 2.25).

Problem 2 (cf. [4], Theorem 2.1). Formulate some period domain, say it were $\Omega_*^{\mathbb{R}\mathbb{F}_4}$, whose connected components (up to equivalence) are in **bijective correspondence** with the connected components of the moduli space (up to the action of the automorphism group of \mathbb{F}_4 over \mathbb{R}) of real anti-bicanonical curves with one real nondegenerate double point on $\mathbb{R}\mathbb{F}_4$ which yield marked real 2-elementary K3 surfaces $((X, \tau, \varphi_-), \alpha)$ satisfying $\alpha \circ (\varphi_-)_* \circ \alpha^{-1} = \psi$.

Remark 4.7. For (\mathbb{L}_{K3}, ψ) , either \mathbb{L}_+ or $\mathbb{L}_{-,S}$ does not contain any element v such that

$$\mathbf{v} \equiv \mathcal{F} \pmod{2\mathbb{L}_{K3}}.$$

For the anti-holomorphic involution " φ_{-} " (recall Definition 3.6), $\mathbb{L}_{-,S}$ contains an element **v** such that $\mathbf{v} \equiv \mathcal{F} \pmod{2\mathbb{L}_{K3}}$. This phenomenon is similar to the argument in [4].

We are also interested in the correspondences of the Coxeter graphs (see [4], [5]) obtained from isometry classes of integral involutions of the K3 lattice \mathbb{L}_{K3} of type $(S, \theta) \cong ((3, 1, 1), -id)$ and the non-increasing simplest degenerations of nonsingular curves. Problems concerning this topic are as follows.

We fix an integral involution (\mathbb{L}_{K3}, ψ) of type $(S, \theta) \cong ((3, 1, 1), -\mathrm{id})$. Recall

$$\mathbb{L}_{\pm} := \{ x \in \mathbb{L}_{K3} \mid \psi(x) = \pm x \},\$$

U, and $\mathbb{L}_{-,\mathbb{U}} := \mathbb{L}_{-} \cap \mathbb{U}^{\perp}$. $\mathbb{L}_{-,\mathbb{U}}$ is also hyperbolic. Let $\mathcal{L}_{-,\mathbb{U}}$ be the hyperbolic space obtained from $\mathbb{L}_{-,\mathbb{U}} \otimes \mathbb{R}$. The group generated by the reflections with respect to real hyperplanes \mathbf{v}^{\perp} such that $\mathbf{v} \in \mathbb{L}_{-,\mathbb{U}}$ with $\mathbf{v}^2 = -2$ acts on the hyperbolic space $\mathcal{L}_{-,\mathbb{U}}$. Let $\widetilde{\Omega}_{-}$ be one of its fundamental domains which has a face orthogonal to F. Let C be the Coxeter graph of $\widetilde{\Omega}_{-}$.

Definition 4.8. • Let C' be the graph which is obtained by removing all thick or dotted edges from C.

- Consider the group of symmetries of C' obtained from some automorphism of $(\mathbb{L}_{K3}, \psi, \mathbb{U})$. Let C'' be the quotient graph of C' by the action of the group.
- Let e be the vertex of C corresponding to F and let e' be the class (in C") containing e.
 Let K be the connected component of C" containing e'.

Problem 3 (cf. [4], Proposition 3.1). Does the number (up to equivalence) of connected components of $\Omega_{\mathbb{R}}^{\mathbb{R}_4}$ coincide with that of vertices of the graph K ?

Degenerations and the graph *P*. We fix an isometry class of integral involutions of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ with $(r(\psi), a(\psi), \delta_{\psi}) \neq (10, 8, 0), (10, 10, 0)$. Then we get a real 2-elementary K3 surface (X, τ, φ) of type $(S, \theta) \cong (\mathbb{U}, -\mathrm{id})$ and a real **nonsingular** curve $A = s + A_1$ in $|-2K_{\mathbb{F}_4}|$ where A_1 is a real nonsingular curve in |12c + 3s| on \mathbb{RF}_4 .

We define the graphs P as follows (See Definition 3.2).

- **Definition 4.9.** The vertices of P are all the rigid isotopy classes of real curves A'_1 in |12c+3s| with one nondegenerate double point on \mathbb{F}_4 obtained from the degenerations of types 1)—3) of the real nonsingular curve A_1 with φ .
 - Two vertices of P are connected by an **edge** if one rigid isotopy class is obtained from the conjunction of the ovals \mathcal{E}_1 and the non-contractible component of A_1 (Conjunction 1)), and the other rigid isotopy class is obtained from the contraction of the oval \mathcal{E}_1 (Contraction 3)).

SACHIKO SAITO

• Two vertices of P are connected by an **edge** if one rigid isotopy class is obtained from the conjunction of the ovals \mathcal{E}_1 and \mathcal{E}_2 of A_1 (Conjunction 2)), and the other rigid isotopy class is obtained from the contraction of the oval \mathcal{E}_1 (Contraction 3)).

Problem 4 (cf. [4], Proposition 3.4). Fix a real **nonsingular** curve $A = s + A_1$ in $|-2K_{\mathbb{F}_4}|$ where A_1 is a real nonsingular curve in |12c + 3s| on \mathbb{RF}_4 and φ as above.

Let us get an **arbitrary** real curve A'_1 in |12c + 3s| with one nondegenerate double point on \mathbb{F}_4 obtained from one of the degenerations of types 1)—3) of A_1 with φ , and construct the graph K (see Lemma 3.7 and Definition 4.8) from A'_1 and φ_- . Then, is the graph K isomorphic to the graph P?

References

- [1] V.A. Alexeev, V.V. Nikulin, Del Pezzo and K3 Surfaces, MSJ Memoirs, 2006. DOI: 10.1142/e002
- [2] A. Degtyarev, I. Itenberg, V. Kharlamov, *Real Enriques Surfaces*, Lecture Notes in Math., 1746, Springer, 2000.
- [3] A. Degtyarev, I. Itenberg, V. Zvonilov, Real trigonal curves and real elliptic surfaces of type I, 2011. arXiv:1102.3414v1
- [4] I. Itenberg, Curves of degree 6 with one non-degenerate double point and groups of monodromy of nonsingular curves, Real Algebraic Geometry, Proceedings, Rennes 1991, Lecture Notes in Math., 1524, Springer, (1992), 267–288.
- [5] I. Itenberg, Rigid isotopy classification of curves of degree 6 with one nondegenerate double point, Advances in Soviet Mathematics 18, (1994), 193–208.
- [6] I. Itenberg, Groups of monodromy of non-singular curves of degree 6, Real Analytic and Algebraic Geometry, Proceedings, Trento (Italy) 1992, Walter de Gruyter, (1995), 161–168.
- [7] V.V. Nikulin, Integral symmetric bilinear forms and some of their geometric applications, Math. USSR Izv., 14 (1980), 103–167. DOI: 10.1070/IM1980v014n01ABEH001060
- [8] V.V. Nikulin, On the quotient groups of the automorphism groups of hyperbolic forms by the subgroups generated by 2-reflections, Algebraic-geometric applications, J. Soviet Math., 22 (1983), 1401–1476. DOI: 10.1007/BF01094757
- [9] V.V. Nikulin, Involutions of integral quadratic forms and their applications to real algebraic geometry, Math. USSR Izv., 22 (1984), 99–172. DOI: 10.1070/IM1984v022n01ABEH001435
- [10] V.V. Nikulin, Sachiko Saito, Real K3 surfaces with non-symplectic involution and applications, Proc. London Math. Soc., 90 (2005), 591–654. DOI: 10.1112/S0024611505015212
- [11] V.V. Nikulin, Sachiko Saito, Real K3 surfaces with non-symplectic involution and applications. II, Proc. London Math. Soc., 95 (2007), 20–48. DOI: 10.1112/plms/pdl023
- [12] S. Yu. Orevkov, Riemann existence theorem and construction of real algebraic curves, Ann. Fac. Sci. Toulouse 12-4 (2003), 517-531. DOI: 10.5802/afst.1060

DEPARTMENT OF MATHEMATICS EDUCATION, ASAHIKAWA CAMPUS,, HOKKAIDO UNIVERSITY OF EDUCATION, ASAHIKAWA, JAPAN

 $E\text{-}mail\ address:\ \texttt{saito.sachiko@a.hokkyodai.ac.jp}$

RESONANT BANDS, AOMOTO COMPLEX, AND REAL 4-NETS

MICHELE TORIELLI AND MASAHIKO YOSHINAGA

ABSTRACT. The resonant band is a useful notion for the computation of the nontrivial monodromy eigenspaces of the Milnor fiber of a real line arrangement. In this article, we develop the resonant band description for the cohomology of the Aomoto complex. As an application, we prove that real 4-nets do not exist.

1. INTRODUCTION

Combinatorial decisions of topological invariants are the central problems in the theory of hyperplane arrangements. Milnor fibers and their eigenspace decompositions have received a lot of attention and have been studied by diverse techniques ([23]) (e.g., Alexander polynomials, Hodge theory, nets and multinets, covering spaces, Salvetti complexes, characteristic and resonance varieties etc.) Among others, the authors follow the previous studies using real structures, ([24, 29, 30]) and Aomoto complexes over finite fields, [4, 7, 17, 20].

Concerning the relation between Milnor fibers and Aomoto complexes, two key results were obtained by Papadima and Suciu [20, 21].

Monodromy eigenspaces $\stackrel{(1)}{\longleftrightarrow}$ Aomoto complex $\stackrel{(2)}{\longleftrightarrow}$ Multinets

The first one is an upper bound for the rank of the eigenspaces in terms of the Betti numbers of the Aomoto complexes over finite fields [20]. It was subsequently used by many authors to prove vanishing theorems [1, 2, 8, 17]. The second one is the bijective correspondence between 3-nets and nonzero elements in the cohomology group of the Aomoto complex over \mathbb{F}_3 . A degree one element of the Orlik-Solomon algebra over the finite field \mathbb{F}_q is bijectively corresponding to the coloring (with q-colors) of the arrangement. Papadima and Suciu succeeded to translate the cocycle condition into combinatorics of coloring [21]. A deep relation between nontrivial eigenspaces and multinet structures had been conjectured. Papadima-Suciu's results provide a beautiful framework to understand the nontrivial eigenspaces via multinets.

If we restrict our attention to real arrangements, the real structure contains a lot of information about the topology of the complexification. The resonant band, introduced in [29, 30], is a useful tool for computing nontrivial eigenspaces of the Milnor fibers and local system cohomology groups. The purpose of this paper is to introduce the notion of resonant bands for the Aomoto complex (over any coefficient ring) of a real arrangement. Then combining resonant bands techniques with the above Papadima-Suciu's picture (over \mathbb{F}_2), we prove that real 4-nets do not exist, which is a partial answer to a conjecture that the Hessian arrangement is the only 4-net.

The paper is organized as follows. §2 is a summary of well known facts on multinets and Orlik-Solomon algebras. Especially, we describe in detail the transformation of the Orlik-Solomon algebra when we exchange the hyperplane at infinity, which will be used later. §3 is a summary of the recent work by Papadima-Suciu. The crucial result that we use later is Proposition 3.4. Proposition 3.4 translates the cocycle conditions of the Aomoto complex (over \mathbb{F}_2) into combinatorial structures of subarrangements. §4 is the main part of this paper. After recalling a description of the Aomoto complex in terms of chambers in §4.1 (following [27]), we introduce

the notion of η -resonant band in §4.2. In the main theorem (Theorem 4.8), we prove that the cohomology of the Aomoto complex is isomorphic to a submodule of the free module generated by resonant bands under a certain non-resonant condition at infinity. When the coefficient ring of the Aomoto complex is \mathbb{F}_2 , everything can be described in terms of combinatorics of subarrangements. This translation is done in §4.3. In §5, we prove the non-existence of real 4-nets. The key result is the Non-Separation Theorem 5.1 in §5.1 which concerns subarrangements corresponding to cocycles of the Aomoto complex over \mathbb{F}_2 with differential given by the diagonal element. The Non-Separation Theorem asserts that at an intersection of multiplicity 4, the subarrangement corresponding to a nontrivial cohomology class has a special ordering. This assertion heavily relies on the real structure. Therefore, at this moment, it seems hopeless to generalize our argument to the complex case. Assuming a real 4-net exists, it is easy to construct a subarrangement which contradicts the Non-Separation Theorem. Hence real 4-nets do not exist (§5.2). (This fact was first proved by Cordovil-Forge [6, Lem. 2.4]. Our arguments prove a little bit stronger version. See Remark 5.3.)

2. Preliminaries

2.1. Conventions. In this paper, three types of hyperplane arrangements appear: affine arrangements in \mathbb{K}^{ℓ} , arrangements in the projective space \mathbb{KP}^{ℓ} and central arrangements in $\mathbb{K}^{\ell+1}$. It is better to distinguish them by notations ([18, 19]).

- $\mathcal{A} = \{H_1, \ldots, H_n\}$ denotes an arrangement of affine hyperplanes in the affine ℓ -space \mathbb{K}^{ℓ} .
- $\tilde{\tilde{\mathcal{A}}} = c\mathcal{A} = \{\tilde{H}_0, \tilde{H}_1, \dots, \tilde{H}_n\}$ denotes the coning of \mathcal{A} , which is a central hyperplane arrangement in $\mathbb{K}^{\ell+1}$. The hyperplane \tilde{H}_0 is corresponding to the hyperplane at infinity of \mathcal{A} .
- $\overline{\mathcal{A}} = \{\overline{H}_0, \overline{H}_1, \dots, \overline{H}_n\}$ denotes the projectivization of $\widetilde{\mathcal{A}}$, which is a hyperplane arrangement induced by $\widetilde{\mathcal{A}}$ in the projective ℓ -space \mathbb{KP}^{ℓ} .
- $\mathsf{d}_{\widetilde{H}_{i}}\widetilde{\mathcal{A}} = \{\mathsf{d}_{\widetilde{H}_{i}}\widetilde{H}_{0},\ldots,\mathsf{d}_{\widetilde{H}_{i}}\widetilde{H}_{i},\ldots,\mathsf{d}_{\widetilde{H}_{i}}\widetilde{H}_{n}\}$ denotes the decoming of $\widetilde{\mathcal{A}}$ with respect to the hyperplane \widetilde{H}_{i} . Note that $\mathsf{d}_{\widetilde{H}_{o}}\widetilde{\mathcal{A}} = \mathcal{A}$.

Other frequently used notations are:

- R: a commutative ring (unless stated otherwise),
- \mathbb{K} : a field,
- $M(\mathcal{A})$: the complexified complement of \mathcal{A} .

2.2. Multinets. In this subsection, we recall several facts on multinets.

Definition 2.1. Let $\overline{\mathcal{A}} = \{\overline{H}_0, \dots, \overline{H}_n\}$ be a projective line arrangement in \mathbb{CP}^2 . Let $k \geq 3$ and $d \geq 2$ be integers. A (reduced) (k, d)-multinet (or k-multinet for simplicity) on \mathcal{A} is a pair $(\mathcal{N}, \mathcal{X})$, where \mathcal{N} is a partition of \mathcal{A} into k classes $\overline{\mathcal{A}} = \overline{\mathcal{A}}_1 \sqcup \cdots \sqcup \overline{\mathcal{A}}_k$ and $\mathcal{X} \subset \mathbb{CP}^2$ is a set of multiple points (called the base locus) such that

- (i) $|\overline{\mathcal{A}}_1| = \cdots = |\overline{\mathcal{A}}_k| =: d;$
- (*ii*) $\overline{H} \in \overline{\mathcal{A}}_i$ and $\overline{H}' \in \overline{\mathcal{A}}_j$ $(i \neq j)$ imply that $\overline{H} \cap \overline{H}' \in \mathcal{X}_j$:
- (iii) for all $p \in \mathcal{X}$, $n_p := |\{\overline{H} \in \overline{\mathcal{A}}_i \mid \overline{H} \ni p\}|$ is constant and independent of i;
- (iv) for any $\overline{H}, \overline{H}' \in \overline{\mathcal{A}}_i$ (i = 1, ..., k), there is a sequence $\overline{H} = \overline{H}'_0, \overline{H}'_1, ..., \overline{H}'_r = \overline{H}'$ in $\overline{\mathcal{A}}_i$ such that $\overline{H}'_{i-1} \cap \overline{H}'_i \notin \mathcal{X}$ for $1 \leq j \leq r$.

If $n_p = 1$ for every $p \in \mathcal{X}$, then $(\mathcal{N}, \mathcal{X})$ is called a net.

4-NETS

Note that in the previous Definition (*ii*) and (*iii*) implies (*i*), and, by [13], \mathcal{N} and \mathcal{X} determine each other. Moreover, note that if $(\mathcal{N}, \mathcal{X})$ is a (k, d)-net, then each $p \in \mathcal{X}$ has multiplicity k.

The next theorem, which combines results of Pereira and Yuzvinsky [22, 33], summarizes what is known about the existence of non-trivial multinets on arrangements (see also [23, 3, 32] for more results).

Theorem 2.2. Let $\overline{\mathcal{A}}$ be a k-multinet, with base locus \mathcal{X} . Then

- (1) If $|\mathcal{X}| > 1$, then k = 3 or 4.
- (2) If there is a hyperplane $\overline{H} \in \overline{\mathcal{A}}$ such that $m_H > 1$, then k = 3.
- (3) If k = 4, then $|\mathcal{X}| = d^2$ and $\overline{\mathcal{A}}$ is a (4, d)-net.

Although several infinite families of multinets with k = 3 are known, only one multinet with k = 4 is known to exist: the (4,3)-net on the Hessian arrangement (which is defined over $\mathbb{Q}(\sqrt{-3})$). It is conjectured that the only (4, d)-net is the Hessian arrangement. In [10], it is proved that the Hessian is the unique (4, d)-net for $d \leq 6$ (hence for $|\mathcal{A}| \leq 24$). We will later prove that real (4, d)-nets do not exist, for any d.

2.3. Orlik-Solomon algebra and Aomoto complex. Let $\mathcal{A} = \{H_1, \ldots, H_n\}$ be an arrangement of affine hyperplanes in \mathbb{K}^{ℓ} and R be a commutative ring. Let $E_1 = \bigoplus_{j=1}^n Re_j$ be the free module generated by e_1, e_2, \ldots, e_n , where e_i is a symbol corresponding to the hyperplane H_i . Let $E = \wedge E_1$ be the exterior algebra over R. The algebra E is graded via $E = \bigoplus_{p=0}^n E_p$, where $E_p = \wedge^p E_1$. The R-module E_p is free and has the distinguished basis consisting of monomials $e_S := e_{i_1} \wedge \cdots \wedge e_{i_p}$, where $S = \{i_1, \ldots, i_p\}$ is running through all the subsets of $\{1, \ldots, n\}$ of cardinality p and $i_1 < i_2 < \cdots < i_p$. The graded algebra E is a commutative DGA with respect to the differential ∂ of degree -1 uniquely defined by the conditions $\partial e_i = 1$ for all $i = 1, \ldots, n$ and the graded Leibniz formula. Then for every $S \subset \{1, \ldots, n\}$ of cardinality p

$$\partial e_S = \sum_{j=1}^p (-1)^{j-1} e_{S_j}$$

where S_j is the complement in S to its *j*-th element.

For every $S \subset \{1, \ldots, n\}$, put $\cap S = \bigcap_{i \in S} H_i$ (possibly $\cap S = \emptyset$). The set of all intersections $L(\mathcal{A}) := \{\cap S \mid S \subset \{1, \ldots, n\}\}$ is called the intersection poset. The subset $S \subset \{1, \ldots, n\}$ is called *dependent* if $\cap S \neq \emptyset$ and the set of linear polynomials $\{\alpha_i \mid i \in S\}$ with $H_i = \alpha_i^{-1}(0)$, is linearly dependent.

Definition 2.3. The Orlik-Solomon ideal of \mathcal{A} is the ideal $I = I(\mathcal{A})$ of E generated by

- (1) all e_S with $\cap S = \emptyset$ and
- (2) all ∂e_S with S dependent.

The algebra $A := A_B^{\bullet}(\mathcal{A}) = E/I(\mathcal{A})$ is called the Orlik-Solomon algebra of \mathcal{A} .

Clearly *I* is a homogeneous ideal of *E* whence *A* is a graded algebra and we can write $A = \bigoplus_{p\geq 0} A_R^p$, where $A_R^p = E_p/(I \cap E_p)$. If *A* is central, then for any $S \subset A$, we have $\cap S \neq \emptyset$. Therefore, the Orlik-Solomon ideal is generated by the elements of type (2) from Definition 2.3. In this case, the map ∂ induces a well-defined differential $\partial: A_R^{\bullet}(A) \longrightarrow A_R^{\bullet-1}(A)$.

Recall that [18, Cor. 3.73], for each p, we can write (Brieskorn decomposition)

(1)
$$A_R^p(\mathcal{A}) = \bigoplus_{X \in L_p(\mathcal{A})} A_R^p(\mathcal{A}_X),$$

where $L_p(\mathcal{A}) := \{X \in L(\mathcal{A}) \mid \operatorname{codim} X = p\}$ and $\mathcal{A}_X := \{H \in \mathcal{A} \mid X \subset H\}$. See for example [18, Corollary 3.73].

Recall that the coning $\widetilde{\mathcal{A}} = c\mathcal{A} = \{\widetilde{H}_0, \widetilde{H}_1, \dots, \widetilde{H}_n\}$ of \mathcal{A} is a central arrangement in $\mathbb{K}^{\ell+1}$. We denote the corresponding generators of the Orlik-Solomon algebra $A_R^{\bullet}(\widetilde{\mathcal{A}})$ by $\widetilde{e}_0, \widetilde{e}_1, \dots, \widetilde{e}_n$. The map

$$\iota \colon A^1_R(\mathcal{A}) \longrightarrow A^1_R(\widetilde{\mathcal{A}}) : e_i \longmapsto \widetilde{e}_i - \widetilde{e}_0,$$

induces an injective *R*-algebra homomorphism $\iota: A_R^{\bullet}(\mathcal{A}) \longrightarrow A_R^{\bullet}(\widetilde{\mathcal{A}})$ ([31]). The image of the embedding ι is equal to the subalgebra

$$A_R^{\bullet}(\hat{\mathcal{A}})_0 := \{ \omega \in A_R^{\bullet}(\hat{\mathcal{A}}) \mid \partial(\omega) = 0 \}$$

of $A_R^{\bullet}(\widetilde{\mathcal{A}})$.

Consider the decoming $\mathcal{A}' := \mathsf{d}_{\widetilde{H}_i} \widetilde{\mathcal{A}} = \{H'_0, \dots, \widehat{H'_i}, \dots, H'_n\}$ with respect to the hyperplane $\widetilde{H}_i \in \widetilde{\mathcal{A}}$. We denote the generators of the Orlik-Solomon algebra $A^{\bullet}_R(\mathcal{A}')$ by $e'_0, \dots, \widehat{e'_i}, \dots, e'_n$. Then the Orlik-Solomon algebras of \mathcal{A} and \mathcal{A}' are isomorphic. The explicit isomorphism is given by

$$e_j \longmapsto \begin{cases} e'_j - e'_0, & \text{if } 1 \le j \le n, j \ne i, \\ -e'_0, & \text{if } j = i. \end{cases}$$

Let us fix an element $\eta = \sum_{i=1}^{n} a_i e_i \in A^1_R(\mathcal{A})$. Since $\eta \wedge \eta = 0$,

$$0 \longrightarrow A^1_R(\mathcal{A}) \xrightarrow{\eta} A^2_R(\mathcal{A}) \xrightarrow{\eta} \cdots \xrightarrow{\eta} A^\ell_R(\mathcal{A}) \xrightarrow{\eta} 0$$

forms a cochain complex, $(A_R^{\bullet}(\mathcal{A}), \eta)$, which is called the *Aomoto complex*. By the above embedding ι , we can identify the Aomoto complex $(A_R^{\bullet}(\mathcal{A}), \eta)$ with $(A_R^{\bullet}(\widetilde{\mathcal{A}})_0, \widetilde{\eta})$, where

$$\widetilde{\eta} = \iota(\eta) = \sum_{i=1}^{n} a_i \widetilde{e}_i - (a_1 + \dots + a_n) \widetilde{e}_0 \quad ([11]).$$

3. Mod p Aomoto complex and the Papadima-Suciu correspondence

In this section, we recall recent results by Papadima and Suciu [21]. They found a way of constructing a 3-net from a non-trivial element of the cohomology of Aomoto complex over \mathbb{F}_3 . Let $\overline{\mathcal{A}} = \{\overline{H}_0, \ldots, \overline{H}_n\}$ be a line arrangement on the projective plane \mathbb{KP}^2 with $3||\overline{\mathcal{A}}|$. Assume that there do not exist multiple points of multiplicity $\{3r \mid r \in \mathbb{Z}, r > 1\}$. Let

$$\widetilde{\eta}_0 := \sum_{i=0}^n \widetilde{e}_i \in A^1_{\mathbb{F}_3}(\widetilde{\mathcal{A}})_0$$

be the diagonal element. Then there is a natural bijective correspondence:

(2)
$$(H^1(A^{\bullet}_{\mathbb{F}_3}(\widetilde{\mathcal{A}})_0, \widetilde{\eta}_0) \setminus \{0\}) / \mathbb{F}_3^{\times} \xrightarrow{\simeq} \left\{ \begin{array}{c} \text{Isomorphism classes of} \\ 3\text{-net structures on } \overline{\mathcal{A}} \end{array} \right\}.$$

The correspondence is explicitly given by $H^1(A_{\mathbb{F}_3}^{\bullet}(\widetilde{\mathcal{A}})_0, \widetilde{\eta}_0) \ni \omega = \sum_{i=0}^n a_i \widetilde{e}_i \longmapsto (\overline{\mathcal{A}}_0, \overline{\mathcal{A}}_1, \overline{\mathcal{A}}_2)$, where $\overline{\mathcal{A}}_m = \{\overline{H}_i \mid a_i = m\}$ (m = 0, 1, 2). The point of the above correspondence is that by using the local structures of the Orlik-Solomon algebra, we can translate the cocycle condition into the combinatorial conditions of $(\overline{\mathcal{A}}_0, \overline{\mathcal{A}}_1, \overline{\mathcal{A}}_2)$, which turn out to be exactly the defining conditions of 3-nets. Later we will employ a similar consideration for the Aomoto complex over \mathbb{F}_2 which we summarize in this section. 3.1. A local lemma. The next lemma (cf. [16, §3]) is useful for analyzing the map $\eta: A_R^1(\mathcal{A}) \longrightarrow A_R^2(\mathcal{A})$ by the Brieskorn decomposition (1).

Lemma 3.1. Let $C_s = \{H_1, \ldots, H_s\}$ be a central arrangement in \mathbb{K}^2 (Figure 1). Let R be a commutative ring and $\eta = a_1e_1 + \cdots + a_se_s \in A^1_R(\mathcal{C}_s)$ be a degree one element of Orlik-Solomon algebra.

- (1) $\eta \wedge (e_i e_j) = -(\sum_{i=1}^s a_i) \cdot e_i \wedge e_j.$
- (2) Let $\omega = b_1 e_1 + \dots + b_s e_s \in A_R^1(\mathcal{A})$ be another element. Assume that ω and η are linearly independent (i.e., $c_1\eta + c_2\omega = 0, (c_1, c_2 \in R) \Longrightarrow c_1 = c_2 = 0$). Then $\eta \wedge \omega = 0$ if and only if $\sum_{i=1}^s a_i = \sum_{i=1}^s b_i = 0$.



FIGURE 1. Central arrangement C_s

Proof. (1) It is straightforward from the relation $e_{ij} = e_{ik} - e_{jk}$, where $e_{ij} := e_i \wedge e_j$.

(2) (Cf. [34, Proposition 2.1]) If $\sum_{i=1}^{s} b_i = 0$, then $\omega = b_1(e_1 - e_s) + \dots + b_{s-1}(e_1 - e_{s-1})$. Then applying (1), we have $\eta \wedge \omega = -(\sum_{i=1}^{s} a_i) \cdot (b_1e_{1s} + \dots + b_{s-1}e_{1,s-1})$. This is zero if $\sum_{i=1}^{s} a_i = 0$. Conversely, suppose $\eta \wedge \omega = 0$. Since \mathcal{C}_s is central, we can apply the differential ∂ . We have

$$0 = \partial(\eta \wedge \omega) = (\partial \eta)\omega - (\partial \omega)\eta.$$

By the assumption that η and ω are linearly independent, $\partial \eta = \partial \omega = 0$.

3.2. Aomoto complex over \mathbb{F}_p . Let $\mathcal{A} = \{H_1, \ldots, H_n\}$ be an arrangement of affine lines in \mathbb{K}^2 . Choose a prime p such that p|(n+1). Consider the Aomoto complex over $R = \mathbb{F}_p$ and the embedding $\iota: A^{\bullet}_{\mathbb{F}_p}(\mathcal{A}) \xrightarrow{\simeq} A^{\bullet}_{\mathbb{F}_p}(\widetilde{\mathcal{A}})_0 \subset A^{\bullet}_{\mathbb{F}_p}(\widetilde{\mathcal{A}})$. Since n is equal to -1 in \mathbb{F}_p , the image of the diagonal element $\eta_0 := e_1 + \cdots + e_n \in A^1_{\mathbb{F}_p}(\mathcal{A})$ is

$$\widetilde{\eta}_0 := \iota(\eta_0) = \widetilde{e}_0 + \widetilde{e}_1 + \dots + \widetilde{e}_n \in A^1_{\mathbb{F}_p}(\widetilde{\mathcal{A}})_0.$$

We consider the first cohomology group of the Aomoto complex $(A^{\bullet}_{\mathbb{F}_p}(\mathcal{A}), \eta_0) \simeq (A^{\bullet}_{\mathbb{F}_p}(\mathcal{A})_0, \tilde{\eta}_0)$. Let $\widetilde{\omega} = \sum_{i=0}^n a_i \widetilde{e}_i \in A^1_{\mathbb{F}_p}(\mathcal{A})_0$. Let us translate the relation $\widetilde{\eta} \wedge \widetilde{\omega} = 0$ in terms of the coefficients a_i of $\widetilde{\omega}$ by using the Brieskorn decomposition (1). For an intersection $X \in L_2(\mathcal{A})$ of codimension two, let us define the localization at X by

(3)
$$\widetilde{\omega}|_X := \sum_{\widetilde{H}_i \in \widetilde{\mathcal{A}}_X} a_i \widetilde{e}_i.$$

Proposition 3.2. With notation as above, $\tilde{\eta}_0 \wedge \tilde{\omega} = 0$ if and only if the following (i) and (ii) hold, for any $X \in L_2(\widetilde{\mathcal{A}})$.

(i) If $|\widetilde{\mathcal{A}}_X|$ is divisible by p, then $\sum_{\widetilde{H}_i \in \widetilde{\mathcal{A}}_X} a_i = 0$ in \mathbb{F}_p .

(ii) If $|\widetilde{\mathcal{A}}_X|$ is not divisible by p, then

$$a_{i_1}=a_{i_2}=\cdots=a_{i_t},$$

where $\widetilde{\mathcal{A}}_X = {\{\widetilde{H}_{i_1}, \widetilde{H}_{i_2}, \dots, \widetilde{H}_{i_t}\}}$. (This is equivalent to that $\widetilde{\omega}|_X$ and $\widetilde{\eta}_0|_X$ are linearly dependent.)

Proof. By the Brieskorn decomposition (1), $\tilde{\eta}_0 \wedge \tilde{\omega} = 0$ if and only if $\tilde{\eta}_0|_X \wedge \omega|_X = 0$ for all $X \in L_2(\widetilde{A})$. Using the Lemma 3.1 (2), it is equivalent to (i) and (ii) above.

3.3. Aomoto complex over \mathbb{F}_2 and subarrangements. Now we consider the Aomoto complex over $\mathbb{F}_2 = \mathbb{Z}/2\mathbb{Z}$. Since the coefficient is either 0 or $1 \in \mathbb{F}_2$, elements of $A^1_{\mathbb{F}_2}(\widetilde{\mathcal{A}})$ can be identified with subarrangements of $\widetilde{\mathcal{A}}$.

Definition 3.3. Let $\widetilde{S} \subset \widetilde{A}$ be a subset. Let us define an element corresponding to the subset by

$$\widetilde{e}(\widetilde{\mathcal{S}}) := \sum_{\widetilde{H}_i \in \widetilde{\mathcal{S}}} \widetilde{e}_i \in A^1_{\mathbb{F}_2}(\widetilde{\mathcal{A}}).$$

For an affine arrangement $\mathcal{A} = \{H_1, \ldots, H_n\}$ and a subset $\mathcal{S} \subset \mathcal{A}$, similarly we define

$$e(\mathcal{S}) := \sum_{H_i \in \mathcal{S}} e_i \in A^1_{\mathbb{F}_2}(\mathcal{A}).$$

Obviously the diagonal element is $\tilde{\eta}_0 = \tilde{e}(\tilde{\mathcal{A}})$ and $\tilde{e}(\tilde{\mathcal{S}}) + \tilde{\eta}_0 = \tilde{e}(\tilde{\mathcal{A}} \setminus \tilde{\mathcal{S}})$. Applying Proposition 3.2 for p = 2, we have the following.

Proposition 3.4. Let $\widetilde{\mathcal{A}} = \{\widetilde{H}_0, \widetilde{H}_1, \dots, \widetilde{H}_n\}$ be central arrangement in \mathbb{K}^3 . Let $\widetilde{\mathcal{S}} \subset \widetilde{\mathcal{A}}$ be a subset. Then $\widetilde{\eta}_0 \wedge \widetilde{e}(\widetilde{\mathcal{S}}) = 0$ if and only if the following (i) and (ii) hold, for any $X \in L_2(\widetilde{\mathcal{A}})$.

- (i) If $|\widetilde{\mathcal{A}}_X|$ is even, then $|\widetilde{\mathcal{S}}_X|$ is also even.
- (ii) If $|\widetilde{\mathcal{A}}_X|$ is odd, then either $\widetilde{\mathcal{A}}_X = \widetilde{\mathcal{S}}_X$ or $\widetilde{\mathcal{S}}_X = \emptyset$.

Remark 3.5. The existence of $\widetilde{\omega} \in A_{\mathbb{F}_2}^1(\widetilde{\mathcal{A}})$ such that $\widetilde{\omega} \neq 0, \widetilde{\omega} \neq \widetilde{\eta}_0$ and $\widetilde{\eta}_0 \wedge \widetilde{\omega} = 0$ is equivalent to the existence of a partition $\widetilde{\mathcal{A}} = \widetilde{\mathcal{A}}_1 \sqcup \widetilde{\mathcal{A}}_2$ such that at each intersection $X \in L_2(\widetilde{\mathcal{A}})$ of codimension 2, (at least) one of the following is satisfied:

- (1) \mathcal{A}_X is included in \mathcal{A}_1 or in \mathcal{A}_2 ,
- (2) $|(\widetilde{\mathcal{A}}_1)_X|$ and $|(\widetilde{\mathcal{A}}_2)_X|$ are both even.

The authors do not know any real essential arrangement which possesses the above partition. Hence, we do not know any real essential arrangement which satisfies $H^1(A^{\bullet}_{\mathbb{F}_2}(\widetilde{\mathcal{A}})_0, \widetilde{\eta}_0) \neq 0$.

Example 3.6. Suppose that $\overline{\mathcal{A}} = \overline{\mathcal{A}}_1 \sqcup \overline{\mathcal{A}}_2 \sqcup \overline{\mathcal{A}}_3 \sqcup \overline{\mathcal{A}}_4$ is a 4-net. Then

$$\begin{split} \widetilde{\eta}_0 \wedge \widetilde{e}(\widetilde{\mathcal{A}}_1 \cup \widetilde{\mathcal{A}}_2) &= \widetilde{\eta}_0 \wedge \widetilde{e}(\widetilde{\mathcal{A}}_1 \cup \widetilde{\mathcal{A}}_3) \\ &= \widetilde{\eta}_0 \wedge \widetilde{e}(\widetilde{\mathcal{A}}_1 \cup \widetilde{\mathcal{A}}_4) \\ &- 0 \end{split}$$

These three elements satisfy a linear relation,

$$\widetilde{e}(\widetilde{\mathcal{A}}_1 \cup \widetilde{\mathcal{A}}_2) + \widetilde{e}(\widetilde{\mathcal{A}}_1 \cup \widetilde{\mathcal{A}}_3) + \widetilde{e}(\widetilde{\mathcal{A}}_1 \cup \widetilde{\mathcal{A}}_4) = \widetilde{\eta}_0,$$

and span a two dimensional subspace in $H^1(A^{\bullet}_{\mathbb{F}_2}(\widehat{\mathcal{A}})_0, \widetilde{\eta}_0)$. We obtain a well-known inequality:

$$\dim H^1(A^{\bullet}_{\mathbb{F}_2}(\mathcal{A})_0, \widetilde{\eta}_0) \ge 2 \quad ([9, 20]).$$

4-NETS

4. Resonant bands description of Aomoto complex

Resonant bands provide effective tools to compute local system cohomology groups and eigenspaces of Milnor monodromies. In this section, we give a description of the cohomology of the Aomoto complex in terms of resonant bands.

4.1. Aomoto complex via chambers. We first introduce several notions related to the real structure of line arrangements. (The notions are summarized in Example 4.4 and Figure 2.) Let $\mathcal{A} = \{H_1, \ldots, H_n\}$ be an arrangement of affine lines in \mathbb{R}^2 . A connected component of $\mathbb{R}^2 \setminus \bigcup_{H \in \mathcal{A}} H$ is called a chamber. The set of all chambers is denoted by $ch(\mathcal{A})$. Let $C, C' \in ch(\mathcal{A})$. The line $H \in \mathcal{A}$ is said to separate C and C' when they are contained in opposite sides of H. The set of all lines separating C and C' is denoted by Sep(C, C'). The set of chambers $ch(\mathcal{A})$ is provided with a natural metric, the so-called adjacency distance, d(C, C') = |Sep(C, C')|.

Let us fix a flag

$$\emptyset = \mathcal{F}^{-1} \subset \mathcal{F}^0 \subset \mathcal{F}^1 \subset \mathcal{F}^2 = \mathbb{R}^2,$$

(of affine subspaces with dim $\mathcal{F}^i = i$, we also fix orientations of subspaces) satisfying the following conditions:

- (i) (genericity) \mathcal{F}^0 is not contained in $\bigcup_{i=1}^n H_i$, and \mathcal{F}^1 intersects with $\bigcup_{i=1}^n H_i$ at distinct n points.
- (ii) (near to ∞)
 - \mathcal{F}^0 does not separate *n* points $\mathcal{F}^1 \cap H_i$ (i = 1, ..., n) in \mathcal{F}^1 .
 - \mathcal{F}^1 does not separate intersections of \mathcal{A} in \mathbb{R}^2 .

(See Figure 2 for example.) Each line H_i determines two half spaces H_i^{\pm} . We choose H_i^{\pm} so that $\mathcal{F}^0 \in H_i^-$ for all $i = 1, \ldots, n$. We also fix an orientation of \mathcal{F}^1 and after re-numbering the lines, if necessary, we may assume the following

$$\mathcal{F}^0 < H_1 \cap \mathcal{F}^1 < H_2 \cap \mathcal{F}^1 < \dots < H_n \cap \mathcal{F}^1,$$

with respect to the ordering of \mathcal{F}^1 .

Associated to such a flag $\mathcal{F} = \{\mathcal{F}^{\bullet}\}$, we define a subset of $\mathsf{ch}(\mathcal{A})$ as follows.

$$\mathsf{ch}^{i}_{\mathcal{F}}(\mathcal{A}) = \{ C \in \mathsf{ch}(\mathcal{A}) \mid C \cap \mathcal{F}^{i-1} = \emptyset, C \cap \mathcal{F}^{i} \neq \emptyset \}.$$

We denote by $R[\mathsf{ch}_{\mathcal{F}}^{i}(\mathcal{A})] = \bigoplus_{C \in \mathsf{ch}_{\mathcal{F}}^{i}(\mathcal{A})} R \cdot [C]$, the free *R*-module generated by $C \in \mathsf{ch}_{\mathcal{F}}^{i}(\mathcal{A})$, where *R* is a commutative ring. It is known that $\operatorname{rank}_{R} A_{R}^{i}(\mathcal{A}) = |\mathsf{ch}_{\mathcal{F}}^{i}(\mathcal{A})|$ ([25]). We fix notations as follows.

Assumption 4.1. Let us set

 $\mathsf{ch}^{0}_{\mathcal{F}}(\mathcal{A}) = \{C_{0}\}, \ \mathsf{ch}^{1}_{\mathcal{F}}(\mathcal{A}) = \{C_{1}, \dots, C_{n}\}, \ \text{ and } \ \mathsf{ch}^{2}_{\mathcal{F}}(\mathcal{A}) = \{D_{1}, D_{2}, \dots, D_{b}\},$

where $b = |\mathsf{ch}_{\mathcal{F}}^2(\mathcal{A})|$. We can choose C_1, \ldots, C_n such that $\operatorname{Sep}(C_0, C_i) = \{H_1, H_2, \ldots, H_i\}$, or equivalently, $C_i = H_1^+ \cap \cdots \cap H_i^+ \cap H_{i+1}^- \cap \cdots \cap H_n^-$, for all $i = 0, 1, \ldots, n$ (see Figure 2).

When $1 \leq i < n$, the boundary of $C_i \cap \mathcal{F}^1$ consists of two points, $H_i \cap \mathcal{F}^1$ and $H_{i+1} \cap \mathcal{F}^1$, while $C_n \cap \mathcal{F}^1$ is a half-line and its boundary consists of a point $H_n \cap \mathcal{F}^1$.

Definition 4.2. We use the notations above. Consider $\eta = \sum_{i=1}^{n} a_i e_i \in A_R^1(\mathcal{A})$.

(1) Define the R-homomorphisms $\nabla_{\eta} \colon R[\mathsf{ch}^{0}_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^{1}_{\mathcal{F}}(\mathcal{A})]$ as follows.

$$\nabla_{\eta}([C_0]) = \sum_{C \in \mathsf{ch}_{\mathcal{F}}^1(\mathcal{A})} \left(\sum_{H_i \in \operatorname{Sep}(C_0, C)} a_i \right) \cdot [C]$$
$$= \sum_{i=1}^n (a_1 + \dots + a_i) \cdot [C_i].$$

(2) Define the map

deg:
$$\operatorname{ch}^{1}_{\mathcal{F}}(\mathcal{A}) \times \operatorname{ch}^{2}_{\mathcal{F}}(\mathcal{A}) \longrightarrow \{\pm 1, 0\},\$$

as follows.

(i) If i < n, then the segment $C_i \cap \mathcal{F}^1$ has two boundaries, say, $H_i \cap \mathcal{F}^1$ and $H_{i+1} \cap \mathcal{F}^1$.

$$\deg(C_i, D) = \begin{cases} 1 & \text{if } D \subset H_i^- \cap H_{i+1}^+, \\ -1 & \text{if } D \subset H_i^+ \cap H_{i+1}^-, \\ 0 & \text{otherwise.} \end{cases}$$

(*ii*) If i = n,

$$\deg(C_n, D) = \begin{cases} -1 & \text{if } D \subset H_n^+, \\ 0 & \text{if } D \subset H_n^-. \end{cases}$$

(3) Define the R-homomorphisms $\nabla_{\eta} \colon R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})]$ as follows.

$$\nabla_{\eta}([C]) = \sum_{D \in \mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})} \deg(C, D) \left(\sum_{H_i \in \operatorname{Sep}(C, D)} a_i \right) \cdot [D].$$

Proposition 4.3. ([27]) $(R[ch_{\mathcal{F}}^{\bullet}(\mathcal{A})], \nabla_{\eta})$ is a cochain complex. Furthermore, there is a natural isomorphism of cochain complexes,

(4)
$$\varphi \colon (R[\mathsf{ch}^{\bullet}_{\mathcal{F}}(\mathcal{A})], \nabla_{\eta}) \xrightarrow{\simeq} (A^{\bullet}_{R}(\mathcal{A}), \eta).$$

In degree 1, the isomorphism is explicitly given by

(5)
$$R[\mathsf{ch}^{1}_{\mathcal{F}}(\mathcal{A})] \xrightarrow{\simeq} A^{1}_{R}(\mathcal{A}), \ [C_{i}] \longmapsto \varphi([C_{i}]) = \begin{cases} e_{i} - e_{i+1} & \text{if } i < n, \\ e_{n} & \text{if } i = n. \end{cases}$$

In particular, we have

$$H^1(R[\mathsf{ch}^{\bullet}_{\mathcal{F}}(\mathcal{A})], \nabla_{\eta}) \simeq H^1(A^{\bullet}_R(\mathcal{A}), \eta).$$

The isomorphism (4) is natural in the sense that it respects Borel-Moore homology [27, 14, 28]. Recall that each chamber $C \in ch_{\mathcal{F}}^2(\mathcal{A})$ (with suitable orientation) determines a Borel-Moore 2-homology cycle $[C] \in H_2^{BM}(M(\mathcal{A}), R)$ of the complexified complement $M(\mathcal{A})$. The isomorphism (4), for i = 2, is obtained by the composition

(6)
$$R[\mathsf{ch}_{\mathcal{F}}^2(\mathcal{A})] \longrightarrow H_2^{BM}(M(\mathcal{A}), R) \xrightarrow{\simeq} H^2(M(\mathcal{A}), R) \simeq A_R^i(\mathcal{A}).$$

Example 4.4. Let $\mathcal{A} = \{H_1, \ldots, H_6\}$ be six affine lines as in Figure 2. We also fix a flag (with orientation) $\mathcal{F} = \{\mathcal{F}^0 \subset \mathcal{F}^1\}$ (as in Figure 2). There are 16 chambers. We have

$$\begin{aligned} \mathsf{ch}_{\mathcal{F}}^{0}(\mathcal{A}) &= \{C_{0}\},\\ \mathsf{ch}_{\mathcal{F}}^{1}(\mathcal{A}) &= \{C_{1}, C_{2}, \dots, C_{6}\},\\ \mathsf{ch}_{\mathcal{F}}^{2}(\mathcal{A}) &= \{D_{1}, D_{2}, \dots, D_{9}\}. \end{aligned}$$

4-NETS

The degree maps are computed, as follows.

$\deg(C_i, D_j)$	D_1	D_2	D_3	D_4	D_5	D_6	D_7	D_8	D_9
C_1	0	-1	0	0	1	0	1	1	0
C_2	-1	0	0	-1	-1	0	0	-1	0
C_3	1	1	1	1	1	1	0	1	1
C_4	-1	$^{-1}$	$^{-1}$	0	0	0	0	0	0
C_5	0	0	0	$^{-1}$	$^{-1}$	-1	0	0	0
C_6	0	0	0	0	0	0	-1	-1	-1

Consider $\eta = \sum_{i=1}^{6} a_i e_i \in A^1_R(\mathcal{A})$. We will compute the map ∇_{η} . The first one $\nabla_{\eta} \colon R[\mathsf{ch}^0_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})]$ is, by definition,

$$\nabla_{\eta}([C_0]) = a_1 \cdot [C_1] + a_{12} \cdot [C_2] + a_{123} \cdot [C_3] + a_{1234} \cdot [C_4] + a_{12345} \cdot [C_5] + a_{123456} \cdot [C_6],$$

where $a_{ijk} = a_i + a_j + a_k$, etc. The second one $\nabla_\eta \colon R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})]$ is given as follows.



FIGURE 2. Example 4.4

4.2. Aomoto complex via resonant bands. Let $\mathcal{A} = \{H_1, \ldots, H_n\}$ be an arrangement of affine lines in \mathbb{R}^2 . We fix the flag \mathcal{F} as in §4.1. The cohomology of the Aomoto complex can be computed using chambers. In this subsection, we introduce the notion " η -resonant bands" which enables us to simplify the computation of cohomology. This can be regarded as "the Aomoto complex version" of the results in [29, 30].

Definition 4.5. A band B is a region bounded by a pair of consecutive parallel lines H_i and H_{i+1} .

Each band *B* contains two unbounded chambers $U_1(B), U_2(B) \in \mathsf{ch}(\mathcal{A})$. Since *B* intersects \mathcal{F}^1 , we may assume that $B \cap \mathcal{F}^1 = U_1(B) \cap \mathcal{F}^1$ and $U_2(B) \cap \mathcal{F}^1 = \emptyset$. In other words, $U_1(B) \in \mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})$ and $U_2(B) \in \mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})$. The distance $d(U_1(B), U_2(B))$ is called the *length* of the band *B*.

Definition 4.6. Let $\eta = \sum_{i=1}^{n} a_i e_i \in A^1_R(\mathcal{A})$. A band B is called η -resonant if

$$\sum_{H_i \in \operatorname{Sep}(U_1(B), U_2(B))} a_i = 0.$$

We denote by $\operatorname{RB}_{\eta}(\mathcal{A})$ the set of all η -resonant bands.

We can extend U_1 to an injective *R*-module homomorphism $U_1: R[\operatorname{RB}_{\eta}(\mathcal{A})] \hookrightarrow R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})].$ We denote by

$$\widetilde{\nabla}_{\eta} := -\nabla_{\eta} \circ U_1 \colon R[\operatorname{RB}_{\eta}(\mathcal{A})] \longrightarrow R[\operatorname{ch}^2_{\mathcal{F}}(\mathcal{A})]$$

the composition of U_1 and ∇_{η} (multiplied by -1).

More precisely, to each η -resonant band $B \in \operatorname{RB}_{\eta}(\mathcal{A})$, we associate an element

$$\widetilde{\nabla}_{\eta}(B) \in R[\mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})]$$

as follows:

$$\widetilde{\nabla}_{\eta}(B) := -\nabla_{\eta}(U_1(B)) = \sum_{D \in \mathsf{ch}(\mathcal{A}), D \subset B} \left(\sum_{H_i \in \operatorname{Sep}(U_1(B), D)} a_i \right) \cdot [D].$$

Example 4.7. Let $\mathcal{A} = \{H_1, \ldots, H_6\}$ be an arrangement of lines as in Figure 2. There are three bands B_1, B_2, B_3 , i.e., those defined by (H_2, H_3) , (H_4, H_5) and (H_5, H_6) , respectively. We have

$$U_1(B_1) = C_2, U_2(B_1) = D_8, U_1(B_2) = C_4, U_2(B_2) = D_3, \text{ and } U_1(B_3) = C_5, U_2(B_3) = D_6.$$

The band B_1 has length 4, while B_2 and B_3 have length 3. Let $\eta = a_1e_1 + \cdots + a_6e_6 \in A_R^1(\mathcal{A})$. The band B_1 is η -resonant if and only if $a_1 + a_4 + a_5 + a_6 = 0$. Then we have

$$\nabla_{\eta}([B_1]) = a_4[D_1] + (a_4 + a_5)[D_4] + (a_1 + a_4 + a_5)[D_5]$$

Obviously the map U_1 induces $U_1: \operatorname{Ker}(\widetilde{\nabla}_{\eta}) \longrightarrow \operatorname{Ker}(\nabla_{\eta} : R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})])$. Thus we have a natural map

(7)
$$\widetilde{U}_1 \colon \operatorname{Ker}(\widetilde{\nabla}_{\eta}) \longrightarrow H^1(R[\mathsf{ch}_{\mathcal{F}}^{\bullet}(\mathcal{A})], \nabla_{\eta}).$$

The above map \widetilde{U}_1 is neither injective nor surjective in general. The following is the main result concerning resonant bands which asserts that the map \widetilde{U}_1 above (7) is isomorphic under certain non-resonant assumption at infinity. This provides an effective way to compute $H^1(R[\mathsf{ch}^{\bullet}_{\mathcal{F}}(\mathcal{A})], \nabla_{\eta})$. Indeed, normally, $|\mathrm{RB}_{\eta}(\mathcal{A})|$ is much smaller than $|\mathsf{ch}^{1}_{\mathcal{F}}(\mathcal{A})|$.

Theorem 4.8. Let R be a commutative ring and $\eta = \sum_{i=1}^{n} a_i e_i \in A_R^1(\mathcal{A})$.

- (i) Suppose that $\alpha := \sum_{i=1}^{n} a_i \in \mathbb{R}^{\times}$ is invertible. Then the natural map \widetilde{U}_1 is injective.
- (ii) We assume that R is an integral domain and $\alpha := \sum_{i=1}^{n} a_i \in R^{\times}$. Then \widetilde{U}_1 is an isomorphism.
- (iii) Let R be an arbitrary commutative ring. If $\alpha := \sum_{i=1}^{n} a_i \in R^{\times}$ and all bands are η -resonant, then the natural map \widetilde{U}_1 is an isomorphism.

4-NETS

Proof. (i) Let

$$\sum r_B \cdot [B] := \sum_{B \in \mathrm{RB}_\eta(\mathcal{A})} r_B \cdot [B] \in R[\mathrm{RB}_\eta(\mathcal{A})], \ r_B \in R$$

Suppose $\sum r_B \cdot [B] \in \operatorname{Ker} \widetilde{U}_1$, that is, $U_1(\sum r_B \cdot [B]) \in \operatorname{Im} \left(\nabla_\eta \colon R[\mathsf{ch}^0_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})] \right)$. Since $R[\mathsf{ch}^0_{\mathcal{F}}(\mathcal{A})] = R \cdot [C_0]$, there exists an element $s \in R$ such that

(8)
$$\sum r_B \cdot [U_1(B)] = s \cdot \nabla_\eta([C_0]).$$

Note that in the left hand side of (8), the chamber C_n does not appear, because C_n is not bounded by two parallel lines. By Definition 4.2, $\nabla_{\eta}([C_0]) = \sum_{i=1}^{n} (a_1 + \cdots + a_i) \cdot [C_i]$. The coefficient of $[C_n]$ is equal to $s \cdot (a_1 + \cdots + a_n) = s \cdot \alpha$. By the assumption that α is invertible, we have s = 0. Hence $\sum r_B \cdot U_1(B) = s \cdot \nabla_{\eta}([C_0]) = 0$, and, since U_1 is injective, we have $\sum r_B \cdot [B] = 0$.

Next we show the surjectivity of (7). Suppose that

$$\beta = \sum_{i=1}^{n} b_i \cdot [C_i] \in \operatorname{Ker}(\nabla_\eta \colon R[\mathsf{ch}^1_{\mathcal{F}}(\mathcal{A})] \longrightarrow R[\mathsf{ch}^2_{\mathcal{F}}(\mathcal{A})])$$

Consider the following element,

(9)
$$\beta' = \beta - \frac{b_n}{\alpha} \cdot \nabla_{\eta}([C_0])$$
$$= \sum_{i=1}^{n-1} b'_i \cdot [C_i].$$

Obviously, β and β' represent the same element in $H^1(R[\mathsf{ch}^{\bullet}_{\mathcal{F}}(\mathcal{A})], \nabla_{\eta})$. It is sufficient to show that $\beta' \in \operatorname{Im} U_1$.

Next we consider a chamber C_i (i < n) such that H_i and H_{i+1} are not parallel. Then there is a unique chamber $D_p \in ch_{\mathcal{F}}^2(\mathcal{A})$ such that $\operatorname{Sep}(C_i, D_p) = \mathcal{A}$, which is called the "opposite chamber of C_i " in [26, Def. 2.1] and denoted by $D_p = C_i^{\vee}$. Then we consider the coefficient $c_{C_i^{\vee}}$ of $[C_i^{\vee}]$ in $\nabla_{\eta}(\beta') = \sum c_D \cdot [D]$. Since C_i^{\vee} appears only in $\nabla_{\eta}([C_i])$ and $\nabla_{\eta}([C_n])$, and the coefficient of $[C_n]$ is already zero, we have $c_{C_i^{\vee}} = \alpha \cdot b'_i$. By the assumption that $\alpha \in \mathbb{R}^{\times}$, $\nabla_{\eta}(\beta') = 0$, in particular $c_{C_i^{\vee}} = 0$, implies that $b'_i = 0$. So $\beta' = \sum_{i=1}^{n-1} b'_i \cdot [C_i]$ is a linear combination of C_i 's (i < n) such that H_i and H_{i+1} are parallel. So far, we only use the fact $\alpha \in \mathbb{R}^{\times}$. If all bands are η -resonant, then we have already proved that β' is generated by $U_1(B)$ with $B \in \operatorname{RB}_{\eta}(\mathcal{A})$. Thus (iii) is proved.

Now we assume that R is an integral domain. We will prove (ii). Let C_i be a chamber such that the walls H_i and H_{i+1} are parallel. Let B be the corresponding band defined by H_i and H_{i+1} . Note that $C_i = U_1(B)$ and its opposite chamber is $U_2(B)$. Suppose that B is not an η -resonant band, that is, $\alpha' := \sum_{H_j \in \text{Sep}(U_1(B), U_2(B))} a_j \neq 0$. Again consider the coefficient of $[U_2(B)]$ in $\nabla_{\eta}(\beta')$. $[U_2(B)]$ appears in $\nabla_{\eta}([C_i])$ and some other terms $\nabla_{\eta}([C_k])$ for k such that H_k and H_{k+1} are not parallel. However the coefficients of chambers of the second type in β' are already zero. Therefore the coefficient of $[U_2(B)]$ in $\nabla_{\eta}(\beta')$ is $-\alpha' \cdot b'_i$, which is zero. Since R is an integral domain, we have $b'_i = 0$. Hence β' is a linear combination of $U_1(B)$'s where $B \in \text{RB}_{\eta}(\mathcal{A})$. This completes the proof of the surjectivity. \Box **Remark 4.9.** Equation (7) and Theorem 4.8 are concerning the following homomorphism of cochains.

The map \widetilde{U}_1 is nothing but the homomorphism $\operatorname{Ker}(\widetilde{\nabla}_\eta) \longrightarrow H^1(A^{\bullet}_R(\mathcal{A}), \eta)$ induced by φ_1 . By Proposition 4.3 (especially, the explicit map (5)), the map φ_1 above is given by

$$[B] \longmapsto e_i - e_{i+1},$$

where B is a η -resonant band bounded by the lines H_i and H_{i+1} .

Example 4.10. Let $R = \mathbb{F}_2$. Let $\mathcal{A} = \{H_1, \ldots, H_6\}$ be an arrangement of affine lines as in Figure 3 (which is $\mathcal{A}(7,1)$ in [12]). Let $\eta = e_2 + e_3 + e_6 \in A_R^1(\mathcal{A})$ (the supporting lines of η are colored blue). There are three bands B_1 (bounded by H_1 and H_2), B_2 (bounded by H_3 and H_4), and B_3 (bounded by H_5 and H_6). Sep $(U_1(B_1), U_2(B_1)) = \{H_3, H_4, H_5, H_6\}$ and two of the lines, H_3 and H_6 , have non-zero coefficients in η . Hence B_1 is an η -resonant band. Similarly, B_2 and B_3 are η -resonant, and we have $\operatorname{RB}_{\eta}(\mathcal{A}) = \{B_1, B_2, B_3\}$. By definition, $\widetilde{\nabla}_{\eta}(B_1) = \widetilde{\nabla}_{\eta}(B_2) = \widetilde{\nabla}_{\eta}(B_3) = [D_1]$. Hence the kernel

$$\operatorname{Ker}(\widetilde{\nabla}_{\eta} \colon \mathbb{F}_2[\operatorname{RB}_{\eta}(\mathcal{A})] \longrightarrow \mathbb{F}_2[\operatorname{ch}(\mathcal{A})])$$

is 2-dimensional (generated by $[B_1] - [B_2]$ and $[B_2] - [B_3]$). By Theorem 4.8, $H^1(A^{\bullet}_{\mathbb{F}_2}(\mathcal{A}), \eta) \simeq \mathbb{F}_2^2$.



FIGURE 3. Example 4.10

Example 4.11. We consider $\overline{\mathcal{A}} = \mathcal{A}(16, 1) = \{\overline{H}_1, \ldots, \overline{H}_{16}\}$ from the Grünbaum's catalogue [12], see Figure 4. Let us denote by $\mathcal{A} = \{H_2, H_3, \ldots, H_{16}\}$ the decoming $\mathsf{d}_{\widetilde{H}_1}\widetilde{\mathcal{A}}$, the lower-left one in Figure 4. The affine arrangement \mathcal{A} has 7 bands B_1, \ldots, B_7 . To indicate the choice of $U_1(B)$ and $U_2(B)$, we always put the name B of the band in the unbounded chamber $U_1(B)$.

Let $R = \mathbb{Z}/8\mathbb{Z}$. Define $\tilde{\eta}_1, \tilde{\eta}_2 \in A^1_R(\mathcal{A})_0$ by

$$\widetilde{\eta}_1 = \widetilde{e}_1 + \widetilde{e}_3 + \widetilde{e}_5 + \widetilde{e}_7 + \widetilde{e}_9 + \widetilde{e}_{11} + \widetilde{e}_{13} + \widetilde{e}_{15},$$

$$\widetilde{\eta}_2 = \widetilde{e}_2 + \widetilde{e}_4 + \widetilde{e}_6 + \widetilde{e}_8 + \widetilde{e}_{10} + \widetilde{e}_{12} + \widetilde{e}_{14} + \widetilde{e}_{16},$$

and set $\widetilde{\eta} := \widetilde{\eta}_1 + 6\widetilde{\eta}_2$.

Let $\eta = (e_3 + e_5 + e_7 + \dots + e_{15}) + 6(e_2 + e_4 + \dots + e_{16}) \in A^1_R(\mathcal{A})$. Then all 7 bands are η -resonant. Thus we can apply theorem Theorem 4.8 (iii). The kernel $\operatorname{Ker}(\widetilde{\nabla}_{\eta} : R[\operatorname{RB}_{\eta}(\mathcal{A})] \longrightarrow R[\operatorname{ch}(\mathcal{A})])$ is a free R-module generated by

$$[B_1] + 2[B_2] + 3[B_3] + 4[B_4] + 5[B_5] + 6[B_6] + 7[B_7].$$

The corresponding element (via the correspondence Remark 4.9) in $A^1_R(\widetilde{\mathcal{A}})_0$ is

 $4(\widetilde{e}_2+\widetilde{e}_3)+3(\widetilde{e}_4-\widetilde{e}_7+\widetilde{e}_{13}-\widetilde{e}_{16})+2(\widetilde{e}_6+\widetilde{e}_9-\widetilde{e}_{11}-\widetilde{e}_{14})+(\widetilde{e}_5+\widetilde{e}_8-\widetilde{e}_{12}-\widetilde{e}_{15}).$

By Theorem 4.8 (iii), the cohomology of the Aomoto complex

$$H^1(A^1_R(\widetilde{\mathcal{A}})_0,\widetilde{\eta}) \simeq H^1(A^1_R(\mathcal{A}),\eta) \simeq \operatorname{Ker}(\widetilde{\nabla}_\eta) \simeq R \simeq \mathbb{Z}/8\mathbb{Z}$$

is non-vanishing.



FIGURE 4. $\mathcal{A}(16, 1)$ and decomings with respect to H_1 and H_{10} .

Remark 4.12. Let us point out a possible relation between $\mathbb{Z}/8\mathbb{Z}$ -resonance in Example 4.11 and isolated torsion points of order 8 in the characteristic variety of $\mathcal{A}(16, 1)$.

Let us denote $M = M(\mathcal{A}(16, 1)) = \mathbb{CP}^2 \setminus \bigcup_{H \in \mathcal{A}(16, 1)} H_{\mathbb{C}}$ the complexified complement. Recall that the character torus of M is

$$\mathbb{T} := \operatorname{Hom}(\pi_1(M), \mathbb{C}^{\times}) \simeq \{ \boldsymbol{t} = (t_1, t_2, \dots, t_{16}) \in (\mathbb{C}^{\times})^{16} \mid \prod_{i=1}^{16} t_i = 1 \}$$

We also define the essential open subset of \mathbb{T} by

 $\mathbb{T}^{\circ} := \{ \boldsymbol{t} = (t_1, \dots, t_{16}) \in \mathbb{T} \mid t_i \neq 1, \forall i = 1, \dots, 16 \}.$

The characteristic variety $\mathcal{V}^1(\mathcal{A}(16,1))$ of $\mathcal{A}(16,1)$ is the set of points in the character torus \mathbb{T} such that the associated local system has non-vanishing first cohomology, i.e.,

$$\mathcal{V}^1(\mathcal{A}(16,1)) = \{ \boldsymbol{t} \in \mathbb{T} \mid \dim H^1(M, \mathcal{L}_{\boldsymbol{t}}) \ge 1 \}$$

Let $\zeta = e^{2\pi i/8}$ and consider the following point,

$$\rho = (\zeta, \zeta^6, \zeta, \zeta^6) \in \mathbb{T}^{\circ}.$$

Let us recall quickly the resonant band algorithm for computing local system cohomology groups (see [30] for details). For a given local system $\mathcal{L}_{\mathbf{t}}$, we define the set $\operatorname{RB}_{\mathcal{L}_{\mathbf{t}}}(\mathcal{A})$ of $\mathcal{L}_{\mathbf{t}}$ -resonant bands and the map $\nabla_{\mathcal{L}_{\mathbf{t}}} : \mathbb{C}[\operatorname{RB}_{\mathcal{L}_{\mathbf{t}}}(\mathcal{A})] \longrightarrow \mathbb{C}[\operatorname{ch}(\mathcal{A})]$. If $\mathcal{L}_{\mathbf{t}}$ has non-trivial monodromy around the line at infinity, then we have the isomorphism $H^1(M, \mathcal{L}_{\mathbf{t}}) \simeq \operatorname{Ker}(\nabla_{\mathcal{L}_{\mathbf{t}}})$.

Since \mathcal{L}_{ρ} defined above has non trivial monodromy around any line, we can apply resonant band algorithm to any decomings. Here we exhibit two cases (although the results coincide logically), $\mathsf{d}_{\widetilde{H}_1}\widetilde{\mathcal{A}}$ and $\mathsf{d}_{\widetilde{H}_1\circ}\widetilde{\mathcal{A}}$. (See Figure 4.)

• The affine arrangement $\mathsf{d}_{\widetilde{H}_1}\mathcal{A}$ has seven bands B_1, \ldots, B_7 , which are all \mathcal{L}_{ρ} -resonant. Then $\operatorname{Ker}(\nabla_{\mathcal{L}_{\rho}})$ is one dimensional and generated by the following element,

$$\sin\left(\frac{\pi}{8}\right)[B_1] - \sin\left(\frac{\pi}{4}\right)[B_2] + \sin\left(\frac{3\pi}{8}\right)[B_3] - \sin\left(\frac{\pi}{2}\right)[B_4] + \sin\left(\frac{3\pi}{8}\right)[B_5] - \sin\left(\frac{\pi}{4}\right)[B_6] + \sin\left(\frac{\pi}{8}\right)[B_7].$$

 The affine arrangement d_{H̃10} *Ã* has nine bands B'₁,..., B'₉, which are all L_ρ-resonant. Then Ker(∇_{L_ρ}) is one dimensional and generated by the following element,

$$[B_1] + \sqrt{2}[B_2] + [B_3]$$

$$-[B_4] + (1 + \sqrt{2})[B_5] - (2 + \sqrt{2})[B_6] + (2 + \sqrt{2})[B_7] + (1 + \sqrt{2})[B_8] + [B_9].$$

Hence we have dim $H^1(M, \mathcal{L}_{\rho}) = 1$. Furthermore, we can prove that ρ generates the essential part of the characteristic variety. More precisely, we have the following,

(10)
$$\mathcal{V}^{1}(\mathcal{A}(16,1)) \cap \mathbb{T}^{\circ} = \{\rho, \rho^{2}, \rho^{3}, \rho^{5}, \rho^{6}, \rho^{7}\}.$$

4.3. Resonant bands over \mathbb{F}_2 and subarrangements. Let $\mathcal{A} = \{H_1, \ldots, H_n\}$ be an arrangement of affine lines in \mathbb{R}^2 . Let $\mathcal{S} \subset \mathcal{A}$ be a subset. Denote $e(\mathcal{S}) := \sum_{H_i \in \mathcal{S}} e_i \in A^1_{\mathbb{F}_2}(\mathcal{A})$.

Clearly, $e(S) + e(A) = e(A \setminus S)$. Below is the summary of "subarrangement description of resonant band algorithm":

(a) Let B be a band of \mathcal{A} . Then $B \in \operatorname{RB}_{e(\mathcal{S})}(\mathcal{A})$ if and only if the number of lines in \mathcal{S} separating $U_1(B)$ and $U_2(B)$ is even, i.e., $2||\mathcal{S} \cap \operatorname{Sep}(U_1(B), U_2(B))|$.

(b) $\widetilde{\nabla}_{e(\mathcal{S})} \colon \mathbb{F}_2[\operatorname{RB}_{e(\mathcal{S})}(\mathcal{A})] \longrightarrow \mathbb{F}_2[\operatorname{ch}^2_{\mathcal{F}}(\mathcal{A})]$ is given by the following formula.

$$\widetilde{\nabla}_{e(\mathcal{S})}(B) = \sum_{C \in \mathsf{ch}(\mathcal{A}), C \subset B} |\mathcal{S} \cap \operatorname{Sep}(U_1(B), C)| \cdot [C].$$

(See Example 4.10). In particular, if we consider $\eta_0 = e(\mathcal{A}) = e_1 + e_2 + \cdots + e_n$, then we have

$$\widetilde{\nabla}_{\eta_0}(B) = \sum_{C \in \mathsf{ch}(\mathcal{A}), C \subset B} d(U_1(B), C) \cdot [C].$$

(c) Suppose that $|\mathcal{S}|$ is odd. Then we can apply Theorem 4.8, and we have an isomorphism

$$\Psi_{e(S)} \colon \operatorname{Ker}(\stackrel{\simeq}{\nabla}_{e(S)}) \xrightarrow{\simeq} H^1(A^{\bullet}_{\mathbb{F}_2}(\mathcal{A}), e(\mathcal{S})).$$

(d) Using Remark 4.9 (and Proposition 4.3 (especially, the explicit map (5))), the above isomorphism is given by

$$\Psi_{e(S)}([B]) = e_i + e_{i+1} \in A^1_{\mathbb{F}_2}(\mathcal{A}),$$

where B is a e(S)-resonant band determined by the lines H_i and H_{i+1} .

5. Non-existence of real 4-nets

5.1. Aomoto complex for the diagonal element. Let $\mathcal{A} = \{H_1, \ldots, H_n\}$ be an arrangement of affine lines in \mathbb{R}^2 with odd n. Let $\widetilde{\mathcal{A}} = \{\widetilde{H}_0, \widetilde{H}_1, \ldots, \widetilde{H}_n\}$ be the coning of \mathcal{A} and $\overline{\mathcal{A}} = \{\overline{H}_0, \overline{H}_1, \ldots, \overline{H}_n\}$ be the projectivization. Recall that

$$\widetilde{\eta}_0 := \widetilde{e}(\widetilde{\mathcal{A}}) = \widetilde{e}_0 + \widetilde{e}_1 + \dots + \widetilde{e}_n \in A^1_{\mathbb{F}_2}(\widetilde{\mathcal{A}})_0$$

is the diagonal element and $\eta_0 = e(\mathcal{A}) = e_1 + \cdots + e_n \in A^1_{\mathbb{F}_2}(\mathcal{A})$. Notice that *n* odd implies that the map Ψ_{η_0} is an isomorphism by §4.3 (c).

Choose a subset $\widetilde{\mathcal{S}} \subset \widetilde{\mathcal{A}}$. In the figures below, the lines in $\widetilde{\mathcal{S}}$ are colored in red. The other lines are black.

As we saw in Proposition 3.4, the relation $\tilde{\eta}_0 \wedge \tilde{e}(\tilde{S}) = 0$ is equivalent to " $|\overline{\mathcal{A}}_X|$ is even $\Longrightarrow |\overline{\mathcal{S}}_X|$ is even" and " $|\overline{\mathcal{A}}_X|$ is odd \Longrightarrow either $\overline{\mathcal{S}}_X = \emptyset$ or $\overline{\mathcal{S}}_X = \overline{\mathcal{A}}_X$ " for $\forall X \in L_2(\overline{\mathcal{A}})$. From this, it is easily seen that if the multiplicity is $|\overline{\mathcal{A}}_X| \leq 3$, then $\overline{\mathcal{A}}_X$ is monocolor (either all red $\overline{\mathcal{S}}_X = \overline{\mathcal{A}}_X$ or all black $\overline{\mathcal{S}}_X = \emptyset$). However, when $|\overline{\mathcal{A}}_X| = 4$, there are four cases (Figure 5):

- (i) $\overline{\mathcal{S}}_X = \emptyset$.
- (*ii*) $\overline{\mathcal{S}}_X = \overline{\mathcal{A}}_X$.
- (*iii*) $|\overline{\mathcal{S}}_X| = 2$ and the lines in $\overline{\mathcal{S}}_X$ are adjacent.
- (iv) $|\overline{\mathcal{S}}_X| = 2$ and the lines in $\overline{\mathcal{S}}_X$ are separated by lines in $\overline{\mathcal{A}}_X \setminus \widetilde{\mathcal{S}}_X$.



FIGURE 5. Local structures of \overline{S}_X . (Members of \overline{S}_X are red, and \overline{S}_X equals $\{\overline{H}_1, \overline{H}_3\}$ in (*iii*) and (*iv*)).

The cases (iii) and (iv) are combinatorially identical. However, the real structures are different. This difference is crucial, actually, by using resonant bands, we can prove that (iv) can not happen ("Non Separation Theorem").

Theorem 5.1. Let $\overline{S} \subset \overline{A}$. Suppose that $\tilde{\eta}_0 \wedge \tilde{e}(\widetilde{S}) = 0$. Let $X \in \mathbb{RP}^2$ be an intersection of \overline{A} such that $|\overline{A}_X| = 4$ and $|\overline{S}_X| = 2$. Then the two lines of \overline{S}_X are adjacent as Figure 5 (iii). In particular, (iv) does not happen.

Proof. Suppose that there exists $X \in \mathbb{RP}^2$ such that $\overline{\mathcal{A}}_X = \{\overline{H}_0, \overline{H}_1, \overline{H}_2, \overline{H}_3\}$ with $\overline{\mathcal{S}}_X = \{\overline{H}_1, \overline{H}_3\}$ arranging as (iv) in Figure 5.

First consider the deconing with respect to \overline{H}_0 , we have $\mathcal{A} = \mathsf{d}_{H_0} \widetilde{\mathcal{A}} = \{H_1, \ldots, H_n\}$. Then



FIGURE 6. Deconings $\mathsf{d}_{\widetilde{H}_0}\widetilde{\mathcal{A}}$ and $\mathsf{d}_{\widetilde{H}_1}\widetilde{\mathcal{A}}$

$$\mathcal{S} = \{H_1, H_3, \dots\} \subset \mathcal{A}.$$

The lines H_1, H_2, H_3 are parallel (the left of Figure 6) and determine two bands B_1 (bounded by H_1 and H_2) and B_2 (bounded by H_2 and H_3). Note that $e(\mathcal{S}) = e_1 + e_3 + \cdots \in A^1_{\mathbb{F}_2}(\mathcal{A})$. By the correspondence in §4.3 (d), we have

$$\Psi_{n_0}^{-1}(e(\mathcal{S})) = [B_1] + [B_2] + \dots$$

in particular, both $[B_1]$ and $[B_2]$ appear. (Otherwise, e_1, e_3 can not appear.) On the other hand, we have the following relation

(11)
$$\widetilde{\nabla}_{\eta_0}(\Psi_{\eta_0}^{-1}(e(\mathcal{S})) = \widetilde{\nabla}_{\eta_0}([B_1]) + \widetilde{\nabla}_{\eta_0}([B_2]) + \dots = 0.$$

Choose a chamber C such that $C \subset B_2$ and $d(U_1(B_2), C) = 1$. Let $\text{Sep}(U_1(B_2), C) = \{H_{i_0}\}$. The chamber C is adjacent to an unbounded chamber $U_1(B_2)$, hence, C is contained in at most two bands B_2 and B_{j_0} . Since $\widetilde{\nabla}_{\widetilde{\eta}}([B_2]) = [C] + \cdots \in \mathbb{F}_2[\text{RB}_{\widetilde{\eta}}(\mathcal{A})]$, by (11), [C] must be cancelled by another resonant band B_{j_0} which appears in $\Psi_{\eta_0}^{-1}(e(\mathcal{S}))$. Thus we have

$$\Psi_{\eta_0}^{-1}(e(\mathcal{S})) = [B_1] + [B_2] + \dots + [B_{j_0}] + \dots$$

Let H_{i_0} and H_{i_0+1} be walls of B_{j_0} . Then applying Ψ , we have

$$e(\mathcal{S}) = (e_1 + e_2) + (e_2 + e_3) + \dots + (e_{i_0} + e_{i_0+1}) + \dots$$
$$= e_1 + e_3 + \dots + e_{i_0} + \dots$$

4-NETS

Here note that e_{i_0} survives because B_{j_0} is the only band which has H_{i_0} as a wall. This implies $H_{i_0} \in S$. Therefore, if $C \subset B_2$ and $d(U_1(B_2), C) = 1$, then $\text{Sep}(U_1(B_2), C) \subset S$. (Left hand side of Figure 6.) The same assertion holds for the opposite unbounded chamber $U_2(B_2)$.

Next we consider $\overline{\mathcal{S}}' := \overline{\mathcal{A}} \setminus \overline{\mathcal{S}}$. Since $\widetilde{e}(\widetilde{\mathcal{S}}') = \widetilde{\eta}_0 + \widetilde{e}(\widetilde{\mathcal{S}})$, $\widetilde{\eta}_0 \wedge \widetilde{e}(\widetilde{\mathcal{S}}') = 0$. In Figure 5 (*iv*), the roles of black and red lines exchange. Black lines are the members of $\overline{\mathcal{S}}'$ and red lines are not. We take the decoming with respect to \widetilde{H}_1 , we have $d_{\widetilde{H}_1}\widetilde{\mathcal{A}} = \{H'_0, H'_2, H'_3, \ldots, H'_n\}$ (Right hand side of Figure 6). Then $\mathcal{S}' = \{H'_0, H'_2, \ldots\} \subset d_{\widetilde{H}_1}\widetilde{\mathcal{A}}$. The lines H'_0, H'_2, H'_3 are parallel and determine two bands B'_2 (bounded by H'_2 and H'_3) and B'_3 (bounded by H'_3 and H'_0). By a similar argument to the previous case (decoming with respect to \widetilde{H}_0), we can conclude that if $C' \subset B'_2$ and $d(U_1(B'_2), C') = 1$, then $\operatorname{Sep}(U_1(B'_2), C') \subset \mathcal{S}'$. (Right hand side of Figure 6.) The same assertion holds for the opposite unbounded chamber $U_2(B'_2)$.

The bands B_2 and B'_2 are identical in the projective plane \mathbb{RP}^2 . However, the colors of boundaries of unbounded chambers are different. This is a contradiction. Thus the case (iv) can not happen.

5.2. Real 4-nets do not exist.

Theorem 5.2. There does not exist a real arrangement \overline{A} that supports a 4-net structure.

Proof. Suppose $\overline{\mathcal{A}}$ supports a 4-net structure with partition $\overline{\mathcal{A}} = \overline{\mathcal{A}}_1 \sqcup \overline{\mathcal{A}}_2 \sqcup \overline{\mathcal{A}}_3 \sqcup \overline{\mathcal{A}}_4$. There exists a multiple point $X \in \mathbb{RP}^2$ of $\overline{\mathcal{A}}$ with multiplicity 4 such that X is the intersection point of 4 lines $H_i \in \mathcal{A}_i$. Suppose that the lines are ordered as in Figure 7.



FIGURE 7. Local structure of a 4-net.

We can now define $\widetilde{S} = \widetilde{\mathcal{A}}_1 \sqcup \widetilde{\mathcal{A}}_3$. Then as in Example 3.6, we have $\widetilde{\eta}_0 \wedge \widetilde{e}(\widetilde{S}) = 0$. By definition, $\overline{\mathcal{S}}_X = \{\overline{H}_1, \overline{H}_3\}$ consists of two lines separated by the other two lines \overline{H}_2 and \overline{H}_4 . Therefore (iv) in Figure 5 happens. This contradicts the Non-separation Theorem 5.1.

Remark 5.3. The non-existence of real 4-nets was proved in [6, Lem. 2.4]. Their proof relies on the metric structure of \mathbb{R}^2 . So it does not apply to oriented matroids. Our arguments actually prove that there do not exist rank 3 oriented matroids (equivalently, pseudo-line arrangements in \mathbb{RP}^2) which have 4-net structures. The details are omitted.

Acknowledgments: The authors thank Alex Suciu for explaining the ideas of the preprint [21], Mike Falk and Mustafa Hakan Gunturkun for telling us of the reference [6]. The authors also thank the referee for the careful reading and many suggestions. During the preparation of this paper, the authors were partially supported by JSPS Postdoctoral Fellowship For Foreign Researchers, the Grant-in-Aid for Scientific Research (C) 25400060, and the Grant-in-aid (No. 23224001 (S)) for Scientific Research, JSPS.

References

- P. Bailet, On the monodromy of Milnor fibers of hyperplane arrangements. Canad. Math. Bull. 57 (2014), 697-707. DOI: 10.4153/CMB-2014-032-4
- P. Bailet, M. Yoshinaga, Degeneration of Orlik-Solomon algebras and Milnor fibers of complex line arrangements. To appear in *Geometriae Dedicata*.
- [3] J. Bartz, S. Yuzvinsky, Multinets in ℙ². Bridging Algebra, Geometry, and Topology, Springer Proceedings in Mathematics & Statistics 96 (2014), 21-35. DOI: 10.1007/978-3-319-09186-0_3
- [4] D. C. Cohen, P. Orlik, Arrangements and local systems. Math. Res. Lett. 7 (2000), no. 2-3, 299-316.
- [5] D. C. Cohen, A. Suciu, On Milnor fibrations of arrangements. J. London Math. Soc. 51 (1995), no. 2, 105-119.
- [6] R. Cordovil, D. Forge, A note on Tutte polynomials and Orlik-Solomon algebras. European J. Combin. 24 (2003), no. 8, 1081-1087.
- [7] G. Denham, The Orlik-Solomon complex and Milnor fibre homology. Arrangements in Boston: a Conference on Hyperplane Arrangements (1999). Topology Appl. 118 (2002), no. 1-2, 45-63.
- [8] A. Dimca, D. Ibadula, A. D. Măcinic, Pencil type line arrangements of low degree: classification and monodromy. To appear in Ann. Scuola Norm. Sup. Pisa.
- [9] A. Dimca, S. Papadima, Finite Galois covers, cohomology jump loci, formality properties, and multinets. Ann. Scuola Norm. Sup. Pisa, 10 (2011), 253-268.
- [10] C. Dunn, M. S. Miller, M. Wakefield, S. Zwicknagl, Equivalence classes of Latin squares and nets in CP². Ann. Fac. Sci. Toulouse Math. (6) 23 (2014), no. 2, 335-351.
- [11] M. Falk, Arrangements and cohomology. Ann. Comb. 1 (1997), no. 2, 135-157.
- [12] B. Grünbaum, A catalogue of simplicial arrangements in the real projective plane. Ars Math. Contemp. 2 (2009), no. 1, 1-25.
- [13] M. Falk, S. Yuzvinsky, Multinets, resonance varieties, and pencils of plane curves. Compos. Math. 143 (2007), no. 4, 1069-1088.
- [14] K. Ito, M. Yoshinaga, Semi-algebraic partition and basis of Borel-Moore homology of hyperplane arrangements. Proc. Amer. Math. Soc. 140 (2012), no. 6, 2065-2074.
- [15] A. Libgober, On combinatorial invariance of the cohomology of the Milnor fiber of arrangements and the Catalan equation over function fields. Arrangements of hyperplanes-Sapporo 2009, 175-187, Adv. Stud. Pure Math., 62, Math. Soc. Japan, Tokyo, 2012.
- [16] A. Libgober, S. Yuzvinsky, Cohomology of the Orlik-Solomon algebras and local systems. Compositio Math. 121 (2000), no. 3, 337-361.
- [17] A. D. Măcinic, S. Papadima, On the monodromy action on Milnor fibers of graphic arrangements. *Topology Appl.* 156 (2009), no. 4, 761-774.
- [18] P. Orlik, Peter, H. Terao, Arrangements of hyperplanes. Grundlehren der Mathematischen Wissenschaften, 300. Springer-Verlag, Berlin, 1992. xviii+325 pp.
- [19] P. Orlik, H. Terao, Arrangements and hypergeometric integrals. MSJ Memoirs, 9. Mathematical Society of Japan, Tokyo, 2001. x+112 pp.
- [20] S. Papadima, A. Suciu, The spectral sequence of an equivariant chain complex and homology with local coefficients. Trans. A. M. S., 362 (2010), no. 5, 2685-2721.
- [21] S. Papadima, A. Suciu, The Milnor fibration of a hyperplane arrangement: from modular resonance to algebraic monodromy. arXiv: 1401.0868
- [22] J. V. Pereira, S. Yuzvinsky, Completely reducible hypersurfaces in a pencil. Adv. Math. 219 (2008), no. 2, 672-688.
- [23] A. Suciu, Hyperplane arrangements and Milnor fibrations, Ann. Fac. Sci. Toulouse Math. (6) 23 (2014), no. 2, 417-481.
- [24] K. Williams, The homology groups of the Milnor fiber associated to a central arrangement of hyperplanes in C³. Topology Appl. 160 (2013), no. 10, 1129-1143.
- [25] M. Yoshinaga, Hyperplane arrangements and Lefschetz's hyperplane section theorem. Kodai Math. J. 30, no. 2 (2007), 157–194. DOI: 10.2996/kmj/1183475510
- [26] M. Yoshinaga, Minimality of hyperplane arrangements and basis of local system cohomology. Singularities in geometry and topology, 345-362, IRMA Lect. Math. Theor. Phys., 20, Eur. Math. Soc., Zürich, 2012.
- [27] M. Yoshinaga, The chamber basis of the Orlik-Solomon algebra and Aomoto complex. Ark. Mat. 47 (2009), no. 2, 393-407.
- [28] M. Yoshinaga, Minimal stratifications for line arrangements and positive homogeneous presentations for fundamental groups. *Configuration Spaces: Geometry, Combinatorics and Topology*, 503-533, CRM Series, 14, Ed. Norm., Pisa, 2012.

4-NETS

- [29] M. Yoshinaga, Milnor fibers of real line arrangements. Journal of Singularities, 7 (2013), 220-237.
- [30] M. Yoshinaga, Resonant bands and local system cohomology groups for real line arrangements. Vietnam Journal of Mathematics, 42, 3, (2014) 377-392.
- [31] S. Yuzvinsky, Orlik-Solomon algebras in algebra and topology. Russian Math. Surveys 56 (2001), no. 2, 293-364.
- [32] S. Yuzvinsky, Realization of finite Abelian groups by nets in P². Compositio Math. 140 (2004), 1614-1624. DOI: 10.1112/S0010437X04000600
- [33] S. Yuzvinsky, A new bound on the number of special fibers in a pencil of curves. Proc. Amer. Math. Soc. 137 (2009), no. 5, 1641-1648.
- [34] S. Yuzvinsky, Cohomology of the Brieskorn-Orlik-Solomon algebras. Comm. Alg. 23 (1998), no. 14, 5339-5354.

MICHELE TORIELLI, DEPARTMENT OF MATHEMATICS "GIUSEPPE PEANO", UNIVERSITY OF TURIN, VIA CARLO ALBERTO 10, 10123, TURIN, ITALY

E-mail address: michele.torielli@unito.it

Masahiko Yoshinaga, Department of Mathematics, Hokkaido University, Kita 10, Nishi 8, Kita-Ku, Sapporo 060-0810, Japan

E-mail address: yoshinaga@math.sci.hokudai.ac.jp

ON THE ŁOJASIEWICZ EXPONENTS OF QUASI-HOMOGENEOUS FUNCTIONS

ALAIN HARAUX AND TIEN SON PHAM

ABSTRACT. Quasi-homogeneous functions, and especially polynomials, enjoy some specific properties around the origin which allow to estimate the so-called Lojasiewicz exponents in a way quite similar to homogeneous functions. In particular we generalize a previous result for polynomials of two variables concerning the optimal Lojasiewicz gradient inequality at the origin.

1. INTRODUCTION

In his pioneering papers [13], [14], Lojasiewicz established that any analytic function f of n real variables satisfies an inequality of the form

(1.1)
$$\|\nabla f(x)\| \ge c|f(x) - f(a)|^{\beta}$$

for ||x - a|| small enough with c > 0, $\beta \in (0, 1)$. This inequality, known as the Lojasiewicz gradient inequality, is useful to establish trend to equilibrium of the general solutions of gradient systems

(1.2)
$$u' + \nabla f(u) = 0$$

and can also be used to evaluate the rate of convergence. It is therefore of interest to know as precisely as possible the connection between f and its gradient and in particular to determine the best(smallest) possible value of β in (1.1) when a is a critical point of f. This value is called the Lojasiewicz exponent at a. In [7] for instance, it was shown that if f is a homogeneous polynomial with degree $d \geq 2$, the Lojasiewicz exponent at the origin is exactly $1 - \frac{1}{d}$ when n = 2. This property is no longer true if n > 2.

On the other hand, Gwoździewicz [6] (see also [16]) considered the case of a real analytic function at an isolated zero and also found, in this case, an interesting relationship between various Lojasiewicz exponents, relative to different Lojasiewicz inequalities. In addition the case of general polynomials has been thoroughly investigated by D'Acunto and Kurdyka in [2] (see also [11, 12, 17]).

Our paper is concerned to the extension of the result from [7] and several estimates of Lojasiewicz exponents at the origin when f is a quasi-homogeneous map (see, for example, [1]). It is divided in 5 sections. In Section 2, we state and prove some preliminary results, mainly concerning the local behavior of quasi-homogeneous maps near the origin. Section 3 contains more information in the specific case where the origin is an isolated zero of f. Section 4 deals with the Lojasiewicz gradient exponent of quasi-homogeneous polynomials, in particular we generalize the main result of [7] (see also [4, 5, 9] for related results). Section 5 contains

²⁰¹⁰ Mathematics Subject Classification. 14P10, 14P15, 32S05, 34A26.

Key words and phrases. quasi-homogeneous, gradient inequality, Lojasiewicz exponents.

The second author was partially supported by Vietnam National Foundation for Science and Technology Development (NAFOSTED) grant number 101.04-2013.07.

more precise estimates for quasi-homogeneous polynomials of two variables. These results are illustrated by typical examples and completed by a few remarks.

2. Definitions and preliminary results

We first recall the concept of a quasi-homogeneous map. Let $\mathbb{K} = \mathbb{C}$ or \mathbb{R} . We say that $f: \mathbb{K}^n \to \mathbb{K}^k$ is a (positively) quasi-homogeneous map with weight

$$w := (w_1, w_2, \dots, w_n) \in (\mathbb{R}_+ - \{0\})^n$$

and quasi-degree $d := (d_1, d_2, ..., d_k) \in (\mathbb{R}_+ - \{0\})^k$ if

(2.1)
$$f_i(t^{w_1}x_1, t^{w_2}x_2, \dots, t^{w_n}x_n) = t^{d_i}f_i(x_1, x_2, \dots, x_n)$$

for each i = 1, 2, ..., k, and all t > 0. Note that if $w_j = 1$ for j = 1, 2, ..., n then the above definition means that the components f_i are homogeneous functions of degree d_i . When k = 1, a scalar function $f = f_1$ satisfying (2.1) is called a (positively) quasi-homogeneous function with weight $w = (w_1, w_2, ..., w_n)$ and quasi-degree $d = d_1$. In the sequel we shall drop for simplicity the word "positively". Note that any monomial $x^{\alpha} := x_1^{\alpha_1} x_2^{\alpha_2} \dots x_n^{\alpha_n}$ is a quasihomogeneous function with arbitrary weight $w = (w_1, w_2, \dots, w_n) \in (\mathbb{N} - \{0\})^n$ and quasidegree $\langle w, \alpha \rangle := w_1 \alpha_1 + w_2 \alpha_2 + \dots + w_n \alpha_n$. Moreover, we have

Proposition 2.1. Let $f: \mathbb{K}^n \to \mathbb{K}$ be a polynomial function. Then f is quasi-homogeneous with weight $w \in (\mathbb{N} - \{0\})^n$ and quasi-degree $m \in \mathbb{N} - \{0\}$ if and only if all its constitutive monomials are quasi-homogeneous functions with weight w and quasi-degree m.

Proof. Suppose that f is a quasi-homogeneous polynomial with weight w and quasi-degree m. We have the following finite expansion

$$f(x) := \sum_{\alpha} a_{\alpha} x^{\alpha}.$$

Then

$$\sum_{\alpha} a_{\alpha} t^{\langle w, \alpha \rangle} x^{\alpha} = t^m \sum_{\alpha} a_{\alpha} x^{\alpha}.$$

This gives

$$\sum_{\alpha} a_{\alpha} [t^{\langle w, \alpha \rangle - m} - 1] x^{\alpha} = 0$$

for all $x \in \mathbb{K}^n$ and for all t > 0. Since the field \mathbb{K} is of characteristic 0, we get

$$\langle w, \alpha \rangle - m = 0$$

for all α provided $a_{\alpha} \neq 0$. In other words, all constitutive monomials of f are quasi-homogeneous functions with weight w and quasi-degree m. The converse is clear.

For a fixed weight $w := (w_1, w_2, \dots, w_n) \in (\mathbb{R}_+ - \{0\})^n$ we set

$$||x||_w := \max_{j=1,2,\dots,n} |x_j|^{\frac{1}{w_j}}$$

for $x := (x_1, x_2, ..., x_n) \in \mathbb{K}^n$. In the special case $w = w_0 = (1, ...1)$, we recover the usual l^{∞} norm on \mathbb{K}^n and we set

$$||x|| := ||x||_{w_0} = \max_{j=1,2,\dots,n} |x_j|.$$

Remark 2.2. (i) It is worth noting that, except in the special case $w = w_0 = (1, ...1), \|\cdot\|_w$ is not a norm since it is not homogeneous of degree 1.

(ii) It is easy to see that $\|\cdot\|_w$ is a quasi-homogeneous function with weight w and with quasi-degree 1.

The following basic properties will be used throughout the text.

Proposition 2.3. Let

$$w_* := \min_{j=1,2,\dots,n} w_j,$$

 $w^* := \max_{j=1,2,\dots,n} w_j.$

Then the following hold

(i) For all $||x|| \ge 1$ we have

$$||x||^{\frac{1}{w_*}} \ge ||x||_w \ge ||x||^{\frac{1}{w^*}}.$$

In particular, $||x|| \to \infty$ if and only if $||x||_w \to \infty$.

(ii) For all $||x|| \leq 1$ we have

$$\|x\|^{\frac{1}{w_*}} \le \|x\|_w \le \|x\|^{\frac{1}{w^*}},$$

In particular, $||x|| \to 0$ if and only if $||x||_w \to 0$.

t

Proof. The proof of the proposition is straightforward from the definitions.

In the sequel for t > 0, for any $w := (w_1, w_2, \dots, w_n) \in (\mathbb{R}_+ - \{0\})^n$ and $x := (x_1, x_2, \dots, x_n) \in \mathbb{K}^n$

we denote

•
$$x := (t^{w_1}x_1, t^{w_2}x_2, \dots, t^{w_n}x_n),$$

and for $d := (d_1, d_2, ..., d_k) \in (\mathbb{R}_+ - \{0\})^k$ we set

$$d_* := \min_{i=1,2,...,k} d_i, d^* := \max_{i=1,2,...,k} d_i.$$

Let $f: \mathbb{K}^n \to \mathbb{K}^k$ be quasi-homogeneous with weight $w := (w_1, w_2, \ldots, w_n) \in (\mathbb{R}_+ - \{0\})^n$ and quasi-degree $d := (d_1, d_2, \ldots, d_k) \in (\mathbb{R}_+ - \{0\})^k$. If $f_i \equiv 0$ for some $i \in \{1, 2, \ldots, k\}$, then d_i can be replaced by any positive number. In the sequel we shall assume

$$(2.2) \qquad \forall i \in \{1, 2, \dots, k\}, \quad f_i \neq 0.$$

It is easy to check that in this case d_i is uniquely defined by (2.1) for all $i \in \{1, 2, ..., k\}$. Then d_* and d^* are well defined.

The next two results summarize some important consequences of the quasi homogeneity property.

Proposition 2.4. Let $f := (f_1, f_2, \ldots, f_k) \colon \mathbb{K}^n \to \mathbb{K}^k$ be a quasi-homogeneous map with weight $w := (w_1, w_2, \ldots, w_n)$ and quasi-degree $d := (d_1, d_2, \ldots, d_k)$ satisfying (2.2). Then the following properties are equivalent.

- (i) The origin is an isolated zero of f.
- (ii) $f^{-1}(0) = \{0\}.$

Proof. It is clear that (ii) implies (i). Conversely if (ii) is not satisfied, let $a \neq 0$ be such that f(a) = 0. This implies that $f_1(a) = f_2(a) = \cdots = f_k(a) = 0$. Hence

$$f_i(t \bullet a) = t^{d_i} f_i(a) = 0$$

for i = 1, 2, ..., k and all t > 0. Note that $||t \bullet a|| \to 0$ as $t \to 0$. Thus, the origin is not an isolated zero of f, which is a contradiction.

Proposition 2.5. Let $f : \mathbb{R}^n \to \mathbb{R}$ be a continuous quasi-homogeneous function with weight w and quasi-degree m. Suppose that $n \ge 2$. Then the following conditions are equivalent

- (i) $f^{-1}(0) = \{0\}.$
- (ii) f has a strict global extremum at the origin.
- (iii) For each $\epsilon \ge 0$, $\{x \in \mathbb{R}^n \mid |f(x)| = \epsilon\}$ is a non-empty compact set.
- (iv) $\min_{\|x\|=1} |f(x)| > 0.$

Proof. (i) \Rightarrow (ii) If $f^{-1}(0) = \{0\}$, then by connectedness f has a constant sign (for instance f > 0) on the unit Euclidian sphere $\mathbb{S}^{n-1} := \{x \in \mathbb{R}^n \mid ||x|| = 1\}$ of dimension (n-1) if n > 1. But for any $x \neq 0$, there is clearly t > 0 such that $y := t \bullet x$ is in \mathbb{S}^{n-1} . Indeed the euclidian norm of $t \bullet x$ is 0 for t = 0, tends to infinity with t and is a continuous function of t, hence it must take the value 1 for some finite positive t. Then $f(x) = f(t^{-1} \bullet y) = t^{-m}f(y) > 0$, which proves (ii).

(ii) \Rightarrow (iv) Suppose, by contradiction, that $\min_{\|x\|=1} |f(x)| = 0$. Then there exists a point $a \in \mathbb{R}^n$ such that $\|a\| = 1$ and f(a) = 0. This implies that $f(t \bullet a) = 0$ for all t > 0, which contradicts (ii).

(iv) \Rightarrow (iii) By contradiction, assume that the set $\{x \in \mathbb{R}^n \mid |f(x)| = \epsilon\}$ is not compact for some $\epsilon \geq 0$. This means that there exists a sequence $x^p \in \mathbb{R}^n, p \in \mathbb{N}$, such that $||x^p|| \to \infty$ as $p \to \infty$ and $|f(x^p)| = \epsilon$. Let $t_p := \frac{1}{||x^p||_w} \to 0$. Then the sequence $|f(t_p \bullet x^p)| = t_p^m \epsilon$ tends to zero as $p \to \infty$. From the sequence of points $t_p \bullet x^p$ lying on the compact set $\{||x||_w = 1\}$ one can choose a subsequence convergent to some $a, ||a||_w = 1$. Clearly, f(a) = 0 and $a \neq 0$, which contradicts (iv).

(iii) \Rightarrow (i) If f(a) = 0 for some $a \neq 0$, then $f(t \bullet a) = 0$ for all t > 0. Consequently, by letting t tend to infinity, we obtain that the set $\{x \in \mathbb{R}^n \mid |f(x)| = 0\}$ is not compact, which contradicts (iii).

3. The Lojasiewicz inequality for a quasi-homogeneous map which vanishes only at the origin

In this section we are interested in the first Lojasiewicz inequality which relates in general the size of f(u) and the distance of u to the set $f^{-1}(0)$. However we essentially restrict our study to the case where this set is reduced to 0.

Proposition 3.1. Let $f := (f_1, f_2, \ldots, f_k) \colon \mathbb{K}^n \to \mathbb{K}^k$ be a continuous quasi homogeneous map with weight $w := (w_1, w_2, \ldots, w_n)$ and quasi-degree $d := (d_1, d_2, \ldots, d_k)$. Then the following statements hold.

(i) There exists a positive constant c_1 such that

 $||f(x)|| \le c_1 ||x||_w^{d_*}, \quad as \quad ||x|| \le 1.$

(ii) If $f^{-1}(0) = \{0\}$, then there exists a positive constant c_2 such that

$$c_2 \|x\|_w^{d^*} \le \|f(x)\|, \quad as \quad \|x\| \le 1.$$

Proof. Consider the family of topological closed spheres

$$S_t := \{ (x_1, x_2, \dots, x_n) \in \mathbb{K}^n \mid ||x||_w = t \}.$$

Then, by Proposition 2.3(i), for each t > 0 we have that S_t is a compact set. Let $x \in \mathbb{K}^n$ be such that $x \neq 0$ and $||x|| \leq 1$. By Proposition 2.3(ii), we have $t := \frac{1}{||x||_w} \geq 1$. Note that $t \bullet x \in S_1$. (i) Let

$$c_1 := \max_{i=1,2,\dots,k} \max_{y \in S_1} |f_i(y)|.$$

We have

$$c_1 \ge |f_i(t \bullet x)| = |t^{d_i} f_i(x)|$$
 for $i = 1, 2, \dots, k$.

It follows that

 $c_1 \|x\|_w^{d_i} \ge |f_i(x)|$ for $i = 1, 2, \dots, k$.

Consequently,

$$c_1 \|x\|_w^{d_*} = c_1 \max_{i=1,2,\dots,k} \|x\|_w^{d_i} \ge \|f(x)\|.$$

which proves (i).

(ii) Let $c_2 := \min_{y \in S_1} ||f(y)|| > 0$. By definition, we have

$$c_{2} \leq \|f(t \bullet x)\| = \max_{i=1,2,\dots,k} |t^{d_{i}}f_{i}(x)| \leq \max_{i=1,2,\dots,k} |t^{d_{i}}| \max_{i=1,2,\dots,k} |f_{i}(x)|$$

$$\leq |t^{d^{*}}|\|f(x)\| = \frac{1}{\|x\|_{w}^{d^{*}}} \|f(x)\|,$$

which proves (ii). The proposition is proved.

Theorem 3.2. Let $f := f^0 + f^1 + \cdots + f^l : \mathbb{K}^n \to \mathbb{K}^k$, where f^0, f^1, \ldots, f^l are continuous quasihomogeneous maps with weight $w = (w_1, w_2, \ldots, w_n)$ and quasi-degrees d^0, d^1, \ldots, d^l respectively such that

$$(d^0)^* < (d^1)_* \le (d^2)_* \le \dots \le (d^l)_*$$

If the origin is an isolated zero of f^0 then there exists a positive constant c such that

$$c \|x\|_w^{(d^0)^*} \le \|f(x)\|, \quad as \quad \|x\| \ll 1.$$

Proof. By Proposition 3.1(ii), there exists a positive constant c_0 such that

$$|x_0||x||_w^{(d^0)^*} \le ||f^0(x)||, \text{ as } ||x|| \ll 1.$$

On the other hand, from Proposition 3.1(i), there exist positive constants c_1, c_2, \ldots, c_l such that for $i = 1, 2, \ldots, l$,

$$||f^{i}(x)|| \le c_{i}||x||_{w}^{(d^{i})_{*}}, \text{ as } ||x|| \ll 1.$$

We have for $||x|| \ll 1$ the next estimate

$$\begin{aligned} \|f^{1}(x) + f^{2}(x) + \dots + f^{l}(x)\| &\leq \|f^{1}(x)\| + \|f^{2}(x)\| + \dots + \|f^{l}(x)\| \\ &\leq c_{1}\|x\|_{w}^{(d^{1})_{*}} + c_{2}\|x\|_{w}^{(d^{2})_{*}} + \dots + c_{l}\|x\|_{w}^{(d^{l})_{*}}. \end{aligned}$$

Thus it follows from $(d^0)^* < (d^1)_* \le (d^2)_* \le \cdots \le (d^l)_*$ that

$$||f^1(x) + f^2(x) + \dots + f^l(x)|| \ll ||x||_w^{(d^0)^*}, \text{ as } ||x|| \ll 1.$$

Therefore, we have for $||x|| \ll 1$ the next inequality

$$\begin{aligned} \|f(x)\| &\geq \|f^{0}(x)\| - \|f^{1}(x) + f^{2}(x) + \dots + f^{l}(x)\| \\ &\geq c_{0}\|x\|_{w}^{(d^{0})^{*}} - c'\|x\|_{w}^{(d^{0})^{*}} \quad (0 < c' \ll c_{0}). \end{aligned}$$

This gives

$$||f(x)|| \ge (c_0 - c') ||x||_w^{(d^0)^*}, \text{ as } ||x|| \ll 1$$

which proves the theorem.

The following is a direct consequence from Proposition 3.1 and Theorem 3.2:

Corollary 3.3. Under the hypothesis of Theorem 3.2, there exists a positive constant c such that

$$c||x||^{\frac{(d^0)^*}{w_*}} \le ||f(x)||, \quad as \quad ||x|| \ll 1.$$

We define the Lojasiewicz exponent $\alpha_0(f)$ of the map f at the origin $0 \in \mathbb{K}^n$ as the infimum of the set of all real numbers l > 0 which satisfy the condition: there exists a positive constant c such that

$$c||x||^{l} \le ||f(x)||, \text{ as } ||x|| \ll 1.$$

If the set of all the exponents is empty we put $\alpha_0(f) := +\infty$.

Corollary 3.4. Let $f := (f_1, f_2, \ldots, f_k) \colon \mathbb{K}^n \to \mathbb{K}^k$ be a continuous quasi-homogeneous map with weight $w := (w_1, w_2, \ldots, w_n)$ and quasi-degree $d := (d_1, d_2, \ldots, d_k)$. Suppose that $f^{-1}(0) = \{0\}$. Then

$$\frac{d_{*}}{v_{*}} \leq \alpha_{0}(f) \leq \frac{d^{*}}{w_{*}}$$

Proof. It follows from Corollary 3.3 that

$$\alpha_0(f) \leq \frac{d^*}{w_*}.$$

In order to prove the left inequality, let i, j be such that $d_i = d_*$ and $w_j = w_*$. Take $a \in \mathbb{K}^n$ with the property that $a_j f_i(a) \neq 0$. Then, asymptotically as $t \to 0$, we have¹

$$\begin{aligned} \|f(t \bullet a)\| &\simeq t^{d_*}, \\ \|t \bullet a\| &\simeq t^{w_*}. \end{aligned}$$

Consequently,

$$\|f(t \bullet a)\| \simeq \|t \bullet a\|^{\frac{d_*}{w_*}}.$$

By the definition of the Lojasiewicz exponent $\alpha_0(f)$, we find that

$$\frac{d_*}{w_*} \leq \alpha_0(f).$$

Example 3.5. (i) Let $f := (f_1 := x^2 + y^4, f_2 := (x^2 - y^4)^2) \colon \mathbb{R}^2 \to \mathbb{R}^2$. It is easy to check that f is a positive quasi-homogeneous map with weight w := (2, 1) and quasi-degree d := (4, 8). Moreover, $\alpha_0(f) = 4 \ (= \frac{d_*}{w_*})$.

Moreover, $\alpha_0(f) = 4 \ (= \frac{d_*}{w_*})$. (ii) Let $f := (f_1 := x^2 - y^4, f_2 := (x^2 + y^4)^2)$: $\mathbb{R}^2 \to \mathbb{R}^2$. Then f is a positive quasi-homogeneous map with weight w := (2, 1) and quasi-degree d := (4, 8). Moreover,

$$\alpha_0(f) = 8 \ (= \frac{d^*}{w_*}).$$

Corollary 3.6. Let $f: \mathbb{K}^n \to \mathbb{K}$ be a continuous quasi-homogeneous function with weight w and quasi-degree m. If $f^{-1}(0) = \{0\}$ then

$$\alpha_0(f) = \frac{m}{w_*}.$$

 \Box

¹Where $A \simeq B$ means that A/B lies between two positive constants.

Proof. The claim comes from $d^* = d_* = m$.

4. The Łojasiewicz gradient inequality for quasi-homogeneous polynomials

We now consider the case k = 1 and let $f \colon \mathbb{K}^n \to \mathbb{K}$ be a C^1 quasi-homogeneous function with weight $w := (w_1, w_2, \ldots, w_n)$ and quasi-degree m:

$$f(t^{w_1}x_1, t^{w_2}x_2, \dots, t^{w_n}x_n) = t^m f(x_1, x_2, \dots, x_n).$$

We define the Lojasiewicz gradient exponent $\beta_0(f)$ of the map f at the origin $0 \in \mathbb{K}^n$ as the infimum of the set of all real numbers l > 0 which satisfy the condition: there exists a positive constant c such that

$$|c|f(x)|^{l} \le ||\nabla f(x)||$$
 for $||x|| \ll 1$.

If the set of all the exponents is empty we put $\beta_0(f) := +\infty$. It is well-known (see [14]) that if f is analytic, then $\beta_0(f) < 1$.

We start with a general result valid for C^1 functions.

Theorem 4.1. Let $f : \mathbb{K}^n \to \mathbb{K}$ be a C^1 quasi-homogeneous function with weight w and quasidegree m. Then

$$\max\{0, 1 - \frac{w^*}{m}\} \le \beta_0(f) \le 1.$$

Proof. Since f is a C¹-positive quasi-homogeneous function with weight w and quasi-degree $m \ge w^*$,

$$mt^{m-1}f(x) = \sum_{j=1}^{n} w_j t^{w_j - 1} x_j \frac{\partial f}{\partial x_j} (t \bullet x).$$

In particular, we have the generalized Euler identity

(4.1)
$$mf(x) = \sum_{j=1}^{n} w_j x_j \frac{\partial f}{\partial x_j}(x).$$

As a consequence, there exists a positive constant c_1 such that

 $c_1|f(x)| \leq ||x|| ||\nabla f(x)||$ for all $||x|| \ll 1$.

This implies that $\beta_0(f) \leq 1$.

Next, let l > 0 and $c_2 > 0$ be such that

(4.2)
$$c_2 |f(x)|^l \leq ||\nabla f(x)|| \text{ for all } ||x|| \ll 1.$$

We have for all t > 0 the following relation

$$t^{w_j} \frac{\partial f}{\partial x_j}(t \bullet x) = t^m \frac{\partial f}{\partial x_j}(x) \text{ for } j = 1, 2, \dots, n.$$

This shows that

(4.3)
$$\frac{\partial f}{\partial x_j}(t \bullet x) = t^{m-w_j} \frac{\partial f}{\partial x_j}(x) \quad \text{for} \quad j = 1, 2, \dots, n.$$

Then we have for all t > 0 the following equations

$$f(t \bullet x) = t^m f(x),$$

$$\frac{\partial f}{\partial x_j}(t \bullet x) = t^{m-w_j} \frac{\partial f}{\partial x_j}(x) \text{ for } j = 1, 2, \dots, n.$$

Since $f \neq 0$ there is $a \in \mathbb{K}^n$ such that $0 \neq \nabla f^2(a) = 2f(a)\nabla f(a)$. This implies $f(a) \neq 0$ and $\nabla f(a) \neq 0$. Then, asymptotically as $t \to +0$, we have

$$\begin{aligned} |f(t \bullet a)| &\simeq t^m, \\ \|\nabla f(t \bullet a)\| &\simeq t^{m-w_j} \quad \text{for some} \quad j \in \{1, 2, \dots, n\}, \end{aligned}$$

Therefore (4.2) implies the existence of $c_3 > 0$ such that

$$c_3 t^{ml} \leq t^{m-w_j} \leq t^{m-w^*}$$

as $t \to +0$. This in turn implies that $ml \ge m - w^*$, which is equivalent to

$$l \ge 1 - \frac{w^*}{m}.$$

This, together with the definition of $\beta_0(f)$, implies the desired result.

In the special case where 0 is the only critical point of f we have a more precise estimation as follows.

Corollary 4.2. Let $f: \mathbb{K}^n \to \mathbb{K}$ be a C^1 quasi-homogeneous function with weight w and quasidegree $m \ge w^*$. Suppose that $\nabla f^{-1}(0) = \{0\}$. Then

$$1 - \frac{w^*}{m} \leq \beta_0(f) \leq 1 - \frac{w_*}{m}.$$

Proof. One has only to show that

$$\beta_0(f) \leq 1 - \frac{w_*}{m}$$

Indeed, by the generalized Euler identity (4.1), there exists a positive constant c_1 such that

$$|f(x)| \le c_1 \|\nabla f(x)\| \|x\|$$

On the other hand, it follows from (4.3) that the following

$$\nabla f(x) \colon \mathbb{K}^n \to \mathbb{K}^n, \quad x \mapsto \left(\frac{\partial f}{\partial x_1}, \frac{\partial f}{\partial x_2}, \dots, \frac{\partial f}{\partial x_n}\right),$$

is a continuous quasi-homogeneous map with weight w and quasi-degree

$$(m-w_1,m-w_2,\ldots,m-w_n)$$

Therefore, by Proposition 2.3 and then Proposition 3.1(ii), for all $||x|| \ll 1$ we have

$$\begin{aligned} |f(x)| &\leq c_1 \|\nabla f(x)\| \|x\|_w^{w_*} \\ &\leq c_2 \|\nabla f(x)\| \|\nabla f(x)\|^{\frac{w_*}{m-w_*}} = c_2 \|\nabla f(x)\|^{1+\frac{w_*}{m-w_*}}, \end{aligned}$$

for some $c_2 > 0$. Hence there exists a positive constant c such that

$$||x||^{1-\frac{w_*}{m}} \le ||\nabla f(x)|| \text{ for } ||x|| \ll 1.$$

Consequently, by the definition of the Lojasiewicz gradient exponent $\beta_0(f)$, we obtain

$$\beta_0(f) \leq 1 - \frac{w_*}{m},$$

which completes the proof.

Example 4.3. (i) (see [10]). Let $f: \mathbb{C}^2 \to \mathbb{C}$, $(x, y) \mapsto x^3 + 3xy^k$, $k \ge 3$, be a complex polynomial. It is clear that f is a quasi-homogeneous polynomial with weight w = (k, 2) and quasi-degree m = 3k. A direct computation shows that the origin in \mathbb{C}^2 is an isolated critical point of f. Moreover, it follows from the results in [10] and [18] that

$$\begin{aligned} \alpha_0(\nabla f) &= \frac{3k}{2} - 1, \\ \beta_0(f) &= \frac{\alpha_0(\nabla f)}{1 + \alpha_0(\nabla f)} = 1 - \frac{2}{3k}. \end{aligned}$$

(ii) Let $f: \mathbb{C}^2 \to \mathbb{C}, (x, y) \mapsto x^4 - 4xy$, be a complex polynomial. It is clear that f is a quasihomogeneous polynomial with weight w = (1, 3) and quasi-degree m = 4. A direct computation shows that the origin in \mathbb{C}^2 is an isolated critical point of f. Moreover, it follows from the results in [10] and [18] that

$$\begin{aligned} \alpha_0(\nabla f) &= 1, \\ \beta_0(f) &= \frac{\alpha_0(\nabla f)}{1 + \alpha_0(\nabla f)} = \frac{1}{2} \end{aligned}$$

Remark 4.4. Let $f: \mathbb{C}^n \to \mathbb{C}$ be a complex polynomial function with an isolated singularity at 0. Then from the works of Teissier [18, Corollary 2] we have the following equation

$$\beta_0(f) = \frac{\alpha_0(\nabla f)}{1 + \alpha_0(\nabla f)}.$$

Moreover, Gwoździewicz has remarked, [6], that the above relation fails to hold for some real polynomial functions with an isolated singularity at 0. However, we have the following.

Corollary 4.5. (see also [6, Theorem 1.3]) Let $f : \mathbb{R}^n \to \mathbb{R}$ be a quasi-homogeneous polynomial function with weight w and quasi-degree $m \ge w^*$. If $f^{-1}(0) = \{0\}$, then

$$\begin{aligned} \alpha_0(f) &= \frac{m}{w_*}, \\ \alpha_0(\nabla f) &= \alpha_0(f) - 1 = \frac{m}{w_*} - 1, \\ \beta_0(f) &= \frac{\alpha_0(\nabla f)}{1 + \alpha_0(\nabla f)} = 1 - \frac{w_*}{m}. \end{aligned}$$

Proof. In fact, by Corollary 3.6, we have

$$\alpha_0(f) = \frac{m}{w_*}.$$

Then the remained relations follow from [6, Theorem 1.3]. We will give below a direct proof in order to keep our paper self-contained.

We first note that the origin is an isolated critical point of f. Indeed, if $\nabla f(a) = 0$ for some $a \neq 0$, then it follows easily from the generalized Euler identity (4.1) that f(a) = 0, which is a contradiction.

Without loss of generality, we may suppose that $f(x) \ge 0$ for all $x \in \mathbb{R}^n$ with equality if and only if x = 0. For each $\delta > 0$, the restriction of f to the sphere $\{x \in \mathbb{R}^n \mid ||x||_E = \delta\}$ attains its minimum at least one point, where $||x||_E := \sqrt{\sum_{i=1}^n x_i^2}$. Let

$$\Gamma := \{ u \in \mathbb{R}^n \mid f(u) = \min_{\|x\|_E = \|u\|_E} f(x) \}.$$

It follows from the Tarski-Seidenberg theorem (see, for example, [3, Theorem 2.3.4]), that Γ is semi-algebraic. Hence the Curve Selection Lemma [15] is applicable. Together with Lagrange's

Multipliers Theorem, this implies that there exists an analytic map

$$(\lambda,\varphi)\colon (-\epsilon,\epsilon)\to \mathbb{R}\times\mathbb{R}^n, \tau\mapsto (\lambda(\tau),\varphi(\tau)),$$

such that

- (i) $\varphi(\tau) = 0$ if and only if $\tau = 0$;
- (ii) $\varphi(\tau) \in \Gamma$ for all $\tau \in [0, \epsilon)$; and
- (iii) $\nabla f(\varphi(\tau)) = \lambda(\tau)\varphi(\tau)$ for all $\tau \in [0, \epsilon)$.

Let $at^p, a > 0$, be the leading term of the Taylor expansion of $\|\varphi(\tau)\|^2$, and $bt^q, b \neq 0$, be that of $|f(\varphi(\tau))|^2$. Then, asymptotically as $t \to 0$, we have

$$|f(\varphi(\tau))| \simeq \|\varphi(\tau)\|^{\frac{q}{p}}.$$

Consequently, by the definition of $\alpha_0(f)$, we get

$$\alpha_0(f) \ge \frac{q}{p}.$$

On the other hand, we may assume (taking $\epsilon > 0$ small enough if necessary) that the function $\tau \mapsto \|\varphi(\tau)\|$ is strictly increasing. Together with the condition (i), we find that for each $x \in \mathbb{R}^n, \|x\| \ll 1$, there exists a positive number $\tau \in [0, \epsilon)$ satisfying the relation $\|\varphi(\tau)\|_E = \|x\|_E$. Hence,

$$|f(x)| = f(x) \ge \min_{\|u\|_E = \|x\|_E} f(u) = f(\varphi(\tau)) \simeq \|\varphi(\tau)\|_{p}^{\frac{q}{p}} = \|x\|_{p}^{\frac{q}{p}}.$$

By the definition, thus

$$\alpha_0(f) \le \frac{q}{p}$$

Therefore,

$$\alpha_0(f) = \frac{q}{p}$$

Moreover, it follows from the generalized Euler identity (4.1) that

$$mf(\varphi(\tau)) = \sum_{j=1}^{n} w_j \varphi_j(\tau) \frac{\partial f}{\partial x_j}(\varphi(\tau)).$$

By the condition (iv), hence

$$|mf(\varphi(\tau))| = |\lambda(\tau)| \sum_{j=1}^{n} w_j [\varphi_j(\tau)]^2$$
$$= \frac{\|\nabla f(\varphi(\tau))\|}{\|\varphi(\tau)\|} \sum_{j=1}^{n} w_j [\varphi_j(\tau)]^2$$
$$\simeq \|\varphi(\tau)\| \|\nabla f(\varphi(\tau))\|.$$

In particular, we get

$$\|\nabla f(\varphi(\tau))\| \simeq \|\varphi(\tau)\|^{\frac{q}{p}-1} \simeq |f(\varphi(\tau))|^{1-\frac{p}{q}}.$$

By definitions, hence

$$\begin{aligned} \alpha_0(\nabla f) &\geq \quad \frac{q}{p} - 1 = \alpha_0(f) - 1 = \frac{m}{w_*} - 1, \\ \beta_0(f) &\geq \quad 1 - \frac{p}{q} = 1 - \frac{1}{\alpha_0(f)} = 1 - \frac{w_*}{m}. \end{aligned}$$

Then the corollary follows immediately from Corollaries 3.4 and 4.2.

The following result is of general interest but we shall only use it to prove Theorem 4.7 below.

Lemma 4.6. Let $f: \mathbb{K}^n \to \mathbb{K}$ be a C^1 -function. For each k positive integer, consider the function $\widetilde{f}: \mathbb{K}^n \to \mathbb{K}, x \mapsto [f(x)]^k$. Suppose that there exist c > 0 and $\theta \in (0, 1]$ such that

 $c|\widetilde{f}(x)|^{1-\theta} \leq \|\nabla \widetilde{f}(x)\| \quad for \quad \|x\| \ll 1.$

Then

$$\frac{c}{k}|f(x)|^{1-k\theta} \le \|\nabla f(x)\| \quad for \quad \|x\| \ll 1$$

Proof. We have

$$\nabla \tilde{f}(x) = k[f(x)]^{k-1} \nabla f(x).$$

Hence

$$c|[f(x)]^k|^{1-\theta} \le k|[f(x)]^{k-1}|||\nabla f(x)||$$

This implies

$$\frac{c}{k}|f(x)|^{1-k\theta} \le \|\nabla f(x)\|,$$

concluding the proof of the lemma.

The following is a generalization of [7, Theorem 2.1].

Theorem 4.7. Let $f: \mathbb{R}^2 \to \mathbb{R}$ be a quasi-homogeneous polynomial function with weight $w := (w_1, w_2)$ and quasi-degree m. Then there exists a positive constant c such that

(4.4) $c|f(x,y)|^{1-\frac{w_*}{m}} \le \|\nabla f(x,y)\|, \quad as \quad \|(x,y)\| \ll 1.$

Proof. Without loss of generality, we may suppose that

$$1 \le w_* = w_1 \le w_2.$$

There are two cases to be considered.

Case 1. *m* is divisible by w_1 ; i.e., $q := \frac{m}{w_1}$ is a positive integer number.

Consider the following function

$$g(x,y) := f(x,y^{\frac{w_2}{w_1}}).$$

Then, by Proposition 2.1, we can see that g is a homogeneous polynomial on $\mathbb{R} \times \mathbb{R}_+$ of degree $q = \frac{m}{w_1}$. Indeed we can write for some finite set $S \subset \mathbb{N} \times \mathbb{N}$:

$$f(x,y) = \sum_{\alpha:=(\alpha_1,\alpha_2)\in S} a_{\alpha} x^{\alpha_1} y^{\alpha_2}$$

with

$$w_1\alpha_1 + w_2\alpha_2 = m = qw_1.$$

Hence

$$\alpha_2 = (q - \alpha_1) \frac{w_1}{w_2};$$

and therefore

$$f(x,y) = \sum_{\alpha \in S} a_{\alpha} x^{\alpha_1} y^{(q-\alpha_1)\frac{w_1}{w_2}}$$

which provides

$$g(x,y) = \sum_{\alpha \in S} a_{\alpha} x^{\alpha_1} y^{q-\alpha_1}.$$

It now follows from [7, Theorem 2.1] that there exists a positive constant c such that

$$c|g(x,y)|^{1-\frac{w_1}{m}} \le \|\nabla g(x,y)\|$$
, as $\|(x,y)\| \ll 1$ and $y \ge 0$.

On the other hand, by the definition

$$\begin{array}{lcl} \frac{\partial g}{\partial x}(x,y) & = & \frac{\partial f}{\partial x}(x,y^{\frac{w_2}{w_1}}),\\ \frac{\partial g}{\partial y}(x,y) & = & \frac{w_2}{w_1}y^{\frac{w_2}{w_1}-1}\frac{\partial f}{\partial y}(x,y^{\frac{w_2}{w_1}}) \end{array}$$

Therefore, asymptotically as $(x, y) \rightarrow (0, 0)$ and $y \ge 0$,

$$\begin{aligned} c|f(x,y^{\frac{w_2}{w_1}})|^{1-\frac{w_1}{m}} &\leq & \left\| \left(\frac{\partial f}{\partial x}(x,y^{\frac{w_2}{w_1}}), \frac{w_2}{w_1}y^{\frac{w_2}{w_1}-1} \frac{\partial f}{\partial y}(x,y^{\frac{w_2}{w_1}}) \right) \right\|. \\ &\leq & \left\| \left(\frac{\partial f}{\partial x}(x,y^{\frac{w_2}{w_1}}), \frac{\partial f}{\partial y}(x,y^{\frac{w_2}{w_1}}) \right) \right\|, \end{aligned}$$

because $w_2 \ge w_1$. Let $u := y^{\frac{w_2}{w_1}} \ge 0$. Then

$$c|f(x,u)|^{1-\frac{w_1}{m}} \leq \left\| \left(\frac{\partial f}{\partial x}(x,u), \frac{\partial f}{\partial y}(x,u) \right) \right\|.$$

By an entirely analogous argument but replacing $g(x,y) = f(x,y^{\frac{w_2}{w_1}})$ by $f(x,-y^{\frac{w_2}{w_1}})$ we can show that the above inequality also holds for all $u \leq 0$. These prove the theorem in Case 1.

Case 2. m is not divisible by w_1 .

Let $\tilde{f}(x,y) := [f(x,y)]^{w_1}$. Then it is clear that $\tilde{f}(x,y)$ is a positive quasi-homogeneous polynomial with weight (w_1, w_2) and quasi-degree $\tilde{m} := mw_1$. Since $\frac{\tilde{m}}{w_1} = m$ is an integer number, by applying Case 1 for the polynomial \tilde{f} we get

$$\widetilde{c}|\widetilde{f}(x,y)|^{1-\frac{w_1}{m}} \le \|\nabla \widetilde{f}(x,y)\|, \text{ as } \|(x,y)\| \ll 1,$$

for some $\tilde{c} > 0$.

By Lemma 4.6, we get

$$\frac{\widetilde{c}}{w_1}|f(x,y)|^{1-\frac{w_1}{m}} \le \|\nabla f(x,y)\|,$$

which completes the proof of the theorem.

Remark 4.8. As we see in the next proposition, the result of Theorem 4.7 is no longer valid in dimensions n > 2.

Proposition 4.9. (Compare with [7, Remark 2.4]) Let $f: \mathbb{R}^3 \to \mathbb{R}$ be given by

$$f(x, y, z) := x^4 + x^2 z^2 - 2xy^2 z + y^4 = x^4 + (xz - y^2)^2.$$

Then there exists a curve $\varphi \colon [0, \epsilon) \to \mathbb{R}^n, t \mapsto \varphi(t)$, such that

$$\|\nabla f[\varphi(t)]\| \ll |f[\varphi(t)]|^{1-\frac{1}{4}} \quad for \quad 0 < t \ll 1.$$

In particular, $\beta_0(f) > 1 - \frac{w_*}{m} = 1 - \frac{1}{4}$.

Proof. It is clear that f is a weighted quasi-homogeneous polynomial with weight w = (1, 1, 1)and quasi-degree m = 4. Moreover, f has non-isolated zero at the origin; namely,

$$f^{-1}(0) = \{(0,0,t) \mid t \in \mathbb{R}\}\$$

Define the polynomial curve $\varphi \colon [0,\epsilon) \to \mathbb{R}^3, t \mapsto (x(t),y(t),z(t)),$ by

$$\begin{array}{rcl} x(t) & := & t^2, \\ y(t) & := & t + t^5, \\ z(t) & := & 1. \end{array}$$

One easily verifies that

$$\begin{aligned} f[\varphi(t)] &= t^8 + 4t^{12} + 4t^{16} + t^{20}, \\ \frac{\partial f}{\partial x}[\varphi(t)] &= -2t^{10}, \\ \frac{\partial f}{\partial y}[\varphi(t)] &= 8t^7 + 12t^{11} + 4t^{15}, \\ \frac{\partial f}{\partial z}[\varphi(t)] &= -4t^8 - 2t^{12}. \end{aligned}$$

Hence, asymptotically as $t \to 0$,

$$\|\nabla f[\varphi(t)]\| \simeq t^7 \ll t^6 \simeq |f[\varphi(t)]|^{1-\frac{1}{4}}.$$

This completes the proof.

Remark 4.10. The polynomial $x^4 + x^2z^2 - 2xy^2z + y^4$ in the above proposition is a homogenization of $x^4 + (x - y^2)^2$ by the new variable z. The last one is a polynomial in the class of polynomials which was considered by János Kollár ([8]).

The following is a direct consequence of Theorem 4.1 and Theorem 4.7.

Corollary 4.11. Let $f: \mathbb{R}^2 \to \mathbb{R}$ be a quasi-homogeneous polynomial function with weight $w := (w_1, w_2)$ and quasi-degree m. Then

$$\max\{0, 1 - \frac{w^*}{m}\} \le \beta_0(f) \le 1 - \frac{w_*}{m}$$

5. Additional results, remarks and examples in dimension 2

In this section we will denote by $f: \mathbb{R}^2 \to \mathbb{R}$ a quasi-homogeneous polynomial with weight $w = (w_1, w_2)$ and quasi-degree m such that

$$w_* = w_1 \le w_2 = w^* \le m.$$

We now apply Corollary 4.11 in special cases.

Corollary 5.1. If the origin is an isolated zero of f_y then

$$\beta_0(f) = 1 - \frac{w_2}{m}$$

Proof. It is well known that f_y is quasi-homogeneous polynomial with weight $w = (w_1, w_2)$ and quasi-degree $m - w_2$. Since $f_y^{-1}(0) = \{0\}$, it follows from Proposition 2.5 that the polynomial f_y has a strict global extremum at the origin. Thus we can assume that $f_y > 0$ on $\mathbb{R}^n - \{0\}$. By again Proposition 2.5, the set $\{f_y = 1\}$ is nonempty compact. Hence

$$\infty > c := \max_{f_y(u,v)=1} |f(u,v)| > 0.$$

Take any $(x,y) \in \mathbb{R}^2$. Let $\epsilon := [f_y(x,y)]^{\frac{1}{m-w_2}}$. Then $\{(u,v) \in \mathbb{R}^2 \mid f_y(u,v) = \epsilon^{m-w_2}\}$ is a non-empty compact set. Moreover,

$$\begin{aligned} |f(x,y)| &\leq \max_{f_y(u,v)=\epsilon^{m-w_2}} |f(u,v)| &= \max_{f_y(\epsilon^{-w_1}u,\epsilon^{-w_2}v)=1} |f(u,v)| \\ &= \max_{f_y(\tilde{u},\tilde{v})=1} |f(\epsilon^{w_1}\tilde{u},\epsilon^{w_2}\tilde{v})| \\ &= \max_{f_y(\tilde{u},\tilde{v})=1} |f(\tilde{u},\tilde{v})|\epsilon^m \\ &= c[f_y(x,y)]^{\frac{m}{m-w_2}}. \end{aligned}$$

64

This gives

$$f_y(x,y) \ge c' |f(x,y)|^{1-\frac{w_2}{m}},$$

here $c' := c^{1-\frac{w_2}{m}} > 0$. Therefore

$$\|\nabla f(x,y)\| \ge |f_y(x,y)| \ge c' |f(x,y)|^{1-\frac{w_2}{m}}.$$

By the definition of $\beta_0(f)$, we get

$$\beta_0(f) \le 1 - \frac{w_2}{m}.$$

Then, by Corollary 4.11, $\beta_0(f) = 1 - \frac{w_2}{m}$.

Example 5.2. Let $f: \mathbb{R}^2 \to \mathbb{R}, (x, y) \mapsto y^3 + 3x^4y + 2x^6$ be a real polynomial. It is clear that f is a quasi-homogeneous polynomial with weight w = (1, 2) and quasi-degree m = 6. A direct computation shows that $f_y^{-1}(0) = \{(0, 0)\}$. Hence $\beta_0(f) = 1 - \frac{2}{6} = \frac{2}{3}$.

Corollary 5.3. Suppose that the origin is not an isolated zero of f_y . If there exists $(a,b) \in \mathbb{R}^2$ such that $f(a,b) \neq 0$, $f_x(a,b) \neq 0$ and $f_y(a,b) = 0$, then

$$\beta_0(f) = 1 - \frac{w_1}{m}.$$

Proof. By the hypothesis, we have for all t > 0

$$\begin{array}{rcl} f(t^{w_1}a, t^{w_2}b) &\simeq t^m, \\ f_x(t^{w_1}a, t^{w_2}b) &\simeq t^{m-w_1}, \\ f_y(t^{w_1}a, t^{w_2}b) &\equiv 0. \end{array}$$

Asymptotically as $t \to 0$, hence

$$|f(t^{w_1}a, t^{w_2}b)| \simeq t^m, \|\nabla f(t^{w_1}a, t^{w_2}b)\| \simeq t^{m-w_1}.$$

This implies that

$$\|\nabla f(t^{w_1}a, t^{w_2}b)\| \simeq \|f(t^{w_1}a, t^{w_2}b)\|^{1-\frac{w_1}{m}}$$

Then, by the definition of $\beta_0(f)$,

$$\beta_0(f) \ge 1 - \frac{w_1}{m}.$$

On the other hand, by Corollary 4.11, $\beta_0(f) \leq 1 - \frac{w_1}{m}$. Therefore $\beta_0(f) = 1 - \frac{w_1}{m}$.

Example 5.4. Let $f: \mathbb{R}^2 \to \mathbb{R}, (x, y) \mapsto x^2y - y^2$ be a real polynomial. It is clear that f is a quasi-homogeneous polynomial with weight w = (1, 2) and quasi-degree m = 4. A direct computation shows that $f_y^{-1}(0) = \{x^2 - 2y = 0\}$ and the origin in \mathbb{R}^2 is an isolated critical point of f. Moreover, it is easy to see that the conditions of Corollary 5.3 are satisfied. Hence, $\beta_0(f) = 1 - \frac{1}{4} = \frac{3}{4}$.

Remark 5.5. (i) All results in this paper allow to compute the Lojasiewicz exponents for some functions which are not quasi-homogeneous, for instance, the function f(x) := P(Ax), where P is a quasi-homogeneous polynomial of two variables and A a nonsingular 2×2 square matrix. As an example the polynomial $P(x, y) := ax^4 + by^2 + cx^2y$ is quasi-homogeneous with weight (1, 2) and quasi-degree 4. The polynomial Q(x, y) = P(x, x + y) is not quasi-homogeneous if $bc \neq 0$.

(ii) On the other hand, there are of course polynomials of two variables which cannot be put in the form = P(Ax) with, P, A as above. For instance the polynomial $Q(x, y) := x^2(1+y)$ is such that no polynomial $P = Q \circ A$ with A a nonsingular 2×2 square matrix is quasi-homogeneous.

65

ALAIN HARAUX AND TIEN SON PHAM

References

- V. I. Arnold, A. N. Varchenko, and S. M. Gusein-Zade, Singularities of differentiable maps, Birkhauser, 1985. DOI: 10.1007/978-1-4612-5154-5
- D. D'Acunto, K. Kurdyka, Explicit bounds for the Lojasiewicz exponent in the gradient inequality for polynomials, Ann. Pol. Math., Vol. 87 (2005), 51-61. DOI: 10.4064/ap87-0-5
- [3] Benedetti R. and J. J. Risler, Real algebraic and semi-algebraic sets, Hermann, 1990.
- [4] C. Bivia-Ausina, and S. Encinas, The Lojasiewicz exponent of a set of weighted homogeneous ideals, J. Pure Appl. Algebra 215 (2011), no. 4, 578-88.
- [5] S. Brzostowski, T. Krasinski, and G. Oleksik, A conjecture on the Lojasiewicz exponent, J. Singul. 6 (2012), 124-130.
- [6] J. Gwoździewicz, The Lojasiewicz exponent of an analytic function at an isolated zero, Comment. Math. Helv. Vol. 74 (1999), No. 3, 364-375.
- [7] A. Haraux, Positively homogeneous functions and the Lojasiewicz gradient inequality, Ann. Polon. Math. Vol. 87 (2005), 165-174.
- [8] J. Kollar, An effective Lojasiewicz inequality for real polynomials, Period. Math. Hungar., Vol. 38 (1999), N. 3, 213-221.
- T. Krasinski, G. Oleksik, and A. Ploski, The Lojasiewicz exponent of an isolated weighted homogeneous surface singularity, Proc. Amer. Math. Soc. 137 (2009), no. 10, 3387-3397.
- [10] T. C. Kuo and Y. C. Lu, On analytic function germs of two complex variables, Topology, Vol. 16 (1977), 299-310.
- [11] K. Kurdyka, and S. Spodzieja, Separation of real algebraic sets and the Lojasiewicz exponent, Proc. Amer. Math. Soc., 142 (9) S 0002-993912061-2, (2014), 3089-3102.
- [12] C. Li, B. S. Mordukhovich, and T. S. Pham, New fractional error bounds for polynomial systems with applications to Höderian stability in optimization and spectral theory of tensors, Math. Program. Ser. A. DOI: 10.1007/s10107-014-0806-9
- [13] S. Lojasiewicz, Une proprriété topologique des sous ensembles analytiques réels, Colloques internationaux du C.N.R.S, 117. Les Équations aux Dérivées Partielles, (1963) 87-89.
- [14] S. Lojasiewicz, Ensembles semi-analytiques, preprint IHES, 1965.
- [15] J. Milnor, Singular points of complex hypersurfaces, Annals of Mathematics Studies 61, Princeton University Press, 1968.
- [16] T. S. Pham, The Lojasiewicz exponent of a subanalytic mapping at an isolated zero, Proceedings of the American Mathematical Society, 139 (2011), 1-9.
- [17] T. S. Pham, An explicit bound for the Lojasiewicz exponent of real polynomials, Kodai Mathematical Journal, No. 2, 35 (2012), 311-319.
- [18] B. Teissier, Varietés polaires I Invariants polaires des singularités d'hypersufaces, Inventiones Mathematicae, Vol. 40 (1977), 267-293. DOI: 10.1007/BF01425742

ALAIN HARAUX, LABORATOIRE JACQUES-LOUIS LIONS, U.M.R C.N.R.S. 7598, UNIVERSITÉ PIERRE ET MARIE CURIE, BOÎTE COURRIER 187, 75252 PARIS CEDEX 05, FRANCE

E-mail address: haraux@ann.jussieu.fr

TIEN SON PHAM, DEPARTMENT OF MATHEMATICS, DALAT UNIVERSITY, DALAT, VIETNAM *E-mail address*: sonpt@dlu.edu.vn
L²-RIEMANN-ROCH FOR SINGULAR COMPLEX CURVES

JEAN RUPPENTHAL AND MARTIN SERA

ABSTRACT. We present a comprehensive L^2 -theory for the $\overline{\partial}$ -operator on singular complex curves, including L^2 -versions of the Riemann-Roch theorem and some applications.

1. Introduction

The L^2 -theory for the $\overline{\partial}$ -operator is one of the central tools in complex analysis on complex manifolds, but still not very well developed on singular complex spaces. Just recently, considerable progress has been made in understanding the L^2 -cohomology of singular complex spaces with isolated singularities. Let X be a Hermitian complex space of pure dimension n and with isolated singularities only. For simplicity, we assume that X is compact. Let $H_w^{p,q}(X^*)$ be the L^2 -Dolbeault cohomology on the level of (p,q)-forms with respect to the $\overline{\partial}$ -operator in the sense of distributions, denoted by $\overline{\partial}_w$ in the following, computed on $X^* = \operatorname{Reg} X$. Let $\pi : M \to X$ be a resolution of singularities with snc exceptional divisor, $Z := \pi^{-1}(\operatorname{Sing} X)$ the unreduced exceptional divisor. Then it has been shown by Øvrelid, Vassiliadou [ØV13] and the first author [Rup11, Rup14] by different approaches that there exists an effective divisor $D \geq Z - |Z|$ on M such that there are natural isomorphisms

$$H^{n,q}_{w}(X^{*}) \cong H^{n,q}(M) ,
 H^{0,q}_{w}(X^{*}) \cong \frac{H^{q}(M, \mathcal{O}(D))}{H^{q}_{|Z|}(M, \mathcal{O}(D))}$$
(1.1)

for all $0 \le q \le n$. Here, $H_{|Z|}^q$ denotes the cohomology with support on |Z|. If dim $X \le 2$, then (1.1) holds with the divisor D = Z - |Z| and $H_{|Z|}^q (M, \mathcal{O}(Z - |Z|)) = \{0\}$, so that (1.1) gives a nice smooth representation of the L^2 -cohomology groups $H_w^{0,q}(X^*)$. In case dim X > 2, it is conjectured that (1.1) holds with D = Z - |Z| (see [Rup11]).

However, the L^2 -theory for the $\overline{\partial}$ -operator developed in [ØV13] and [Rup11, Rup14] applies only to dim $X \ge 2$ (for dim X = 1, (1.1) has been known before, see [Par89, PS91]). The purpose of the present paper is to give a complete L^2 -theory for the $\overline{\partial}$ -operator on a singular complex curve, including L^2 -versions of the Riemann-Roch theorem, and to understand the appearance of the divisor Z - |Z| in the case dim X = 1.

Let us explain some of our results in detail. Let X be a Hermitian singular complex space¹ of dimension 1, i. e., a Hermitian complex curve, and $L \to X$ a Hermitian holomorphic line bundle. Let $\overline{\partial}_w : L^{p,q}(X^*, L) \to L^{p,q+1}(X^*, L)$ denote the weak extension of the Cauchy-Riemann operator $\overline{\partial} : \mathscr{D}^{p,q}(X^*, L) \to \mathscr{D}^{p,q+1}(X^*, L)$, i. e., the $\overline{\partial}$ -operator in the sense of distributions. Here, $\mathscr{D}^{p,q}(X^*, L) := \mathscr{C}^{\infty}_{cpt}(X^*, \Lambda^{p,q}T^*X^* \otimes L)$ denotes the set of smooth differential forms with compact support in X^* and values in L, and $L^{p,q}(X^*, L)$ is the set of square-integrable forms with values in L and respect to the Hermitian metrics on X^* and L.

Key words and phrases. Dolbeault operator, L^2 -theory, singular complex curves, Riemann-Roch theorem.

²⁰¹⁰ Mathematics Subject Classification. 32W05, 32C36, 14C40.

¹ A Hermitian complex space (X, g) is a reduced complex space X with a metric g on the regular part such that the following holds: If $x \in X$ is an arbitrary point there exist a neighborhood U = U(x), a holomorphic embedding of U into a domain G in \mathbb{C}^N and an ordinary smooth Hermitian metric in G whose restriction to U is $g|_U$.

Let $H^{p,q}_w(X^*,L)$ denote the L^2 -Dolbeault cohomology on X^* with respect to $\overline{\partial}_w$ and

$$h_w^{p,q}(X^*,L) := \dim H_w^{p,q}(X^*,L)$$

Note that the genus g = g(X) of X and the degree $\deg(L)$ of L are well-defined, even in the presence of singularities (see Section 2.2). For a singular point $x \in \operatorname{Sing} X$, we define its modified multiplicity $\operatorname{mult}'_x X$ as follows: Let X_j , j = 1, ..., m, be the irreducible components of X in the singular point x. Then

$$\operatorname{mult}_{x}' X := \sum_{j=1}^{m} \left(\operatorname{mult}_{x} X_{j} - 1 \right).$$

Note that regular irreducible components do not contribute to $\operatorname{mult}'_x X$. In Section 2.2, we recall the definition of the multiplicity $\operatorname{mult}_x X_j$ and present different ways to compute it.

Theorem 1.2 ($\overline{\partial}_w$ -Riemann-Roch). Let X be a compact Hermitian complex curve with m irreducible components and $L \to X$ a holomorphic line bundle. Then

$$h_w^{0,0}(X^*,L) - h_w^{0,1}(X^*,L) = m - g + \deg(L) + \sum_{x \in \operatorname{Sing} X} \operatorname{mult}'_x X,$$
 (1.3)

and

$$h_w^{1,1}(X^*, L) - h_w^{1,0}(X^*, L) = m - g - \deg(L)$$

Theorem 1.2 is a corollary of Theorem 4.4 which we prove in Section 4. We also consider an L^2 -dual version there, i.e., an L^2 -Riemann-Roch theorem for the minimal closed L^2 -extension of the $\overline{\partial}$ -operator which we denote by $\overline{\partial}_s$ (see Section 2.1).

On singular complex curves, the $\overline{\partial}_s$ -operator is of particular importance because of its relation to weakly holomorphic functions. Namely, the weakly holomorphic functions are precisely the $\overline{\partial}_s$ -holomorphic L^2_{loc} -functions (for a localized version of the $\overline{\partial}_s$ -operator, see Section 5). Let $H^{p,q}_{s,\text{loc}}(X^*)$ denote the L^2_{loc} -Dolbeault cohomology on X^* with respect to $\overline{\partial}_s$, and $\widehat{\mathcal{O}}_X$ the sheaf of germs of weakly holomorphic functions on X. Then:

Theorem 1.4. Let X be a Hermitian complex curve. Then

$$\begin{aligned} H^0(X, \widehat{\mathcal{O}}_X) &= H^{0,0}_{s, \text{loc}}(X^*), \\ H^1(X, \widehat{\mathcal{O}}_X) &\cong H^{0,1}_{s, \text{loc}}(X^*). \end{aligned}$$

If X is irreducible and compact, then dim $H^0(X, \widehat{\mathcal{O}}_X) = 1$, dim $H^1(X, \widehat{\mathcal{O}}_X) = g(X)$. We prove Theorem 1.4 in Section 5.

To exemplify the use of L^2 -theory for the $\overline{\partial}$ -operator on a singular complex space, in particular the L^2 -Riemann-Roch theorem, we give in Section 6 two applications. There, we use our L^2 -theory to give alternative proofs of two well-known facts. First, we show that each compact complex curve can be realized as a ramified covering of \mathbb{CP}^1 . Second, we show that a positive holomorphic line bundle over a compact complex curve is ample, yielding that any compact complex curve is projective.

Let us clarify the relation to previous work of others. In the case of complex curves, (1.1) was in essence discovered by Pardon [Par89], and one can deduce parts of Theorem 4.4 and the second statement of Corollary 4.8 from Pardon's work by some additional arguments on the regularity of the $\bar{\partial}$ -operator. The first part of Corollary 4.8 was discovered by Haskell [Has89], and from that one can deduce the second statement of Theorem 1.2 by use of L^2 -Serre duality. Moreover, Theorem 1.2 was proved in essence by Brüning, Peyerimhoff and Schröder in [BPS90] and [Sch89] by computing the indices of the $\bar{\partial}_w$ - and the $\bar{\partial}_s$ -operator.

The new point in the present work is that we can put all the partial results mentioned above in the general framework of a comprehensive L^2 -theory. From that, we draw also a new understanding of weakly holomorphic functions (Theorem 1.4) and of the divisor Z - |Z|. Moreover, all the previous work has been done only for forms with values in the trivial bundle

68

(except of [Sch89]), whereas we incorporate line bundles. This is essential for applications as we illustrate by the examples mentioned above.

Acknowledgment. This research was supported by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation), grant RU 1474/2 within DFG's Emmy Noether Programme. The authors thank Eduardo S. Zeron for interesting and fruitful discussions and are grateful to the unknown referee for several suggestions which helped to improve the paper.

2. Preliminaries

2.1. Closed extensions of the Cauchy-Riemann operator. Let X be a complex curve and $X^* := \operatorname{Reg} X$ the set of regular points. We assume that X is a Hermitian complex space in the sense that X^* carries a Hermitian metric γ which is locally given as the restriction of the metric of the ambient space when X is embedded holomorphically into some complex number space.

We denote by $\mathscr{D}^{p,q}(X^*)$ the smooth differential forms of degree (p,q) with compact support in X^* (test forms) and by $L^{p,q}(X^*)$ the set of square-integrable forms with respect to the metric γ on X^* .

Let $\overline{\partial}_s : L^{p,q}(X^*) \to L^{p,q+1}(X^*)$ be the minimal (strong) closed L^2 -extension of the Cauchy-Riemann operator $\overline{\partial} : \mathscr{D}^{p,q}(X^*) \to \mathscr{D}^{p,q+1}(X^*)$, i. e., $\overline{\partial}_s$ is defined by the closure of the graph of $\overline{\partial}$ in $L^{p,q}(X^*) \times L^{p,q+1}(X^*)$. $\overline{\partial}_w : L^{p,q}(X^*) \to L^{p,q+1}(X^*)$ is the maximal (weak) closed L^2 -extension of $\overline{\partial}$, i. e., $\overline{\partial}_w$ is defined in sense of distributions. We denote by $H^{p,q}_{w/s}(X^*)$ the Dolbeault cohomology with respect to $\overline{\partial}_w$ or $\overline{\partial}_s$, respectively, and by $h^{p,q}_{w/s}(X^*)$ the dimension of $H^{p,q}_{w/s}(X^*)$.

Let $\vartheta : \mathscr{D}^{p,q+1}(X^*) \to \mathscr{D}^{p,q}(X^*)$ be the formal adjoint of $\overline{\partial}$ and $\vartheta_{s/w} := \overline{\partial}_{w/s}^*$ the Hilbert-space adjoint of $\overline{\partial}_{w/s}$. This notation makes sense as $\vartheta_{w/s}$ is in fact the maximal (weak) or minimal (strong), respectively, L^2 -extension of ϑ . Let $\overline{*} : L^{p,q}(X^*) \to L^{1-p,1-q}(X^*)$ be the conjugated Hodge-*-operator with respect to the metric γ . Then we have $\vartheta_{w/s} = -\overline{*} \overline{\partial}_{w/s} \overline{*}$.

Let $L \to X$ be a Hermitian holomorphic line bundle on X with an (arbitrary) metric on L which is smooth on the whole of X. We define $\mathscr{D}^{p,q}(X^*, L) := \mathscr{C}^{\infty}_{cpt}(X^*, \Lambda^{p,q}T^*X^* \otimes L)$ as the smooth (p,q)-forms with compact support and values in L, and $L^{p,q}(X^*, L)$ as the Hilbert space of square-integrable forms with values in L. We consider the Cauchy-Riemann operator $\overline{\partial} : \mathscr{D}^{p,q}(X^*, L) \to \mathscr{D}^{p,q+1}(X^*, L)$ locally given by $\overline{\partial} : \mathscr{D}^{p,q}(X^*) \to \mathscr{D}^{p,q+1}(X^*)$. Since $\overline{\partial}$ commutes with the trivializations of the holomorphic line bundle, $\overline{\partial}$ is well defined. We get the weak and strong extensions $\overline{\partial}_w, \overline{\partial}_s : L^{p,q}(X^*, L) \to L^{p,q+1}(X^*, L)$ and the cohomology $H^{p,q}_{w/s}(X^*, L)$ as above.

In Section 3, we will study also the following other closed extensions of $\overline{\partial}$ besides the minimal $\overline{\partial}_s$ and the maximal $\overline{\partial}_w$. Let $D \in \mathbb{C}^n$ be a domain, and $X \subset D$ an analytic set of dimension one with Sing $X = \{0\}$. We can interpret $\overline{\partial}_s$ as $\overline{\partial}_w$ with certain boundary conditions. The boundary of X^* consists of two parts, the singular point $\{0\}$ and the boundary at ∂D : $\partial X = \partial X^* \setminus \{0\}$. Therefore, there are two boundary conditions. Let $\overline{\partial}_{s,w}$ denote the closed L^2 -extension which satisfies the boundary condition at $\{0\}$, i. e., $f \in \text{dom } \overline{\partial}_{s,w}$ iff $f \in \text{dom } \overline{\partial}_w$ and there is a sequence $\{f_j\}$ in dom $\overline{\partial}_w$ such that $\text{supp } f_j \cap \{0\} = \emptyset$, $f_j \to f$, and $\overline{\partial}_w f_j \to \overline{\partial}_w f$ in L^2 . $\overline{\partial}_{w,s}$ denotes the extension which satisfies the boundary condition at ∂X , i. e., $f \in \text{dom } \overline{\partial}_{w,s}$ iff $f \in \text{dom } \overline{\partial}_w$ and there is a sequence the closed $\{f_j\}$ in dom $\overline{\partial}_w$ such that $\text{supp } f_j \cap \{0\} = \emptyset$, $f_j \to f$, and $\overline{\partial}_w f_j \to \overline{\partial}_w f$ in L^2 . $\overline{\partial}_w f_j \to \overline{\partial}_w f$ in L^2 . We define the adjoint operators

$$\vartheta_{s,w} := -\overline{\ast} \partial_{s,w} \overline{\ast} \quad \text{and} \quad \vartheta_{w,s} := -\overline{\ast} \partial_{w,s} \overline{\ast},$$

which we can realize as Hilbert-space adjoint operators (see [Rup14, Lem. 5.1]):

Lemma 2.1. The Hilbert-space adjoints $\overline{\partial}_{s,w}^*$ and $\overline{\partial}_{w,s}^*$ satisfy the representations

$$\overline{\partial}_{s,w}^* = \vartheta_{w,s} = -\overline{*} \,\overline{\partial}_{w,s} \,\overline{*} \qquad \text{and} \qquad \overline{\partial}_{w,s}^* = \vartheta_{s,w} = -\overline{*} \,\overline{\partial}_{s,w} \,\overline{*},$$

respectively.

2.2. Resolution of complex curves, divisors, line bundles. Every (reduced) complex space X (which is countable at infinity) has a resolution of singularities $\pi : M \to X$, i. e., there are a complex manifold M, a proper complex subspace S of X which contains the singular locus of X and a proper holomorphic map $\pi : M \to X$ such that the restriction $M \setminus \pi^{-1}(S) \to X \setminus S$ of π is biholomorphic, and $\pi^{-1}(S)$ is the locally finite union of smooth hypersurfaces (see [Hir64] and [Hir77, Thm. 7.1]).

If X is a compact complex curve, then such a resolution is given just by the normalization of the curve, and it is unique up to biholomorphism: Let $\pi_1 : M_1 \to X$ and $\pi_2 : M_2 \to X$ be two resolutions of X. Then $\psi := \pi_2^{-1} \circ \pi_1 : M_1 \setminus \pi_1^{-1}(\operatorname{Sing} X) \to M_2 \setminus \pi_2^{-1}(\operatorname{Sing} X)$ is biholomorphic and bounded in the singular locus. Yet, $\pi_i^{-1}(\operatorname{Sing} X)$ consist of isolated points. Therefore, ψ has a (bi-) holomorphic extension.

Let $\pi: M \to X$ be a resolution of a compact complex curve X. We define the *genus* of X by the genus of the resolution

$$g(X) := h^1(M) = \dim H^1(M, \mathcal{O}).$$

If X has more than one irreducible component, then M is not connected and $h^1(M)$ is the sum of the genera of the connected components. Since the resolution is unique up to biholomorphism, this is well-defined.

Throughout the article (except of Section 6.1), we will work with divisors on compact Riemann surfaces only. Therefore, there is no difference between Cartier and Weil divisors, and we can associate to each line bundle a divisor.

Let $L \to X$ be a holomorphic line bundle. Then the pull-back $\pi^*L \to M$ is well-defined by the pull-back of the transition functions of the line bundle. There is a divisor D on M associated to π^*L such that $\mathcal{O}(\pi^*L) \cong \mathcal{O}(D)^2$ and deg $\pi^*L = \deg D$. The uniqueness of the resolution (up to biholomorphism) implies the independence of deg π^*L from π , so that

$$\deg L := \deg \pi^* L$$

is also well-defined.

For any divisor D on M, there exists a holomorphic line bundle $L_D \to M$ associated to Dsuch that $\mathcal{O}(L_D) \cong \mathcal{O}(D)$. The constant function f = 1 induces a meromorphic section s_D of L_D such that $\operatorname{div}(s_D) = D$. One can then identify sections in $\mathcal{O}(D)$ with sections in $\mathcal{O}(L_D)$ by $g \mapsto g \otimes s_D$, and we denote the inverse mapping by $s \mapsto s \cdot s_D^{-1}$. If Y is an effective divisor, then s_Y is a holomorphic section of L_Y and $\mathcal{O} \subset \mathcal{O}(Y)$. Hence, there is the natural inclusion $\mathcal{O}(D) \subset \mathcal{O}(D+Y)$ which induces the inclusion $\mathcal{O}(L_D) \subset \mathcal{O}(L_{D+Y})$ given by $s \mapsto (s \cdot s_D^{-1}) \otimes s_{D+Y}$. For $U \subset M$, we obtain the inclusion

$$L^{p,q}_{\text{loc}}(U, L_D) \hookrightarrow L^{p,q}_{\text{loc}}(U, L_{D+Y}), s \mapsto (s \cdot s_D^{-1}) \otimes s_{D+Y}.$$
(2.2)

Here, $L_{loc}^{p,q}(U, L_D)$ denotes the locally square-integrable forms with values in L_D . This definition is independent of the chosen Hermitian metric on L_D . If M is compact, all metrics are equivalent and we get the inclusion

$$L^{p,q}(M, L_D) \subset L^{p,q}(M, L_{D+Y}).$$
 (2.3)

Let $Z := \pi^{-1}(\operatorname{Sing} X)$ be the unreduced exceptional divisor and |Z| the underlying reduced divisor. Then $\deg(Z - |Z|)$ is independent of the resolution as well. We will discuss some alternative ways to compute $\deg Z$.

² We denote by $\mathcal{O}(D)$ the sheaf of germs of holomorphic functions f such that $\operatorname{div}(f) + D \ge 0$.

Locally, the resolution is given by the Puiseux parametrization: Let A be an analytic set of dimension 1 in $\Omega \in \mathbb{C}^n$ with $\operatorname{Sing} A = \{0\}$ which is irreducible at 0. Shrinking Ω , there are coordinates $z, w_1, ..., w_{n-1}$ around 0 such that A is contained in the cone $||w|| \leq C|z|$, $w = (w_1, ..., w_{n-1})$. The projection $\operatorname{pr}_z : A \to \mathbb{C}_z$ on the z-coordinate is a finite ramified covering. Let s be the number of the sheets of pr_z . Generic choice of the coordinates gives the same number of sheets s, called the multiplicity $\operatorname{mult}_0 A$ of A in $\{0\}$. There exists a parametrization $\pi : \Delta \to A, t \mapsto (t^s, w_1(t), ..., w_{n-1}(t))$, where $\Delta := \{t \in \mathbb{C} : |t| < 1\}$; cf. e. g. [Chi89, Sect. 6.1]. π is called the Puiseux parametrization. The unreduced exceptional divisor is just $Z = (\pi^{-1}(z)) = (t^s)$, and so deg Z = s.

The number of sheets of the covering pr_z is also equal to the Lelong number $\nu([A], 0)$ of the positive closed current [A] given by the integration over A (see [Chi89, Prop. 2 in § 3.15], [Dem12, Thm. 7.7] or [GH78, § 3.2]).

The tangent cone gives another way to compute $\operatorname{mult}_0 A$. For a holomorphic function f on Ω , let $f = \sum_{k=k_0}^{\infty} f_k$ be the decomposition in homogeneous polynomials f_k of degree k with $f_{k_0} \neq 0$ (choosing a smaller Ω) and $f^* := f_{k_0} \neq 0$ be the *initial homogeneous polynomial of* f. If A is given by the ideal sheaf \mathscr{J}_A , then

$$C_0(A) = \{ \alpha \in \mathbb{C}^n : f^*(\alpha) = 0 \ \forall f \in \mathscr{J}_{A,0} \} \subset T_0 \mathbb{C}^n$$

is called the *tangent cone* of A in 0 (cf. [Chi89, Sect. 8.4]). The natural projection $\mathbb{C}^n \setminus 0 \to \mathbb{CP}^{n-1}$ maps $C_0(A)$ on a projective variety $\widetilde{C}_0(A)$. The degree deg Y of a projective variety Y in \mathbb{CP}^{n-1} of dimension p is defined as the class of Y in $H_{2p}(\mathbb{CP}^{n-1}, \mathbb{Z}) \cong \mathbb{Z}$, and $\operatorname{mult}_0 A = \operatorname{deg} \widetilde{C}_0(A)$ (see Sect. 2 of [GH78, §1.3]). In the case of an irreducible complex curve A, note that $\widetilde{C}_0(A)$ is just a point of multiplicity $\operatorname{mult}_0 A$.

All in all, we have

$$\deg Z = \operatorname{mult}_0 A = \nu([A], 0) = \deg C_0(A).$$

2.3. Extension theorems. We need the following extension theorem. Let Δ be the unit disc in \mathbb{C} and $\Delta^* := \Delta \setminus \{0\}$.

Theorem 2.4 (L^2 -extension). If $u \in L^{p,0}_{loc}(\Delta)$ and $v \in L^{p,1}_{loc}(\Delta)$ satisfy $\overline{\partial}u = v$ on Δ^* in the sense of distributions, then $\overline{\partial}u = v$ on Δ .

A more general statement is true for domains in \mathbb{C}^n and proper analytic subsets of arbitrary codimension, cf. e. g. [Rup09, Thm. 3.2].

If $A \subset \Omega \Subset \mathbb{C}^n$ is a pure dimensional analytic set, let $\widehat{\mathcal{O}} = \widehat{\mathcal{O}}_A$ be the normalization sheaf of \mathcal{O}_A which is defined stalk-wise by the integral closure of $\mathcal{O}_{A,x}$ in the sheaf $\mathcal{M}_{A,x}$ of meromorphic functions for all $x \in A$. A function in $\widehat{\mathcal{O}}(U)$, $U \subset A$ open, is called *weakly holomorphic*. Weakly holomorphic functions are holomorphic in regular points of A and bounded in singular points. If A is locally irreducible, then weakly holomorphic functions are continuous in Sing A (cf. e.g. [GR84, § VI.4].)

The classical Riemann extension theorem generalizes to the following result (see e.g. [GR84, Sect. VII.4.1]).

Theorem 2.5 (Riemann extension). Let $A \subset \Omega \in \mathbb{C}^n$ be a pure dimensional analytic set. Every holomorphic function on $A^* := \text{Reg } A$ which is bounded at Sing A is weakly holomorphic on A.

J. RUPPENTHAL AND M. SERA

3. Local L^2 -theory of complex curves

In this section, we study the local L^2 -theory of (locally) irreducible analytic curves in \mathbb{C}^n . By the remarks on the local structure of singular complex curves in Section 2.2 and Section 4, it follows that the studied situation is general enough. We will compute the L^2 -Dolbeault cohomology by use of the Puiseux parametrization and will see why the term $\sum_{x \in \text{Sing } X} \text{mult}'_x X$ occurs in (1.3).

Let A be an irreducible analytic curve in $\Delta^n \subset \mathbb{C}^n_{zw_1\dots w_{n-1}}$ given by the Puiseux parametrization

$$\pi: \Delta \to \mathbb{C}^n$$
, $\pi(t) := (t^s, w(t)),$

where $w = (w_1, ..., w_{n-1}): \Delta \to \Delta^{n-1}$ is a holomorphic map such that each component w_i vanishes at least of the order s+1 in the origin. Here, Δ is the unit disk $\{t \in \mathbb{C} : |t| < 1\}$. We can assume that π is bijective, in particular, a resolution/normalization of A such that mult₀ A = s. Further, we can assume that 0 is the only singular point of A.

For a regular point $(z_0, w_0) \in A^* := \operatorname{Reg} A$, let $t_0 \in \Delta^*$ be the preimage under π . Since π is biholomorphic on $\Delta^* := \Delta \setminus \{0\}, d\pi_{t_0}(\frac{\partial}{\partial t}) = st_0^{s-1}\frac{\partial}{\partial z} + \sum_{k=1}^{n-1} w'_k(t_0)\frac{\partial}{\partial w_k}$ is a non-vanishing tangent vector of A^* in (z_0, w_0) , i.e.,

$$(1 + \|\frac{1}{s}t_0^{1-s}w'(t_0)\|^2)^{-1/2} \left(\frac{\partial}{\partial z} + \sum_{k=1}^{n-1} \frac{1}{s}t_0^{1-s}w'_k(t_0)\frac{\partial}{\partial w_k}\right)$$

is a normalized generator of $T_{(z_0,w_0)}A^*$ and $(1 + \|\frac{1}{s}t_0^{1-s}w'(t_0)\|^2)^{1/2}dz$ is a normalized generator of $T^*_{(z_0,w_0)}A^*$. Since w'_k vanishes at least of order s in the origin, we obtain $1 + \|\frac{1}{s}t^{1-s}w'(t)\|^2 \sim 1$ on Δ and $dV_{A^*} \sim idz \wedge d\overline{z}$, where dV_{A^*} denotes the volume form on A^* induced by the standard Euclidean metric of \mathbb{C}^n . Using $\pi^*dz = d(\pi^*z) = dt^s = st^{s-1}dt$ and $\pi^*(dz \wedge d\overline{z}) = s^2|t|^{2(s-1)}dt \wedge d\overline{t}$, we get

$$\tau^* dV_{A^*} \sim |t|^{2(s-1)} dV_\Delta$$

Let $\iota: A^* \to \Delta^*$ be the inverse of π . Then, $\iota(z, w)$ is the root $t = \sqrt[s]{z}$ with w = w(t). We get $\iota^*(dt) = \frac{1}{s} z^{1/s-1} dz$ and $\iota^*(dt \wedge d\overline{t}) = \frac{1}{s^2} |z|^{2(1/s-1)} dz \wedge d\overline{z}$, i. e.,

$$\iota^* dV_{\Delta^*} \sim |z|^{2(1/s-1)} dV_{A^*}.$$

If g is a measurable function on A^* , we obtain

$$\int_{A^*} |g|^2 dV_{A^*} = \int_{\Delta} |\pi^* g|^2 \pi^* dV_{A^*} \sim \int_{\Delta} |\pi^* g|^2 \cdot |t|^{2(s-1)} dV_{\Delta}$$

Hence,

$$g \in L^{0,0}(A^*) \Leftrightarrow t^{s-1}\pi^*g \in L^{0,0}(\Delta).$$
 forms, we have

For (0, 1)-forms and (1, 1)-forms, we have

$$\pi^*(gd\overline{z}) = \pi^*g \cdot \pi^*(d\overline{z}) = \overline{t}^{s-1}\pi^*(g)d\overline{t},$$
$$\pi^*(gdz \wedge d\overline{z}) = |t|^{2(s-1)}\pi^*(g)dt \wedge d\overline{t}.$$

Thus

$$f \in L^{0,0}(A^*) \Leftrightarrow t^{s-1} \cdot \pi^* f \in L^{0,0}(\Delta),$$

$$f \in L^{1,0}(A^*) \Leftrightarrow \pi^* f \in L^{1,0}(\Delta),$$

$$f \in L^{0,1}(A^*) \Leftrightarrow \pi^* f \in L^{0,1}(\Delta), \text{ and}$$

$$f \in L^{1,1}(A^*) \Leftrightarrow t^{1-s} \cdot \pi^* f \in L^{1,1}(\Delta).$$

(3.1)

On the other hand, if $v \in L^{0,0}(\Delta)$, we get

$$\infty > \int_{\Delta} |v|^2 dV_{\Delta} = \int_{A^*} |\iota^* v|^2 \iota^* dV_{\Delta} \sim \int_{A^*} |\iota^* v|^2 \cdot |z|^{2(1/s-1)} dV_{A^*}$$

Thus, $|z|^{1/s-1}\iota^* v$ is square-integrable on A^* . For each (0,1)-form $vd\bar{t} \in L^{0,1}(\Delta)$, we get

$$s\iota^*(vd\overline{t}) = \overline{z}^{1/_s - 1}\iota^*(v)d\overline{z} \in L^{0,1}(A^*),$$

and for each (1,1)-form $vdt \wedge d\bar{t} \in L^{1,1}(\Delta)$, we get $|z|^{1-\frac{1}{s}}\iota^*(vdt \wedge d\bar{t}) \in L^{1,1}(A^*)$.

So, if $f \in L^{0,1}(A^*)$, then $u := \pi^* f$ is in L^2 , too. Since dim $\Delta = 1$, there exists $v \in L^{0,0}(\Delta)$ with $\overline{\partial}_w v = u$. We set $g := \iota^* v$. Since $|z|^{1/s-1}g$ is in L^2 and $|z|^{2(1-1/s)}$ is bounded,

$$\|g\|_{L^2}^2 = \int_{A^*} |z^{1/s-1}g|^2 \cdot |z|^{2(1-1/s)} dV_{A^*} \le \|z^{1/s-1}g\|_{L^2} \cdot \|z^{2(1-1/s)}\|_{L^{\infty}} < \infty$$

Hence, we get an L^2 -solution for $\overline{\partial}_w g = f$ and

$$H_w^{0,1}(A^*) = L^{0,1}(A^*) / \mathcal{R}(\overline{\partial}_w) = 0$$

In the same way, it is easy to compute

$$H_w^{1,1}(A^*) = 0.$$

We will now determine $H^{p,0}_w(A^*) = \ker(\overline{\partial}_w : L^{p,0} \to L^{p,1})$ by use of the L^2 -extension theorem (Theorem 2.4). For this, let $\mathcal{O}_{L^2}(\Delta)$ be the square-integrable holomorphic functions on Δ , and let $\Omega^1_{L^2}(\Delta)$ be the holomorphic 1-forms with square-integrable coefficient. If $g \in L^{0,0}(A^*)$ and $\overline{\partial}_w g = 0$, then $v := \pi^* g \in |t|^{1-s} L^{0,0}(\Delta)$ and $\overline{\partial}_w v = 0$ on Δ^* . Therefore, $\overline{\partial}(t^{s-1}v) = 0$ on Δ^* and $t^{s-1}v \in L^{0,0}(\Delta)$. The extension theorem implies $\overline{\partial}(t^{s-1}v) = 0$ on Δ , i. e., v is a meromorphic function with a pole of order s - 1 or less at the origin. We say $v \in t^{1-s} \mathcal{O}_{L^2}(\Delta)$. Since, on the other hand, $\iota^*(t^{1-s} \mathcal{O}_{L^2}(\Delta)) \subset \ker \overline{\partial}_w$, we conclude

$$H^{0,0}_w(A^*) \cong t^{1-s}\mathcal{O}_{L^2}(\Delta).$$
 (3.2)

If $f \in L^{1,0}(A^*)$ and $\overline{\partial}_w f = 0$, then $u := \pi^* f \in L^{1,0}(\Delta)$ and $\overline{\partial}_w u = 0$ on Δ (using the extension theorem again). Hence, u is holomorphic on Δ and

$$H^{1,0}_w(A^*) \cong \Omega^1_{L^2}(\Delta).$$

To compute the cohomology groups $H_s^{*,*}(A^*)$, we use L^2 -duality:

Lemma 3.3. Let $\overline{\partial}_e$ denote either the weak or the strong closed extension of $\overline{\partial}$, and $\overline{\partial}_{e^c}$ the other one. For $p \in \{0,1\}$, let the range $\mathcal{R}(\overline{\partial}_e)$ of $\overline{\partial}_e : L^{p,0} \to L^{p,1}$ be closed. Then

$$H^{p,1}_{e}(A^*) \cong H^{1-p,0}_{e^c}(A^*).$$

For the proof see e.g. [Rup14, Thm. 2.3].

Lemma 3.4. For $p \in \{0, 1\}$,

$$H_s^{p,0}(A^*) \cong H_w^{1-p,1}(A^*) = 0$$
 and
 $H_s^{p,1}(A^*) \cong H_w^{1-p,0}(A^*).$

Proof. Recall that $H_w^{1-p,1}(A^*) = 0$. This implies $L^{1-p,1}(A^*) = \mathcal{R}(\overline{\partial}_w)$ and, particularly, that the range of $\overline{\partial}_w : L^{1-p,0} \to L^{1-p,1}$ is closed. As $\vartheta_w = -\overline{*}\overline{\partial}_w \overline{*}$ and $\overline{*}$ is an isometric isomorphism, we conclude that the range of $\vartheta_w : L^{p,1} \to L^{p,0}$ is closed as well. This is equivalent to the range of $\overline{\partial}_s = \vartheta_w^* : L^{p,0} \to L^{p,1}$ being closed (standard functional analysis). Lemma 3.3 implies both isomorphisms.

To get the complete picture, we also need to understand the Dolbeault cohomology groups of the closed extensions $\overline{\partial}_{s,w}$ and $\overline{\partial}_{w,s}$, respectively.

Lemma 3.5. For $p \in \{0, 1\}$,

$$H^{p,0}_{w,s}(A^*) = 0.$$

Proof. Let $f \in \ker \overline{\partial}_{w,s} = H^{p,0}_{w,s}(A^*)$. We have showed $\omega \cdot u := \omega \cdot \pi^* f \in L^{p,0}(\Delta)$ with $\omega(t) = t^{s-1}$ if p = 0 and $\omega(t) \equiv 1$ if p = 1. By the extension theorem, we conclude $\overline{\partial}_s(\omega \cdot u) = 0$ on Δ , where $\overline{\partial}_s$ denotes the (strong) closure of $\overline{\partial}_{cpt} : \mathscr{C}^{\infty}_{cpt;p,0}(\Delta) \to \mathscr{C}^{\infty}_{cpt;p,1}(\Delta)$. The generalized Cauchy condition implies that the trivial extension of ωu to the complex plane is a holomorphic *p*-form with compact support (cf. [LM02, § V.3]). We deduce that $\omega u = 0$ and, hence, f = 0.

Lemma 3.6.

$$H^{0,0}_{s,w}(A^*) \cong \mathcal{O}_{L^2}(\Delta) \text{ and} \\ H^{1,0}_{s,w}(A^*) \cong t^{s-1}\Omega^1_{L^2}(\Delta).$$

As $\mathcal{O}_{L^2}(\Delta) \cong \widehat{\mathcal{O}}_{L^2}(A)$, the first isomorphism implies that the $\overline{\partial}_{s,w}$ -holomorphic functions on a singular complex curve are precisely the square-integrable weakly holomorphic functions.

Proof. First, we prove that $\mathcal{O}_{L^2}(\Delta) = \pi^*(\ker \overline{\partial}_{s,w} : L^{0,0}(A^*) \to L^{0,1}(A^*)).$

i) For $v \in \mathcal{O}_{L^2}(\Delta)$, we claim that $g := \iota^* v \in \ker \overline{\partial}_{s,w}$. To see that, choose smooth functions $\tilde{\chi}_k : \mathbb{R} \to [0,1]$ with $\tilde{\chi}_k|_{(-\infty,k]} = 0$, $\tilde{\chi}_k|_{[k+1,\infty)} = 1$ and $|\tilde{\chi}'_k| \leq 2$. We get

$$(\tilde{\chi}_k \circ \log \circ |\log|)'(\rho) = \frac{\tilde{\chi}'_k(\log|\log\rho|)}{\rho\log\rho}$$

We define $\chi_k : A^* \to [0,1], (z,w) \mapsto \tilde{\chi}_k(\log |\log |z||)$ (which is inspired by [PS91, p. 617]) and get $\sup p \overline{\partial} \chi_k \subset A^* \cap \Delta_{\varepsilon_k}^n$, where $\varepsilon_k := \exp(-\exp(k)) \to 0$ if $k \to \infty$. As $v \in L^{0,0}(\Delta)$, we have $g \in z^{1-\frac{1}{s}} L^{0,0}(A^*) \subset L^{0,0}(A^*)$. Then $g \cdot \chi_k \to g$ in L^2 . As a holomorphic function, v is bounded in a neighborhood of 0. Therefore,

$$\begin{split} \|g\overline{\partial}\chi_k\|_{A^*}^2 &= \left\|g\cdot\frac{\tilde{\chi}_k'(\log|\log|z||)}{|z|\log|z|}\overline{\partial}|z|\right\|_{A^*\cap\Delta_{\varepsilon_k}^n}^2 \lesssim \left\|g\cdot\frac{1}{|z|\log|z|}\right\|_{A^*\cap\Delta_{\varepsilon_k}^n}^2 \\ &\sim \left\|v\cdot\frac{|t|^{s-1}}{|t|^s\log|t|^s}\right\|_{\Delta_{\varepsilon_k}}^2 \lesssim \left\|\frac{1}{|t|\log|t|}\right\|_{\Delta_{\varepsilon_k}}^2 = \int_{\Delta_{\varepsilon_k}}\frac{1}{|t|^2\log^2|t|}dV \\ &= 2\pi\int_0^{\varepsilon_k}\frac{\rho}{\rho^2\log^2\rho}d\rho \sim \left[-\frac{1}{\log\rho}\right]_0^{\varepsilon_k} \to 0, \text{ if } k \to \infty. \end{split}$$

Hence, $\overline{\partial}(g\chi_k) = g\overline{\partial}\chi_k \to 0 = \overline{\partial}_w g$ in L^2 . So, $g \in \operatorname{dom} \overline{\partial}_{s,w}$.

ii) $\pi^*(\ker \overline{\partial}_{s,w}) \subset \mathcal{O}_{L^2}(\Delta)$ (cf. the proof of Lem. 6.2 in [Rup14]): Let g be in $\ker \overline{\partial}_{s,w}$, i.e., there are g_j in $L^2(A^*)$ with $g_j \to g$, $\overline{\partial}g_j \to 0$ in $L^2(A^*)$ and $0 \notin \operatorname{supp} g_j$. Let $\chi \in \mathscr{C}^{\infty}_{\operatorname{cpt}}(\Delta, [0, 1])$ be identically 1 on $\Delta_{1/2}$. We define $u := \chi \pi^* g$ and $u_j := \chi \pi^* g_j$. It follows that $t^{s-1}u_j \to t^{s-1}u$ and $\overline{\partial}u_j \to \overline{\partial}u$ in $L^2(\Delta)$. Let $P: L^2(\Delta) \to L^2(\Delta)$ be the Cauchy-operator on the punctured disc, i.e.,

$$[P(h)](t) := \frac{1}{2\pi i} \int_{\Delta^*} \frac{h(\zeta)}{\zeta - t} d\zeta \wedge d\overline{\zeta}.$$

Since the support of u_j is away from 0 and $\partial \Delta$, we get $u_j = P\left(\frac{\partial u_j}{\partial \overline{\zeta}}\right)$. The L^2 -continuity of P and $\overline{\partial} u_j \to \overline{\partial} u$ in L^2 imply that

$$u_j = P\left(\frac{\partial u_j}{\partial \overline{\zeta}}\right) \to P\left(\frac{\partial u}{\partial \overline{\zeta}}\right)$$

in L^2 . Since t^{s-1} is bounded, we obtain $t^{s-1}u_j \to t^{s-1}P\left(\frac{\partial u}{\partial \overline{\zeta}}\right)$ and, hence, $u = P\left(\frac{\partial u}{\partial \overline{\zeta}}\right)$ in L^2 . That yields $\pi^*g \in L^2(\Delta)$. With $\pi^*g \in t^{1-s}\mathcal{O}_{L^2}(\Delta)$ and the extension theorem, we conclude $\pi^*g \in \mathcal{O}_{L^2}(\Delta)$. Second, we claim that $\ker(\overline{\partial}_{s,w}: L^{1,0}(A^*) \to L^{1,1}(A^*)) \cong t^{s-1}\Omega^1_{L^2}(\Delta)$. f = gdz is in $\ker\overline{\partial}_{s,w}$ iff $g \in \ker\overline{\partial}_{s,w}$. This is equivalent to $\pi^*g \in \mathcal{O}_{L^2}(\Delta)$. Since $\pi^*(dz) = t^{s-1}dt$, we infer that $\pi^*: H^{1,0}_{s,w}(A^*) \to t^{s-1}\Omega^1_{L^2}(\Delta), \pi^*f = t^{s-1}\pi^*gdt$ is an isomorphism.

The Riemann extension theorem (Theorem 2.5) implies $\mathcal{O}_{L^2}(\Delta) \cong \widehat{\mathcal{O}}_{L^2}(A)$ and the last statement.

Lemma 3.7. For $p \in \{0,1\}$, $\mathcal{R}(\overline{\partial}_{w,s}: L^{p,0}(A^*) \to L^{p,1}(A^*))$ and $\mathcal{R}(\overline{\partial}_{s,w}: L^{p,0}(A^*) \to L^{p,1}(A^*))$ both are closed, and

$$H^{p,1-q}_{w,s}(A^*) \cong H^{1-p,q}_{s,w}(A^*) \text{ for } q \in \{0,1\}.$$

Proof. Since $\overline{\partial}_{w,s}^* = \vartheta_{s,w}$, $\overline{\partial}_{s,w}^* = \vartheta_{w,s}$, $\vartheta_{s,w} = -\overline{*}\overline{\partial}_{s,w}\overline{*}$ and $\vartheta_{w,s} = -\overline{*}\overline{\partial}_{w,s}\overline{*}$ (see Lemma 2.1), it is easy to see that Lemma 3.3 holds for $\overline{\partial}_{w,s}$ and $\overline{\partial}_{s,w}$. We remark (cf. the proof of Lemma 3.4) that

$$\mathcal{R}(\overline{\partial}_{s,w}) \text{ is closed } \Leftrightarrow \mathcal{R}(\vartheta_{w,s}) \text{ is closed}$$
$$\Leftrightarrow \mathcal{R}(\overline{\partial}_{w,s}) \text{ is closed}$$

Therefore, it is enough to show that $\mathcal{R}(\overline{\partial}_{w,s})$ is closed.

Let $\varphi : \Delta^* \to \mathbb{R}$ be the smooth function defined by $\varphi(t) := (1-s) \log |t|^2$. Then

$$L^{p,q}(\Delta,\varphi) = t^{1-s} L^{p,q}(\Delta)$$

for the L^2 -space $L^{p,q}(\Delta, \varphi)$ with weight $e^{-\varphi}$.

We set $T_1 := \pi^* \overline{\partial}_{w,s} \iota^* : L^{0,0}(\Delta, \varphi) \to L^{0,1}(\Delta)$. The extension theorem implies that T_1 is the (strong) closure of $\overline{\partial}_{cpt} : L^{0,0}(\Delta, \varphi) \to L^{0,1}(\Delta)$. Therefore, T_1^* is the weak closed extension of $\vartheta_{cpt}^{\varphi} : L^{0,1}(\Delta) \to L^{0,0}(\Delta, \varphi)$ which is defined by

$$(\overline{\partial}_{\rm cpt}\alpha,\beta) = (\alpha,\vartheta_{\rm cpt}^{\varphi}\beta)_{\varphi} = \int \langle \alpha,\vartheta_{\rm cpt}^{\varphi}\beta \rangle e^{-\varphi} dV.$$

We set $\overline{*}_{\varphi} := e^{-\varphi} \overline{*}$. Then $T_2 := -\overline{*}_{\varphi} T_1^* \overline{*}$ is the weak closed extension of

$$\overline{\partial}_{\rm cpt} : L^{1,0}(\Delta) \to L^{1,1}(\Delta, -\varphi)$$

because integration by parts implies $\vartheta_{\rm cpt}^{\varphi} = -\overline{*}_{\varphi} \,\overline{\partial}_{\rm cpt} \,\overline{*}$:

$$(\alpha, \overline{\ast}_{-\varphi} \,\overline{\partial}_{\mathrm{cpt}} \,\overline{\ast} \,\beta)_{\varphi} = \int \alpha \wedge \overline{\ast}_{\varphi} \,\overline{\ast}_{-\varphi} \,\overline{\partial}_{\mathrm{cpt}} \,\overline{\ast} \,\beta = (-1)^{1-p} \int \alpha \wedge \overline{\partial}_{\mathrm{cpt}} \,\overline{\ast} \,\beta$$
$$= -\int \overline{\partial}_{\mathrm{cpt}} \alpha \wedge \overline{\ast} \,\beta = -(\overline{\partial}_{\mathrm{cpt}} \alpha, \beta).$$

Hence, T_2 is $\overline{\partial}_w : L^{1,0}(\Delta) \to L^{1,1}(\Delta, -\varphi)$ in sense of distributions. Since for all

$$u\in L^{1,1}(\Delta,-\varphi)=t^{s-1}L^{1,1}(\Delta)\subset L^{1,1}(\Delta)$$

there is a $v \in L^{1,0}(\Delta)$ with $T_2 v = \overline{\partial}_w v = u$, the range of T_2 is closed. Thus, the range of T_1^* and the range of $\overline{\partial}_{w,s} = \iota^* T_1 \pi^* : L^{0,0}(A^*) \to L^{0,1}(A^*)$ are closed as well.

We set $S_1 := \pi^* \overline{\partial}_{w,s} \iota^* : L^{1,0}(\Delta) \to L^{1,1}(\Delta, -\varphi)$ and $S_2 := -\overline{*} S_1^* \overline{*}_{\varphi}$. Then S_2 is the weak closure of $\overline{\partial}_{cpt} : L^{0,0}(\Delta, \varphi) \to L^{0,1}(\Delta)$.

$$\mathcal{R}(S_2) = \{ u \in L^{0,1}(\Delta) : \exists v \in L^{0,0}(\Delta, \varphi) = t^{1-s}L^{0,0}(\Delta) \text{ with } S_2v = u \}$$
$$\supset \{ u \in L^{0,1}(\Delta) : \exists v \in L^{0,0}(\Delta) \text{ with } \overline{\partial}_w v = u \} = L^{0,1}(\Delta).$$

Therefore, $\mathcal{R}(S_2) = L^{0,1}(\Delta)$ is closed. This implies the claim.

Summarizing, we computed (with $s = \text{mult}_0 A$):

$$\begin{aligned}
H_{w}^{0,0}(A^{*}) &\cong H_{s}^{1,1}(A^{*}) &\cong t^{1-s}\mathcal{O}_{L^{2}}(\Delta), \\
H_{w}^{1,0}(A^{*}) &\cong H_{s}^{0,1}(A^{*}) &\cong \Omega_{L^{2}}^{1}(\Delta), \\
H_{w}^{p,1}(A^{*}) &= H_{s}^{1-p,0}(A^{*}) &= 0, \\
H_{s,w}^{0,0}(A^{*}) &\cong H_{w,s}^{1,1}(A^{*}) &\cong \mathcal{O}_{L^{2}}(\Delta), \\
H_{s,w}^{1,0}(A^{*}) &\cong H_{w,s}^{0,1}(A^{*}) &\cong t^{s-1}\Omega_{L^{2}}^{1}(\Delta), \text{ and} \\
H_{s,w}^{p,1}(A^{*}) &= H_{w,s}^{1-p,0}(A^{*}) &= 0.
\end{aligned}$$
(3.8)

4. L^2 -cohomology of complex curves

We will prove Theorem 1.2 in this section. As a preparation, we consider the following local situation: Let A be a locally irreducible analytic set of dimension one in a domain $\Omega \in \mathbb{C}_{zw_1...w_{n-1}}^n$ with Sing $A = \{0\}$, let dV denote the volume form on $A^* := \operatorname{Reg} A$ which is induced by the Euclidean metric and let $z : A \to \mathbb{C}_z$ be the projection on the first coordinate. Let us mention (cf. e.g. Prop. in [Chi89, Sect. 8.1]):

Theorem 4.1. The set of all tangent vectors at a point of a one-dimensional irreducible analytic set in \mathbb{C}^n is a complex line.

Thus, we can assume that $C_0(A) = \mathbb{C}_z \times \{0\} \subset \mathbb{C}_z \times \mathbb{C}_{w_1 \dots w_{n-1}}^{n-1}$, and, therefore, $dV \sim dz \wedge d\overline{z}$ (by shrinking Ω if necessary).

Let $\pi : M \to A$ be a resolution of A, $x_0 := \pi^{-1}(0)$. Then $Z = (\pi^*(z))$ is the unreduced exceptional divisor of the resolution. After shrinking A and M again, we can assume that M is covered by a single chart $\psi : M \to \mathbb{C}$ with $x_0 \in M$ and $\psi(x_0) = 0$. We set $\zeta := \pi^*(z)$ and get $Z = (\zeta)$. $|Z| = (\psi)$ implies $Z - |Z| = (\frac{\zeta}{\psi})$. We obtain

$$\pi^*(dz) = d(\pi^*z) = \frac{\partial \zeta}{\partial \psi} d\psi \sim \frac{\zeta}{\psi} d\psi.$$

Therefore, $\pi^*(dV) \sim \left|\frac{\zeta}{\psi}\right|^2 d\psi \wedge d\overline{\psi}$, and we conclude (recall the definition of line bundles L_D from Section 2.2):

$$f \in L^{p,q}(A^*) \Leftrightarrow \left| \frac{\zeta}{\psi} \right|^{1-p-q} \cdot \pi^* f \in L^{p,q}(M)$$

$$\Leftrightarrow \pi^* f \in L^{p,q}(M, L_{(1-p-q)(Z-|Z|)}),$$

$$(4.2)$$

Nagase stated this equivalence already in Lem. 5.1 of [Nag90]. By use of the extension Theorem 2.4, we get:

$$f \in \operatorname{dom}\left(\overline{\partial}_{w}: L^{p,0}(A^{*}) \to L^{p,1}(A^{*})\right)$$

$$\Leftrightarrow \pi^{*}f \in \operatorname{dom}\left(\overline{\partial}_{w}: L^{p,0}(M, L_{(1-p)(Z-|Z|)}) \to L^{p,1}(M, L_{p(|Z|-Z)})\right).$$
(4.3)

The essential observation for the proof of Theorem 1.2 is the following:

Theorem 4.4. Let X be a compact complex curve and $L \to X$ a holomorphic line bundle. Let $\pi : M \to X$ be a resolution of X with exceptional divisor Z, and D a divisor on M such that $\pi^*L \cong L_D$, i. e., $\mathcal{O}(\pi^*L) \cong \mathcal{O}(D)$. Then

$$\begin{split} &H^{0,0}_w(X^*,L)\cong H^0(M,\mathcal{O}(Z-|Z|+D)),\\ &H^{0,1}_w(X^*,L)\cong H^1(M,\mathcal{O}(Z-|Z|+D)),\\ &H^{1,0}_w(X^*,L)\cong H^0(M,\Omega^1(D))\cong H^1(M,\mathcal{O}(-D)), \ and\\ &H^{1,1}_w(X^*,L)\cong H^1(M,\Omega^1(D))\cong H^0(M,\mathcal{O}(-D)). \end{split}$$

76

In [Par89, §5], Pardon proved that $H^{0,q}_{(2),sm}(X^*) \cong H^q(M, \mathcal{O}(Z - |Z|))$, where $H^{p,q}_{(2),sm}(X^*)$ denotes the $\overline{\partial}$ -cohomology with respect to smooth L^2 -forms. We will use similar arguments here.

Proof. Let x_0 be in Sing X, and let A be an open neighborhood of $x_0 = 0$ in X embedded locally in \mathbb{C}^n . We assume that $A = A_1 \cup ... \cup A_m$ with at x_0 irreducible analytic sets A_i . We obtain resolutions $\pi_i := \pi|_{\pi^{-1}(A_i)} : M_i \to A_i$ of A_i . The sets M_i are pairwise disjoint in M and, also, the support of the exceptional divisors Z_i of the resolution π_i . We get $Z|_{\pi^{-1}(A)} = \sum_{i=1}^m Z_i$ and $|Z||_{\pi^{-1}(A)} = \sum_{i=1}^m |Z_i|$. Therefore, the consideration in the local case (see (4.3)) implies that $\overline{\partial}_w : L^{p,0}(X^*, L) \to L^{p,1}(X^*, L)$ can be identified with

$$\overline{\partial}_w : L^{p,0}(M, L_{(1-p)(Z-|Z|)+D}) \to L^{p,1}(M, L_{p(|Z|-Z)+D}).$$

Hence,

$$H^{0,0}_w(X^*, L) \cong \ker(\overline{\partial}_w : L^{0,0}(M, L_{Z-|Z|+D}) \to L^{0,1}(M, L_D)) \cong H^0(M, \mathcal{O}(Z-|Z|+D))$$

and

$$H^{1,0}_w(X^*,L) \cong \ker(\overline{\partial}_w : L^{1,0}(M,L_D) \to L^{1,1}(M,L_{|Z|-Z+D})) \cong H^0(M,\Omega^1(D)).$$

Serre duality (see Theorem 2 in [Ser55, § 3.10]) implies

$$H^{1,0}_w(X^*, L) \cong H^0(M, \Omega^1(D)) \cong H^1(M, \mathcal{O}(-D)).$$

To prove the other two isomorphisms, consider the following general situation: Let E be a divisor on M, L_E the associated bundle, and let $\mathcal{L}_E^{p,q}$ denote the sheaf on M which is defined by $\mathcal{L}_E^{p,q}(U) := L_{\text{loc}}^{p,q}(U, L_E)$ for each open set $U \subset M$. Let $E' \leq E$ be another divisor. Consider the $\overline{\partial}$ -operator in the sense of distributions $\overline{\partial}_w : \mathcal{L}_E^{p,0} \to \mathcal{L}_{E'}^{p,1}$. Let $\mathcal{C}_{E,E'}^{p,0}$ denote the sheaf defined by

$$\mathcal{C}^{p,0}_{E,E'}(U) := \operatorname{dom} \left(\overline{\partial}_w : L^{p,0}_{\operatorname{loc}}(U, L_E) \to L^{p,1}_{\operatorname{loc}}(U, L_{E'}) \right).$$

Then $\mathcal{C}^{p,0}_{E,E'}$ is fine and, in particular, $H^1(M, \mathcal{C}^{p,0}_{E,E'}) = 0$. We get the sequence

$$0 \to \Omega^p(E) \to \mathcal{C}^{p,0}_{E,E'} \xrightarrow{\partial_w} \mathcal{L}^{p,1}_{E'} \to 0$$
(4.5)

which is exact by the usual Grothendieck-Dolbeault lemma because there is an embedding $\mathcal{L}_{E'}^{p,q} \subset \mathcal{L}_{E}^{p,q}$ (induced by the natural inclusion $\mathcal{O}(E') \subset \mathcal{O}(E)$, see (2.2)).

This induces the long exact sequence of cohomology groups

$$0 \to \Gamma(M, \Omega^p(E)) \to \mathcal{C}^{p,0}_{E,E'}(M) \xrightarrow{\overline{\partial}_w} \mathcal{L}^{p,1}_{E'}(M) \to H^1(M, \Omega^p(E)) \to H^1(M, \mathcal{C}^{p,0}_{E,E'}) = 0.$$

Hence, $\mathcal{L}_{E'}^{p,1}(M)/\overline{\partial}_w \mathcal{C}_{E,E'}^{p,0}(M) \cong H^1(M,\Omega^p(E))$. We conclude

$$H^{0,1}_w(X^*,L) \cong \mathcal{L}^{0,1}_D(M) / \overline{\partial}_w \mathcal{C}^{0,0}_{Z-|Z|+D,D}(M) \cong H^1(M, \mathcal{O}(Z-|Z|+D))$$

and, using the Serre duality again,

$$H^{1,1}_w(X^*,L) \cong \mathcal{L}^{1,1}_{|Z|-Z+D}(M)/\overline{\partial}_w \mathcal{C}^{1,0}_{D,|Z|-Z+D}(M) \cong H^1(M,\Omega^1(D)) \cong H^0(M,\mathcal{O}(-D)).$$

Theorem 1.2 follows now as a simple corollary by use of the classical Riemann-Roch theorem for each connected component of the Riemann surface M, keeping in mind that by definition g(M) = g(X), deg $L = \deg \pi^* L = \deg D$ and $\operatorname{mult}'_x X = \sum_{p \in \pi^{-1}(x)} \deg_p(Z - |Z|)$.

To deduce also a Riemann-Roch theorem for the $\overline{\partial}_s$ -cohomology, we can use the following L^2 -version of Serre duality:

Theorem 4.6. For each $p \in \{0,1\}$, the range of $\overline{\partial}_w : L^{p,0}(X^*,L) \to L^{p,1}(X^*,L)$ is closed. In particular, we get

$$H_w^{p,q}(X^*, L) \cong H_s^{1-p,1-q}(X^*, L^{-1}).$$

Proof. Recall the following well-known fact. If $P : H_1 \to H_2$ is a densely defined closed operator between Hilbert spaces with range $\mathcal{R}(P)$ of finite codimension, then the range $\mathcal{R}(P)$ is closed in H_2 (see e.g. [HL84], Appendix 2.4).

As M is compact, Theorem 4.4 implies particularly that the range of $\overline{\partial}_w$ is finite codimensional and, therefore, closed. Since $\overline{\partial}_s$ is the adjoint of $-\overline{*}\overline{\partial}_w\overline{*}$, the range of

$$\overline{\partial}_s: L^{1-p,0}(X^*, L^{-1}) \to L^{1-p,1}(X^*, L^{-1})$$

is closed as well. That both ranges are closed implies the L^2 -duality (cf. Lemma 3.3)

$$H^{p,q}_w(X^*,L) \cong \mathscr{H}^{p,q}_w(X^*,L) \cong \mathscr{H}^{1-p,1-q}_s(X^*,L^{-1}) \cong H^{1-p,1-q}_s(X^*,L^{-1}),$$

where $\mathscr{H}_{w/s}^{p,q}(X^*,L) := \ker \overline{\partial}_{w/s} \cap \ker \overline{\partial}_{w/s}^*$ denotes the space of $\overline{\partial}$ -harmonic forms with values in L.

Therefore, Theorem 4.4 yields:

$$\begin{aligned} H^{0,0}_{s}(X^{*},L) &\cong H^{1,1}_{w}(X^{*},L^{-1}) \cong H^{0}(M,\mathcal{O}(D)), \\ H^{0,1}_{s}(X^{*},L) &\cong H^{1,0}_{w}(X^{*},L^{-1}) \cong H^{1}(M,\mathcal{O}(D)), \\ H^{1,0}_{s}(X^{*},L) &\cong H^{0,1}_{w}(X^{*},L^{-1}) \cong H^{1}(M,\mathcal{O}(Z-|Z|-D)), \text{and} \\ H^{1,1}_{s}(X^{*},L) &\cong H^{0,0}_{w}(X^{*},L^{-1}) \cong H^{0}(M,\mathcal{O}(Z-|Z|-D)). \end{aligned}$$

$$(4.7)$$

Haskell computed $H^{0,q}_{\text{cpt}}(X^*) \cong H^q(M, \mathcal{O}_M)$, where $H^{p,q}_{\text{cpt}}(X^*)$ denotes the $\overline{\partial}$ -cohomology with respect to smooth forms with compact support (see Thm. 3.1 in [Has89]). From (4.7), we obtain the dual version of Theorem 1.2, i. e., the Riemann-Roch theorem for the $\overline{\partial}_s$ -cohomology:

Corollary 4.8 ($\overline{\partial}_s$ -Riemann-Roch). Let X be a compact complex curve with m irreducible components, $L \to X$ be a holomorphic line bundle, and $\pi : M \to X$ be a resolution of X. Then,

$$\begin{split} h_s^{0,0}(X^*,L) &- h_s^{0,1}(X^*,L) = m - g + \deg L, \text{and} \\ h_s^{1,1}(X^*,L) &- h_s^{1,0}(X^*,L) = m - g + \deg(Z - |Z|) - \deg L, \end{split}$$

where Z is the exceptional divisor of the resolution.

In [BPS90], Brüning, Peyerimhoff and Schröder proved that $h_s^{0,0}(X^*) - h_s^{0,1}(X^*) = m - g$ and $h_w^{0,0}(X^*) - h_w^{0,1}(X^*) = m - g + \deg Z - |Z|$ by computing the indices of the differential operators $\overline{\partial}_s$ and $\overline{\partial}_w$. Schröder generalized this result for vector bundles in [Sch89].

5. Weakly holomorphic functions

In this section, we will prove Theorem 1.4 by studying weakly holomorphic functions and a localized version of the $\overline{\partial}_s$ -operator.

Recalling the arguments at the beginning of Section 4, it is easy to see that the results of Section 3 generalize to arbitrary complex curves. In particular, the $\overline{\partial}_{s,w}$ -holomorphic functions on a singular complex curve are precisely the square-integrable weakly holomorphic functions (cf. Lemma 3.6), and the $\overline{\partial}_{s,w}$ -equation is locally solvable in the L^2 -sense (combine Lemma 3.5 and Lemma 3.7).

Let X be a singular complex curve, $\mathcal{L}_X^{p,q}$ the sheaf of locally square-integrable forms, and let

$$\overline{\partial}_w : \mathcal{L}^{p,0}_X \to \mathcal{L}^{p,1}_X$$

be the $\overline{\partial}$ -operator in the sense of distributions. For each open set $U \subset X$, we define $\overline{\partial}_{s,\text{loc}}$ on $L^{p,0}_{\text{loc}}(U)$ by $f \in \text{dom } \overline{\partial}_{s,\text{loc}}$ iff $f \in \text{dom } \overline{\partial}_w$ and $f \in \text{dom } (\overline{\partial}_{s,w} : L^{p,0}(V) \to L^{p,1}(V))$ for all $V \in U$ (for more details, see [Rup14, Sect. 6]). Let $\mathcal{F}^{p,0}_X$ be the sheaf of germs defined by $\mathcal{F}_X^{p,0}(U) := \operatorname{dom}\left(\overline{\partial}_{s,\operatorname{loc}}: L^{p,0}_{\operatorname{loc}}(U) \to L^{p,1}_{\operatorname{loc}}(U)\right)$, and let $\widehat{\mathcal{O}}_X$ denote the sheaf of germs of weakly holomorphic functions on X. Then our considerations above yield an exact sequence

$$0 \to \widehat{\mathcal{O}}_X = \ker \overline{\partial}_{s, \text{loc}} \hookrightarrow \mathcal{F}_X^{0, 0} \xrightarrow{\overline{\partial}_{s, \text{loc}}} \mathcal{L}_X^{0, 1} \to 0.$$
(5.1)

The sheaves $\mathcal{F}_X^{0,0}$ and $\mathcal{L}_X^{0,1}$ are fine and so (5.1) is a fine resolution of $\widehat{\mathcal{O}}_X$. Let $H_{s,\text{loc}}^{p,q}(X^*)$ denote the L^2_{loc} -Dolbeault cohomology on X^* with respect to the $\overline{\partial}_{s,\text{loc}}$ -operator. Using $\widehat{\mathcal{O}}_X = \pi_* \mathcal{O}_M$, we deduce from (5.1):

$$H^{0}(M, \mathcal{O}_{M}) \cong H^{0}(X, \widehat{\mathcal{O}}_{X}) = H^{0,0}_{s, \text{loc}}(X^{*}),$$

$$H^{1}(M, \mathcal{O}_{M}) \cong H^{1}(X, \widehat{\mathcal{O}}_{X}) \cong H^{0,1}_{s, \text{loc}}(X^{*}),$$

where $\pi: M \to X$ is a resolution of X. That proves Theorem 1.4.

6. Applications

There are many applications of the classical Riemann-Roch theorem; we will transfer two of them to our situation to exemplify how the L^2 -Riemann-Roch theorem can substitute the classical one on singular spaces.

6.1. Compact complex curves as covering spaces of \mathbb{CP}^1 . Let X be a compact irreducible complex curve with $\operatorname{Sing} X = \{x_1, ..., x_k\}$, let $(h_i)_{x_i} \in \mathcal{O}_{x_i}$ be chosen such that $(h_i)_{x_i} \widehat{\mathcal{O}}_{x_i} \subset \mathcal{O}_{x_i}$ and let $U_i \subset X$ be a (Stein) neighborhood of x_i with $h_i \cdot \widehat{\mathcal{O}}(U_i) \subset \mathcal{O}(U_i)$ (for the existence of the h_i , see e.g. Thm. 6 and its Cor. in [Nar66, §III.2]). Choose an $x_0 \in X^*$ and a (Stein) neighborhood U_0 of x_0 . We can assume that $U_0, ..., U_k$ are pairwise disjoint.

We define a line bundle $L \to X$ as follows. Let $U_{k+1} = X^* \setminus \{x_0\}$ and choose $f_0 \in \mathcal{O}(U_0)$ such that f_0 is vanishing to the order $r := \operatorname{ord}_{x_0} f_0 \ge 1$ in x_0 , which we will determine later, but has no other zeros. We also set $f_i := 1/h_i$ for i = 1, ..., k and $f_{k+1} = 1$, and consider the Cartier divisor $\{(U_i, f_i)\}_{i=0,...,k+1}$ on X. Let $L \to X$ be the line bundle associated to this divisor. As the f_i have no zeros for i > 0, there exists a non-negative integer δ such that deg $L = r - \delta$. Now choose $r := g(X) + \delta + 1$. It follows that deg L = g(X) + 1. Give L an arbitrary smooth Hermitian metric.

There is a canonical way to identify holomorphic sections of L with meromorphic functions on X. A holomorphic section $s \in \mathcal{O}(L)$ is represented by a tuple $\{s_i\}_i$ where $s_j/f_j = s_l/f_l$ on $U_j \cap U_l$. This gives a meromorphic function $\Psi(s)$ by setting $\Psi(s) := s_j/f_j$ on U_j . Note that $\Psi(s)$ has zeros in the singular points $x_1, ..., x_k$ and may have a pole of order r at $x_0 \notin \text{Sing } X$.

We can now apply our L^2 -Riemann-Roch theorem. The $\overline{\partial}_s$ -Riemann-Roch theorem, Corollary 4.8, implies dim $H_s^{0,0}(X^*, L) \geq 1 - g(X) + \deg L = 2$. Therefore, there is a section $\tau \in L^{0,0}(X^*, L)$ with $\overline{\partial}_s \tau = 0$ and τ_{k+1} is non-constant, where $\tau = \{\tau_i\}_{i=0,..,k+1}$ is written in the trivialization as above. This means that $\tau_i \in L^{0,0}(X^* \cap U_i), \overline{\partial}_s \tau_i = 0$, and $\tau_j/f_j = \tau_{k+1}$ is non-constant on $U_j \cap U_{k+1}$. Theorem 1.4 implies that $\tau_i \in \widehat{\mathcal{O}}(U_i), i = 1, ..., k + 1$. Now consider $\Psi(\tau)$ as defined above, i. e., $\Psi(\tau) = \tau_i/f_i$ on U_i . We conclude that $\Psi(\tau)/h_i \in \widehat{\mathcal{O}}(U_i)$, thus $\Psi(\tau) \in \mathcal{O}(U_i)$ for i = 1, ..., k. Moreover, $\Psi(\tau)$ is non-constant, so it cannot be holomorphic on the whole compact space X, thus must have a pole of some order $\leq r$ in x_0 . Thus:

$$\Psi(\tau): X \setminus \{x_0\} \to \mathbb{C}, \text{and}$$
$$\widetilde{\Psi}(\tau): X \to \mathbb{CP}^1, x \mapsto \begin{cases} [\Psi(\tau)(x): 1], & x \neq x_0\\ [1: \frac{1}{\Psi(\tau)(x)}], & x \in U_0 \end{cases}$$

are finite, open and, hence, analytic ramified coverings (Covering Lemma, see [GR84, Sect. VII.2.2]). In particular, $X \setminus \{x_0\}$ is Stein (use e.g. Thm. 1 in [GR79, §V.1]).

6.2. **Projectivity of compact complex curves.** A line bundle $L \to X$ on a compact complex space is called very ample if its global holomorphic sections induce a holomorphic embedding into the projective space \mathbb{CP}^N , i. e., if $s_0, ..., s_N$ is a basis of the space of holomorphic sections $\Gamma(X, \mathcal{O}(L))$, then the map

$$\Phi: X \to \mathbb{CP}^N , \ x \mapsto [s_0(x): \dots: s_N(x)], \tag{6.1}$$

given in local trivializations of the s_i , defines a holomorphic embedding of X in \mathbb{CP}^N . If some positive power of the line bundle has this property, then we say that it is ample. A compact complex space is called projective if there is an ample (and, hence, a very ample) line bundle on it.

A classical application of the Riemann-Roch theorem is that any compact Riemann surface is projective, and a line bundle on a Riemann surface is ample if its degree is positive (cf. e. g. [Nar92, Sect. 10]). This generalizes to singular complex curves:

Theorem 6.2. Let X be a compact locally irreducible complex curve. If $L \to X$ is a holomorphic line bundle with deg $L \gg 0$, then L is very ample. In particular, X is projective and each holomorphic line bundle on X is ample if its degree is positive.

Clearly, this result is well-known and follows from more general sheaf-theoretical methods (vanishing theorems) once one knows that L is positive iff deg L > 0 (cf. e. g. Thm. 4.4 in [Pet94, Sect. V.4.3] or Satz 2 in [Gra62, §3]). Nevertheless, it seems interesting to us to present another proof of Theorem 6.2 which is based on the L^2 -Riemann-Roch of singular complex curves. The assumption that X must be locally irreducible in Theorem 6.2 is not necessary. One can prove the result without this assumption easily by the same technique. Yet, to keep the notation simple, we present here only the locally irreducible case.

Let us make some preparations for the proof of Theorem 6.2. Let X be a connected complex curve and $\pi : M \to X$ a resolution of X. We choose a point $x_0 \in \text{Sing } X$ and a small neighborhood $U \subset X$ of x_0 with $U^* := U \setminus \{x_0\} \subset \text{Reg } X$. Assume X is irreducible at x_0 . We define $p_0 := \pi^{-1}(x_0), V := \pi^{-1}(U)$, and $V^* := V \setminus \{p_0\}$. We can assume that there is a chart $t : V \to \mathbb{C}$ such that the image of t is bounded.

The Riemann extension theorem implies that $\pi^{-1}: U \to V$ is weakly holomorphic or, briefly, $\tau := t \circ \pi^{-1} \in \widehat{\mathcal{O}}(U)$ (see Theorem 2.5). We show that τ generates the weakly holomorphic functions at x_0 in the following sense: Let $f \in \widehat{\mathcal{O}}(U)$. Then $f \circ \pi$ is holomorphic on V^* and bounded in p_0 . This implies that $f \circ \pi$ is holomorphic on V, $f \circ \pi(t) = \sum_{\iota=0}^{\infty} a_{\iota} t^{\iota}$, and $f(x) = \sum a_{\iota} \tau(x)^{\iota}$ (by shrinking U and V if necessary). This allows to define the order $\operatorname{ord}_{x_0} f$ of vanishing of f in x_0 by $r \in \mathbb{N}_0$ if $a_r \neq 0$ and $a_{\iota} = 0$ for $\iota < r$. In particular,

$$\operatorname{ord}_{x_0} f = \operatorname{ord}_{p_0}(f \circ \pi)$$

Note that this definition does not depend on the resolution as different resolutions are biholomorphically equivalent.

The L^2 -extension theorem (see Theorem 2.4) and (4.3) imply

$$f \in H^{0,0}_w(U) \Leftrightarrow t^{r_0} \cdot \pi^* f \in \mathcal{O}_{L^2}(V) \Leftrightarrow (\tau^{r_0} \cdot f \in \widehat{\mathcal{O}}(U) \text{ and } f \in L^2(U)),$$
(6.3)

where Z denotes the exceptional divisor of the resolution and $r_0 := \deg_{p_0}(Z - |Z|)$. In particular, we get the representation $f(x) = \sum_{\iota \ge -r_0} a_\iota \tau(x)^\iota$ and $\operatorname{ord}_{x_0} f := \operatorname{ord}_{p_0} \pi^* f \ge -r_0$ is again well-defined. f is weakly holomorphic iff $\operatorname{ord}_{x_0} f \ge 0$.

We denote by L_{x_0} the holomorphic line bundle on X which is trivial on $X \setminus \{x_0\}$ and is given by τ on U, i.e., the line bundle on X given by the open covering $U_1 := X \setminus \{x_0\}, U_0 := U$ and the transition function $g_{01} := \tau : U_0 \cap U_1 \to \mathbb{C}$. Then $\pi^* L_{x_0} \cong L_{p_0}$, where L_{p_0} is the holomorphic line bundle $L_{p_0} \to M$ associated to the divisor $\{p_0\}$.

line bundle $L_{p_0} \to M$ associated to the divisor $\{p_0\}$. Let $L \to X$ be any holomorphic line bundle, $L' := L \otimes L_{x_0}^{-1}$, and let s' be a section in $H_w^{0,0}(X^*, L')$. We can assume that L and L' are given by divisors $\{(U_j, f_j)\}$ and $\{(U_j, f'_j)\}$, respectively, where $\{U_j\}$ is an open covering of X with $U_0 = U$ and $x_0 \notin U_j$ for $j \neq 0$ and where $f_j, f'_j \in \mathcal{M}(U_j)$ with $g_{jk} := f_j/f_k$ and $g'_{jk} := f'_j/f'_k$ in $\mathcal{O}(U_j \cap U_k)$ $(g_{j,k}$ and g'_{jk} are the transition functions of L and L', respectively).

We get $f_0 = f'_0 \cdot \tau$ and $f_j = f'_j$ for $j \neq 0$. There is a meromorphic function $\tilde{s} := \Psi(s') \in \mathcal{M}(X)$ representing s'. This meromorphic function is defined by $\tilde{s} = s'_j/f'_j$ on U_j , where s'_j is the trivialization of s' on U_j . We can define a section $s = \{s_j\} \in H^{0,0}_w(X^*, L)$ by $s_j = \tilde{s} \cdot f_j$. Thus $s_0 = s'_0 \cdot \tau$ and $s_j = s'_j$ for $j \neq 0$. Hence, $\operatorname{ord}_{x_0} s_0 = \operatorname{ord}_{x_0} s'_0 + 1$. Summarizing, we get an injective linear map

$$T: H^{0,0}_w(X^*, L \otimes L^{-1}_{x_0}) \to H^{0,0}_w(X^*, L), \ s' \mapsto s,$$

which we call the *natural inclusion*. It follows from the construction above and by use of (6.3)that each section $s \in H^{0,0}_w(X^*, L)$ with $\operatorname{ord}_{x_0} s_0 > -r_0$ is in the image of T.

As $H^1(M, \mathcal{O}(D')) = 0$ for a divisor D' with deg D' > 2g - 2 by the classical Riemann-Roch theorem (cf. e.g. [Nar92, Sect. 10]), Theorem 4.4 – more precisely,

$$H^{0,1}_w(X^*, L) \cong H^1(M, \mathcal{O}(Z - |Z| + D))$$

- implies the following vanishing theorem.

Theorem 6.4. If $L \to X$ is a holomorphic line bundle on an irreducible compact complex curve X with deg $L > 2g - 2 - \sum_{x \in \text{Sing } X} \text{mult}'_x X$, then $H^{0,1}_w(X^*, L) = 0$.

As a preparation for the proof of Theorem 6.2, we get our main ingredient:

Lemma 6.5. Let $L \to X$ be a holomorphic line bundle on a connected compact locally irreducible complex curve X with deg $L > 2g - 1 - \sum_{x \in \text{Sing } X} \text{mult}'_x X$. Then the natural inclusion

$$T: H^{0,0}_w(X^*, L \otimes L^{-1}_{x_0}) \to H^{0,0}_w(X^*, L)$$

is not surjective. If deg $L > 2g + r_0 - 1 - \sum_{x \in \text{Sing } X} \text{mult}'_x X$, then there is a section $s \in H^{0,0}_w(X^*, L)$ which is weakly holomorphic on $U(x_0)$ and does not vanish in x_0 .

Recall that $r_0 = \operatorname{mult}_{x_0} X - 1 = \deg_{p_0}(Z - |Z|).$

Proof. i) As $\pi^*(L \otimes L_{x_0}^{-1}) \cong \pi^*L \otimes L_{p_0}^{-1}$, we get deg $L \otimes L_{x_0}^{-1} = \deg L - 1 > 2g - 2 - \deg(Z - |Z|)$. The $\overline{\partial}_w$ -Riemann-Roch theorem and $h_w^{0,1}(X^*, L) = 0 = h_w^{0,1}(X^*, L \otimes L_{x_0}^{-1})$ (using Theorem 6.4) yield

$$\begin{aligned} h_w^{0,0}(X^*, L \otimes L_{x_0}^{-1}) &= 1 - g + \deg(Z - |Z|) + \deg L \otimes L_{x_0}^{-1} \\ &< 1 - g + \deg(Z - |Z|) + \deg L = h_w^{0,0}(X^*, L). \end{aligned}$$

Therefore, the natural inclusion T cannot be surjective.

ii) The image of $T^{r_0}: H^{0,0}_w(X^*, L \otimes L^{-r_0}_{x_0}) \to H^{0,0}_w(X^*, L)$ are the sections s with $\operatorname{ord}_{x_0} s_0 \ge 0$, where s_0 is the trivialization of s over $U(x_0)$, i. e., the sections where s_0 is weakly holomorphic on $U(x_0)$. As $H^{0,0}_w(X^*, L \otimes L^{-r_0-1}_{x_0}) \to H^{0,0}_w(X^*, L \otimes L^{-r_0}_{x_0})$ is not surjective (use

$$\deg L \otimes L_{x_0}^{-r_0} = \deg L - r_0 > 2g - 1 - \deg(Z - |Z|)$$

and part (i)), there is a section

 $s' \in H^{0,0}_w(X^*, L \otimes L^{-r_0}_{x_0})$ with $\operatorname{ord}_{x_0} s'_0 = -r_0$ and $\operatorname{ord}_{x_0}(T^{r_0}(s'))_0 = 0$. So, $s := T^{r_0}(s')$ is the section of $H^{0,0}_w(X^*, L)$ we were looking for. \square

Proof of Theorem 6.2. Let X be a connected compact locally irreducible complex curve with Sing $X = \{x_1, ..., x_k\}$, and $L \to X$ a line bundle with deg $L \gg 0$. Following the classical arguments to show that the map Φ in (6.1) is a well-defined holomorphic embedding (see e.g. [Pet94, V.4, Thm. 4.4]), we have to prove:

(i) Φ is well-defined: For $x \in X$, there exists $s \in \Gamma(X, \mathcal{O}(L))$ such that $s(x) \neq 0$.

J. RUPPENTHAL AND M. SERA

- (ii) Φ is injective: For $x, y \in X, x \neq y$, there exists $s \in \Gamma(X, \mathcal{O}(L))$ such that $s(x) \neq 0$ and s(y) = 0.
- (iii) Φ is an immersion: For $x \in X$, the differential $T_x \Phi$ is injective.

Since (obviously) Φ is closed, (ii) and (iii) imply that Φ is an embedding (see e.g. Sect. 1.2.7 in [GR84]).

We will prove the statements (i) and (iii) for singular points $x \in \text{Sing } X$. The case of regular points is simpler and follows easily with the natural inclusion and Lemma 6.5. The statement (ii) can be seen just as (i) by imposing the additional condition that s(y) = 0 in what we do to prove the statement (i).

Let $\pi : M \to X$ be a resolution of singularities. Set $X^* = \operatorname{Reg} X$, $M^* = \pi^{-1}(X^*)$, $p_j := \pi^{-1}(x_j)$, and $r_j := \deg_{p_j}(Z - |Z|)$, where Z is the unreduced exceptional divisor of the resolution. Fix a $\mu \in \{1, ..., k\}$ and choose a neighborhood U_{μ} of x_{μ} such that there exist a resolution of the singularities $\pi : V_{\mu} \to U_{\mu}$ and a chart $t : V_{\mu} \to \mathbb{C}$ with $t \circ \pi^{-1}(x_{\mu}) = 0$, and set $\tau := t \circ \pi^{-1}$.

For each singularity x_j , we can choose a function $h_j \in \mathcal{O}(U_j)$ such that $h_j \cdot \widehat{\mathcal{O}}(U_j) \subset \mathcal{O}(U_j)$ for a neighborhood U_j of x_j small enough (see [Nar66, § III.2]). The number

$$\eta_j := \operatorname{ord}_{x_j} h_j$$

is important for our considerations because of the following fact. If f is a function on U_j with $\operatorname{ord}_{x_j} f \geq \eta_j$, then f/h_j is bounded at x_j ($\operatorname{ord}_{x_j} f/h_j \geq 0$); this implies $f/h_j \in \widehat{\mathcal{O}}(U_j)$ and, hence, $f \in \mathcal{O}(U_j)$. For the maximal ideal in \mathcal{O}_{X,x_j} , we get $\mathfrak{m}_{x_j} = \{f \in \mathcal{O}_{X,x_j} : \operatorname{ord}_{x_j} f > 0\}$ and $\{f \in \mathcal{O}_{X,x_j} : \operatorname{ord}_{x_j} f \geq 2\eta_j\} \subset \mathfrak{m}^2_{x_j}$.

We can choose a weakly holomorphic section $\sigma \in H^{0,0}_w(X^*, L)$ such that σ does not vanish in x_{μ} and $\operatorname{ord}_{x_j} \sigma \geq \eta_j$ for $j \neq \mu$. This section σ exists as we have the natural inclusion (see the construction above)

$$H^{0,0}_w\left(X^*, L \otimes L^{-r_\mu}_{x_\mu} \otimes \bigotimes_{j \neq \mu} L^{-\eta_j - r_j}_{x_j}\right) \to H^{0,0}_w(X^*, L),$$

and $\deg L \gg 0$ implies by Lemma 6.5 that the natural inclusion

$$H^{0,0}_w\left(X^*, L \otimes L^{-r_\mu-1}_{x_\mu} \otimes \bigotimes_{j \neq \mu} L^{-\eta_j-r_j}_{x_j}\right) \to H^{0,0}_w\left(X^*, L \otimes L^{-r_\mu}_{x_\mu} \otimes \bigotimes_{j \neq \mu} L^{-\eta_j-r_j}_{x_j}\right)$$

is not surjective.

Note that σ is holomorphic on $X - \{x_{\mu}\}$ but just weakly holomorphic in x_{μ} . We will now modify σ so that it becomes holomorphic and non-vanishing in x_{μ} . Shrink U_{μ} such that $\sigma = \sum_{\iota \geq 0} a_{\iota} \tau^{\iota}$ on U_{μ} with $a_0 \neq 0$. Let $\sigma' := \sigma/a_0$ so that $\operatorname{ord}_{x_{\mu}}(\sigma'-1) \geq 1$, i. e., $\sigma'-1 = \sum_{\iota \geq 1} a'_{\iota}\tau^{\iota}$ on U_{μ} . Choose as above a $\tilde{\sigma} \in H^{0,0}_w(X^*, L)$ with $\operatorname{ord}_{x_{\mu}} \tilde{\sigma} = 1$ and $\operatorname{ord}_{x_j} \tilde{\sigma} \geq \eta_j$ for $j \neq \mu$. Let $\tilde{\sigma} = \sum_{\iota \geq 1} \tilde{a}_{\iota}\tau^{\iota}$ close to x_{μ} with $\tilde{a}_1 \neq 0$. We define $\sigma'' := \sigma' - \frac{a'_1}{\tilde{a}_1}\tilde{\sigma}$. Then, $\operatorname{ord}_{x_{\mu}}(\sigma''-1) \geq 2$ and $\operatorname{ord}_{x_j} \sigma'' \geq \eta_j$ for $j \neq \mu$. We repeat this procedure recursively to get a section $\xi = \{\xi_j\} \in H^{0,0}_w(X^*, L)$ with $\operatorname{ord}_{x_{\mu}}(\xi_{\mu} - 1) \geq \eta_{\mu}$ and $\operatorname{ord}_{x_j} \xi_j \geq \eta_j$ for $j \neq \mu$. Thus, ξ is a holomorphic section on X, non-vanishing in x_{μ} . That shows (i) for $x = x_{\mu}$.

We will prove (iii) for x_{μ} . Let $v \in T_{x_{\mu}}X = (\mathfrak{m}_{x_{\mu}}/\mathfrak{m}_{x_{\mu}}^2)^*$ satisfy $v \neq 0$, i. e., there exists an $f \in \mathfrak{m}_{x_{\mu}}$ with $v(f + \mathfrak{m}_{x_{\mu}}^2) \neq 0$. We claim there exists a $g \in \mathfrak{m}_{\Phi(x_{\mu})}$ with $g \circ \Phi - f \in \mathfrak{m}_{x_{\mu}}^2$. Then $v(g \circ \Phi + \mathfrak{m}_{x_{\mu}}^2) = v(f + \mathfrak{m}_{x_{\mu}}^2) \neq 0$, i. e., $T_x \Phi(v) \neq 0$.

Proof of the claim: Replacing 1 with $f = \sum_{\iota \ge 1} f_{\iota} \tau^{\iota}$, we can repeat the procedure in (i) to construct a section $\xi = \{\xi_j\} \in H^{0,0}_w(X^*, L)$ with $\operatorname{ord}_{x_{\mu}}(\xi_{\mu} - f) \ge 2\eta_{\mu}$ and $\operatorname{ord}_{x_j} \xi_j \ge \eta_j$ for $j \ne \mu$. We get ξ is holomorphic, $\xi_{\mu} \in \mathfrak{m}_{x_{\mu}}$ and $\xi_{\mu} - f \in \mathfrak{m}^2_{x_{\mu}}$. Let Φ be defined by $\Phi(x) = [s_0(x):\ldots:s_N(x)]$ with holomorphic sections $s_i = \{s_{ij}\}$ (see (6.1)). Hence, we can choose a vector $(g_0, \ldots, g_N) \in \mathbb{C}^{N+1}$

82

such that $\xi = \sum_i g_i s_i$. Because of (i), there exits an i_0 such that $c := s_{i_0}(x_{\mu}) \neq 0$ – we can assume $i_0 = 0$. We set $U := \{x \in U_{\mu} : s_{0\mu}(x) \neq 0\}$ and identify $\{[t_0:...:t_N] : t_0 = 1\} \subset \mathbb{CP}^N$ with \mathbb{C}^N such that $\Phi|_U : U \to \mathbb{C}^N$ is defined by $\Phi(x) = \left(\frac{s_{1\mu}(x)}{s_{0\mu}(x)}, ..., \frac{s_{N\mu}(x)}{s_{0\mu}(x)}\right)$. Let $g : \mathbb{C}^N \to \mathbb{C}$ denote the holomorphic function $g(t_1, ..., t_N) := c \cdot (g_0 + \sum_{i=1}^N g_i t_i)$, i.e.,

$$s_{0\mu} \cdot (g \circ \Phi|_U) = c \sum_{i=0}^N g_i s_{i\mu} = c \cdot \xi_\mu$$

on U. Since $c = s_{0\mu}(x_{\mu}) \neq 0$ and since f and $\frac{c}{s_{0\mu}} - 1$ are in $\mathfrak{m}_{x_{\mu}}$, we get $g \in \mathfrak{m}_{\Phi(x_{\mu})}$ and

$$g \circ \Phi - f = \frac{c}{s_{0\mu}} \left(\xi_{\mu} - f\right) + f \cdot \left(\frac{c}{s_{0\mu}} - 1\right) \in \mathfrak{m}_{x_{\mu}}^{2}.$$

For this proof, L has to satisfy

$$\deg L > 2g + \max\{\eta_j\} + \sum_{j=1}^k (\eta_j + r_j) - \deg(Z - |Z|) = 2g + k + \max\{\eta_j\} + \sum_{j=1}^k \eta_j.$$

References

- [BPS90] Jochen Brüning, Norbert Peyerimhoff and Herbert Schröder, The ∂-operator on algebraic curves, Comm. Math. Phys. 129 (1990), no. 3, 525–534.
- [Chi89] Evgeni M. Chirka, Complex Analytic Sets, Mathematics and its Applications (Soviet Series), vol. 46, Kluwer Academic Publishers Group, Dordrecht, 1989.
- [Dem12] Jean-Pierre Demailly, Complex Analytic and Differential Geometry, Institut Fourier, Université de Grenoble I.
- [Gra62] Hans Grauert, Über Modifikationen und Exzeptionelle Analytische Mengen, Math. Ann. 146 (1962), 331–368. DOI: 10.1007/BF01441136
- [GR84] Hans Grauert and Reinhold Remmert, Coherent Analytic Sheaves, Grundlehren der Mathematischen Wissenschaften, no. 265, Springer, Berlin, 1984.
- [GR79] _____, Theory of Stein Spaces, Grundlehren der Mathematischen Wissenschaften, vol. 236, Springer, Berlin, 1979, Translated from the German by Alan Huckleberry.
- [GH78] Phillip Griffiths and Joseph Harris, Principles of Algebraic Geometry, Wiley-Interscience [John Wiley & Sons], New York, 1978, Pure and Applied Mathematics.
- [Has89] Peter Haskell, L²-Dolbeault complexes on singular curves and surfaces, Proc. Amer. Math. Soc. 107 (1989), no. 2, 517–526.
- [HL84] Gennadi Henkin and Jürgen Leiterer, Theory of functions on complex manifolds, Monographs in Mathematics, vol. 79, Birkhäuser Verlag, Basel, 1984.
- [Hir77] Heisuke Hironaka, Bimeromorphic smoothing of a complex-analytic space, Acta Math. Vietnam. 2 (1977), no. 2, 103–168.
- [Hir64] _____, Resolution of singularities of an algebraic variety over a field of characteristic zero I, II, Ann. of Math. (2) 79 (1964), 109–203, 205–326. DOI: 10.2307/1970486
- [LM02] Ingo Lieb and Joachim Michel, The Cauchy-Riemann Complex, Aspects of Mathematics, no. E 34, Vieweg, Braunschweig, 2002. DOI: 10.1007/978-3-322-91608-2
- [Nag90] Masayoshi Nagase, Remarks on the L²-Dolbeault cohomology groups of singular algebraic surfaces and curves, Publ. Res. Inst. Math. Sci. 26 (1990), no. 5, 867–883.
- [Nar92] Raghavan Narasimhan, Compact Riemann Surfaces, Lectures in Mathematics ETH Zürich, Birkhäuser Verlag, Basel, 1992.
- [Nar66] _____, Introduction to the Theory of Analytic Spaces, Lecture Notes in Mathematics, No. 25, Springer-Verlag, Berlin, 1966.
- [ØV13] Nils Øvrelid and Sophia Vassiliadou, L²-∂-cohomology groups of some singular complex spaces, Invent. Math. 192 (2013), no. 2, 413–458.
- [Pet94] Thomas Peternell, Pseudoconvexity, the Levi problem and vanishing theorems, Several complex variables, VII, Encyclopaedia Math. Sci., vol. 74, Springer, Berlin, 1994, pp. 221–257. DOI: 10.1007/978-3-662-09873-8_6
- [Par89] William L. Pardon, The L_2 - $\bar{\partial}$ -cohomology of an algebraic surface, Topology **28** (1989), no. 2, 171–195.
- [PS91] William L. Pardon and Mark A. Stern, L²-∂-cohomology of complex projective varieties, J. Amer. Math. Soc. 4 (1991), no. 3, 603–621.

J. RUPPENTHAL AND M. SERA

- [Rup09] Jean Ruppenthal, About the $\overline{\partial}$ -equation at isolated singularities with regular exceptional set, Internat. J. Math. **20** (2009), no. 4, 459–489.
- [Rup14] $_$, L^2 -theory for the $\overline{\partial}$ -operator on compact complex spaces, Duke Math. J. **163** (2014), no. 15, 2887–2934.
- [Rup11] _____, L²-theory for the $\overline{\partial}$ -operator on complex spaces with isolated singularities, Preprint 2011. arXiv: 1110.2373
- [Sch89] Herbert Schröder, *The Riemann-Roch theorem on algebraic curves*, Séminaire de Théorie Spectrale et Géométrie, No. 7, Année 1988–1989, Univ. Grenoble I, Saint-Martin-d'Hères, 1989, pp. 115–121.
- [Ser55] Jean-Pierre Serre, Un théorème de dualité, Comm. Math. Helv. 29 (1955), 9–26.

DEPARTMENT OF MATHEMATICS, UNIVERSITY OF WUPPERTAL, GAUSS STR. 20, 42119 WUPPERTAL, GERMANY. *E-mail address:* ruppenthal@uni-wuppertal.de, sera@math.uni-wuppertal.de

SYMMETRIES AND STABILIZATION FOR SHEAVES OF VANISHING CYCLES

C. BRAV, V. BUSSI, D. DUPONT, D. JOYCE, AND B. SZENDRŐI, WITH AN APPENDIX BY JÖRG SCHÜRMANN

ABSTRACT. We study symmetries and stabilization properties of perverse sheaves of vanishing cycles $\mathcal{PV}_{U,f}^{\bullet}$ of a regular function $f: U \to \mathbb{C}$ on a smooth \mathbb{C} -scheme U, with critical locus $X = \operatorname{Crit}(f)$. We prove four main results:

(a) If $\Phi: U \to U$ is an isomorphism fixing X and compatible with f, then the action of Φ_* on $\mathcal{PV}_{U,f}^{\bullet}$ is multiplication by det $(d\Phi|_{X^{red}}) = \pm 1$.

(b) $\mathcal{PV}_{U,f}^{\bullet}$ depends up to canonical isomorphism only on $(X^{(3)}, f^{(3)})$, for $X^{(3)}$ the third-order thickening of X in U, and $f^{(3)} = f|_{X^{(3)}} : X^{(3)} \to \mathbb{C}$.

(c) If U, V are smooth \mathbb{C} -schemes, $f: U \to \mathbb{C}, g: V \to \mathbb{C}$ are regular, $X = \operatorname{Crit}(f), Y = \operatorname{Crit}(g)$, and $\Phi: U \to V$ is an embedding with $f = g \circ \Phi$ and $\Phi|_X: X \to Y$ an isomorphism, there is a natural isomorphism $\Theta_{\Phi}: \mathcal{PV}_{U,f}^{\bullet} \to \Phi|_X^*(\mathcal{PV}_{V,g}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi}$, for P_{Φ} a principal $\mathbb{Z}/2\mathbb{Z}$ -bundle on X.

(d) If (X, s) is an oriented d-critical locus in the sense of Joyce [23], there is a natural perverse sheaf $P_{X,s}^{\bullet}$ on X, such that if (X, s) is locally modelled on $\operatorname{Crit}(f : U \to \mathbb{C})$ then $P_{X,s}^{\bullet}$ is locally modelled on $\mathcal{PV}_{U,f}^{\bullet}$.

We also generalize our results to replace U, X by complex analytic spaces, and $\mathcal{PV}_{U,f}^{\bullet}$ by \mathscr{D} -modules or mixed Hodge modules.

We discuss applications of (d) to categorifying Donaldson–Thomas invariants of Calabi– Yau 3-folds, and to defining a 'Fukaya category' of Lagrangians in a complex symplectic manifold using perverse sheaves.

CONTENTS

1.	Inti	roduction	86
2.	Bac	ckground on perverse sheaves	88
	2.1.	Constructible complexes on \mathbb{C} -schemes	89
	2.2.	Perverse sheaves on \mathbb{C} -schemes	90
	2.3.	Nearby cycles and vanishing cycles on C-schemes	93
	2.4.	Perverse sheaves of vanishing cycles on \mathbb{C} -schemes	94
	2.5.	Summary of the properties we use in this paper	98
	2.6.	Perverse sheaves on complex analytic spaces	99
	2.7.	\mathscr{D} -modules on \mathbb{C} -schemes and complex analytic spaces	100
	2.8.	Mixed Hodge modules: basics	100
	2.9.	Monodromic mixed Hodge modules	102
	2.10.	Mixed Hodge modules of vanishing cycles	104
3.	Act	tion of symmetries on vanishing cycles	106
	3.1.	Proof of Proposition 3.4	108
	3.2.	Part (a): $\det(d\Psi _{Y^{\text{red}}}^{-1} \circ d\Phi _{X^{\text{red}}}) = \pm 1$	113
	3.3.	Part (b): $\mathcal{PV}_{\Phi} = \det(d\Psi _{\mathbf{Y}_{red}}^{-1} \circ d\Phi _{\mathbf{X}_{red}}) \cdot \mathcal{PV}_{\Psi}$	114
	3.4.	Demodules and mixed Hodge modules	117

4. Dep	pendence of $\mathcal{PV}_{U,f}^{\bullet}$ on f						
4.1.	Proof of Proposition 4.3						
4.2.	Proof of Theorem 4.2 for \mathbb{C} -schemes						
4.3.	<i>Image</i> -modules and mixed Hodge modules						
5. Sta	bilizing vanishing cycles						
5.1.	Theorem 5.4(a): the isomorphism Θ_{Φ}	130					
5.2.	Theorem 5.4(b): Θ_{Φ} depends only on $\Phi _X : X \to Y$						
5.3.	Theorem 5.4(c): composition of the Θ_{Φ}	134					
5.4.	<i>Image</i> -modules and mixed Hodge modules						
6. Per	verse sheaves on oriented d-critical loci						
6.1.	Background material on d-critical loci						
6.2.	The main result, and applications						
6.3.	Proof of Theorem 6.9 for C-schemes						
6.4.	<i>Image</i> -modules and mixed Hodge modules						
Append	lix A. Compatibility results, by Jörg Schürmann						
References							

1. INTRODUCTION

Let U be a smooth \mathbb{C} -scheme and $f: U \to \mathbb{C}$ a regular function, and write $X = \operatorname{Crit}(f)$, as a \mathbb{C} -subscheme of U. Then one can define the *perverse sheaf of vanishing cycles* $\mathcal{PV}_{U,f}^{\bullet}$ on X. Formally, $X = \coprod_{c \in f(X)} X_c$, where $X_c \subseteq X$ is the open and closed \mathbb{C} -subscheme of points $x \in X$ with f(x) = c, and

$$\mathcal{PV}_{U,f}^{\bullet}|_{X_c} = \phi_{f-c}^p(A_U[\dim U])|_{X_c}$$

for each $c \in f(X)$, where $A_U[\dim U]$ is the constant perverse sheaf on U over a base ring A, and

$$\phi_{f-c}^p : \operatorname{Perv}(U) \longrightarrow \operatorname{Perv}(f^{-1}(c))$$

is the vanishing cycle functor for $f - c : U \to \mathbb{C}$. See §2 for an introduction to perverse sheaves, and an explanation of this notation.

This paper will prove four main results, Theorems 3.1, 4.2, 5.4 and 6.9. The first three give properties of the $\mathcal{PV}_{U,f}^{\bullet}$, which we may summarize as follows:

(a) Let U, f, X be as above, and write X^{red} for the reduced \mathbb{C} -subscheme of X. Suppose $\Phi: U \to U$ is an isomorphism with $f \circ \Phi = f$ and $\Phi|_X = \text{id}_X$. Then Φ induces a natural isomorphism $\Phi_*: \mathcal{PV}_{U,f}^{\bullet} \to \mathcal{PV}_{U,f}^{\bullet}$.

Theorem 3.1 implies that $d\Phi|_{TU|_{X^{red}}}: TU|_{X^{red}} \to TU|_{X^{red}}$ has determinant

$$\det(\mathrm{d}\Phi|_{X^{\mathrm{red}}}): X^{\mathrm{red}} \longrightarrow \mathbb{C} \setminus \{0\},\$$

which is a locally constant map $X^{\text{red}} \to \{\pm 1\}$, and $\Phi_* : \mathcal{PV}^{\bullet}_{U,f} \to \mathcal{PV}^{\bullet}_{U,f}$ is multiplication by $\det(d\Phi|_{X^{\text{red}}})$.

In fact Theorem 3.1 proves a more complicated statement, which only requires Φ to be defined étale locally on U.

(b) Let U, f, X be as above, and write $I_X \subseteq \mathcal{O}_U$ for the sheaf of ideals of regular functions $U \to \mathbb{C}$ vanishing on X. For each $k = 1, 2, \ldots$, write $X^{(k)}$ for the k^{th} order thickening of X in U, that is, $X^{(k)}$ is the closed \mathbb{C} -subscheme of U defined by the vanishing of the sheaf of ideals I_X^k in \mathcal{O}_U . Write $f^{(k)} := f|_{X^{(k)}} : X^{(k)} \to \mathbb{C}$.

0	0
8	b
-	~

Theorem 4.2 says that the perverse sheaf $\mathcal{PV}_{U,f}^{\bullet}$ depends only on the third-order thickenings $(X^{(3)}, f^{(3)})$ up to canonical isomorphism.

As in Remark 4.5, étale locally, $\mathcal{PV}_{U,f}^{\bullet}$ depends only on $(X^{(2)}, f^{(2)})$ up to noncanonical isomorphism, with isomorphisms natural up to sign.

(c) Let U, V be smooth \mathbb{C} -schemes, $f: U \to \mathbb{C}$, $g: V \to \mathbb{C}$ be regular, and $X = \operatorname{Crit}(f)$, $Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V. Let $\Phi: U \hookrightarrow V$ be a closed embedding of \mathbb{C} schemes with $f = g \circ \Phi: U \to \mathbb{C}$, and suppose $\Phi|_X: X \to Y$ is an isomorphism. Then
Theorem 5.4 constructs a natural isomorphism of perverse sheaves on X:

$$\Theta_{\Phi}: \mathcal{PV}_{U,f}^{\bullet} \longrightarrow \Phi|_X^* \left(\mathcal{PV}_{V,g}^{\bullet} \right) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi}, \tag{1.1}$$

where $\pi_{\Phi} : P_{\Phi} \to X$ is a certain principal $\mathbb{Z}/2\mathbb{Z}$ -bundle on X. Writing N_{UV} for the normal bundle of U in V, then the Hessian Hess g induces a nondegenerate quadratic form q_{UV} on $N_{UV}|_X$, and P_{Φ} parametrizes square roots of $\det(q_{UV}) : K_U^2|_X \to \Phi|_X^*(K_V^2)$.

Theorem 5.4 also shows that the Θ_{Φ} in (1.1) are functorial in a suitable sense under compositions of embeddings $\Phi: U \hookrightarrow V, \Psi: V \hookrightarrow W$.

Here (c) is proved by showing that étale locally there exist equivalences $V \simeq U \times \mathbb{C}^n$ identifying $\Phi(U)$ with $U \times \{0\}$ and $g: V \to \mathbb{C}$ with $f \boxplus z_1^2 + \cdots + z_n^2: U \times \mathbb{C}^n \to \mathbb{C}$, and applying étale local isomorphisms of perverse sheaves

$$\mathcal{PV}_{U,f}^{\bullet} \cong \mathcal{PV}_{U,f}^{\bullet} \boxtimes \mathcal{PV}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}^{\bullet} \cong \mathcal{PV}_{U \times \mathbb{C}^n, f \boxplus z_1^2 + \dots + z_n^2}^{\bullet} \cong \mathcal{PV}_{V,g}^{\bullet},$$

using $\mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} \cong A_{\{0\}}$ in the first step, and the Thom–Sebastiani Theorem for perverse sheaves in the second.

Passing from $f: U \to \mathbb{C}$ to $g = f \boxplus z_1^2 + \cdots + z_n^2 : U \times \mathbb{C}^n \to \mathbb{C}$ is an important idea in singularity theory, as in Arnold et al. [1] for instance. It is known as *stabilization*, and f and g are called *stably equivalent*. So, Theorem 5.4 concerns the behaviour of perverse sheaves of vanishing cycles under stabilization.

Our fourth main result, Theorem 6.9, concerns a new class of geometric objects called *d-critical* loci, introduced in Joyce [23], and explained in §6.1. An (algebraic) d-critical locus (X, s) over \mathbb{C} is a \mathbb{C} -scheme X with a section s of a certain natural sheaf \mathcal{S}_X^0 on X. A d-critical locus (X, s)may be written Zariski locally as a critical locus $\operatorname{Crit}(f : U \to \mathbb{C})$ of a regular function f on a smooth \mathbb{C} -scheme U, and s records some information about U, f (in the notation of (b) above, s remembers $f^{(2)}$). There is also a complex analytic version.

Algebraic d-critical loci are classical truncations of the *derived critical loci* (more precisely, -1-shifted symplectic derived schemes) introduced in derived algebraic geometry by Pantev, Toën, Vaquié and Vezzosi [42]. Theorem 6.9 roughly says that if (X, s) is an algebraic d-critical locus over \mathbb{C} with an 'orientation', then we may define a natural perverse sheaf $P_{X,s}^{\bullet}$ on X, such that if (X, s) is locally modelled on $\operatorname{Crit}(f : U \to \mathbb{C})$ then $P_{X,s}^{\bullet}$ is locally modelled on $\mathcal{PV}_{U,f}^{\bullet}$. The proof uses Theorem 5.4.

These results have exciting applications in the categorification of Donaldson–Thomas theory on Calabi–Yau 3-folds, and in defining a new kind of 'Fukaya category' of complex Lagrangians in complex symplectic manifold, which we will discuss at length in Remarks 6.14 and 6.15.

Although we have explained our results only for \mathbb{C} -schemes and perverse sheaves upon them, the proofs are quite general and work in several contexts:

- (i) Perverse sheaves on C-schemes or complex analytic spaces with coefficients in any wellbehaved commutative ring A, such as Z, Q or C.
- (ii) \mathscr{D} -modules on \mathbb{C} -schemes or complex analytic spaces.
- (iii) Saito's mixed Hodge modules on C-schemes or complex analytic spaces.

We discuss all these in §2, before proving our four main results in §3–§6. Appendix A, by Jörg Schürmann, proves two compatibility results between duality and Thom–Sebastiani type isomorphisms needed in the main text.

This is one of six linked papers [6, 9-11, 23], with more to come. The best logical order is that the first is Joyce [23] defining d-critical loci, and the second Bussi, Brav and Joyce [9], which proves Darboux-type theorems for the k-shifted symplectic derived schemes of Pantev et al. [42], and defines a truncation functor from -1-shifted symplectic derived schemes to algebraic d-critical loci.

This paper is the third in the sequence. Combining our results with [23, 42] gives new results on categorifying Donaldson–Thomas invariants of Calabi–Yau 3-folds, as in Remark 6.14. In the fourth paper Bussi, Joyce and Meinhardt [11] will generalize the ideas of this paper to motivic Milnor fibres (we explain the relationship between the motivic and cohomological approaches below in Remark 6.10), and deduce new results on motivic Donaldson–Thomas invariants using [23,42]. In the fifth, Ben-Bassat, Brav, Bussi and Joyce [6] generalize [9,11] and this paper from (derived) schemes to (derived) Artin stacks.

Sixthly, Bussi [10] will show that if (S, ω) is a complex symplectic manifold, and L, M are complex Lagrangians in S, then the intersection $X = L \cap M$, as a complex analytic subspace of S, extends naturally to a complex analytic d-critical locus (X, s). If the canonical bundles K_L, K_M have square roots $K_L^{1/2}, K_M^{1/2}$ then (X, s) is oriented, and so Theorem 6.9 below defines a perverse sheaf $P_{L,M}^{\bullet}$ on X, which Bussi also constructs directly.

As in Remark 6.15, we hope in future work to define a 'Fukaya category' of complex Lagrangians in (X, ω) in which $\operatorname{Hom}(L, M) \cong \mathbb{H}^{-n}(P^{\bullet}_{L,M})$.

Conventions. All \mathbb{C} -schemes are assumed separated and of finite type. All complex analytic spaces are Hausdorff and locally of finite type.

Acknowledgements. We would like to thank Oren Ben-Bassat, Alexandru Dimca, Young-Hoon Kiem, Jun Li, Kevin McGerty, Sven Meinhardt, Pierre Schapira, and especially Morihiko Saito and Jörg Schürmann for useful conversations and correspondence, and Jörg Schürmann for a very careful reading of our manuscript, leading to many improvements, as well as providing the Appendix. This research was supported by EPSRC Programme Grant EP/I033343/1.

2. Background on perverse sheaves

Perverse sheaves, and the related theories of \mathscr{D} -modules and mixed Hodge modules, make sense in several contexts, both algebraic and complex analytic:

- (a) Perverse sheaves on C-schemes with coefficients in a ring A (usually Z, Q or C), as in Beilinson, Bernstein and Deligne [5] and Dimca [14].
- (b) Perverse sheaves on complex analytic spaces with coefficients in a ring A (usually Z, Q or C), as in Dimca [14].
- (c) \mathscr{D} -modules on \mathbb{C} -schemes, as in Borel [8] in the smooth case, and Saito [48] in general.
- (d) D-modules on complex manifolds as in Björk [7], and on complex analytic spaces as in Saito [48].
- (e) Mixed Hodge modules on \mathbb{C} -schemes, as in Saito [45, 47].
- (f) Mixed Hodge modules on complex analytic spaces, as in Saito [45, 47].

All our main results and proofs work, with minor modifications, in all six settings (a)–(f). As (a) is arguably the simplest and most complete theory, we begin in $\S2.1-\S2.4$ with a general introduction to constructible complexes and perverse sheaves on \mathbb{C} -schemes, the nearby and

vanishing cycle functors, and perverse sheaves of vanishing cycles $PV_{U,f}^{\bullet}$ on \mathbb{C} -schemes, following Dimca [14].

Several important properties of perverse sheaves in (a) either do not work, or become more complicated, in settings (b)–(f). Section 2.5 lists the parts of $\S2.1-\S2.4$ that we will use in proofs in this paper, so the reader can check that they do work in (b)–(f). Then $\S2.6-\S2.10$ give brief discussions of settings (b)–(f), focussing on the differences with (a) in $\S2.1-\S2.4$.

A good introductory reference on perverse sheaves on \mathbb{C} -schemes and complex analytic spaces is Dimca [14]. Three other books are Kashiwara and Schapira [27], Schürmann [50], and Hotta, Tanisaki and Takeuchi [21]. Massey [36] and Rietsch [43] are surveys on perverse sheaves, and Beilinson, Bernstein and Deligne [5] is an important primary source, who cover both \mathbb{Q} -perverse sheaves on \mathbb{C} -schemes as in (a), and \mathbb{Q}_l -perverse sheaves on \mathbb{K} -schemes as in (g) below.

Remark 2.1. Two further possible settings, in which not all the results we need are available in the literature, are the following.

- (g) Perverse sheaves on K-schemes with coefficients in $\mathbb{Z}/l^n\mathbb{Z}$, \mathbb{Z}_l , \mathbb{Q}_l , or $\overline{\mathbb{Q}}_l$ for $l \neq \operatorname{char} \mathbb{K} \neq 2$ a prime, as in Beilinson et al. [5].
- (h) \mathscr{D} -modules on \mathbb{K} -schemes for \mathbb{K} an algebraically closed field, as in Borel [8].

The issue is that the Thom–Sebastiani theorem is not available in these contexts in the generality we need it. Once an appropriate form of this result becomes available, our main theorems will hold also in these two contexts, sometimes under the further assumption that char $\mathbb{K} = 0$, needed for the results quoted from [9, 42]. We leave the details to the interested reader.

2.1. Constructible complexes on \mathbb{C} -schemes. We begin by discussing constructible complexes, following Dimca [14, §2–§4].

Definition 2.2. Fix a well-behaved commutative base ring A (where 'well-behaved' means that we need assumptions on A such as A is regular noetherian, of finite global dimension or finite Krull dimension, a principal ideal domain, or a Dedekind domain, at various points in the theory), to study sheaves of A-modules. For some results A must be a field. Usually we take $A = \mathbb{Z}, \mathbb{Q}$ or \mathbb{C} .

Let X be a \mathbb{C} -scheme, always assumed of finite type. Write X^{an} for the set of \mathbb{C} -points of X with the complex analytic topology. Consider sheaves of A-modules S on X^{an} . A sheaf S is called (*algebraically*) constructible if all the stalks S_x for $x \in X^{\mathrm{an}}$ are finite type A-modules, and there is a finite stratification $X^{\mathrm{an}} = \coprod_{j \in J} X_j^{\mathrm{an}}$ of X^{an} , where $X_j \subseteq X$ for $j \in J$ are \mathbb{C} -subschemes of X and $X_j^{\mathrm{an}} \subseteq X^{\mathrm{an}}$ the corresponding subsets of \mathbb{C} -points, such that $S|_{X_j^{\mathrm{an}}}$ is an A-local system for all $j \in J$.

Write D(X) for the derived category of complexes \mathcal{C}^{\bullet} of sheaves of A-modules on X^{an} . Write $D_c^b(X)$ for the full subcategory of bounded complexes \mathcal{C}^{\bullet} in D(X) whose cohomology sheaves $\mathcal{H}^m(\mathcal{C}^{\bullet})$ are constructible for all $m \in \mathbb{Z}$. Then $D(X), D_c^b(X)$ are triangulated categories. An example of a constructible complex on X is the *constant sheaf* A_X on X with fibre A at each point.

Grothendieck's "six operations on sheaves" $f^*, f^!, Rf_*, Rf_!, \mathcal{R}Hom, \overset{L}{\otimes}$ act on D(X) preserving the subcategory $D_c^b(X)$. That is, if $f: X \to Y$ is a morphism of \mathbb{C} -schemes, then we have two different pullback functors $f^*, f^!: D(Y) \to D(X)$, which also map $D_c^b(Y) \to D_c^b(X)$. Here f^* is called the *inverse image* [14, §2.3], and $f^!$ the *exceptional inverse image* [14, §3.2].

We also have two different pushforward functors

$$Rf_*, Rf_! : D(X) \longrightarrow D(Y)$$

mapping $D_c^b(X) \to D_c^b(Y)$, where Rf_* is called the *direct image* [14, §2.3] and is right adjoint to $f^*: D(Y) \to D(X)$, and $Rf_!$ is called the *direct image with proper supports* [14, §2.3] and is left adjoint to $f^!: D(Y) \to D(X)$. We need the assumptions from §1 that X, Y are separated and of finite type for $Rf_*, Rf_!: D_c^b(X) \to D_c^b(Y)$ to be defined for arbitrary morphisms $f: X \to Y$.

For $\mathcal{B}^{\bullet}, \mathcal{C}^{\bullet}$ in $D_c^b(X)$, we may form their derived Hom $\mathcal{R}Hom(\mathcal{B}^{\bullet}, \mathcal{C}^{\bullet})$ [14, §2.1], and left derived tensor product $\mathcal{B}^{\bullet} \overset{L}{\otimes} \mathcal{C}^{\bullet}$ in $D_c^b(X)$, [14, §2.2]. Given $\mathcal{B}^{\bullet} \in D_c^b(X)$ and $\mathcal{C}^{\bullet} \in D_c^b(Y)$, we define $\mathcal{B}^{\bullet} \overset{L}{\boxtimes} \mathcal{C}^{\bullet} = \pi_X^*(\mathcal{B}^{\bullet}) \overset{L}{\otimes} \pi_Y^*(\mathcal{C}^{\bullet})$ in $D_c^b(X \times Y)$, where $\pi_X : X \times Y \to X, \pi_Y : X \times Y \to Y$ are the projections.

If X is a \mathbb{C} -scheme, there is a functor $\mathbb{D}_X : D^b_c(X) \to D^b_c(X)^{\mathrm{op}}$ with

$$\mathbb{D}_X \circ \mathbb{D}_X \cong \mathrm{id} : D^b_c(X) \longrightarrow D^b_c(X),$$

called *Verdier duality*. It reverses shifts, that is, $\mathbb{D}_X(\mathcal{C}^{\bullet}[k]) = (\mathbb{D}_X(\mathcal{C}^{\bullet}))[-k]$ for \mathcal{C}^{\bullet} in $D_c^b(X)$ and $k \in \mathbb{Z}$.

Remark 2.3. Note how Definition 2.2 mixes the complex analytic and the complex algebraic: we consider sheaves on X^{an} in the *analytic* topology, which are constructible with respect to an *algebraic* stratification $X = \coprod_j X_j$.

Here are some properties of all these:

Theorem 2.4. In the following, X, Y, Z are \mathbb{C} -schemes, and f, g are morphisms, and all isomorphisms ' \cong ' of functors or objects are canonical.

(i) For $f: X \to Y$ and $g: Y \to Z$, there are natural isomorphisms of functors

(ii) If $f: X \to Y$ is proper then $Rf_* \cong Rf_!$.

(iii) If $f: X \to Y$ is étale then $f^* \cong f^!$. More generally, if $f: X \to Y$ is smooth of relative (complex) dimension d, then $f^*[d] \cong f^![-d]$, where $f^*[d], f^![-d]$ are the functors $f^*, f^!$ shifted by $\pm d$.

(iv) If $f: X \to Y$ then $Rf_! \cong \mathbb{D}_Y \circ Rf_* \circ \mathbb{D}_X$ and $f^! \cong \mathbb{D}_X \circ f^* \circ \mathbb{D}_Y$.

(v) If U is a smooth \mathbb{C} -scheme then $\mathbb{D}_U(A_U) \cong A_U[2 \dim U]$.

If X is a \mathbb{C} -scheme and $\mathcal{C}^{\bullet} \in D^b_c(X)$, the hypercohomology $\mathbb{H}^*(\mathcal{C}^{\bullet})$ and compactly-supported hypercohomology $\mathbb{H}^*_c(\mathcal{C}^{\bullet})$, both graded A-modules, are

$$\mathbb{H}^{k}(\mathcal{C}^{\bullet}) = H^{k}(R\pi_{*}(\mathcal{C}^{\bullet})) \quad \text{and} \quad \mathbb{H}^{k}_{c}(\mathcal{C}^{\bullet}) = H^{k}(R\pi_{!}(\mathcal{C}^{\bullet})) \quad \text{for } k \in \mathbb{Z},$$
(2.1)

where $\pi: X \to *$ is projection to a point.

If X is proper then $\mathbb{H}^*(\mathcal{C}^{\bullet}) \cong \mathbb{H}^*_{c}(\mathcal{C}^{\bullet})$ by Theorem 2.4(ii). They are related to usual cohomology by $\mathbb{H}^k(A_X) \cong H^k(X; A)$ and $\mathbb{H}^k_{c}(A_X) \cong H^k_{c}(X; A)$. If A is a field then under Verdier duality we have $\mathbb{H}^k(\mathcal{C}^{\bullet}) \cong \mathbb{H}^{-k}_{c}(\mathbb{D}_X(\mathcal{C}^{\bullet}))^*$.

2.2. Perverse sheaves on C-schemes. Next we review perverse sheaves, following Dimca [14, §5].

Definition 2.5. Let X be a \mathbb{C} -scheme, and for each $x \in X^{\mathrm{an}}$, let $i_x : * \to X$ map $i_x : * \mapsto x$. If $\mathcal{C}^{\bullet} \in D^b_c(X)$, then the support supp^m \mathcal{C}^{\bullet} and cosupport cosupp^m \mathcal{C}^{\bullet} of $\mathcal{H}^m(\mathcal{C}^{\bullet})$ for $m \in \mathbb{Z}$ are

$$\operatorname{supp}^{m} \mathcal{C}^{\bullet} = \left\{ x \in X^{\operatorname{an}} : \mathcal{H}^{m}(i_{x}^{*}(\mathcal{C}^{\bullet})) \neq 0 \right\},$$
$$\operatorname{cosupp}^{m} \mathcal{C}^{\bullet} = \overline{\left\{ x \in X^{\operatorname{an}} : \mathcal{H}^{m}(i_{x}^{!}(\mathcal{C}^{\bullet})) \neq 0 \right\}},$$

where $\overline{\{\cdots\}}$ means the closure in X^{an} . If A is a field then $\operatorname{cosupp}^m \mathcal{C}^{\bullet} = \operatorname{supp}^{-m} \mathbb{D}_X(\mathcal{C}^{\bullet})$. We call \mathcal{C}^{\bullet} perverse, or a perverse sheaf, if $\dim_{\mathbb{C}} \operatorname{supp}^{-m} \mathcal{C}^{\bullet} \leq m$ and $\dim_{\mathbb{C}} \operatorname{cosupp}^m \mathcal{C}^{\bullet} \leq m$ for all $m \in \mathbb{Z}$, where by convention $\dim_{\mathbb{C}} \emptyset = -\infty$. Write $\operatorname{Perv}(X)$ for the full subcategory of perverse sheaves in $D_c^b(X)$. Then $\operatorname{Perv}(X)$ is an abelian category, the heart of a t-structure on $D_c^b(X)$.

Perverse sheaves have the following properties:

Theorem 2.6. (a) If A is a field then Perv(X) is noetherian and artinian.

(b) If A is a field then $\mathbb{D}_X : D^b_c(X) \to D^b_c(X)$ maps $\operatorname{Perv}(X)$ to $\operatorname{Perv}(X)$.

(c) If $i : X \hookrightarrow Y$ is inclusion of a closed \mathbb{C} -subscheme, then $Ri_*, Ri_!$ (which are naturally isomorphic) map Perv(X) to Perv(Y).

Write $\operatorname{Perv}(Y)_X$ for the full subcategory of objects in $\operatorname{Perv}(Y)$ supported on X. Then $Ri_* \cong Ri_!$ are equivalences of categories $\operatorname{Perv}(X) \xrightarrow{\sim} \operatorname{Perv}(Y)_X$. The restrictions $i^*|_{\operatorname{Perv}(Y)_X}$, $i^!|_{\operatorname{Perv}(Y)_X}$ which map $\operatorname{Perv}(Y)_X$ to $\operatorname{Perv}(X)$, are naturally isomorphic, and are quasi-inverses for

$$Ri_*, Ri_! : \operatorname{Perv}(X) \to \operatorname{Perv}(Y)_X$$

(d) If $f: X \to Y$ is étale then f^* and $f^!$ (which are naturally isomorphic) map $\operatorname{Perv}(Y)$ to $\operatorname{Perv}(X)$. More generally, if $f: X \to Y$ is smooth of relative dimension d, then $f^*[d] \cong f^![-d]$ map $\operatorname{Perv}(Y)$ to $\operatorname{Perv}(X)$.

(e) $\stackrel{\sim}{\boxtimes} : D^b_c(X) \times D^b_c(Y) \to D^b_c(X \times Y) \text{ maps } \operatorname{Perv}(X) \times \operatorname{Perv}(Y) \text{ to } \operatorname{Perv}(X \times Y).$

(f) Let U be a smooth \mathbb{C} -scheme. Then $A_U[\dim U]$ is perverse, where A_U is the constant sheaf on U with fibre A, and $[\dim U]$ means shift by $\dim U$ in the triangulated category $D_c^b(X)$. Note that Theorem 2.4(v) gives a canonical isomorphism $\mathbb{D}_U(A_U[\dim U]) \cong A_U[\dim U]$.

When $A = \mathbb{Q}$, so that $\operatorname{Perv}(X)$ is noetherian and artinian by Theorem 2.6(a), the simple objects in $\operatorname{Perv}(X)$ admit a complete description: they are all isomorphic to intersection cohomology complexes $IC_{\bar{V}}(\mathcal{L})$ for $V \subseteq X$ a smooth locally closed \mathbb{C} -subscheme and $\mathcal{L} \to V$ an irreducible \mathbb{Q} -local system, [14, §5.4]. Furthermore, if $f : X \to Y$ is a proper morphism of \mathbb{C} -schemes, then the Decomposition Theorem [5, 6.2.5], [14, Th. 5.4.10], [45, Cor. 3] says that, in case $IC_{\bar{V}}(\mathcal{L})$ is of geometric origin, $Rf_*(IC_{\bar{V}}(\mathcal{L}))$ is isomorphic to a finite direct sum of shifts of simple objects $IC_{\bar{V}'}(\mathcal{L}')$ in $\operatorname{Perv}(Y)$.

The next theorem is proved by Beilinson et al. [5, Cor. 2.1.23, §2.2.19, & Th. 3.2.4]. The analogue for $D_c^b(X)$ or D(X) rather than Perv(X) is false. One moral is that perverse sheaves behave like sheaves, rather than like complexes.

Theorem 2.7(i) will be used throughout 3-6. Theorem 2.7(ii) will be used only once, in the proof of Theorem 6.9 in 6.3, and we only need Theorem 2.7(ii) to hold in the Zariski topology, rather than the étale topology.

Theorem 2.7. Let X be a \mathbb{C} -scheme. Then perverse sheaves on X form a stack (a kind of sheaf of categories) on X in the étale topology.

Explicitly, this means the following. Let $\{u_i : U_i \to X\}_{i \in I}$ be an **étale open cover** for X, so that $u_i : U_i \to X$ is an étale morphism of \mathbb{C} -schemes for $i \in I$ with $\coprod_i u_i$ surjective. Write $U_{ij} = U_i \times_{u_i, X, u_j} U_j$ for $i, j \in I$ with projections

$$\pi_{ij}^i: U_{ij} \longrightarrow U_i, \quad \pi_{ij}^j: U_{ij} \longrightarrow U_j, \quad u_{ij} = u_i \circ \pi_{ij}^i = u_j \circ \pi_{ij}^j: U_{ij} \longrightarrow X.$$

Similarly, write $U_{ijk} = U_i \times_X U_j \times_X U_k$ for $i, j, k \in I$ with projections

$$\pi_{ijk}^{ij}: U_{ijk} \longrightarrow U_{ij}, \quad \pi_{ijk}^{ik}: U_{ijk} \longrightarrow U_{ik}, \quad \pi_{ijk}^{jk}: U_{ijk} \longrightarrow U_{jk},$$

$$\pi_{ijk}^{i}: U_{ijk} \longrightarrow U_{i}, \quad \pi_{ijk}^{j}: U_{ijk} \longrightarrow U_{j}, \quad \pi_{ijk}^{k}: U_{ijk} \longrightarrow U_{k}, \quad u_{ijk}: U_{ijk} \longrightarrow X,$$

so that $\pi_{ijk}^i = \pi_{ij}^i \circ \pi_{ijk}^{ij}$, $u_{ijk} = u_{ij} \circ \pi_{ijk}^{ij} = u_i \circ \pi_{ijk}^i$, and so on. All these morphisms $u_i, \pi_{ij}^i, \ldots, u_{ijk}$ are étale, so by Theorem 2.6(d) $u_i^* \cong u_i^i$ maps $\operatorname{Perv}(X) \to \operatorname{Perv}(U_i)$, and similarly for $\pi_{ij}^i, \ldots, u_{ijk}$. With this notation:

(i) Suppose $\mathcal{P}^{\bullet}, \mathcal{Q}^{\bullet} \in \text{Perv}(X)$, and we are given $\alpha_i : u_i^*(\mathcal{P}^{\bullet}) \to u_i^*(\mathcal{Q}^{\bullet})$ in $\text{Perv}(U_i)$ for all $i \in I$ such that for all $i, j \in I$ we have

$$(\pi_{ij}^i)^*(\alpha_i) = (\pi_{ij}^i)^*(\alpha_j) : u_{ij}^*(\mathcal{P}^{\bullet}) \longrightarrow u_{ij}^*(\mathcal{Q}^{\bullet}).$$

Then there is a unique $\alpha : \mathcal{P}^{\bullet} \to \mathcal{Q}^{\bullet}$ in $\operatorname{Perv}(X)$ with $\alpha_i = u_i^*(\alpha)$ for all $i \in I$. (ii) Suppose we are given objects $\mathcal{P}_i^{\bullet} \in \operatorname{Perv}(U_i)$ for all $i \in I$ and isomorphisms

$$\alpha_{ij}: (\pi^i_{ij})^*(\mathcal{P}^{\bullet}_i) \longrightarrow (\pi^j_{ij})^*(\mathcal{P}^{\bullet}_j)$$

in $\operatorname{Perv}(U_{ij})$ for all $i, j \in I$ with $\alpha_{ii} = \operatorname{id}$ and

$$(\pi_{ijk}^{jk})^*(\alpha_{jk}) \circ (\pi_{ijk}^{ij})^*(\alpha_{ij}) = (\pi_{ijk}^{ik})^*(\alpha_{ik}) : (\pi_{ijk}^i)^*(\mathcal{P}_i) \longrightarrow (\pi_{ijk}^k)^*(\mathcal{P}_k)$$

in $\operatorname{Perv}(U_{ijk})$ for all $i, j, k \in I$. Then there exists \mathcal{P}^{\bullet} in $\operatorname{Perv}(X)$, unique up to canonical isomorphism, with isomorphisms $\beta_i : u_i^*(\mathcal{P}^{\bullet}) \to \mathcal{P}_i^{\bullet}$ for each $i \in I$, satisfying

$$\alpha_{ij} \circ (\pi^i_{ij})^* (\beta_i) = (\pi^j_{ij})^* (\beta_j) : u^*_{ij} (\mathcal{P}^{\bullet}) \longrightarrow (\pi^j_{ij})^* (\mathcal{P}^{\bullet}_j)$$

for all $i, j \in I$.

We will need the following proposition in §3.3 to prove Theorem 3.1(b). Most of it is setting up notation, only the last part $\alpha|_{X'} = \beta|_{X'}$ is nontrivial.

Proposition 2.8. Let W, X be \mathbb{C} -schemes, $x \in X$, and $\pi_{\mathbb{C}} : W \to \mathbb{C}, \pi_X : W \to X, \iota : \mathbb{C} \to W$ morphisms, such that $\pi_{\mathbb{C}} \times \pi_X : W \to \mathbb{C} \times X$ is étale, and $\pi_{\mathbb{C}} \circ \iota = \mathrm{id}_{\mathbb{C}} : \mathbb{C} \to \mathbb{C}$, and $\pi_X \circ \iota(t) = x$ for all $t \in \mathbb{C}$. Write $W_t = \pi_{\mathbb{C}}^{-1}(t) \subset W$ for each $t \in \mathbb{C}$, and $j_t : W_t \to W$ for the inclusion. Then $\pi_X|_{W_t} = \pi_X \circ j_t : W_t \to X$ is étale, and $\iota(t) \in W_t$ with $\pi_X|_{W_t}(\iota(t)) = x$, so we may think of W_t for $t \in \mathbb{C}$ as a 1-parameter family of étale open neighbourhoods of x in X.

Let $\mathcal{P}^{\bullet}, \mathcal{Q}^{\bullet} \in \text{Perv}(X)$, so that by Theorem 2.6(d) as π_X is smooth of relative dimension 1 and $\pi_X|_{W_t}$ is étale, we have $\pi_X^*[1](\mathcal{P}^{\bullet}) \in \text{Perv}(W)$ and

$$\pi_X|_{W_t}^*(\mathcal{P}^{\bullet}) = j_t^*[-1]\big(\pi_X^*[1](\mathcal{P}^{\bullet})\big) \in \operatorname{Perv}(W_t),$$

and similarly for \mathcal{Q}^{\bullet} .

Suppose $\alpha, \beta : \mathcal{P}^{\bullet} \to \mathcal{Q}^{\bullet}$ in $\operatorname{Perv}(X)$ and $\gamma : \pi_X^*[1](\mathcal{P}^{\bullet}) \to \pi_X^*[1](\mathcal{Q}^{\bullet})$ in $\operatorname{Perv}(W)$ are morphisms such that $\pi_X|_{W_0}^*(\alpha) = j_0^*[-1](\gamma)$ in $\operatorname{Perv}(W_0)$ and $\pi_X|_{W_1}^*(\beta) = j_1^*[-1](\gamma)$ in $\operatorname{Perv}(W_1)$. Then there exists a Zariski open neighbourhood X' of x in X such that

 $\alpha|_{X'} = \beta|_{X'} : \mathcal{P}^{\bullet}|_{X'} \longrightarrow \mathcal{Q}^{\bullet}|_{X'}.$

Here we should think of $j_t^*[-1](\gamma)$ for $t \in \mathbb{C}$ as a family of perverse sheaf morphisms $\mathcal{P}^{\bullet} \to \mathcal{Q}^{\bullet}$, defined near x in X locally in the étale topology. But morphisms of perverse sheaves are discrete (to see this, note that we can take $A = \mathbb{Z}$), so as $j_t^*[-1](\gamma)$ depends continuously on t, it should be locally constant in t near x, in a suitable sense. The conclusion $\alpha|_{X'} = \beta|_{X'}$ essentially says that $j_0^*[-1](\gamma) = j_1^*[-1](\gamma)$ near x.

If $P \to X$ is a principal $\mathbb{Z}/2\mathbb{Z}$ -bundle on a \mathbb{C} -scheme X, and $\mathcal{Q}^{\bullet} \in \operatorname{Perv}(X)$, we will define a perverse sheaf $\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P$, which will be important in §5–§6.

Definition 2.9. Let X be a \mathbb{C} -scheme. A principal $\mathbb{Z}/2\mathbb{Z}$ -bundle $P \to X$ is a proper, surjective, étale morphism of \mathbb{C} -schemes $\pi : P \to X$ together with a free involution $\sigma : P \to P$, such that the orbits of $\mathbb{Z}/2\mathbb{Z} = \{1, \sigma\}$ are the fibres of π . We will use the ideas of isomorphism of principal

bundles $\iota: P \to P'$, section $s: X \to P$, tensor product $P \otimes_{\mathbb{Z}/2\mathbb{Z}} P'$, and pullback $f^*(P) \to W$ under a \mathbb{C} -scheme morphism $f: W \to X$, all of which are defined in the obvious ways.

Let $P \to X$ be a principal $\mathbb{Z}/2\mathbb{Z}$ -bundle. Write $\mathcal{L}_P \in D_c^b(X)$ for the rank one A-local system on X induced from P by the nontrivial representation of $\mathbb{Z}/2\mathbb{Z} \cong \{\pm 1\}$ on A. It is characterized by $\pi_*(A_P) \cong A_X \oplus \mathcal{L}_P$. For each $\mathcal{Q}^{\bullet} \in D_c^b(X)$, write $\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P \in D_c^b(X)$ for $\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}} \mathcal{L}_P$, and call it \mathcal{Q}^{\bullet} twisted by P. If \mathcal{Q}^{\bullet} is perverse then $\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P$ is perverse.

Perverse sheaves and complexes twisted by principal $\mathbb{Z}/2\mathbb{Z}$ -bundles have the obvious functorial behaviour. For example, if $P \to X$, $P' \to X$ are principal $\mathbb{Z}/2\mathbb{Z}$ -bundles and $\mathcal{Q}^{\bullet} \in D_c^b(X)$ there is a canonical isomorphism $(\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P) \otimes_{\mathbb{Z}/2\mathbb{Z}} P' \cong \mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} (P \otimes_{\mathbb{Z}/2\mathbb{Z}} P')$, and if $f: W \to X$ is a \mathbb{C} -scheme morphism there is a canonical isomorphism

$$f^*(\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P) \cong f^*(\mathcal{Q}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} f^*(P).$$

2.3. Nearby cycles and vanishing cycles on \mathbb{C} -schemes. We explain nearby cycles and vanishing cycles, as in Dimca [14, §4.2]. The definition is complex analytic, $\widetilde{X_*^{\mathrm{an}}}, \widetilde{\mathbb{C}^*}$ in (2.2) do not come from \mathbb{C} -schemes.

Definition 2.10. Let X be a \mathbb{C} -scheme, and let $f : X \to \mathbb{C}$ be a regular function. Define $X_0 = f^{-1}(0)$, as a \mathbb{C} -subscheme of X, and $X_* = X \setminus X_0$. Consider the commutative diagram of complex analytic spaces:

$$X_{0}^{\mathrm{an}} \xrightarrow{i} X^{\mathrm{an}} \underbrace{\leqslant}_{j}^{\pi} X_{*}^{\mathrm{an}} \underbrace{\leqslant}_{p}^{p} \widetilde{X}_{*}^{\mathrm{an}} \\ \downarrow^{f} \qquad \downarrow^{f} \qquad \downarrow^{f} \qquad \downarrow^{f} \\ \{0\} \longrightarrow \mathbb{C} \xleftarrow{\mathbb{C}^{*}} \underbrace{\leqslant}_{\rho}^{\rho} \widetilde{\mathbb{C}^{*}}.$$

$$(2.2)$$

Here $X^{\mathrm{an}}, X^{\mathrm{an}}_0, X^{\mathrm{an}}_*$ are the complex analytic spaces associated to the \mathbb{C} -schemes X_0, X, X_* , and $i : X^{\mathrm{an}}_0 \hookrightarrow X^{\mathrm{an}}, j : X^{\mathrm{an}}_* \hookrightarrow X^{\mathrm{an}}$ are the inclusions, $\rho : \widetilde{\mathbb{C}^*} \to \mathbb{C}^*$ is the universal cover of $\mathbb{C}^* = \mathbb{C} \setminus \{0\}$, and $\widetilde{X^{\mathrm{an}}_*} = X^{\mathrm{an}}_* \times_{f,\mathbb{C}^*,\rho} \widetilde{\mathbb{C}^*}$ the corresponding cover of X^{an}_* , with covering map $p : \widetilde{X^{\mathrm{an}}_*} \to X^{\mathrm{an}}_*$, and $\pi = j \circ p$.

As in §2.6, the triangulated categories D(X), $D_c^b(X)$ and six operations f^* , $f^!$, Rf_* , $Rf_!$, $\mathcal{R}Hom$, $\overset{L}{\otimes}$ also make sense for complex analytic spaces. So we can define the *nearby cycle* functor $\psi_f : D_c^b(X) \to D_c^b(X_0)$ to be $\psi_f = i^* \circ R\pi_* \circ \pi^*$. Since this definition goes via $\widetilde{X_*}^{an}$ which is not a \mathbb{C} -scheme, it is not obvious that ψ_f maps to (algebraically) constructible complexes $D_c^b(X_0)$ rather than just to $D(X_0)$, but it does [14, p. 103], [27, p. 352].

There is a natural transformation $\Xi: i^* \Rightarrow \psi_f$ between the functors

$$i^*, \psi_f : D^b_c(X) \longrightarrow D^b_c(X_0).$$

The vanishing cycle functor $\phi_f : D^b_c(X) \to D^b_c(X_0)$ is a functor such that for every \mathcal{C}^{\bullet} in $D^b_c(X)$ we have a distinguished triangle

$$i^*(\mathcal{C}^{\bullet}) \xrightarrow{\Xi(\mathcal{C}^{\bullet})} \psi_f(\mathcal{C}^{\bullet}) \longrightarrow \phi_f(\mathcal{C}^{\bullet}) \xrightarrow{[+1]} i^*(\mathcal{C}^{\bullet})$$

in $D_c^b(X_0)$. Following Dimca [14, p. 108], we write ψ_f^p, ϕ_f^p for the shifted functors $\psi_f[-1], \phi_f[-1]: D_c^b(X) \to D_c^b(X_0)$.

The generator of $\mathbb{Z} = \pi_1(\mathbb{C}^*)$ on $\widetilde{\mathbb{C}^*}$ induces a deck transformation $\delta_{\mathbb{C}^*} : \widetilde{\mathbb{C}^*} \to \widetilde{\mathbb{C}^*}$ which lifts to a deck transformation $\delta_{X^*} : \widetilde{X^*} \to \widetilde{X^*}$ with $p \circ \delta_{X^*} = p$ and $\tilde{f} \circ \delta_{X^*} = \delta_{\mathbb{C}^*} \circ \tilde{f}$. As in [14, p. 103,

p. 105], we can use δ_{X^*} to define natural transformations $M_{X,f}: \psi_f^p \Rightarrow \psi_f^p$ and $M_{X,f}: \phi_f^p \Rightarrow \phi_f^p$, called *monodromy*.

Alternative definitions of ψ_f , ϕ_f in terms of specialization and microlocalization functors are given by Kashiwara and Schapira [27, Prop. 8.6.3]. Here are some properties of nearby and vanishing cycles. Parts (i),(ii) can be found in Dimca [14, Th. 5.2.21 & Prop. 4.2.11]. Part (iv) is proved by Massey [37]; compare also Proposition A.1 in the Appendix.

Theorem 2.11. (i) If X is a \mathbb{C} -scheme and $f: X \to \mathbb{C}$ is regular, then the functors $\psi_f^p, \phi_f^p: D_c^b(X) \to D_c^b(X_0)$ both map $\operatorname{Perv}(X)$ to $\operatorname{Perv}(X_0)$.

(ii) Let $\Phi: X \to Y$ be a proper morphism of \mathbb{C} -schemes, and $g: Y \to \mathbb{C}$ be regular. Write $f = g \circ \Phi: X \to \mathbb{C}, X_0 = f^{-1}(0) \subseteq X, Y_0 = g^{-1}(0) \subseteq Y$, and $\Phi_0 = \Phi|_{X_0}: X_0 \to Y_0$. Then we have natural isomorphisms

$$R(\Phi_0)_* \circ \psi_f^p \cong \psi_q^p \circ R\Phi_* \quad and \quad R(\Phi_0)_* \circ \phi_f^p \cong \phi_q^p \circ R\Phi_*.$$
(2.3)

Note too that $R\Phi_* \cong R\Phi_!$ and $R(\Phi_0)_* \cong R(\Phi_0)_!$, as Φ, Φ_0 are proper.

(iii) Let $\Phi: X \to Y$ be an étale morphism of \mathbb{C} -schemes, and $g: Y \to \mathbb{C}$ be regular. Write $f = g \circ \Phi: X \to \mathbb{C}, X_0 = f^{-1}(0) \subseteq X, Y_0 = g^{-1}(0) \subseteq Y$, and $\Phi_0 = \Phi|_{X_0}: X_0 \to Y_0$. Then we have natural isomorphisms

$$\Phi_0^* \circ \psi_f^p \cong \psi_q^p \circ \Phi^* \quad and \quad \Phi_0^* \circ \phi_f^p \cong \phi_q^p \circ \Phi^*.$$
(2.4)

Note too that $\Phi^* \cong \Phi^!$ and $\Phi_0^* \cong \Phi_0^!$, as Φ, Φ_0 are étale.

More generally, if $\Phi: X \to Y$ is smooth of relative (complex) dimension d and g, f, X_0, Y_0, Φ_0 are as above, then we have natural isomorphisms

$$\Phi_0^*[d] \circ \psi_f^p \cong \psi_g^p \circ \Phi^*[d] \quad and \quad \Phi_0^*[d] \circ \phi_f^p \cong \phi_g^p \circ \Phi^*[d].$$

$$(2.5)$$

Note too that $\Phi^*[d] \cong \Phi^![-d]$ and $\Phi_0^*[d] \cong \Phi_0^![-d]$.

(iv) If X is a \mathbb{C} -scheme and $f : X \to \mathbb{C}$ is regular, then there are natural isomorphisms $\psi_f^p \circ \mathbb{D}_X \cong \mathbb{D}_{X_0} \circ \psi_f^p$ and $\phi_f^p \circ \mathbb{D}_X \cong \mathbb{D}_{X_0} \circ \phi_f^p$.

2.4. Perverse sheaves of vanishing cycles on \mathbb{C} -schemes. We can now define the main subject of this paper, the perverse sheaf of vanishing cycles $\mathcal{PV}_{U,f}^{\bullet}$ for a regular function $f: U \to \mathbb{C}$.

Definition 2.12. Let U be a smooth \mathbb{C} -scheme, and $f : U \to \mathbb{C}$ a regular function. Write $X = \operatorname{Crit}(f)$, as a closed \mathbb{C} -subscheme of U.

Then as a map of topological spaces, $f|_X : X \to \mathbb{C}$ is locally constant, with finite image f(X), so we have a decomposition $X = \coprod_{c \in f(X)} X_c$, for $X_c \subseteq X$ the open and closed \mathbb{C} -subscheme with f(x) = c for each \mathbb{C} -point $x \in X_c$.

(Note that if X is non-reduced, then $f|_X : X \to \mathbb{C}$ need not be locally constant as a morphism of \mathbb{C} -schemes, but $f|_{X^{\text{red}}} : X^{\text{red}} \to \mathbb{C}$ is locally constant, where X^{red} is the reduced \mathbb{C} -subscheme of X. Since X, X^{red} have the same topological space, $f|_X : X \to \mathbb{C}$ is locally constant on topological spaces.)

For each $c \in \mathbb{C}$, write $U_c = f^{-1}(c) \subseteq U$. Then as in §2.3, we have a vanishing cycle functor ϕ_{f-c}^p : Perv $(U) \to$ Perv (U_c) . So we may form $\phi_{f-c}^p(A_U[\dim U])$ in Perv (U_c) , since $A_U[\dim U] \in$ Perv(U) by Theorem 2.6(f). One can show $\phi_{f-c}^p(A_U[\dim U])$ is supported on the closed subset $X_c = \operatorname{Crit}(f) \cap U_c$ in U_c , where $X_c = \emptyset$ unless $c \in f(X)$. That is, $\phi_{f-c}^p(A_U[\dim U])$ lies in Perv $(U_c)_{X_c}$. But Theorem 2.6(c) says $\operatorname{Perv}(U_c)_{X_c}$ and $\operatorname{Perv}(X_c)$ are equivalent categories, so we may regard $\phi_{f-c}^p(A_U[\dim U])$ as a perverse sheaf on X_c . That is, we can consider

$$\phi_{f-c}^p(A_U[\dim U])|_{X_c} = i_{X_c,U_c}^*(\phi_{f-c}^p(A_U[\dim U]))$$

in Perv (X_c) , where $i_{X_c,U_c}: X_c \to U_c$ is the inclusion morphism.

As $X = \coprod_{c \in f(X)} X_c$ with each X_c open and closed in X, we have

$$\operatorname{Perv}(X) = \bigoplus_{c \in f(X)} \operatorname{Perv}(X_c)$$

Define the perverse sheaf of vanishing cycles $\mathcal{PV}_{U,f}^{\bullet}$ of U, f in $\operatorname{Perv}(X)$ to be

$$\mathcal{PV}^{\bullet}_{U,f} = \bigoplus_{c \in f(X)} \phi^p_{f-c}(A_U[\dim U])|_{X_c}$$

That is, $\mathcal{PV}_{U,f}^{\bullet}$ is the unique perverse sheaf on $X = \operatorname{Crit}(f)$ with

$$\mathcal{PV}_{U,f}^{\bullet}|_{X_c} = \phi_{f-c}^p(A_U[\dim U])|_{X_c}$$

for all $c \in f(X)$.

Under Verdier duality, we have $A_U[\dim U] \cong \mathbb{D}_U(A_U[\dim U])$ by Theorem 2.6(f), so

$$\phi_{f-c}^p(A_U[\dim U]) \cong \mathbb{D}_{U_c}(\phi_{f-c}^p(A_U[\dim U]))$$

by Theorem 2.11(iv). Applying $i^*_{X_c,U_c}$ and using $\mathbb{D}_{X_c} \circ i^*_{X_c,U_c} \cong i^!_{X_c,U_c} \circ \mathbb{D}_{U_c}$ by Theorem 2.4(iv) and $i^!_{X_c,U_c} \cong i^*_{X_c,U_c}$ on $\operatorname{Perv}(U_c)_{X_c}$ by Theorem 2.6(c) also gives

$$\phi_{f-c}^p(A_U[\dim U])|_{X_c} \cong \mathbb{D}_{X_c}(\phi_{f-c}^p(A_U[\dim U])|_{X_c}).$$

Summing over all $c \in f(X)$ yields a canonical isomorphism

$$\sigma_{U,f}: \mathcal{PV}^{\bullet}_{U,f} \xrightarrow{\cong} \mathbb{D}_X(\mathcal{PV}^{\bullet}_{U,f}).$$
(2.6)

For $c \in f(X)$, we have a monodromy operator

$$M_{U,f-c}: \phi_{f-c}^p(A_U[\dim U]) \longrightarrow \phi_{f-c}^p(A_U[\dim U]),$$

which restricts to $\phi_{f-c}^p(A_U[\dim U])|_{X_c}$. Define the twisted monodromy operator

$$\tau_{U,f}: \mathcal{PV}_{U,f}^{\bullet} \longrightarrow \mathcal{PV}_{U,f}^{\bullet}$$

by

$$\tau_{U,f}|_{X_c} = (-1)^{\dim U} M_{U,f-c}|_{X_c} : \phi_{f-c}^p(A_U[\dim U])|_{X_c} \longrightarrow \phi_{f-c}^p(A_U[\dim U])|_{X_c},$$
(2.7)

for each $c \in f(X)$.

Here 'twisted' refers to the sign $(-1)^{\dim U}$ in (2.7). We include this sign change as it makes monodromy act naturally under transformations which change dimension — without it, equation (5.15) below would only commute up to a sign $(-1)^{\dim V - \dim U}$, not commute — and it normalizes the monodromy of any nondegenerate quadratic form to be the identity, as in (2.13). The sign $(-1)^{\dim U}$ also corresponds to the twist ' $(\frac{1}{2} \dim U)$ ' in the definition (2.24) of the mixed Hodge module of vanishing cycles $\mathcal{HV}_{U,f}^{\bullet}$ in §2.10.

The (compactly-supported) hypercohomology $\mathbb{H}^*(\mathcal{PV}_{U,f}^{\bullet}), \mathbb{H}^*_{c}(\mathcal{PV}_{U,f}^{\bullet})$ from (2.1) is an important invariant of U, f. If A is a field then the isomorphism $\sigma_{U,f}$ in (2.6) implies that $\mathbb{H}^k(\mathcal{PV}_{U,f}^{\bullet}) \cong \mathbb{H}^{-k}_{c}(\mathcal{PV}_{U,f}^{\bullet})^*$, a form of Poincaré duality.

We defined $\mathcal{PV}_{U,f}^{\bullet}(A)$ in perverse sheaves over a base ring A. Writing $\mathcal{PV}_{U,f}^{\bullet}(A)$ to denote the base ring, one can show that

$$\mathcal{PV}^{\bullet}_{U,f}(A) \cong \mathcal{PV}^{\bullet}_{U,f}(\mathbb{Z}) \overset{\scriptscriptstyle{L}}{\otimes}_{\mathbb{Z}} A.$$

Thus, we may as well take $A = \mathbb{Z}$, or $A = \mathbb{Q}$ if we want A to be a field, since the case of general A contains no more information.

There is a Thom–Sebastiani Theorem for perverse sheaves, due to Massey [35] and Schürmann [50, Cor. 1.3.4]. Applied to $\mathcal{PV}_{U,f}^{\bullet}$, it yields:

Theorem 2.13. Let U, V be smooth \mathbb{C} -schemes and $f: U \to \mathbb{C}$, $g: V \to \mathbb{C}$ be regular, so that $f \boxplus g: U \times V \to \mathbb{C}$ is regular with $(f \boxplus g)(u, v) := f(u) + g(v)$. Set $X = \operatorname{Crit}(f)$ and $Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V, so that $\operatorname{Crit}(f \boxplus g) = X \times Y$. Then there is a natural isomorphism

$$\mathcal{TS}_{U,f,V,g}: \mathcal{PV}_{U\times V,f\boxplus g}^{\bullet} \longrightarrow \mathcal{PV}_{U,f}^{\bullet} \boxtimes \mathcal{PV}_{V,g}^{\bullet}$$
(2.8)

in $Perv(X \times Y)$, such that the following diagrams commute:

The next example will be important later.

Example 2.14. Define $f : \mathbb{C}^n \to \mathbb{C}$ by

$$f(z_1,\ldots,z_n) = z_1^2 + \cdots + z_n^2$$

for n > 1. Then $\operatorname{Crit}(f) = \{0\}$, so $\mathcal{PV}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}^{\bullet} = \phi_f^p(A_{\mathbb{C}^n}[n])|_{\{0\}}$ is a perverse sheaf on the point $\{0\}$. Following Dimca [14, Prop. 4.2.2, Ex. 4.2.3 & Ex. 4.2.6], we find that there is a canonical isomorphism

$$\mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} \cong H^{n-1} \big(MF_f(0); A \big) \otimes_A A_{\{0\}}, \tag{2.11}$$

where $MF_f(0)$ is the Milnor fibre of f at 0, as in [14, p. 103]. Since $f(z) = z_1^2 + \cdots + z_n^2$ is homogeneous, we see that

$$MF_f(0) \cong \left\{ (z_1, \dots, z_n) \in \mathbb{C}^n : f(z_1, \dots, z_n) = 1 \right\} \cong T^* \mathcal{S}^{n-1},$$

so that $H^{n-1}(MF_f(0); A) \cong H^{n-1}(\mathcal{S}^{n-1}; A) \cong A$. Therefore we have

$$\mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} \cong A_{\{0\}}.$$
(2.12)

This isomorphism (2.12) is natural up to sign (unless the base ring A has characteristic 2, in which case (2.12) is natural), as it depends on the choice of isomorphism $H^{n-1}(\mathcal{S}^{n-1}, A) \cong A$, which corresponds to an orientation for \mathcal{S}^{n-1} . This uncertainty of signs will be important in §5–§6.

We can also use Milnor fibres to compute the monodromy operator on $\mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}$. There is a monodromy map $\mu_f : MF_f(0) \to MF_f(0)$, natural up to isotopy, which is the monodromy in the Milnor fibration of f at 0. Under the identification $MF_f(0) \cong T^* \mathcal{S}^{n-1}$ we may take μ_f to be the map $d(-1): T^* \mathcal{S}^{n-1} \to T^* \mathcal{S}^{n-1}$ induced by $-1: \mathcal{S}^{n-1} \to \mathcal{S}^{n-1}$ mapping

$$-1: (x_1, \ldots, x_n) \mapsto (-x_1, \ldots, -x_n).$$

This multiplies orientations on \mathcal{S}^{n-1} by $(-1)^n$. Thus, $\mu_{f*}: H^{n-1}(\mathcal{S}^{n-1}, A) \to H^{n-1}(\mathcal{S}^{n-1}, A)$ multiplies by $(-1)^n$.

By [14, Prop. 4.2.2], equation (2.11) identifies the action of the monodromy operator $M_{\mathbb{C}^n, f}|_{\{0\}}$ on $\mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}$ with the action of μ_{f*} on $H^{n-1}(\mathcal{S}^{n-1}, A)$. So $M_{\mathbb{C}^n, f}|_{\{0\}}$ is multiplication by $(-1)^n$. Combining this with the sign change $(-1)^{\dim U}$ in (2.7) for $U = \mathbb{C}^n$ shows that the twisted monodromy is

$$\pi_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} = \mathrm{id} : \mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} \longrightarrow \mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}.$$
(2.13)

Equations (2.12)–(2.13) also hold for n = 0, 1, though (2.11) does not.

Note also that these results are compatible with the Thom–Sebastiani Theorem 2.13, and can be deduced from it and the case n = 1.

We introduce some notation for pullbacks of $\mathcal{PV}_{V,q}^{\bullet}$ by étale morphisms.

Definition 2.15. Let U, V be smooth \mathbb{C} -schemes, $\Phi : U \to V$ an étale morphism, and $g : V \to \mathbb{C}$ a regular function. Write $f = g \circ \Phi : U \to \mathbb{C}$, and $X = \operatorname{Crit}(f), Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V. Then $\Phi|_X : X \to Y$ is étale. Define an isomorphism

$$\mathcal{PV}_{\Phi}: \mathcal{PV}_{U,f}^{\bullet} \longrightarrow \Phi|_X^* \left(\mathcal{PV}_{V,g}^{\bullet} \right) \quad \text{in Perv}(X)$$
 (2.14)

by the commutative diagram for each $c \in f(X) \subseteq g(Y)$:

$$\mathcal{P}\mathcal{V}_{U,f}^{\bullet}|_{X_c} = \phi_{f-c}^p (A_U[\dim U])|_{X_c} \xrightarrow{\alpha} \phi_{f-c}^p \circ \Phi^*(A_V[\dim V]))|_{X_c}$$

$$\downarrow^{\mathcal{P}\mathcal{V}_{\Phi}|_{X_c}} \qquad \beta \downarrow \qquad (2.15)$$

$$\Phi|_{X_c}^*(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}) = \Phi_0^* \circ \phi_{g-c}^p \circ (A_V[\dim V]))|_{X_c}.$$

Here α is ϕ_{f-c}^p applied to the canonical isomorphism $A_U \to \Phi^*(A_V)$, noting that, as Φ is étale, dim $U = \dim V$ and β is induced by (2.4).

By naturality of the isomorphisms α, β in (2.15) we find the following commute, where $\sigma_{U,f}, \tau_{U,f}$ are as in (2.6)–(2.7):

$$\mathcal{PV}_{U,f}^{\bullet} \xrightarrow{\sigma_{U,f}} \mathbb{D}_{X}(\mathcal{PV}_{U,f}^{\bullet}) \\
\downarrow^{\mathcal{P}V_{\Phi}} \xrightarrow{\mathbb{D}_{X}(\mathcal{P}V_{\Phi})} \\
\Phi|_{X}^{*}(\mathcal{PV}_{V,g}^{\bullet}) \xrightarrow{\Phi|_{X}^{*}(\sigma_{V,g})} \Phi|_{X}^{*}(\mathbb{D}_{Y}(\mathcal{PV}_{V,g}^{\bullet})) \xrightarrow{\cong} \mathbb{D}_{X}(\Phi|_{X}^{*}(\mathcal{PV}_{V,g}^{\bullet})), \quad (2.16)$$

If U = V, f = g and $\Phi = \mathrm{id}_U$ then $\mathcal{PV}_{\mathrm{id}_U} = \mathrm{id}_{\mathcal{PV}_{U,f}^{\bullet}}$.

If W is another smooth \mathbb{C} -scheme, $\Psi : V \to W$ is étale, and $h : W \to \mathbb{C}$ is regular with $g = h \circ \Psi : V \to \mathbb{C}$, then composing (2.15) for Φ with $\Phi|_{X_c}^*$ of (2.15) for Ψ shows that

$$\mathcal{PV}_{\Psi\circ\Phi} = \Phi|_X^*(\mathcal{PV}_\Psi) \circ \mathcal{PV}_\Phi : \mathcal{PV}_{U,f}^{\bullet} \longrightarrow (\Psi \circ \Phi)|_X^*(\mathcal{PV}_{W,h}^{\bullet}).$$
(2.18)

That is, the isomorphisms \mathcal{PV}_{Φ} are functorial.

Example 2.16. In Definition 2.15, set $U = V = \mathbb{C}^n$ and

$$f(z_1, \ldots, z_n) = g(z_1, \ldots, z_n) = z_1^2 + \cdots + z_n^2,$$

so that $Y = Z = \{0\} \subset \mathbb{C}^n$. Let $M \in O(n, \mathbb{C})$ be an orthogonal matrix, so that $M : \mathbb{C}^n \to \mathbb{C}^n$ is an isomorphism with $f = g \circ M$ and $M|_{\{0\}} = \mathrm{id}_{\{0\}}$. As $M|_Y = \mathrm{id}_Y$, Definition 2.15 defines an isomorphism

$$\mathcal{PV}_M: \mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} \longrightarrow \mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}.$$
(2.19)

Equation (2.11) describes $\mathcal{PV}^{\bullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2}$ in terms of $MF_f(0) \cong T^* \mathcal{S}^{n-1}$. Now

$$M|_{MF_f(0)} : MF_f(0) \longrightarrow MF_f(0)$$

multiplies orientations on \mathcal{S}^{n-1} by det M, so

$$(M|_{MF_f(0)})_*: H^{n-1}(MF_f(0); A) \longrightarrow H^{n-1}(MF_f(0); A)$$

is multiplication by det M. Thus (2.11) implies that \mathcal{PV}_M in (2.19) is multiplication by det $M = \pm 1$.

2.5. Summary of the properties we use in this paper. Since parts of $\S2.1-\S2.4$ do not work for the other kinds of perverse sheaves, \mathscr{D} -modules and mixed Hodge modules in $\S2.6-\S2.10$, we list what we will need for $\S3-\S6$, to make it easy to check they are also valid in the settings of $\S2.6-\S2.10$.

- (i) There should be an A-linear abelian category $\mathcal{P}(X)$ of \mathcal{P} -objects defined for each scheme or complex analytic space X, over a fixed, well-behaved base ring A. We do not require A to be a field.
- (ii) There should be a Verdier duality functor \mathbb{D}_X with $\mathbb{D}_X \circ \mathbb{D}_X \cong$ id, defined on a suitable subcategory of \mathcal{P} -objects on X which includes the objects we are interested in. We do not need \mathbb{D}_X to be defined on all objects in $\mathcal{P}(X)$.
- (iii) If U is a smooth scheme or complex manifold, then there should be a canonical object $A_U[\dim U] \in \mathcal{P}(U)$, with a canonical isomorphism

$$\mathbb{D}_U(A_U[\dim U]) \cong A_U[\dim U].$$

- (iv) Let $f : X \hookrightarrow Y$ be a closed embedding of schemes or complex analytic spaces; this implies f is proper. Then $f_*, f_! : \mathcal{P}(X) \to \mathcal{P}(Y)$ should exist, inducing an equivalence of categories $\mathcal{P}(X) \xrightarrow{\sim} \mathcal{P}_X(Y)$ as in Theorem 2.6(c), where $\mathcal{P}_X(Y)$ is the full subcategory of objects in $\mathcal{P}(Y)$ supported on X.
- (v) Let $f: X \to Y$ be an étale morphism. Then the pullbacks $f^*, f^!: \mathcal{P}(Y) \to \mathcal{P}(X)$ should exist. More generally, if $f: X \to Y$ is smooth of relative dimension d, then there should be pullbacks $f^*[d], f^![-d]$ mapping $\mathcal{P}(Y) \to \mathcal{P}(X)$. If X, Y are smooth, there should be a canonical isomorphism $f^*[d](A_Y[\dim Y]) \cong A_X[\dim X]$. We do not need pullbacks to exist for general morphisms $f_r: X \to Y$, though see (xi) below.
- (vi) An external tensor product $\boxtimes : \mathcal{P}(X) \times \mathcal{P}(Y) \to \mathcal{P}(X \times Y)$ should exist for all X, Y.
- (vii) If X is a scheme or complex analytic space, $P \to X$ a principal $\mathbb{Z}/2\mathbb{Z}$ -bundle, and $\mathcal{Q}^{\bullet} \in \mathcal{P}(X)$, the twisted perverse sheaf $\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P \in \mathcal{P}(X)$ should make sense as in Definition 2.9, and have the obvious functorial properties.
- (viii) A vanishing cycle functor $\phi_f^p: \mathcal{P}(U) \to \mathcal{P}(U_0)$ and a monodromy transformation

$$M_{U,f}: \phi_f^p \Rightarrow \phi_f^p$$

in §2.4 should exist for all smooth U and regular/holomorphic $f: U \to \mathbb{A}^1$.

- (ix) The functors $\mathbb{D}_X, f^*, f^!, f_*, f_!, \phi_f^p$ should satisfy the natural isomorphisms in Theorems 2.4 and 2.11, provided they exist. They should have the obvious compatibilities with $\stackrel{L}{\boxtimes}$, and restriction to (Zariski) open sets.
- (x) There should be suitable subcategories of \mathcal{P} -objects which form a stack in the étale or complex analytic topologies, as in Theorem 2.7. In the algebraic case we only need Theorem 2.7(ii) to hold for Zariski open covers, not étale open covers.
- (xi) Proposition 2.8 must hold. This involves pullbacks j_t^* by a morphism $j_t : W_t \hookrightarrow W$ which is not étale or smooth, as in (v) above. But on objects we only consider

$$j_t^*(\pi_X^*(\mathcal{P}^\bullet)) = \pi_X|_{W_t}^*(\mathcal{P}^\bullet)$$

which exists in $\mathcal{P}(W_t)$ by (v) as $\pi_X|_{W_t}$ is étale, so j_t^* is defined on the objects we need.

(xii) There should be a Thom–Sebastiani Theorem for \mathcal{P} -objects, so that the analogue of Theorem 2.13 holds.

Remark 2.17. The existence of a (bounded) derived category of \mathcal{P} -objects will not be assumed, or used, in this paper. On the other hand, in all the cases we consider, there will be a realization functor from the category of \mathcal{P} -objects to an appropriate category of constructible complexes, and the notation used above reflects this. So in (iii),(v) above, [1] does not stand for a shift in any derived category; the notation means a \mathcal{P} -object or morphism whose realization is the appropriate constructible object or morphism. See Remark 2.20 below.

2.6. Perverse sheaves on complex analytic spaces. Next we discuss perverse sheaves on complex analytic spaces, as in Dimca [14]. The theory follows $\S2.1-\S2.4$, replacing (smooth) \mathbb{C} -schemes by complex analytic spaces (complex manifolds), and regular functions by holomorphic functions.

Let X be a complex analytic space, always assumed locally of finite type (that is, locally embeddable in \mathbb{C}^n). In the analogue of Definition 2.2, we fix a well-behaved commutative ring A, and consider sheaves of A-modules S on X in the complex analytic topology. A sheaf S is called (*analytically*) constructible if all the stalks S_x for $x \in X$ are finite type A-modules, and there is a locally finite stratification $X = \coprod_{j \in J} X_j$ of X, where now $X_j \subseteq X$ for $j \in J$ are complex analytic subspaces of X, such that $S|_{X_j}$ is an A-local system for all $j \in J$.

Write D(X) for the derived category of complexes \mathcal{C}^{\bullet} of sheaves of A-modules on X, exactly as in §2.1, and $D_c^b(X)$ for the full subcategory of bounded complexes \mathcal{C}^{\bullet} in D(X) whose cohomology sheaves $\mathcal{H}^m(\mathcal{C}^{\bullet})$ are analytically constructible for all $m \in \mathbb{Z}$. Then $D(X), D_c^b(X)$ are triangulated categories.

When we wish to distinguish the complex algebraic and complex analytic theories, we will write $D_c^b(X)^{\text{alg}}$, $\text{Perv}(X)^{\text{alg}}$ for the algebraic versions in §2.1–§2.2 with X a \mathbb{C} -scheme, and $D_c^b(X)^{\text{an}}$, $\text{Perv}(X)^{\text{an}}$ for the analytic versions.

Here are the main differences between the material of $\S2.1-\S2.4$ for perverse sheaves on \mathbb{C} -schemes and on complex analytic spaces:

(a) If $f: X \to Y$ is an arbitrary morphism of \mathbb{C} -schemes, then as in §2.1 the pushforwards $Rf_*, Rf_!: D(X) \to D(Y)$ also map $D^b_c(X)^{\text{alg}} \to D^b_c(Y)^{\text{alg}}$.

However, if $f: X \to Y$ is a morphism of complex analytic spaces, then

$$Rf_*, Rf_! : D(X) \longrightarrow D(Y)$$

need not map $D_c^b(X)^{\mathrm{an}} \to D_c^b(Y)^{\mathrm{an}}$ without extra assumptions on f, for example, if $f: X \to Y$ is proper.

(b) The analogue of Theorem 2.7 says that perverse sheaves on a complex analytic space X form a stack in the complex analytic topology. This is proved in the subanalytic context

in [27, Th. 10.2.9]; the analytic case follows upon noting that a sheaf is complex analytically constructible if and only if is locally at all points, as proved in [14, Prop. 4.1.13]. See also [21, Prop. 8.1.26].

The analogues of (i)–(xii) in §2.5 work for complex analytic perverse sheaves, and so our main results hold in this context.

If X is a \mathbb{C} -scheme, and X^{an} the corresponding complex analytic space, then D(X) in §2.1 for X a \mathbb{C} -scheme coincides with $D(X^{an})$ for X^{an} a complex analytic space, and

$$D_c^b(X)^{\operatorname{alg}} \subset D_c^b(X^{\operatorname{an}})^{\operatorname{an}}, \quad \operatorname{Perv}(X)^{\operatorname{alg}} \subset \operatorname{Perv}(X^{\operatorname{an}})^{\operatorname{an}}$$

are full subcategories, and the six functors f^* , $f^!$, Rf_* , $Rf_!$, $Rf_!$, $\mathcal{R}Hom$, $\overset{\scriptscriptstyle \perp}{\otimes}$ for \mathbb{C} -scheme morphisms $f: X \to Y$ agree in the algebraic and analytic cases.

2.7. \mathscr{D} -modules on \mathbb{C} -schemes and complex analytic spaces. \mathscr{D} -modules on a smooth \mathbb{C} -scheme or smooth complex analytic space X are sheaves of modules over a certain sheaf of rings of differential operators \mathscr{D}_X on X. Some books on them are Borel et al. [8], Coutinho [12], and Hotta, Takeuchi and Tanisaki [21] in the \mathbb{C} -scheme case, and Björk [7] and Kashiwara [26] in the complex analytic case. For a singular complex \mathbb{C} -scheme or complex analytic space X, the definition of a well-behaved category of \mathscr{D} -modules is given by Saito [48], via locally embedding X into a smooth scheme or space.

The analogue of perverse sheaves on X are called *regular holonomic* \mathscr{D} -modules, which form an abelian category $\operatorname{Mod}_{\operatorname{rh}}(\mathscr{D}_X)$, the heart in the derived category $D^b_{\operatorname{rh}}(\operatorname{Mod}(\mathscr{D}_X))$ of bounded complexes of \mathscr{D}_X -modules with regular holonomic cohomology modules. The whole package of $\S2.1-\S2.4$ works for \mathscr{D} -modules. Our next theorem is known as the *Riemann-Hilbert correspondence* [7, $\SV.5$], [21, Th. 7.2.1], see Borel [8, $\S14.4$] for \mathbb{C} -schemes, Kashiwara [25] for complex manifolds, and Saito [48, $\S6$] for complex analytic spaces, and also Maisonobe and Mekhbout [34].

Theorem 2.18. Let X be a \mathbb{C} -scheme or complex analytic space. Then there is a **de Rham** functor $\mathrm{DR} : D^b_{\mathrm{rh}}(\mathrm{Mod}(\mathscr{D}_X)) \xrightarrow{\sim} D^b_c(X, \mathbb{C})$, which is an equivalence of categories, restricts to an equivalence $\mathrm{Mod}_{\mathrm{rh}}(\mathscr{D}_X) \xrightarrow{\sim} \mathrm{Perv}(X, \mathbb{C})$, and commutes with $f^*, f^!, Rf_*, Rf_!, RHom, \overset{L}{\otimes}$, and also with ψ^p_f, ϕ^p_f for X smooth. Here $D^b_c(X, \mathbb{C}), \mathrm{Perv}(X, \mathbb{C})$ are constructible complexes and perverse sheaves over the base ring $A = \mathbb{C}$.

Because of the Riemann–Hilbert correspondence, all our results on perverse sheaves of vanishing cycles on \mathbb{C} -schemes and complex analytic spaces in §3–§6 over a well-behaved base ring A, translate immediately when $A = \mathbb{C}$ to the corresponding results for \mathscr{D} -modules of vanishing cycles, with no extra work.

2.8. Mixed Hodge modules: basics. We write this section in the minimal generality needed for our applications. The statements made work equally well in the category of (algebraic) \mathbb{C} -schemes and the category of complex analytic spaces. By space, we will mean an object in either of these categories. The theory of mixed Hodge modules works with reduced spaces; should a space X be non-reduced, the following constructions are taken by definition on its reduction.

For a space X, let $\operatorname{HM}(X)$ denote Saito's category [45] of polarizable pure Hodge modules, (locally) a direct sum of subcategories $\operatorname{HM}(X)^w$ of pure Hodge modules of fixed weight w. On a smooth X, a pure Hodge module M^{\bullet} consists of a triple of data: a filtered holonomic \mathscr{D} -module (M, F), a Q-perverse sheaf, and a comparison map identifying the former with the complexification of the latter under the Riemann-Hilbert correspondence; see [45, §5.1.1, p. 952] and [47, §4]. This triple has to satisfy many other properties; in particular, the underlying holonomic \mathscr{D} module is automatically regular, and algebraic Hodge modules are asked to be extendable to an algebraic compactification. Thus there is a forgetful functor $\operatorname{HM}(X) \to \operatorname{Mod}_{\mathrm{rh}}(\mathscr{D}_X)$ from Hodge modules to regular holonomic (algebraic) \mathscr{D} -modules. Hodge modules on singular spaces are defined, similarly to \mathscr{D} -modules, via embeddings into smooth varieties; see Saito [47] and also Maxim, Saito and Schürmann [39, §1.8].

There is a duality functor \mathbb{D}_X^H : HM(X) \to HM(X). Pure Hodge modules also admit a Tate twist functor $M^{\bullet} \mapsto M^{\bullet}(1)$, see [45, §5.1.3, p. 952]. This functor shifts the filtration and rotates the rational structure on the underlying perverse sheaf: the \mathscr{D} -module filtration (M, F) is shifted to (M, F[n]) with $(F[n])_i = F_{i-n}$; the underlying perverse sheaf is tensored by $\mathbb{Z}(n) = (2\pi i)^n \mathbb{Z} \subset \mathbb{C}$, as in [45, (2.0.2), p. 876].

A polarization of weight w on a pure Hodge module $M^{\bullet} \in HM(X)^w$ is a morphism of pure Hodge modules

$$\sigma: M^{\bullet} \longrightarrow \mathbb{D}_X^H(M^{\bullet})(-w),$$

satisfying the extra conditions using vanishing cycles described on [45, (5.1.6.2) on p. 956 and (5.2.10.2) on p. 968], as well as the condition that on points it should correspond to the classical notion of a polarization of a pure Hodge structure (including positive definiteness).

Next, let MHM(X) denote the category of graded polarizable mixed Hodge modules [45, 47]. A graded polarizable mixed Hodge module carries a functorial weight filtration W, with graded pieces being polarizable pure Hodge modules, see [45, §5.2.10, p. 967-8]. The forgetful functor **rat** : $MHM(X) \rightarrow Perv(X)$ to the appropriate category of perverse Q-sheaves on X is faithful and exact; faithfulness in particular means that a morphism in MHM(X) is uniquely determined by the underlying morphism of perverse sheaves. The Tate twist functor extends to MHM(X); under this functor, the weight filtration W of the mixed Hodge module is changed to W[2n]with $W[2n]_i = W_{i+2n}$ as on [45, p. 855]. The duality functor \mathbb{D}_X^H also extends to MHM(X) and is compatible with Verdier duality on the perverse realization. There is also a forgetful functor $MHM(X) \rightarrow Mod_{rh}(\mathscr{D}_X)$ to regular holonomic \mathscr{D} -modules, even for singular spaces.

Theorem 2.19. The categories of graded polarizable mixed Hodge modules have the following properties:

- (i) By [47, Th. 3.9, p. 288], the category of mixed Hodge modules for X a point is canonically equivalent to Deligne's category of graded polarizable mixed Hodge structures.
- (ii) For a smooth space U, we have a canonical object of weight dim U

$$\mathbb{Q}_U^H[\dim U] \in \mathrm{HM}(U) \subset \mathrm{MHM}(U),$$

which by [45, Prop. 5.2.16, p. 971] possesses a canonical polarization

$$\sigma: \mathbb{Q}_U^H[\dim U] \longrightarrow \mathbb{D}_U^H \mathbb{Q}_U^H[\dim U](-\dim U).$$

(iii) For an open inclusion $f: Y \hookrightarrow X$ of spaces, there is a pullback functor

 $f^* = f^! : \mathrm{MHM}(X) \longrightarrow \mathrm{MHM}(Y).$

More generally, by [47, Prop. 2.19, p. 258], for an arbitrary morphism $f: Y \to X$, there exist cohomological pullback functors $L^j f^*, L^j f^!$: MHM(X) \to MHM(Y) compatible with (perverse) cohomological pullback on the perverse sheaf level.

(iv) For a closed embedding $i: X \hookrightarrow Y$, there is a pushforward functor

$$i_* = i_! : \operatorname{MHM}(X) \longrightarrow \operatorname{MHM}(Y),$$

whose essential image is the full subcategory $MHM_X(Y)$ of objects in MHM(Y) supported on X. Its inverse is $i^* = i^! : MHM_X(Y) \to MHM(X)$. More generally, by [45, Th. 5.3.1, p. 977] and [47, Th. 2.14, p. 252], for a **projective** map $f : X \to Y$ there are cohomological pushforward functors

$$R^{j}f_{*}: \mathrm{MHM}(X) \longrightarrow \mathrm{MHM}(Y).$$

(v) There is an external tensor product functor

$$\overline{\boxtimes} : \mathrm{MHM}(X) \times \mathrm{MHM}(Y) \longrightarrow \mathrm{MHM}(X \times Y),$$

which is compatible with duality in the sense that for all $M^{\bullet} \in MHM(X)$ and $N^{\bullet} \in MHM(Y)$, there is a natural isomorphism

$$\mathbb{D}_X^H M^{\bullet} \stackrel{\scriptscriptstyle L}{\boxtimes} \mathbb{D}_Y^H N^{\bullet} \cong \mathbb{D}_{X \times Y}^H (M^{\bullet} \stackrel{\scriptscriptstyle L}{\boxtimes} N^{\bullet}).$$

Remark 2.20. We will not need to use any derived category D^2 MHM(X) of mixed Hodge modules in this paper, which is just as well since on singular analytic X, the appropriate boundedness conditions do not appear to be well understood, and the general pullback and pushforward functors of Theorem 2.19(iii),(iv) are not known to exist as derived functors outside of the algebraic context of [47, §4]. Hence, in part (ii) above, [1] does not stand for a shift in the derived category; $\mathbb{Q}_U^H[\dim U]$ just denotes a mixed Hodge module whose realization is the perverse sheaf $\mathbb{Q}_U[\dim U]$ on U. Compare Remark 2.17 above.

Using the functors above, we can now define the twist of a mixed Hodge module by a principal $\mathbb{Z}/2\mathbb{Z}$ -bundle. In the setup of Definition 2.9, given a $\mathbb{Z}/2\mathbb{Z}$ -bundle $\pi : P \to X$, and an object $M^{\bullet} \in \text{MHM}(X)$ on a space X, we have a natural map $M^{\bullet} \to \pi_*\pi^*M^{\bullet}$, which is an injection by faithfulness of the realization functor and the fact that it is an injection on the perverse sheaf level. The quotient object will be denoted, by abuse of notation, by $M^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P$ in MHM(X).

2.9. Monodromic mixed Hodge modules. To discuss nearby and vanishing cycle functors in a way consistent with monodromy, we need an extension of the category of mixed Hodge modules. For a space X, following Saito [49, §4.2] denote by MHM(X; T_s, N) the category of mixed Hodge modules M^{\bullet} on X with commuting actions of a finite order operator $T_s: M^{\bullet} \to M^{\bullet}$ and a locally nilpotent operator $N: M^{\bullet} \to M^{\bullet}(-1)$. There is an embedding of categories MHM(X) \to MHM(X; T_s, N) defined by setting T = id and N = 0. As proved by [49, (4.6.2)], the category MHM(X; T_s, N) is equivalent to the category MHM($X \times \mathbb{C}$)_{mon,!} of monodromic mixed Hodge modules on $X \times \mathbb{C}^*$ extended by zero to $X \times \mathbb{C}$; compare also [33, §4.2].

Every object $M^{\bullet} \in \text{MHM}(X; T_s, N)$ decomposes into a direct sum $M^{\bullet} = M_1^{\bullet} \oplus M_{\neq 1}^{\bullet}$ of the T_s -invariant part and its T_s -equivariant complement. The Tate twist, and appropriate cohomological pullback and pushforward functors continue to exist. There is a duality functor

$$\mathbb{D}_X^T : \mathrm{MHM}(X; T_s, N) \longrightarrow \mathrm{MHM}(X; T_s, N)$$

defined by

$$\mathbb{D}_X^T(M^{\bullet}) = \mathbb{D}_X^H(M_1^{\bullet}) \oplus \mathbb{D}_T^H(M_{\neq 1}^{\bullet})(1),$$

equipped with the finite-order operator $\mathbb{D}_X(T_s)^{-1}$ and the nilpotent operator $-\mathbb{D}_X(N)$. This duality functor still satisfies $\mathbb{D}_X^T \circ \mathbb{D}_X^T = \mathrm{id}$.

Saito [49, §5.1] also defines an external tensor product

$$\stackrel{^{T}}{\boxtimes} : \mathrm{MHM}(X_1; T_s, N) \times \mathrm{MHM}(X_2; T_s, N) \longrightarrow \mathrm{MHM}(X_1 \times X_2; T_s, N).$$

defined on the monodromic category as follows. The addition map on fibres

$$\pi: (X_1 \times \mathbb{C}) \times (X_2 \times \mathbb{C}) \longrightarrow (X_1 \times X_2) \times \mathbb{C}$$

induces the additive convolution

$$\pi_*(-\boxtimes -): \mathrm{MHM}(X_1 \times \mathbb{C})_{\mathrm{mon},!} \times \mathrm{MHM}(X_2 \times \mathbb{C})_{\mathrm{mon},!} \longrightarrow \mathrm{MHM}(X_1 \times x_2 \times \mathbb{C})_{\mathrm{mon},!}$$

One can translate this external tensor product \boxtimes to the MHM $(X; T_s, N)$ defined by concrete data (M^{\bullet}, T_s, N) . On the underlying \mathscr{D} -modules and perverse sheaves, it is just the usual product

102
\boxtimes . The operators are defined by $T_s = T_s \boxtimes T_s$ and $N = N \boxtimes id + id \boxtimes N$. However, the Hodge and weight filtrations on the underlying \mathscr{D} -modules and perverse sheaves are shifted using the finite order endomorphisms T_s ; for details, see [49, (5.1.1)–(5.1.2)]. Note that as a consequence of these definitions, the forgetful functors

$$MHM(-; T_s, N) \longrightarrow MHM(-)$$

do not map \boxtimes to \boxtimes . Twisted duality and the twisted tensor product commute in the sense that given $M^{\bullet} \in \text{MHM}(X; T_s, N)$ and $N^{\bullet} \in \text{MHM}(Y; T_s, N)$, we have a natural isomorphism in $\text{MHM}(X \times Y; T_s, N)$:

$$\mathbb{D}_X^T(M^{\bullet}) \stackrel{'}{\boxtimes} \mathbb{D}_Y^T(N^{\bullet}) \cong \mathbb{D}_{X \times Y}^T(M^{\bullet} \stackrel{'}{\boxtimes} N^{\bullet}).$$
(2.20)

For an object $M^{\bullet} \in \text{MHM}(X; T_s, N)$ whose weight filtration is a (suitable shifted) monodromy filtration of the nilpotent morphism N, there is a stronger notion of polarization which will be useful for us. A strong polarization of weight w of such an object M^{\bullet} is a morphism

$$\sigma: M^{\bullet} \longrightarrow \mathbb{D}_X^T(M^{\bullet})(-w)$$

in MHM(X), compatible with T_s and N, such that σ defines polarizations on the N-primitive parts of M^{\bullet} , compatible with Hodge filtrations; for precise conditions, see [45, p. 855]. A polarization on a pure Hodge module is a strong polarization (with N = 0); a strongly polarized mixed Hodge module is graded polarizable. The partial twist in the definition of \mathbb{D}_X^T implies that M^{\bullet} is of weight w if and only if M_1^{\bullet} , respectively $M_{\neq 1}^{\bullet}$ are of weights w, w - 1 in the sense of [45, p. 855].

Given strongly polarized mixed Hodge modules $M_i^{\bullet} \in \text{MHM}(X_i; T_s, N)$ of weight w_i for i = 1, 2, polarized by $\sigma_i : M_i^{\bullet} \to \mathbb{D}_{X_i}^T(M_i^{\bullet})(-w_i)$, there is an induced morphism σ in a commutative diagram

where the top map is $\sigma_1 \boxtimes \sigma_2$ and the right is the isomorphism (2.20). In general, it is *not clear* whether this morphism is a strong polarization of the tensor product $M_1^{\bullet} \boxtimes M_2^{\bullet}$; this result is not available in the literature. However, in this paper we only use this construction in cases where one of the monodromic mixed Hodge modules is essentially trivial, living on $X_1 = \text{pt}$ with N = 0, in which case it is easy to check that the resulting σ is a strong polarization.

Note also that if M^{\bullet} is strongly polarized by $\sigma: M^{\bullet} \to \mathbb{D}_X^T(M^{\bullet})(-w)$, then its Tate twist is also strongly polarized by the composition

$$M^{\bullet}(1) \xrightarrow{\sigma(1)} \mathbb{D}_X^T(M^{\bullet})(-w+1) \xrightarrow{\sim} \mathbb{D}_X^T(M^{\bullet}(1))(-w+2).$$
(2.21)

The notion of strong polarization leads to gluing, in the following way.

Theorem 2.21. Let $X = \bigcup_i U_i$ be an open cover of a space X, in any of the Zariski, étale or complex analytic topologies. Then:

(i) Suppose we are given mixed Hodge modules $M^{\bullet}, N^{\bullet} \in \text{MHM}(X)$, with morphisms $f_i : M^{\bullet}|_{U_i} \to N^{\bullet}|_{U_i}$ in $\text{MHM}(U_i)$ which agree on overlaps U_{ij} . Then there is a unique $f \in \text{Hom}_{\text{MHM}(X)}(M^{\bullet}, N^{\bullet})$ with $f|_{U_i} = f_i$.

(ii) Suppose we are given mixed Hodge modules $M_i^{\bullet} \in \text{MHM}(U_i; T_s, N)$, each equipped with a strong polarization σ_i . Suppose also that we are given isomorphisms

$$\alpha_{ij}: M_i^{\bullet}|_{U_{ij}} \longrightarrow M_j^{\bullet}|_{U_{ij}}$$

on intersections, commuting with the restrictions of the maps T_{si} , N_i and σ_i , with

$$\alpha_{jk}|_{U_{ijk}} \circ \alpha_{ij}|_{U_{ijk}} = \alpha_{ik}|_{U_{ijk}}$$

on triple intersections. Then there is a strongly polarized mixed Hodge module

$$M^{\bullet} \in \mathrm{MHM}(X; T_s, N),$$

restricting to M_i^{\bullet} on U_i .

Proof. To prove (i), it is enough to note that the f_i glue on the perverse sheaf and \mathscr{D} -module levels, respecting filtrations.

To prove (ii), we begin with the case of pure Hodge modules of fixed weight w. The data of a pure Hodge module consists of a pair of a filtered holonomic \mathscr{D} -module and a \mathbb{Q} -perverse sheaf, with an identification of the former with the complexification of the latter under the Riemann–Hilbert correspondence. Since both filtered holonomic \mathscr{D} -modules and \mathbb{Q} -perverse sheaves form stacks (both in the algebraic and the analytic case), this data glues over X. As for the (strong) polarization, $T_{si} = \text{id}$ and $N_i = 0$ glue to $T_s = \text{id}$ and N = 0, whereas the map σ on the perverse sheaf level glues from the maps σ_i once again from the stack property (now for morphisms) of perverse sheaves.

The conditions [45, §5.1.6, p. 955] which make such a pair a pure Hodge module come from local conditions as well as conditions on vanishing cycles; the latter glue by induction on the dimension. So strongly polarized pure Hodge modules form a stack. The case of mixed Hodge modules is similar: we need to glue filtrations and polarizations, as well as the maps T_{si} , N_i and σ_i , first on the level of perverse sheaves, and then checking the axioms, which are local or follow by induction.

Remark 2.22. Given a projective \mathbb{C} -scheme X, and a polarizable mixed Hodge module M^{\bullet} in MHM(X) on it, the second part of Theorem 2.19(iv) applied to $f : X \to \text{pt}$ shows that the hypercohomology $\mathbb{H}^*(X, M^{\bullet})$ carries a mixed Hodge structure. In particular, it carries a weight filtration and therefore has a weight polynomial, which will be useful in refinements of Donaldson–Thomas theory, see the discussion in Remark 6.14 below. So we need to glue polarizable objects from local data. On the other hand, graded polarizable mixed Hodge modules may fail to form a stack in the analytic category unless the polarizations glue. This is the reason for using the stronger form of polarization, which allows for gluing as shown above.

2.10. Mixed Hodge modules of vanishing cycles. By Saito's work [45–47], for a regular function $f: U \to \mathbb{C}$ on a smooth space U, the perverse nearby and vanishing cycle functors ψ_f^p, ϕ_f^p defined on perverse sheaves in §2.3 lift to functors

$$\psi_f^H, \phi_f^H : \mathrm{MHM}(U) \longrightarrow \mathrm{MHM}(U_0; T_s, N),$$

where $U_0 = f^{-1}(0)$. The actions of the finite order and nilpotent operators T_s , N are given by the semisimple part of the monodromy operator, and the logarithm of its unipotent part. The analogue

$$\phi_f^H \circ \mathbb{D}_U^T \cong \mathbb{D}_{U_0}^T \circ \phi_f^H$$

of Theorem 2.11(iv) is proved in [46]; to make this isomorphism work is the reason for the the twist in the definition of \mathbb{D}_U^T . Note also that [46, Th. 1.6] fits with the convention that T_s and N are defined on dual objects as $\mathbb{D}_U(T_s)^{-1}$ and $\mathbb{D}_U(N)$, respectively.

By [45, §5.2], if $M^{\bullet} \in \text{HM}(U)$ is a pure Hodge module, then a polarization of M^{\bullet} induces a strong polarization on the (mixed) Hodge module of vanishing cycles $\phi_f^H(M^{\bullet})$, of the same weight. In particular, if $M^{\bullet} = \mathbb{Q}_U^H[\dim U]$ is the canonical object with its canonical polarization from Theorem 2.19(ii), then $\phi_f^H(\mathbb{Q}_U^H[\dim U]) \in \text{MHM}(\text{Crit}(f); T_s, N)$ is a strongly polarized mixed Hodge module on the critical locus of f, with polarization

$$\sigma: \phi_f^H \left(\mathbb{Q}_U^H[\dim U] \right) \longrightarrow \mathbb{D}_{\operatorname{Crit}(f)}^T \circ \phi_f^H \left(\mathbb{Q}_U^H[\dim U] \right) (-\dim U).$$
(2.22)

Example 2.23. Define $f : \mathbb{C} \to \mathbb{C}$ by $f(z) = z^2$. Then $\operatorname{Crit}(f) = \{0\}$, and we obtain an object $\phi_f^H(\mathbb{Q}_{\mathbb{C}}^H[1])$ in MHM(pt; T_s, N), a one-dimensional polarized mixed Hodge structure with monodromy acting by $T_s = -\operatorname{id}$ and N = 0. For $g : \mathbb{C}^2 \to \mathbb{C}$ given by $g(z_1, z_2) = z_1^2 + z_2^2$, it is well known that

$$\phi_{z_1^2+z_2^2}^H \left(\mathbb{Q}_{\mathbb{C}^2}^H[2] \right) \cong \mathbb{Q}(-1),$$

with trivial monodromy action. Applying the Thom–Sebastiani formula for mixed Hodge modules [49, Th. 5.4], we see that

$$\phi_{z^2}^H \left(\mathbb{Q}_{\mathbb{C}}^H [1] \right) \stackrel{^{\mathrm{T}}}{\boxtimes} \phi_{z^2}^H \left(\mathbb{Q}_{\mathbb{C}}^H [1] \right) \cong \mathbb{Q}(-1)$$

in the category MHM(pt; T_s, N). The objects $\mathbb{Q}(1)$ and $\mathbb{Q}(-1)$ thus admit square roots under $\stackrel{\tau}{\boxtimes}$ in this category, which we will denote by $\mathbb{Q}(\frac{1}{2})$ and $\mathbb{Q}(-\frac{1}{2})$, where

$$\phi_{z^2}^H \left(\mathbb{Q}_{\mathbb{C}}^H [1] \right) = \mathbb{Q}(-\frac{1}{2}). \tag{2.23}$$

More explicitly, we have

and

$$\mathbb{Q}(-\frac{1}{2}) = (\mathbb{Q}(0), -\mathrm{id}, 0)$$
$$\mathbb{Q}(\frac{1}{2}) = (\mathbb{Q}(1), -\mathrm{id}, 0).$$

Define an object $\mathbb{Q}(\frac{n}{2}) \in \text{MHM}(\text{pt}; T_s, N)$ for each $n \in \mathbb{Z}$ by $\mathbb{Q}(\frac{n}{2}) = \mathbb{Q}(\frac{1}{2})^{\overset{T}{\boxtimes}^n}$ for $n \ge 0$, and $\mathbb{Q}(\frac{n}{2}) = \mathbb{Q}(-\frac{1}{2})^{\overset{T}{\boxtimes}^{-n}}$ for n < 0. For any space X with structure morphism $\pi : X \to \text{pt}$, and any $M^{\bullet} \in D^b \text{ MHM}(X; T_s, N)$, we define the $\frac{n}{2}$ twist of M^{\bullet} to be $M^{\bullet}(\frac{n}{2}) = M^{\bullet} \boxtimes^{\overset{T}{\boxtimes}} (\mathbb{Q}(\frac{n}{2}))$. If M^{\bullet} is strongly polarized, then this tensor product is also strongly polarized by the tensor polarization by our comments above.

Let U be a smooth space, $f: U \to \mathbb{C}$ a regular function, and $X = \operatorname{Crit}(f)$ its critical locus, as a subspace of U. The perverse sheaf of vanishing cycles $\mathcal{PV}_{U,f}^{\bullet} \in \operatorname{Perv}(X)$ from §2.4 has a lift to a mixed Hodge module $\mathcal{HV}_{U,f}^{\bullet}$ in $\operatorname{MHM}(X; T_s, N)$, defined for each $c \in f(X)$ by

$$\mathcal{HV}^{\bullet}_{U,f}|_{X_c} = \phi^H_{f-c} \left(\mathbb{Q}^H_U[\dim U] \right) \Big|_{X_c} \left(\frac{1}{2} \dim U \right) \in \mathrm{MHM}(X_c; T_s, N).$$
(2.24)

This mixed Hodge module inherits a strong polarization of weight 0 (compare (2.21) and (2.22))

$$\sigma_{U,f}^{H}: \mathcal{HV}_{U,f}^{\bullet} \longrightarrow \mathbb{D}_{X}^{T} (\mathcal{HV}_{U,f}^{\bullet}).$$
(2.25)

The twist $(\frac{1}{2} \dim U)$ in (2.24), using the notation of Example 2.23, is included for the same reason as the $(-1)^{\dim U}$ in the definition (2.7) of $\tau_{U,f}$. It makes $\mathcal{HV}_{U,f}^{\bullet}$ act naturally under transformations which change dimension — without it, the mixed Hodge module version of (5.15) below would have to include a twist $(\frac{1}{2}n)$ for $n = \dim V - \dim U$. Then

$$\mathcal{HV}_{U,f}^{\bullet}, \quad T_s: \mathcal{HV}_{U,f}^{\bullet} \to \mathcal{HV}_{U,f}^{\bullet}, \quad N: \mathcal{HV}_{U,f}^{\bullet} \to \mathcal{HV}_{U,f}^{\bullet}(-1), \quad \text{and} \quad \sigma_{U,f}^H: \mathcal{HV}_{U,f}^{\bullet} \to \mathbb{D}_X^T (\mathcal{HV}_{U,f}^{\bullet})$$

are related to

$$\mathcal{PV}_{U,f}^{\bullet}, \ \tau_{U,f}: \mathcal{PV}_{U,f}^{\bullet} \to \mathcal{PV}_{U,f}^{\bullet}, \ \text{and} \ \sigma_{U,f}: \mathcal{PV}_{U,f}^{\bullet} \xrightarrow{\cong} \mathbb{D}_X(\mathcal{PV}_{U,f}^{\bullet})$$

in §2.4 by

$$\mathcal{PV}_{U,f}^{\bullet} = \operatorname{rat}(\mathcal{HV}_{U,f}^{\bullet}), \ \tau_{U,f} = \operatorname{rat}(T_s) \circ \exp(2\pi i \operatorname{rat}(N)), \ \sigma_{U,f} = \operatorname{rat}(\sigma_{U,f}^H);$$

for the last statement, see Proposition A.1 in the Appendix.

The following Thom–Sebastiani type result is the analogue of Theorem 2.13.

Theorem 2.24. Let U, V be smooth spaces and $f: U \to \mathbb{C}$, $g: V \to \mathbb{C}$ be regular functions, so that $f \boxplus g: U \times V \to \mathbb{C}$ is given by $(f \boxplus g)(u, v) := f(u) + g(v)$. Set $X = \operatorname{Crit}(f)$ and $Y = \operatorname{Crit}(g)$ as subspaces of U, V, so that $\operatorname{Crit}(f \boxplus g) = X \times Y$. Then there is a natural isomorphism

$$\mathcal{TS}^{H}_{U,f,V,g}: \mathcal{HV}^{\bullet}_{U \times V, f \boxplus g} \xrightarrow{\cong} \mathcal{HV}^{\bullet}_{U,f} \boxtimes \mathcal{HV}^{\bullet}_{V,g} \text{ in } \mathrm{MHM}(X \times Y; T_s, N),$$
(2.26)

so that the following diagram commutes:

$$\begin{aligned}
\mathcal{H}\mathcal{V}_{U\times V,f\boxplus g}^{\bullet} \xrightarrow{\sigma_{U\times V,f\boxplus g}} & \mathbb{D}_{X\times Y}^{T}(\mathcal{H}\mathcal{V}_{U\times V,f\boxplus g}^{\bullet}) \\
\downarrow^{\mathcal{T}S_{U,f,V,g}^{H}} & \mathbb{D}_{X}^{T}(\mathcal{H}\mathcal{V}_{U,f}^{\bullet}) & \stackrel{\wedge}{\boxtimes} \\
\mathcal{H}\mathcal{V}_{V,g}^{\bullet} \xrightarrow{\sigma_{U,f}\boxtimes\sigma_{V,g}} & \mathbb{D}_{X}^{T}(\mathcal{H}\mathcal{V}_{U,f}^{\bullet}) & \cong & \mathbb{D}_{X\times Y}^{T}(\mathcal{H}\mathcal{V}_{U,f}^{\bullet}\boxtimes\mathcal{H}\mathcal{H}\mathcal{V}_{V,g}^{\bullet}).
\end{aligned}$$
(2.27)

Proof. The existence of the isomorphism (2.26) follows from the Thom–Sebastiani Theorem for mixed Hodge modules due to Saito [49, Th. 5.4], applied to $\mathcal{HV}_{U,f}^{\bullet}$. The diagram (2.27) exists by (2.20); its commutativity can be checked on the level of the underlying perverse sheaves which is (2.9), in light of Propositions A.1–A.2 in the Appendix. Note that (2.26) also includes the analogue of (2.10) in Theorem 2.13, according to which we have a matching of the monodromy actions

$$\tau_{U\times V,f\boxplus g}\cong \tau_{U,f}\boxtimes^{L}\tau_{V,g},$$

In this paper we will only ever apply Theorem 2.24 when $V = \mathbb{C}^n$, $g = z_1^2 + \cdots + z_n^2$ and $Y = \{0\}$. Combining (2.23) and (2.24) shows that

$$\mathcal{HV}^{\bullet}_{\mathbb{C},z^2} = \left(\mathbb{Q}(-\frac{1}{2})\right)(\frac{1}{2}) \cong \mathbb{Q}(0) \cong \mathbb{Q}^H_{\{0\}}.$$

Thus, by Theorem 2.24, $\mathbb{Q}_{\{0\}}^H \boxtimes \mathbb{Q}_{\{0\}}^H \cong \mathbb{Q}_{\{0\}}^H$, and induction on n, we see that

as (2.26) holds in MHM $(X \times Y; T_s, N)$ rather than just MHM $(X \times Y)$.

$$\mathcal{HV}^{ullet}_{\mathbb{C}^n, z_1^2 + \dots + z_n^2} \cong \mathbb{Q}^H_{\{0\}}.$$

As for (2.12), this isomorphism is natural up to sign, depending on a choice of orientation for the complex Euclidean space $(\mathbb{C}^n, dz_1^2 + \cdots + dz_n^2)$.

3. Action of symmetries on vanishing cycles

Here is our first main result.

Theorem 3.1. Let U, V be smooth \mathbb{C} -schemes, $\Phi, \Psi : U \to V$ étale morphisms, and $f: U \to \mathbb{C}, g: V \to \mathbb{C}$ regular functions with $g \circ \Phi = f = g \circ \Psi$. Write $X = \operatorname{Crit}(f)$ and $Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V, so that $\Phi|_X, \Psi|_X : X \to Y$ are étale morphisms. Suppose $\Phi|_X = \Psi|_X$. Then:

106

(a) As Φ, Ψ are étale, $d\Phi : TU \to \Phi^*(TV)$, $d\Psi : TU \to \Psi^*(TV)$ are isomorphisms of vector bundles. Restricting to the reduced \mathbb{C} -subscheme X^{red} of X, and using $\Phi|_{X^{\text{red}}} = \Psi|_{X^{\text{red}}}$ as $\Phi|_X = \Psi|_X$, gives isomorphisms

$$\begin{split} \mathrm{d}\Phi|_{X^{\mathrm{red}}},\mathrm{d}\Psi|_{X^{\mathrm{red}}}:TU|_{X^{\mathrm{red}}} &\longrightarrow \Phi|_{X^{\mathrm{red}}}^{*}(TV),\\ and thus \quad \mathrm{d}\Psi|_{X^{\mathrm{red}}}^{-1}\circ\mathrm{d}\Phi|_{X^{\mathrm{red}}}:TU|_{X^{\mathrm{red}}} \longrightarrow TU|_{X^{\mathrm{red}}}. \end{split}$$

Hence

$$\det\left(\mathrm{d}\Psi|_{X^{\mathrm{red}}}^{-1}\circ\mathrm{d}\Phi|_{X^{\mathrm{red}}}\right):X^{\mathrm{red}}\longrightarrow\mathbb{C}\setminus\{0\}$$

is a regular function. Then $\det(d\Psi|_{X^{red}}^{-1} \circ d\Phi|_{X^{red}})$ is a locally constant map

$$X^{\mathrm{red}} \longrightarrow \{\pm 1\} \subset \mathbb{C} \setminus \{0\}.$$

(b) Definition 2.15 defines isomorphisms $\mathcal{PV}_{\Phi}, \mathcal{PV}_{\Psi} : \mathcal{PV}_{U,f}^{\bullet} \to \Phi|_X^* (\mathcal{PV}_{V,g}^{\bullet})$ in $\operatorname{Perv}(X)$. These are related by

$$\mathcal{PV}_{\Phi} = \det\left(\mathrm{d}\Psi|_{X^{\mathrm{red}}}^{-1} \circ \mathrm{d}\Phi|_{X^{\mathrm{red}}}\right) \cdot \mathcal{PV}_{\Psi},\tag{3.1}$$

regarding det $\left(d\Psi|_{X^{\text{red}}}^{-1} \circ d\Phi|_{X^{\text{red}}}\right)$: $X \to \{\pm 1\}$ as a locally constant map of topological spaces, where X, X^{red} have the same topological space.

The analogues of these results also hold for \mathscr{D} -modules and mixed Hodge modules on \mathbb{C} -schemes, and (with Φ, Ψ local biholomorphisms and f, g analytic functions) for perverse sheaves, \mathscr{D} -modules and mixed Hodge modules on complex analytic spaces, as in §2.6–§2.10.

By taking U = V, f = g, Φ an isomorphism and $\Psi = id_U$, we deduce a result on the action of symmetries on perverse sheaves of vanishing cycles:

Corollary 3.2. Let U be a smooth \mathbb{C} -scheme, $\Phi: U \to U$ an isomorphism, and $f: U \to \mathbb{C}$ be regular with $f \circ \Phi = f$. Write $X = \operatorname{Crit}(f)$ as a \mathbb{C} -subscheme of U and X^{red} for its reduced \mathbb{C} -subscheme, and suppose $\Phi|_X = \operatorname{id}_X$. Then $\det(\mathrm{d}\Phi|_{X^{\operatorname{red}}} : TU|_{X^{\operatorname{red}}} \to TU|_{X^{\operatorname{red}}})$ is a locally constant map $X^{\operatorname{red}} \to \{\pm 1\}$, and $\mathcal{PV}_{\Phi}: \mathcal{PV}_{U,f}^{\bullet} \xrightarrow{\cong} \mathcal{PV}_{U,f}^{\bullet}$ in $\operatorname{Perv}(X)$ from Definition 2.15 is multiplication by $\det(\mathrm{d}\Phi|_{X^{\operatorname{red}}}) = \pm 1$. The analogues hold in the settings of §2.6–§2.10.

Example 3.3. Let $U = V = \mathbb{C}^n$ and $f(z_1, \ldots, z_n) = g(z_1, \ldots, z_n) = z_1^2 + \cdots + z_n^2$, so that $X = Y = \{0\} \subset \mathbb{C}^n$. Let $\Phi, \Psi \in O(n, \mathbb{C})$ be orthogonal matrices, so that $\det \Phi, \det \Psi \in \{\pm 1\}$ and $\Phi, \Psi : \mathbb{C}^n \to \mathbb{C}^n$ are isomorphisms with $f = g \circ \Phi = g \circ \Phi$ and $\Phi|_{\{0\}} = \Psi|_{\{0\}} = \mathrm{id}_{\{0\}}$. In Theorem 3.1(a) we have

$$\mathrm{d}\Psi|_{X^{\mathrm{red}}}^{-1} \circ \mathrm{d}\Phi|_{X^{\mathrm{red}}} = \Psi^{-1} \circ \Phi : \mathbb{C}^n \longrightarrow \mathbb{C}^n,$$

so that $\det(d\Psi|_{X^{\mathrm{red}}}^{-1} \circ d\Phi|_{X^{\mathrm{red}}}) = \det \Psi^{-1} \det \Phi = \pm 1.$

For Theorem 3.1(b), Example 2.16 shows that $\mathcal{PV}_{\Phi}, \mathcal{PV}_{\Psi} : A_{\{0\}} \to A_{\{0\}}$ are multiplication by det Φ , det Ψ , so $\mathcal{PV}_{\Phi} = (\det \Psi^{-1} \det \Phi) \cdot \mathcal{PV}_{\Psi}$, as in (3.1).

The proof of Theorem 3.1(b) uses the following proposition. To interpret it, pretend for simplicity that the étale morphisms $\pi_U|_{W_t} : W_t \to U$ in (b) are invertible. Then

$$\Theta_t := \pi_V |_{W_t} \circ \pi_U |_{W_t}^{-1}$$

for $t \in \mathbb{C}$ are a 1-parameter family of morphisms $U \to V$, which satisfy $f = g \circ \Theta_t$ and $\Theta_t|_X = \Phi|_X = \Psi|_X$ for $t \in \mathbb{C}$, with $\Theta_0 = \Phi$ and $\Theta_1 = \Psi$. Thus, modulo taking étale covers of U, the family $\{\Theta_t : t \in \mathbb{C}\}$ interpolates between Φ and Ψ .

Proposition 3.4. Let U, V be smooth \mathbb{C} -schemes, let $\Phi, \Psi : U \to V$ be étale morphisms, and let $f : U \to \mathbb{C}, g : V \to \mathbb{C}$ be regular functions with $g \circ \Phi = f = g \circ \Psi$. Write $X = \operatorname{Crit}(f)$ and $Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V, so that $\Phi|_X, \Psi|_X : X \to Y$ are étale. Suppose $\Phi|_X = \Psi|_X$, and $x \in X$ such that $d\Psi|_x^{-1} \circ d\Phi|_x : T_x U \to T_x U$ satisfies

$$\left(\mathrm{d}\Psi|_x^{-1} \circ \mathrm{d}\Phi|_x - \mathrm{id}_{T_xU}\right)^2 = 0$$

Then there exist a smooth \mathbb{C} -scheme W and morphisms $\pi_{\mathbb{C}} : W \to \mathbb{C}, \pi_U : W \to U, \pi_V : W \to V$ and $\iota : \mathbb{C} \to W$ such that:

- (a) $\pi_{\mathbb{C}} \circ \iota(t) = t$, $\pi_U \circ \iota(t) = x$ and $\pi_V \circ \iota(t) = \Phi(x)$ for all $t \in \mathbb{C}$;
- (b) $\pi_{\mathbb{C}} \times \pi_U : W \to \mathbb{C} \times U$ and $\pi_{\mathbb{C}} \times \pi_V : W \to \mathbb{C} \times V$ are étale. Thus, $W_t := \pi_{\mathbb{C}}^{-1}(t)$ is a smooth \mathbb{C} -scheme for each $t \in \mathbb{C}$, and $\pi_U|_{W_t} : W_t \to U$, $\pi_V|_{W_t} : W_t \to V$ are étale, and $\iota(t) \in W_t$ with $\pi_U : \iota(t) \mapsto x, \pi_V : \iota(t) \mapsto \Phi(x)$;
- (c) $h := f \circ \pi_U = g \circ \pi_V : W \to \mathbb{C}$. Thus,

$$(\pi_{\mathbb{C}} \times \pi_U)|_Z : Z \longrightarrow \mathbb{C} \times X$$
 and $(\pi_{\mathbb{C}} \times \pi_V)|_Z : Z \longrightarrow \mathbb{C} \times Y$

are étale, where $Z := \operatorname{Crit}(h)$;

- (d) $\Phi|_X \circ \pi_U|_Z = \Psi|_X \circ \pi_U|_Z = \pi_V|_Z : Z \to Y \subseteq V$; and
- (e) $\Phi \circ \pi_U|_{W_0} = \pi_V|_{W_0}$ and $\Psi \circ \pi_U|_{W_1} = \pi_V|_{W_1}$, for W_0, W_1 as in (b).

We will prove Proposition 3.4 in $\S3.1$, and Theorem 3.1 in $\S3.2-\S3.4$.

3.1. **Proof of Proposition 3.4.** Let $U, V, \Phi, \Psi, f, g, X, Y, x$ be as in Proposition 3.4. Choose a Zariski open neighbourhood V' of $\Phi(x) = \Psi(x)$ in V and étale coordinates $(z_1, \ldots, z_n) : V' \to \mathbb{C}^n$ on V', with

$$z_1 = \dots = z_n = 0$$

at $\Phi(x)$. Let *m* be the rank of the symmetric matrix $\left(\frac{\partial^2 g}{\partial z_i \partial z_j}|_{\Phi(x)}\right)_{i,j=1}^n$, so that $m \in \{0, \ldots, n\}$. By applying an element of $\operatorname{GL}(n, \mathbb{C})$ to the coordinates (z_1, \ldots, z_n) we can suppose that

$$\frac{\partial^2 g}{\partial z_i \partial z_j}\Big|_{\Phi(x)} = \begin{cases} 1, & i = j \in \{1, \dots, m\}, \\ 0, & \text{otherwise.} \end{cases}$$
(3.2)

Then $\frac{1}{2} \frac{\partial g}{\partial z_i}$ agrees with z_i to first order at $\Phi(x)$ for $i = 1, \ldots, m$, so replacing z_i by $\frac{1}{2} \frac{\partial g}{\partial z_i}$ for $i = 1, \ldots, m$ and making V' smaller, we can suppose (3.2) holds and z_1, \ldots, z_m lie in the ideal $\left(\frac{\partial g}{\partial z_i}, i = 1, \ldots, n\right)$ in $\mathcal{O}_{V'}$. Thus we may write

$$z_i = \sum_{j=1}^n A_{ij} \cdot \frac{\partial g}{\partial z_j}, \qquad i = 1, \dots, m,$$
(3.3)

where $A_{ij}: V' \to \mathbb{A}^1$ are regular functions for i = 1, ..., m and j = 1, ..., n. Taking $\frac{\partial}{\partial z_j}$ of (3.3) for j = 1, ..., m and using (3.2) gives

$$A_{ij}|_{\Phi(x)} = \begin{cases} 1, & i = j, \ i, j \in \{1, \dots, m\}, \\ 0, & i \neq j, \ i, j \in \{1, \dots, m\}. \end{cases}$$
(3.4)

Set $U' = \Phi^{-1}(V') \cap \Psi^{-1}(V')$, so that U' is a Zariski open neighbourhood of x in U. Define étale coordinates $(x_1, \ldots, x_n) : U' \to \mathbb{C}^n$ and $(y_1, \ldots, y_n) : U' \to \mathbb{C}^n$ by $x_i = z_i \circ \Phi$ and $y_i = z_i \circ \Psi$, so that $x_1 = \cdots = x_n = y_1 = \cdots = y_n = 0$ at x. Since $f = g \circ \Phi = g \circ \Psi$ we have $\frac{\partial f}{\partial x_j} = \frac{\partial g}{\partial z_j} \circ \Phi$ and $\frac{\partial f}{\partial y_i} = \frac{\partial g}{\partial z_i} \circ \Psi$. Thus (3.2) and (3.3) imply that

$$\frac{\partial^2 f}{\partial x_i \partial x_j}\Big|_x = \frac{\partial^2 f}{\partial y_i \partial y_j}\Big|_x = \begin{cases} 1, & i = j \in \{1, \dots, m\}, \\ 0, & \text{otherwise,} \end{cases}$$
(3.5)

$$x_i = \sum_{j=1}^n (A_{ij} \circ \Phi) \cdot \frac{\partial f}{\partial x_j}, \quad y_i = \sum_{j=1}^n (A_{ij} \circ \Psi) \cdot \frac{\partial f}{\partial y_j}, \quad i = 1, \dots, m.$$
(3.6)

Now $d\Phi|_x : T_x U \to T_{\Phi(x)} V$ maps $\frac{\partial}{\partial x_j} \mapsto \frac{\partial}{\partial z_j}$, as $x_j = z_j \circ \Phi$, and $d\Psi|_x : T_x U \to T_{\Phi(x)} V$ maps $\frac{\partial}{\partial y_j} \mapsto \frac{\partial}{\partial z_j}$. Hence $d\Psi|_x^{-1} \circ d\Phi|_x : T_x U \to T_x U$ maps $\frac{\partial}{\partial x_j} \mapsto \frac{\partial}{\partial y_j} = \sum_{i=1}^n \frac{\partial x_i}{\partial y_j} \cdot \frac{\partial}{\partial x_i}$. Define $B_{ij} \in \mathbb{C}$ for i, j = 1, ..., n by

$$\delta_{ij} + B_{ij} = \frac{\partial x_i}{\partial y_j}\Big|_x. \tag{3.7}$$

Then $(\delta_{ij} + B_{ij})_{i,j=1}^n$ is the matrix of $d\Psi|_x^{-1} \circ d\Phi|_x$ w.r.t. the basis $\frac{\partial}{\partial x_1}, \ldots, \frac{\partial}{\partial x_n}$, and $(B_{ij})_{i,j=1}^n$ is the matrix of $d\Psi|_x^{-1} \circ d\Phi|_x - \mathrm{id}_{T_xU}$, so by assumption $(B_{ij})^2 = 0$. Therefore the inverse matrix of $(\delta_{ij} + B_{ij})$ is $(\delta_{ij} - B_{ij})$, so (3.7) gives

$$\delta_{ij} - B_{ij} = \frac{\partial y_i}{\partial x_j}\Big|_x. \tag{3.8}$$

More generally, $(\delta_{ij} + tB_{ij})$ is invertible for $t \in \mathbb{C}$, with inverse $(\delta_{ij} - tB_{ij})$. Now $\Phi|_X = \Psi|_X$ implies that $d\Psi|_x^{-1} \circ d\Phi|_x$ is the identity on $T_xX \subseteq T_xU$, and $T_xX = \text{Ker}(\text{Hess}_x f) = \langle \frac{\partial}{\partial x_{m+1}}, \dots, \frac{\partial}{\partial x_n} \rangle$ by (3.5), so

$$B_{ij} = 0$$
 for all $i = 1, ..., n$ and $j = m + 1, ..., n$. (3.9)

We have $\frac{\partial^2 f}{\partial y_i \partial y_l}\Big|_x = \sum_{j,k} \frac{\partial x_j}{\partial x_i} \frac{\partial^2 f}{\partial x_j \partial x_k} \frac{\partial x_k}{\partial y_l}\Big|_x$, $\frac{\partial^2 f}{\partial x_i \partial x_l}\Big|_x = \sum_{j,k} \frac{\partial y_j}{\partial x_i} \frac{\partial^2 f}{\partial y_j \partial y_k} \frac{\partial y_k}{\partial x_l}\Big|_x$, which by (3.5) and (3.7)-(3.9) give equations equivalent to

$$B_{ij} + B_{ji} = \sum_{k=1}^{n} B_{ki} B_{kj} = 0$$
 for all $i, j = 1, \dots, m$. (3.10)

Define regular $t', x'_1, \ldots, x'_n, y'_1, \ldots, y'_n, z'_1, \ldots, z'_n : \mathbb{C} \times U' \times V' \to \mathbb{C}$ by

$$\begin{split} t'(t,u,v) &= t, & x'_i(t,u,v) = x_i(u) = z_i \circ \Phi(u), \\ y'_i(t,u,v) &= y_i(u) = z_i \circ \Psi(u), & z'_i(t,u,v) = z_i(v). \end{split}$$

Then $(t', y'_1, \ldots, y'_n, z'_1, \ldots, z'_n)$ are étale coordinates on $\mathbb{C} \times U' \times V'$.

Let S be an affine Zariski open neighbourhood of $\mathbb{C} \times (x, \Phi(x))$ in $\mathbb{C} \times U' \times V'$, satisfying a series of smallness conditions we will give during the proof. Regard $(t', y'_1, \ldots, y'_n, z'_1, \ldots, z'_n)$ as étale coordinates on S, and write $\pi_{\mathbb{C}}: S \to \mathbb{C}, \pi_U: S \to U, \pi_V: S \to V$ for the projections. We will work with (sheaves of) ideals in \mathcal{O}_S , using notation $(x'_i - z'_i, i = 1, \dots, n)$ to denote the ideal generated by the functions $x'_1 - z'_1, \ldots, x'_n - z'_n$, and

$$f \circ \pi_U - g \circ \pi_V \in \left(x'_i - z'_i, \ i = 1, \dots, n\right)$$

to mean that $f \circ \pi_U - g \circ \pi_V \in H^0(\mathcal{O}_S)$ is a section of the ideal $(x'_i - z'_i, i = 1, \dots, n)$. Write $I_X \subset \mathcal{O}_U, I_Y \subset \mathcal{O}_V$ for the ideals of functions on U, V vanishing on X, Y, and

$$\pi_U^{-1}(I_X), \pi_V^{-1}(I_Y) \subset \mathcal{O}_S$$

for the preimage ideals.

Since $x_i = z_i \circ \Phi$, the functions $x'_i - z'_i$ for i = 1, ..., n vanish on the smooth, closed \mathbb{C} subscheme $(\mathbb{C} \times (\mathrm{id} \times \Phi)(U)) \cap S$ in S, and locally these functions cut out this \mathbb{C} -subscheme. So making S smaller we can suppose $(\mathbb{C} \times (\mathrm{id} \times \Phi)(U)) \cap S$ is the \mathbb{C} -subscheme

$$x_1' - z_1' = \dots = x_n' - z_n' = 0$$

in S. As $f = g \circ \Phi$, the function $f \circ \pi_U - g \circ \pi_V$ is zero on $(\mathbb{C} \times (\mathrm{id} \times \Phi)(U)) \cap S$. Hence

$$f \circ \pi_U - g \circ \pi_V \in \left(x'_i - z'_i, \ i = 1, \dots, n\right) \subset \mathcal{O}_S.$$

$$(3.11)$$

Lifting (3.11) from $(x'_i - z'_i, i = 1, ..., n)$ to $(x'_i - z'_i, i = 1, ..., n)^2$, making S smaller if necessary, we may choose regular $C_i : S \to \mathbb{C}$ for i = 1, ..., n with

$$f \circ \pi_U - g \circ \pi_V - \sum_{i=1}^n C_i \cdot (x'_i - z'_i) \in (x'_i - z'_i, \ i = 1, \dots, n)^2.$$
(3.12)

Apply $\frac{\partial}{\partial z'_i}$ to (3.12), using the étale coordinates $(t', y'_1, \ldots, y'_n, z'_1, \ldots, z'_n)$ on S. Since

$$\frac{\partial}{\partial z'_i} (g \circ \pi_V) = \frac{\partial g}{\partial z_i} \circ \pi_V \quad \text{and} \quad \frac{\partial}{\partial z'_i} (f \circ \pi_U) = 0 = \frac{\partial x'_j}{\partial z'_i},$$

this gives

$$C_i - \frac{\partial g}{\partial z_i} \circ \pi_V \in \left(x'_i - z'_i, \ i = 1, \dots, n \right).$$

Combining this with (3.12) yields

$$f \circ \pi_U - g \circ \pi_V - \sum_{i=1}^n \left(\frac{\partial g}{\partial z_i} \circ \pi_V\right) \cdot \left(x'_i - z'_i\right) \in \left(x'_i - z'_i, \ i = 1, \dots, n\right)^2$$

So making S smaller we can choose regular $D_{ij}: S \to \mathbb{C}$ for $i, j = 1, \ldots, n$ with $D_{ij} = D_{ji}$ and

$$f \circ \pi_U - g \circ \pi_V = \sum_{i=1}^n \left(\frac{\partial g}{\partial z_i} \circ \pi_V \right) \cdot (x'_i - z'_i) + \sum_{i,j=1}^n D_{ij} \cdot (x'_i - z'_i)(x'_j - z'_j).$$
(3.13)

Similarly, starting from $y_i = z_i \circ \Psi$ and $f = g \circ \Psi$ we may choose regular $E_{ij} : S \to \mathbb{C}$ for i, j = 1, ..., n with $E_{ij} = E_{ji}$ and

$$f \circ \pi_U - g \circ \pi_V = \sum_{i=1}^n \left(\frac{\partial g}{\partial z_i} \circ \pi_V\right) \cdot (y'_i - z'_i) + \sum_{i,j=1}^n E_{ij} \cdot (y'_i - z'_i)(y'_j - z'_j).$$
(3.14)

Applying $\frac{\partial^2}{\partial z'_i \partial z'_j}$ to (3.13) and (3.14), restricting to $(t, x, \Phi(x))$ for $t \in \mathbb{C}$, noting that $x'_i = y'_i = z'_i = 0$ at $(t, x, \Phi(x))$, and using (3.2), we deduce that

$$D_{ij}(t, x, \Phi(x)) = E_{ij}(t, x, \Phi(x)) = \frac{1}{2} \frac{\partial^2 g}{\partial z_i \partial z_j} \Big|_{\Phi(x)}$$

$$= \begin{cases} \frac{1}{2}, & i = j \in \{1, \dots, m\}, \\ 0, & \text{otherwise.} \end{cases}$$
(3.15)

Summing 1 - t' times (3.13) with t' times (3.14) and rearranging yields

$$f \circ \pi_{U} - g \circ \pi_{V} = \sum_{i=1}^{n} \left[\frac{\partial g}{\partial z_{i}} \circ \pi_{V} + 2t'(1-t') \sum_{j=1}^{n} (D_{ij} - E_{ij})(x'_{j} - y'_{j}) \right] \\ \cdot \left((1-t')x'_{i} + t'y'_{i} - z'_{i} \right) \\ + \sum_{i,j=1}^{n} \left[(1-t')D_{ij} + t'E_{ij} \right] \cdot \left((1-t')x'_{i} + t'y'_{i} - z'_{i} \right) \left((1-t')x'_{j} + t'y'_{j} - z'_{j} \right) \\ + \sum_{i,j=1}^{n} t'(1-t') \left[t'D_{ij} + (1-t')E_{ij} \right] \cdot (x'_{i} - y'_{i})(x'_{j} - y'_{j}).$$
(3.16)

Since $x'_i - y'_i = (x_i - y_i) \circ \pi_U$, and $(x_i - y_i)|_X = z_i \circ \Phi|_X - z_i \circ \Psi|_X = 0$ as $\Phi|_X = \Psi|_X$, we see that $x'_i - y'_i \in \pi_U^{-1}(I_X)$. Thus making S smaller if necessary, we may choose regular $F_{ij} : S \to \mathbb{C}$ such that

$$x'_{i} - y'_{i} = \sum_{j=1}^{n} F_{ij} \cdot \left(\frac{\partial f}{\partial y_{j}} \circ \pi_{U}\right) \quad \text{for } i = 1, \dots, n.$$

$$(3.17)$$

110

Furthermore, by (3.6) when $i = 1, \ldots, m$ we may take

$$F_{ij} = \left(\sum_{k=1}^{n} (A_{ik} \circ \Phi) \cdot \frac{\partial y_j}{\partial x_k} - A_{ij} \circ \Psi\right) \circ \pi_U.$$

Restricting to $(t, x, \Phi(x))$ and using (3.4), (3.8), (3.9) and $\Phi(x) = \Psi(x)$ gives

$$F_{ij}(t, x, \Phi(x)) = -B_{ji}$$
 for $i = 1, \dots, m$ and $j = 1, \dots, n$. (3.18)

Applying $\frac{\partial}{\partial x'_i}$ to equation (3.13) shows that

$$\frac{\partial g}{\partial z_i} \circ \pi_V - \frac{\partial f}{\partial x_i} \circ \pi_U \in (x'_j - z'_j, \ j = 1, \dots, n).$$

Thus we may write

$$\frac{\partial g}{\partial z_i} \circ \pi_V = \sum_{j=1}^n \left(\frac{\partial y_j}{\partial x_i} \circ \pi_U \right) \cdot \left(\frac{\partial f}{\partial y_j} \circ \pi_U \right) + \sum_{j=1}^n G_{ij} \cdot (x'_j - z'_j), \tag{3.19}$$

where $G_{ij}: S \to \mathbb{C}$ are regular. Applying $\frac{\partial}{\partial z'_j}$ to (3.19), restricting to $(t, x, \Phi(x))$ and using (3.2) yields

$$G_{ij}(t, x, \Phi(x)) = \begin{cases} -1, & i = j \in \{1, \dots, m\}, \\ 0, & \text{otherwise.} \end{cases}$$
(3.20)

From (3.17) and (3.19) we see that

$$\frac{\partial g}{\partial z_i} \circ \pi_V = \sum_{j=1}^n H_{ij} \cdot \left(\frac{\partial f}{\partial y_j} \circ \pi_U\right) + \sum_{j=1}^n G_{ij} \cdot \left((1-t')x_j' + t'y_j' - z_j'\right),\tag{3.21}$$

where $H_{ij} = \frac{\partial y_j}{\partial x_i} \circ \pi_U + t' \sum_{k=1}^n G_{ik} F_{kj}$, so that from equations (3.8), (3.9), (3.18) and (3.20) we deduce that

$$H_{ij}(t, x, \Phi(x)) = \delta_{ij} - (1-t)B_{ji}.$$
(3.22)

Combining (3.16), (3.17) and (3.21) gives

$$f \circ \pi_{U} - g \circ \pi_{V} - \sum_{i=1}^{n} I_{i} \cdot \left((1 - t')x_{i}' + t'y_{i}' - z_{i}' \right) - \sum_{i,j,k,l=1}^{n} t'(1 - t') \left[t'D_{ij} + (1 - t')E_{ij} \right] F_{ik}F_{jl} \cdot \left(\frac{\partial f}{\partial y_{k}} \circ \pi_{U} \right) \left(\frac{\partial f}{\partial y_{l}} \circ \pi_{U} \right) \in \left((1 - t')x_{i}' + t'y_{i}' - z_{i}', \ i = 1, \dots, n \right)^{2},$$
(3.23)

where
$$I_i = \sum_{j=1}^{n} \left[H_{ij} + 2t'(1-t') \sum_{k=1}^{n} (D_{ik} - E_{ik}) F_{kj} \right] \cdot \left(\frac{\partial f}{\partial y_j} \circ \pi_U \right).$$
 (3.24)

Consider the matrix of functions $[H_{ij} + \cdots]_{i,j=1}^n$ appearing in (3.24). Equations (3.15) and (3.22) imply that at $(t, x, \Phi(x))$ this reduces to $(\delta_{ij} - (1-t)B_{ji})$, which is invertible from above. Thus, making S smaller, we can suppose that $[H_{ij} + \cdots]_{i,j=1}^n$ in (3.24) is an invertible matrix on S. Write $[J_{ij}]_{i,j=1}^n$ for the inverse matrix. Then we have

$$\sum_{i,j,k,l=1}^{n} t'(1-t') \left[t'D_{ij} + (1-t')E_{ij} \right] F_{ik}F_{jl} \cdot \left(\frac{\partial f}{\partial y_k} \circ \pi_U \right) \left(\frac{\partial f}{\partial y_l} \circ \pi_U \right)$$

= $t'(1-t') \sum_{i,j=1}^{n} K_{ij} \cdot I_i I_j$, where
$$K_{ij} = \sum_{k,l,p,q=1}^{n} t'(1-t') \left[t'D_{kl} + (1-t')E_{kl} \right] F_{kp}F_{lq}J_{pi}J_{qj}.$$
 (3.25)

Using (3.9), (3.10), (3.15) and (3.18) we find that

$$K_{ij}(t, x, \Phi(x)) = 0 \quad \text{for all } t \in \mathbb{C}.$$
(3.26)

Combining (3.23) and (3.25), making S smaller if necessary we may write

$$f \circ \pi_U - g \circ \pi_V = \sum_{i=1}^n I_i \cdot \left((1 - t') x'_i + t' y'_i - z'_i \right) + t' (1 - t') \sum_{i,j=1}^n K_{ij} \cdot I_i I_j$$

+
$$\sum_{i,j=1}^n L_{ij} \cdot \left((1 - t') x'_i + t' y'_i - z'_i \right) \left((1 - t') x'_j + t' y'_j - z'_j \right), \qquad (3.27)$$

for regular $L_{ij}: S \to \mathbb{C}$ for $i, j = 1, \ldots, n$.

Write $(r_{ij})_{i,j=1}^n$ for the coordinates on \mathbb{C}^{n^2} . Let T be a Zariski open neighbourhood of

 $\mathbb{C} \times \left(x, \Phi(x), (0)_{i,j=1}^n\right)$

in $S \times \mathbb{C}^{n^2}$ to be chosen shortly, and let W be the closed C-subscheme of T defined by

$$W = \left\{ \left(t, u, v, (r_{ij})_{i,j=1}^{n}\right) \in T \subseteq S \times \mathbb{C}^{n^{2}} \subseteq \mathbb{C} \times U \times V \times \mathbb{C}^{n^{2}} : \\ \left((1-t')x'_{i}+t'y'_{i}-z'_{i}\right)(t, u, v) = \sum_{j=1}^{n} r_{ij} \cdot I_{j}(t, u, v), \quad i=1,\dots,n, \\ r_{ij}+t(1-t)K_{ij}(t, u, v) + \sum_{k,l=1}^{n} L_{kl}(t, u, v) \cdot r_{ki}r_{lj} = 0, \quad i, j = 1,\dots,n \right\}.$$
(3.28)

Define \mathbb{C} -scheme morphisms $\pi_{\mathbb{C}} : W \to \mathbb{C}, \pi_U : W \to U$, and $\pi_V : W \to V$ to map $(t, u, v, (r_{ij})_{i,j=1}^n)$ to t, u, v, respectively.

At $(t, x, \Phi(x)) \in S$ for $t \in \mathbb{C}$ we have $x'_i = y'_i = z'_i = 0$, and $I_i = 0$ by (3.24) as $\frac{\partial f}{\partial y_j}\Big|_x = 0$, and $K_{ij} = 0$ by (3.26). Hence $(t, x, \Phi(x), (0)_{i,j=1}^n)$ satisfies the equations of (3.28), and lies in W. Define $\iota : \mathbb{C} \to W$ by

$$\iota(t) = (t, x, \Phi(x), (0)_{i,j=1}^n).$$
(3.29)

Now $T \subseteq \mathbb{C} \times U \times V \times \mathbb{C}^{n^2}$ is smooth of dimension $1 + n + n + n^2$, and in (3.28) we impose $n + n^2$ equations, so the expected dimension of W is $(1 + 2n + n^2) - (n + n^2) = n + 1$. The linearizations of the $n + n^2$ equations in (3.28) at

$$(t, u, v, (r_{ij})_{i,j=1}^n) = (t, x, \Phi(x), (0)_{i,j=1}^n) = \iota(t)$$

are

$$dy_i|_x(\delta u) - dz_i|_{\Phi(x)}(\delta v) = 0, \quad i = 1, \dots, n,$$

$$\delta r_{ij} + dK_{ij}|_{(t,x,\Phi(x))}(\delta t \oplus \delta u \oplus \delta v) = 0, \quad i, j = 1, \dots, n,$$
(3.30)

for $\delta t \in T_t \mathbb{C}$, $\delta u \in T_x U$, $\delta v \in T_{\Phi(x)} V$, and $(\delta r_{ij})_{i,j=1}^n \in T_{(0)_{i,j=1}^n} \mathbb{C}^{n^2}$, where we have used $dx'_i = dy'_i$ and $I_j = K_{ij} = 0$ at $(t, x, \Phi(x))$. As $dy_1|_x, \ldots, dy_n|_x$ are a basis for $T_x^* U$, equations (3.30) are transverse, so W is smooth of dimension n+1 near $\iota(t)$. Hence, taking T small enough, we can suppose W is smooth.

It remains to prove Proposition 3.4(a)–(e). Part (a) is immediate from (3.29). For (b), the vector space of solutions $(\delta t, \delta u, \delta v, (\delta r_{ij})_{i,j=1}^n)$ to (3.30) is $T_{\iota(t)}W$, where

$$d(\pi_{\mathbb{C}} \times \pi_U)|_{\iota(t)} : T_{\iota(t)}W \to T_{(t,x)}(\mathbb{C} \times U)$$

and $d(\pi_{\mathbb{C}} \times \pi_V)|_{\iota(t)} : T_{\iota(t)}W \to T_{(t,\Phi(x))}(\mathbb{C} \times V) \text{ map } (\delta t, \delta u, \delta v, (\delta r_{ij})_{i,j=1}^n)$ to $(\delta t, \delta u)$ and $(\delta t, \delta v)$. By (3.30), these are isomorphisms, so $\pi_{\mathbb{C}} \times \pi_U$ and $\pi_{\mathbb{C}} \times \pi_V$ are étale near $\iota(\mathbb{C})$. Making T, W smaller, we can suppose $\pi_{\mathbb{C}} \times \pi_U$ and $\pi_{\mathbb{C}} \times \pi_V$ are étale.

112

For (c), we have

$$\left(f \circ \pi_W - g \circ \pi_U \right) \left(t, u, v, (r_{ij})_{i,j=1}^n \right) = f(u) - g(v) = (f \circ \pi_U - g \circ \pi_V) (t, u, v)$$

$$= \sum_{i=1}^n I_i \cdot \left((1 - t') x'_i + t' y'_i - z'_i \right) + t' (1 - t') \sum_{i,j=1}^n K_{ij} I_i I_j$$

$$+ \sum_{i,j=1}^n L_{ij} \cdot \left((1 - t') x'_i + t' y'_i - z'_i \right) \left((1 - t') x'_j + t' y'_j - z'_j \right)$$

$$= \sum_{i=1}^n I_i \cdot \left(\sum_{j=1}^n r_{ij} \cdot I_j \right) + t' (1 - t') \sum_{i,j=1}^n K_{ij} I_i I_j$$

$$+ \sum_{i,j=1}^n L_{ij} \cdot \left(\sum_{k=1}^n r_{ik} \cdot I_k \right) \left(\sum_{l=1}^n r_{jl} \cdot I_l \right)$$

$$= \sum_{i,j=1}^n I_i I_j \cdot \left[r_{ij} + t' (1 - t') K_{ij} + \sum_{k,l=1}^n L_{kl} \cdot r_{ki} r_{lj} \right] = 0,$$

using (3.27) in the third step, the first equation of (3.28) in the fourth, rearranging and exchanging labels i, k and j, l in the fifth, and the second equation of (3.28) in the sixth. Hence $f \circ \pi_U - g \circ \pi_V = 0$: $W \to \mathbb{C}$, proving (c).

For (d), from (3.28) we can show that $(\mathbb{C} \times (\mathrm{id} \times \Phi)(X) \times \mathbb{C}^{n^2}) \cap W$ is open and closed in $Z = \mathrm{Crit}(h)$, and contains $\iota(\mathbb{C})$. So making T, W smaller we can take

$$Z = \left(\mathbb{C} \times (\mathrm{id} \times \Phi)(X) \times \mathbb{C}^{n^2}\right) \cap W,$$

and then (d) follows as $\Phi|_X = \Psi|_X$.

For (e), observe that when t = 0 in (3.28), the second equation reduces to $r_{ij} = 0$ near $\iota(\mathbb{C})$ as $t(1-t)K_{ij}(t, u, v) = 0$, so making T, W smaller gives

$$W_0 = \left\{ \left(0, u, v, (0)_{i,j=1}^n\right) \in T : \left(x'_i - z'_i\right)(0, u, v) = 0, \ i = 1, \dots, n \right\} \\ = \left\{ \left(0, u, v, (0)_{i,j=1}^n\right) \in T : v = \Phi(u) \right\}.$$

Hence $\Phi \circ \pi_W|_{W_0} = \pi_U|_{W_0}$. Similarly, when t = 1 we have

$$W_1 = \left\{ \left(1, u, v, (0)_{i,j=1}^n\right) \in T : \left(y'_i - z'_i\right)(0, u, v) = 0, \ i = 1, \dots, n \right\}$$
$$= \left\{ \left(0, u, v, (0)_{i,j=1}^n\right) \in T : v = \Psi(u) \right\},$$

so that $\Psi \circ \pi_W|_{W_1} = \pi_U|_{W_1}$. This proves (e), and Proposition 3.4.

3.2. Part (a): det $(d\Psi|_{X^{red}}^{-1} \circ d\Phi|_{X^{red}}) = \pm 1$. We work in the situation of Theorem 3.1. For each $x \in X \subseteq U$, consider the diagram of linear maps of vector spaces:

where $T_x X$ is the Zariski tangent space of X, and $\operatorname{Hess}_x f = (\partial^2 f)|_x$ the Hessian of f at x. The rows of (3.31) are exact, and the columns isomorphisms. The outer squares of (3.31) clearly commute. We can show the central square commutes by taking second derivatives of $f = g \circ \Phi$ to get $\partial^2 f|_x = \partial^2 g|_{\Phi(x)} \circ (\mathrm{d}\Phi|_x \otimes \mathrm{d}\Phi|_x)$, and composing with $\mathrm{id} \otimes \mathrm{d}\Phi|_x^{-1}$. Thus (3.31) is commutative. There is also an analogue of (3.31) for Ψ . Since $\Psi(x) = \Phi(x)$, we may compose the columns of (3.31) for Φ with the inverses of the columns of (3.31) for Ψ to get a commutative diagram

where the outer morphisms are identities as $\Phi|_X = \Psi|_X$.

Choose a complementary vector subspace N_x to T_xX in T_xU , which we think of as the normal to X in U at x, so that $T_xU = T_xX \oplus N_x$. Write $\operatorname{Hess}'_x f$ for the restriction of $\operatorname{Hess}_x f$ to a symmetric bilinear form on N_x . Since $T_xX = \operatorname{Ker}(\operatorname{Hess}_x f)$, we see that $\operatorname{Hess}'_x f$ is a nondegenerate symmetric bilinear form on N_x . We may write equation (3.32) as

for some linear $A : N_x \to T_x X$ and $B : N_x \to N_x$. Then (3.33) commuting implies that *B* preserves the nondegenerate symmetric bilinear form $\operatorname{Hess}'_x f$ on N_x , and $\det B = \pm 1$. So $\det(\operatorname{d}\Psi|_x^{-1} \circ \operatorname{d}\Phi|_x) = \det(\operatorname{d}_0^{\operatorname{id}} B) = \det B = \pm 1$ for $x \in X$.

Thus, as a map of topological spaces, det $(d\Psi|_{X^{\text{red}}}^{-1} \circ d\Phi|_{X^{\text{red}}}) : X^{\text{red}} \to \mathbb{C} \setminus \{0\}$ actually maps $X^{\text{red}} \to \{\pm 1\}$. Since it is continuous, it is locally constant. Now if $f, g: Y \to Z$ are morphisms of \mathbb{C} -schemes with Y reduced, then f = g if and only if f(y) = g(y) for each point $y \in Y$. Applying this to compare det $(d\Psi|_{X^{\text{red}}}^{-1} \circ d\Phi|_{X^{\text{red}}}) : X^{\text{red}} \to \mathbb{C} \setminus \{0\}$ locally with the constant maps 1 or -1 on X^{red} shows that det $(d\Psi|_{X^{\text{red}}}^{-1} \circ d\Phi|_{X^{\text{red}}})$ is a locally constant map $X^{\text{red}} \to \{\pm 1\} \subset \mathbb{C} \setminus \{0\}$ as a \mathbb{C} -scheme morphism. This proves Theorem 3.1(a).

3.3. Part (b): $\mathcal{PV}_{\Phi} = \det(d\Psi|_{X^{\text{red}}}^{-1} \circ d\Phi|_{X^{\text{red}}}) \cdot \mathcal{PV}_{\Psi}$. For Theorem 3.1(b), we begin with the following proposition.

Proposition 3.5. Let $U, V, \Phi, \Psi, f, g, X, Y$ be as in Theorem 3.1, and suppose $x \in X$ with $(d\Psi|_x^{-1} \circ d\Phi|_x - id_{T_xU})^2 = 0$. Then there exists a Zariski open neighbourhood X' of x in X such that $\mathcal{PV}_{\Phi}|_{X'} = \mathcal{PV}_{\Psi}|_{X'}$.

Proof. Apply Proposition 3.4 to get $W, \pi_{\mathbb{C}}, \pi_U, \pi_V, \iota, h, Z$. Then apply Proposition 2.8 with $Z, X, x, \pi_{\mathbb{C}}|_Z, \pi_U|_Z, \iota, \mathcal{PV}_{U,f}^{\bullet}, \Phi|_X^* (\mathcal{PV}_{V,g}^{\bullet}), \mathcal{PV}_{\Phi}, \mathcal{PV}_{\Psi}$ in place of $W, X, x, \pi_{\mathbb{C}}, \pi_X, \iota, \mathcal{P}^{\bullet}, \mathcal{Q}^{\bullet}, \alpha$,

 β , respectively, and with γ defined by the commuting diagram of isomorphisms:

$$\begin{array}{cccc} \mathcal{P}\mathcal{V}_{W,h}^{\bullet} & & (\pi_{\mathbb{C}} \times \pi_{V})|_{Z}^{*} \left(\mathcal{P}\mathcal{V}_{\mathbb{C} \times V,0\boxplus g}^{\bullet}\right) \\ & & \downarrow^{\mathcal{P}\mathcal{V}_{\pi_{\mathbb{C}} \times \pi_{V}}} & (\pi_{\mathbb{C}} \times \pi_{V})|_{Z}^{*} (\mathcal{T}S_{\mathbb{C},0,V,g}) \downarrow \\ (\pi_{\mathbb{C}} \times \pi_{U})|_{Z}^{*} \left(\mathcal{P}\mathcal{V}_{\mathbb{C} \times U,0\boxplus f}^{\bullet}\right) & (\pi_{\mathbb{C}} \times \pi_{V})|_{Z}^{*} \left(\mathcal{P}\mathcal{V}_{\mathbb{C},0}^{\bullet} \boxtimes \mathcal{P}\mathcal{V}_{V,g}^{\bullet}\right) \\ & & \downarrow^{(\pi_{\mathbb{C}} \times \pi_{V})|_{Z}^{*} (\mathcal{T}S_{\mathbb{C},0,U,f})} & \delta' \downarrow \\ (\pi_{\mathbb{C}} \times \pi_{U})|_{Z}^{*} \left(\mathcal{P}\mathcal{V}_{\mathbb{C},0}^{\bullet} \boxtimes \mathcal{P}\mathcal{V}_{U,f}^{\bullet}\right) & \left[\pi_{\mathbb{C}}|_{Z}^{*} \left(A_{\mathbb{C}}[1]\right)\right] \bigotimes^{L} \left[\pi_{V}|_{Z}^{*} \left(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}\right)\right] \\ & & \downarrow^{\delta} & \epsilon' \downarrow \\ \left[\pi_{\mathbb{C}}|_{Z}^{*} \left(A_{\mathbb{C}}[1]\right)\right] \bigotimes^{L} \left[\pi_{U}|_{Z}^{*} \left(\mathcal{P}\mathcal{V}_{U,f}^{\bullet}\right)\right] & \pi_{V}|_{Z}^{*}[1] \left(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}\right) \\ & & \downarrow^{\epsilon} & & & & \\ \pi_{U}|_{Z}^{*}[1] \left(\mathcal{P}\mathcal{V}_{U,f}^{\bullet}\right) & & \gamma & \pi_{U}|_{Z}^{*}[1] \left(\Phi|_{X}^{*} \left(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}\right)\right). \end{array} \right)$$

Here $\mathcal{TS}_{\mathbb{C},0,U,f}, \mathcal{TS}_{\mathbb{C},0,V,q}$ are as in (2.8), δ, δ' come from $\mathcal{PV}_{\mathbb{C},0} \cong A_{\mathbb{C}}[1]$, and ϵ, ϵ' come from $\pi_{\mathbb{C}}|_{Z}^{*}(A_{\mathbb{C}}) \cong A_{Z}$ and $A_{Z} \otimes \mathcal{P}^{\bullet} \cong \mathcal{P}^{\bullet}$ for $\mathcal{P}^{\bullet} \in \operatorname{Perv}(Z)$.

Then the hypothesis $\pi_X|_{W_0}^*(\alpha) = j_0^*[-1](\gamma)$ in Proposition 2.8 follows from comparing $j_0^*[-1]$ applied to (3.34) with the commuting diagram

where $j_0 : Z \cap W_0 \hookrightarrow Z$ is the inclusion, and the bottom square commutes by Proposition 3.4(e) and (2.18). Similarly $\pi_X|_{W_1}^*(\beta) = j_1^*[-1](\gamma)$. Hence Proposition 2.8 gives Zariski open $x \in X' \subseteq X$ with $\mathcal{PV}_{\Phi}|_{X'} = \mathcal{PV}_{\Psi}|_{X'}$.

Now to prove Theorem 3.1(b), let $x \in X$ be arbitrary. As in §3.2, we can choose a splitting $T_x U = T_x X \oplus N_x$ such that

$$d\Psi|_x^{-1} \circ d\Phi|_x = \begin{pmatrix} \mathrm{id} & A\\ 0 & B \end{pmatrix} : \frac{T_x X \oplus}{N_x} \longrightarrow \frac{T_x X \oplus}{N_x}$$
(3.35)

for linear $A: N_x \to T_x X$ and $B: N_x \to N_x$, where B preserves the nondegenerate symmetric bilinear form $\operatorname{Hess}'_x f$ on N_x .

Choose a Zariski open neighbourhood U' of x in U and a splitting $TU' = E \oplus F$ for algebraic vector subbundles $E, F \subseteq TU$ with $E|_x = T_x X$ and $F|_x = N_x$. Then $df|_{U'} = \alpha \oplus \beta$ for unique $\alpha \in H^0(E)$ and $\beta \in H^0(F)$, and $X \cap U'$ is defined by $\alpha = \beta = 0$.

Since $\operatorname{Hess}_x f = \partial(\mathrm{d}f)|_x$ is nondegenerate on N_x , we see that $\nabla\beta|_x : T_xU \to F|_x$ induces an isomorphism $N_x \to F|_x$, so $\nabla\beta|_x$ is surjective. Therefore $S := \beta^{-1}(0)$ is a smooth \mathbb{C} -subscheme of U' near x, and making U' smaller, we can suppose S is smooth.

Set $e = f|_S : S \to \mathbb{C}$. Then the isomorphism $T^*S \cong E|_S$ identifies $de \in H^0(T^*S)$ with $\alpha|_S \in H^0(E|_S)$. Hence $\operatorname{Crit}(e: S \to \mathbb{C}) = \operatorname{Crit}(f|_{U'}: U' \to \mathbb{C}) = X \cap U'$, as \mathbb{C} -subschemes of U.

By [23, Prop. 2.23] quoted in Theorem 5.1(i) below, there exist a smooth \mathbb{C} -scheme R, morphisms $\gamma: R \to U', \ \delta: R \to S, \ \epsilon: R \to \mathbb{C}^n$ where $n = \dim U' - \dim S$, and $r \in R$, such that

115

 $\gamma(r) = x, \ \gamma|_Q = \delta|_Q, \ f \circ \gamma = e \circ \delta + (z_1^2 + \dots + z_n^2) \circ \epsilon : R \to \mathbb{C}$, and the following commutes with horizontal morphisms étale:

$$S \xleftarrow{\gamma|_Q} Q := \gamma^{-1}(S) \xrightarrow{\gamma|_Q} S$$

$$\downarrow^{\subset} \qquad \downarrow^{\subset} \qquad \downarrow^{\subset} \qquad \overset{\gamma|_Q}{\overset{\text{id}_S \times 0}{}} \qquad (3.36)$$

$$U' \xleftarrow{\gamma} R \xrightarrow{\delta \times \epsilon} S \times \mathbb{C}^n.$$

Taking derivatives at $r \in Q \subseteq R$ in (3.36) gives a commutative diagram

$$\begin{array}{ccc} T_xX = T_xS \xleftarrow{\cong} & T_rQ \xrightarrow{\cong} & T_xX = T_xS \\ \bigvee \subset & & \downarrow \subset & & \downarrow C \\ T_xX \oplus N_x = T_xU \xleftarrow{d\gamma|_r} & T_rR \xrightarrow{d(\gamma|_Q)|_r} & T_xX \oplus T_0\mathbb{C}^n. \end{array}$$

Therefore $d(\delta \times \epsilon)|_r \circ d\gamma|_r^{-1} : T_x X \oplus N_x \to T_x X \oplus T_0 \mathbb{C}^n$ is the identity on $T_x X$, and induces an isomorphism $N_x \to T_0 \mathbb{C}^n$, which as $f \circ \gamma = e \circ \delta + (z_1^2 + \dots + z_n^2) \circ \epsilon$ identifies $\operatorname{Hess}'_x f$ on N_x with $\operatorname{Hess}_0(z_1^2 + \dots + z_n^2) = dz_1 \otimes dz_1 + \dots + dz_n \otimes dz_n$ on $T_0 \mathbb{C}^n$. Thus, the linear isomorphism $B : N_x \to N_x$ above preserving $\operatorname{Hess}'_x f$ is identified with a linear isomorphism $M : \mathbb{C}^n \to \mathbb{C}^n$ preserving $dz_1 \otimes dz_1 + \dots + dz_n \otimes dz_n$, that is, $M \in O(n, \mathbb{C})$ satisfies

$$\begin{pmatrix} \mathrm{id} & 0\\ 0 & B \end{pmatrix} \circ \mathrm{d}\gamma|_r \circ \mathrm{d}(\delta \times \epsilon)|_r^{-1} = \mathrm{d}\gamma|_r \circ \mathrm{d}(\delta \times \epsilon)|_r^{-1} \circ \begin{pmatrix} \mathrm{id} & 0\\ 0 & M \end{pmatrix}.$$
(3.37)

Define P to be the \mathbb{C} -scheme fibre product $P = R \times_{\delta \times (M \circ \epsilon), S \times \mathbb{C}^n, \delta \times \epsilon} R$, with projections $\pi_1, \pi_2 : P \to R$. Then P is smooth and π_1, π_2 are étale, as $R, S \times \mathbb{C}^n$ are smooth and

$$\delta \times (M \circ \epsilon), \delta \times \epsilon : R \to S \times \mathbb{C}^n$$

are étale. As $r \in R$ with $(\delta \times (M \circ \epsilon))(r) = (x, 0) = (\delta \times \epsilon)(r)$, we have a point $p \in P$ with $\pi_1(p) = \pi_2(p) = r$. Define $d = f \circ \gamma \circ \pi_1 : P \to \mathbb{C}$ and $Z = \operatorname{Crit}(d)$. Then

$$d = f \circ \gamma \circ \pi_1 = (e \boxplus z_1^2 + \dots + z_n^2) \circ (\delta \times \epsilon) \circ \pi_1$$

= $(e \boxplus z_1^2 + \dots + z_n^2) \circ (\delta \times (M \circ \epsilon)) \circ \pi_1$
= $(e \boxplus z_1^2 + \dots + z_n^2) \circ (\delta \times \epsilon) \circ \pi_2 = f \circ \gamma \circ \pi_2.$ (3.38)

Consider the étale morphisms $\Phi \circ \gamma \circ \pi_1, \Psi \circ \gamma \circ \pi_2 : P \to V$. Both map $p \mapsto \Phi(x)$, and satisfy $g \circ (\Phi \circ \gamma \circ \pi_1) = d = g \circ (\Psi \circ \gamma \circ \pi_2)$ by (3.38) and $g \circ \Phi = f = g \circ \Psi$. Taking derivatives at p to get linear maps $T_p P \to T_{\Phi(x)} V$, we find that

$$\begin{aligned} \mathrm{d}(\Psi \circ \gamma \circ \pi_{2})|_{p} &= \mathrm{d}\Psi|_{x} \circ \mathrm{d}\gamma|_{r} \circ \mathrm{d}(\delta \times \epsilon)|_{r}^{-1} \circ \mathrm{d}((\delta \times \epsilon) \circ \pi_{2})|_{p} \\ &= \mathrm{d}\Psi|_{x} \circ \mathrm{d}\gamma|_{r} \circ \mathrm{d}(\delta \times \epsilon)|_{r}^{-1} \circ \mathrm{d}((\delta \times (M \circ \epsilon)) \circ \pi_{1})|_{p} \\ &= \mathrm{d}\Psi|_{x} \circ \mathrm{d}\gamma|_{r} \circ \mathrm{d}(\delta \times \epsilon)|_{r}^{-1} \circ \binom{\mathrm{id} \quad 0}{0 \quad M} \circ \mathrm{d}(\delta \times \epsilon)|_{r} \circ \mathrm{d}\pi_{1}|_{p} \\ &= \mathrm{d}\Psi|_{x} \circ \binom{\mathrm{id} \quad 0}{0 \quad B} \circ \mathrm{d}\gamma|_{r} \circ \mathrm{d}(\delta \times \epsilon)|_{r}^{-1} \circ \mathrm{d}(\delta \times \epsilon)|_{r} \circ \mathrm{d}\pi_{1}|_{p} \\ &= \mathrm{d}\Psi|_{x} \circ \binom{\mathrm{id} \quad -AB^{-1}}{0 \quad \mathrm{id}} \binom{\mathrm{id} \quad A}{0 \quad B} \circ \mathrm{d}\gamma|_{r} \circ \mathrm{d}\pi_{1}|_{p} \end{aligned}$$
(3.39)

using $(\delta \times (M \circ \epsilon)) \circ \pi_1 = (\delta \times \epsilon) \circ \pi_2$ in the second step, (3.37) in the fourth, and (3.35) in the sixth. Since $\left[\begin{pmatrix} \mathrm{id} & -AB^{-1} \\ 0 & \mathrm{id} \end{pmatrix}^2 = 0$, equation (3.39) implies that

$$\left(\mathrm{d}(\Psi\circ\gamma\circ\pi_2)|_p^{-1}\circ\mathrm{d}(\Phi\circ\gamma\circ\pi_1)|_p-\mathrm{id}_{T_pP}\right)^2=0,$$

and thus Proposition 3.5 gives a Zariski open neighbourhood P' of p in P such that

$$\mathcal{PV}_{\Phi\circ\gamma\circ\pi_1}|_{P'} = \mathcal{PV}_{\Psi\circ\gamma\circ\pi_2}|_{P'} : \mathcal{PV}_{P,d}^{\bullet}|_{P'} \longrightarrow (\Phi\circ\gamma\circ\pi_1)|_Z^* (\mathcal{PV}_{V,g}^{\bullet})|_{P'}.$$
 (3.40)

Since $(\delta \times (M \circ \epsilon)) \circ \pi_1 = (\delta \times \epsilon) \circ \pi_2 : P \to S \times \mathbb{C}^n$ are étale with

$$(e \boxplus z_1^2 + \dots + z_n^2) \circ (\delta \times (M \circ \epsilon)) \circ \pi_1 = d = (e \boxplus z_1^2 + \dots + z_n^2) \circ (\delta \times \epsilon) \circ \pi_2,$$

we see using (2.8) and (2.18) that

$$\pi_{1}|_{Z}^{*} \left[\mathcal{P}\mathcal{V}_{\delta} \stackrel{L}{\boxtimes} \left(M|_{\{0\}}^{*} (\mathcal{P}\mathcal{V}_{\epsilon}) \circ \mathcal{P}\mathcal{V}_{M} \right) \right] \circ \mathcal{P}\mathcal{V}_{\pi_{1}} = \pi_{1}|_{Z}^{*} \left[\mathcal{P}\mathcal{V}_{\delta} \stackrel{L}{\boxtimes} \mathcal{P}\mathcal{V}_{M\circ\epsilon} \right] \circ \mathcal{P}\mathcal{V}_{\pi_{1}}$$

$$\cong \pi_{1}|_{Z}^{*} (\mathcal{P}\mathcal{V}_{\delta\times(M\circ\epsilon)}) \circ \mathcal{P}\mathcal{V}_{\pi_{1}} = \mathcal{P}\mathcal{V}_{(\delta\times(M\circ\epsilon))\circ\pi_{1}} = \mathcal{P}\mathcal{V}_{(\delta\times\epsilon)\circ\pi_{2}}$$

$$= \pi_{2}|_{Z}^{*} (\mathcal{P}\mathcal{V}_{\delta\times\epsilon}) \circ \mathcal{P}\mathcal{V}_{\pi_{2}} \cong \pi_{2}|_{Z}^{*} \left[\mathcal{P}\mathcal{V}_{\delta} \stackrel{L}{\boxtimes} \mathcal{P}\mathcal{V}_{\epsilon} \right] \circ \mathcal{P}\mathcal{V}_{\pi_{2}},$$

$$(3.41)$$

where ' \cong ' are equalities after identifying both sides of (2.8). Since $\pi_1|_Z = \pi_2|_Z$, and $M|_{\{0\}} = \mathrm{id}_{\{0\}}$, and Example 2.16 shows that \mathcal{PV}_M in (2.19) is multiplication by det M, equation (3.41) implies that

$$\det M \cdot \pi_1 |_Z^* \left[\mathcal{P} \mathcal{V}_\delta \stackrel{L}{\boxtimes} \mathcal{P} \mathcal{V}_\epsilon \right] \circ \mathcal{P} \mathcal{V}_{\pi_1} = \pi_1 |_Z^* \left[\mathcal{P} \mathcal{V}_\delta \stackrel{L}{\boxtimes} \mathcal{P} \mathcal{V}_\epsilon \right] \circ \mathcal{P} \mathcal{V}_{\pi_2}.$$

As $\pi_1|_Z^* \left[\mathcal{PV}_\delta \boxtimes^L \mathcal{PV}_\epsilon \right]$ is an isomorphism, this gives

$$\det M \cdot \mathcal{PV}_{\pi_1} = \mathcal{PV}_{\pi_2} : \mathcal{PV}_{P,d}^{\bullet} \longrightarrow \pi_1 |_Z^* (\mathcal{PV}_{R,e}^{\bullet}).$$
(3.42)

Writing $Z' = Z \cap P'$, we now have

$$\begin{aligned} (\gamma \circ \pi_1)|_{Z'}^* (\mathcal{P}\mathcal{V}_{\Phi}) \circ \mathcal{P}\mathcal{V}_{\gamma \circ \pi_1}|_{P'} &= \mathcal{P}\mathcal{V}_{\Phi \circ \gamma \circ \pi_1}|_{P'} = \mathcal{P}\mathcal{V}_{\Psi \circ \gamma \circ \pi_2}|_{P'} \\ &= \pi_2|_{Z'}^* (\mathcal{P}\mathcal{V}_{\Psi \circ \gamma}) \circ \mathcal{P}\mathcal{V}_{\pi_2}|_{P'} \\ &= \det M \cdot \pi_1|_{Z'}^* (\mathcal{P}\mathcal{V}_{\Psi \circ \gamma}) \circ \mathcal{P}\mathcal{V}_{\pi_1}|_{P'} = \det M \cdot \mathcal{P}\mathcal{V}_{\Psi \circ \gamma \circ \pi_1}|_{P'} \\ &= \det M \cdot (\gamma \circ \pi_1)|_{Z'}^* (\mathcal{P}\mathcal{V}_{\Psi}) \circ \mathcal{P}\mathcal{V}_{\gamma \circ \pi_1}|_{P'}, \end{aligned}$$
(3.43)

using (3.40) in the second step, (3.42) and $\pi_1|_{Z'} = \pi_2|_{Z'}$ in the fourth, and (2.18) in the rest. As $\mathcal{PV}_{\gamma \circ \pi_1}|_{P'}$ is an isomorphism, (3.43) implies that

$$(\gamma \circ \pi_1)|_{Z'}^*(\mathcal{PV}_{\Phi}) = \det M \cdot (\gamma \circ \pi_1)|_{Z'}^*(\mathcal{PV}_{\Psi}),$$

and by Theorem 2.7(i) this implies that

$$\mathcal{PV}_{\Phi}|_{X'} = \det M \cdot \mathcal{PV}_{\Phi}|_{X'}, \qquad (3.44)$$

where $X' = (\gamma \circ \pi_1)(Z')$ is a Zariski open neighbourhood of x in X, since $(\gamma \circ \pi_1)|_{Z'} : Z' \to X$ is étale with $\gamma \circ \pi_1(p) = x$. Now (3.35) and (3.37) give

$$\det\left(\mathrm{d}\Psi|_x^{-1}\circ\mathrm{d}\Phi|_x\right) = \det\left(\begin{smallmatrix}\mathrm{id}&A\\0&B\end{smallmatrix}\right) = \det B = \det M.$$

So (3.44) proves that (3.1) holds near x in X. As this is true for all $x \in X$, Theorem 3.1(b) follows.

3.4. \mathscr{D} -modules and mixed Hodge modules. The proof of Proposition 3.4 applies verbatim also in the analytic context. Theorem 3.1(a),(b) then follow from Proposition 3.4 and the argument given above, using §2.5, including the Sheaf Property (x) for morphisms. Hence all these results carry over to our other contexts §2.6–§2.10.

4. Dependence of $\mathcal{PV}_{Uf}^{\bullet}$ on f

We will use the following notation:

Definition 4.1. Let U be a smooth \mathbb{C} -scheme, let $f: U \to \mathbb{C}$ be a regular function, and let X equal $\operatorname{Crit}(f)$ as a closed \mathbb{C} -subscheme of U. Write $I_X \subseteq \mathcal{O}_U$ for the sheaf of ideals of regular functions $U \to \mathbb{C}$ vanishing on X, so that $I_X = I_{\mathrm{d}f}$. For each $k = 1, 2, \ldots$, write $X^{(k)}$ for the k^{th} order thickening of X in U, that is, $X^{(k)}$ is the closed \mathbb{C} -subscheme of U defined by the sheaf of ideals I_X^k in \mathcal{O}_U . Also write X^{red} for the reduced \mathbb{C} -subscheme of U.

Then we have a chain of inclusions of closed $\mathbb{C}\text{-subschemes}$

$$X^{\text{red}} \subseteq X = X^{(1)} \subseteq X^{(2)} \subseteq X^{(3)} \subseteq \dots \subseteq U.$$
(4.1)

Write $f^{(k)} := f|_{X^{(k)}} : X^{(k)} \to \mathbb{C}$, and $f^{\text{red}} := f|_{X^{\text{red}}} : X^{\text{red}} \to \mathbb{C}$, so that $f^{(k)}, f^{\text{red}}$ are regular functions on the \mathbb{C} -schemes $X^{(k)}, X^{\text{red}}$. Note that $f^{\text{red}} : X^{\text{red}} \to \mathbb{C}$ is locally constant, since X = Crit(f).

We also use the same notation for complex analytic spaces.

In §2.4 we defined the perverse sheaf of vanishing cycles $\mathcal{PV}_{U,f}^{\bullet}$ in $\operatorname{Perv}(X)$. So we can ask: how much of the sequence (4.1) does $\mathcal{PV}_{U,f}^{\bullet}$ depend on? That is, is $\mathcal{PV}_{U,f}^{\bullet}$ (canonically?) determined by $(X^{\operatorname{red}}, f^{\operatorname{red}})$, or by $(X^{(k)}, f^{(k)})$ for some $k \geq 1$, as well as by (U, f)? Our next theorem shows that $\mathcal{PV}_{U,f}^{\bullet}$ is determined up to canonical isomorphism by $(X^{(3)}, f^{(3)})$, and hence a fortiori also by $(X^{(k)}, f^{(k)})$ for k > 3:

Theorem 4.2. Let U, V be smooth \mathbb{C} -schemes, $f : U \to \mathbb{C}$, $g : V \to \mathbb{C}$ be regular functions, and $X = \operatorname{Crit}(f)$, $Y = \operatorname{Crit}(g)$ as closed \mathbb{C} -subschemes of U, V, so that §2.4 defines perverse sheaves $\mathcal{PV}_{U,f}^{\bullet}, \mathcal{PV}_{V,g}^{\bullet}$ on X, Y. Define $X^{(3)}, f^{(3)}$ and $Y^{(3)}, g^{(3)}$ as in Definition 4.1, and suppose $\Phi : X^{(3)} \to Y^{(3)}$ is an isomorphism with $g^{(3)} \circ \Phi = f^{(3)}$, so that $\Phi|_X : X \to Y \subseteq Y^{(3)}$ is an isomorphism.

Then there is a canonical isomorphism in Perv(X)

$$\Omega_{\Phi}: \mathcal{PV}^{\bullet}_{U,f} \longrightarrow \Phi|_X^*(\mathcal{PV}^{\bullet}_{V,g}), \tag{4.2}$$

which is characterized by the property that if T is a smooth \mathbb{C} -scheme and $\pi_U : T \to U$, $\pi_V : T \to V$ are étale morphisms with $e := f \circ \pi_U = g \circ \pi_V : T \to \mathbb{C}$, so that $\pi_U|_Q : Q \to X$, $\pi_V|_Q : Q \to Y$ are étale for $Q := \operatorname{Crit}(e)$, and $\Phi \circ \pi_U|_{Q^{(2)}} = \pi_V|_{Q^{(2)}} : Q^{(2)} \to Y^{(2)}$, then

$$\pi_U|_Q^*(\Omega_\Phi) \circ \mathcal{PV}_{\pi_U} = \mathcal{PV}_{\pi_V} : \mathcal{PV}_{T,e}^{\bullet} \longrightarrow \pi_V|_Q^*(\mathcal{PV}_{U,f}^{\bullet}).$$
(4.3)

Also the following commute, where $\sigma_{U,f}, \sigma_{V,g}, \tau_{U,f}, \tau_{V,g}$ are as in (2.6)–(2.7):

If there exists an étale morphism $\Xi: U \to V$ with

$$g \circ \Xi = f : U \to \mathbb{C}$$
 and $\Xi|_{X^{(3)}} = \Phi : X^{(3)} \to Y^{(3)}$,

then $\Omega_{\Phi} = \mathcal{PV}_{\Xi}$, for \mathcal{PV}_{Ξ} as in (2.14).

If W is another smooth \mathbb{C} -scheme, $h: W \to \mathbb{C}$ is a regular function, $Z = \operatorname{Crit}(h)$, and $\Psi: Y^{(3)} \to Z^{(3)}$ is an isomorphism with $h^{(3)} \circ \Psi = q^{(3)}$, then

$$\Omega_{\Psi\circ\Phi} = \Phi|_X^*(\Omega_{\Psi}) \circ \Omega_{\Phi} : \mathcal{PV}_{U,f}^{\bullet} \longrightarrow (\Psi \circ \Phi)|_X^*(\mathcal{PV}_{W,h}^{\bullet}).$$
(4.6)

If U = V, f = g, X = Y and $\Phi = \mathrm{id}_{X^{(3)}}$ then $\Omega_{\mathrm{id}_{X^{(3)}}} = \mathrm{id}_{\mathcal{PV}_{U,f}^{\bullet}}$.

The analogues of all the above also hold with appropriate modifications for \mathcal{D} -modules on \mathbb{C} schemes, for perverse sheaves and \mathcal{D} -modules on complex analytic spaces, and for mixed Hodge modules on \mathbb{C} -schemes and complex analytic spaces, as in §2.6–§2.10.

We will prove Theorem 4.2 in §4.2–§4.3. The proof for \mathbb{C} -schemes depends on the case k=2of the following proposition, proved in $\S4.1$:

Proposition 4.3. Let U, V be smooth \mathbb{C} -schemes, $f: U \to \mathbb{C}$, $g: V \to \mathbb{C}$ be regular functions, and $X = \operatorname{Crit}(f) \subseteq U, Y = \operatorname{Crit}(g) \subseteq V$. Using the notation of Definition 4.1, suppose $\Phi: X^{(k+1)} \to Y^{(k+1)}$ is an isomorphism with $g^{(k+1)} \circ \Phi = f^{(k+1)}$ for some $k \ge 2$. Then for each $x \in X$ we can choose a smooth \mathbb{C} -scheme T and étale morphisms $\pi_U : T \to U, \pi_V : T \to V$ such that

- (a) $e := f \circ \pi_U = g \circ \pi_V : T \to \mathbb{C};$
- (b) setting $Q = \operatorname{Crit}(e)$, then $\pi_U|_{Q^{(k)}} : Q^{(k)} \to X^{(k)} \subseteq U$ is an isomorphism with a Zariski open neighbourhood $\tilde{X}^{(k)}$ of x in $X^{(k)}$; and (c) $\Phi \circ \pi_U|_{Q^{(k)}} = \pi_V|_{Q^{(k)}} : Q^{(k)} \to Y^{(k)}$.

The proof of Proposition 4.3 is similar to that of Proposition 3.5 in §3.1. One can also prove an analogue of Proposition 4.3 when k = 1, but in part (b) $\pi_U|_{O^{(1)}} : Q^{(1)} \to X^{(1)}$ must be étale rather than a Zariski open inclusion.

In Proposition 4.3, we start with $\Phi: X^{(k+1)} \xrightarrow{\cong} Y^{(k+1)}$, but we construct T, π_U, π_V with $\Phi \circ \pi_U|_{O^{(k)}} = \pi_V|_{O^{(k)}}$. One might expect to find T, π_U, π_V with $\Phi \circ \pi_U|_{O^{(k+1)}} = \pi_V|_{O^{(k+1)}}$, but the next example shows this is not possible.

Example 4.4. Let U, V be open neighbourhoods of 0 in \mathbb{C} , and $f: U \to \mathbb{C}, g: V \to \mathbb{C}$ be regular functions given as power series by $f(x) = x^{m+1}$ and $g(y) = y^{m+1} + Ay^{(k+1)m} + \cdots$, for $k, m \ge 2$ and $0 \ne A \in \mathbb{C}$, where 0 is the only critical point of g.

Then $X := \operatorname{Crit}(f) = \operatorname{Spec}(\mathbb{C}[x]/(x^m))$ and $Y := \operatorname{Crit}(g) = \operatorname{Spec}(\mathbb{C}[y]/(y^m))$, so $X^{(k+1)} = \operatorname{Spec}(\mathbb{C}[x]/(x^{(k+1)m})), \ f^{(k+1)} = x^{m+1} + (x^{(k+1)m}), \ Y^{(k+1)} = \operatorname{Spec}(\mathbb{C}[y]/(y^{(k+1)m})),$ and $g^{(k+1)} = y^{m+1} + (y^{(k+1)m})$. Thus $\Phi : X^{(k+1)} \to Y^{(k+1)}$ acting on functions by $y + (y^{(k+1)m}) \mapsto x + (x^{(k+1)m})$ is an

isomorphism with $f^{(k+1)} = q^{(k+1)} \circ \Phi$.

Suppose T, π_U, π_V are as in Proposition 4.3, and use $w = x \circ \pi_U$ as a coordinate on T. Then $e(w) = w^{m+1}$, and $Q = \operatorname{Crit}(e) = \operatorname{Spec}(\mathbb{C}[w]/(w^m))$. We have $\pi_U(w) = w$, so $\Phi \circ \pi_U(w) = w$, but $w^{m+1} = \pi_V(w)^{m+1} + A\pi_V(w)^{(k+1)m} + \cdots$, so that $\pi_V(w) = w - \frac{1}{m+1}Aw^{km} + \cdots$. Thus, $\Phi \circ \pi_U : T \to V$ and $\pi_V : T \to V$ differ by $-\frac{1}{m+1}Aw^{km} + \cdots$, which is zero on $Q^{(k)}$ but not on $Q^{(k+1)}$. Hence in this example there do not exist T, π_U, π_V with $\Phi \circ \pi_U|_{Q^{(k+1)}} = \pi_V|_{Q^{(k+1)}}$.

Remark 4.5. We can also ask: can we improve $(X^{(3)}, f^{(3)})$ in Theorem 4.2 to $(X^{(2)}, f^{(2)})$ or $(X^{(1)}, f^{(1)})$ or $(X^{\text{red}}, f^{\text{red}})$? Here are some thoughts on this.

(a) The analogue of Proposition 4.3 for k = 1 mentioned above implies that étale or complex analytically locally on X, (U, f) and hence $\mathcal{PV}_{U, f}^{\bullet}$ are determined up to non-canonical isomorphism by $(X^{(2)}, f^{(2)})$. Using the ideas of $\S5-\S6$, one can show that these non-canonical isomorphisms of $\mathcal{PV}_{U,f}^{\bullet}$ are unique up to sign.

119

(b) Consider the following example: let $U = (\mathbb{C} \setminus \{0\}) \times \mathbb{C} = V$, and define $f : U \to \mathbb{C}$ and $g : V \to \mathbb{C}$ by $f(x, y) = y^2$ and $g(x, y) = xy^2$. Then $X := \operatorname{Crit}(f) = \{y = 0\} = \operatorname{Crit}(g) =: Y$, and $f^{(2)} = g^{(2)} = 0$, so that $(X^{(2)}, f^{(2)}) = (Y^{(2)}, g^{(2)})$. However, as in Example 5.5 below, $\mathcal{PV}_{U,f}^{\bullet} \ncong \mathcal{PV}_{V,g}^{\bullet}$. Thus, globally, $\mathcal{PV}_{U,f}^{\bullet}$ is not determined up to isomorphism by $(X^{(2)}, f^{(2)})$.

(c) Suppose U is a complex manifold and $f: U \to \mathbb{C}$ is holomorphic, with $\operatorname{Crit}(f)$ a single (not necessarily reduced) point x. The Mather-Yau Theorem [38] shows that the germ of (U, f) at x is determined up to non-canonical isomorphism by the complex analytic subspace $f^{(1)} = 0$ in $X^{(1)}$, and hence by the pair $(X^{(1)}, f^{(1)})$. Therefore, for *isolated* singularities, $\mathcal{PV}_{U,f}^{\bullet}$ is determined up to non-canonical isomorphism by $(X^{(1)}, f^{(1)})$.

(d) Define $f: U \to \mathbb{C}$ by $U = \mathbb{C}$ and $f(z) = cz^n$ for $0 \neq c \in \mathbb{C}$ and n > 2. This has an isolated singularity at 0, and $(X^{(1)}, f^{(1)})$ is independent of c. By moving c in a circle round zero, we see that in this example $\mathcal{PV}_{U,f}^{\bullet}$ is determined up to a $\mathbb{Z}/n\mathbb{Z}$ group of automorphisms. So the non-canonical isomorphisms of $\mathcal{PV}_{U,f}^{\bullet}$ are not unique up to sign, in contrast to (a).

(e) Parts (a)–(d) leave open the question of whether $\mathcal{PV}_{U,f}^{\bullet}$ is determined locally up to noncanonical isomorphism by $(X^{(1)}, f^{(1)})$ for non-isolated singularities. We do not have a counterexample to this.

However, Gaffney and Hauser [19, §4] give examples of complex manifolds U and holomorphic $f: U \to \mathbb{C}$ with $X = \operatorname{Crit}(f)$ non-isolated, such that the germ of (U, f) at $x \in X$ is not determined up to non-canonical isomorphism by the germ of $(X^{(1)}, f^{(1)})$ at x, in contrast to the Mather–Yau Theorem, and continuous families of distinct germs [U, f, x] can have the same germ $[X^{(1)}, f^{(1)}, x]$. It seems likely that in examples of this kind, the mixed Hodge module $\mathcal{HV}^{\bullet}_{U,f}$ (which contains continuous Hodge-theoretic information) is not locally determined up to non-canonical isomorphism by $(X^{(1)}, f^{(1)})$.

(f) For the example in (d), $\mathcal{PV}_{U,f}^{\bullet}$ depends on $n = 3, 4, \ldots$, but $(X^{\text{red}}, f^{\text{red}}) = (\{0\}, 0)$ is independent of n. So $\mathcal{PV}_{U,f}^{\bullet}$ is not determined even locally up to non-canonical isomorphism by $(X^{\text{red}}, f^{\text{red}})$.

4.1. **Proof of Proposition 4.3.** The \mathbb{C} -subscheme $X^{(k+1)}$ in U is the zeroes of the ideal $I_X^{k+1} \subset \mathcal{O}_U$, which vanishes to order $k+1 \ge 2$ at $x \in X \subseteq X^{(k+1)} \subseteq U$. Hence $T_x X^{(k+1)} = T_x U$. As $\Phi : X^{(k+1)} \to Y^{(k+1)}$ is an isomorphism, it follows that

$$T_x U = T_x X^{(k+1)} \cong T_{\Phi(x)} Y^{(k+1)} = T_{\Phi(x)} V.$$
(4.7)

Therefore $n := \dim U = \dim V$.

Choose a Zariski open neighbourhood V' of $\Phi(x)$ in V and étale coordinates

$$(y_1,\ldots,y_n):V'\to\mathbb{C}^n$$

on V'. Write $g' = g|_{V'}$ and $Y' = \operatorname{Crit}(g') = Y \cap V'$, so that $Y'^{(k+1)} = Y^{(k+1)} \cap V'$. Then $y_a \circ \Phi$ are regular functions on the open neighbourhood $\Phi^{-1}(V') \subseteq X^{(k+1)}$ of x in $X^{(k+1)}$, so they extend Zariski locally from $X^{(k+1)}$ to U. Thus we can choose a Zariski open neighbourhood U' of x in U with $\Phi(X^{(k+1)} \cap U') \subseteq Y^{(k+1)} \cap V'$, and regular functions $x_i : U' \to \mathbb{C}$ with $x_i|_{X^{(k+1)} \cap U'} = y_i \circ \Phi|_{X^{(k+1)} \cap U'}$ for $i = 1, \ldots, n$.

Write $f' = f|_{U'}$ and $X' = \operatorname{Crit}(f') = X \cap U'$, so that $X'^{(k+1)} = X^{(k+1)} \cap X'$. Since (y_1, \ldots, y_n) are étale coordinates,

$$\mathrm{d}y_1|_{\Phi(x)},\ldots,\mathrm{d}y_n|_{\Phi(x)}$$

are a basis for $T^*_{\Phi(x)}V$, so $dx_1|_x, \ldots, dx_n|_x$ are a basis for T^*_xX by (4.7). Hence by making U' smaller, we can suppose (x_1, \ldots, x_n) are étale coordinates on U'.

Consider the \mathbb{C} -scheme $U' \times V'$, with projections

 $\pi_{U'}: U' \times V' \longrightarrow U'$ and $\pi_{V'}: U' \times V' \longrightarrow V'$,

and write

$$x'_i = x_i \circ \pi_{U'} : U' \times V' \to \mathbb{C}, \quad y'_i = y_i \circ \pi_{V'} : U' \times V' \to \mathbb{C},$$

so that $(x'_1, \ldots, x'_n, y'_1, \ldots, y'_n)$ are étale coordinates on $U' \times V'$. We have a morphism

$$\mathrm{id} \times \Phi|_{X'^{(k+1)}} : X'^{(k+1)} \longrightarrow U' \times V'$$

which embeds $X'^{(k+1)}$ as a closed \mathbb{C} -subscheme of $U' \times V'$. The image $(\mathrm{id} \times \Phi|_{X'^{(k+1)}})(X'^{(k+1)})$ is locally the zeroes of the sheaf of ideals

$$(x'_i - y'_i, i = 1, \dots, n) + \pi_{U'}^{-1}(I_X^{k+1}) \subset \mathcal{O}_{U' \times V'},$$

where $(x'_i - y'_i, i = 1, ..., n)$ denotes the ideal generated by $x'_i - y'_i : U' \times V' \to \mathbb{C}$ for i = 1, ..., n, and $\pi_{U'}^{-1}(I_X^{k+1}) \subset \mathcal{O}_{U' \times V'}$ the preimage ideal of $I_X^{k+1}|_{U'} \subset \mathcal{O}_{U'}$.

Now $(f \circ \pi_{U'} - g \circ \pi_{V'})|_{(\mathrm{id} \times \Phi)(X'^{(k+1)})} = 0$ as $f^{(k+1)} = g^{(k+1)} \circ \Phi$. Hence

$$f \circ \pi_{U'} - g \circ \pi_{V'} \in \left(x'_i - y'_i, \ i = 1, \dots, n \right) + \pi_{U'}^{-1} \left(I_X^{k+1} \right).$$

$$(4.8)$$

Lifting (4.8) from $(x'_i - y'_i, i = 1, ..., n)$ to $(x'_i - y'_i, i = 1, ..., n)^2$, after making U', V' smaller if necessary, we can choose regular functions $A_i : U' \times V' \to \mathbb{C}$ for i = 1, ..., n such that

$$f \circ \pi_{U'} - g \circ \pi_{V'} - \sum_{i=1}^{n} A_i \cdot (x'_i - y'_i) \in (x'_i - y'_i, \ i = 1, \dots, n)^2 + \pi_{U'}^{-1} (I_X^{k+1}).$$
(4.9)

Apply $\frac{\partial}{\partial x'_i}$ to (4.9), using the étale coordinates $(x'_1, \ldots, x'_n, y'_1, \ldots, y'_n)$ on $U' \times V'$. Since

$$\frac{\partial}{\partial x_i'} \big(f \circ \pi_{U'} \big) = \frac{\partial f}{\partial x_i} \circ \pi_{U'}$$

and $\frac{\partial}{\partial x'_i} (g \circ \pi_{V'}) = 0$, this gives

$$\frac{\partial f}{\partial x_i} \circ \pi_{U'} - A_i \in \left(x'_i - y'_i, \ i = 1, \dots, n \right) + \pi_{U'}^{-1} \left(I_X^k \right).$$
(4.10)

Changing A_i by an element of $(x'_i - y'_i, i = 1, ..., n)$ can be absorbed in the ideal

$$(x'_i - y'_i, \ i = 1, \dots, n)^2$$

in (4.9), so we can suppose $\frac{\partial f}{\partial x_i} \circ \pi_{U'} - A_i \in \pi_{U'}^{-1}(I_X^k)$. As
$$I_X = \left(\frac{\partial f}{\partial x_i}, \ j = 1, \dots, n\right),$$

after making U', V' smaller we may write

$$A_{i} = \frac{\partial f}{\partial x_{i}} \circ \pi_{U'} + \sum_{j=1}^{n} B_{ij} \cdot \frac{\partial f}{\partial x_{j}} \circ \pi_{U'}, \qquad (4.11)$$

with $B_{ij} \in \pi_{U'}^{-1}(I_X^{k-1})$ for i, j = 1, ..., n. Consider the matrix of functions $(\delta_{ij} + B_{ij})_{i,j=1}^n$ on $U' \times V'$. At the point $(x, \Phi(x))$ in $U' \times V'$ this matrix is the identity, since $B_{ij}(x, \Phi(x)) = 0$ as $B_{ij} \in \pi_{U'}^{-1}(I_X^{k-1})$ with $k \ge 2$, so $(\delta_{ij} + B_{ij})_{i,j=1}^n$ is invertible near $(x, \Phi(x))$, and making U', V' smaller we can suppose $(\delta_{ij} + B_{ij})_{i,j=1}^n$ is invertible on $U' \times V'$. But in matrix notation we have

$$(A_i)_{i=1}^n = (\delta_{ij} + B_{ij})_{i,j=1}^n (\frac{\partial f}{\partial x_j} \circ \pi_{U'})_{j=1}^n.$$

Hence in ideals in $\mathcal{O}_{U' \times V'}$ we have

$$(A_i, i = 1, \dots, n) = \left(\frac{\partial f}{\partial x_j} \circ \pi_{U'}, j = 1, \dots, n\right) = \pi_{U'}^{-1}(I_X) \subset \mathcal{O}_{U' \times V'}.$$
(4.12)

Now by (4.9), after making U', V' smaller if necessary, we may write

$$f \circ \pi_{U'} - g \circ \pi_{V'} = \sum_{i=1}^{n} A_i \cdot (x'_i - y'_i) + \sum_{i,j=1}^{n} C_{ij} \cdot (x'_i - y'_i) (x'_j - y'_j) + \sum_{i,j=1}^{n} D_{ij} \cdot A_i A_j,$$
(4.13)

for regular functions $C_{ij}, D_{ij}: U' \times V' \to \mathbb{C}$ with $D_{ij} \in H^0(\pi_{U'}^{-1}(I_X^{k-1}))$ for $i, j = 1, \ldots, n$, where

in the last term we have used (4.12) to write two factors of $\pi_{U'}^{-1}(I_X)$ in terms of A_1, \ldots, A_n . Write $(z_{ij})_{i,j=1}^n$ for the coordinates on \mathbb{C}^{n^2} . Let W be a Zariski open neighbourhood of $(x, \Phi(x), (0)_{i,j=1}^n)$ in $U' \times V' \times \mathbb{C}^{n^2}$ to be chosen shortly, and let T be the \mathbb{C} -subscheme of W defined by

$$T = \left\{ \left(u, v, (z_{ij})_{i,j=1}^{n} \right) \in W \subseteq U' \times V' \times \mathbb{C}^{n^{2}} : \\ x_{i}(u) - y_{i}(v) = \sum_{j=1}^{n} z_{ij} \cdot A_{j}(u, v), \quad i = 1, \dots, n, \\ z_{ij} + \sum_{l,m=1}^{n} C_{lm}(u, v) \cdot z_{li} z_{mj} + D_{ij}(u, v) = 0, \quad i, j = 1, \dots, n \right\}.$$

$$(4.14)$$

Define \mathbb{C} -scheme morphisms $\pi_U: T \to U$ by $\pi_U: (u, v, (z_{ij})_{i,j=1}^n) \mapsto u$ and $\pi_V: T \to V$ by $\pi_V : (u, v, (z_{ij})_{i,j=1}^n) \mapsto v.$

Now $W \subseteq U' \times V' \times \mathbb{C}^{n^2}$ is smooth of dimension $n + n + n^2$, and in (4.14) we impose $n + n^2$ equations, so the expected dimension of T is $(2n + n^2) - (n + n^2) = n$. The linearizations of the $n + n^2$ equations in (4.14) at $(u, v, (z_{ij})_{i,j=1}^n) = (x, \Phi(x), (0)_{i,j=1}^n)$ are

$$dx_{i}|_{x}(\delta u) - dy_{i}|_{\Phi(x)}(\delta v) = 0, \quad i = 1, \dots, n,$$

$$\delta z_{ij} + dD_{ij}|_{(x,\Phi(x))}(\delta u \oplus \delta v) = 0, \quad i, j = 1, \dots, n,$$

(4.15)

for $\delta u \in T_x U'$, $\delta v \in T_{\Phi(x)} V'$, and $(\delta z_{ij})_{i,j=1}^n \in T_{(0)_{i,j=1}^n} \mathbb{C}^{n^2}$. As $dx_1|_x, \ldots, dx_n|_x$ are a basis for $T_x^* U'$, the equations (4.15) are transverse, so that T is smooth of dimension n near $(x, \Phi(x), (0)_{i,j=1}^n).$

The vector space of solutions $(\delta u, \delta v, (\delta z_{ij})_{i,j=1}^n)$ to (4.15) is $T_{(x,\Phi(x),(0))}T$, where

$$\mathrm{d}\pi_U|_{(x,\Phi(x),(0))}: T_{(x,\Phi(x),(0))}T \longrightarrow T_xU$$

maps $(\delta u, \delta v, (\delta z_{ij})_{i,j=1}^n) \mapsto \delta u$, and

$$d\pi_V|_{(x,\Phi(x),(0))}:T_{(x,\Phi(x),(0))}T\longrightarrow T_{\Phi(x)}V$$

maps $(\delta u, \delta v, (\delta z_{ij})_{i,j=1}^n) \mapsto \delta v$. Clearly, $d\pi_U|_{(x,\Phi(x),(0))}, d\pi_V|_{(x,\Phi(x),(0))}$ are isomorphisms, so as T is smooth near $(x, \Phi(x), (0)_{i,i=1}^n)$ and U, V are smooth, we see that π_U, π_V are étale near $(x, \Phi(x), (0))$. Thus, by choosing the open neighbourhood $(x, \Phi(x), (0)) \in W \subseteq U' \times V' \times \mathbb{C}^{n^2}$ sufficiently small, we can suppose that T is smooth of dimension n and π_U : $T \to U$ and $\pi_V: T \to V$ are étale.

It remains to prove Proposition 4.3(a)-(c). For (a), we have

$$\begin{split} \left(f \circ \pi_{U} - g \circ \pi_{V}\right) \left(u, v, (z_{ij})_{i,j=1}^{n}\right) &= f(u) - g(v) = (f \circ \pi_{U'} - g \circ \pi_{V'})(u, v) \\ &= \sum_{i=1}^{n} A_{i}(u, v) \cdot \left(x_{i}(u) - y_{i}(v)\right) + \sum_{i,j=1}^{n} C_{ij}(u, v) \cdot \left(x_{i}(u) - y_{i}(v)\right) \left(x_{j}(u) - y_{j}(v)\right) \\ &+ \sum_{i,j=1}^{n} D_{ij}(u, v) \cdot A_{i}(u, v)A_{j}(u, v) \\ &= \sum_{i=1}^{n} A_{i}(u, v) \cdot \left(\sum_{j=1}^{n} z_{ij} \cdot A_{j}(u, v)\right) + \sum_{i,j=1}^{n} C_{ij}(u, v) \cdot \left(\sum_{l=1}^{n} z_{il} \cdot A_{l}(u, v)\right) \left(\sum_{m=1}^{n} z_{jm} \cdot A_{m}(u, v)\right) \\ &+ \sum_{i,j=1}^{n} D_{ij}(u, v) \cdot A_{i}(u, v)A_{j}(u, v) \\ &= \sum_{i,j=1}^{n} A_{i}(u, v)A_{j}(u, v) \left[z_{ij} + \sum_{l,m=1}^{n} C_{lm}(u, v) \cdot z_{li}z_{mj} + D_{ij}(u, v)\right] = 0, \end{split}$$

using (4.13) in the third step, the first equation of (4.14) in the fourth, rearranging and exchanging labels i, l and j, m in the fifth, and the second equation of (4.14) in the sixth. Hence $f \circ \pi_U - g \circ \pi_V = 0: T \to \mathbb{C}$, proving (a).

For (b), using the morphism $\operatorname{id} \times \Phi|_X \times (0) : X \to U \times V \times \mathbb{C}^{n^2} \supseteq W$, define

$$\tilde{X} = \left(\operatorname{id} \times \Phi |_X \times (0) \right)^{-1} (W),$$

so that \tilde{X} is a Zariski open neighbourhood of x in X. Then $(\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$ is a closed \mathbb{C} -subscheme of W. We claim that:

- (i) $(\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$ is a closed \mathbb{C} -subscheme of $T \subseteq W$; and
- (ii) $(\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$ is open and closed in $Q := \operatorname{Crit}(e) \subseteq T$,

where $e := f \circ \pi_U = g \circ \pi_V : T \to \mathbb{C}$. To prove (i), we have to show that the equations of (4.14) hold on $(\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$, which is true as $x_i|_{\tilde{X}} = y_i \circ \Phi|_{\tilde{X}}$, and $z_{ij} \circ (0) = 0$, and $D_{ij} \circ (\operatorname{id} \times \Phi|_{\tilde{X}}) = 0$ as $D_{ij} \in H^0(\pi_{U'}^{-1}(I_X^{k-1}))$ for $k \ge 2$. For (ii), as $\pi_U : T \to U$ is étale with $e = f \circ \pi_U$, we see that $\pi_U|_Q : Q \to X$ is étale. But

For (ii), as $\pi_U: T \to U$ is étale with $e = f \circ \pi_U$, we see that $\pi_U|_Q: Q \to X$ is étale. But $\pi_U|_Q \circ (\operatorname{id} \times \Phi|_X \times (0))|_{\tilde{X}} = \operatorname{id}_{\tilde{X}}$. Hence $(\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$ is open in Q, and is also closed in Q as it is closed in T. Thus, by making W, T smaller to delete other components of Q, we can suppose that $Q = (\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$. Then $\pi_U|_Q: Q \to \tilde{X}$ is an isomorphism with the Zariski open neighbourhood \tilde{X} of x in X. Since $\pi_U: T \to U$ is étale with $e = f \circ \pi_U$, this extends to the k^{th} order thickenings, so $\pi_U|_{Q^{(k)}}: Q^{(k)} \to \tilde{X}^{(k)}$ is an isomorphism, proving (b).

For (c), first note that $Q = (\operatorname{id} \times \Phi|_X \times (0))(\tilde{X})$, so $\Phi \circ \pi_U|_Q = \pi_V|_Q$ is immediate. We have to extend this to the thickening $Q^{(k)}$. Write $I_Q \subset \mathcal{O}_T$ for the ideal of functions vanishing on Q. Then $I_Q = \pi_U^{-1}(I_X)$ as π_U identifies Q with $\tilde{X} \subseteq X$. We have

$$A_i \circ \pi_{U' \times V'} \in I_Q$$
 and $D_{ij} \circ \pi_{U' \times V'} \in I_Q^{k-1}$,

as $A_i \in H^0(\pi_{U'}^{-1}(I_X))$, $D_{ij} \in H^0(\pi_{U'}^{-1}(I_X^{k-1}))$. The second equation of (4.14) then shows that $z_{ij} \circ \pi_{\mathbb{C}^{n^2}} \in I_Q^{k-1}$,

since $Q = (\mathrm{id} \times \Phi|_X \times (0))(X)$ implies that $z_{ij} \circ \pi_{\mathbb{C}^{n^2}} = 0$ on Q, so we can neglect the terms $\sum_{l,m=1}^n C_{lm}(u,v) \cdot z_{li}z_{mj}$. Hence the first equation of (4.14) gives

$$x_i \circ \pi_U - y_i \circ \pi_V \in I_Q^k$$

As I_Q^k vanishes on $Q^{(k)}$, and $x_i|_{X'} = y_i \circ \Phi|_{X'}$, this gives

$$y_i \circ (\Phi \circ \pi_U|_{Q^{(k)}}) = x_i \circ \pi_U|_{Q^{(k)}} = y_i \circ (\pi_V|_{Q^{(k)}}).$$

Thus $\Phi \circ \pi_U|_{Q^{(k)}} = \pi_V|_{Q^{(k)}}$ follows, as (y_1, \ldots, y_n) are étale coordinates on V near $\pi_V(Q)$ and $\Phi \circ \pi_U|_Q = \pi_V|_Q$. This proves (c), and Proposition 4.3.

4.2. **Proof of Theorem 4.2 for** \mathbb{C} -schemes. Let U, V, f, g, X, Y and $\Phi : X^{(3)} \to Y^{(3)}$ be as in Theorem 4.2. Pick $x \in X$, and apply Proposition 4.3 with k = 2. This gives a smooth \mathbb{C} -scheme T and étale morphisms $\pi_U : T \to U, \pi_V : T \to V$ with $e := f \circ \pi_U = g \circ \pi_V : T \to \mathbb{C}$ and $Q := \operatorname{Crit}(e)$, such that $\pi_U|_{Q^{(2)}} : Q^{(2)} \to X^{(2)}$ is an étale open neighbourhood of x in $X^{(2)}$, and

$$\Phi \circ \pi_U|_{Q^{(2)}} = \pi_V|_{Q^{(2)}} : Q^{(2)} \longrightarrow Y^{(2)}.$$

Actually Proposition 4.3 proves more, that $\pi_U|_{Q^{(2)}}: Q^{(2)} \to X^{(2)}$ is an isomorphism with a Zariski open set $x \in \tilde{X}^{(2)} \subseteq X^{(2)}$, but we will not use this.

Thus, we can choose $\{(T^a, \pi^a_U, \pi^a_V, e^a, Q^a) : a \in A\}$, where A is an indexing set, such that $T^a, \pi^a_U, \pi^a_V, e^a, Q^a$ satisfy the conditions above for each $a \in A$, and $\{\pi^a_U|_{Q^a} : Q^a \to X\}_{a \in A}$ is an étale open cover of X. Then for each $a \in A$, by Definition 2.15 we have isomorphisms

$$\mathcal{PV}_{\pi_U^a}: \mathcal{PV}_{T^a,e^a}^{\bullet} \longrightarrow \pi_U^a|_{Q^a}^* \big(\mathcal{PV}_{U,f}^{\bullet}\big), \quad \mathcal{PV}_{\pi_V^a}: \mathcal{PV}_{T^a,e^a}^{\bullet} \longrightarrow \pi_V^a|_{Q^a}^* \big(\mathcal{PV}_{V,f}^{\bullet}\big).$$

Noting that $\pi_V^a|_{Q^a} = \Phi|_X \circ \pi_U^a|_{Q^a}$, we may define an isomorphism

$$\Omega^{a} = \mathcal{P}\mathcal{V}_{\pi^{a}_{V}} \circ \mathcal{P}\mathcal{V}_{\pi^{a}_{U}}^{-1} : \pi^{a}_{U}|_{\mathcal{Q}^{a}}^{*} \left(\mathcal{P}\mathcal{V}_{U,f}^{\bullet}\right) \longrightarrow \pi^{a}_{U}|_{\mathcal{Q}^{a}}^{*} \left(\Phi|_{X}^{*}\left(\mathcal{P}\mathcal{V}_{V,f}^{\bullet}\right)\right).$$
(4.16)

For $a, b \in A$, define $T^{ab} = T^a \times_{\pi^a_U, U, \pi^b_U} T^b$ to be the \mathbb{C} -scheme fibre product, so that T^{ab} is a smooth \mathbb{C} -scheme and the projections $\Pi_{T^a} : T^{ab} \to T^a$, $\Pi_{T^b} : T^{ab} \to T^b$ are étale. Define $e^{ab} = e^a \circ \Pi_{T^a} : T^a \to \mathbb{C}$. Then

$$e^{ab} = e^a \circ \Pi_{T^a} = g \circ \pi^a_V \circ \Pi_{T^a} = f \circ \pi^a_U \circ \Pi_{T^a}$$

= $f \circ \pi^b_U \circ \Pi_{T^b} = g \circ \pi^b_V \circ \Pi_{T^b} = e^b \circ \Pi_{T^b}.$ (4.17)

Write $Q^{ab} = \operatorname{Crit}(e^{ab})$. Then $\Pi_{T^a}|_{Q^{ab}} : Q^{ab} \to Q^a$ and $\Pi_{T^b}|_{Q^{ab}} : Q^{ab} \to Q^b$ are étale. Now $\pi^a_U \circ \Pi_{T^a} = \pi^b_U \circ \Pi_{T^b}$ and $\Phi \circ \pi^a_U|_{Q^a(2)} = \pi^a_V|_{Q^a(2)}$ imply that

$$(\pi_{V}^{a} \circ \Pi_{T^{a}})|_{Q^{ab}(2)} = \pi_{V}^{a}|_{Q^{a}(2)} \circ \Pi_{T^{a}}|_{Q^{ab}(2)}$$

$$= \Phi|_{X^{(2)}} \circ \pi_{U}^{a}|_{Q^{a}(2)} \circ \Pi_{T^{a}}|_{Q^{ab}(2)} = \Phi|_{X^{(2)}} \circ (\pi_{U}^{a} \circ \Pi_{T^{a}})|_{Q^{ab}(2)}$$

$$= \Phi|_{X^{(2)}} \circ (\pi_{U}^{b} \circ \Pi_{T^{b}})|_{Q^{ab}(2)} = \Phi|_{X^{(2)}} \circ \pi_{U}^{a}|_{Q^{b}(2)} \circ \Pi_{T^{b}}|_{Q^{ab}(2)}$$

$$= \pi_{V}^{a}|_{Q^{b}(2)} \circ \Pi_{T^{b}}|_{Q^{ab}(2)} = (\pi_{V}^{b} \circ \Pi_{T^{b}})|_{Q^{ab}(2)}.$$

$$(4.18)$$

Hence $(\pi_V^a \circ \Pi_{T^a})|_{Q^{ab}} = (\pi_V^b \circ \Pi_{T^b})|_{Q^{ab}}$. Moreover, as $TQ^{ab\,(2)}|_{Q^{ab}} = T(T^{ab})|_{Q^{ab}}$, we see that $d(\pi_V^a \circ \Pi_{T^a})|_{Q^{ab}} = d(\pi_V^b \circ \Pi_{T^b})|_{Q^{ab}}$, so that

$$d(\pi_V^b \circ \Pi_{T^b})|_{Q^{ab}}^{-1} \circ d(\pi_V^a \circ \Pi_{T^a})|_{Q^{ab}} = id: T(T^{ab})|_{Q^{ab}} \longrightarrow T(T^{ab})|_{Q^{ab}}$$

So det $\left(d(\pi_V^b \circ \Pi_{T^b})|_{Q^{ab}}^{-1} \circ d(\pi_V^a \circ \Pi_{T^a})|_{Q^{ab}}\right) = 1$. Thus, applying Theorem 3.1 with T^{ab} , V, Q^{ab} , $\pi_V^a \circ \Pi_{T^a}$, $\pi_V^b \circ \Pi_{T^b}$, e^{ab} , f in place of V, W, X, Φ , Ψ , f, g gives

$$\mathcal{PV}_{\pi_V^a \circ \Pi_{T^a}} = \mathcal{PV}_{\pi_V^b \circ \Pi_{T^b}} : \mathcal{PV}_{T^{ab}, e^{ab}}^{\bullet} \longrightarrow (\pi_V^a \circ \Pi_{T^a})|_{Q^{ab}}^* (\mathcal{PV}_{V,g}^{\bullet}).$$
(4.19)

124

Now

$$\Pi_{T^{a}}|_{Q^{ab}}^{*}(\Omega^{a}) = \Pi_{T^{a}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{V}^{a}}) \circ \Pi_{T^{a}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{U}^{a}})^{-1}$$

$$= \left[\Pi_{T^{a}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{V}^{a}}) \circ \mathcal{PV}_{\Pi_{T^{a}}}\right] \circ \left[\Pi_{T^{a}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{U}^{a}}) \circ \mathcal{PV}_{\Pi_{T^{a}}}\right]^{-1}$$

$$= \mathcal{PV}_{\pi_{V}^{b} \circ \Pi_{T^{a}}} \circ \mathcal{PV}_{\pi_{U}^{a} \circ \Pi_{T^{a}}}^{-1} = \mathcal{PV}_{\pi_{V}^{b} \circ \Pi_{T^{b}}} \circ \mathcal{PV}_{\pi_{U}^{b} \circ \Pi_{T^{b}}}^{-1}$$

$$= \left[\Pi_{T^{b}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{V}^{a}}) \circ \mathcal{PV}_{\Pi_{T^{b}}}\right] \circ \left[\Pi_{T^{b}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{U}^{b}}) \circ \mathcal{PV}_{\Pi_{T^{b}}}\right]^{-1}$$

$$= \Pi_{T^{b}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{V}^{b}}) \circ \Pi_{T^{b}}|_{Q^{ab}}^{*}(\mathcal{PV}_{\pi_{U}^{b}})^{-1} = \Pi_{T^{b}}|_{Q^{ab}}^{*}(\Omega^{b}),$$

$$(4.20)$$

using (4.16) in the first and seventh steps, (2.18) in the third and fifth, and (4.19) and

$$\pi^a_U \circ \Pi_{T^a} = \pi^b_U \circ \Pi_{T^b}$$

in the fourth. Therefore Theorem 2.7(i) applied to the étale open cover $\{\pi^a_U|_{Q^a}: Q^a \to X\}_{a \in A}$ of X shows that there is a unique isomorphism Ω_{Φ} in (4.2) with $\pi^a_U|^*_{Q^a}(\Omega_{\Phi}) = \Omega^a$ for all $a \in A$.

Suppose $\{(T^a, \ldots, Q^a) : a \in A\}$ and $\{(T'^a, \ldots, Q'^a) : a \in A'\}$ are alternative choices above, yielding morphisms Ω_{Φ} and Ω'_{Φ} in (4.2). By running the same construction using the family $\{(T^a, \ldots, Q^a) : a \in A\} \amalg \{(T'^a, \ldots, Q'^a) : a \in A'\}$, we get a third morphism Ω''_{Φ} in (4.2), such that $\pi^a_U|^*_{Q^a}(\Omega_{\Phi}) = \Omega^a = \pi^a_U|^*_{Q^a}(\Omega'_{\Phi})$ for $a \in A$, giving $\Omega_{\Phi} = \Omega''_{\Phi}$, and

$$\pi_U'^a|_{Q'^a}^*(\Omega_{\Phi}') = \Omega'^a = \pi_U'^a|_{Q'^a}^*(\Omega_{\Phi}'')$$

for $a \in A'$, which forces $\Omega'_{\Phi} = \Omega''_{\Phi}$. Thus $\Omega_{\Phi} = \Omega'_{\Phi}$, so Ω_{Φ} is independent of the choice of $\{(T^a, \ldots, Q^a) : a \in A\}$ above.

Let T, π_U, π_V, e, Q be as in Theorem 4.2. Applying the argument above using the family $\{(T^a, \ldots, Q^a) : a \in A\} \amalg \{(T, \pi_U, \pi_V, e, Q)\}$ shows that Ω_{Φ} satisfies

$$\pi_U|_Q^*(\Omega_\Phi) = \mathcal{PV}_{\pi_V} \circ \mathcal{PV}_{\pi_U}^{-1}$$

by (4.16). Thus (4.3) holds.

To show that (4.4)-(4.5) commute, we can combine equations (2.16)-(2.17), (4.16) and $\pi^a_U|^*_{Q^a}(\Omega_{\Phi}) = \Omega^a$ to show that $\pi^a_U|^*_{Q^a}$ applied to (4.4)-(4.5) commute in Perv (Q^a) for each $a \in A$, so (4.4)-(4.5) commute by Theorem 2.7(i).

Suppose there exists an étale morphism $\Xi: U \to V$ with $f = g \circ \Xi: U \to \mathbb{C}$ and

$$\Xi|_{X^{(3)}} = \Phi : X^{(3)} \to Y^{(3)}$$

Then as we have to prove, we have

$$\mathcal{PV}_{\Xi} = \operatorname{id}_X^*(\Omega_{\Phi}) \circ \mathcal{PV}_{\operatorname{id}_U} = \Omega_{\Phi} \circ \operatorname{id}_{\mathcal{PV}_{U,f}^{\bullet}} = \Omega_{\Phi},$$

where in the first step we use (4.3) with T = U, $\pi_U = \mathrm{id}_U$, $\pi_V = \Xi$, e = f, and Q = X, and in the second we use $\mathcal{PV}_{\mathrm{id}_U} = \mathrm{id}_{\mathcal{PV}_{U,f}^{\bullet}}$ from Definition 2.15.

Suppose W is another smooth \mathbb{C} -scheme, $h: W \to \mathbb{C}$ is regular, $Z = \operatorname{Crit}(h)$, and $\Psi: Y^{(3)} \to Z^{(3)}$ is an isomorphism with $h^{(3)} \circ \Psi = g^{(3)}$. Let $x \in X$, and set $y = \Phi(x) \in Y$. Proposition 4.3 for x, Φ gives a smooth T and étale $\pi_U: T \to U, \pi_V: T \to V$ with $e := f \circ \pi_U = g \circ \pi_V$ and $Q := \operatorname{Crit}(e)$, such that $\pi_U|_{Q^{(2)}}: Q^{(2)} \to X^{(2)}$ is an étale open neighbourhood of x, and $\Phi \circ \pi_U|_{Q^{(2)}} = \pi_V|_{Q^{(2)}}$. Proposition 4.3 for y, Ψ gives smooth \tilde{T} and étale $\tilde{\pi}_V: \tilde{T} \to V$, $\tilde{\pi}_W: \tilde{T} \to W$ with $\tilde{e} := g \circ \tilde{\pi}_V = h \circ \tilde{\pi}_W$ and $\tilde{Q} := \operatorname{Crit}(\tilde{e})$, such that $\tilde{\pi}_V|_{\tilde{Q}^{(2)}}: \tilde{Q}^{(2)} \to Y^{(2)}$ is an étale open neighbourhood of y, and $\Psi \circ \tilde{\pi}_V|_{\tilde{Q}^{(2)}} = \tilde{\pi}_W|_{\tilde{Q}^{(2)}}$.

Define $\hat{T} = T \times_{\pi_V, V, \tilde{\pi}_V} \tilde{T}$ with projections $\Pi_T : \hat{T} \to T$, $\Pi_{\tilde{T}} : \hat{T} \to \tilde{T}$. Then \hat{T} is smooth and $\Pi_T, \Pi_{\tilde{T}}$ are étale, as T, \tilde{T}, V are smooth and $\pi_V, \tilde{\pi}_V$ étale. Define $\hat{\pi}_U = \pi_U \circ \Pi_T : \hat{T} \to U$ and $\hat{\pi}_W = \tilde{\pi}_W \circ \Pi_{\tilde{T}} : \hat{T} \to W$. Then $\hat{\pi}_U, \hat{\pi}_W$ are étale. Set $\hat{e} = f \circ \hat{\pi}_U : \hat{T} \to \mathbb{C}$, and write $\hat{Q} = \operatorname{Crit}(\hat{e})$. Then

$$\hat{e} = f \circ \hat{\pi}_U = f \circ \pi_U \circ \Pi_T = g \circ \pi_V \circ \Pi_T = g \circ \tilde{\pi}_V \circ \Pi_{\tilde{T}} = h \circ \tilde{\pi}_W \circ \Pi_{\tilde{T}} = h \circ \hat{\pi}_W.$$

Also $\hat{\pi}_U|_{\hat{Q}^{(2)}}: \hat{Q}^{(2)} \to X^{(2)}$ is an étale open neighbourhood of x, and

$$\begin{split} (\Psi \circ \Phi) \circ \hat{\pi}_U |_{\hat{Q}^{(2)}} &= \Psi \circ \Phi \circ \pi_U |_{Q^{(2)}} \circ \Pi_T |_{\hat{Q}^{(2)}} = \Psi \circ \pi_V |_{Q^{(2)}} \circ \Pi_T |_{\hat{Q}^{(2)}} \\ &= \Psi \circ \tilde{\pi}_V |_{\hat{Q}^{(2)}} \circ \Pi_{\tilde{T}} |_{\hat{Q}^{(2)}} = \tilde{\pi}_W |_{\hat{Q}^{(2)}} \circ \Pi_{\tilde{T}} |_{\hat{Q}^{(2)}} = \hat{\pi}_W |_{\hat{Q}^{(2)}} \end{split}$$

Thus we may apply (4.3) for Ω_{Φ} with $T, \pi_U, \pi_V, \ldots, Q$, and for Ω_{Ψ} with $\hat{T}, \hat{\pi}_V, \hat{\pi}_W, \ldots, \hat{Q}$, and for $\Omega_{\Psi \circ \Phi}$ with $\hat{T}, \hat{\pi}_U, \hat{\pi}_W, \ldots, \hat{Q}$. This yields

$$\pi_{U}|_{Q}^{*}(\Omega_{\Phi}) = \mathcal{P}\mathcal{V}_{\pi_{V}} \circ \mathcal{P}\mathcal{V}_{\pi_{U}}^{-1}, \quad \tilde{\pi}_{V}|_{\tilde{Q}}^{*}(\Omega_{\Psi}) = \mathcal{P}\mathcal{V}_{\tilde{\pi}_{W}} \circ \mathcal{P}\mathcal{V}_{\tilde{\pi}_{V}}^{-1}, \\ \hat{\pi}_{U}|_{\tilde{Q}}^{*}(\Omega_{\Psi\circ\Phi}) = \mathcal{P}\mathcal{V}_{\hat{\pi}_{W}} \circ \mathcal{P}\mathcal{V}_{\hat{\pi}_{U}}^{-1}.$$

$$(4.21)$$

Now

$$\begin{split} \hat{\pi}_{U}|_{\hat{Q}}^{*}\left(\Omega_{\Psi\circ\Phi}\right) &= \mathcal{P}\mathcal{V}_{\hat{\pi}_{W}}\circ\mathcal{P}\mathcal{V}_{\hat{\pi}_{U}}^{-1} = \mathcal{P}\mathcal{V}_{\tilde{\pi}_{W}\circ\Pi_{\bar{T}}}\circ\mathcal{P}\mathcal{V}_{\pi_{U}\circ\Pi_{T}}^{-1} \\ &= \left[\Pi_{\bar{T}}\right]_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\tilde{\pi}_{W}}\right)\circ\mathcal{P}\mathcal{V}_{\Pi_{\bar{T}}}\right]\circ\left[\Pi_{T}\right]_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\pi_{U}}\right)\circ\mathcal{P}\mathcal{V}_{\Pi_{T}}\right]^{-1} \\ &= \Pi_{\bar{T}}|_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\tilde{\pi}_{W}}\circ\mathcal{P}\mathcal{V}_{\tilde{\pi}_{V}}^{-1}\right)\circ\Pi_{\bar{T}}|_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\pi_{V}}\right)\circ\mathcal{P}\mathcal{V}_{\Pi_{\bar{T}}}\circ\mathcal{P}\mathcal{V}_{\Pi_{T}}^{-1}\circ\Pi_{T}|_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\pi_{U}}^{-1}\right) \\ &= \Pi_{\bar{T}}|_{\hat{Q}}^{*}\left(\tilde{\pi}_{V}|_{\hat{Q}}^{*}\left(\Omega_{\Psi}\right)\right)\circ\mathcal{P}\mathcal{V}_{\tilde{\pi}_{V}\circ\Pi_{\bar{T}}}\circ\mathcal{P}\mathcal{V}_{\Pi_{T}}^{-1}\circ\Pi_{T}|_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\pi_{U}}^{-1}\right) \\ &= \left[\tilde{\pi}_{V}|_{\hat{Q}}\circ\Pi_{\bar{T}}|_{\hat{Q}}\right]^{*}\left(\Omega_{\Psi}\right)\circ\mathcal{P}\mathcal{V}_{\pi_{V}}\circ\mathcal{P}\mathcal{V}_{\Pi_{T}}\circ\mathcal{P}\mathcal{V}_{\Pi_{T}}^{-1}\circ\Pi_{T}|_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\pi_{U}}^{-1}\right) \\ &= \left[\Phi_{|_{X}}\circ\pi_{U}|_{Q}\circ\Pi_{T}|_{\hat{Q}}\right]^{*}\left(\Omega_{\Psi}\right)\circ\Pi_{T}|_{\hat{Q}}^{*}\left(\mathcal{P}\mathcal{V}_{\pi_{V}}\circ\mathcal{P}\mathcal{V}_{\pi_{U}}^{-1}\right) \\ &= \left[\pi_{U}\circ\Pi_{T}\right]|_{\hat{O}}^{*}\left(\Phi_{|_{X}}^{*}\left(\Omega_{\Psi}\right)\right)\circ\Pi_{T}|_{\hat{O}}^{*}\left(\pi_{U}|_{Q}^{*}\left(\Omega_{\Phi}\right)\right) = \hat{\pi}_{U}|_{\hat{O}}^{*}\left(\Phi_{|_{X}}^{*}\left(\Omega_{\Psi}\right)\circ\Omega_{\Phi}\right), \end{split}$$

using (4.21) in the first, fifth and ninth steps, (2.18) in the third, fifth and seventh steps, $\pi_V \circ \Pi_T = \tilde{\pi}_V \circ \Pi_{\tilde{T}}$ in the sixth and seventh, and $\Phi \circ \pi_U|_Q = \pi_V|_Q$ in the eighth. Thus, for each $x \in X$, we have constructed an étale open neighbourhood $\hat{\pi}_U|_{\hat{Q}} : \hat{Q} \to X$ such that $\hat{\pi}_U|_{\hat{Q}}^*$ applied to (4.6) holds. Equation (4.6) follows by Theorem 2.7(i). Finally, if U = V, f = g, X = Y and $\Phi = \operatorname{id}_{X^{(3)}}$ then $\Omega_{\operatorname{id}_X^{(3)}} = \operatorname{id}_{\mathcal{P}V_{U,f}^{\bullet}}$ follows by taking $\Xi = \operatorname{id}_U$ in the fourth paragraph of the theorem. This proves Theorem 4.2 for perverse sheaves on \mathbb{C} -schemes.

4.3. \mathscr{D} -modules and mixed Hodge modules. Once again, the proof of Proposition 4.3 is completely algebraic, so applies in the other contexts of §2.6–§2.10. Theorem 4.2 then follows for our other contexts from that and the general framework of §2.5.

5. Stabilizing vanishing cycles

To set up notation for our main result, which is Theorem 5.4 below, we need the following theorem, which is proved in Joyce [23, Prop.s 2.22, 2.23 & 2.25].

Theorem 5.1 (Joyce [23]). Let U, V be smooth \mathbb{C} -schemes, $f : U \to \mathbb{C}$, $g : V \to \mathbb{C}$ be regular, and $X = \operatorname{Crit}(f)$, $Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V. Let $\Phi : U \hookrightarrow V$ be a closed embedding of \mathbb{C} -schemes with $f = g \circ \Phi : U \to \mathbb{C}$, and suppose $\Phi|_X : X \to V \supseteq Y$ is an isomorphism $\Phi|_X : X \to Y$. Then:

(i) For each $x \in X \subseteq U$ there exist smooth \mathbb{C} -schemes U', V', a point $x' \in U'$ and morphisms $\iota: U' \to U, j: V' \to V, \Phi': U' \to V', \alpha: V' \to U$ and $\beta: V' \to \mathbb{C}^n$, where $n = \dim V - \dim U$,

such that $\iota(x') = x$, and ι, \jmath and $\alpha \times \beta : V \to U \times \mathbb{C}^n$ are étale, and the following diagram commutes

and $g \circ j = f \circ \alpha + (z_1^2 + \dots + z_n^2) \circ \beta : V' \to \mathbb{C}$. Thus, setting $f' := f \circ \iota : U' \to \mathbb{C}$, $g' := g \circ j : V' \to \mathbb{C}, X' := \operatorname{Crit}(f') \subseteq U'$, and $Y' := \operatorname{Crit}(g') \subseteq V'$, then $f' = g' \circ \Phi' : U' \to \mathbb{C}$, and $\Phi'|_{X'} : X' \to Y', \iota|_{X'} : X' \to X, j|_{Y'} : Y' \to Y, \alpha|_{Y'} : Y' \to X$ are étale. We also require that $\Phi \circ \alpha|_{Y'} = j|_{Y'} : Y' \to Y$.

(ii) Write N_{UV} for the normal bundle of $\Phi(U)$ in V, regarded as an algebraic vector bundle on U in the exact sequence of vector bundles on U:

$$0 \longrightarrow TU \xrightarrow{d\Phi} \Phi^*(TU) \xrightarrow{\Pi_{UV}} N_{UV} \longrightarrow 0.$$
 (5.2)

Then there exists a unique $q_{UV} \in H^0(S^2 N_{UV}^*|X)$ which is a nondegenerate quadratic form on $N_{UV}|_X$, such that whenever $U', V', \iota, \jmath, \Phi', \beta, n, X'$ are as in (i), writing $\langle dz_1, \ldots, dz_n \rangle_{U'}$ for the trivial vector bundle on U' with basis dz_1, \ldots, dz_n , there is a natural isomorphism $\hat{\beta} : \langle dz_1, \ldots, dz_n \rangle_{U'} \to \iota^*(N_{UV}^*)$ making the following diagram commute:

$$\iota^{*}(N_{UV}^{*}) \xrightarrow{\iota^{*}(\Pi_{UV}^{*})} \iota^{*} \circ \Phi^{*}(T^{*}V) = \Phi'^{*} \circ j^{*}(T^{*}V)$$

$$\stackrel{\wedge}{\beta} \qquad \Phi'^{*}(\mathrm{d}j^{*}) \downarrow \qquad (5.3)$$

$$\langle \mathrm{d}z_{1}, \dots, \mathrm{d}z_{n} \rangle_{U'} = \Phi'^{*} \circ \beta^{*}(T_{0}^{*}\mathbb{C}^{n}) \xrightarrow{\Phi'^{*}(\mathrm{d}\beta^{*})} \Phi'^{*}(T^{*}V'),$$

and
$$\iota|_{X'}^*(q_{UV}) = (S^2\hat{\beta})|_{X'}(\mathrm{d} z_1 \otimes \mathrm{d} z_1 + \dots + \mathrm{d} z_n \otimes \mathrm{d} z_n).$$
 (5.4)

(iii) Now suppose W is another smooth \mathbb{C} -scheme, $h: W \to \mathbb{C}$ is regular, $Z = \operatorname{Crit}(h)$ as a \mathbb{C} -subscheme of W, and $\Psi: V \to W$ is a closed embedding of \mathbb{C} -schemes with $g = h \circ \Psi: V \to \mathbb{C}$ and $\Psi|_Y: Y \to Z$ an isomorphism. Define N_{VW}, q_{VW} and N_{UW}, q_{UW} using $\Psi: V \to W$ and $\Psi \circ \Phi: U \to W$ as in (ii) above. Then there are unique morphisms $\gamma_{UVW}, \delta_{UVW}$ which make the following diagram of vector bundles on U commute, with straight lines exact:



Restricting to X gives an exact sequence of vector bundles:

$$0 \longrightarrow N_{UV}|_X \xrightarrow{\gamma_{UVW}|_X} N_{UW}|_X \xrightarrow{\delta_{UVW}|_X} \Phi|_X^*(N_{VW}) \longrightarrow 0.$$
(5.6)

Then there is a natural isomorphism of vector bundles on X

$$N_{UW}|_X \cong N_{UV}|_X \oplus \Phi|_X^*(N_{VW}), \tag{5.7}$$

compatible with the exact sequence (5.6), which identifies

$$q_{UW} \cong q_{UV} \oplus \Phi|_X^*(q_{VW}) \oplus 0 \quad under \ the \ splitting$$
$$S^2 N_{UW}|_X^* \cong S^2 N_{UV}|_X^* \oplus \Phi|_X^*(S^2 N_{VW}^*|_Y) \oplus \left(N_{UV}^*|_X \otimes \Phi|_X^*(N_{VW}^*)\right).$$
(5.8)

(iv) Analogues of (i)–(iii) hold for complex analytic spaces, replacing the smooth \mathbb{C} -schemes U, V, W by complex manifolds, the regular functions f, g, h by holomorphic functions, the \mathbb{C} -schemes X, Y, Z by complex analytic spaces, the étale open sets $\iota : U' \to U, j : V' \to V$ by complex analytic open sets $U' \subseteq U, V' \subseteq V$, and with $\alpha \times \beta : V' \to U \times \mathbb{C}^n$ a biholomorphism with a complex analytic open neighbourhood of (x, 0) in $U \times \mathbb{C}^n$.

Following [23, Def.s 2.26 & 2.34], we define:

Definition 5.2. Let U, V be smooth \mathbb{C} -schemes, $f : U \to \mathbb{C}, g : V \to \mathbb{C}$ be regular, and $X = \operatorname{Crit}(f), Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V. Suppose $\Phi : U \to V$ is a closed embedding of \mathbb{C} -schemes with $f = g \circ \Phi : U \to \mathbb{C}$ and $\Phi|_X : X \to Y$ an isomorphism. Then Theorem 5.1(ii) defines the normal bundle N_{UV} of U in V, a vector bundle on U of rank $n = \dim V - \dim U$, and a nondegenerate quadratic form $q_{UV} \in H^0(S^2 N_{UV}^*|_X)$. Taking top exterior powers in the dual of (5.2) gives an isomorphism of line bundles on U

$$\rho_{UV}: K_U \otimes \Lambda^n N^*_{UV} \xrightarrow{\cong} \Phi^*(K_V),$$

where K_U, K_V are the canonical bundles of U, V.

Write X^{red} for the reduced \mathbb{C} -subscheme of X. As q_{UV} is a nondegenerate quadratic form on $N_{VW}|_X$, its determinant $\det(q_{VW})$ is a nonzero section of $(\Lambda^n N^*_{VW})|_X^{\otimes^2}$. Define an isomorphism of line bundles on X^{red} :

$$J_{\Phi} = \rho_{UV}^{\otimes^2} \circ \left(\operatorname{id}_{K_U^2|_{X^{\operatorname{red}}}} \otimes \operatorname{det}(q_{UV})|_{X^{\operatorname{red}}} \right) : K_U^{\otimes^2}|_{X^{\operatorname{red}}} \xrightarrow{\cong} \Phi|_{X^{\operatorname{red}}}^* \left(K_V^{\otimes^2} \right).$$
(5.9)

Since principal $\mathbb{Z}/2\mathbb{Z}$ -bundles $\pi: P \to X$ in the sense of Definition 2.9 are an (étale or complex analytic) topological notion, and X^{red} and X have the same topological space (even in the étale or complex analytic topology), principal $\mathbb{Z}/2\mathbb{Z}$ -bundles on X^{red} and on X are equivalent. Define $\pi_{\Phi}: P_{\Phi} \to X$ to be the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle which parametrizes square roots of J_{Φ} on X^{red} . That is, (étale or complex analytic) local sections $s_{\alpha}: X \to P_{\Phi}$ of P_{Φ} correspond to local isomorphisms $\alpha: K_U|_{X^{\text{red}}} \to \Phi|_{X^{\text{red}}}^*(K_V)$ on X^{red} with $\alpha \otimes \alpha = J_{\Phi}$.

Now suppose W is another smooth \mathbb{C} -scheme, $h: W \to \mathbb{C}$ is regular, $Z = \operatorname{Crit}(h)$ as a \mathbb{C} -subscheme of W, and $\Psi: V \hookrightarrow W$ is a closed embedding of \mathbb{C} -schemes with $g = h \circ \Psi: V \to \mathbb{C}$ and $\Psi|_Y: Y \to Z$ an isomorphism. Then Theorem 5.1(iii) applies, and from (5.7)–(5.8) we can deduce that

$$J_{\Psi \circ \Phi} = \Phi|_{X^{\mathrm{red}}}^* (J_{\Psi}) \circ J_{\Phi} : K_U^{\otimes^2}|_{X^{\mathrm{red}}} \xrightarrow{\cong} (\Psi \circ \Phi)|_{X^{\mathrm{red}}}^* (K_W^{\otimes^2})$$
$$= \Phi|_{X^{\mathrm{red}}}^* [\Psi|_{Y^{\mathrm{red}}}^* (K_W^{\otimes^2})].$$
(5.10)

For the principal $\mathbb{Z}/2\mathbb{Z}$ -bundles $\pi_{\Phi} : P_{\Phi} \to X, \pi_{\Psi} : P_{\Psi} \to Y, \pi_{\Psi \circ \Phi} : P_{\Psi \circ \Phi} \to X$, equation (5.10) implies that there is a canonical isomorphism

$$\Xi_{\Psi,\Phi}: P_{\Psi\circ\Phi} \xrightarrow{\cong} \Phi|_X^*(P_\Psi) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_\Phi.$$
(5.11)

It is also easy to see that these $\Xi_{\Psi,\Phi}$ have an associativity property under triple compositions, that is, given another smooth \mathbb{C} -scheme T, regular $e: T \to \mathbb{C}$ with $Q := \operatorname{Crit}(e)$, and $\Upsilon: T \hookrightarrow U$ a closed embedding with $e = f \circ \Upsilon : T \to \mathbb{C}$ and $\Upsilon|_Q : Q \to X$ an isomorphism, then

$$(\mathrm{id}_{(\Phi\circ\Upsilon)|_{Q}^{*}(P_{\Psi})} \otimes \Xi_{\Phi,\Upsilon}) \circ \Xi_{\Psi,\Phi\circ\Upsilon} = (\Upsilon)|_{Q}^{*}(\Xi_{\Psi,\Phi}) \otimes \mathrm{id}_{P_{\Upsilon}}) \circ \Xi_{\Psi\circ\Phi,\Upsilon} : P_{\Psi\circ\Phi\circ\Upsilon} \longrightarrow (\Phi\circ\Upsilon)|_{Q}^{*}(P_{\Psi}) \otimes_{\mathbb{Z}/2\mathbb{Z}} \Upsilon|_{Q}^{*}(P_{\Phi}) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Upsilon}.$$

$$(5.12)$$

Analogues of all the above also work for complex manifolds and complex analytic spaces, as in Theorem 5.1(v).

The reason for restricting to X^{red} above is the following [23, Prop. 2.27], whose proof uses the fact that X^{red} is reduced in an essential way.

Lemma 5.3. In Definition 5.2, the isomorphism J_{Φ} in (5.9) and the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle $\pi_{\Phi}: P_{\Phi} \to X$ depend only on U, V, X, Y, f, g and $\Phi|_X: X \to Y$. That is, they do not depend on $\Phi: U \to V$ apart from $\Phi|_X: X \to Y$.

Using the notation of Definition 5.2, we can state our main result:

Theorem 5.4. (a) Let U, V be smooth \mathbb{C} -schemes, $f : U \to \mathbb{C}$, $g : V \to \mathbb{C}$ be regular, and $X = \operatorname{Crit}(f), Y = \operatorname{Crit}(g)$ as \mathbb{C} -subschemes of U, V. Let $\Phi : U \hookrightarrow V$ be a closed embedding of \mathbb{C} -schemes with $f = g \circ \Phi : U \to \mathbb{C}$, and suppose $\Phi|_X : X \to V \supseteq Y$ is an isomorphism $\Phi|_X : X \to Y$. Then there is a natural isomorphism of perverse sheaves on X:

$$\Theta_{\Phi}: \mathcal{PV}_{U,f}^{\bullet} \longrightarrow \Phi|_X^* \left(\mathcal{PV}_{V,g}^{\bullet} \right) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi}, \tag{5.13}$$

where $\mathcal{PV}_{U,f}^{\bullet}, \mathcal{PV}_{V,g}^{\bullet}$ are the perverse sheaves of vanishing cycles from §2.4, and P_{Φ} the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle from Definition 5.2, and if \mathcal{Q}^{\bullet} is a perverse sheaf on X then $\mathcal{Q}^{\bullet} \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi}$ is as in Definition 2.9. Also the following diagrams commute, where $\sigma_{U,f}, \sigma_{V,g}, \tau_{U,f}, \tau_{V,g}$ are as in (2.6)–(2.7):



If U = V, f = g, $\Phi = \operatorname{id}_U$ then $\pi_{\Phi} : P_{\Phi} \to X$ is trivial, and Θ_{Φ} corresponds to $\operatorname{id}_{\mathcal{PV}_{U,f}^{\bullet}}$ under the natural isomorphism $\operatorname{id}_X^*(\mathcal{PV}_{U,f}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi} \cong \mathcal{PV}_{U,f}^{\bullet}$.

(b) The isomorphism Θ_{Φ} in (5.13) depends only on U, V, X, Y, f, g and $\Phi|_X : X \to Y$. That is, if $\tilde{\Phi} : U \to V$ is an alternative choice for Φ with $\Phi|_X = \tilde{\Phi}|_X : X \to Y$, then $\Theta_{\Phi} = \Theta_{\tilde{\Phi}}$, noting that $P_{\Phi} = P_{\tilde{\Phi}}$ by Lemma 5.3.

(c) Now suppose W is another smooth \mathbb{C} -scheme, $h : W \to \mathbb{C}$ is a regular function, $Z = \operatorname{Crit}(h)$, and $\Psi : V \hookrightarrow W$ is a closed embedding with $g = h \circ \Psi : V \to \mathbb{C}$ and $\Psi|_Y : Y \to Z$ an isomorphism. Then Definition 5.2 defines principal $\mathbb{Z}/2\mathbb{Z}$ -bundles

$$\pi_{\Phi}: P_{\Phi} \longrightarrow X, \quad \pi_{\Psi}: P_{\Psi} \longrightarrow Y, \quad \pi_{\Psi \circ \Phi}: P_{\Psi \circ \Phi} \longrightarrow X$$

and an isomorphism $\Xi_{\Psi,\Phi}$ in (5.11), and part (a) defines isomorphisms of perverse sheaves $\Theta_{\Phi}, \Theta_{\Psi\circ\Phi}$ on X and Θ_{Ψ} on Y. Then the following commutes in Perv(X):

(d) The analogues of (a)–(c) also hold for \mathscr{D} -modules on \mathbb{C} -schemes, for perverse sheaves and \mathscr{D} -modules on complex analytic spaces, and for mixed Hodge modules on \mathbb{C} -schemes and complex analytic spaces, as in §2.6–§2.10.

Example 5.5. Let $U = \mathbb{C} \setminus \{0\}$ and $V = (\mathbb{C} \setminus \{0\}) \times \mathbb{C}$ as smooth \mathbb{C} -schemes, define regular $f: U \to \mathbb{C}$ and $g: V \to \mathbb{C}$ by f(x) = 0 and $g(x, y) = x^k y^2$ for fixed $k \in \mathbb{Z}$, and define $\Phi: U \to V$ by $\Phi: x \mapsto (x, 0)$, so that $f = g \circ \Phi: U \to \mathbb{C}$. Then $X := \operatorname{Crit}(f) = U$, and $Y := \operatorname{Crit}(g) = \{(x, y) \in V: kx^{k-1}y^2 = 2x^k y = 0\} = \{(x, y) \in V: y = 0\}$, as $x \neq 0$. Thus $\Phi|_X: X \to Y$ is an isomorphism.

In Theorem 5.1(ii), N_{UV}^* is the trivial line bundle on U with basis dy, and $q_{UV} = x^k dy \otimes dy$. In Definition 5.2, $K_U|_X$ and $\Phi|_X^*(K_V)$ are the trivial line bundles on $X = X^{\text{red}} = U$ with bases dx and $dx \wedge dy$, and J_{Φ} in (5.9) maps

$$J_{\Phi}: \mathrm{d} x \otimes \mathrm{d} x \longmapsto x^k \, (\mathrm{d} x \wedge \mathrm{d} y) \otimes (\mathrm{d} x \wedge \mathrm{d} y).$$

The principal $\mathbb{Z}/2\mathbb{Z}$ -bundle $\pi_{\Phi} : P_{\Phi} \to X$ in Definition 5.2 parametrizes $\alpha : K_U|_X \to \Phi|_X^*(K_V)$ with $\alpha \otimes \alpha = J_{\Phi}$. Writing $\alpha : dx \mapsto p \, dx \wedge dy$ for p a local function on $X = \mathbb{C} \setminus \{0\}, \alpha \otimes \alpha = J_{\Phi}$ reduces to $p^2 = x^k$. Thus, P_{Φ} parametrizes (étale local) square roots p of $x^k : \mathbb{C} \setminus \{0\} \to \mathbb{C} \setminus \{0\}$.

reduces to $p^2 = x^k$. Thus, P_{Φ} parametrizes (étale local) square roots p of $x^k : \mathbb{C} \setminus \{0\} \to \mathbb{C} \setminus \{0\}$. If k is even then x^k has a global square root $p = x^{k/2}$, so the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle P_{Φ} has a global section, and is trivial. If k is odd then x^k has no global square root on $X = \mathbb{C} \setminus \{0\}$, so P_{Φ} has no global section, and is nontrivial.

Thus, Theorem 5.4 implies that if k is even then $\mathcal{PV}_{V,g}^{\bullet} \cong A_Y[1]$ is the constant perverse sheaf on Y, but if k is odd then $\mathcal{PV}_{V,g}^{\bullet}$ is the twist of $A_Y[1]$ by the unique nontrivial principal $\mathbb{Z}/2\mathbb{Z}$ -bundle on $Y \cong \mathbb{C} \setminus \{0\}$.

5.1. Theorem 5.4(a): the isomorphism Θ_{Φ} . Let U, V, f, g, X, Y, Φ be as in Theorem 5.4(a), and use the notation N_{UV}, q_{UV} from Theorem 5.1(ii) and J_{Φ}, P_{Φ} from Definition 5.2. We will show that there exists a unique perverse sheaf morphism Θ_{Φ} in (5.13) which is characterized by the property that whenever $U', V', \iota, \jmath, \Phi', \alpha, \beta, X', Y', f', g'$ are as in Theorem 5.1(i) then the following diagram of isomorphisms in Perv(X') commutes:

$$\begin{split}
\iota|_{X'}^{*}(\mathcal{P}\mathcal{V}_{U,f}^{\bullet}) & \longrightarrow \iota|_{X'}^{*} \circ (\operatorname{id}_{X} \times 0)^{*}(\mathcal{P}\mathcal{V}_{U,f}^{\bullet} \boxtimes \mathcal{P}\mathcal{V}_{\mathbb{C}^{n},z_{1}^{2}+\dots+z_{n}^{2}}^{L}) \\
\downarrow \iota|_{X'}^{*}(\Theta_{\Phi}) & \iota|_{X'}^{*} \circ (\operatorname{id}_{X} \times 0)^{*}(\mathcal{T}\mathcal{S}_{U,f,\mathbb{C}^{n},z_{1}^{2}+\dots+z_{n}^{2}}^{-1}) \\
\iota|_{X'}^{*} \circ \Phi|_{X}^{*}(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} \iota|_{X'}^{*}(\mathcal{P}_{\Phi}) & \iota|_{X'}^{*} \circ (\operatorname{id}_{X} \times 0)^{*}(\mathcal{P}\mathcal{V}_{U \times \mathbb{C}^{n},f \boxplus z_{1}^{2}+\dots+z_{n}^{2}}) \\
\downarrow^{\delta} & \Phi'|_{X'}^{*} \circ \Phi|_{X}^{*}(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}) = \underbrace{\Phi'|_{X'}^{*}(\mathcal{P}\mathcal{V}_{J})} \\
\Psi'|_{X'}^{*} \circ \Phi|_{X}^{*}(\mathcal{P}\mathcal{V}_{V,g}^{\bullet}) = \underbrace{\Phi'|_{X'}^{*}(\mathcal{P}\mathcal{V}_{J})} & \Phi'|_{X'}^{*}(\mathcal{P}\mathcal{V}_{o \times \beta}^{\bullet}), \end{split} \tag{5.17}$$

where $\mathcal{TS}_{U,f,\mathbb{C}^n,z_1^2+\cdots+z_n^2}$ is as in (2.8), and γ, δ are defined as follows:

- (A) γ : PV[•]_{U,f} → (id_X×0)*(PV[•]_{U,f} ⊠ PV[•]_{Cⁿ,z²₁+...+z²_n}) in Perv(X) comes from the isomorphism PV[•]_{Cⁿ,z²₁+...+z²_n} ≅ A_{0} in (2.12).
 (B) The principal Z/2Z-bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}, q_{UV}|_{X^{red}}), as the bundle P_Φ → X comes from (N_{UV}|_{X^{red}}) = X f^U = X
- (B) The principal $\mathbb{Z}/2\mathbb{Z}$ -bundle $P_{\Phi} \to X$ comes from $(N_{UV}|_{X^{\text{red}}}, q_{UV}|_{X^{\text{red}}})$, as the bundle of square roots of $\det(q_{UV}|_{X^{\text{red}}})$. Thus, the pullback $\iota|_{X'}^*(P_{\Phi}) \to X'$ comes from $(\iota|_{X'^{\text{red}}}^*(N_{UV}), \iota|_{X'^{\text{red}}}^*(q_{UV}))$. Now Theorem 5.1(ii) defines

$$\hat{\beta}|_{X'^{\mathrm{red}}} : \langle \mathrm{d}z_1, \dots, \mathrm{d}z_n \rangle_{X'^{\mathrm{red}}} \xrightarrow{\cong} \iota|_{X'^{\mathrm{red}}}^*(N_{UV}^*)$$

identifying $\sum_{j=1}^{n} dz_j^2$ with $\iota|_{X'^{\text{red}}}^*(q_{UV})$. Thus, $\hat{\beta}|_{X'^{\text{red}}}$ induces a trivialization of

$$\iota|_{X'}^*(P_\Phi) \longrightarrow X'.$$

Then $\delta : \iota|_{X'}^* \circ \Phi|_X^* (\mathcal{PV}_{V,g}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} \iota|_{X'}^* (P_{\Phi}) \to \iota|_{X'}^* \circ \Phi|_X^* (\mathcal{PV}_{V,g}^{\bullet})$ in $\operatorname{Perv}(X')$ comes from this trivialization of the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle $\iota|_{X'}^* (P_{\Phi}) \to X'$.

Since Theorem 5.1(i) holds for each $x \in X$, we may choose a family

$$\left\{ (U'_a, V'_a, \iota_a, \jmath_a, \Phi'_a, \alpha_a, \beta_a, f'_a, g'_a, X'_a, Y'_a) : a \in A \right\}$$

such that U'_a, V'_a, \ldots, Y'_a satisfy Theorem 5.1(i) for each $a \in A$, and $\{\iota'_a | X'_a : X'_a \to X\}_{a \in A}$ is an étale open cover of X. For each $a \in A$, define an isomorphism

$$\Theta_a: \iota_a|_{X'_a}^* \left(\mathcal{PV}_{U,f}^{\bullet} \right) \longrightarrow \iota_a|_{X'_a}^* \circ \Phi|_X^* \left(\mathcal{PV}_{V,g}^{\bullet} \right)$$

to make the following diagram of isomorphisms commute:

$$\begin{aligned}
\iota_{a}|_{X_{a}^{\prime}}^{*}\left(\mathcal{PV}_{U,f}^{\bullet}\right) & \longrightarrow \iota_{a}|_{X_{a}^{\prime}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*}\left(\mathcal{PV}_{U,f}^{\bullet} \boxtimes \mathcal{PV}_{\mathbb{C}^{n}, z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{V,g}^{\bullet}\right) & \iota_{a}|_{X_{a}^{\prime}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*}\left(\mathcal{TS}_{U,f,\mathbb{C}^{n}, z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{V,g}^{\bullet}\right) & \iota_{a}|_{X_{a}^{\prime}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*}\left(\mathcal{PV}_{U \times \mathbb{C}^{n}, f \boxplus z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\otimes_{\mathbb{Z}/2\mathbb{Z}}\iota_{a}|_{X_{a}^{\prime}}^{*}\left(\mathcal{P}\right) & \Phi_{a}^{\prime}|_{X_{a}^{\prime}}^{*} \circ (\alpha_{a} \times \beta_{a})|_{Y_{a}^{\prime}}^{*}\left(\mathcal{PV}_{U \times \mathbb{C}^{n}, f \boxplus z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{V,g}^{\bullet}\right) & \Phi_{a}^{\prime}|_{X_{a}^{\prime}}^{*}\left(\mathcal{PV}_{u \times \mathbb{C}^{n}, f \boxplus z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{V,g}^{\bullet}\right) & = \underbrace{\Phi_{a}^{\prime}|_{X_{a}^{\prime}}^{*}\left(\mathcal{PV}_{a} \times \beta_{a}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{u \times \mathbb{C}^{n}, f \boxplus z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{V,g}^{\bullet}\right) & = \underbrace{\Phi_{a}^{\prime}|_{X_{a}^{\prime}}^{*}\left(\mathcal{PV}_{a} \times \beta_{a}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{u \times \mathbb{C}^{n}, f \boxplus z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{V,g}^{\bullet}\right) & = \underbrace{\Phi_{a}^{\prime}|_{X_{a}^{\prime}}^{*}\left(\mathcal{PV}_{a} \times \beta_{a}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{u \times \mathbb{C}^{n}, f \boxplus z_{1}^{2} + \dots + z_{n}^{2}\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{X}^{*}\left(\mathcal{PV}_{v,g}^{\bullet}\right) & = \underbrace{\Phi_{a}^{\prime}|_{X_{a}^{\prime}}^{*}\left(\mathcal{PV}_{a} \times \beta_{a}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\left(\mathcal{PV}_{a} \times \beta_{a}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\left(\mathcal{PV}_{a}^{\bullet}\right)\right) \\
\downarrow_{a}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{A}|_{X_{a}^{\prime}}\left(\mathcal{PV}_{v,g}^{\bullet}\right) & = \underbrace{\Phi_{a}^{\prime}|_{X_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\left(\mathcal{PV}_{a}^{\bullet}\right)\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\left(\mathcal{PV}_{a}^{\bullet}\right)\right) \\
\downarrow_{a}|_{A}|_{X_{a}^{\prime}}^{*} \circ \Phi|_{A}|_{X_{a}^{\prime}}\left(\mathcal{PV}_{v,g}^{\bullet}\right) & = \underbrace{\Phi_{a}^{\prime}|_{X_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\left(\mathcal{PV}_{a}^{\bullet}\right)\right)}_{\Psi_{a}^{\prime}}\left(\mathcal{PV}_{a}|_{X_{a}^{\prime}}\left(\mathcal{PV}_{a}^{\bullet}\right)\right) \\
\downarrow_{a}|_{A}|_{A}|_{A}|_{A}|_{A}^{\bullet}\left(\mathcal{PV}_{a}|_{A}^{\bullet}\right)\right) \\
\downarrow_{a}|_{A}|_{A}|_{A}|_{A}^{\bullet}\left(\mathcal{PV}_{a$$

where γ is as in (A), and δ_a defined as in (B) above.

For $a, b \in A$, define

$$U'_{ab} = U'_a \times_{\iota_a, U, \iota_b} U'_b \quad \text{and} \quad V'_{ab} = V'_a \times_{\jmath_a, V, \jmath_b} V'_b,$$

with projections $\Pi_{U'_a} : U'_{ab} \to U'_a$, $\Pi_{U'_b} : U'_{ab} \to U'_b$, $\Pi_{V'_a} : V'_{ab} \to V'_a$, $\Pi_{V'_b} : V'_{ab} \to V'_b$. Then U'_{ab}, V'_{ab} are smooth and $\Pi_{U'_a}, \Pi_{U'_b}, \Pi_{V'_a}, \Pi_{V'_b}$ étale. The universal property of $V'_a \times_{\mathcal{J}_a, V, \mathcal{J}_b} V'_b$ gives a unique morphism $\Phi'_{ab} : U'_{ab} \to V'_{ab}$ with

$$\Pi_{V'_a} \circ \Phi'_{ab} = \Phi'_a \circ \Pi_{U'_a} \quad \text{and} \quad \Pi_{V'_b} \circ \Phi'_{ab} = \Phi'_b \circ \Pi_{U'_b}.$$
(5.19)

Set $f'_{ab} = f'_a \circ \Pi_{U'_a} : U'_{ab} \to \mathbb{C}, g'_{ab} = g'_a \circ \Pi_{V'_a} : V'_{ab} \to \mathbb{C}$ and $X'_{ab} = \operatorname{Crit}(f'_{ab}) \subseteq U'_{ab}, Y'_{ab} = \operatorname{Crit}(g'_{ab}) \subseteq V'_{ab}$. As for (4.17) we have

$$\begin{aligned} f'_{ab} &= f'_a \circ \Pi_{U'_a} = f \circ \iota_a \circ \Pi_{U'_a} = f \circ \iota_b \circ \Pi_{U'_b} = f'_b \circ \Pi_{U'_b}, \\ g'_{ab} &= g'_a \circ \Pi_{V'_a} = g \circ j_a \circ \Pi_{V'_a} = g \circ j_b \circ \Pi_{V'_b} = g'_b \circ \Pi_{V'_b} \\ &= (f \boxplus z_1^2 + \dots + z_n^2) \circ (\alpha_a \times \beta_a) \circ \Pi_{V'_a} \\ &= (f \boxplus z_1^2 + \dots + z_n^2) \circ (\alpha_b \times \beta_b) \circ \Pi_{V'_b}. \end{aligned}$$

Apply Theorem 3.1 with $V'_{ab}, U \times \mathbb{C}^n, (\alpha_a \times \beta_a) \circ \prod_{V'_a}, (\alpha_b \times \beta_b) \circ \prod_{V'_b}, g'_{ab}, \text{ and } f \boxplus z_1^2 + \dots + z_n^2$ in place of V, W, Φ, Ψ, f, g . The analogue of $\Phi|_X = \Psi|_X$ is

$$\begin{aligned} (\alpha_a \times \beta_a) \circ \Pi_{V'_a}|_{Y'_{ab}} &= ((\Phi|_X^{-1} \circ \Phi|_X \circ \alpha_a)|_{Y'_a} \times 0) \circ \Pi_{V'_a}|_{Y'_{ab}} \\ &= (\Phi|_X^{-1} \circ j_a \circ \Pi_{V'_a}|_{Y'_{ab}}) \times 0 = (\Phi|_X^{-1} \circ j_b \circ \Pi_{V'_b}|_{Y'_{ab}}) \times 0 \\ &= ((\Phi|_X^{-1} \circ \Phi|_X \circ \alpha_b)|_{Y'_b} \times 0) \circ \Pi_{V'_b}|_{Y'_{ab}} = (\alpha_b \times \beta_b) \circ \Pi_{V'_b}|_{Y'_{ab}}, \end{aligned}$$
(5.20)

using $\Phi|_X : X \to Y$ an isomorphism and $\beta_a|_{Y'_a} = 0$ in the first step, $j_a|_{Y'_a} = \Phi|_X \circ \alpha_a|_{Y'_a}$ in the second, $j_a \circ \prod_{V'_a} = j_b \circ \prod_{V'_b}$ in the third, $j_b|_{Y'_b} = \Phi|_X \circ \alpha_b|_{Y'_b}$ in the fourth, and $\beta_b|_{Y'_b} = 0$ in the fifth. Thus Theorem 3.1 gives

$$\mathcal{PV}_{(\alpha_{a}\times\beta_{a})\circ\Pi_{V_{a}'}} = \det\left[d\left((\alpha_{b}\times\beta_{b})\circ\Pi_{V_{b}'}\right)|_{Y_{ab}'^{red}}^{-1}\circ d\left((\alpha_{a}\times\beta_{a})\circ\Pi_{V_{a}'}\right)|_{Y_{ab}'}\right] \cdot \mathcal{PV}_{(\alpha_{b}\times\beta_{b})\circ\Pi_{V_{b}'}} : \mathcal{PV}_{V_{ab}',g_{ab}'} \longrightarrow (\alpha_{a}\times\beta_{a})\circ\Pi_{V_{a}'}|_{Y_{ab}'}^{*} \left(\mathcal{PV}_{U\times\mathbb{C}^{n},f\boxplus z_{1}^{2}+\dots+z_{n}^{2}}^{\bullet}\right)$$
(5.21)

in Perv (Y'_{ab}) , where det $[\cdots]$ maps $Y'^{\text{red}}_{ab} \to \{\pm 1\}$.

Consider the morphisms

$$\Pi_{U'_{a}}|_{X'_{ab}}^{*}(\delta_{a}), \Pi_{U'_{b}}|_{X'_{ab}}^{*}(\delta_{b}) : (\Phi \circ \iota_{a} \circ \Pi_{U'_{a}})|_{X'_{ab}}^{*} \left(\mathcal{PV}_{V,g}^{\bullet}\right) \otimes_{\mathbb{Z}/2\mathbb{Z}} (\iota_{a} \circ \Pi_{U'_{a}})|_{X'_{ab}}^{*} \left(\mathcal{P}_{\Phi}\right) \\ \longrightarrow (\Phi \circ \iota_{a} \circ \Pi_{U'_{a}})|_{X'_{ab}}^{*} \left(\mathcal{PV}_{V,g}^{\bullet}\right).$$

$$(5.22)$$

As in (B) above, these are defined using two different trivializations of the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle $(\iota_a \circ \Pi_{U'_a})|^*_{X'_{ab}}(P_{\Phi}) \to X'_{ab}$, defined using

$$\Pi_{U'_a}|_{X'^{\mathrm{red}}_{ab}}^*(\hat{\beta}_a), \Pi_{U'_b}|_{X'^{\mathrm{red}}_{ab}}^*(\hat{\beta}_b) : \langle \mathrm{d}z_1, \dots, \mathrm{d}z_n \rangle_{X'^{\mathrm{red}}_{ab}} \longrightarrow (\iota_a \circ \Pi_{U'_a})|_{X'^{\mathrm{red}}_{ab}}^*(N^*_{UV}),$$

which are isomorphisms of vector bundles on $X_{ab}^{\prime red}$ identifying the nondegenerate quadratic forms $\sum_{j=1}^{n} dz_{j}^{2}$ on $\langle dz_{1}, \ldots, dz_{n} \rangle_{X_{ab}^{\prime \prime red}}$ and $(\iota_{a} \circ \Pi_{U_{a}^{\prime}})|_{X_{ab}^{\prime \prime red}}^{*}(q_{UV})$ on $(\iota_{a} \circ \Pi_{U_{a}^{\prime}})|_{X_{ab}^{\prime \prime red}}^{*}(N_{UV}^{*})$, for $\hat{\beta}_{a}, \hat{\beta}_{b}$ as in (5.3). Thus we see that

$$\Pi_{U'_{a}}|_{X'_{ab}}^{*}(\delta_{a}) = \det\left[\Pi_{U'_{a}}|_{X'^{red}_{ab}}^{*}(\hat{\beta}_{a}) \circ \Pi_{U'_{b}}|_{X'^{red}_{ab}}^{*}(\hat{\beta}_{b})^{-1}\right] \cdot \Pi_{U'_{b}}|_{X'_{ab}}^{*}(\delta_{b}),$$
(5.23)

where det $[\cdots]$ maps $X_{ab}^{\prime red} \rightarrow \{\pm 1\}$ since both isomorphisms in (5.22) identify the same nondegenerate quadratic forms.

We have an exact sequence of vector bundles on $X_{ab}^{\prime red}$:

$$0 \longrightarrow TU'_{ab}|_{X_{ab}^{\prime \text{red}}} \longrightarrow \Phi'_{ab}|_{X_{ab}^{\prime \text{red}}}^* (TV'_{ab}) \longrightarrow (\iota_a \circ \Pi_{U'_a})|_{X_{ab}^{\prime \text{red}}}^* (N_{UV}) \longrightarrow 0.$$

Choosing a local splitting of this sequence, we may identify

$$\begin{aligned} \Phi_{ab}'|_{X_{ab}'^{\text{red}}}^* \Big[\mathrm{d} \Big((\alpha_b \times \beta_b) \circ \Pi_{V_b'} \Big) |_{Y_{ab}'^{\text{red}}}^{-1} \circ \mathrm{d} \Big((\alpha_a \times \beta_a) \circ \Pi_{V_a'} \Big) |_{Y_{ab}'^{\text{red}}} \Big] \\ & \cong \begin{pmatrix} \mathrm{id}_{TU_{ab}'|_{X_{ab}''^{\text{red}}}} & * \\ 0 & (\Pi_{U_a'}|_{X_{ab}'^{\text{red}}}^* (\hat{\beta}_a) \circ \Pi_{U_b'}|_{X_{ab}'^{\text{red}}}^* (\hat{\beta}_b)^{-1} \Big) * \end{pmatrix} \end{aligned}$$

Therefore

$$\Phi'_{ab}|_{X_{ab}^{\prime red}}^{X} \left(\det \left[d\left((\alpha_b \times \beta_b) \circ \Pi_{V_b^{\prime}} \right) |_{Y_{ab}^{\prime red}}^{-1} \circ d\left((\alpha_a \times \beta_a) \circ \Pi_{V_a^{\prime}} \right) |_{Y_{ab}^{\prime red}} \right] \right) \\
= \det \left[\Pi_{U_a^{\prime}} |_{X_{ab}^{\prime red}}^{*} (\hat{\beta}_a) \circ \Pi_{U_b^{\prime}} |_{X_{ab}^{\prime red}}^{*} (\hat{\beta}_b)^{-1} \right] : X_{ab}^{\prime red} \longrightarrow \{\pm 1\}.$$
(5.24)

Now

$$\begin{split} \Pi_{U_{a}'}|_{X_{ab}'}^{*}(\Theta_{a}) &= \Pi_{U_{a}'}|_{X_{ab}'}^{*}(\delta_{a}^{-1}) \circ (\Phi_{a}' \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\mathcal{PV}_{J_{a}}) \circ (\Phi_{a}' \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\mathcal{PV}_{\alpha_{a} \times \beta_{a}}) \\ &\circ ((\mathrm{id}_{X} \times 0) \circ \iota_{a} \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\mathcal{TS}_{U,f,\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}}^{-1}) \circ (\iota_{a} \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\gamma) \\ &= \Pi_{U_{a}'}|_{X_{ab}'}^{*}(\delta_{a}^{-1}) \circ (\Pi_{V_{a}'} \circ \Phi_{ab}')|_{X_{ab}'}^{*}(\mathcal{PV}_{J_{a}}) \circ \Phi_{ab}'|_{X_{ab}'}^{*}(\mathcal{PV}_{\Pi_{V_{a}'}}) \circ \Phi_{ab}'|_{X_{ab}'}^{*}(\mathcal{PV}_{\Pi_{V_{a}'}}) \\ &\circ (\Pi_{V_{a}'} \circ \Phi_{ab}')|_{X_{ab}'}^{*}(\mathcal{PV}_{a_{a} \times \beta_{a}}) \circ ((\mathrm{id}_{X} \times 0) \circ \iota_{a} \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\mathcal{TS}_{U,f,\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}}^{-1}) \circ (\iota_{a} \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\gamma) \\ &= \Pi_{U_{a}'}|_{X_{ab}'}^{*}(\delta_{a}^{-1}) \circ \Phi_{ab}'|_{X_{ab}'}^{*}(\mathcal{PV}_{J_{a} \circ \Pi_{V_{a}'}}) \circ \Phi_{ab}'|_{X_{ab}'}^{*}(\mathcal{PV}_{(\alpha_{a} \times \beta_{a}) \circ \Pi_{V_{a}'}'}) \\ &\circ ((\mathrm{id}_{X} \times 0) \circ \iota_{a} \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\mathcal{TS}_{U,f,\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}}^{-1}) \circ (\iota_{a} \circ \Pi_{U_{a}'})|_{X_{ab}'}^{*}(\gamma) \\ &= \det[\Pi_{U_{a}'}|_{X_{ab}'}^{*}(\mathrm{red}(\hat{\beta}_{a}) \circ \Pi_{U_{b}'})|_{X_{ab}'}^{*}(\mathrm{red}(\hat{\beta}_{b})^{-1}]^{-1} \cdot \\ \Phi_{ab}'|_{X_{ab}''}^{*}(\det[\operatorname{d}((\alpha_{b} \times \beta_{b}) \circ \Pi_{V_{b}'})]_{Y_{ab}''}^{*}(\alpha) \circ \operatorname{d}((\alpha_{a} \times \beta_{a}) \circ \Pi_{V_{a}'})|_{Y_{ab}'''}^{*}(\gamma) \\ &= \operatorname{det}[\Pi_{U_{b}'}|_{X_{ab}'}^{*}(\delta_{b}^{-1}) \circ (\Psi_{ab}'|_{X_{ab}'}^{*}(\mathcal{TS}_{U,f,\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}}) \circ (\iota_{b} \circ \Pi_{U_{b}'})|_{Y_{ab}''}^{*}(\gamma) \\ &= \Pi_{U_{b}'}|_{X_{ab}'}^{*}(\delta_{b}^{-1}) \circ (\Pi_{V_{b}'} \circ \Phi_{ab}'|_{X_{ab}'}^{*}(\mathcal{PV}_{J_{b}}) \circ \Phi_{ab}'|_{X_{ab}'}^{*}(\mathcal{PV}_{\Pi_{b}'}) \\ &\circ ((\Pi_{V_{b}'} \circ \Phi_{ab}')|_{X_{ab}'}^{*}(\mathcal{PV}_{ab}^{-1}) \circ ((\operatorname{id}_{X} \times 0) \circ \iota_{b} \circ \Pi_{U_{b}'})|_{X_{ab}'}^{*}(\mathcal{TS}_{U,f,\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}}^{-1}) \circ (\iota_{b} \circ \Pi_{U_{b}'})|_{X_{ab}'}^{*}(\mathcal{PV}_{ab}^{-1}) \\ &\circ ((\Pi_{V_{b}'} \circ \Phi_{ab}')|_{X_{ab}'}^{*}(\mathcal{PV}_{J_{b}}) \circ ((\operatorname{id}_{X} \times 0) \circ \iota_{b} \circ \Pi_{U_{b}'})|_{X_{ab}'}^{*}(\mathcal{PV}_{ab}^{-1}) \\ &\circ ((\operatorname{id}_{X} \times 0) \circ \iota_{b} \circ \Pi_{U_{b}'})|_{X_{ab}'}^{*}(\mathcal{TS}_{U,f,\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}}) \circ (\iota_{b} \circ \Pi_{U_{b}'})|_{X_{ab}'$$

using (5.18) in the first and seventh steps, (5.19) in the second and sixth, (2.18) in the third, (5.21), (5.23), $\iota_a \circ \Pi_{U'_a} = \iota_b \circ \Pi_{U'_b}$ and $\jmath_a \circ \Pi_{V'_a} = \jmath_b \circ \Pi_{V'_b}$ in the fourth, and (2.18) and (5.24) in the fifth. Therefore Theorem 2.7(i) applied to the étale open cover $\{\iota_a|_{X'_a}: X'_a \to X\}_{a \in A}$ of X shows that there is a unique isomorphism Θ_{Φ} in (5.13) with $\iota_a|_{X'_a}^*(\Theta_{\Phi}) = \Theta_a$ for all $a \in A$.

Suppose $\{(U'_a, \ldots, Y'_a) : a \in A\}$ and $\{(\tilde{U}'_a, \ldots, \tilde{Y}'_a) : a \in \tilde{A}\}$ are alternative choices above, yielding morphisms Θ_{Φ} and $\tilde{\Theta}_{\Phi}$ in (5.13). By running the same construction using the family $\{(U'_a, \ldots, Y'_a) : a \in A\}$ II $\{(\tilde{U}'_a, \ldots, \tilde{Y}'_a) : a \in \tilde{A}\}$, we can show that $\Theta_{\Phi} = \tilde{\Theta}_{\Phi}$, so Θ_{Φ} is independent of the choice of $\{(U'_a, \ldots, Y'_a) : a \in A\}$ above. Let $U', V', \iota, \jmath, \Phi', \alpha, \beta, X', Y', f', g'$ be as in Theorem 5.1(i). Constructing Θ_{Φ} using $\{(U'_a, \ldots, Y'_a) : a \in A\}$ III $\{(U'_a, \ldots, Y'_a) : a \in A\}$ we construction of Θ_{Φ} .

To see that (5.14)-(5.15) commute, in the situation of (5.17) we show that Verdier duality and monodromy operators commute with each morphism in (5.17). Going clockwise from the top left corner, $\iota|_{X'}^*(\gamma)$ is compatible with Verdier duality and monodromy because of the commutative diagrams

$$\begin{array}{cccc} A_{\{0\}} & & & & & & & \\ \downarrow_{\Gamma} & & & & & \\ \mathcal{P}\mathcal{V}^{\bullet}_{\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}} & & & & \\ \mathcal{P}\mathcal{V}^{\bullet}_{\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}} & & & \\ \mathcal{P}\mathcal{V}^{\bullet}_{\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}} & & & \\ \end{array} \\ \begin{array}{c} A_{\{0\}} & & & & \\ \downarrow_{\Gamma} & & & & \\ \psi_{\Gamma} & & & & \\ \tau_{\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}} & & & \\ \tau_{\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}} & & & \\ \mathcal{P}\mathcal{V}^{\bullet}_{\mathbb{C}^{n},\Sigma_{j}z_{j}^{2}} & & \\ \end{array} \\ \end{array}$$

where $\Gamma : A_{\{0\}} \to \mathcal{PV}_{\mathbb{C}^n, \Sigma_j z_j^2}^{\bullet}$ is the isomorphism used to define γ in (A) above. Equations (2.9)–(2.10) imply that $\iota|_{X'}^* \circ (\operatorname{id}_X \times 0)^* (\mathcal{TS}_{U,f,\mathbb{C}^n,\Sigma_j z_j^2}^{-1})$ is compatible with Verdier duality and monodromy, and (2.16)–(2.17) imply that $\Phi'|_{X'}^* (\mathcal{PV}_j), \Phi'|_{X'}^* (\mathcal{PV}_{\alpha \times \beta})$ are. Also δ is compatible with Verdier duality and monodromy, since these do not affect the trivialization of $\iota|_{X'}^* (P_{\Phi}) \to X'$ used to define δ in (B) above.

Thus by (5.17) we see that $\iota|_{X'}^*(\Theta_{\Phi})$ is compatible with Verdier duality and monodromy, that is, $\iota|_{X'}^*$ applied to (5.14)–(5.15) commute. Since we can form an étale open cover of X by such $\iota|_{X'}: X' \to X$, Theorem 2.7(i) implies that (5.14)–(5.15) commute.

Finally, if U = V, f = g and $\Phi = \mathrm{id}_U$ then $J_{\Phi} = \mathrm{id} : K_U^2|_{X^{\mathrm{red}}} \to K_U^2|_{X^{\mathrm{red}}}$ in (5.9), which has a natural square root $\alpha = \mathrm{id} : K_U|_{X^{\mathrm{red}}} \to K_U|_{X^{\mathrm{red}}}$, so $\pi_{\Phi} : P_{\Phi} \to X$ is trivial in Definition 5.2. In (5.17) we may put U' = V' = U, $\iota = \jmath = \alpha = \mathrm{id}_U$, n = 0, $\beta = 0$, X' = Y' = X, f' = g' = f, and then each morphism in (5.17) is essentially the identity on $\mathcal{PV}_{U,f}^{\bullet}$, so $\Theta_{\Phi} = \mathrm{id}_X^*(\Theta_{\Phi}) = \mathrm{id}_{\mathcal{PV}_{U,f}^{\bullet}}$. This proves Theorem 5.4(a).

5.2. Theorem 5.4(b): Θ_{Φ} depends only on $\Phi|_X : X \to Y$. Suppose $\Phi, \tilde{\Phi} : U \to V$ are alternative choices in Theorem 5.4(a) with

$$\Phi|_X = \Phi|_X : X \longrightarrow Y,$$

so that $P_{\Phi} = P_{\tilde{\Phi}}$ by Lemma 5.3. Fix $x \in X$, let $a \neq b$ be labels, and let $U'_a, V'_a, \iota_a, \jmath_a, \Phi'_a, \alpha_a, \beta_a, X'_a, Y'_a, f'_a, g'_a$ be as in Theorem 5.1(i) for x, Φ and U'_b, V'_b, \ldots, g'_b as in Theorem 5.1(i) for $x, \tilde{\Phi}$. As in §5.1, define Θ_a, Θ_b and $U'_{ab}, V'_{ab}, \Pi_{U'_a}, \Pi_{U'_b}, \Pi_{V'_a}, \Pi_{V'_b}, \Phi'_{ab}, f'_{ab}, g'_{ab}, X'_{ab}, Y'_{ab}$, and follow the proof in §5.1 from (5.19) as far as (5.25).

This proof does not actually need $U'_a, \ldots, g_a, \Theta_a$ and $U'_b, \ldots, g_b, \Theta_b$ to be defined using the same $\Phi: U \to V$, it only uses in (5.20)–(5.22) that $\Phi|_X: X \to Y$ is the same for U'_a, \ldots, Θ_a and U'_b, \ldots, Θ_b . Thus we can apply it with U'_a, \ldots, Θ_a defined using Φ , and U'_b, \ldots, Θ_b defined using $\tilde{\Phi}$. Hence

$$\begin{aligned} (\iota_a \circ \Pi_{U'_a})|_{X'_{ab}}^*(\Theta_{\Phi}) &= \Pi_{U'_a}|_{X'_{ab}}^*(\Theta_a) = \Pi_{U'_b}|_{X'_{ab}}^*(\Theta_b) \\ &= (\iota_b \circ \Pi_{U'_b})|_{X'_{ab}}^*(\Theta_{\Phi}) = (\iota_a \circ \Pi_{U'_a})|_{X'_{ab}}^*(\Theta_{\Phi}) \end{aligned}$$

using $\iota_a|_{X'_a}^*(\Theta_{\Phi}) = \Theta_a$ in the first step, (5.25) in the second, $\iota_b|_{X'_b}^*(\Theta_{\Phi}) = \Theta_b$ in the third, and $\iota_a \circ \Pi_{U'_a} = \iota_b \circ \Pi_{U'_b}$ in the fourth. As such $\iota_a \circ \Pi_{U'_a}|_{X'_{ab}} : X'_{ab} \to X$ form an étale open cover of X, this implies that $\Theta_{\Phi} = \Theta_{\Phi}$ by Theorem 2.7(i).

5.3. Theorem 5.4(c): composition of the Θ_{Φ} . Let $U, V, W, f, g, h, X, Y, Z, \Phi, \Psi$ be as in Theorem 5.4(c). Let $x \in X$, and set $y = \Phi(x) \in Y$. Apply Theorem 5.1(i) to $U, V, f, g, X, Y, \Phi, x$ to get \mathbb{C} -schemes U', V', a point $x' \in U'$, morphisms $\iota : U' \to U, j : V' \to V, \Phi' : U' \to V',$ $\alpha : V' \to U$ and $\beta : V' \to \mathbb{C}^m$ where $m = \dim V - \dim U$, and $f' := f \circ \iota : U' \to \mathbb{C}$, $g' := g \circ j : V' \to \mathbb{C}, X' := \operatorname{Crit}(f') \subseteq U', Y' := \operatorname{Crit}(g') \subseteq V'$, satisfying conditions including $\iota, j, \alpha \times \beta$ étale, (5.1) commutes, and $\iota(x') = x$.

Similarly, apply Theorem 5.1(i) to $V, W, g, h, Y, Z, \Psi, y$ to get \mathbb{C} -schemes \tilde{V}, \tilde{W} , a point $\tilde{y} \in \tilde{V}$, morphisms $\tilde{\iota} : \tilde{V} \to V, \tilde{\jmath} : \tilde{W} \to W, \tilde{\Psi} : \tilde{V} \to \tilde{W}, \tilde{\alpha} : \tilde{W} \to V$ and $\tilde{\beta} : \tilde{W} \to \mathbb{C}^n$ where $n = \dim W - \dim V$, and $\tilde{g} := g \circ \tilde{\iota} : \tilde{V} \to \mathbb{C}, \tilde{h} := h \circ \tilde{\jmath} : \tilde{W} \to \mathbb{C}, \tilde{Y} := \operatorname{Crit}(\tilde{g}) \subseteq \tilde{V},$ $\tilde{Z} := \operatorname{Crit}(\tilde{h}) \subseteq \tilde{W}$, satisfying conditions.

Define $\hat{U} = U' \times_{\Phi \circ \iota, V, \tilde{\iota}} \tilde{V}$ and $\hat{W} = V' \times_{j, V, \tilde{\alpha}} \tilde{W}$, with projections $\Pi_{U'} : \hat{U} \to U', \Pi_{\tilde{V}} : \hat{U} \to \tilde{V}$, $\Pi_{V'} : \hat{W} \to V', \Pi_{\tilde{W}} : \hat{W} \to \tilde{W}$. As $x' \in U'$ and $\tilde{y} \in \tilde{V}$ with $\Phi \circ \iota(x') = y = \tilde{\iota}(\tilde{y})$, there exists $\hat{x} \in \hat{U}$ with $\Pi_{U'}(\hat{x}) = x'$ and $\Pi_{\tilde{V}}(\hat{x}) = \tilde{y}$. Set $\hat{f} := f' \circ \Pi_{U'} : \hat{U} \to \mathbb{C}$ and $\hat{h} := \tilde{h} \circ \Pi_{\tilde{W}} : \hat{W} \to \mathbb{C}$, and $\hat{X} := \operatorname{Crit}(\hat{f}) \subseteq \hat{U}, \, \hat{Z} := \operatorname{Crit}(\hat{h}) \subseteq \hat{W}$. The morphisms $\Phi' \circ \Pi_{U'} : \hat{U} \to V', \, \tilde{\Psi} \circ \Pi_{\tilde{V}} : \hat{U} \to \tilde{W}$ satisfy

$$\jmath \circ (\Phi' \circ \Pi_{U'}) = \Phi \circ \iota \circ \Pi_{U'} = \tilde{\iota} \circ \Pi_{\tilde{V}} = \tilde{\alpha} \circ (\tilde{\Psi} \circ \Pi_{\tilde{V}}).$$

Hence there exists a unique morphism $\widehat{\Psi \circ \Phi} : \widehat{U} \to \widehat{W}$ such that $\Pi_{V'} \circ \widehat{\Psi \circ \Phi} = \Phi' \circ \Pi_{U'}$ and $\Pi_{\widetilde{W}} \circ \widehat{\Psi \circ \Phi} = \widetilde{\Psi} \circ \Pi_{\widetilde{V}}$. Then the following diagram

is the analogue of (5.1) for $U, W, f, h, X, Z, \Psi \circ \Phi, x$, and the conclusions of Theorem 5.1(i) hold. Thus (5.17) holds for Θ_{Φ} using $U', V', X', Y'\iota, j, \Phi', \alpha, \beta, m$, and for Θ_{Ψ} using $\tilde{V}, \tilde{W}, \tilde{Y}, \tilde{Z}, \tilde{\iota}, \tilde{j}, \tilde{\Psi}, \tilde{\alpha}, \tilde{\beta}, n$, and for $\Theta_{\Psi \circ \Phi}$ using $\hat{U}, \hat{W}, \hat{X}, \hat{Z}, \hat{\iota}, \hat{j}, \Psi \circ \Phi, \hat{\alpha}, \hat{\beta}, m + n$.

We have a commutative diagram in $Perv(\hat{X})$:

$$\begin{split} \hat{\iota}|_{\hat{X}}^{*} \circ (\Psi \circ \Phi)|_{X}^{*} (\mathcal{PV}_{W,h}) & \xrightarrow{\Psi \circ \Phi|_{\hat{X}}^{*} (\mathcal{PV}_{j \circ \mathcal{P}V_{a \times \hat{\beta}}^{-1})}} \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} (\mathcal{PV}_{U \times \mathbb{C}^{m+n}, f \boxplus \Sigma_{j} y_{j}^{2} + \Sigma_{j} z_{j}^{2})} \\ & \uparrow (\Psi' \circ \Pi_{\hat{V}})|_{\hat{X}}^{*} (\mathcal{PV}_{j \circ \mathcal{P}V_{a \times \hat{\beta}}^{-1})} & \uparrow (TS_{U,f,\mathbb{C}^{m+n},\Sigma_{j} y_{j}^{2} + \Sigma_{j} z_{j}^{2})} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} (\mathcal{PV}_{V \times \mathbb{C}^{n}, g \boxplus \Sigma_{j} z_{j}^{2})} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} (\mathcal{PV}_{U,f}^{\bullet}) \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} (TS_{V,g,\mathbb{C}^{n},\Sigma_{j} z_{j}^{2})} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} (\mathcal{PV}_{U,S,\mathbb{C}^{m+n},\Sigma_{j} y_{j}^{2} + \Sigma_{j} z_{j}^{2})} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} (TS_{V,g,\mathbb{C}^{n},\Sigma_{j} z_{j}^{2})} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} (TS_{U,S,\mathbb{C}^{m},\Sigma_{j} y_{j}^{2})} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ \Phi|_{X}^{*} \circ (\operatorname{id}_{Y} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ \Phi|_{X}^{*} (\mathcal{P}V_{V,g}^{*}) & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} & \hat{\iota}|_{\hat{X}}^{*} \circ (\operatorname{id}_{X} \times 0)^{*} \\ & \hat{\iota}|_{\hat{X}}^{*} \circ \Phi|_{X}^{*} \circ \Phi|_{X}^{*} \otimes \Phi|_{X}^{*$$

where the top right quadrilateral commutes because of associativity in the Thom–Sebastiani Theorem for $\mathcal{PV}_{V,f}^{\bullet}$, Theorem 2.13.

Also we have

which commutes because the trivializations of $\iota|_{X'}(P_{\Phi}), \tilde{\iota}|_{\tilde{X}}(P_{\Psi}), \hat{\iota}|_{\hat{X}}(P_{\Psi\circ\Phi})$ used to define $\delta, \tilde{\delta}, \hat{\delta}$ are compatible with $\Xi_{\Psi,\Phi}$.

Combining (5.26) and (5.27) with $\Pi_{U'}|_{\hat{X}}^*$ applied to (5.17) for Θ_{Φ} , and $\Pi_{\tilde{V}}|_{\hat{X}}^*$ applied to (5.17) for $\Theta_{\Psi,\Phi}$, and (5.17) for $\Theta_{\Psi\circ\Phi}$, we can show that the following diagram commutes in Perv(\hat{X}):

which is $\hat{\iota}|_{\hat{X}}^*$ applied to (5.16). Since such $\hat{\iota}|_{\hat{X}} : \hat{X} \to X$ form an étale cover of X, equation (5.16) commutes by Theorem 2.7(i). This proves Theorem 5.4(c).

5.4. \mathscr{D} -modules and mixed Hodge modules. By Theorem 5.1(iv),(v), the earlier parts of that result hold for our other contexts in §2.6–§2.10. Once again, the proofs of Theorem 5.4(a)–(c) then carry over to the other contexts using the general framework of §2.5, now also making use of property (vii).

6. Perverse sheaves on oriented D-critical loci

6.1. Background material on d-critical loci. Here are some of the main definitions and results on d-critical loci, from Joyce [23, Th.s 2.1, 2.20, 2.28 & Def.s 2.5, 2.18, 2.31]. For the algebraic case we work with C-schemes.

Theorem 6.1. Let X be a \mathbb{C} -scheme. Then there exists a sheaf S_X of \mathbb{C} -vector spaces on X, unique up to canonical isomorphism, which is uniquely characterized by the following two properties:

(i) Suppose $R \subseteq X$ is Zariski open, U is a smooth \mathbb{C} -scheme, and $i : R \hookrightarrow U$ is a closed embedding. Then we have an exact sequence of sheaves of \mathbb{C} -vector spaces on R:

$$0 \longrightarrow I_{R,U} \longrightarrow i^{-1}(\mathcal{O}_U) \xrightarrow{i^{\sharp}} \mathcal{O}_X|_R \longrightarrow 0,$$

where $\mathcal{O}_X, \mathcal{O}_U$ are the sheaves of regular functions on X, U, and i^{\sharp} is the morphism of sheaves of \mathbb{C} -algebras on R induced by i.

There is an exact sequence of sheaves of \mathbb{C} -vector spaces on R:

$$0 \longrightarrow \mathcal{S}_X|_R \xrightarrow{\iota_{R,U}} \frac{i^{-1}(\mathcal{O}_U)}{I_{R,U}^2} \xrightarrow{\mathrm{d}} \frac{i^{-1}(T^*U)}{I_{R,U} \cdot i^{-1}(T^*U)},$$

136

where d maps $f + I_{R,U}^2 \mapsto df + I_{R,U} \cdot i^{-1}(T^*U)$.

(ii) Let $R \subseteq S \subseteq X$ be Zariski open, U, V be smooth \mathbb{C} -schemes, $i : R \hookrightarrow U, j : S \hookrightarrow V$ closed embeddings, and $\Phi : U \to V$ a morphism with $\Phi \circ i = j|_R : R \to V$. Then the following diagram of sheaves on R commutes:

Here $\Phi: U \to V$ induces $\Phi^{\sharp}: \Phi^{-1}(\mathcal{O}_V) \to \mathcal{O}_U$ on U, so we have

$$i^{-1}(\Phi^{\sharp}): j^{-1}(\mathcal{O}_V)|_R = i^{-1} \circ \Phi^{-1}(\mathcal{O}_V) \longrightarrow i^{-1}(\mathcal{O}_U), \tag{6.2}$$

a morphism of sheaves of \mathbb{C} -algebras on R.

As $\Phi \circ i = j|_R$, equation (6.2) maps $I_{S,V}|_R \to I_{R,U}$, and so maps $I_{S,V}^2|_R \to I_{R,U}^2$. Thus (6.2) induces the morphism in the second column of (6.1).

Similarly, $d\Phi: \Phi^{-1}(T^*V) \to T^*U$ induces the third column of (6.1).

There is a natural decomposition $S_X = S_X^0 \oplus \mathbb{C}_X$, where \mathbb{C}_X is the constant sheaf on X with fibre \mathbb{C} , and $S_X^0 \subset S_X$ is the kernel of the composition

$$\mathcal{S}_X \xrightarrow{\beta_X} \mathcal{O}_X \xrightarrow{i_X^{\sharp}} \mathcal{O}_{X^{\mathrm{red}}}$$

with X^{red} the reduced \mathbb{C} -subscheme of X, and $i_X : X^{\mathrm{red}} \hookrightarrow X$ the inclusion.

The analogue of all the above also holds for complex analytic spaces.

Definition 6.2. An algebraic d-critical locus over \mathbb{C} is a pair (X, s), where X is a \mathbb{C} -scheme, and $s \in H^0(\mathcal{S}^0_X)$ for \mathcal{S}^0_X as in Theorem 6.1, satisfying the condition that for each $x \in X$, there exists a Zariski open neighbourhood R of x in X, a smooth \mathbb{C} -scheme U, a regular function $f: U \to \mathbb{A}^1 = \mathbb{C}$, and a closed embedding $i: R \to U$, such that $i(R) = \operatorname{Crit}(f)$ as \mathbb{C} -subschemes of U, and $\iota_{R,U}(s|_R) = i^{-1}(f) + I^2_{R,U}$.

Similarly, a complex analytic d-critical locus is a pair (X, s), where X is a complex analytic space, and $s \in H^0(\mathcal{S}^0_X)$ for \mathcal{S}_X as in Theorem 6.1, such that each $x \in X$ has an open neighbourhood $R \subset X$ with a closed embedding $i : R \hookrightarrow U$ into a complex manifold U and a holomorphic function $f : U \to \mathbb{C}$, such that $i(R) = \operatorname{Crit}(f)$, and $\iota_{R,U}(s|_R) = i^{-1}(f) + I^2_{R,U}$.

In both cases we call the quadruple (R, U, f, i) a critical chart on (X, s).

Let (X, s) be a d-critical locus (either algebraic or complex analytic), and (R, U, f, i) be a critical chart on (X, s). Let $U' \subseteq U$ be (Zariski) open, and set

$$R' = i^{-1}(U') \subseteq R, \quad i' = i|_{R'} : R' \hookrightarrow U', \quad \text{and} \quad f' = f|_{U'}.$$

Then (R', U', f', i') is also a critical chart on (X, s), and we call it a *subchart* of (R, U, f, i). As a shorthand we write $(R', U', f', i') \subseteq (R, U, f, i)$.

Let (R, U, f, i), (S, V, g, j) be critical charts on (X, s), with $R \subseteq S \subseteq X$. An *embedding* of (R, U, f, i) in (S, V, g, j) is a locally closed embedding $\Phi : U \hookrightarrow V$ such that $\Phi \circ i = j|_R$ and $f = g \circ \Phi$. As a shorthand we write $\Phi : (R, U, f, i) \hookrightarrow (S, V, g, j)$.

If $\Phi : (R, U, f, i) \hookrightarrow (S, V, g, j)$ and $\Psi : (S, V, g, j) \hookrightarrow (T, W, h, k)$ are embeddings, then $\Psi \circ \Phi : (R, U, f, i) \hookrightarrow (T, W, h, k)$ is also an embedding.

Theorem 6.3. Let (X, s) be a d-critical locus (either algebraic or complex analytic), and let (R, U, f, i), (S, V, g, j) be critical charts on (X, s).

Then, for each $x \in R \cap S \subseteq X$, there exist subcharts $(R', U', f', i') \subseteq (R, U, f, i)$, and $(S', V', g', j') \subseteq (S, V, g, j)$ with $x \in R' \cap S' \subseteq X$, a critical chart (T, W, h, k) on (X, s), and embeddings $\Phi : (R', U', f', i') \hookrightarrow (T, W, h, k)$, and $\Psi : (S', V', g', j') \hookrightarrow (T, W, h, k)$.

Theorem 6.4. Let (X, s) be a d-critical locus (either algebraic or complex analytic), and $X^{\text{red}} \subseteq X$ the associated reduced \mathbb{C} -scheme or reduced complex analytic space. Then there exists an (algebraic or holomorphic) line bundle $K_{X,s}$ on X^{red} which we call the **canonical bundle** of (X, s), which is natural up to canonical isomorphism, and is characterized by the following properties:

(i) If (R, U, f, i) is a critical chart on (X, s), there is a natural isomorphism

$$\iota_{R,U,f,i}: K_{X,s}|_{R^{\mathrm{red}}} \longrightarrow i^* (K_U^{\otimes^2})|_{R^{\mathrm{red}}}, \tag{6.3}$$

where $K_U = \Lambda^{\dim U} T^* U$ is the canonical bundle of U in the usual sense.

(ii) Let $\Phi : (R, U, f, i) \hookrightarrow (S, V, g, j)$ be an embedding of critical charts on (X, s). Then (5.9) defines an isomorphism of line bundles on $\operatorname{Crit}(f)^{\operatorname{red}}$:

$$J_{\Phi}: K_U^{\otimes^2}|_{\operatorname{Crit}(f)^{\operatorname{red}}} \xrightarrow{\cong} \Phi|_{\operatorname{Crit}(f)^{\operatorname{red}}}^* \left(K_V^{\otimes^2}\right).$$

Since $i: R \to \operatorname{Crit}(f)$ is an isomorphism with $\Phi \circ i = j|_R$, this gives

$$i|_{R^{\mathrm{red}}}^*(J_{\Phi}):i|_{R^{\mathrm{red}}}^*(K_U^{\otimes^2}) \xrightarrow{\cong} j|_{R^{\mathrm{red}}}^*(K_V^{\otimes^2}).$$

and we must have

$$\iota_{S,V,g,j}|_{R^{\mathrm{red}}} = i|_{R^{\mathrm{red}}}^*(J_{\Phi}) \circ \iota_{R,U,f,i} : K_{X,s}|_{R^{\mathrm{red}}} \longrightarrow j^*(K_V^{\otimes^2})|_{R^{\mathrm{red}}}.$$
(6.4)

Definition 6.5. Let (X, s) be a d-critical locus (either algebraic or complex analytic), and $K_{X,s}$ its canonical bundle from Theorem 6.4. An *orientation* on (X, s) is a choice of square root line bundle $K_{X,s}^{1/2}$ for $K_{X,s}$ on X^{red} . That is, an orientation is an (algebraic or holomorphic) line bundle L on X^{red} , together with an isomorphism $L^{\otimes^2} = L \otimes L \cong K_{X,s}$. A d-critical locus with an orientation will be called an *oriented d-critical locus*.

In [9, Th. 6.6] we show that algebraic d-critical loci are classical truncations of objects in derived algebraic geometry known as -1-shifted symplectic derived schemes, introduced by Pantev, Toën, Vaquié and Vezzosi [42].

Theorem 6.6 (Bussi, Brav and Joyce [9]). Suppose (\mathbf{X}, ω) is a -1-shifted symplectic derived scheme in the sense of Pantev et al. [42] over \mathbb{C} , and let $X = t_0(\mathbf{X})$ be the associated classical \mathbb{C} -scheme of \mathbf{X} . Then X extends naturally to an algebraic d-critical locus (X, s). The canonical bundle $K_{X,s}$ from Theorem 6.4 is naturally isomorphic to the determinant line bundle $\det(\mathbb{L}_{\mathbf{X}})|_{X^{red}}$ of the cotangent complex $\mathbb{L}_{\mathbf{X}}$ of \mathbf{X} .

Now Pantev et al. [42] show that derived moduli schemes of coherent sheaves, or complexes of coherent sheaves, on a Calabi–Yau 3-fold Y have -1-shifted symplectic structures. Using this, in [9, Cor. 6.7] we deduce:

Corollary 6.7. Suppose Y is a Calabi–Yau 3-fold over \mathbb{C} , and \mathcal{M} is a classical moduli \mathbb{C} -scheme of simple coherent sheaves in $\operatorname{coh}(Y)$, or simple complexes of coherent sheaves in $D^b \operatorname{coh}(Y)$, with (symmetric) obstruction theory $\phi : \mathcal{E}^{\bullet} \to \mathbb{L}_{\mathcal{M}}$ as in Behrend [2], Thomas [52], or Huybrechts and Thomas [22]. Then \mathcal{M} extends naturally to an algebraic d-critical locus (\mathcal{M}, s) . The canonical bundle $K_{\mathcal{M},s}$ from Theorem 6.4 is naturally isomorphic to $\det(\mathcal{E}^{\bullet})|_{\mathcal{M}^{red}}$.
Here we call $F \in \operatorname{coh}(Y)$ simple if $\operatorname{Hom}(F, F) = \mathbb{C}$, and we call $F^{\bullet} \in D^b \operatorname{coh}(Y)$ simple if $\operatorname{Hom}(F^{\bullet}, F^{\bullet}) = \mathbb{C}$ and $\operatorname{Ext}^{<0}(F^{\bullet}, F^{\bullet}) = 0$. Thus, d-critical loci will have applications in Donaldson-Thomas theory for Calabi-Yau 3-folds [24, 32, 33, 52]. Orientations on (\mathcal{M}, s) are closely related to orientation data in the work of Kontsevich and Soibelman [32, 33].

Pantev et al. [42] also show that derived intersections $L \cap M$ of algebraic Lagrangians L, M in an algebraic symplectic manifold (S, ω) have -1-shifted symplectic structures, so that Theorem 6.6 gives them the structure of algebraic d-critical loci. Bussi [10, §3] will prove a complex analytic version of this:

Theorem 6.8 (Bussi [10]). Suppose (S, ω) is a complex symplectic manifold, and L, M are complex Lagrangian submanifolds in S. Then the intersection $X = L \cap M$, as a complex analytic subspace of S, extends naturally to a complex analytic d-critical locus (X, s). The canonical bundle $K_{X,s}$ from Theorem 6.4 is naturally isomorphic to $K_L|_{X^{red}} \otimes K_M|_{X^{red}}$.

6.2. The main result, and applications. Here is our main result, which will be proved in $\frac{6.3-6.4}{10}$.

Theorem 6.9. Let (X, s) be an oriented algebraic d-critical locus over \mathbb{C} , with orientation $K_{X,s}^{1/2}$. Then for any well-behaved base ring A, such as \mathbb{Z}, \mathbb{Q} or \mathbb{C} , there exists a perverse sheaf $P_{X,s}^{\bullet}$ in Perv(X) over A, which is natural up to canonical isomorphism, and Verdier duality and monodromy isomorphisms

$$\Sigma_{X,s}: P^{\bullet}_{X,s} \longrightarrow \mathbb{D}_X (P^{\bullet}_{X,s}), \qquad \mathcal{T}_{X,s}: P^{\bullet}_{X,s} \longrightarrow P^{\bullet}_{X,s}, \tag{6.5}$$

which are characterized by the following properties:

(i) If (R, U, f, i) is a critical chart on (X, s), there is a natural isomorphism

$$\omega_{R,U,f,i}: P_{X,s}^{\bullet}|_R \longrightarrow i^* \left(\mathcal{PV}_{U,f}^{\bullet} \right) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}, \tag{6.6}$$

where $\pi_{R,U,f,i}: Q_{R,U,f,i} \to R$ is the principal $\mathbb{Z}/2\mathbb{Z}$ -bundle parametrizing local isomorphisms $\alpha: K_{X,s}^{1/2} \to i^*(K_U)|_{R^{red}}$ with $\alpha \otimes \alpha = \iota_{R,U,f,i}$, for $\iota_{R,U,f,i}$ as in (6.3). Furthermore the following commute in Perv(R):

$$P_{X,s}^{\bullet}|_{R} \xrightarrow{\omega_{R,U,f,i}} i^{*}(\mathcal{PV}_{U,f}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}$$

$$\downarrow^{\Sigma_{X,s}|_{R}} \xrightarrow{i^{*}(\sigma_{U,f}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}} \downarrow$$

$$\mathbb{D}_{R}(P_{X,s}^{\bullet}|_{R}) \xleftarrow{\mathbb{D}_{R}(\omega_{R,U,f,i})} \stackrel{i^{*}(\mathbb{D}_{\mathrm{Crit}(f)}(\mathcal{PV}_{U,f}^{\bullet})) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}}{i^{*}(\mathcal{PV}_{U,f}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}},$$

$$P_{X,s}^{\bullet}|_{R} \xrightarrow{\omega_{R,U,f,i}} \stackrel{i^{*}(\mathcal{PV}_{U,f}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}}{i^{*}(\tau_{U,f}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}},$$

$$P_{X,s}^{\bullet}|_{R} \xrightarrow{\omega_{R,U,f,i}} \stackrel{i^{*}(\mathcal{PV}_{U,f}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}}{i^{*}(\tau_{U,f}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}},$$

$$(6.8)$$

(ii) Let $\Phi : (R, U, f, i) \hookrightarrow (S, V, g, j)$ be an embedding of critical charts on (X, s). Then there is a natural isomorphism of principal $\mathbb{Z}/2\mathbb{Z}$ -bundles

$$\Lambda_{\Phi}: Q_{S,V,g,j}|_R \xrightarrow{\cong} i^*(P_{\Phi}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}$$

$$(6.9)$$

on R, for P_{Φ} as in Definition 5.2, defined as follows: local isomorphisms

$$\begin{aligned} \alpha : K_{X,s}^{1/2}|_{R^{\mathrm{red}}} &\longrightarrow i^*(K_U)|_{R^{\mathrm{red}}}, \quad \beta : K_{X,s}^{1/2}|_{R^{\mathrm{red}}} \longrightarrow j^*(K_V)|_{R^{\mathrm{red}}}, \\ and & \gamma : i^*(K_U)|_{R^{\mathrm{red}}} \longrightarrow j^*(K_V)|_{R^{\mathrm{red}}} \end{aligned}$$

with $\alpha \otimes \alpha = \iota_{R,U,f,i}, \ \beta \otimes \beta = \iota_{S,V,g,j}|_{R^{red}}, \ \gamma \otimes \gamma = i|_{R^{red}}^*(J_{\Phi})$ correspond to local sections $s_{\alpha} : R \to Q_{R,U,f,i}, s_{\beta} : R \to Q_{S,V,g,j}|_R, s_{\gamma} : R \to i^*(P_{\Phi}).$ Equation (6.4) shows that $\beta = \gamma \circ \alpha$ is a possible solution for β , and we define Λ_{Φ} in (6.9) such that $\Lambda_{\Phi}(s_{\beta}) = s_{\gamma} \otimes_{\mathbb{Z}/2\mathbb{Z}} s_{\alpha} \text{ if and only if } \beta = \gamma \circ \alpha.$

Then the following diagram commutes in Perv(R), for Θ_{Φ} as in (5.13):

$$P_{X,s}^{\bullet}|_{R} \xrightarrow{\omega_{R,U,f,i}} i^{*}(\mathcal{PV}_{U,f}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}$$

$$\downarrow^{\omega_{S,V,g,j}|_{R}} \xrightarrow{i^{*}(\Theta_{\Phi}) \otimes \operatorname{id}_{Q_{R,U,f,i}}} \downarrow^{*}(\Theta_{\Phi}) \otimes \operatorname{id}_{Q_{R,U,f,i}} \downarrow$$

$$j^{*}(\mathcal{PV}_{V,g}^{\bullet})|_{R} \xrightarrow{\operatorname{id}_{j^{*}(\mathcal{PV}_{V,g}^{\bullet}) \otimes \Lambda_{\Phi}}} i^{*}(\Phi^{*}(\mathcal{PV}_{V,g}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi})$$

$$\otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{S,V,g,j}|_{R} \xrightarrow{i^{*}(\Phi^{*}(\mathcal{PV}_{V,g}) \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi})} \otimes_{\mathbb{Z}/2\mathbb{Z}} P_{\Phi}$$

$$\otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R,U,f,i}.$$

$$(6.10)$$

The analogues of all the above also hold for \mathscr{D} -modules on oriented algebraic d-critical loci over \mathbb{C} , for perverse sheaves and \mathscr{D} -modules on oriented complex analytic d-critical loci, and for mixed Hodge modules on oriented algebraic d-critical loci over $\mathbb C$ and oriented complex analytic d-critical loci, as in §2.6–§2.10.

Remark 6.10. This sheaf-theoretic result is compatible with the motivic result of Bussi, Joyce and Meinhardt in [11]. Given (X, s) an oriented algebraic d-critical locus over \mathbb{C} , [11] proves the existence of a natural motivic element $MF_{X,s} \in \overline{\mathcal{M}}_X^{\mu}$ in a version of the relative Grothendieck ring of varieties over X, equivariant with respect to suitable actions of the group $\hat{\mu}$ of all roots of unity (for detailed definitions, see [11]). Since the mixed Hodge module realization factorizes over the additional relation one has to impose in [11] on the Grothendieck group, the ring \mathcal{M}^{μ}_{X} has a map to $K_0(MHM(X;T_s))$, the K-group of algebraic mixed Hodge modules on X with a finite order automorphism (note that the Grothendieck group only sees the semisimple part T_s of the monodromy and not the nilpotent part N). By a Cech-type argument using the corresponding comparison result of [20, Prop. 3.17], the image of $MF_{X,s}$ in $K_0(\text{MHM}(X;T_s))$ agrees with the image of the mixed Hodge module realization of $P^{\bullet}_{X,s}$, since both sides are Zariski locally modelled by the same vanishing cycles. Thus, for example, they give the same weight polynomial for global cohomology with compact support.

From Theorem 6.6, Corollary 6.7 and Theorem 6.8 we deduce:

Corollary 6.11. Let (\mathbf{X}, ω) be a -1-shifted symplectic derived scheme over \mathbb{C} in the sense of Pantev et al. [42], and $X = t_0(\mathbf{X})$ the associated classical \mathbb{C} -scheme. Suppose we are given a square root det $(\mathbb{L}_{\mathbf{X}})|_X^{1/2}$ for det $(\mathbb{L}_{\mathbf{X}})|_X$. Then we may define $P^{\bullet}_{\mathbf{X},\omega} \in \operatorname{Perv}(X)$, uniquely up to canonical isomorphism, and isomorphisms $\Sigma_{\mathbf{X},\omega} : P^{\bullet}_{\mathbf{X},\omega} \to \mathbb{D}_X(P^{\bullet}_{\mathbf{X},\omega}), \ \mathrm{T}_{\mathbf{X},\omega} : P^{\bullet}_{\mathbf{X},\omega} \to P^{\bullet}_{\mathbf{X},\omega}$. The same applies for \mathscr{D} -modules and mixed Hodge modules on X.

Corollary 6.12. Let Y be a Calabi–Yau 3-fold over \mathbb{C} , and \mathcal{M} a classical moduli \mathbb{C} -scheme of simple coherent sheaves in coh(Y), or simple complexes of coherent sheaves in $D^b coh(Y)$, with natural (symmetric) obstruction theory $\phi : \mathcal{E}^{\bullet} \to \mathbb{L}_{\mathcal{M}}$ as in Behrend [2], Thomas [52], or Huybrechts and Thomas [22]. Suppose we are given a square root $\det(\mathcal{E}^{\bullet})^{1/2}$ for $\det(\mathcal{E}^{\bullet})$. Then we may define $P^{\bullet}_{\mathcal{M}} \in \operatorname{Perv}(\mathcal{M})$, uniquely up to canonical isomorphism, and isomorphisms $\Sigma_{\mathcal{M}}: P^{\bullet}_{\mathcal{M}} \to \mathbb{D}_{\mathcal{M}}(P^{\bullet}_{\mathcal{M}}), \ \mathbf{T}_{\mathcal{M}}: P^{\bullet}_{\mathcal{M}} \to P^{\bullet}_{\mathcal{M}}.$ The same applies for \mathscr{D} -modules and mixed Hodge modules on \mathcal{M} .

Corollary 6.13. Let (S, ω) be a complex symplectic manifold and L, M complex Lagrangian submanifolds in S, and write $X = L \cap M$, as a complex analytic subspace of S. Suppose we are given square roots $K_L^{1/2}, K_M^{1/2}$ for K_L, K_M . Then we may define $P_{L,M}^{\bullet} \in \text{Perv}(X)$, uniquely up to canonical isomorphism, and isomorphisms $\Sigma_{L,M} : P^{\bullet}_{L,M} \to \mathbb{D}_X(P^{\bullet}_{L,M})$, and $T_{L,M} : P^{\bullet}_{L,M} \to P^{\bullet}_{L,M}$.

The same applies for \mathcal{D} -modules and mixed Hodge modules on X.

The next two remarks discuss applications of Corollaries 6.12 and 6.13 to Donaldson–Thomas theory, and to Lagrangian Floer cohomology.

Remark 6.14. If Y is a Calabi–Yau 3-fold over \mathbb{C} and τ a suitable stability condition on coherent sheaves on Y, the *Donaldson–Thomas invariants* $DT^{\alpha}(\tau)$ are integers which 'count' the moduli schemes $\mathcal{M}_{st}^{\alpha}(\tau)$ of τ -stable coherent sheaves on Y with Chern character $\alpha \in H^{even}(Y; \mathbb{Q})$, provided there are no strictly τ -semistable sheaves in class α on Y. They were defined by Thomas [52], who showed they are unchanged under deformations of Y, following a suggestion of Donaldson and Thomas [16].

Behrend [2] showed that $DT^{\alpha}(\tau)$ may be written as a weighted Euler characteristic $\chi(\mathcal{M}_{st}^{\alpha}(\tau),\nu)$, where $\nu: \mathcal{M}_{st}^{\alpha}(\tau) \to \mathbb{Z}$ is a certain constructible function called the *Behrend func*tion. Joyce and Song [24] extended the definition of $DT^{\alpha}(\tau)$ to classes α including τ -semistable sheaves (with $DT^{\alpha}(\tau) \in \mathbb{Q}$), and proved a wall-crossing formula for $DT^{\alpha}(\tau)$ under change of stability condition τ . Kontsevich and Soibelman [32] gave a (partly conjectural) motivic generalization of Donaldson–Thomas invariants, also with a wall-crossing formula.

Corollary 6.12 is relevant to the *categorification* of Donaldson–Thomas theory. As in [2, §1.2], the perverse sheaf $P^{\bullet}_{\mathcal{M}^{\alpha}_{st}(\tau)}$ has pointwise Euler characteristic $\chi(P^{\bullet}_{\mathcal{M}^{\alpha}_{st}(\tau)}) = \nu$. This implies that when A is a field, say $A = \mathbb{Q}$, the (compactly-supported) hypercohomologies $\mathbb{H}^*(P^{\bullet}_{\mathcal{M}^{\alpha}_{st}(\tau)}), \mathbb{H}^*_{c}(P^{\bullet}_{\mathcal{M}^{\alpha}_{st}(\tau)})$ from (2.1) satisfy

$$\sum_{k \in \mathbb{Z}} (-1)^k \dim \mathbb{H}^k \left(P^{\bullet}_{\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)} \right) = \sum_{k \in \mathbb{Z}} (-1)^k \dim \mathbb{H}^k_{\mathrm{c}} \left(P^{\bullet}_{\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)} \right) = \chi \left(\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau), \nu \right) = DT^{\alpha}(\tau),$$

where $\mathbb{H}^{k}(P^{\bullet}_{\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)}) \cong \mathbb{H}^{-k}_{c}(P^{\bullet}_{\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)})^{*}$ by Verdier duality. That is, we have produced a natural graded Q-vector space $\mathbb{H}^{*}(P^{\bullet}_{\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)})$, thought of as some kind of generalized cohomology of $\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)$, whose graded dimension is $DT^{\alpha}(\tau)$. This gives a new interpretation of the Donaldson-Thomas invariant $DT^{\alpha}(\tau)$.

In fact, as discussed at length in [51, §3], the first natural "refinement" or "quantization" direction of a Donaldson–Thomas invariant $DT^{\alpha}(\tau) \in \mathbb{Z}$ is not the Poincaré polynomial of this cohomology, but its weight polynomial

$$w\big(\mathbb{H}^*(P^{\bullet}_{\mathcal{M}^{\alpha}_{+}(\tau)}),t\big) \in \mathbb{Z}\big[t^{\pm\frac{1}{2}}\big],$$

defined using the mixed Hodge structure on the cohomology of the mixed Hodge module version of $P^{\bullet}_{\mathcal{M}^{\alpha}(\tau)}$ (which exists assuming that $\mathcal{M}^{\alpha}_{\mathrm{st}}(\tau)$ is projective, for example, see Remark 2.22).

The material above is related to work by other authors. The idea of categorifying Donaldson– Thomas invariants using perverse sheaves or \mathscr{D} -modules is probably first due to Behrend [2], and for Hilbert schemes Hilbⁿ(Y) of a Calabi–Yau 3-fold Y is discussed by Dimca and Szendrői [15] and Behrend, Bryan and Szendrői [3, §3.4], using mixed Hodge modules. Corollary 6.12 answers a question of Joyce and Song [24, Question 5.7(a)].

As in [24, 32] representations of quivers with superpotentials (Q, W) give 3-Calabi–Yau triangulated categories, and one can define Donaldson–Thomas type invariants $DT^{\alpha}_{Q,W}(\tau)$ 'counting' such representations, which are simple algebraic 'toy models' for Donaldson–Thomas invariants of Calabi–Yau 3-folds. Kontsevich and Soibelman [33] explain how to categorify these quiver invariants $DT^{\alpha}_{Q,W}(\tau)$, and define an associative multiplication on the categorification to make a *Cohomological Hall Algebra*. This paper was motivated by the aim of extending [33] to define Cohomological Hall Algebras for Calabi–Yau 3-folds. The square root det $(\mathcal{E}^{\bullet})^{1/2}$ required in Corollary 6.12 corresponds roughly to *orientation data* in the work of Kontsevich and Soibelman [32, §5], [33].

In a paper written independently of our programme [9,11,23], Kiem and Li [31] have recently proved an analogue of Corollary 6.12 by complex analytic methods, beginning from Joyce and Song's result [24, Th. 5.4], proved using gauge theory, that $\mathcal{M}_{st}^{\alpha}(\tau)$ is locally isomorphic to $\operatorname{Crit}(f)$ as a complex analytic space, for V a complex manifold and $f: V \to \mathbb{C}$ holomorphic.

Remark 6.15. In the situation of Corollary 6.13, with $\dim_{\mathbb{C}} S = 2n$, we claim that there ought morally to be some kind of approximate comparison

$$\mathbb{H}^{k}(P_{L,M}^{\bullet}) \approx HF^{k+n}(L,M), \tag{6.11}$$

where $HF^*(L, M)$ is the Lagrangian Floer cohomology of Fukaya, Oh, Ohta and Ono [18]. We can compare and contrast the two sides of (6.11) as follows:

- (a) $\mathbb{H}^*(P^{\bullet}_{L,M})$ is defined over any well-behaved base ring A, e.g. $A = \mathbb{Z}$ or \mathbb{Q} , but $HF^*(L, M)$ is defined over a Novikov ring of power series Λ_{nov} .
- (b) $\mathbb{H}^*(P_{L,M}^{\bullet})$ has extra structure not visible in $HF^*(L, M)$, from Verdier duality and monodromy operators $\Sigma_{L,M}$, $T_{L,M}$, plus the mixed Hodge module version has a mixed Hodge structure.
- (c) $\mathbb{H}^*(P^{\bullet}_{L,M})$ is defined for arbitrary complex Lagrangians L, M, not necessarily compact or closed in S, but $HF^*(L, M)$ is only defined for L, M compact, or at least for L, Mclosed and well-behaved at infinity.
- (d) To define $HF^*(L, M)$ one generally assumes L, M intersect transversely, or at least cleanly. But $\mathbb{H}^*(P_{L,M}^{\bullet})$ is defined when $L \cap M$ is arbitrarily singular, and the construction is only really interesting for singular $L \cap M$.
- (e) To define $HF^*(L, M)$ we need L, M to be oriented and spin, to orient moduli spaces of *J*-holomorphic curves. When L, M are complex Lagrangians they are automatically oriented, and spin structures on L, M correspond to choices of square roots $K_L^{1/2}, K_M^{1/2}$, as used in Corollary 6.13.

Some of the authors are working on defining a 'Fukaya category' of complex Lagrangians in a complex symplectic manifold, using $\mathbb{H}^*(P_{L,M}^{\bullet})$ as morphisms.

We now discuss related work. Nadler and Zaslow [40, 41] show that if X is a real analytic manifold (for instance, a complex manifold), then the derived category $D_c^b(X)$ of constructible sheaves on X is equivalent to a certain derived Fukaya category $D^b \mathcal{F}(T^*X)$ of exact Lagrangians in T^*X .

Let L, M be complex Lagrangians in a complex symplectic manifold (S, ω) . Regarding $\mathcal{O}_L, \mathcal{O}_M$ as coherent sheaves on S, Behrend and Fantechi [4, Th.s 4.3 & 5.2] claim to construct canonical \mathbb{C} -linear (not \mathcal{O}_S -linear) differentials

$$d: \mathcal{E}xt^{i}_{\mathcal{O}_{S}}(\mathcal{O}_{L}, \mathcal{O}_{M}) \longrightarrow \mathcal{E}xt^{i+1}_{\mathcal{O}_{S}}(\mathcal{O}_{L}, \mathcal{O}_{M})$$

with $d^2 = 0$, such that $(\mathcal{E}xt^*_{\mathcal{O}_S}(\mathcal{O}_L, \mathcal{O}_M), d)$ is a constructible complex. There is a mistake in the proof of [4, Th. 4.3]. To fix this one should instead work with $\mathcal{E}xt^*_{\mathcal{O}_S}(K_L^{1/2}, K_M^{1/2})$ for square roots $K_L^{1/2}, K_M^{1/2}$ as in Corollary 6.13. Also the proof of the constructibility of $(\mathcal{E}xt^*_{\mathcal{O}_S}(K_L^{1/2}, K_M^{1/2}), d)$ in [4, Th. 5.2] depended on a result of Kapranov, which later turned out to be false.

Our $P_{L,M}^{\bullet}$ over $A = \mathbb{C}$ should be the natural perverse sheaf on $L \cap M$ conjectured by Behrend and Fantechi [4, Conj. 5.16], who also suggest there should be a spectral sequence from $(\mathcal{E}xt^*_{\mathcal{O}_S}(K_L^{1/2}, K_M^{1/2}), \mathbf{d})[n]$ to $P_{L,M}^{\bullet}$. (See Sabbah [44, Th. 1.1] for a related result.) In [4, §5.3], Behrend and Fantechi discuss how to define a 'Fukaya category' using their ideas. Kashiwara and Schapira [29] develop a theory of deformation quantization modules, or DQmodules, on a complex symplectic manifold (S, ω) , which roughly may be regarded as symplectic versions of \mathscr{D} -modules. Holonomic DQ-modules \mathcal{D}^{\bullet} are supported on (possibly singular) complex Lagrangians L in S. If L is a smooth, closed, complex Lagrangian in S and $K_L^{1/2}$ a square root of K_L , D'Agnolo and Schapira [13] show that there exists a simple holonomic DQ-module \mathcal{D}^{\bullet} supported on L.

If $\mathcal{D}^{\bullet}, \mathcal{E}^{\bullet}$ are simple holonomic DQ-modules on S supported on smooth Lagrangians L, M, then Kashiwara and Schapira [28] show that $\mathcal{RHom}(\mathcal{D}^{\bullet}, \mathcal{E}^{\bullet})[n]$ is a perverse sheaf on S over the field $\mathbb{C}((\hbar))$, supported on $X = L \cap M$. Pierre Schapira explained to the authors how to prove that $\mathcal{RHom}(\mathcal{D}^{\bullet}, \mathcal{E}^{\bullet})[n] \cong P^{\bullet}_{L,M}$, when $P^{\bullet}_{L,M}$ is defined over the base ring $A = \mathbb{C}((\hbar))$.

Now let L, M, N be Lagrangians in S, with square roots $K_L^{1/2}, K_M^{1/2}, K_N^{1/2}$. We have a product $HF^k(L, M) \times HF^l(M, N) \to HF^{k+l}(L, N)$ from composition of morphisms in $D^b\mathcal{F}(S)$. So (6.11) suggests there should be a product

$$\mathbb{H}^{k}(P_{L,M}^{\bullet}) \times \mathbb{H}^{l}(P_{M,N}^{\bullet}) \longrightarrow \mathbb{H}^{k+l+n}(P_{L,N}^{\bullet}), \tag{6.12}$$

which would naturally be induced by a morphism in $D_c^b(S)$

$$\mu_{L,M,N}: P^{\bullet}_{L,M} \overset{L}{\otimes} P^{\bullet}_{M,N} \longrightarrow P^{\bullet}_{L,N}[n].$$
(6.13)

Observe that the work of Behrend–Fantechi and Kashiwara–Schapira cited above supports the existence of (6.12)–(6.13): there are natural products

$$\begin{split} \mathcal{E}xt^k_{\mathcal{O}_S}(K_L^{1/2},K_M^{1/2}) \otimes_{\mathcal{O}_S} \mathcal{E}xt^l_{\mathcal{O}_S}(K_M^{1/2},K_N^{1/2}) &\longrightarrow \mathcal{E}xt^{k+l}_{\mathcal{O}_S}(K_L^{1/2},K_N^{1/2}) \\ R\mathscr{H}om(\mathcal{D}^{\bullet},\mathcal{E}^{\bullet}) \overset{L}{\otimes} R\mathscr{H}om(\mathcal{E}^{\bullet},\mathcal{F}^{\bullet}) &\longrightarrow R\mathscr{H}om(\mathcal{D}^{\bullet},\mathcal{F}^{\bullet}). \end{split}$$

But since (6.13) is a morphism of complexes, not of perverse sheaves, Theorem 2.7(i) does not apply, so we cannot construct $\mu_{L,M,N}$ by naïvely gluing data on an open cover, as we have been doing in §3–§6.

6.3. **Proof of Theorem 6.9 for** \mathbb{C} -schemes. Let (X, s) be an oriented algebraic d-critical locus over \mathbb{C} , with orientation $K_{X,s}^{1/2}$. By Definition 6.2 we may choose a family

$$\{(R_a, U_a, f_a, i_a) : a \in A\}$$

of critical charts (R_a, U_a, f_a, i_a) on (X, s) such that $\{R_a : a \in A\}$ is a Zariski open cover of the \mathbb{C} -scheme X. Then for each $a \in A$ we have a perverse sheaf

$$i_a^* \left(\mathcal{PV}_{U_a, f_a}^{\bullet} \right) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R_a, U_a, f_a, i_a} \in \operatorname{Perv}(R_a), \tag{6.14}$$

for Q_{R_a,U_a,f_a,i_a} as in Theorem 6.9(i). The idea of the proof is to use Theorem 2.7(ii) to glue the perverse sheaves (6.14) on the Zariski open cover $\{R_a : a \in A\}$ to get a global perverse sheaf $P_{X,s}^{\bullet}$ on X. Note that Theorem 2.7(ii) is written for étale open covers, but this immediately implies the simpler Zariski version.

To do this, for all $a, b \in A$ we have to construct isomorphisms

$$\alpha_{ab} : \begin{bmatrix} i_a^* (\mathcal{PV}_{U_a, f_a}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R_a, U_a, f_a, i_a} \end{bmatrix} \Big|_{R_a \cap R_b} \longrightarrow \\ \begin{bmatrix} i_b^* (\mathcal{PV}_{U_b, f_b}^{\bullet}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R_b, U_b, f_b, i_b} \end{bmatrix} \Big|_{R_a \cap R_b} \in \operatorname{Perv}(R_a \cap R_b),$$
(6.15)

satisfying $\alpha_{aa} = \text{id for all } a \in A$ and

$$\alpha_{bc}|_{R_a \cap R_b \cap R_c} \circ \alpha_{ab}|_{R_a \cap R_b \cap R_c} = \alpha_{ac}|_{R_a \cap R_b \cap R_c} \quad \text{for all } a, b, c \in A.$$
(6.16)

Fix $a, b \in A$. By applying Theorem 6.3 to the critical charts (R_a, U_a, f_a, i_a) , (R_b, U_b, f_b, i_b) at each $x \in R_a \cap R_b$, we can choose an indexing set D_{ab} and for each $d \in D_{ab}$ subcharts $(R_a'^d, U_a'^d, f_a'^d, i_a'^d) \subseteq (R_a, U_a, f_a, i_a)$ and $(R_b'^d, U_b'^d, f_b'^d, i_b'^d) \subseteq (R_b, U_b, f_b, i_b)$, a critical chart (S^d, V^d, g^d, j^d) on (X, s), and embeddings

$$\Phi^{d}: (R_{a}^{\prime d}, U_{a}^{\prime d}, f_{a}^{\prime d}, i_{a}^{\prime d}) \hookrightarrow (S^{d}, V^{d}, g^{d}, j^{d}) \quad \text{and} \quad \Psi^{d}: (R_{b}^{\prime d}, U_{b}^{\prime d}, f_{b}^{\prime d}, i_{b}^{\prime d}) \hookrightarrow (S^{d}, V^{d}, g^{d}, j^{d})$$

such that $\{R_a^{\prime d} \cap R_b^{\prime d} : d \in D_{ab}\}$ is a Zariski open cover of $R_a \cap R_b$. For each $d \in D_{ab}$, define an isomorphism

$$\alpha_{ab}^{d}: \left[i_{a}^{*}\left(\mathcal{PV}_{U_{a},f_{a}}^{\bullet}\right)\otimes_{\mathbb{Z}/2\mathbb{Z}}Q_{R_{a},U_{a},f_{a},i_{a}}\right]\Big|_{R_{a}^{\prime d}\cap R_{b}^{\prime d}} \longrightarrow \left[i_{b}^{*}\left(\mathcal{PV}_{U_{b},f_{b}}^{\bullet}\right)\otimes_{\mathbb{Z}/2\mathbb{Z}}Q_{R_{b},U_{b},f_{b},i_{b}}\right]\Big|_{R_{a}^{\prime d}\cap R_{b}^{\prime d}}$$

by the commutative diagram



where $\Theta_{\Phi^d}, \Theta_{\Psi^d}$ are as in Theorem 5.4, and $\Lambda_{\Phi^d}, \Lambda_{\Psi^d}$ as in (6.9).

We claim that for all $d, e \in D_{ab}$ we have

$$\alpha_{ab}^{d}|_{R_{a}^{\prime d} \cap R_{b}^{\prime d} \cap R_{a}^{\prime e} \cap R_{b}^{\prime e}} = \alpha_{ab}^{e}|_{R_{a}^{\prime d} \cap R_{b}^{\prime d} \cap R_{a}^{\prime e} \cap R_{b}^{\prime e}}.$$
(6.18)

To see this, let $x \in R'^d_a \cap R'^d_b \cap R'^e_a \cap R'^e_b$, and apply Theorem 6.3 to the critical charts $(S^d, V^d, g^d, j^d), (S^e, V^e, g^e, j^e)$ and point $x \in S^d \cap S^e$. This gives subcharts

$$(S'^{d}, V'^{d}, g'^{d}, j'^{d}) \subseteq (S^{d}, V^{d}, g^{d}, j^{d}) \quad \text{ and } \quad (S'^{e}, V'^{e}, g'^{e}, j'^{e}) \subseteq (S^{e}, V^{e}, g^{e}, j^{e})$$

with $x \in S'^d \cap S'^e$, a critical chart (T, W, h, k) on (X, s), and embeddings

$$\Omega: (S'^d, V'^d, g'^d, j'^d) \hookrightarrow (T, W, h, k) \qquad \text{and} \qquad \Upsilon: (S'^e, V'^e, g'^e, j'^e) \hookrightarrow (T, W, h, k) \in \mathbb{C}$$



Here we have given two expressions for the top left diagonal morphism in (6.19). To see these are equal, set $R'^{de}_a = R'^d_a \cap R'^e_a \cap S'^d \cap S'^e$, $U'^{de}_a = (\Phi^d)^{-1}(V'^d) \cap (\Phi^e)^{-1}(V'^e)$, $f'^{de}_a = f_a|_{U'^{de}_a}$, and $i'^{de}_a = i_a|_{R'^{de}_a}$. Then $(R'^{de}_a, U'^{de}_a, f'^{de}_a, i'^{de}_a) \subseteq (R_a, U_a, f_a, i_a)$ is a subchart and

$$\Omega \circ \Phi^d|_{U_a^{\prime de}}, \Upsilon \circ \Phi^e|_{U_a^{\prime de}} : (R_a^{\prime de}, U_a^{\prime de}, f_a^{\prime de}, i_a^{\prime de}) \longrightarrow (T, W, h, k)$$

are embeddings.

As $\Omega \circ \Phi^d \circ i'^{de}_a = k|_{R'^{de}_a} = \Upsilon \circ \Phi^e \circ i'^{de}_a$, Theorem 5.4(b) gives $\Theta_{\Omega \circ \Phi^d}|_{i_a(R'^{de}_a)} = \Theta_{\Upsilon \circ \Phi^e}|_{i_a(R'^{de}_a)}$, so that $i_a|^*_{R^{de}}(\Theta_{\Omega \circ \Phi^d}) = i_a|^*_{R^{de}}(\Theta_{\Upsilon \circ \Phi^e})$ as $R^{de} \subseteq R'^{de}_a$. Also $\Lambda_{\Omega \circ \Phi^d} = \Lambda_{\Upsilon \circ \Phi^e}$ as these are defined in Theorem 6.9(ii) using $J_{\Omega \circ \Phi^d}, J_{\Upsilon \circ \Phi^e}$, which are equal by Lemma 5.3. So the two expressions are equal, and similarly for the bottom right diagonal morphism.

The upper triangle in (6.19) commutes because (5.16) gives

$$(\mathrm{id}\otimes\Xi_{\Omega,\Phi^d})\circ\Theta_{\Omega\circ\Phi^d}|_{i_a(R^{de})}=(\Phi^d|_{i_a(R^{de})}^*(\Theta_\Omega)\otimes\mathrm{id})\circ\Theta_{\Phi^d}|_{i_a(R^{de})},$$

and the definitions of Ξ_{Ω,Φ^d} in (5.11) and $\Lambda_{\Omega}, \Lambda_{\Phi^d}, \Lambda_{\Omega\circ\Phi^d}$ in (6.9) imply that

$$\begin{split} & \left(i_a|_{R^{de}}^*(\Xi_{\Omega,\Phi^d})\otimes \mathrm{id}\right)\circ\Lambda_{\Omega\circ\Phi^d}|_{R^{de}} = (\mathrm{id}\otimes\Lambda_{\Phi^d})\circ\Lambda_{\Omega}|_{R^{de}} : \\ & Q_{T,W,h,k}|_{R^{de}}\longrightarrow j^d|_{R^{de}}^*(P_{\Omega})\otimes_{\mathbb{Z}/2\mathbb{Z}} i_a|_{R^{de}}^*(P_{\Phi^d})\otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R_a,U_a,f_a,i_a}|_{R^{de}} . \end{split}$$

Similarly, the other three triangles in (6.19) commute, so (6.19) commutes.

By (6.17), the two routes round the outside of (6.19) are $\alpha_{ab}^d|_{R^{de}}$ and $\alpha_{ab}^e|_{R^{de}}$, which are equal as (6.19) commutes. As we can cover $R'^d_a \cap R'^e_b \cap R'^e_a \cap R'^e_b$ by such Zariski open R^{de} , equation (6.18) follows. Therefore by the Zariski open cover version of Theorem 2.7(i), there is a unique isomorphism α_{ab} in (6.15) such that $\alpha_{ab}|_{R'^d_a \cap R'^d_b} = \alpha_{ab}^d$ for all $d \in D_{ab}$.

If $D_{ab}, R_a^{\prime d}, \ldots, \Phi^d, \Psi^d$ are used to define α_{ab} and $\tilde{D}_{ab}, \tilde{R}_a^{\prime d}, \ldots, \tilde{\Phi}^d, \tilde{\Psi}^d$ are alternative choices yielding $\tilde{\alpha}_{ab}$, then by our usual argument using $D_{ab} \amalg \tilde{D}_{ab}$ and both sets of data we see that $\alpha_{ab} = \tilde{\alpha}_{ab}$, so α_{ab} is independent of choices.

Because the Θ_{Φ^d} , Θ_{Ψ^d} used to define α_{ab} are compatible with Verdier duality and monodromy by (5.14)–(5.15), and the Λ_{Φ^d} , Λ_{Ψ^d} affect only the principal $\mathbb{Z}/2\mathbb{Z}$ -bundles rather than the perverse sheaves, we can show α_{ab} is compatible with Verdier duality and monodromy, in that the following commute:



When a = b we can take $\Psi^d = \Phi^d$, so (6.17) gives $\alpha^d_{aa} = id$, and $\alpha_{aa} = id$.

To prove (6.16), let $a, b, c \in A$, and $x \in R_a \cap R_b \cap R_c$. Applying Theorem 6.3 twice and composing the embeddings, we can construct subcharts $(R'_a, U'_a, f'_a, i'_a) \subseteq (R_a, U_a, f_a, i_a)$, $(R'_b, U'_b, f'_b, i'_b) \subseteq (R_b, U_b, f_b, i_b)$, and $(R'_c, U'_c, f'_c, i'_c) \subseteq (R_c, U_c, f_c, i_c)$ with $x \in R'_a \cap R'_b \cap R'_c$, a critical chart (S, V, g, j) on (X, s), and embeddings $\Phi : (R'_a, U'_a, f'_a, i'_a) \hookrightarrow (S, V, g, j)$, $\Psi : (R'_b, U'_b, f'_b, i'_b) \hookrightarrow (S, V, g, j), \Upsilon : (R'_c, U'_c, f'_c, i'_c) \hookrightarrow (S, V, g, j)$. Then the construction of α_{ab} above yields

$$\begin{split} \alpha_{ab}|_{R'_{a}\cap R'_{b}\cap R'_{c}} &= \left((i^{*}_{b}(\Theta_{\Psi}^{-1})\otimes \mathrm{id})\circ(\mathrm{id}\otimes\Lambda_{\Psi})\circ(\mathrm{id}\otimes\Lambda_{\Phi}^{-1})\circ(i^{*}_{a}(\Theta_{\Phi})\otimes \mathrm{id}) \right)|_{R'_{a}\cap R'_{b}\cap R'_{c}},\\ \alpha_{bc}|_{R'_{a}\cap R'_{b}\cap R'_{c}} &= \left((i^{*}_{c}(\Theta_{\Upsilon}^{-1})\otimes \mathrm{id})\circ(\mathrm{id}\otimes\Lambda_{\Upsilon})\circ(\mathrm{id}\otimes\Lambda_{\Psi}^{-1})\circ(i^{*}_{b}(\Theta_{\Psi})\otimes \mathrm{id}) \right)|_{R'_{a}\cap R'_{b}\cap R'_{c}},\\ \alpha_{ac}|_{R'_{a}\cap R'_{b}\cap R'_{c}} &= \left((i^{*}_{c}(\Theta_{\Upsilon}^{-1})\otimes \mathrm{id})\circ(\mathrm{id}\otimes\Lambda_{\Upsilon})\circ(\mathrm{id}\otimes\Lambda_{\Phi}^{-1})\circ(i^{*}_{a}(\Theta_{\Phi})\otimes \mathrm{id}) \right)|_{R'_{a}\cap R'_{b}\cap R'_{c}}, \end{split}$$

so that $\alpha_{bc}|_{R'_a \cap R'_b \cap R'_c} \circ \alpha_{ab}|_{R'_a \cap R'_b \cap R'_c} = \alpha_{ac}|_{R'_a \cap R'_b \cap R'_c}$. As we can cover $R_a \cap R_b \cap R_c$ by such Zariski open $R'_a \cap R'_b \cap R'_c$, equation (6.16) follows by Theorem 2.7(i).

The Zariski open cover version of Theorem 2.7(ii) now implies that there exists $P_{X,s}^{\bullet}$ in Perv(X), unique up to canonical isomorphism, with isomorphisms

$$\omega_{R_a,U_a,f_a,i_a}: P^{\bullet}_{X,s}|_{R_a} \longrightarrow i^*_a \left(\mathcal{PV}^{\bullet}_{U_a,f_a}\right) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R_a,U_a,f_a,i_a}$$

as in (6.6) for each $a \in A$, with $\alpha_{ab} \circ \omega_{R_a,U_a,f_a,i_a}|_{R_a \cap R_b} = \omega_{R_b,U_b,f_b,i_b}|_{R_a \cap R_b}$ for all $a, b \in A$. Also, (6.7)–(6.8) with (R_a, U_a, f_a, i_a) in place of (R, U, f, i) define isomorphisms $\Sigma_{X,s}|_{R_a}$, $T_{X,s}|_{R_a}$ for each $a \in A$. Equations (6.20)–(6.21) imply that the prescribed values for $\Sigma_{X,s}|_{R_a}$, $T_{X,s}|_{R_a}$ and $\Sigma_{X,s}|_{R_b}$, $T_{X,s}|_{R_b}$ agree when restricted to $R_a \cap R_b$ for all $a, b \in A$. Hence, Theorem 2.7(i) gives unique isomorphisms $\Sigma_{X,s}$, $T_{X,s}$ in (6.5) such that (6.7)–(6.8) commute with (R_a, U_a, f_a, i_a) in place of (R, U, f, i) for all $a \in A$.

Suppose $\{(R_a, U_a, f_a, i_a) : a \in A\}$ and $\{(\tilde{R}_a, \tilde{U}_a, \tilde{f}_a, \tilde{i}_a) : a \in \tilde{A}\}$ are alternative choices above, yielding $P_{X,s}^{\bullet}, \Sigma_{X,s}, \mathcal{T}_{X,s}$ and $\tilde{P}_{X,s}^{\bullet}, \tilde{\Sigma}_{X,s}, \tilde{\mathcal{T}}_{X,s}$. Then applying the same construction to the family $\{(R_a, U_a, f_a, i_a) : a \in A\}$ II $\{(\tilde{R}_a, \tilde{U}_a, \tilde{f}_a, \tilde{i}_a) : a \in \tilde{A}\}$ to get $\hat{P}_{X,s}^{\bullet}$, we have

146

canonical isomorphisms $P_{X,s}^{\bullet} \cong \hat{P}_{X,s}^{\bullet} \cong \tilde{P}_{X,s}^{\bullet}$, which identify $\Sigma_{X,s}, T_{X,s}$ with $\tilde{\Sigma}_{X,s}, \tilde{T}_{X,s}$. Thus $P_{X,s}^{\bullet}, \Sigma_{X,s}, T_{X,s}$ are independent of choices up to canonical isomorphism.

Now fix $\{(R_a, U_a, f_a, i_a) : a \in A\}$, $P_{X,s}^{\bullet}, \Sigma_{X,s}, T_{X,s}$ and $\omega_{R_a, U_a, f_a, i_a}$ for $a \in A$ above for the rest of the proof. Suppose (R, U, f, i) is a critical chart on (X, s). Running the construction above with the family $\{(R_a, U_a, f_a, i_a) : a \in A\}$ II $\{(R, U, f, i)\}$, we can suppose it yields the same (not just isomorphic) $P_{X,s}^{\bullet}, \Sigma_{X,s}, T_{X,s}$ and $\omega_{R_a, U_a, f_a, i_a}$, but it also yields a unique $\omega_{R, U, f, i}$ in (6.6) which makes (6.7)–(6.8) commute. This proves Theorem 6.9(i).

Let $\Phi : (R, U, f, i) \hookrightarrow (S, V, g, j)$ be an embedding of critical charts on (X, s). The definition of Λ_{Φ} in Theorem 6.9(ii) is immediate. Run the construction above using the family $\{(R_a, U_a, f_a, i_a) : a \in A\}$ II $\{(R, U, f, i), (S, V, g, j)\}$, and follow the definition of α_{ab} with (R, U, f, i), (S, V, g, j) in place of $(R_a, U_a, f_a, i_a), (R_b, U_b, f_b, i_b)$. We can take

$$D_{ab} = \{d\}, \quad (R_a^{\prime d}, U_a^{\prime d}, f_a^{\prime d}, i_a^{\prime d}) = (R, U, f, i), \quad (R_b^{\prime d}, U_b^{\prime d}, f_b^{\prime d}, i_b^{\prime d}) = (S^d, V^d, g^d, j^d) = (S, V, g, j),$$

 $\Phi^d = \Phi$ and $\Psi^d = \mathrm{id}_V$. Then (6.17) gives $\alpha_{ab} = \alpha^d_{ab} = (\mathrm{id} \otimes \Lambda_{\Phi}^{-1}) \circ (i^*(\Theta_{\Phi}) \otimes \mathrm{id})$. Thus,

 $\alpha_{ab} \circ \omega_{R_a, U_a, f_a, i_a}|_{R_a \cap R_b} = \omega_{R_b, U_b, f_b, i_b}|_{R_a \cap R_b}$

implies that (6.10) commutes, proving Theorem 6.9(ii).

6.4. \mathscr{D} -modules and mixed Hodge modules. Once again, the proof of Theorem 6.9 carries over to our other contexts in §2.6–§2.10 using the general framework of §2.5, now also making use of the Stack Property (x) for objects. For the case of mixed Hodge modules, we use Theorem 2.21(ii) to glue the $i_a^*(\mathcal{HV}_{U_a,f_a}) \otimes_{\mathbb{Z}/2\mathbb{Z}} Q_{R_a,U_a,f_a,i_a}$ on $R_a \subseteq X$ for $a \in A$ with their natural strong polarizations (2.25), which are preserved by the isomorphisms α_{ab} in §6.3 on overlaps $R_a \cap R_b$.

APPENDIX A. COMPATIBILITY RESULTS, BY JÖRG SCHÜRMANN

In the main body of the paper, when comparing results for mixed Hodge modules to those involving perverse sheaves, we rely on the compatibility between duality and Thom–Sebastiani type isomorphisms of perverse sheaves and mixed Hodge modules. These compatibility statements cannot easily be read off from the existing literature, so we provide proofs here.

Proposition A.1. If X is a \mathbb{C} -scheme and $f : X \to \mathbb{C}$ is regular, then Massey's natural isomorphisms from [37] quoted as Theorem 2.11(iv) coincide with the image under the realization functor of Saito's analogous isomorphisms [45] between functors on mixed Hodge modules. There is also an analogous compatibility result for X a complex analytic space equipped with an analytic function f.

Proof. Massey's construction in [37] of the duality isomorphisms uses the definition of the vanishing cycle functor in terms of the local cohomology of suitable real half-spaces, compare also [50]. Using their notation, the compatibility comes down to compatibility of the diagram

Here the upper, respectively lower horizontal isomorphisms are the ones of Saito, respectively Massey, and the vertical isomorphisms follow for example from [50, Lem. 1.3.2, p. 69]. Saito deduces his duality isomorphism in [45, Lem. 5.2.4, p. 965] from a pairing on nearby cycles induced by a pairing $F \otimes G \to a_X^2 A$ for $F, G \in D_c^b(X)$, with $a_X : X \to pt$ the constant map and $A \subset \mathbb{C}$ a coefficient field. But Massey's duality isomorphism can be also be induced from such a pairing fitting into a commutative diagram, with

$$L_0 = \{ \operatorname{Re}(f) = 0 \}$$
 and $j : \{ \operatorname{Re}(f) = 0, f \neq 0 \} \to L_0$

the open inclusion:

Here the isomorphism $j^*(a'_X A|_{L_0}) \cong j^*(a'_{L_0} A)[1]$ comes from the fact that $\operatorname{Re}(f)$ has no critical points (in a stratified sense) in $X \setminus X_0$, locally near X_0 . But then the commutativity of (A.2) implies by [45, Lem. 5.2.4, p. 965] the commutativity of (A.1), concluding the proof.

A similar compatibility question arises for the Thom–Sebastiani isomorphism. Here the precise statement is the following.

Proposition A.2. Let $f_i : Y_i \to \mathbb{C}$ be regular functions on smooth \mathbb{C} -schemes, for i = 1, 2. Let $f = f_1 \boxplus f_2 : Y_1 \times Y_2 \to \mathbb{C}$ be as in Theorem 2.13. Then the isomorphism (2.8) of Massey [35,50] coincides with the image under the realization functor of Saito's analogous isomorphism (2.26) of [49] for mixed Hodge modules.

Proof. The Thom–Sebastiani isomorphism (2.26) is constructed by Saito [49, Th. 2.6] based on the Verdier specialization [53]. First, let $f: Y \to \mathbb{C}$ be a regular function, with $X = f^{-1}(0)$ of codimension one, so that the normal cone $C_X Y = X \times \mathbb{C}$ becomes a trivial line bundle with $f': C_X Y \to \mathbb{C}$ given by the projection. Let $p: D_X Y \to \mathbb{C}$ be the deformation to the normal cone with $C_X Y = p^{-1}(0) \subset D_X Y$, with $q: D_X Y \to Y$ the natural map. Then f' extends to a function $g: D_X Y \to \mathbb{C}$, with g = f/s on $p \neq 0 = Y \times \mathbb{C}$ for s the usual coordinate on \mathbb{C} . For $F \in D_c^b(Y)$, we get a commutative diagram

$$\begin{split} \phi^p_{f'}(sp_XF) &\longleftarrow \phi^p_f(F) \\ & \downarrow \cong & \cong \downarrow \\ R\Gamma_{\{\operatorname{Re}(f')\geq 0\}}(sp_XF)|_X &\longleftarrow R\Gamma_{\{\operatorname{Re}(f)\geq 0\}}(F)|_X. \end{split}$$

Here the monodromical sheaf complex $sp_X F \in D_c^b(C_X Y)_{\text{mon}}$ is the Verdier specialization of F as in [49,53]. The upper horizontal isomorphism is the one of [49, Lem. 2.2], whereas the vertical isomorphisms are those of [50, Lem. 1.3.2, p. 69]. The lower horizontal map is defined by the natural base change morphism

$$R\Gamma_{\{\operatorname{Re}(f')\geq 0\}}(sp_XF)|_X \longleftarrow \psi_p(R\Gamma_{\{\operatorname{Re}(g)\geq 0\}}(q^*F))|_X \cong R\Gamma_{\{\operatorname{Re}(f)\geq 0\}}(F)|_X,$$

where the last isomorphism follows as in [50, Lem. 1.3.3, p. 70-71].

Consider now the situation in the proposition, with

$$f = f_1 \boxplus f_2 : Y = Y_1 \times Y_2 \to \mathbb{C},$$

also $X_i = f_i^{-1}(0)$ and $X = f^{-1}(0)$; finally let $\mu_{f_i} = R\Gamma_{\{\operatorname{Re}(f_i) \geq 0\}}$ to shorten the notation. Let

$$\pi: C_{X_1}Y_1 \times C_{X_2}(Y_2) = (X_1 \times X_2) \times \mathbb{C}^2 \to (X_1 \times X_2) \times \mathbb{C} \subset C_X(Y)$$

be the map induced by addition in the fibres. Then, for $F_i \in D_c^b(Y_i)$, one gets a commutative diagram

The upper horizontal and left vertical isomorphisms form the Thom–Sebastiani isomorphism (2.26) of [49], whereas the lower horizontal isomorphism is the Thom–Sebastiani isomorphism (2.8) of [35, 50]. This concludes the proof.

References

- V.I. Arnold, S.M. Gusein-Zade and A.N. Varchenko, Singularities of differentiable maps, Volume 1, Monographs in Math. 82, Birkhäuser, 1985.
- K. Behrend, Donaldson-Thomas type invariants via microlocal geometry, Ann. of Math. 170 (2009), 1307– 1338. DOI: 10.4007/annals.2009.170.1307
- K. Behrend, J. Bryan and B. Szendrői, Motivic degree zero Donaldson-Thomas invariants, Invent. Math. 192 (2013), 111–160. DOI: 10.1007/s00222-012-0408-1
- [4] K. Behrend and B. Fantechi, Gerstenhaber and Batalin-Vilkovisky structures on Lagrangian intersections, pages 1–47 in Algebra, arithmetic, and geometry, Progr. Math. 269, Birkhäuser, Boston, MA, 2009.
- [5] A.A. Beilinson, J. Bernstein and P. Deligne, Faisceaux pervers, Astérisque 100, 1982.
- [6] O. Ben-Bassat, C. Brav, V. Bussi, and D. Joyce, A 'Darboux Theorem' for shifted symplectic structures on derived Artin stacks, with applications, Geom. & Top. 19 (2015), 1287–1359. DOI: 10.2140/gt.2015.19.1287
 [7] J. F. Dirich, A. Livi, G. Lindi, K. Bassa, M. Bassa, J. Bassa,
- [7] J.-E. Björk, Analytic D-modules and applications, Kluwer, Dordrecht, 1993.
- [8] A. Borel et al., Algebraic D-modules, Perspectives in Mathematics 2, Academic Press, 1987.
 [9] C. Brav, V. Bussi and D. Joyce, A Darboux theorem for derived schemes with shifted symplectic structure,
 - 2013. arXiv: 1305.6302
- [10] V. Bussi, Categorification of Lagrangian intersections on complex symplectic manifolds using perverse sheaves of vanishing cycles, 2014. arXiv: 1404.1329
- [11] V. Bussi, D. Joyce and S. Meinhardt, On motivic vanishing cycles of critical loci, 2013. arXiv: 1305.6428
- [12] S.C. Coutinho, A primer of algebraic *D*-modules, L.M.S. Student texts 33, Cambridge University Press, 1995. DOI: 10.1017/CBO9780511623653
- [13] A. D'Agnolo and P. Schapira, Quantization of complex Lagrangian submanifolds, Advances in Math. 213 (2007), 358–379. DOI: 10.1016/j.aim.2006.12.009
- [14] A. Dimca, Sheaves in Topology, Universitext, Springer-Verlag, Berlin, 2004. DOI: 10.1007/978-3-642-18868-8
- [15] A. Dimca and B. Szendrői, The Milnor fibre of the Pfaffian and the Hilbert scheme of four points on C³, Math. Res. Lett. 16 (2009), 1037–1055. DOI: 10.4310/MRL.2009.v16.n6.a12
- [16] S.K. Donaldson and R.P. Thomas, Gauge Theory in Higher Dimensions, Chapter 3 in S.A. Huggett, L.J. Mason, K.P. Tod, S.T. Tsou and N.M.J. Woodhouse, editors, The Geometric Universe, Oxford University Press, Oxford, 1998.
- [17] E. Freitag and R. Kiehl, Etale cohomology and the Weil Conjecture, Ergeb. der Math. und ihrer Grenzgebiete 13, Springer-Verlag, 1988. DOI: 10.1007/978-3-662-02541-3
- [18] K. Fukaya, Y.-G. Oh, H. Ohta and K. Ono, Lagrangian intersection Floer theory anomaly and obstruction, Parts I & II. AMS/IP Studies in Advanced Mathematics, 46.1 & 46.2, A.M.S./International Press, 2009.
- T. Gaffney and H. Hauser, Characterizing singularities of varieties and of mappings, Invent. math. 81 (1985), 427–447. DOI: 10.1007/BF01388580
- [20] G. Guibert, F. Loeser and M. Merle, Iterated vanishing cycles, convolution, and a motivic analogue of a conjecture of Steenbrink, Duke Math. J. 132 (2006), 409–457. DOI: 10.1215/S0012-7094-06-13232-5
- [21] R. Hotta, T. Tanisaki and K. Takeuchi, *D-modules, perverse sheaves, and representation theory*, Progr. Math. 236, Birkhäuser, Boston, MA, 2008.
- [22] D. Huybrechts and R.P. Thomas, Deformation-obstruction theory for complexes via Atiyah and Kodaira-Spencer classes, Math. Ann. 346 (2010), 545–569. DOI: 10.1007/s00208-009-0397-6

- [23] D. Joyce, A classical model for derived critical loci, to appear in Journal of Differential Geometry, 2015. arXiv: 1304.4508
- [24] D. Joyce and Y. Song, A theory of generalized Donaldson-Thomas invariants, Mem. Amer. Math. Soc. 217 (2012), no. 1020.
- [25] M. Kashiwara, The Riemann-Hilbert problem for holonomic systems, Publ. RIMS 20 (1984), 319–365. DOI: 10.2977/prims/1195181610
- [26] M. Kashiwara, *D-modules and microlocal calculus*, Trans. Math. Mono. 217, A.M.S., 2003.
- [27] M. Kashiwara and P. Schapira, Sheaves on manifolds, Grundlehren der Math. Wiss. 292, Springer-Verlag, Berlin, 1990. DOI: 10.1007/978-3-662-02661-8
- [28] M. Kashiwara and P. Schapira, Constructibility and duality for simple modules on symplectic manifolds, Amer. J. Math. 130 (2008), 207–237. DOI: 10.1353/ajm.2008.0007
- [29] M. Kashiwara and P. Schapira, Deformation quantization modules, Astérisque 345, 2012.
- [30] R. Kiehl and R. Weissauer, Weil Conjectures, perverse sheaves and l'adic Fourier transform, Springer-Verlag, 2001. DOI: 10.1007/978-3-662-04576-3
- [31] Y.-H. Kiem and J. Li, Categorification of Donaldson-Thomas invariants via perverse sheaves, 2012. arXiv: 1212.6444
- [32] M. Kontsevich and Y. Soibelman, Stability structures, motivic Donaldson-Thomas invariants and cluster transformations, 2008. arXiv: 0811.2435
- [33] M. Kontsevich and Y. Soibelman, Cohomological Hall algebra, exponential Hodge structures and motivic Donaldson-Thomas invariants, Commun. Number Theory Phys. 5 (2011), 231–352. DOI: 10.4310/CNTP.2011.v5.n2.a1
- [34] P. Maisonobe and Z. Mebkhout, Le théorème de comparaison pour les cycles évanescents, pages 311–389 in Éléments de la théorie des systèmes différentiels géométriques, Sémin. Congr., 8, Soc. Math. France, Paris, 2004.
- [35] D. Massey, The Sebastiani-Thom isomorphism in the derived category, Compositio Math. 125 (2001), 353– 362. DOI: 10.1023/A:1002608716514
- [36] D. Massey, Notes on perverse sheaves and vanishing cycles, 1999. arXiv: math.AG/9908107
- [37] D. Massey, Natural commuting of vanishing cycles and the Verdier dual, 2009. arXiv: 0908.2799
- [38] J. Mather and S.S. Yau, Classification of isolated hypersurface singularities by their moduli algebra, Invent. Math. 69 (1982), 243–251. DOI: 10.1007/BF01399504
- [39] L. Maxim, M. Saito and J. Schürmann, Symmetric products of mixed Hodge modules, J. Math. Pures Appl. 96 (2011), 462–483. DOI: 10.1016/j.matpur.2011.04.003
- [40] D. Nadler, Microlocal branes are constructible sheaves, Selecta Math. 15 (2009), 563–619.
 DOI: 10.1007/s00029-009-0008-0
- [41] D. Nadler and E. Zaslow, Constructible sheaves and the Fukaya category, J. Amer. Math. Soc. 22 (2009), 233–286. DOI: 10.1090/S0894-0347-08-00612-7
- [42] T. Pantev, B. Toën, M. Vaquié and G. Vezzosi, *Shifted symplectic structures*, Publ. Math. I.H.E.S. 117 (2013), 271–328. DOI: 10.1007/s10240-013-0054-1
- [43] K. Rietsch, An introduction to perverse sheaves, pages 391–429 in V. Dlab and C.M. Ringel, editors, Representations of finite dimensional algebras and related topics in Lie theory and geometry, Fields Inst. Commun. 40, A.M.S., Providence, RI, 2004.
- [44] C. Sabbah, On a twisted de Rham complex, II, 2010. arXiv: 1012.3818
- [45] M. Saito, Modules de Hodge polarisables, Publ. RIMS 24 (1988), 849–995. DOI: 10.2977/prims/1195173930
- [46] M. Saito, Duality for vanishing cycle functors, Publ. RIMS 25 (1989), 889–921.
 DOI: 10.2977/prims/1195172510
- [47] M. Saito, Mixed Hodge Modules, Publ. RIMS 26 (1990), 221–333. DOI: 10.2977/prims/1195171082
- [48] M. Saito, *D*-modules on analytic spaces, Publ. RIMS 27 (1991), 291–332. DOI: 10.2977/prims/1195169840
- [49] M. Saito, Thom-Sebastiani Theorem for Hodge Modules, preprint, 2010.
- [50] J. Schürmann, Topology of singular spaces and constructible sheaves, Monografie Matematyczne 63, Birkhäuser, Basel, 2003.
- [51] B. Szendrői, Nekrasov's partition function and refined Donaldson-Thomas theory: the rank one case, SIGMA (2012) 088, 16pp.
- [52] R.P. Thomas, A holomorphic Casson invariant for Calabi-Yau 3-folds, and bundles on K3 fibrations, J. Diff. Geom. 54 (2000), 367–438.
- [53] J.-L. Verdier, Spécialization de faisceaux et monodromie modérée, Astérisque 101-102 (1983), 332–364.

Address for Christopher Brav: Faculty of Mathematics, Higher School of Economics, 7 Vavilova Str., Moscow, Russia E-Mail: chris.i.brav@gmail.com. Address for Vittoria Bussi: ICTP, Strada Costiera 11, Trieste, Italy E-Mail: vbussi@ictp.it. Address for Delphine Dupont, Dominic Joyce and Balázs Szendrői: The Mathematical Institute, Woodstock Road, Oxford, UK E-Mails: joyce@maths.ox.ac.uk, szendroi@maths.ox.ac.uk Address for Jörg Schürmann: Mathematische Institut, Universität Münster, Einsteinstrasse 62, 48149 Münster, Ger-Many. E-Mail: jschuerm@math.uni-muenster.de.

THE PUNCTUAL HILBERT SCHEMES FOR THE CURVE SINGULARITIES OF TYPE A_{2d}

YOSHIKI SŌMA AND MASAHIRO WATARI

ABSTRACT. Pfister and Steenbrink studied punctual Hilbert schemes for irreducible curve singularities. In particular, they analyzed the structure of certain punctual Hilbert schemes for monomial curve singularities. In this paper, we generalize their results about the curve singularities of type A_{2d} by clarifying the relationships among the punctual Hilbert schemes for the singularities.

1. INTRODUCTION

Let \mathcal{O} be the complete local ring of an irreducible curve singularity over an algebraically closed field k of characteristic 0. We denote by $\overline{\mathcal{O}}$ and δ the normalization of \mathcal{O} and the δ -invariant of \mathcal{O} respectively. Pfister and Steenbrink [3] defined a special subset \mathcal{M} of the Grassmannian $\operatorname{Gr}(\delta, \overline{\mathcal{O}}/I(2\delta))$ where $I(2\delta)$ is the set of all elements in \mathcal{O} whose orders are greater than or equal to 2δ . It is a projective variety which consists of \mathcal{O} -sub-modules and we call it the *Pfister-Steenbrink variety* (PS variety) for the given singularity. For any positive integer r, the existence of the punctual Hilbert scheme of degree r was also shown there. It is a projective variety which parametrizes the ideals of codimension r in \mathcal{O} and is realized as a connected component of the PS variety.

A numerical semi-group Γ is called *monomial*, if any curve singularity with it has no moduli. Pfister and Steenbrink determined all monomial semi-groups in [3]. Using the intersections with Schubert cells, they also analyzed the structure of the PS varieties for the curve singularities with monomial semi-groups. The cases for the curve singularities of types A_{2d} , E_6 and E_8 were also involved in their study. The punctual Hilbert schemes for the curve singularity of type A_1 and related topics were discussed by Ran in [4, 5]. Kawai [2] computed the Euler characteristic of the Hilbert scheme $C^{[d]}$ of 0-dimensional length d subschemes of a projective curve C with only the A_1 and A_2 singularities. The structures of punctual Hilbert schemes for the curve singularities of types A_1 and A_2 were used in this computation. The result was also discussed in the context of string theory. Recently, by using computational methods, the authors of this paper studied the structure of all punctual Hilbert schemes for the curve singularities of types E_6 and E_8 in [7]. On the other hand, the PS varieties for curve singularities were studied from another point of view. Rego [6] introduced the *compactified Jacobian* of singular curves. He also constructed the *Jacobi factor* for a curve singularity. The Jacobi factor and the PS variety coincide for a given curve singularity.

In this paper, we consider the curve singularities of type A_{2d} (i.e. the curve singularities whose local rings are isomorphic to $k[[t^2, t^{2d+1}]]$ where $d \in \mathbb{N}$). We denote by $\mathcal{M}_{d,r}$ the punctual Hilbert scheme of degree r for the curve singularity of type A_{2d} . Pfister and Steenbrink showed that: the *PS variety* $\mathcal{M}_{d,2d}$ is an irreducible rational projective variety of dim $\mathcal{M}_{d,2d} = d$. In particular, (i) $\mathcal{M}_{1,2} \cong \mathbb{P}^1$, (ii) $\mathcal{M}_{2,4}$ is a quardratic cone in \mathbb{P}^5 , (iii) $\mathcal{M}_{3,6}$ is a threefold with a singular

²⁰⁰⁰ Mathematics Subject Classification. 14C05, 14H20.

Key words and phrases. punctual Hilbert schemes, curve singularities of type A_{2d} .

line with transverse singularity of type A_2 . In general, the following fact also holds: if $r \geq 2\delta$, then the punctual Hilbert scheme of degree r for an irreducible curve singularity coincides with the PS variety for the same singularity (Corollary 11). So it is enough to consider their degree r within $1 \leq r \leq 2\delta$ for the analysis of the structure of punctual Hilbert schemes (Remark 12). Since the δ -invariant for a given A_{2d} singularity equals d, the PS variety coincides with $\mathcal{M}_{d,2d}$ by the above fact. Our main theorems are stated as follows:

Theorem 1. Let d and r be two integers with $1 \leq r \leq 2d$. Putting s := [r/2], the punctual Hilbert scheme $\mathcal{M}_{d,r}$ is a rational projective variety with dim $\mathcal{M}_{d,r} = s$. If $r \geq 2$, then it is isomorphic to the Pfister-Steenbrink variety $\mathcal{M}_{s,2s}$.

Theorem 2. We keep the notations d, r and s as in Theorem 1. The punctual Hilbert schemes for the curve singularity of type A_{2d} have the following structures:

- (i): The punctual Hilbert scheme $\mathcal{M}_{d,1}$ consists of one point.
- (ii): The punctual Hilbert schemes $\mathcal{M}_{d,2}$ and $\mathcal{M}_{d,3}$ are isomorphic to a projective line \mathbb{P}^1 .
- (iii): The punctual Hilbert scheme $\mathcal{M}_{d,r}$ with $4 \leq r \leq 2d$ is a singular projective variety whose singular locus is given by $\mathcal{M}_{d,2s-2} \cap \mathcal{M}_{d,2s-1} \cong \mathcal{M}_{d,2s-3}$.

The present paper is organized as follows: In Section 2 below, we briefly recall Pfister-Steenbrink theory introduced in [3]. In Section 3, we study ideals in the local ring $k[[t^2, t^{2d+1}]]$ of the curve singularities of type A_{2d} . From the point of view of Γ -semi-module structure of orders, we determine the sets of ideals in \mathcal{O} with codimension r ($1 \leq r \leq 2\delta$) and their decompositions. These yield affine cell decompositions of the punctual Hilbert schemes for the curve singularities of type A_{2d} . In Section 4, we first show the irreducibility of the punctual Hilbert schemes. We also prove the following proposition:

Proposition 3. The following relations hold for punctual Hilbert schemes:

- (i): For integers d and s with $1 \leq s \leq d-1$, we have $\mathcal{M}_{d,2s} \cong \mathcal{M}_{d,2s+1}$.
- (ii): For integers d, d' and r with $1 \le r \le \min\{2d, 2d'\}$, we have $\mathcal{M}_{d,r} \cong \mathcal{M}_{d',r}$.

Finally, we prove Theorem 1 and 2 by using them.

Acknowledgement The authors would like to express his sincere gratitude to Professor Fumio Sakai for his valuable advices and warm encouragement during the preparation of the present article.

2. PFISTER-STEENBRINK THEORY FOR PUNCTUAL HILBERT SCHEMES

In the present paper, we restrict ourselves to monomial curve singularities defined below. However, the notions in this section hold in more general situations. See [3] for details.

Definition 4. A monomial curve singularity is an irreducible curve singularity whose local ring is isomorphic to $k[[t^{a_1}, \ldots, t^{a_m}]]$ for some $a_1, \ldots, a_m \in \mathbb{N}$.

Remark 5. Without loss of generality, we may assume that $gcd(a_1, \ldots, a_m) = 1$ in Definition 4.

Let $\mathcal{O} = k[[t^{a_1}, \ldots, t^{a_m}]]$ be the local ring of a monomial curve singularity. Its normalization $\overline{\mathcal{O}}$ is isomorphic to k[[t]]. We call $\Gamma := \{ \operatorname{ord}_t(f) \mid f \in \mathcal{O} \}$ the *semi-group* of \mathcal{O} . The positive integer $\delta := \dim_k(\overline{\mathcal{O}}/\mathcal{O})$ is called the δ -invariant of \mathcal{O} . For $n \in \mathbb{N}$, set $\overline{I}(n) := \{ f \in \overline{\mathcal{O}} \mid \operatorname{ord}_t(f) \geq n \}$ and $I(n) := \overline{I}(n) \cap \mathcal{O}$. Setting $\operatorname{ord}_t(0) = \infty$, we regard $\overline{I}(n)$ (resp. I(n)) as an ideal of $\overline{\mathcal{O}}$ (resp. \mathcal{O}). For an ideal I of \mathcal{O} , we denote by $\Gamma(I) := \{ \operatorname{ord}_t(f) \mid f \in I \}$ the set of orders of all elements in I. Put $G(I) := \Gamma \setminus \Gamma(I)$. For $r \in \mathbb{N}$, define

 $\mathcal{I}_r := \{ I | I \text{ is an ideal of } \mathcal{O} \text{ with } \dim_k \mathcal{O}/I = r \}.$

A subset $\Delta \subset \mathbb{Z}$ is called a Γ -semi-module, if $\Delta + \Gamma \subset \Delta$. Note that if Δ is a Γ -semimodule, then $\Delta - r$ is also a Γ -semi-module for any integer r. We write $\Delta = \langle \alpha_1, \cdots, \alpha_p \rangle_{\Gamma}$ for a Γ -semi-module Δ which is minimally generated by $\alpha_1, \dots, \alpha_p$ (i.e $\Delta = \sum_{i=1}^p (\alpha_i + \Gamma)$ and $\Delta \supseteq \sum_{i=1, i \neq j}^{p} (\alpha_i + \Gamma)$ for $\forall j \in \{1, \dots, p\}$). We also denote by $\mathcal{I}(\Delta)$ the set of all ideals of \mathcal{O} whose set of orders are Δ . Note that $\mathcal{I}(\Delta) \neq \emptyset$ if and only if $\Delta \subset \Gamma$.

The following facts are known:

Lemma 6 ([7], Lemma 5). An ideal I in \mathcal{O} belongs to \mathcal{I}_r if and only if we have $\sharp G(I) = r$.

Proposition 7 ([7], Proposition 7). There exists a finite number of distinct Γ -semi-modules $\Delta_{r,1}, \cdots, \Delta_{r,n_r}$ such that

(1)
$$\mathcal{I}_r = \bigcup_{l=1}^{n_r} \mathcal{I}(\Delta_{r,l}).$$

Remark 8. The set of Γ -semi-modules $\Delta_{r,l}$ in (1) is an invariant for the codimension r.

Let Gr $(\delta, \overline{\mathcal{O}}/I(2\delta))$ be the Grassmannian which consists of δ -dimensional linear subspaces of $\overline{\mathcal{O}}/I(2\delta)$. For $V \in \mathrm{Gr}\left(\delta, \overline{\mathcal{O}}/\mathrm{I}(2\delta)\right)$, we define a multiplication by $\mathcal{O} \times V \ni (f, \overline{v}) \mapsto \overline{fv} \in V$. Set

 $\mathcal{M} := \left\{ V \in \mathrm{Gr}\left(\delta, \overline{\mathcal{O}}/\mathrm{I}(2\delta)\right) \mid V \text{ is an } \mathcal{O}\text{-sub-module w.r.t. the multiplication} \right\}.$

Consider the composition map

(2)
$$\psi: \mathcal{M} \xrightarrow{\psi_1} \mathcal{M}_{\delta,2\delta}(k) / \sim \xrightarrow{\psi_2} \mathbb{P}^N$$

where $M_{\delta,2\delta}(k)$ is the set of all $\delta \times 2\delta$ matrices over k and the equivalence relation \sim is the similarity of matrices. For a formal power series $f = \sum_{j=0}^{\infty} a_j t^j$ in $\overline{\mathcal{O}}$, we denote its coset in $\overline{\mathcal{O}}/I(2\delta)$ by $\overline{f} = \sum_{j=0}^{2\delta-1} a_j \tau^j$. The notation τ signifies the coset of t. Define $\operatorname{ord}_{\tau}(\overline{f})$ by $\operatorname{ord}_t(f)$ (resp. ∞), if $\operatorname{ord}_t(f) \leq 2\delta - 1$ (resp. $\operatorname{ord}_t(f) \geq 2\delta$). In this paper, we use the notation $[v_1, \cdots, v_{\delta}]_k$ for a k-vector space generated by v_1, \dots, v_{δ} . Let $V = [\overline{f}_1, \cdots, \overline{f}_{\delta}]_k$ be an element of \mathcal{M} where $\overline{f}_i = \sum_{j=0}^{2\delta-1} a_{ij} \tau^j$. We identify \overline{f}_i with the point $\mathbf{a}_i = (a_{i0}, \cdots, a_{i2\delta-1})$ in $k^{2\delta}$. Let A_V be the $\delta \times 2\delta$ matrix whose *i*th row is \mathbf{a}_i . We call it the *representation matrix* of V. The first map ψ_1 in (2) is defined by sending a k-vector space V to the coset of A_V . The second map ψ_2 in (2) is the Plücker embedding with $N = {\binom{2\delta}{\delta}} - 1$. Note that ψ_1 and ψ_2 are injective. For $r \in \mathbb{N}$, Pfister and Steenbrink defined a map $\varphi_r : \mathcal{I}_r \to \mathcal{M}$ by $\varphi_r(I) = t^{-r} I/I(2\delta)$.

Proposition 9 ([3], Theorem 3). The map φ_r is injective for any r. Furthermore, it is bijective for $r \geq 2\delta$. The image $(\psi \circ \varphi_r)(\mathcal{I}_r)$ is Zariski closed in $\psi(\mathcal{M})$.

Put $\mathcal{M}_r := \varphi_r(\mathcal{I}_r)$. Since ψ is injective, we identify $\psi(\mathcal{M})$ and $\psi(\mathcal{M}_r)$ with \mathcal{M} and \mathcal{M}_r respectively.

Definition 10. We call \mathcal{M} and \mathcal{M}_r the Pfister-Steenbrink variety (PS variety for short) and the punctual Hilbert scheme of degree r for a given curve singularity respectively.

The following fact follows from Proposition 9:

Corollary 11. Any punctual Hilbert scheme \mathcal{M}_r with $r \geq 2\delta$ coincides with the PS variety \mathcal{M} .

Remark 12. By virtue of Corollary 11, it is enough to consider codimensions r within $1 \le r \le 2\delta$ for the analysis of \mathcal{M}_r .

Set $\mathcal{M}(\Delta_{r,l}) := \varphi_r(\mathcal{I}(\Delta_{r,l}))$ for each component $\mathcal{I}(\Delta_{r,l})$ in (1). Since ψ is injective, we also identify $\psi(\mathcal{M}(\Delta_{r,l}))$ with $\mathcal{M}(\Delta_{r,l})$. Namely, we regard $\mathcal{M}(\Delta_{r,l})$ as a subset of the punctual Hilbert scheme \mathcal{M}_r parametrizing ideals in $\mathcal{I}(\Delta_{r,l})$. Set $[a,b] := \{x \in \mathbb{Z}_{>0} | a \leq x \leq b\}$. For

a Γ -semi-module $\Delta_{r,l} = \langle \alpha_1, \cdots, \alpha_{p_l} \rangle_{\Gamma}$, we have $\Delta_{r,l} - r = \langle \alpha_1 - r, \cdots, \alpha_{p_l} - r \rangle_{\Gamma}$. Define $A := \{\alpha_1 - r, \cdots, \alpha_{p_l} - r\} \cap [0, 2\delta - 1]$ and $J_{\alpha} := [\alpha + 1, 2\delta - 1] \setminus \{\Delta_{r,l} - r\}$ for $\alpha \in A$. The following facts are known:

Proposition 13 ([3], Theorem 7). Let I be an element of $\mathcal{I}(\Delta_{r,l})$. There exist uniquely determined $b_{\alpha j} \in k$ such that the \mathcal{O} -sub-module $\varphi_r(I)$ is generated by

$$\overline{f}_{\alpha} := \tau^{\alpha} + \sum_{j \in J_{\alpha}} b_{\alpha j} \tau^{j} \quad (\alpha \in A).$$

Corollary 14 ([3], Corollary of Theorem 11). The component $\mathcal{M}(\Delta_{r,l})$ is isomorphic to the affine space k^N where $N = \sum_{\alpha \in A} \sharp J_{\alpha}$.

We obtain an affine cell decomposition of \mathcal{M}_r by Proposition 7 and Corollary 14.

Proposition 15. The punctual Hilbert scheme \mathcal{M}_r of degree r has an affine cell decomposition

(3)
$$\mathcal{M}_r = \bigcup_{l=1}^{n_r} \mathcal{M}(\Delta_{r,l})$$

The following fact also follows from Corollary 14:

Proposition 16. If \mathcal{M}_r is irreducible, then it is a rational projective variety.

The 2δ -dimensional k-vector space $\overline{\mathcal{O}}/I(2\delta)$ has the canonical flag

$$0 \subset V_1 \subset V_2 \subset \cdots \subset V_{2\delta} = \mathcal{O}/I(2\delta)$$

where $V_i = \overline{I}(2\delta - i)/I(2\delta)$ for $1 \le i \le 2\delta$. This induces a partition of $\operatorname{Gr}(\delta, \overline{\mathcal{O}}/I(2\delta))$ into Schubert cells W_{a_1,\ldots,a_δ} for $\delta \ge a_1 \ge \cdots \ge a_\delta \ge 0$, which is defined by

$$W_{a_1,\dots,a_{\delta}} := \left\{ W \in \operatorname{Gr}(\delta,\overline{\mathcal{O}}/I(2\delta)) \middle| \begin{array}{l} \dim(W \cap V_{\delta+i-a_i}) = i \text{ for } 1 \le i \le \delta \\ \dim(W \cap V_j) < i \text{ for } j < \delta+i-a_i \end{array} \right\}.$$

For an index set $\Lambda = \{a_1, \ldots, a_{\delta}\}$, we sometimes write W_{Λ} instead of $W_{a_1, \ldots, a_{\delta}}$.

Proposition 17. We have $W_{b_1,...,b_{\delta}} \subset \overline{W_{a_1,...,a_{\delta}}}$ if and only if $b_i \geq a_i$ holds for $1 \leq i \leq \delta$.

For the details about Schubert cells, see [1, p.195].

Lemma 18. Let $\mathcal{M}(\Delta_{r,l})$ be a component in (3) and write $\{b_1, \ldots, b_{\delta}\} = (\Delta_{r,l} - r) \cap [0, 2\delta - 1]$ where $0 \leq b_1 < \cdots < b_{\delta} < 2\delta$. Setting $a_{\delta-i+1} = b_i - i + 1$ for $1 \leq i \leq \delta$, we have

$$\mathcal{M}(\Delta_{r,l}) = \mathcal{M}_r \cap W_{a_1,\dots,a_\delta}.$$

Proof. It is known that our assertion is true for $r = 2\delta$ (see Lemma 5 in [3]). So we consider the case where $r < 2\delta$. Since $\mathcal{M}(\Delta_{r,l}) \subset \mathcal{M}_r \subset \mathcal{M}_{2\delta}$, there exists a Γ -semi-module $\Delta_{2\delta}$ such that $\mathcal{M}(\Delta_{2\delta})$ is a component of $\mathcal{M}_{2\delta}$ and $\Delta_{2\delta} - 2\delta = \Delta_{r,l} - r$. It follows from the above fact for $r = 2\delta$ that $\mathcal{M}(\Delta_{r,l}) = \mathcal{M}_r \cap \mathcal{M}(\Delta_{2\delta}) = \mathcal{M}_r \cap (\mathcal{M}_{2\delta} \cap W_{a_1,...,a_{\delta}}) = \mathcal{M}_r \cap W_{a_1,...,a_{\delta}}$.

3. Ideals in the local ring of the singularities of type A_{2d}

From this section, we only consider the curve singularities of type A_{2d} and freely use the notations introduced in the previous section. Let \mathcal{O} be the local ring $k[[t^2, t^{2d+1}]]$ for some $d \in \mathbb{N}$. The semi-group Γ of \mathcal{O} is generated by 2 and 2d+1. Note that any Γ -semi-module Δ contained in Γ is generated by at at most two elements $\alpha_1 := \min\{\Gamma(I)\}$ and $\alpha_2 := \min\{\Gamma(I) \setminus (\alpha_1 + \Gamma)\}$ as Γ -semi-module. We have $\alpha_2 < \alpha_1 + 2d + 1$. We use the notation $\mathcal{I}(\alpha_1, \alpha_2)$ instead of $\mathcal{I}(\Delta)$.

Lemma 19. For any element I of $\mathcal{I}(\alpha_1, \alpha_2)$, there exist two generators f_1 and f_2 such that

$$f_1 = t^{\alpha_1} + \sum_{j \in G(I), j > \alpha_1} c_j t^j, \quad f_2 = t^{\alpha_2}$$

Proof. Since $\Gamma(I)$ is generated by at most two positive integers as Γ -semi-module, the ideal I is generated by at most two elements. Let $g_1 = t^{\alpha_1} + \sum_{j>\alpha_1} c_j t^j$ and $g_2 = t^{\alpha_2} + \sum_{j>\alpha_2} d_j t^j$ be such generators. For any $j \in \Gamma(I)$, there exists an element $h_j = t^j$ + terms of higher order in I. Reducing g_1 and g_2 by h_j 's successively, we obtain the desired generators. \Box

By Lemma 19, normal forms of all ideals in $\mathcal{I}(\alpha_1, \alpha_2)$ are described as follows:

Proposition 20. If $\alpha_1 = 2p$ and $\alpha_2 = 2q + 1$, then we have

$$\mathcal{I}(2p, 2q+1) = \begin{cases} \left\{ \left(t^{2p}, t^{2d+1}\right)\right\} \text{ for } p \leq d = q, \\ \left\{ \left(t^{2p} + \sum_{i=d}^{q-1} a_i t^{2i+1}, t^{2q+1}\right) \middle| a_i \in k \right\} \text{ for } p \leq d < q, \\ \left\{ \left(t^{2p}, t^{2p+1}\right)\right\} \text{ for } d < p = q, \\ \left\{ \left(t^{2p} + \sum_{i=p}^{q-1} a_i t^{2i+1}, t^{2q+1}\right) \middle| a_i \in k \right\} \text{ for } d < p < q. \end{cases}$$

On the other hand, if $\alpha_1 = 2p + 1$ and $\alpha_2 = 2q$, then we have

$$\mathcal{I}(2p+1,2q) = \begin{cases} \left\{ \left(t^{2p+1}, t^{2p+2}\right) \right\} \text{ for } p \ge d, \ q = p+1, \\ \left\{ \left(t^{2p+1} + \sum_{i=p+1}^{q-1} a_i t^{2i}, t^{2q}\right) \middle| a_i \in k \end{cases} \text{ for } p \ge d, \ q > p+1. \end{cases}$$

Proof. Let I be a non-zero ideal in \mathcal{O} . For the set G(I), we have the following two cases: (Case 1): α_1 is even and α_2 is odd. Write $\alpha_1 = 2p$ and $\alpha_2 = 2q + 1$. It follows from the definitions of α_1 and α_2 that $p \leq q$. We easily see that $\Gamma(I) = \{2i | p \leq i \leq q\} \cup \{n | n \geq 2q + 1\}$ and c(I) = 2q. Since $2d + 1 \leq \alpha_2 \leq \alpha_1 + 2d + 1$ hold, we also obtain $d \leq q \leq p + d$. In terms of p, q and d, the set G(I) is described as follows:

$$(4) \qquad G(I) = \begin{cases} \{2i \mid 0 \le i \le p-1\} \text{ for } p \le d = q, \\ \{2i \mid 0 \le i \le p-1\} \cup \{2j+1 \mid d \le j \le q-1\} \text{ for } p \le d < q, \\ \{2i \mid 0 \le i \le d\} \cup [2d+1, 2p-1] \text{ for } d < p = q, \\ \{2i \mid 0 \le i \le d\} \cup [2d+1, 2p-1] \cup \{2j+1 \mid p \le j \le q-1\} \text{ for } d < p < q. \end{cases}$$

(Case 2): α_1 is odd and α_2 is even. Put $\alpha_1 = 2p + 1$ and $\alpha_2 = 2q$. It follows from the relations $2d + 1 \leq \alpha_1 < \alpha_2 \leq 2(p + d + 1)$ that $d \leq p < q \leq p + d + 1$. For $I \in \mathcal{I}(\alpha_1, \alpha_2)$, we have $\Gamma(I) = \{2i + 1 | p \leq i \leq q - 1\} \cup \{n | n \geq 2q\}$. For this case, the following four cases occur:

$$(5) \quad G(I) = \begin{cases} \{2i \mid 0 \le i \le d\} \text{ for } p = d, \ q = d+1, \\ \{2i \mid 0 \le i \le q-1\} \text{ for } p = d, \ q > d+1, \\ \{2i \mid 0 \le i \le d\} \cup [2d+1, 2p] \text{ for } p > d, \ q = p+1, \\ \{2i \mid 0 \le i \le d\} \cup [2d+1, 2p] \cup \{2j \mid p+1 \le j \le q-1\} \text{ for } p > d, \ q > p+1. \end{cases}$$

Our assertions follow from Lemma 19 with (4) and (5).

Lemma 21. If I belongs to $\mathcal{I}(2p, 2q+1)$ or $\mathcal{I}(2p+1, 2q)$, then its codimension r is given by (6)r = p + q - d.

Proof. This relation follows from Lemma 6 with (4) and (5).

The decomposition of \mathcal{I}_r is determined in terms of generators of $\Delta_{r,l}$.

Proposition 22. The sets \mathcal{I}_r for $1 \leq r \leq 2d$ are decomposed as follows:

(A):
$$1 \le r \le d$$
 and $r = 2s + 1$. $\mathcal{I}_{2s+1} = \bigcup_{l=0}^{s} \mathcal{I}(r+2l+1, r+2(d-l))$.
(B): $2 \le r \le d$ and $r = 2s$. $\mathcal{I}_{2s} = \bigcup_{l=0}^{s} \mathcal{I}(r+2l, r+2(d-l)+1)$.

For the cases where $d + 1 \le r \le 2d - 1$, the decompositions depend on d.

(C-i):
$$d+1 \le r \le 2d-1, r=2s+1 \text{ and } d=2h.$$

$$\mathcal{I}_{2s+1} = \left\{ \bigcup_{l=0}^{h-1} \mathcal{I}\left(r+2l+1, r+2(d-l)\right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{I}\left(r+2l, r+2(d-l)+1\right) \right\}$$

(C-ii): $d+1 \le r \le 2d-1$, r = 2s+1 and d = 2h+1.

$$\mathcal{I}_{2s+1} = \left\{ \bigcup_{l=0}^{h} \mathcal{I}\left(r+2l+1, r+2(d-l)\right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{I}\left(r+2l, r+2(d-l)+1\right) \right\}.$$

(D-i): $d + 1 \le r \le 2d$, r = 2s and d = 2h.

$$\mathcal{I}_{2s} = \left\{ \bigcup_{l=0}^{h} \mathcal{I}\left(r+2l, r+2(d-l)+1\right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h-1} \mathcal{I}\left(r+2l+1, r+2(d-l)\right) \right\}.$$

(D-ii): $d + 1 \le r \le 2d$, r = 2s and d = 2h + 1.

$$\mathcal{I}_{2s} = \bigg\{ \bigcup_{l=0}^{h} \mathcal{I}\left(r+2l, r+2(d-l)+1\right) \bigg\} \cup \bigg\{ \bigcup_{l=d-s}^{h} \mathcal{I}\left(r+2l+1, r+2(d-l)\right) \bigg\}.$$

Proof. We infer from the relation (6) in Lemma 21 that

(7)
$$\alpha_2 = -\alpha_1 + 2r + 2d + 1.$$

We first consider the case where $1 \le r \le d$. In this case, the positive integer α_1 must be even. Indeed, if not, then we have $\sharp G(I) \ge d+1$ since $\alpha_1 \ge d+1$. So we conclude that $r \ge d+1$ by Lemma 6. It is a contradiction. It follows from the definitions of α_1 and α_2 that

(8)
$$\alpha_1 \le \alpha_2 \le \alpha_1 + 2d + 1.$$

Note that $\mathcal{I}(\alpha_1, \alpha_2)$ is a component of \mathcal{I}_r if and only if α_1 and α_2 satisfy both of (7) and (8). According to r, the sets of all pairs (α_1, α_2) which satisfy (7) and (8) are determined as follows: (A) r = 2s + 1. $\{(r + 2l + 1, r + 2(d - l))\}_{l=0,\dots,s}$

(B) $r = 2s. \{(r+2l, r+2(d-l)+1)\}_{l=0,\dots,s}$

Next we consider the case in which $d + 1 \leq r \leq 2d$. According to r and d, we obtain the following sets of the pair (α_1, α_2) which satisfy (7) and (8): (C-i) r = 2s + 1 and d = 2h.

 $\{(r+2l+1,r+2(d-l))\}_{l=0,\ldots,h-1}\cup\{(r+2l,r+2(d-l)+1)\}_{d=s,\ldots,h}$

(C-ii) r = 2s + 1 and d = 2h + 1.

$$\{(r+2l+1,r+2(d-l))\}_{l=0,\ldots,h}\cup\{(r+2l,r+2(d-l)+1)\}_{d-s,\ldots,h}$$

(D-i) r = 2s and d = 2h.

$$\{(r+2l,r+2(d-l)+1)\}_{l=0,\ldots,h} \cup \{(r+2l+1,r+2(d-l))\}_{l=d-s,\ldots,h-1}$$

(D-ii) r = 2s and d = 2h + 1.

$$\{(r+2l,r+2(d-l)+1)\}_{l=0,\ldots,h}\cup\{(r+2l+1,r+2(d-l))\}_{l=d-s,\ldots,h}$$

Our assertions follow from Proposition 20 with the above datum.

Let $\Delta_{r,l}$ be a Γ -semi-module in (3) and take an element I from $\mathcal{I}(\Delta_{r,l})$. For the \mathcal{O} -sub-module $\varphi_r(I)$, define the set of orders of $\varphi_r(I)$ by $\Gamma(\varphi_r(I)) := \{ \operatorname{ord}_\tau(\overline{f}) | \overline{f} \in \varphi_r(I) \}$. It is clear that $\Gamma(\varphi_r(I))$ has a Γ -semi-module structure. Furthermore, if the Γ -semi-module $\Delta_{r,l}$ is generated by α_1 and α_2 , then $\Gamma(\varphi_r(I))$ is generated by $\alpha_1 - r$ and $\alpha_2 - r$. So we write $\mathcal{M}(\alpha_1 - r, \alpha_2 - r)$ instead of $\mathcal{M}(\Delta_{r,l})$ for such case. For each r, the decomposition (3) of \mathcal{M}_r follows from Proposition 6.

Corollary 23. The punctual Hilbert schemes \mathcal{M}_r $(1 \leq r \leq 2d)$ are decomposed as follows:

$$\begin{aligned} \mathbf{(A):} \ 1 &\leq r \leq d \ and \ r = 2s + 1. \ \mathcal{M}_{2s+1} = \bigcup_{l=0}^{n} \mathcal{M}_{2s+1} \left(2l+1, 2(d-l)\right). \\ \mathbf{(B):} \ 2 &\leq r \leq d \ and \ r = 2s. \ \mathcal{M}_{2s} = \bigcup_{l=0}^{s} \mathcal{M}_{2s} \left(2l, 2(d-l)+1\right). \\ \mathbf{(C-i):} \ d+1 &\leq r \leq 2d-1, \ r = 2s+1 \ and \ d = 2h. \\ \mathcal{M}_{2s+1} = \left\{ \bigcup_{l=0}^{h-1} \mathcal{M}_{2s+1} \left(2l+1, 2(d-l)\right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{M}_{2s+1} \left(2l, 2(d-l)+1\right) \right\} \\ \mathbf{(C-ii):} \ d+1 &\leq r \leq 2d-1, \ r = 2s+1 \ and \ d = 2h+1. \\ \mathcal{M}_{2s+1} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{2s+1} \left(2l+1, 2(d-l)\right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{M}_{2s+1} \left(2l, 2(d-l)+1\right) \right\}. \end{aligned}$$

(D-i): $d + 1 \le r \le 2d$, r = 2s and d = 2h.

$$\mathcal{M}_{2s} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{2s} \left(2l, 2(d-l) + 1 \right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h-1} \mathcal{M}_{2s} \left(2l+1, 2(d-l) \right) \right\}.$$

(D-ii): $d + 1 \le r \le 2d$, r = 2s and d = 2h + 1.

$$\mathcal{M}_{2s} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{2s} \left(2l, 2(d-l) + 1 \right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{M}_{2s} \left(2l+1, 2(d-l) \right) \right\}.$$

Remark 24. The punctual Hilbert scheme \mathcal{M}_1 consists of one point which corresponds to the maximal ideal of \mathcal{O} .

158

4. Proof of Main Theorems

In this section, we prove Theorem 1 and 2. To emphasis d, we use notations $\mathcal{M}_{d,r}$ and $\mathcal{M}_{d,r}(\alpha_1 - r, \alpha_2 - r)$ instead of \mathcal{M}_r and $\mathcal{M}_r(\alpha_1 - r, \alpha_2 - r)$.

Lemma 25. If the codimension r is odd (resp. even), then the component $\mathcal{M}_{d,r}(1, 2d)$ (resp. $\mathcal{M}_{d,r}(0, 2d+1)$) of the decomposition in Corollary 23 is the open dense subset of $\mathcal{M}_{d,r}$.

Proof. We should check our assertion for each case in Corollary 23. However, we only consider the case (D-ii) in Corollary 23. The other cases are treated in the same way. For r = 2s, set

$$\Lambda_1(l) := \{\underbrace{d, \dots, d}_{2l}, \underbrace{d-1, d-2, \dots, 2l+1, 2l}_{d-2l}\} \text{ for } l = 0, \dots, h,$$

$$\Lambda_2(l) := \{\underbrace{d, \dots, d}_{2l+1}, \underbrace{d-1, d-2, \dots, 2d-r+l}_{d-2l-1}\} \text{ for } l = d-s, \dots, h.$$

By Lemma 18, we see that

(9)
$$\mathcal{M}_{d,2s}(2l,2(d-l)+1) = \mathcal{M}_{d,2s} \cap W_{\Lambda_1(l)} \quad \text{for } l = 0,\dots,h, \\ \mathcal{M}_{d,2s}(2l+1,2(d-s)) = \mathcal{M}_{d,2s} \cap W_{\Lambda_2(l)} \quad \text{for } l = d-s,\dots,h.$$

The following inclusions also follows from Proposition 17:

(10)
$$W_{\Lambda_1(l+1)} \subset \overline{W_{\Lambda_1(l)}}, \ W_{\Lambda_2(l+1)} \subset \overline{W_{\Lambda_2(l)}}, \ W_{\Lambda_2(d-s)} \subset \overline{W_{\Lambda_1(0)}}$$

It follows from (9), (10) and (D-ii) in Corollary 23 that

$$\mathcal{M}_{d,2s} = \left\{ \mathcal{M}_{d,2s} \cap \left(\bigcup_{l=0}^{h} W_{\Lambda_{1}(l)} \right) \right\} \cup \left\{ \mathcal{M}_{d,2s} \cap \left(\bigcup_{l=d-s}^{h} W_{\Lambda_{2}(l)} \right) \right\}$$
$$\subset \mathcal{M}_{d,2s} \cap \left(\overline{W_{\Lambda_{1}(0)}} \cup \overline{W_{\Lambda_{2}(d-s)}} \right) = \mathcal{M}_{d,2s} \cap \overline{W_{\Lambda_{1}(0)}}$$
$$= \overline{\mathcal{M}_{d,2s}(0,2d+1)} \subset \mathcal{M}_{d,2s}.$$

Hence we conclude that $\mathcal{M}_{d,2s} = \overline{\mathcal{M}_{d,2s}(0,2d+1)}$.

Next we prove Proposition 3.

Proof of Proposition 3. We first prove (i) by constructing an isomorphism between $\mathcal{M}_{d,2s}$ and $\mathcal{M}_{d,2s+1}$. We have the following combination of the decomposition types of $\mathcal{M}_{d,2s}$ and $\mathcal{M}_{d,2s+1}$.

Decomposition type of $\mathcal{M}_{d,2s}$ (B)(B)(D-i)(D-ii)Decomposition type of $\mathcal{M}_{d,2s+1}$ (A)(C-i)(C-i)(C-ii)

We referred to Corollary 23 for the decomposition types. We only prove our assertion for the pair (D-ii) and (C-ii) here. The other cases can be treated in the same way. Since d is odd in this case, we put d = 2h + 1. It follows from Corollary 23 that

$$\mathcal{M}_{d,2s} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2s} \left(2l, 2(d-l) + 1 \right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{M}_{d,2s} \left(2l+1, 2(d-l) \right) \right\},$$
$$\mathcal{M}_{d,2s+1} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2s+1} \left(2l+1, 2(d-l) \right) \right\} \cup \left\{ \bigcup_{l=d-s}^{h} \mathcal{M}_{d,2s+1} \left(2l, 2(d-l) + 1 \right) \right\}.$$

159

By Propositions 20, 22, Corollary 23 and the definition of $\mathcal{M}_{d,r}$, we obtain the following explicit descriptions of the components of $\mathcal{M}_{d,2s}$ and $\mathcal{M}_{d,2s+1}$:

$$\mathcal{M}_{d,2s}(2l, 2(d-l)+1) = \left\{ \begin{cases} \left(\tau^{2l} + \sum_{j=d-s}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1} \right) & b_j \in k \\ \left(\tau^{2l} + \sum_{j=l}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1} \right) & b_j \in k \end{cases} \text{ for } l = 0 \dots, d-s, \end{cases} \right\}$$

$$\mathcal{M}_{d,2s}(2l+1,2(d-l)) = \left\{ \left\{ \left(\tau^{2l+1} + \sum_{j=l+1}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)} \right) \middle| b_j \in k \right\} \text{ for } l = d-s \dots, h-1, \\ \left\{ (\tau^d, \tau^{d+1}) \right\} \text{ for } l = h, \end{cases}$$

$$\mathcal{M}_{d,2s+1}(2l+1,2(d-l)) = \left\{ \begin{cases} \left(\tau^{2l+1} + \sum_{j=d-s}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)} \right) \ \left| b_j \in k \end{cases} \text{ for } l = 0, \dots, d-s-1, \\ \left\{ \left(\tau^{2l+1} + \sum_{j=l+1}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)} \right) \ \left| b_j \in k \right\} \text{ for } l = d-s \dots, h-1, \\ \left\{ (\tau^d, \tau^{d+1}) \right\} \text{ for } l = h, \end{cases} \right\}$$

$$\mathcal{M}_{d,2s+1}(2l,2(d-l)+1) = \left\{ \left(\tau^{2l} + \sum_{j=l}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1} \right) \middle| b_j \in k \right\} \text{ for } l = d-s \dots, h$$

Both of $\mathcal{M}_{d,2s}$ and $\mathcal{M}_{d,2s+1}$ have s+1 components. Furthermore, the numbers of coefficients involved in the elements of their components are $0, 1, \ldots, s-1, s$. So, for two components of $\mathcal{M}_{d,2s}$ and $\mathcal{M}_{d,2s+1}$ which have same number of coefficients, we can define a bijection by sending an element of $\mathcal{M}_{d,2s}$ to that of $\mathcal{M}_{d,2s+1}$ which has same coefficients. In this way, we obtain s+1bijections between the components of $\mathcal{M}_{d,2s}$ and $\mathcal{M}_{d,2s+1}$. It is clear that the union of them is an isomorphism from $\mathcal{M}_{d,2s}$ to $\mathcal{M}_{d,2s+1}$.

Next we prove (ii). If r = 1, then we have $\mathcal{M}_{d,1} = \{\text{one point}\}\$ for any $d \in \mathbb{N}$, as mentioned in Remark 24. So we consider the case where $r \geq 2$. For any $d, d' \in \mathbb{N}$, we can construct an isomorphism between $\mathcal{M}_{d,r}$ and $\mathcal{M}_{d',r}$ by the same argument in the proof of (i). \Box

The following fact is known:

Theorem 26 ([3]). We have dim $(\mathcal{M}_{d,2d}) = d$ for any $d \in \mathbb{N}$.

Proof of Theorem 1. The rationality of $\mathcal{M}_{d,r}$ is an immediate consequence of Lemma 25 and Proposition 16. The relation $\mathcal{M}_{d,r} \cong \mathcal{M}_{s,2s}$ also follows from (i) and (ii) in Proposition 3. Hence, we obtain dim $\mathcal{M}_{d,r} = \dim \mathcal{M}_{s,2s} = s$ by Theorem 26. \Box

Next we prove Theorem 2.

Proof of Theorem 2. The statement (i) already was mentioned in Remark 24. For (ii), the relation $\mathcal{M}_{2,2} \cong \mathcal{M}_{2,3} \cong \mathbb{P}^1$ proved in [3]. Hence it follows from (ii) in Proposition 3 that $\mathcal{M}_{d,2} \cong \mathcal{M}_{d,3} \cong \mathbb{P}^1$ for any d. The statement (ii) is proved. Since it was shown that

$$\operatorname{Sing}(\mathcal{M}_{2,4}) = \mathcal{M}_{2,2} \cap \mathcal{M}_{2,3} = \mathcal{M}_{2,1} = \{\text{one point}\}\$$

in [3], we only have to consider the cases where $d \geq 3$ to prove (iii). We may assume that r is even by (i) in Proposition 3. Moreover, since $\mathcal{M}_{d,r}$ is isomorphic to $\mathcal{M}_{s,2s}$ by Theorem 1, it is enough to prove our assertion for some PS variety $\mathcal{M}_{d,2d}$ ($d \in \mathbb{N}$). We divide the rest of the proof of (iii) into the following three cases:

Case 1 :
$$d = 3$$
, Case 2 : d is even and $d \ge 4$, Case 3 : d is odd and $d \ge 5$

Here we only consider Case 3. The other cases can be treated in a similar manner. Put d = 2h+1. By using Propositions 20, 22, Corollary 23 and the definition of $\mathcal{M}_{d,r}$, we have

$$\mathcal{M}_{d,2d-3} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2d-3} \left(2l+1, 2(d-l) \right) \right\} \cup \left\{ \bigcup_{l=2}^{h} \mathcal{M}_{d,2d-3} \left(2l, 2(d-l)+1 \right) \right\}$$
$$\mathcal{M}_{d,2d-2} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2d-2} \left(2l, 2(d-l)+1 \right) \right\} \cup \left\{ \bigcup_{l=1}^{h} \mathcal{M}_{d,2d-2} \left(2l+1, 2(d-l) \right) \right\}$$
$$\mathcal{M}_{d,2d-1} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2d-1} \left(2l+1, 2(d-l) \right) \right\} \cup \left\{ \bigcup_{l=1}^{h} \mathcal{M}_{d,2d-1} \left(2l, 2(d-l)+1 \right) \right\}$$
$$\mathcal{M}_{d,2d} = \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2d} \left(2l, 2(d-l)+1 \right) \right\} \cup \left\{ \bigcup_{l=0}^{h} \mathcal{M}_{d,2d} \left(2l+1, 2(d-l) \right) \right\}$$

where

$$\mathcal{M}_{d,2d-3}(2l+1,2(d-l)) = \left\{ \begin{cases} \left(\tau + \sum_{j=2}^{d-1} b_j \tau^{2j}, \tau^{2d}\right) \middle| b_j \in k \\ \left(\tau^{2l+1} + \sum_{j=l+1}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)}\right) \middle| b_j \in k \end{cases} \text{ for } l = 1..., h-1, \\ \left\{ (\tau^d, \tau^{d+1}) \right\} \text{ for } l = h, \end{cases} \right\}$$

$$\mathcal{M}_{d,2d-3}(2l,2(d-l)+1) = \left\{ \left(\tau^{2l} + \sum_{j=l}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1} \right) \middle| b_j \in k \right\} \text{ for } l = 2\dots, h,$$

$$\mathcal{M}_{d,2d-2}(2l,2(d-l)+1) = \left\{ \left\{ \left(1 + \sum_{j=1}^{d-1} b_j \tau^{2j+1}, \tau^{2d+1}\right) \middle| b_j \in k \right\} \text{ for } l = 0, \\ \left\{ \left(\tau^{2l} + \sum_{j=l}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1}\right) \middle| b_j \in k \right\} \text{ for } l = 1..., h, \end{cases} \right\}$$

$$\mathcal{M}_{d,2d-2}(2l+1,2(d-l)) = \left\{ \left\{ \left(\tau^{2l+1} + \sum_{j=l+1}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)} \right) \middle| b_j \in k \right\} \text{ for } l = 1..., h-1, \\ \left\{ (\tau^d, \tau^{d+1}) \right\} \text{ for } l = h, \end{cases}$$

$$\mathcal{M}_{d,2d-1}(2l+1,2(d-l)) = \left\{ \left\{ \left(\tau^{2l+1} + \sum_{j=l+1}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)} \right) \middle| b_j \in k \right\} \text{ for } l = 0 \dots, h-1, \\ \left\{ (\tau^d, \tau^{d+1}) \right\} \text{ for } l = h, \end{cases} \right\}$$

$$\mathcal{M}_{d,2d-1}(2l,2(d-l)+1) = \left\{ \left(\tau^{2l} + \sum_{j=l}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1} \right) \middle| b_j \in k \right\} \text{ for } l = 1...,h,$$

$$\mathcal{M}_{d,2d}(2l,2(d-l)+1) = \left\{ \left(\tau^{2l} + \sum_{j=l}^{d-l-1} b_j \tau^{2j+1}, \tau^{2(d-l)+1} \right) \, \middle| \, b_j \in k \right\} \text{ for } l = 0 \dots, h,$$

$$\mathcal{M}_{d,2d}(2l+1,2(d-l)) = \left\{ \left\{ \left(\tau^{2l+1} + \sum_{j=l+1}^{d-l-1} b_j \tau^{2j}, \tau^{2(d-l)} \right) \middle| b_j \in k \right\} \text{ for } l = 0 \dots, h-1, \\ \left\{ (\tau^d, \tau^{d+1}) \right\} \text{ for } l = h \right\}$$

It follows from the above descriptions of components that

$$\mathcal{M}_{d,2d} = (\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1}) \cup \mathcal{M}_{d,2d}(1,2d) \cup \mathcal{M}_{d,2d}(0,2d+1)$$

where

$$(\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1}) \cap \mathcal{M}_{d,2d}(1,2d) = \emptyset,$$

$$(\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1}) \cap \mathcal{M}_{d,2d}(0,2d+1) = \emptyset,$$

$$\mathcal{M}_{d,2d}(1,2d) \cap \mathcal{M}_{d,2d}(0,2d+1) = \emptyset.$$

Furthermore, $\mathcal{M}_{d,2d}(1,2d)$ and $\mathcal{M}_{d,2d}(0,2d+1)$ are affine spaces by Corollary 14, we must have $\operatorname{Sing}(\mathcal{M}_{d,2d}) = \operatorname{Sing}(\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1})$. Since both of

$$\mathcal{M}_{d,2d-2} \setminus (\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1}) = \mathcal{M}_{d,2d-2}(0,2d+1),$$

$$\mathcal{M}_{d,2d-1} \setminus (\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1}) = \mathcal{M}_{d,2d-1}(1,2d)$$

are affine spaces again, we conclude that $\operatorname{Sing}(\mathcal{M}_{d,2d}) = \mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1}$. Finally, the relation $\mathcal{M}_{d,2d-2} \cap \mathcal{M}_{d,2d-1} \cong \mathcal{M}_{d,2d-3}$ can be shown by constructing an isomorphism. For the construction, refer to the proof of (i) in Proposition 3. \Box

References

- [1] Ph. Griffiths, J. Harris: Principles of Algebraic Geometry. Wiley, New York, (1978).
- [2] T. Kawai: Abelian vortices on nodal and cuspidal curves. J. High Energy Phys. 11(2009)11.
 DOI: 10.1088/1126-6708/2009/11/111
- G. Pfister, J.H.M. Steenbrink: Reduced Hilbert schemes for irreducible curve singularities. J. Pure and Applied Algebra. 77, 103-116, (1992). DOI: 10.1016/0022-4049(92)90033-C
- [4] Z. Ran: A note on Hilbert schemes of nodal curves. J. Algebra. 292 no.2, 429-446, (2005). DOI: 10.1016/j.jalgebra.2005.06.028
- [5] Z. Ran: Geometry of nodal curves. Compos. Math. 141 no 5, 1191-1212, (2005). DOI: 10.1112/S0010437X05001466
- [6] C. Rego: The compactified Jacobian. Ann. Sci. Éc. Norm. Supér. IV. Sér. 13, 211-223, (1980).
- [7] Y. Sōma, M. Watari: Punctual Hilbert schemes for irreducible curve singularities of types E₆ and E₈. J. Singularities. 8, 135-145, (2014).

Yoshiki Sōma, Chōfu Minami Metroporitan High School, 6-2-1 Tamagawa, Chōfushi Tokyo 182-0025 Japan

MASAHIRO WATARI, DEPARTMENT OF INTEGRATED ARTS AND SCIENCE, OKINAWA NATIONAL COLLEGE OF TECHNOLOGY, 905 AZA HENOKO, NAGO OKINAWA 905-2192 JAPAN

E-mail address: watari01@okinawa-ct.ac.jp

FAMILIES OF DISTRIBUTIONS AND PFAFF SYSTEMS UNDER DUALITY

FEDERICO QUALLBRUNN

ABSTRACT. A singular distribution on a non-singular variety X can be defined either by a subsheaf $D \subseteq TX$ of the tangent sheaf, or by the zeros of a subsheaf $D^0 \subseteq \Omega^1_X$ of 1-forms, that is, a Pfaff system. Although both definitions are equivalent under mild conditions on D, they give rise, in general, to non-equivalent notions of flat families of distributions. In this work we investigate conditions under which both notions of flat families are equivalent. In the last sections we focus on the case where the distribution is *integrable*, and we use our results to generalize a theorem of Cukierman and Pereira.

1. INTRODUCTION

Among the several motivations for studying moduli problems in geometry, there is the study of the topological and geometrical properties of differential equations on manifolds. Such problems are studied in the works of R. Thom, V. Arnold, and Kodaira-Spencer in the 1950's and were an influence in the work of Zariski about deformation of singularities; we refer to the introduction and Section 3.10 of [7] for a historical account on the subject. Algebraic geometric techniques were used in the study of (integrable) differential equations on non-singular algebraic varieties at least since Jouanolou [13], moduli and deformation problems of integrable differential equations were studied by Gomez-Mont in [9]. Since then, much has been done in the study of the geometry of integrable 1-forms on algebraic varieties, especially describing singularities of the differential equations they define (see, for instance [15]). Also, much has been done in the determination of irreducible components of the moduli space of integrable 1-forms (e.g.: [5, 6, 14] and references therein).

Many of these works were carried out in a 'naïve' way, without concern for representability of functors or formalization of deformation problems. When trying to formalize the moduli problem for integrable differential equations in non-singular algebraic varieties, an issue appears at the very beginning: although its equivalent to describe a distribution by its tangent sheaf or by the sheaf of 1-forms that vanishes on it, these two descriptions give rise to inequivalent notions of a flat family of distributions. Thus we are led to two different moduli problems, one for (involutive) sub-sheaves of the tangent sheaf, and the other for (integrable) sub-sheaves of the sheaf of 1-forms. In the present work, we prove in Proposition 6.3 and Proposition 6.5 that, under suitable assumptions, these two moduli problems are representable and, in Corollary 7.15, that their irreducible components are moreover birationally equivalent. Using this we can generalize previous theorems in the literature such as Theorem 1 of [5] and the theorems of [16, 17].

In order to better explain our results, we now introduce some notation and definitions. Let X be a non-singular projective algebraic variety. A non-singular k-dimensional distribution D on a non-singular variety X consists of a k-dimensional subspace $D_x \subseteq T_x X$ of the tangent space of X at x varying continuously with x. This notion can be formalized by saying that a k-dimensional

Key words and phrases. Coherent Sheaves, Flat Families, Algebraic Foliations, Moduli Spaces, Kupka Singularities.

distribution is a rank k sub-bundle of the tangent bundle $D \hookrightarrow TX$, and taking $D_x \subseteq T_x X$ to be $D \otimes k(x)$. Equivalently, we can say that a k-dimensional distribution is determined by a sub-bundle $I_D \hookrightarrow \Omega^1_X$, and take D_x to be $\{v \in T_x X \mid \omega(v) = 0, \forall \omega \in I_D \otimes k(x)\}$.

When X is an algebraic variety, it is often the case that there are no algebraic sub-bundles of TX or Ω_X^1 of a given rank, i.e., there are no non-singular distributions. Nevertheless, the definition can be readily generalized to allow $D \subseteq TX$ and $I_D \subseteq \Omega_X^1$ to be subsheaves. In this way, we describe distributions D which are k-dimensional on a dense open subset, but may present singularities along proper subvarieties. Again, a singular distribution is equivalently defined either by a subsheaf $D \subseteq TX$ or by its annihilator $I_D \subseteq \Omega_X^1$. However, as was already observed by Pourcin in [16], when studying families of distributions parametrized by a base scheme S, it may happen that while a family of distributions $D \subseteq T_S(X \times S)$ is flat (in the sense that the quotient $T_S(X \times S)/D$ is a flat sheaf over S), its annihilator $I_D \subseteq \Omega_{X|S}^1$ may not be flat. This gives us two different notions of flat family of (singular) distributions, and therefore two different moduli problems for them.

In this work, we prove that the above two notions of flatness coincide as long as the singular set of the distribution (endowed with a convenient scheme structure) is also flat over S (Proposition 7.6 and Theorem 7.14). We focus on the case of *integrable* distributions; in Proposition 6.3, we give constructions for the moduli space Inv^X of involutive (in the sense of the Frobenius theorem) subsheaves of TX and, in Proposition 6.5, for the moduli space iPf^X of integrable subsheaf of Ω^1_X and conclude in Corollary 7.15 that taking annihilators defines a rational map between Inv^X and iPf^X that is a birational equivalence in each irreducible component of Inv^X . Moreover we give in Theorem 8.13 a sufficient criterion in terms of the singularity of the foliation to know when an involutive subsheaf $T\mathcal{F} \subseteq TX$ represents a point in the dense open set $U \subseteq \text{Inv}^X$ where taking duals gives an isomorphism with an open set $V \subseteq \text{iPf}^X$. Using this criterion, we can generalize the main theorem of [5] by specializing our results to the case where $X = \mathbb{P}^n$ and $T\mathcal{F} \cong \bigoplus_i \mathcal{O}_{\mathbb{P}^n}(d_i)$.

In Section 2, we show a particular example of a flat family of Pfaff systems being dual to a non-flat family of distributions.

In Section 3, we treat some preliminary general notions on sheaves and criterion for flatness that will be useful later.

In Section 4, we study the effect of applying the functor $\mathcal{H}om(-, \mathcal{O}_X)$ to a short exact sequence of sheaves. We also include in this section some observations on exterior powers, which are relevant for the study of distributions of codimension higher than 1.

Section 5 consists mainly of definitions of families of distributions and Pfaff systems, and related notions.

In Section 6, the construction of the moduli spaces of involutive distributions and integrable Pfaff systems is given as subschemes of certain Quot schemes.

In Section 7, the main results of the paper are proved. First the singular scheme of a family of distributions is defined, as well as the analogous notion for family of Pfaff systems and it is proved that the singular scheme of a family of distributions is the same as the singular scheme of its dual family (which is a family of Pfaff systems). In Section 7.1, the codimension 1 version of the main result is proved: if the singular scheme of a flat family of codimension 1 Pfaff systems is itself flat, then the dual family is flat as well. In Section 7.2, an analogous statement is proved for arbitrary codimension. In this case, however, flatness of the singular scheme is not enough to assure flatness of the dual family. To obtain a valid criterion, we define a stratification of the singular scheme, if each stratum is flat over the base, we can assure the dual family will be flat as well.

FEDERICO QUALLBRUNN

In Section 8, we give a sufficient condition to know when the singular scheme of a family of codimension 1 foliations is flat over the base. This condition is related to two of the betterstudied types of singularities of foliations, the Kupka singularities and the Reeb singularities. We prove that, if the singularities of a foliation given by a distribution $T\mathcal{F} \hookrightarrow TX$ are only of these two types, then every flat family

$$0 \to T\mathcal{F}_S \to T_S(X \times S) \to N_\mathcal{F} \to 0,$$

such that there is an $s \in S$ with $T\mathcal{F}_s = T\mathcal{F}$, is such that $\operatorname{sing}(\mathcal{F})$ is flat in a neighborhood of s.

In Section 9, we apply the theorem of Section 8 to recover the main result of [5] as a special case of Theorem 7.8, where $X = \mathbb{P}^n$ and $T\mathcal{F}$ splits as direct sum of line bundles.

The content of this work is part of the author's doctoral thesis, for the degree of Doctor de la Universidad de Buenos Aires, under the advice of Fernando Cukierman. The author was supported by a doctoral grant of CONICET. The author is grateful to Fernando Cukierman and Fernando Sancho de Salas for useful ideas and to Aroldo Kaplan, Alicia Dickenstein and the anonymous referee for a thorough reading of this paper and useful suggestions. The author is also grateful to Universidad de Buenos Aires, where the work was made.

2. An example

Set $X = \mathbb{C}^4$, $S = \mathbb{C}^2$ and consider the family of 1-forms on X

$$\omega_{(t,s)} = xdx + ydy + t \cdot zdz + s \cdot wdw,$$

 $\omega_{(t,s)} = x dx + y dy + t \cdot z dz$ parametrized by $(t,s) \in S = \mathbb{C}^2,$ and the related family

$$I = I_{(t,s)} = \mathcal{O}_{X \times S} \cdot (\omega_{(t,s)}) \subseteq \Omega^1_{X \times S \mid S}$$

of Pfaff systems on $X = \mathbb{C}^4$.

It is easy to see that the sheaf I is flat over $S = \mathbb{C}^2$. On the other hand, if we look at the family of distributions this Pfaff systems define, we will find that it is not flat. Indeed, let Ddenote the annihilator of I:

$$D = \operatorname{Ann}(I) \subseteq T(X \times S).$$

To see that D is not flat over S, we can use Artin's criterion for flatness [2, Corollary to Proposition 3.9].

Proposition 2.1 (Artin's criterion for flatness.). Let (A, \mathcal{M}) and (B, \mathcal{N}) be local rings, $B \to A$ a flat morphism of local rings, and M a finitely generated A-module. Suppose we have generators of M, $M = (f_1, \ldots, f_r)$. Let $\overline{f_i}$ be the class of f_i on $M \otimes_B B/\mathcal{N}$. Then M is flat over B if and only if every relation among $(\overline{f_1}, \ldots, \overline{f_r})$ lifts to a relation among (f_1, \ldots, f_r) .

Note that, while this is not exactly the same context in which Artin states the proposition, his proof extends mutatis mutandi to this setting.

Now observe that $D = \operatorname{Ann}(I)$ has 6 generators:

$$f_1 = y \frac{\partial}{\partial x} - x \frac{\partial}{\partial y}, \quad f_2 = tz \frac{\partial}{\partial x} - x \frac{\partial}{\partial z}, \quad f_3 = sw \frac{\partial}{\partial x} - x \frac{\partial}{\partial w},$$

$$f_4 = tz \frac{\partial}{\partial y} - y \frac{\partial}{\partial z}, \quad f_5 = sw \frac{\partial}{\partial y} - y \frac{\partial}{\partial w}, \quad f_6 = sw \frac{\partial}{\partial z} - tz \frac{\partial}{\partial w}.$$

These generators have, in turn, numerous relations between them, too many to list completely here. We note, however, that all relations are generated by relations of the form

$$af_i + bf_j + cf_k = 0,$$

where a, b and c are in the ideal $(x, y, tz, sw) \subseteq \mathcal{O}_{X \times S}$.

We apply Artin's criterion with $A = \mathcal{O}_{X \times S,(0,0)}$, $B = \mathcal{O}_{S,0}$ and $M = D_{(0,0)}$. In this case, the $\overline{f_i}$'s consist of evaluating the f_i on $(x, y, z, w, 0, 0) \in \mathbb{C}^4 \times (0, 0)$. We get in this way a particular relation on $M \otimes_B B/\mathcal{N}$, namely $\overline{f_6} = 0$. It is then easy to see that this relation does not lift to a relation on (f_1, \ldots, f_6) .

This example shows that there is, in general, no morphism between the moduli spaces $\operatorname{Quot}(X, TX)$ and $\operatorname{Quot}(X, \Omega_X^1)$ induced by taking annihilators. We will show later that taking annihilators does define a *birational* equivalence between irreducible components of $\operatorname{Quot}_X(TX)$ and those of $\operatorname{Quot}_X(\Omega_X^1)$.

The example also hints that a key aspect to understand whether flatness is preserved or not under duality is to look at how the singular set varies. Indeed, we can see that D is flat over $S \setminus \{(0,0)\}$. For each point $(t,s) \in S \setminus \{(0,0)\}$, the singular set of $I|_{(t,s)}$ (i.e., the set of points in $p \in X$ such that $\omega_p = 0$) is just the origin of $\mathbb{C}^4 = X$. On the other hand, the singular set of $I|_{(0,0)}$ is the plane $\{(0,0,z,w): z, w \in \mathbb{C}\} \subset \mathbb{C}^4$. So D ceases to be flat exactly when the singular set of I ceases to be flat as well. This illustrates the typical behavior of singular codimension 1 Pfaff systems and the distributions they determine. In the arbitrary codimension case, the situation is a bit more subtle, but the singular set still plays a decisive role. Of course, to make sense out of this, we have to define a scheme structure on the singular set of a Pfaff system. This will be done in the course of the present work.

3. Preliminaries

Here we gather known facts of algebraic geometry that will be used later. We include proofs of some of these facts for lack (to the author's best knowledge) of a better reference.

3.1. Reflexive sheaves and Serre's property S_2 . Property S_2 can be viewed as an algebraic analog of Hartog's theorem on complex holomorphic functions. For this reason, it will be extremely useful to us, for it will allow us to conclude global statements on sheaves that hold, a priori, for the restriction of these sheaves to (suitably large) open sets. Here we recall some known facts about sheaves with the S_2 property, and sheaves with the *relative* S_2 property as defined in [3].

Definition 3.1. Let $p: X \to T$ be a morphism of schemes and, for each point $x \in X$, let $d_T(x)$ equal the codimension of x in its fiber over T. We say that a sheaf \mathscr{F} on X satisfies the *relative* Serre condition S_k with respect to p if and only if

$$\operatorname{depth}\mathscr{F}_x \ge \min(k, d_T(x)),$$

for all $x \in X$.

The proof of the next proposition works exactly as in the non-relative version.

Proposition 3.2. Let $p: X \to T$ be a morphism of noetherian schemes and \mathscr{F} a torsion-free coherent sheaf with relative property S_2 with respect to p. Let $Y \subset X$ be a closed subset such that $d_T(Y) \ge 2$. Then the restriction map $\rho: \Gamma(X, \mathscr{F}) \to \Gamma(X \setminus Y, \mathscr{F})$ is an isomorphism.

Corollary 3.3. Let $p: X \to T$ be a morphism of noetherian schemes and \mathscr{F} a torsion-free coherent sheaf with property S_2 with respect to p. Let $Y \subset X$ be a closed subset such that $d_T(Y) \geq 2$. Let $U = X \setminus Y$ and let $j: U \to X$ be the inclusion. Then $\mathscr{F} = j_*(\mathscr{F}|_U)$.

This corollary motivates the following definition, which is the relative analogue of a notion due to Grothendieck [10, 5.10].

FEDERICO QUALLBRUNN

Definition 3.4. Let $p: X \to T$ be a morphism of noetherian schemes and \mathscr{F} a coherent sheaf. If, for each closed subset $Y \subset X$ such that $d_T(Y) \ge 2$, with $U = X \setminus Y$ and $j: U \to X$ the inclusion, the natural map

$$\rho_U:\mathscr{F}\to j_*(\mathscr{F}|_U)$$

is an epimorphism, then we say \mathscr{F} is $Z^{(2)}$ -closed relative to p; if ρ_U is an isomorphism, we say that it is $Z^{(2)}$ -pure relative to p.

Proposition 3.5 ([12, Proposition 1.7]). Let X be a quasi-projective integral scheme. A coherent sheaf \mathscr{F} is reflexive if and only if it can be included in an exact sequence

$$0 \to \mathscr{F} \to \mathscr{E} \to \mathscr{G} \to 0.$$

where \mathcal{E} is locally free and \mathcal{G} is torsion-free.

Corollary 3.6. Under the above circumstances, the dual of a coherent sheaf is always reflexive.

Proposition 3.7 (c.f.:[12, Theorem 1.9]). Let $p: X \to T$ be a morphism of noetherian schemes with normal integral fibers, and \mathscr{F} a coherent sheaf on X. Then, if \mathscr{F} is reflexive, it has relative property S_2 with respect to p.

Proof. The statement being local, we can assume X is quasi-projective. Given a reflexive sheaf \mathscr{F} , we take an exact sequence

$$0 \to \mathscr{F} \to \mathscr{L} \to \mathscr{G} \to 0,$$

with \mathscr{L} locally free and \mathscr{G} torsion-free. Since p have normal fibers, \mathcal{O}_X satisfies relative S_2 with respect to p, and so does \mathscr{L} , being locally free. Let $x \in X$ be a point of relative dimension ≥ 2 with respect to its fiber $X_{p(x)}$. Then depth $\mathscr{L}_x \geq 2$ by S_2 and, as \mathscr{G} is torsion-free, depth $\mathscr{G}_x \geq 1$. This in turn implies depth $\mathscr{F}_x \geq 2$.

3.2. Support of a sheaf, zeros of a section. Recall that, given a quasi-coherent sheaf \mathscr{F} on a scheme X, we define the *support* of \mathscr{F} , $\operatorname{supp}(\mathscr{F})$ as the closed sub-scheme defined by the ideal sheaf given locally by

$$\mathcal{I}(\mathscr{F})_x := \operatorname{Ann}(\mathscr{F}_x) \subset \mathcal{O}_{X,x}$$

We have the following useful characterization of the support of a sheaf in terms of a universal property:

Proposition 3.8. The support of a sheaf \mathscr{F} represents the functor

$$\begin{aligned} \mathbf{S}_{\mathscr{F}} : Sch &\longrightarrow Sets \\ T &\mapsto \{f \in \hom(T, X) : \operatorname{Ann}(f^*\mathscr{F}) = 0 \subset \mathcal{O}_T \}. \end{aligned}$$

Proof. A morphism $f: T \to X$ factors through $\operatorname{supp}(\mathscr{F})$ if and only if the map

$$f^{\sharp}: f^{-1}\mathcal{O}_X \to \mathcal{O}_T$$

factors through $f^{-1}(\mathcal{O}_X/\operatorname{Ann}(\mathscr{F}))$. But this happens if and only if $f^{-1}(\operatorname{Ann}(\mathscr{F})) = 0$.

On the other hand, we have the equality

$$\operatorname{Ann}(f^*\mathscr{F}) = \mathcal{O}_T \cdot f^{-1}(\operatorname{Ann}(\mathscr{F}));$$

indeed we may check this in every localization at any point $p \in T$, so if $t \in \operatorname{Ann}(f^*\mathscr{F})_p$ in particular t annihilates every element of the form $m \otimes 1 \in \mathscr{F}_x$, so $t = f^{-1}(x)t'$ where $x \in \operatorname{Ann}(\mathscr{F})_{f(p)}$.

So
$$\operatorname{Ann}(f^*\mathscr{F}) = 0$$
 if and only if $f^{-1}(\operatorname{Ann}(\mathscr{F})) = 0$ and we are done.

168

In other words, we just proved that $\operatorname{supp}(\mathscr{F})$ is the universal scheme with the property that $f^*\mathscr{F}$ is not a torsion module. This simple observation will be very useful when discussing the scheme structure on the singular set of a foliation.

A special case of support of a sheaf is the scheme-theoretic image of a morphism. Remember that the scheme-theoretic image of a morphism $f: X \to Y$ is the sub-scheme $\operatorname{supp}(f_*\mathcal{O}_X) \subseteq Y$.

Now we turn our attention to sections and their zeros. So let X be a scheme and \mathscr{E} a locally free sheaf. Having a global section $s \in \Gamma(X, \mathscr{E})$ is the same as having a morphism (that, by abuse of notation, we also call s)

$$s: \mathcal{O}_X \longrightarrow \mathscr{E}.$$

Now, $s: \mathcal{O}_X \to \mathscr{E}$ defines a dual morphism

$$s^{\vee}: \mathscr{E}^{\vee} \longrightarrow \mathcal{O}_X^{\vee} = \mathcal{O}_X.$$

Definition 3.9. We define the zero scheme Z(s) of the section s as the closed sub-scheme of X defined by the ideal sheaf $\operatorname{Im}(s^{\vee}) \subseteq \mathcal{O}_X$.

We'll apply this definition in the well-behaved situation where \mathcal{O}_X (and therefore \mathscr{E}) is torsion-free.

Proposition 3.10. Let \mathscr{E} be a locally free sheaf on X and $s \in \Gamma(X, \mathscr{E})$ a global section. The scheme Z(s) represents the functor

$$\begin{array}{rcl} \mathbf{Z}_s:Sch & \longrightarrow & Sets \\ & T & \mapsto & \{f\in \hom(T,X)\colon s\otimes 1=0\in \Gamma(T,f^*\mathscr{E})\}. \end{array}$$

Proof. A morphism $f: T \to X$ factors through Z(s) if and only if the map

$$(\mathscr{E}^{\vee}) \otimes \mathcal{O}_T \xrightarrow{s^{\vee} \otimes 1} \mathcal{O}_T$$

is identically 0. Beign locally free, we have

$$\mathscr{E}^{\vee} \otimes \mathcal{O}_T = \mathscr{H}om(\mathscr{E}, \mathcal{O}_X) \otimes \mathcal{O}_T \cong \mathscr{H}om(\mathscr{E} \otimes \mathcal{O}_T, \mathcal{O}_T).$$

So then we have

$$(f^*\mathscr{E})^{\vee} \xrightarrow{s^{\vee} \otimes 1} \mathcal{O}_T$$

is identically 0, as $f^*\mathscr{E}$ is locally free over T; this means $s \otimes 1 = 0 \in \Gamma(T, f^*\mathscr{E})$.

3.3. A criterion for flatness. For lack of a better reference, we provide here a criterion that will become handy when dealing with both reduced and non-reduced base schemes over an algebraically closed field.

Lemma 3.11. Let A be a ring of finite type over an algebraically closed field k, \mathcal{M} a maximal ideal in A, and $f \in \mathcal{M}^n \setminus \mathcal{M}^{n+1}$. Then there is a morphism $\psi : A \to k[T]/(T^{n+1})$ such that $\psi^{-1}((T)) = \mathcal{M}$ and $\psi(f) \neq 0$.

Proof. Fix a presentation $A \cong k[y_1, \ldots, y_r]/I$. By the Nullstelensatz, we can assume that $\mathcal{M} = (x_1, \ldots, x_r)$, where x_i is the class of $y_i \mod I$. Write the class of f in $\mathcal{M}^n/\mathcal{M}^{n+1}$ as

$$\overline{f} = \sum_{|\alpha|=n} a_{\alpha} \overline{x}^{\alpha} \in \mathcal{M}^n / \mathcal{M}^{n+1}$$

where $\alpha = (\alpha_1, \ldots, \alpha_r)$ and $\overline{x}^{\alpha} = (\overline{x_1}^{\alpha_1}, \ldots, \overline{x_r}^{\alpha_r})$.

As $f \notin \mathcal{M}^{n+1}$, the polynomial $q(y_1, \ldots, y_r) := \sum_{|\alpha|=n} a_{\alpha} y^{\alpha}$ is not in I. Now, k being algebraically closed, there is an r-tuple $(\lambda_1, \ldots, \lambda_r) \in k^r$ such that $p(\lambda_1, \ldots, \lambda_r) = 0$ for every $p \in I$ and $q(\lambda_1, \ldots, \lambda_r) \neq 0$.

Finally we can define $\psi: A \to k[T]/(T^{n+1})$ as follows:

$$\psi(x_i) = \lambda_i T.$$

The morphism is well-defined because $p(\lambda_1, ..., \lambda_r) = 0$ for every $p \in I$, moreover $\psi^{-1}(T) = \mathcal{M}$, and $\psi(f) = q(\lambda_1, ..., \lambda_r)T^n \neq 0$.

Proposition 3.12. Let $f: X \to Y$ a projective morphism between schemes of finite type over an algebraically closed field, \mathscr{F} a coherent sheaf over $X, x \in X$ a point, and y = f(x). Then \mathscr{F}_x is f-flat if and only if the following conditions hold:

(1) For every discrete valuation ring A' and every morphism $\mathcal{O}_{Y,y} \to A'$, the following holds: Taking the pull-back diagram



the $\mathcal{O}_{X'}$ -module $\mathscr{F}' = \mathscr{F} \otimes \mathcal{O}_{Y'}$ is f'-flat at every point $x' \in X'$ lying over x.

(2) For every $n \in \mathbb{N}$ and every morphism $\mathcal{O}_{Y,y} \to k[T]/(T^{n+1})$, if we take the diagram analogous to the one above (with $k[T]/(T^{n+1})$ instead of A'), then the $\mathcal{O}_{X'}$ -module $\mathscr{F}' = \mathscr{F} \otimes \mathcal{O}_{Y'}$ is f'-flat at every point $x' \in X'$ lying over x.

Proof. Clearly Conditions 1 and 2 are necessary. Suppose then that 1 and 2 are satisfied.

Take the flattening stratification (see [8, Section 5.4.2]) of Y with respect to \mathscr{F} , $Y = \coprod_P Y_P$. As Condition 1 is satisfied for \mathscr{F} over Y, so it is satisfied for $\iota^*\mathscr{F}$ over Y_{red} , where $\iota: Y_{\text{red}} \to Y$ is the closed immersion of the reduced structure. Then, by the valuative criterion for flatness of [10, 11.8], $\iota^*\mathscr{F}$ is flat over Y_{red} , so by the universal property of the flattening stratification, there is a factorization



As Y_{red} and Y share the same underlying topological set, the above factorization is telling us that the flattening factorization consist on a single stratum Y_P and that $Y_{\text{red}} \rightarrow Y_P$ is a closed immersion.

Assume, by way of contradiction, $Y_P \subsetneq Y$; then there is an affine open sub-scheme $U \subseteq Y$ such that $V = Y_P \cap U \neq U$. Now take the coordinate rings k[U] and k[V] and the morphism between them induced by the inclusion $\phi : k[U] \twoheadrightarrow k[V]$. Let's take $f \in k[U]$ such that $\phi(f) = 0$. By Lemma 3.11, there exists, for some $n \in \mathbb{N}$, a morphism $\psi : k[U] \to k[T]/(T^{n+1})$ such that $\psi(f) \neq 0$; so ψ doesn't factorize through ϕ .

On the other hand, let $Z = \operatorname{Spec}(k[T]/(T^{n+1}))$ and $g: Z \to Y$ be the morphism induced by ψ . As Condition 2 is satisfied, the pull-back $g^* \mathscr{F}$ is flat over $Z = \operatorname{Spec}(k[T]/(T^{n+1}))$. So, by the universal property of the flattening stratification, g factors as



contradicting the statement of the above paragraph, thus proving the proposition.

170

Note that the hypotheses of this property on X and Y (aside from reducedness) are quite stronger than the ones of the original theorem of Grothendieck (the valuative criterion for flatness in [10]); such is the price we have paid to allow a criterion for possibly non-reduced schemes. The price paid is okay with us anyway, considering that we will work mostly with schemes of finite type over \mathbb{C} .

Next we provide a criterion for a $k[T]/(T^{n+1})$ -module to be flat.

Proposition 3.13. Let $A = k[T]/(T^{n+1})$ and M an A-module. Then M is flat if and only if, for every $m \in M$ such that $T^n \cdot m = 0$, there exists $m' \in M$ such that $m = T \cdot m'$.

Proof. Flatness of M is equivalent to the injectivity of the map $M \otimes I \to M$ for every ideal $I \subset A$ (see e.g.:[11, IV.1]). In this case, there are finitely many ideals:

$$\mathcal{M} = (T), \ \mathcal{M}^2, \dots, \ \mathcal{M}^n$$

If M is flat, it is easy to see that the second condition in our statement hold.

Suppose that, for every $m \in M$ such that $T^n \cdot m = 0$, there exists $m' \in M$ such that $m = T \cdot m'$. Let $a \in M \otimes \mathcal{M}^{n-i}$ be in the kernel of $M \otimes \mathcal{M}^{n-i} \to M$. When i = 0, we have $a = m \otimes T^n$, and m is such that $T^n \cdot m = 0$; so, by hypothesis, $m = T \cdot m'$ and then $m \otimes T^n = m' \otimes T^{n+1} = 0$.

When i > 0, we have $a = \sum_{j=n-i}^{i} m_j \otimes T^j$, so $T^i \cdot a = m_{n-i} \otimes T^n \in M \otimes \mathcal{M}^n$. By hypothesis, $m_{n-i} = T \cdot m'$. So $a \in M \otimes \mathcal{M}^{n-i+1}$ and we are done by induction. \Box

The following will be useful in the study of foliations of codimension greater than 1.

Proposition 3.14. Let $p: X \to S$ a projective morphism between schemes of finite type over an algebraically closed field $k, f: S \to Y$ another morphism, with Y of finite type over k, and \mathscr{F} a coherent sheaf over X. Take a stratification $\coprod_i S_i \subseteq S$ of S such that $\mathscr{F}|_{S_i} := \mathscr{F} \otimes_S \mathcal{O}_{S_i}$ is flat for all i. If the composition $\coprod_i S_i \hookrightarrow S \xrightarrow{f} Y$ is a flat morphism, then \mathscr{F} is flat over Y. *Proof.* Invoking Proposition 3.12 we can, after applying base change, reduce to the case where

Y is either the spectrum of a DVR or $Y = \operatorname{Spec}(k[T]/(T^{n+1}))$. (i) Case $Y = \operatorname{Spec}(A)$ with A a DVR. Suppose there is, for some point $x \in X$, a section $s \in \mathscr{F}_x$ that is of torsion over A. Consider $Z = \operatorname{supp}_S(s) \subseteq S$, the support of s over S, that is the support of s as an element of \mathscr{F}_x considered as an $\mathcal{O}_{S,p(x)}$ -module. Now take any stratum

the support of s as an element of \mathscr{F}_x considered as an $\mathscr{C}_{S,p(x)}$ -module. Now take any stratum S_i and suppose $Z \cap S_i \neq \emptyset$. Then there is a section of the pullback \mathscr{F}_{S_i} that is of torsion over A. But \mathscr{F}_{S_i} is flat over S_i which is in turn flat over A, so \mathscr{F}_{S_i} is flat and $Z \cap S_i$ must be empty for every stratum S_i , i.e., s = 0.

(ii) Case $Y = \text{Spec}(k[T]/(T^{n+1}))$. One can essentially repeat the argument above, now taking the section s to be such that $T^n s = 0$ but $s \notin T \cdot \mathscr{F}_x$.

Corollary 3.15. Take the flattening stratification $\coprod_P S_P \subseteq S$, of S with respect to \mathscr{F} . If the composition $\coprod_P S_P \hookrightarrow S \xrightarrow{f} Y$ is a flat morphism, then \mathscr{F} is flat over Y.

4. FAMILIES OF SUB-SHEAVES AND THEIR DUAL FAMILIES

Definition 4.1. Given a short exact sequence of sheaves

 $0 \to \mathscr{G} \xrightarrow{\iota} \mathscr{F} \to \mathscr{H} \to 0,$

we apply to it the left-exact contravariant functor $\mathscr{F} \mapsto \mathscr{F}^{\vee} := \mathcal{H}om_X(\mathscr{F}, \mathcal{O}_X)$ to obtain exact sequences:

(1) $0 \to \mathscr{H}^{\vee} \to \mathscr{F}^{\vee} \to \operatorname{Im}(\iota^{\vee}) \to 0, \\ 0 \to \operatorname{Im}(\iota^{\vee}) \to \mathscr{G}^{\vee} \to \mathscr{E}xt^{1}_{X}(\mathscr{H}, \mathcal{O}_{X}) \to \mathscr{E}xt^{1}_{X}(\mathscr{F}, \mathcal{O}_{X}).$

FEDERICO QUALLBRUNN

We say that the exact sequence (1), is the *dual exact sequence* of $0 \to \mathscr{G} \xrightarrow{\iota} \mathscr{F} \to \mathscr{H} \to 0$.

Lemma 4.2. Let $0 \to N \xrightarrow{\iota} T \xrightarrow{\pi} M \to 0$ be a short exact sequence of *R*-modules such that *T* is reflexive and *M* is torsion free. Then $\operatorname{Im}(\iota^{\vee})^{\vee} = N$ and $M = \operatorname{Im}(\pi^{\vee\vee})$.

Proof. First we take the duals in the short exact sequence to get a sequence

$$0 \to \hom_R(M, R) \xrightarrow{\pi^{\vee}} \hom_R(T, R) \xrightarrow{\iota^{\vee}} \operatorname{Im}(\iota^{\vee}) \to 0.$$

Then we take duals one more time and, given that T is reflexive and that M is torsion-free, we get the diagram

whose rows are exact.

Chasing arrows, we readily see that the leftmost vertical arrow must be an isomorphism. Indeed, since the monomorphism $N \to T^{\vee \vee}$ factors as

$$N \to \operatorname{Im}(\iota^{\vee})^{\vee} \to T^{\vee \vee}$$

the second arrow being a monomorphism, so must $N \to \operatorname{Im}(\iota^{\vee})^{\vee}$ be. On the other hand, given $a \in \operatorname{Im}(\iota^{\vee})^{\vee}$, we can regard it, via the inclusion, as an element in $T^{\vee\vee} = T$, so we can compute $\pi(a)$. As the canonical map $\theta: M \to M^{\vee\vee}$ is an inclusion we have that, $\theta \circ \pi(a) = \pi^{\vee\vee}(a) = 0$, then $\pi(a) = 0$, so $a \in N$. From this, we have $N \cong \operatorname{Im}(\iota^{\vee})^{\vee}$, wich implies $M = \operatorname{Im}(\pi^{\vee\vee})$. \Box

4.1. Exterior Powers. When dealing with foliations of codimension/dimension greater than 1 is usually convenient to work with *p*-forms. We'll need then to compare sub-sheaves $I \subset \Omega^1_X$ with their exterior powers $\wedge^p I \subset \Omega^p_X$. In order to do that we include the following statements, valid in a wider context.

We'll concentrate on flat modules and their exterior powers. This will be important when dealing with flat families of Pfaff systems of codimension higher than 1 (see Remark 5.5).

Lemma 4.3. Let A be a ring containing the field \mathbb{Q} of rational numbers, and let M be a flat A-module. Then, for every $p, \wedge^p M$ is also flat.

Proof. If tensoring with M is an exact functor, so is its iterate $- \otimes M \otimes \cdots \otimes M$. So $M^{\otimes p}$ is flat. As A contains \mathbb{Q} , there is an anti-symmetrization operator

$$M^{\otimes p} \to \wedge^p M,$$

which is a retraction of the canonical inclusion $\wedge^p M \subset M^{\otimes p}$. This makes $\wedge^p M$ a direct summand of $M^{\otimes p}$; set $M^{\otimes p} = \wedge^p M \oplus R$ for some module R. As the tensor power distributes over direct sums (i.e., $(\wedge^p M \oplus R) \otimes N \cong (\wedge^p M \otimes N) \oplus (R \otimes N)$), so does its derived functors. In particular we have, for every module N,

$$0 = \operatorname{Tor}_1(M^{\otimes p}, N) = \operatorname{Tor}_1(\wedge^p M, N) \oplus \operatorname{Tor}_1(R, N)$$

So $\wedge^p M$ is flat.

Finally, we draw some conclusions regarding flat quotients. When dealing with Pfaff systems, we'll be interested in short exact sequences of the form

$$0 \to \wedge^p I \to \Omega^p_X \to \mathcal{G} \to 0,$$

arising from short exact sequences of flat modules

$$0 \to I \to \Omega^1_X \to \Omega \to 0.$$

Note that, in general $\mathcal{G} \neq \wedge^p \Omega$. Nevertheless, we can state:

Proposition 4.4. Let A be a ring containing \mathbb{Q} . Given an exact sequence

$$0 \to M \to P \to N \to 0$$

of flat A-modules, we have an associated exact sequence

$$0 \to \wedge^p M \to \wedge^p P \to Q \to 0.$$

Then Q is also flat.

Proof. Q inherits a filtration from $\wedge^p P$:

$$Q = \wedge^p P / \wedge^p M = \overline{F}^0 \supseteq \overline{F}^1 \supseteq \cdots \supseteq \overline{F}^p = 0,$$

with quotients

$$F^i/F^{i+1} \cong \wedge^i M \otimes \wedge^{p-i} N.$$

Then Q has a filtration all of whose quotients are flat, so Q itself is flat.

5. Families of distributions and PFAFF systems

We will consider subsheaves of the relative tangent sheaf $T_S X$ and the relative differentials $\Omega^1_{X|S}$.

Definition 5.1. A family of distributions is a short exact sequence

$$0 \to T\mathcal{F} \to T_S X \to N_\mathcal{F} \to 0.$$

The family is called flat if $N_{\mathcal{F}}$ is flat over the base S. A family of distributions is called *involu*tive if it's closed under the Lie bracket operation, that is, if for every pair of local sections X, $Y \in T\mathcal{F}(V)$, we have $[X, Y] \in T\mathcal{F}(V)$, where [-, -] is the Lie bracket in $T_SX(V)$.

Likewise, a family of Pfaff systems is just a short exact sequence

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0.$$

It's called flat if $\Omega^1_{\mathcal{F}}$ is flat.

We will say that a family of Pfaff systems is *integrable* if

$$d(I(\mathcal{F})) \wedge \bigwedge^{r} I(\mathcal{F}) = 0 \subset \Omega^{r+2}_{X|S};$$

where $d: \Omega^{j}_{X|S} \to \Omega^{j+1}_{X|S}$ is the <u>relative</u> de Rham differential, and r is the generic rank of the sheaf $\Omega^{1}_{\mathcal{F}}$.

Remark 5.2. Observe that the relative differential $d : \Omega_{X|S}^j \to \Omega_{X|S}^{j+1}$ is not an \mathcal{O}_X -linear morphism. It is, however, $f^{-1}\mathcal{O}_S$ -linear, so the sheaf $d(I(\mathcal{F})) \wedge \bigwedge^r I(\mathcal{F})$, whose annihilation encodes the integrability of the Pfaff system, is actually a sheaf of $f^{-1}\mathcal{O}_S$ -modules.

In particular, the dual to a family of distributions is a family of Pfaff systems and vice-versa.

Remark 5.3. The dual of an involutive family of distributions is an integrable family of Pfaff systems. Reciprocally, the dual of an integrable family of Pfaff systems is a family of involutive distributions. This is just a consequence of the Cartan-Eilenberg formula for the de Rham differential of a 1-form applied to vector fields

$$d\omega(X,Y) = X(\omega(Y)) - Y(\omega(X)) - \omega([X,Y]).$$

Indeed, as involutiveness and integrability can be checked locally over sections, we can proceed as in [19, Prop. 2.30].

Definition 5.4. The *dimension* of a family of distributions is the generic rank of $T\mathcal{F}$. Likewise, the dimension of a family of Pfaff systems is the generic rank of $\Omega^1_{\mathcal{F}}$.

If $p: X \to S$ is moreover projective, S is connected, and the family is flat, so $T\mathcal{F}$ is a flat sheaf over S. Then for every $s \in S$ the Hilbert polynomial of $T\mathcal{F}_s$ is the same, and so is its generic rank (being encoded in the principal coefficient of the polynomial). The same occurs with families of Pfaff systems.

Remark 5.5. Frequently, in the study of foliations of codimension higher than 1, it is more convenient and better adapted to calculations to work with an alternative description of foliations. Namely, one can define a codimension q foliation on a variety X as in [15], with a global section ω of $\Omega^q_X \otimes \mathcal{L}$ such that:

• ω is locally decomposable, i.e., there is, for all $x \in X$, an open set such that

$$\omega = \eta_1 \wedge \cdots \wedge \eta_q,$$

with $\eta_i \in \Omega^1_X$. • ω is integrable, i.e., $\omega \wedge d\eta_i = 0, \ 1 \leq i \leq q$.

With this setting, studying flat families of codimension q foliations (meaning here families of integrable Pfaff systems) as in [5] and [6], parametrized by a scheme S, amounts to studying short exact sequences of flat sheaves

$$0 \to \mathcal{L}^{-1} \to \Omega^q_{X|S} \to \mathcal{G} \to 0$$

that are locally decomposable and integrable. By the results of Section 4.1, a flat family of codimension q Pfaff systems given as a sub-sheaf of $\Omega^1_{X|S}$ gives rise to a flat family in the above sense.

6. Universal families

Now let's take a non-singular projective scheme X, a polynomial $P \in \mathbb{Q}[t]$, and consider the following functor

$$\begin{aligned}
\mathfrak{Inv}^{P}(X) : Sch &\longrightarrow Sets \\
S &\mapsto \begin{cases}
\text{flat families } 0 \to T\mathcal{F} \to T_{S}(X \times S) \to N_{\mathcal{F}} \to 0 \text{ of in-} \\
\text{volutive distributions such that } N_{\mathcal{F}} \text{ have Hilbert} \\
\text{polynomial } P(t).
\end{aligned}$$

Say $p: X \times S \to X$ is the projection, so $T_S(X \times S) = p^*TX$. Clearly one has $\mathfrak{Im}^P(X)$ is a sub-functor of $\mathfrak{Quot}^P(X,TX)$. We are going to show that $\mathfrak{Inv}^P(X)$ is actually a *closed* sub-functor of $\mathfrak{Quot}^{P}(X, TX)$ and therefore also representable.

So take the smooth morphism given by the projection

$$p_1: \operatorname{Quot}_P(X, TX) \times X \to \operatorname{Quot}_P(X, TX).$$
Here we are taking as base scheme $S = \text{Quot}_{P}(X, TX)$, then on the total space,

$$S \times X = \operatorname{Quot}_P(X, TX) \times X$$

we have the natural short exact sequence

$$0 \to \mathscr{F} \to p_2^* TX = T_S(S \times X) \to \mathscr{Q} \to 0.$$

Now we consider the push-forward of these sheaves by p_1 ; as X is proper, these push-forwards are coherent sheaves over S. In particular, we have maps of coherent sheaves over $\text{Quot}^P(X, TX)$

$$p_{1*}\mathscr{F} \otimes_S p_{1*}\mathscr{F} \xrightarrow{[-,-]} p_{1*}T_S(S \times X) \to p_{1*}\mathscr{Q}$$

induced by the maps over $S \times X$. Note that while the Lie bracket on $T_S(S \times X)$ is only $p_1^{-1}\mathcal{O}_{S^{-1}}$ linear, the map induced on the push-forwards is \mathcal{O}_S -linear, and so is a morphism of coherent sheaves. We then also have for all $m, n \in \mathbb{Z}$ the twisted morphisms

$$p_{1*}\mathscr{F}(m)\otimes_S p_{1*}\mathscr{F}(n) \xrightarrow{[-,-]} p_{1*}T_S(S \times X)(m+n) \to p_{1*}\mathscr{Q}(m+n).$$

Note also that, as p_1 is a projective morphism, then there exists an $n \in \mathbb{Z}$ such that, for every $m \geq n$, the natural sheaves morphism over $S \times X$, $p_1^* p_{1*}(\mathscr{F})(m) \to \mathscr{F}(m)$ is an epimorphism. So, if for some $f: Z \to S$ and some $m \geq n$, one has that the composition

$$f^*p_{1*}\mathscr{F}(m) \otimes_Z f^*p_{1*}\mathscr{F}(m) \xrightarrow{[-,-]} f^*p_{1*}T_S(S \times X)(2m) \to f^*p_{1*}\mathscr{Q}(2m)$$

is zero, then the map

$$(f \times id)^* \mathscr{F}(m) \otimes_{\pi_1^{-1} \mathcal{O}_Z} (f \times id)^* \mathscr{F}(m) \xrightarrow{[-,-]} T_Z(Z \times X)(2m) \to (f \times id)^* \mathscr{Q}(2m)$$

is zero as well. Here $\pi_1: Z \times X \to Z$ is the projection, which is by the way the pull-back of p_1 . Now to conclude the representability of $\mathfrak{Inv}^P(X)$ we need one important lemma.

Lemma 6.1. Let S be a noetherian scheme, $p: X \to S$ a projective morphism, and \mathscr{F} a coherent sheaf on X. Then \mathscr{F} is flat over S if and only if there exists some integer N such that, for all $m \geq N$, the push-forwards $p_*\mathscr{F}(m)$ are locally free.

Proof. This is [8, Lemma 5.5].

We can then take $m \in Z$ big enough so $p_{1*}\mathscr{F}(m)$ and $p_{1*}\mathscr{Q}(2m)$ are locally free and the morphism $p_1^*p_{1*}(\mathscr{F})(m) \to \mathscr{F}(m)$ is epimorphism. Then we can regard the composition

$$p_{1*}\mathscr{F}(m) \otimes_S p_{1*}\mathscr{F}(m) \xrightarrow{[-,-]} p_{1*}T_S(S \times X)(2m) \to p_{1*}\mathscr{Q}(2m)$$

as a global section σ of the locally free sheaf $\mathcal{H}om_S(p_{1*}\mathscr{F}(m) \otimes_S p_{1*}\mathscr{F}(m), \mathscr{Q}(2m))$. We can then make the following definition.

Definition 6.2. We define the scheme $\operatorname{Inv}^{P}(X)$ to be the zero scheme $Z(\sigma)$ (cf.: Definition 3.9) of the section σ defined above.

A direct application of Proposition 3.10 to this definition together with the discussion so far immediately gives us the following.

Proposition 6.3. The subscheme $\operatorname{Inv}^{P}(X) \subseteq \operatorname{Quot}^{P}(X, TX)$ represents the functor $\mathfrak{Inv}^{P}(X)$.

Similarly we can consider the sub-functor $\mathfrak{ipf}^P(X)$ of $\mathfrak{Quot}^P(X, \Omega^1_X)$.

$$\begin{split} \mathfrak{iPf}^{P}(X) : Sch &\longrightarrow Sets \\ S &\mapsto \begin{cases} \text{flat families } 0 \to I(\mathcal{F}) \to \Omega^{1}X | S \to \Omega^{1}_{\mathcal{F}} \to 0 \text{ of integrable Pfaff systems such that } \Omega^{1}_{\mathcal{F}} \text{ have Hilbert poly-} \\ \text{nomial } P(t). \end{cases} . \end{split}$$

Then as before we take $S = \text{Quot}^P(X, \Omega^1_X)$ and consider the map

$$p_{1*}(d(\mathscr{I}) \wedge \bigwedge^r \mathscr{I})(m) \longrightarrow p_{1*}\Omega^{r+2}_{S \times X|S}(m),$$

which is, for large enough m, a morphism between locally free sheaves on S.

Definition 6.4. We define the scheme $iPf^{P}(X)$ to be the zero scheme of the above morphism, viewed as a global section of the locally free sheaf $\mathcal{H}om(p_{1*}(d(\mathscr{I}) \wedge \bigwedge^{r} \mathscr{I})(m), p_{1*}\Omega_{S \times X \mid S}^{r+2}(m)).$

And then, by Proposition 3.10, we have representability.

Proposition 6.5. The subscheme $iPf^{P}(X) \subset Quot^{P}(X, \Omega^{1}_{X})$ represents the functor $i\mathfrak{P}f^{P}(X)$.

7. DUALITY

Definition 7.1. The singular locus of a family of distributions $0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0$ is the (scheme-theoretic) support of $\mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)$. Intuitively, its points are the points where $N_{\mathcal{F}}$ fails to be a fiber bundle.

Similarly, for a family of Pfaff systems $0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$, its singular locus is $\operatorname{supp}(\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X)).$

Remark 7.2. Call $i: T\mathcal{F} \to T_S X$ the inclusion. We have an open non-empty set U where, for every $x \in U$, dim $(Im(i \otimes k(x)))$ is maximal. More precisely, U is the open set where $Tor_1^X(N_{\mathcal{F}}, k(x)) = 0$, which is the maximal open set such that $N_{\mathcal{F}}|_U$ is locally free, and therefore so is $T\mathcal{F}$. Then, when restricted to $U, T\mathcal{F}$ can be given locally as the subsheaf of $T_S X$ generated by k linearly independent relative vector fields, i.e., $T\mathcal{F}$ defines a family of non-singular foliations. In U, one has that $\mathcal{E}xt_X^1(N_{\mathcal{F}}, \mathcal{O}_X) = 0$. Then, the underlying topological space of the singular locus of the family given by $T\mathcal{F}$ is the singular set of the foliation in a classical (topologial space) sense.

The above discussion translates verbatim to families of Pfaff systems.

Proposition 7.3. Let

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

be a family of Pfaff systems such that $\Omega^1_{\mathcal{F}}$ is torsion-free. Its singular locus and the singular locus of the dual family

$$0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0$$

are the same sub-scheme of X. We denote this sub-scheme by $sing(\mathcal{F})$

Proof. We are going to show that the immersions

$$Y_1 := \operatorname{supp}(\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X)) \subseteq X$$
 and $Y_2 := \operatorname{supp}(\mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)) \subseteq X$

represent the same sub-functor of Hom(-, X), thus proving the proposition.

First note that, if $\mathcal{E}xt_X^1(N_{\mathcal{F}}, \mathcal{O}_X) = 0$, then $\mathcal{E}xt_X^1(\Omega_{\mathcal{F}}^1, \mathcal{O}_X) = 0$. Indeed, if $\mathcal{E}xt_X^1(N_{\mathcal{F}}, \mathcal{O}_X) = 0$, $N_{\mathcal{F}}$ is locally free and then so is $T\mathcal{F}$. Moreover, since $\Omega_{\mathcal{F}}^1$ is torsion free, we can dualize the short exact sequence $0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0$ and, by Lemma 4.2, obtain the equality

 $\Omega^1_{\mathcal{F}} = T\mathcal{F}^{\vee}$. So $\Omega^1_{\mathcal{F}}$ is locally free and $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) = 0$.

Now, given a quasi-coherent sheaf \mathscr{G} of X, its support supp $(\mathscr{G}) \subseteq X$ represents the following sub-functor of Hom(-, X):

$$T \longmapsto \{f: T \to X \text{ s.t.: } f^*\mathscr{G} \text{ is not a torsion sheaf}\} \subseteq \operatorname{Hom}(T, X).$$

So, let's take a morphism $f: T \to Y_1 \subseteq X$.

(i) $f : T \to Y_1$ is an immersion: Suppose $f^* \mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)$ is a torsion sheaf. Then there's a point $t \in T$ such that

$$\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)\otimes k(t)=0$$

By Nakayama's lemma this implies that there's an open subset $U \subseteq X$ containing t such that $\mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)|_U = 0$. This in turn implies $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X)|_U = 0$, contradicting the fact that $t \in T \subseteq Y_1$. Then $T \subseteq Y_2$.

Similarly one proves that if $T \subseteq Y_2$ then $T \subseteq Y_1$.

(ii) General case: Taking the scheme-theoretic image of f, we can reduce to the above case where T is a sub-scheme of X.

7.1. The codimension 1 case. We now treat the case of families of codimension 1 foliations. From now on, we'll suppose that $X \to S$ is a smooth morphism.

Definition 7.4. A family of involutive distributions

$$0 \to T\mathcal{F} \to T_S X \to N_\mathcal{F} \to 0,$$

is of *codimension* 1 iff $N_{\mathcal{F}}$ is a sheaf of generic rank 1.

Likewise a family of Pfaff systems

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0,$$

is of codimension 1 if the sheaf $I(\mathcal{F})$ have generic rank 1.

Lemma 7.5. Consider a family of codimension 1 Pfaff systems

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

over an integral scheme X, such that $\Omega^1_{\mathcal{F}}$ is torsion-free. Then $I(\mathcal{F})$ is a line-bundle over X.

Proof. If $\Omega_{\mathcal{F}}^1$ is torsion-free, by Lemma 4.2 we have $I(\mathcal{F}) \cong N_{\mathcal{F}}^{\vee}$. In particular $I(\mathcal{F})$ is the dual of a sheaf, and thus is reflexive and observes property S_2 . Write $I = I(\mathcal{F})$ and consider now the sheaf $I^{\vee} \otimes I$ together with the canonical morphism

$$I^{\vee} \otimes I \to \mathcal{O}_X.$$

The generic rank of $I^{\vee} \otimes I$ is 1. As I is reflexive, $I^{\vee} \otimes I$ is self-dual. So the canonical morphism above induces the dual morphism $\mathcal{O}_X \to I^{\vee} \otimes I$. The composition

$$\mathcal{O}_X \to I^{\vee} \otimes I \to \mathcal{O}_X$$

must be invertible, otherwise the image of $I^{\vee} \otimes I$ in \mathcal{O}_X would be a torsion sub-sheaf. Then I is an invertible sheaf.

Proposition 7.6. In the case of codimension 1 Pfaff systems, if $\Omega^1_{\mathcal{F}}$ is torsion-free over X and the inclusion $I(\mathcal{F}) \to \Omega^1_{X|S}$ is nowhere trivial on S (meaning that $I(\mathcal{F}) \otimes \mathcal{O}_T \to \Omega^1_{X|S} \otimes \mathcal{O}_T$ is never the zero morphism for any $T \to S$), then the family is automatically flat.

Proof. Indeed, $\Omega^1_{\mathcal{F}}$ being torsion free implies that the rank 1 sheaf $I(\mathcal{F})$ must be a line bundle. Then, if we take any morphism $f: T \to S$ and take pull-backs, we'll have an exact sequence

$$0 \to \operatorname{Tor}_1^S(\Omega^1_{\mathcal{F}}, T) \to f^*I(\mathcal{F}) \to f^*\Omega^1_{X|S} \to f^*\Omega^1 \to 0.$$

Now, as $I(\mathcal{F})$ is a line bundle, the cokernel $f^*I(\mathcal{F})/\operatorname{Tor}_1^S(\Omega^1_{\mathcal{F}},T)$ must be a torsion sheaf over X_T . But, X being smooth over S, the annihilator $f^*\Omega^1_{X|S}$ is of the form $p^*(J)$, with $J \subset \mathcal{O}_T$, so $f^*I(\mathcal{F}) \to f^*\Omega^1_{X|S}$ must be the zero morphism when restricted to \mathcal{O}_T/J , contradicting the nowhere triviality assumption.

Remark 7.7. In the codimension 1 case, we can calculate $\operatorname{sing}(\mathcal{F})$ by noting that $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X)$ is the cokernel in the exact sequence

$$T_S X \to I(\mathcal{F})^{\vee} \to \mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \to 0.$$

We can then tensor the sequence by $I(\mathcal{F})$ and obtain

$$T_S X \otimes I(\mathcal{F}) \to \mathcal{O}_X \to \mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \otimes I(\mathcal{F}) \to 0.$$

Now, $I(\mathcal{F})$ being a line bundle, the support of $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X)$ and that of $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \otimes I(\mathcal{F})$ are exactly the same. Note then that, in the second exact sequence, the cokernel is the scheme-theoretic zero locus of the twisted 1-form given by

$$\mathcal{O}_X \xrightarrow{\omega} \Omega^1_{X|S} \otimes I(\mathcal{F})^{\vee},$$

as defined in Section 3.2. So, if we have a family of codimension 1 Pfaff systems given locally by a twisted form

$$\omega = \sum_{i=1}^{n} f_i(x) dx_i,$$

then sing(\mathcal{F}) is the scheme defined by the ideal (f_1, \ldots, f_n) .

The above proposition and remark tell us that our definition of flat family for Pfaff systems of codimension 1 is essentially the same as the one used in the now classical works of Lins-Neto, Cerveau, et.al.

Theorem 7.8. Assume we have two families

(2)
$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

(3)
$$0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0,$$

satisfying the following conditions:

- The families 2 and 3 are dual to each other.
- $N_{\mathcal{F}}$ is torsion free (or, equivalently, $\Omega^{1}_{\mathcal{F}}$ is torsion free).
- They are codimension 1 families.
- $\operatorname{sing}(\mathcal{F})$ is flat over S.

Then 2 is flat if and only if 3 is flat.

Proof. Let $\Sigma = \operatorname{sing}(\mathcal{F})$.

Let's suppose first that the family

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

is flat. We have to prove that $N_{\mathcal{F}}$ is also flat. To do this we note that applying the functor $\mathcal{H}om_X(-, \mathcal{O}_X)$ to the family of distributions not only gives us the family of Pfaff systems but also the exact sequence

$$0 \to N_{\mathcal{F}} \to I(\mathcal{F})^{\vee} \to \mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \to 0.$$

Being $\Omega^1_{\mathcal{F}}$ torsion-free, $I(\mathcal{F})$ must be a line bundle, and so must $I(\mathcal{F})^{\vee}$; let's call $I(\mathcal{F})^{\vee} = \mathcal{L}$ to ease the notation. Now \mathcal{L} has $N_{\mathcal{F}}$ as a sub-sheaf generically of rank 1, so $N_{\mathcal{F}} = \mathcal{I} \cdot \mathcal{L}$ for some ideal sheaf \mathcal{I} . Then $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \cong \mathcal{L} \otimes \mathcal{O}_X/\mathcal{I}$. As $\Sigma = \operatorname{supp}(\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X))$, one necessarily has $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \cong \mathcal{L}_{\Sigma}$. Then \mathcal{L}_{Σ} , being a locally free sheaf over Σ which is flat over S, is itself flat over S. Therefore, as \mathcal{L} is also flat over S, flatness for $N_{\mathcal{F}}$ follows.

Let's suppose now that the family

$$0 \to T\mathcal{F} \to T_S X \to N_\mathcal{F} \to 0$$

is flat. We have to prove that $\Omega^1_{\mathcal{F}}$ is also flat. By the above proposition, it's enough to show that the morphism $I(\mathcal{F}) \xrightarrow{\iota} \Omega^1_{X|S}$ is nowhere zero. Suppose there is $T \to S$ such that $\iota_T = 0$. Take an open set $U \subset X$ where $\Omega^1_{\mathcal{F}}$ is locally free. In that open set, we can apply base change with respect to the functor $\mathcal{H}om_X(-, \mathcal{O}_X)$ ([1] or [4]); so, restricting everything to U we have $(\iota_T)^{\vee} \cong (\iota^{\vee})_T$. But, in U, ι^{\vee} is the morphism $T_S X \to N_{\mathcal{F}}$ and so it cannot become the zero morphism under any base change. \Box

7.2. The arbitrary codimension case. To give an analogous theorem to 7.8 in arbitrary codimension, we'll have to deal with finer invariants than the singular locus of the foliation. In the scheme X, we'll consider a stratification naturally associated with \mathcal{F} . This stratification have been already studied and described by Suwa in [18]. To deal with flatness issues, we have to provide a scheme structure to Suwa's stratification; this will be a particular case of a flattening stratification. Before going into that, we begin with some generalities. Remember that we are working over a smooth morphism $X \to S$.

Lemma 7.9. Let $X \to S$ be a smooth morphism and \mathscr{F} a coherent sheaf on X that is relatively $Z^{(2)}$ -closed over S. Then, for all $s \in S$, the sheaf $\mathscr{F}_s = \mathscr{F} \otimes k(s)$ is $Z^{(2)}$ -closed over X_s .

Proof. We have to show that, for every $U \subset X_s$ such that $\operatorname{codim}(X \setminus U) \ge 2$, the restriction

$$\mathscr{F}_s \xrightarrow{\rho_U} \mathscr{F}_s|_U$$

is surjective. As the formal completion $\widehat{\mathcal{O}}_{X_s,x}$ of \mathcal{O}_{X_s} with respect to any closed point x is faithfully flat over \mathcal{O}_{X_s} [11, IV.3.2], we can check surjectivity of ρ_U by looking at every formal completion. As $X \to S$ is smooth formally around a point x, we have $\mathcal{O}_X \cong \mathcal{O}_S \otimes_k k[z_1, \ldots, z_d]$ and so we can take an open subset $V \subseteq X$ to be $V = U \times S$. Then, with this choice of V, we have an epimorphism

$$\widehat{\mathscr{F}}|_V \to \widehat{\mathscr{F}}_s|_U \to 0.$$

Then we have a diagram with exact rows and columns

$$\begin{array}{c} \widehat{\mathscr{F}} \longrightarrow \widehat{\mathscr{F}}_{s} \longrightarrow 0 \\ & \bigvee_{\rho_{V}} & \bigvee_{\rho_{U}} \\ \widehat{\mathscr{F}}|_{V} \longrightarrow \widehat{\mathscr{F}}_{s}|_{U} \longrightarrow 0 \\ & & \downarrow \\ & 0 \end{array}$$

So ρ_U must be an epimorphism as well.

Lemma 7.10. Let $p: X \to S$ be a smooth morphism. Consider a family of distributions

 $0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0.$

If the codimension of $sing(\mathcal{F})$ with respect to $X_{p(sing(\mathcal{F}))}$ is greater than 2, then, for every map $T \to S$, one has

$$\mathcal{H}om_X(T\mathcal{F}, \mathcal{O}_X) \otimes \mathcal{O}_T \cong \mathcal{H}om_{X_T}(T\mathcal{F}_T, \mathcal{O}_T).$$

The analogous statement is true for $I(\mathcal{F})^{\vee}$ in a flat family of Pfaff systems.

Proof. By [1, Theorem 1.9], we only have to prove that, for every closed point $s \in S$, the natural map

$$\mathcal{H}om_X(T\mathcal{F},\mathcal{O}_X)\otimes k(s)\to\mathcal{H}om_{X_s}(T\mathcal{F}\otimes k(s),\mathcal{O}_X\otimes k(s))$$

is surjective. Being the dual of some sheaves, both $\mathcal{H}om_X(T\mathcal{F},\mathcal{O}_X)$ and

 $\mathcal{H}om_{X_s}(T\mathcal{F}\otimes k(s), \mathcal{O}_X\otimes k(s))$

possess the relative property S_2 with respect to p (Proposition 3.7), and so are relatively $Z^{(2)}$ closed with respect to p, and so is $\mathcal{H}om_X(T\mathcal{F}, \mathcal{O}_X) \otimes k(s)$ by the above lemma.

Let $U = X \setminus \operatorname{sing} \mathcal{F}$ and $j : U \hookrightarrow X$ be the inclusion. As $T\mathcal{F}|_U$ is locally free over U, so is $T\mathcal{F}^{\vee}|_U$. Then, in U, we have

$$\mathcal{E}xt^1(T\mathcal{F}|_U, \mathcal{O}_X|_U \otimes_S \mathcal{G}) = 0$$

for every $\mathcal{G} \in Coh(S)$. Then, from the exchange property for local Ext's [1, Theorem 1.9], we get surjectivity on

$$\mathcal{H}om_X(T\mathcal{F}|_U, \mathcal{O}_X|_U) \otimes k(s) \to \mathcal{H}om_{X_s}(T\mathcal{F}|_U \otimes k(s), \mathcal{O}_X|_U \otimes k(s)).$$

But, as $\operatorname{codim}(\operatorname{sing}(\mathcal{F})) > 1$ and both sheaves are S_2 , then surjectivity holds in all of X_s .

Lemma 7.11. Suppose that a flat family

$$0 \to T\mathcal{F} \to T_S X \to N_\mathcal{F} \to 0$$

is such that the relative codimension $d_S(\operatorname{sing}(\mathcal{F}))$ of $\operatorname{sing}(\mathcal{F})$ over S verifies $d_S(\operatorname{sing}(\mathcal{F})) \geq 2$. Suppose further that the flattening stratification of X over $T\mathcal{F}$ is flat over S (c.f., Proposition 3.14). Then $T\mathcal{F}^{\vee}$ is also a flat \mathcal{O}_S -module.

The analogous statement is true for $I(\mathcal{F})^{\vee}$ in a flat family of Pfaff systems.

Proof. The proof works exactly the same for distributions of Pfaff systems mutatis mutandi.

Take $\coprod_P X_P$ to be the flattening stratification of X with respect to $T\mathcal{F}$. The restriction $T\mathcal{F}_{X_P}$ (being coherent and flat over X_P) is locally free over X_P , and hence so is its dual $\mathcal{H}om_{X_P}(T\mathcal{F}_{X_P}, \mathcal{O}_{X_P})$. By Lemma 7.10, in each stratum X_P , we have the isomorphism

$$\mathcal{H}om_{X_P}(T\mathcal{F}_{X_P}, \mathcal{O}_{X_P}) \cong \mathcal{H}om_X(T\mathcal{F}, \mathcal{O}_X) \otimes \mathcal{O}_{X_P} = T\mathcal{F}^{\vee} \otimes \mathcal{O}_{X_P}$$

So $T\mathcal{F}^{\vee}$ is flat when restricted to the filtration $\coprod_P X_P$, which is in turn flat over S. Then, by [8, Section 5.4.2], $T\mathcal{F}^{\vee}$ is flat over S.

Definition 7.12. For a family of distributions consider the flattening stratification

$$\prod_{P(\mathcal{F})} X_{P(\mathcal{F})} \subseteq X$$

of X with respect to $T\mathcal{F} \oplus N_{\mathcal{F}}$. We call this the rank stratification of X with respect to $T\mathcal{F}$.

Remark 7.13. Note that the flattening stratification of $T\mathcal{F} \oplus N_{\mathcal{F}}$ is the (scheme-theoretic) intersection of the flattening stratification of $T\mathcal{F}$ with that of $N_{\mathcal{F}}$. This is because

$$(T\mathcal{F}\oplus N_{\mathcal{F}})\otimes \mathcal{O}_Y$$

is flat if and only if both $T\mathcal{F} \otimes \mathcal{O}_Y$ and $N_{\mathcal{F}} \otimes \mathcal{O}_Y$ are.

This tells us, in particular, that each stratum is indexed by two natural numbers r and k such that

$$x \in X_{r,k} \iff \dim(T\mathcal{F} \otimes k(x)) = r \text{ and } \dim(N_{\mathcal{F}} \otimes k(x)) = k$$

In [18], Suwa studied a related stratification associated to a foliation. Given a distribution $D \subset TM$ on a complex manifold M, he defines the strata $M^{(l)}$ as

$$M^{(l)} = \{x \in M \text{ such that } D_x \subset T_x M \text{ is a sub-space of dimension } l \}$$

Here D is spanned point-wise by vector fields v_1, \ldots, v_r , and $D_x = \langle v_i(x) \rangle$. Clearly if D is of generic rank r, the open stratum is $M^{(r)}$.

Note that, in the setting of distribution as sub-sheafs $i: T\mathcal{F} \hookrightarrow TX$ of the tangent sheaf of a variety, the vector space $T_x\mathcal{F}$ is actually the image of the map

$$T\mathcal{F}\otimes k(x)\xrightarrow{\imath\otimes\kappa(x)}TX\otimes k(x),$$

whose kernel is $\operatorname{Tor}_{1}^{X}(N_{\mathcal{F}}, k(x))$. Moreover we have the exact sequence

$$0 \to T_x \mathcal{F} = \operatorname{Im}(i \otimes k(x)) \to TX \otimes k(x) \to N_{\mathcal{F}} \otimes k(x) \to 0.$$

In particular, in a variety X of dimension n, if $\dim(T_x\mathcal{F}) = l$, then $\dim(N_{\mathcal{F}} \otimes k(x)) = n - l$. So what we call the rank stratification of X is actually a refinement of the stratification studied in [18].

Our main motivation for defining this refinement of the stratification of [18] is the following result.

Theorem 7.14. Assume we have dual families

(4)
$$0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0,$$

(5) $0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0,$

parametrized by a scheme S of finite type over an algebraically closed field, such that

- $N_{\mathcal{F}}$ is torsion free
- The relative codimension of sing(\mathcal{F}) over S (that is $d_S(sing(\mathcal{F}))$) verifies $d_S(sing(\mathcal{F}) \ge 2$.
- Each stratum $X_{r,k}$ of the rank stratification is flat over S.

Then 4 is flat over S if and only if 5 is. Moreover, for each point $s \in S$, we have

$$I(\mathcal{F})_s = (N_{\mathcal{F}s})^{\vee},$$

in other terms "the dual family is the family of the duals".

Proof. We prove one of the implications; the proof of the other is identical.

Consider the exact sequence

$$0 \to \Omega^1_{\mathcal{F}} \to T\mathcal{F}^{\vee} \to \mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X) \to 0.$$

It is clear that, to prove flatness of the dual family, it's enough to show $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)$ is flat over S.

Also, by Lemma 7.10, we have, for every $s \in S$, the diagram with exact rows and columns,

So $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X) \otimes k(s) \to \mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}} \otimes k(s), \mathcal{O}_{X_s})$ is surjective for every $s \in S$ and so, by [1, Theorem 1.9], the exchange property is valid for $\mathcal{E}xt^1_X(\Omega^1_{\mathcal{F}}, \mathcal{O}_X)$. If moreover $\mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)$ is flat over S, then, again by [1, Theorem 1.9],

$$I(\mathcal{F})_s = \mathcal{H}om_X(N_{\mathcal{F}}, \mathcal{O}_X) \otimes k(s) \cong \mathcal{H}om_X(N_{\mathcal{F}s}, \mathcal{O}_{X_s}) = (N_{\mathcal{F}s})^{\vee}$$

By Proposition 3.14, it is enough to show the restriction of $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)$ to every rank stratum is flat over S. So let $Y \subseteq X$ be a rank stratum. If we can show that $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X) \otimes \mathcal{O}_Y$ is locally free, then we're set.

By hypothesis, one has the isomorphism $\mathcal{H}om_X(T\mathcal{F}, \mathcal{O}_X) \otimes \mathcal{O}_Y \cong \mathcal{H}om_Y(T\mathcal{F}_Y, \mathcal{O}_Y)$. So we can express $\mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X) \otimes \mathcal{O}_Y$ as the cokernel in the \mathcal{O}_Y -modules exact sequence

$$\mathcal{H}om_Y(TX_Y, \mathcal{O}_Y) \to \mathcal{H}om_Y(T\mathcal{F}_Y, \mathcal{O}_Y) \to \mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)_Y \to 0.$$

So, localizing at a point $y \in Y$, we can realize the local $\mathcal{O}_{Y,y}$ -module $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)_{Y,y}$ as the set of maps $T\mathcal{F}_{Y,y} \to \mathcal{O}_{Y,y}$ modulo the ones that factors as



To study $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)_{Y,y}$ this way, note that we have the following exact sequence.

$$0 \to \operatorname{Tor}_1^X(N_{\mathcal{F}}, \mathcal{O}_{Y,y}) \to T\mathcal{F}_{Y,y} \to TX_{Y,y} \to (N_{\mathcal{F}})_{Y,y} \to 0.$$

which we split into two short exact sequences,

(6)
$$0 \to \mathcal{K} \to TX_{Y,y} \to (N_{\mathcal{F}})_{Y,y} \to 0$$
 and

(7)
$$0 \to \operatorname{Tor}_{1}^{X}(N_{\mathcal{F}}, \mathcal{O}_{Y,y}) \to T\mathcal{F}_{Y,y} \to \mathcal{K} \to 0.$$

Now, as Y is a rank stratum, $\mathcal{Q}_{\mathcal{Y}}$ and $T\mathcal{F}_Y$ are flat over Y, and coherent, and so they are locally free. As a consequence, short exact sequence (6) splits, so $TX_{Y,y} \cong (N_{\mathcal{F}})_{Y,y} \oplus \mathcal{K}$. So

$$\mathcal{H}om_Y(TX_Y, \mathcal{O}_Y)_y \cong \mathcal{H}om_Y(\mathcal{K}, \mathcal{O}_{Y,y}) \oplus \mathcal{H}om_Y((N_{\mathcal{F}})_Y, \mathcal{O}_Y)_y$$

and we get $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)_{Y,y}$ as the cokernel in

(8)
$$\mathcal{H}om_Y(\mathcal{K}, \mathcal{O}_{Y,y}) \to \mathcal{H}om_Y(T\mathcal{F}_Y, \mathcal{O}_Y)_y \to \mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)_{Y,y} \to 0.$$

Since $(N_{\mathcal{F}})_Y$ and TX_Y are locally free over Y, so is \mathcal{K} . Then short exact sequence (7) splits, so $T\mathcal{F}_{Y,y} \cong \operatorname{Tor}_1^X(N_{\mathcal{F}}, \mathcal{O}_{Y,y}) \oplus \mathcal{K}$. Also, as $T\mathcal{F}_Y$ and \mathcal{K} are locally free over Y, so is $\operatorname{Tor}_1^X(N_{\mathcal{F}}, \mathcal{O}_Y)$. Sequence (8) now reads

$$\mathcal{H}om_Y(\mathcal{K}, \mathcal{O}_{Y,y}) \to \mathcal{H}om_Y(\operatorname{Tor}_1^X(N_{\mathcal{F}}, \mathcal{O}_{Y,y}), \mathcal{O}_Y)_y \oplus \mathcal{H}om_Y(\mathcal{K}, \mathcal{O}_{Y,y}) \to \mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)_{Y,y} \to 0.$$

So we have

$$\mathcal{H}om_Y(\mathrm{Tor}_1^X(N_{\mathcal{F}},\mathcal{O}_{Y,y}),\mathcal{O}_Y)_y \cong \mathcal{E}xt_X^1(N_{\mathcal{F}},\mathcal{O}_X)_{Y,y}.$$

Now, as $\operatorname{Tor}_1^X(N_{\mathcal{F}}, \mathcal{O}_{Y,y})$ is locally free over Y, so is its dual. In other words, we just proved $\mathcal{E}xt_X^1(N_{\mathcal{F}}, \mathcal{O}_X)_Y$ is locally free over Y, which settles the theorem.

Now, by generic flatness, we can conclude the following:

Corollary 7.15. Every irreducible component of the scheme Inv_P is birationally equivalent to an irreducible component of iPf_P .

Remark 7.16. During the proof of Theorem 7.14, we have actually obtained this result:

Proposition 7.17. $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)$ is flat over the rank stratification.

In particular, if $\coprod X_Q$ denotes the flattening stratification of $\mathcal{E}xt^1_X(N_{\mathcal{F}},\mathcal{O}_X)$, there is a morphism

$$\coprod_{P(\mathcal{F})} X_{P(\mathcal{F})} \to \coprod_Q X_Q.$$

Now, by the construction of the flattening stratification, $\coprod X_Q$ consist of an open stratum U such that $\mathcal{E}xt^1_X(N_{\mathcal{F}}, \mathcal{O}_X)|_U = 0$, and closed strata whose closure is $\operatorname{sing}(\mathcal{F})$. So the morphism $\coprod_{P(\mathcal{F})} X_{P(\mathcal{F})} \to \coprod_Q X_Q$ actually defines a stratification of $\operatorname{sing}(\mathcal{F})$.

8. Singularities

Theorem 7.8 gives a condition for a flat family of integrable Pfaff systems to give rise to a flat family of involutive distributions in terms of the flatness of the singular locus. We have then to be able to decide when can we apply the theorem. More precisely, say

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

is a flat family of codimension 1 integrable Pfaff systems, and let $s \in S$. How do we know when $\operatorname{sing}(\mathcal{F})$ is flat around s? In this section, we address this question and give a sufficient condition for $\operatorname{sing}(\mathcal{F})$ to be flat at s in terms of the classification of singular points of the Pfaff system $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$.

From now on, we will consider only Pfaff systems such that $\Omega^1_{\mathcal{F}}$ is torsion-free.

Remember that, if we have a Pfaff system of codimension 1, $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$, such that $\Omega^1_{\mathcal{F}_s}$ is torsion-free, we can consider, locally on X, that it is given by a single 1-form ω and that it is integrable iff $\omega \wedge d\omega = 0$.

Definition 8.1. Let $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$ be a codimension 1 integrable Pfaff system, and let $x \in X$. We say that x is a Kupka singularity of the Pfaff system if, for some 1-form ω locally defining $I(\mathcal{F})$ around x, we have $\omega_x = 0$ and $d\omega_x \neq 0$.

We also say that x is a Kupka singularity of the foliation defined by $I(\mathcal{F})$.

Proposition 8.2. The set of Kupka singularities of a codimension 1 foliation, if non-void, has a natural structure of a codimension 2 sub-scheme of X.

Proof. See [14]

Definition 8.3. Let $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$ be as before. Say X has dimension n over \mathbb{C} . Then we call $x \in X$ a *Reeb singularity* if there exists an analytic neighborhood U of x such that $I(\mathcal{F})$ may be generated by a form ω with the property that ω can be written, locally

in U, in the form $\omega = \sum_{i=1}^{n} f_i dz_i$ with $f_i(x) = 0$ for all i, and $(df_1)_x, \ldots, (df_n)_x$ are linearly independent in $T_x^* X$.

Remark 8.4. Note that Kupka singularities and Reeb singularities are singularities in the sense of 7.1, i.e., they are points in $sing(\mathcal{F})$.

We now give a version for families of the fundamental result of Kupka.

Proposition 8.5. Let

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

be a flat family of integrable Pfaff systems of codimension 1, and let $\Sigma = \operatorname{sing}(\mathcal{F}) \subset X$. Let $s \in S$, and $x \in \Sigma_s$ be such that x is a Kupka singularity of $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$. Then, locally around x, $I(\mathcal{F})$ can be given by a relative 1-form $\omega(z,s) \in \Omega^1_{X \setminus S}$ such that

$$\omega = f_1(z,s)dz_1 + f_2(z,s)dz_2$$

that is, ω is locally the pull-back of a relative form $\eta \in \Omega^1_{Y|S}$, where $Y \to S$ is of relative dimension 2.

The proof is essentially the same as the proof of the classical Kupka theorem, as in [15]. One only needs to note that every ingredient there can be generalized to a relative setup.

For this we note that, as $p: X \to S$ is a smooth morphism, the relative tangent sheaf $T_S X$ is locally free and is the dual sheaf of the locally free sheaf $\Omega^1_{X|S}$. We note also that, if $v \in T_S X(U)$, and $\omega \in \Omega^1_{X|S}(U)$, the relative Lie derivative $L_v(\omega)$ is well-defined by Cartan's formula

$$L_v^S = d_S \iota_v(\omega) + \iota_v(d_S \omega),$$

where $\iota_v(\omega) = \langle v, \omega \rangle$ is the pairing of dual spaces (and by extension also the map $\Omega_{X|S}^q \to \Omega_{X|S}^{q-1}$ determined by v), and d_S is the *relative* de Rham differential. Also $\Omega_{X|S}^q = \wedge^q \Omega_{X|S}^1$.

Finally we observe that, if $p: X \to S$ is of relative dimension d and X is a regular variety over \mathbb{C} of total dimension n, a family of integrable Pfaff systems gives rise to a foliation on X whose leaves are tangent to the fibers of p. Indeed, we can pull back the subsheaf $I(\mathcal{F}) \subset \Omega^1 X | S$ by the natural epimorphism

$$f^*\Omega^1_S \to \Omega^1_X \to \Omega^1_{X|S} \to 0,$$

and get $J = I(\mathcal{F}) + f^* \Omega^1_S \subset \Omega^1_X$, which is an integrable Pfaff system in X, determining a foliation $\hat{\mathcal{F}}$. As $f^* \Omega^1_S \subset J$, the leaves of the foliation $\hat{\mathcal{F}}$ are contained in the fibers X_s of p.

In the general case, where p is smooth but S and X need not to be regular over \mathbb{C} , the Frobenius theorem still gives foliations \mathcal{F}_s in each fiber X_s . Indeed, as $p: X \to S$ is smooth, each fiber X_s is regular over \mathbb{C} and, $\Omega^1_{\mathcal{F}}$ being flat, $I(\mathcal{F})_s \subset \Omega^1_{X_s}$ is an integrable Pfaff system on X_s .

Proposition 8.6. Let $p: X \to S$ a smooth morphism over \mathbb{C} and

$$0 \to I(\mathcal{F}) \to \Omega^1 X | S \to \Omega^1_{\mathcal{F}} \to 0$$

a codimension 1 flat family of Pfaff systems. Let $\omega \in \Omega^1_{X|S}(U)$ be an integrable 1-form such that $I(\mathcal{F})(U) = (\omega)$ in a neighborhood U of a point $x \in X$. Then $d\omega$ is locally decomposable.

Proof. As $T_S X = (\Omega^1 X | S)^{\vee}$ and $\Omega^q_{X|S} = \wedge^q \Omega^1_{X|S}$, we can apply Plücker relations to determine if $d\omega$ is locally decomposable and proceed as in [15, Lemma 2.5].

Lemma 8.7. Suppose that $d\omega_x \neq 0$. Consider \mathcal{G}_s the codimension 2 foliations defined by $d\omega$ in X_s . In the neighborhood V of $x \in X$ where \mathcal{G}_s is non-singular for every s, we have that the leaves of \mathcal{G}_s are integral manifolds of $\omega|_{X_s}$.

Proof. We have only to prove, for every $v \in T_S X$ such that $\iota_v(d\omega) = 0$, that $\iota_v(\omega) = 0$. We can do this exactly as in [15, Lemma 2.6].

Lemma 8.8. Assume the same hypothesis as Lemma 8.7. Let v be a vector field tangent to \mathcal{G} . Then the relative Lie derivative of ω with respect to v is zero.

Proof. Like the proof of [15, Lemma 2.7].

Lemma 8.9. Under the same hypothesis as Lemma 8.7 and 8.8, $\operatorname{sing}(\omega)$ is saturated by leaves of $(\mathcal{G}_s)_{s \in S}$ (i.e., take $y \in V$ a zero of ω such that p(y) = s, and L the leaf of \mathcal{G}_s going through y. Then the inclusion $L \to V$ factors through $\operatorname{sing}(\omega)$).

Proof. We can do this entirely on X_s . Then this reduces to [15, Lemma 2.7].

Proof of Proposition 8.5. We take an analytic neighborhood V of $x \in X$ such that $V \cong U \times D^d$ with $U \subseteq S$ an open set, D^d a complex polydisk, and

$$p|V:V \cong U \times D^d \longrightarrow U.$$

(s, z₁,..., z_d) \mapsto s

By the Frobenius theorem, we can choose the coordinates z_i in such a way that

$$w_i = \frac{\partial}{\partial z_i} \in T_S X(V), \qquad 3 \le i \le d,$$

are tangent to $d\omega$. Then, as $L_{v_i}^S \omega = 0$ and $\iota_{v_i} \omega = 0$, we can write ω as

$$\omega = f_1(z,s)dz_1 + f_2(z,s)dz_2.$$

Proposition 8.10. Let

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

be a flat family of integrable Pfaff systems of codimension 1, and let $\Sigma = \operatorname{sing}(\mathcal{F}) \subset X$. Let $s \in S$, and $x \in \Sigma_s$ be such that x is a Kupka singularity of $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$. Then $\Sigma \to S$ is smooth around x.

Proof. By 8.5 above, we can determine Σ around x as the common zeroes of $f_1(z, s)$ and $f_2(z, s)$. The condition $d\omega \neq 0$ implies Σ is *smooth* over S (remember that we are using the *relative* de Rham differential and that means the variable s counts as a constant).

Proposition 8.11. Let

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

be a flat family of integrable Pfaff systems of codimension 1, $\Sigma = \operatorname{sing}(\mathcal{F}) \subset X$, and $s \in S$. Suppose $x \in \Sigma_s$ is such that x is a Reeb singularity of $0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$. Then $\Sigma \to S$ is étalé around x.

Proof. The condition on x means we can actually give $I(\mathcal{F})$ locally by a relative 1-form $\omega \in \Omega^1_{X|S}$, $\omega = \sum_{i=1}^n f_i(z, s) dz_i$, with n the relative dimension of X over S and the df_i 's linearly independent on x. Then Σ is given by the equations $f_1 = \cdots = f_n = 0$ and is therefore étalé over S. \Box

With these two propositions, we are almost in a position to state our condition for flatness of the dual family, we just need a general:

Lemma 8.12. Let $X \xrightarrow{p} S$ be a morphism between schemes of finite type over an algebraically closed field k. Let $U \subseteq X$ be the maximal open sub-scheme such that $U \xrightarrow{p} S$ is flat, and $s \in S$ a point such that X_s is without embedded components. If $U_s \subseteq X_s$ is dense, then $U_s = X_s$.

Proof. By Proposition 3.12, we must check that, for A either a discrete valuation domain or an Artin ring of the form $k[T]/(T^{n+1})$, and every arrow $\operatorname{Spec}(A) \to S$, the pull-back scheme $X_{\operatorname{Spec}}(A)$ is flat over $\operatorname{Spec}(A)$. In this way, the problem reduces to the case where $S = \operatorname{Spec}(A)$.

(i) Case A is a DVD. In this case, A being a principal domain, flatness of X over Spec(A) is equivalent to the local rings $\mathcal{O}_{X,x}$ being torsion-free $A_{p(x)}$ -modules for every point $x \in X$ ([11, IV.1.3], so it suffices to consider the case $A_{p(x)} = A$.

Now, let $f \in \mathcal{O}_{X,x}$ and $J = \operatorname{Ann}_A(f) \subseteq A$. Suppose $J \neq (0)$ and consider $V(J) \subseteq \operatorname{Spec}(A)$; clearly $\operatorname{supp}(f) \subseteq p^{-1}(V(J)) \subseteq X$. So $U \cap \operatorname{supp}(f) = \emptyset$. But then the restriction f|s of f to X_s has support disjoint with U_s . On the other hand

$$\operatorname{supp}(f|_s) = \overline{\{\mathfrak{P}_1\}} \cup \cdots \cup \overline{\{\mathfrak{P}_m\}} \subseteq X_s,$$

where $\{\mathfrak{P}_1, \ldots, \mathfrak{P}_m\} = \operatorname{Ass}(\mathcal{O}_{X_s,x}/(f|_s)) \subseteq \operatorname{Ass}(\mathcal{O}_{X_s,x})$. As X_s is without immersed components, the \mathfrak{P}_i 's are all minimal, so $X_k \cap \overline{\mathfrak{P}_i}$ is an irreducible component of X_k . But

$$U_s \cap \overline{\mathfrak{P}_i} = \emptyset,$$

contradicting the hypothesis that U_s is dense in X_s .

(ii) Case $A = k[T]/(T^{n+1})$. Using Proposition 3.13, this works just as the first case taking $f \in \mathcal{O}_{X,x}$ as a section such that $T^n f = 0$ but $f \notin T\mathcal{O}_{X,x}$.

We have already said that, in a Pfaff system, Kupka singularities, if they exist, form a codimension 2 sub-scheme of X. We will call $\mathcal{K}(\mathcal{F})$ this sub-scheme, and $\overline{\mathcal{K}}(\mathcal{F})$ its closure.

Theorem 8.13. Let

$$0 \to I(\mathcal{F}) \to \Omega^1_{X|S} \to \Omega^1_{\mathcal{F}} \to 0$$

be a flat family of integrable Pfaff systems, for $s \in S$. Consider the Pfaff system

$$0 \to I(\mathcal{F})_s \to \Omega^1_{X_s} \to \Omega^1_{\mathcal{F}_s} \to 0$$

If sing(\mathcal{F}_s) is without embedded components and sing(\mathcal{F}_s) = $\overline{\mathcal{K}}(\mathcal{F}_s) \cup \{p_1, \ldots, p_m\}$, where the p_i 's are Reeb-type singularities, then sing(\mathcal{F}) $\rightarrow S$ is flat in a neighborhood of $s \in S$.

Proof. Indeed, by Proposition 8.10, $\operatorname{sing}(\mathcal{F})$ is flat in a neighborhood of $\mathcal{K}(\mathcal{F}_s)$, and, as $\operatorname{sing}(\mathcal{F}_s)$ is without embedded components, we can apply Lemma 8.12 to conclude that $\operatorname{sing}(\mathcal{F})$ is flat in a neighborhood of $\overline{\mathcal{K}(\mathcal{F}_s)}$.

Lastly, from Proposition 8.11, it follows that $sing(\mathcal{F})$ is flat in a neighborhood of $\{p_1, \ldots, p_m\}$.

9. Applications

Let $X = \mathbb{P}^n(\mathbb{C})$. It's well-known that the class of sheaves \mathscr{F} that split as a direct sum of line bundles $\mathscr{F} \cong \bigoplus_i \mathcal{O}(k_i)$ has no non-trivial deformations. Indeed, as deformation theory teaches us, first-order deformations of \mathscr{F} are parametrized by $\operatorname{Ext}^1(\mathscr{F}, \mathscr{F})$, in this case we have

$$\operatorname{Ext}^{1}(\mathscr{F},\mathscr{F}) \cong \bigoplus_{i,j} \operatorname{Ext}^{1}(\mathcal{O}(k_{i}), \mathcal{O}(k_{j})) \cong \bigoplus_{i,j} \operatorname{Ext}^{1}(\mathcal{O}, \mathcal{O}(k_{j} - k_{i})) \cong \bigoplus_{i,j} H^{1}(\mathbb{P}^{n}, \mathcal{O}(k_{j} - k_{i})) = 0$$

In particular, given a flat family of distributions

$$0 \to T\mathcal{F} \to T_S(\mathbb{P}^n \times S) \to N_\mathcal{F} \to 0,$$

such that, for some $s \in S$, $T\mathcal{F}_s \cong \bigoplus_i \mathcal{O}(k_i)$, the same decomposition holds true for the rest of the members of the family.

When we deal with codimension 1 foliations, it's more common, however, to work with Pfaff systems or, more concretely, with integrable twisted 1-forms $\omega \in \Omega_{\mathbb{P}^n}^1(d)$, $\omega \wedge d\omega = 0$ (see [14]). It's then than the following question emerged: Given a form $\omega \in \Omega_{\mathbb{P}^n}^1(d)$ such that the vector fields that annihilate ω generate a split sheaf (i.e., a sheaf that decomposes as direct sum of line bundles), will the same feature hold for every deformation of ω ? Such a question was addressed by Cukierman and Pereira in [5]. Here, we use our results to recover the theorem of Cukierman-Pereira as a special case.

As was observed before, every time we have a codimension 1 Pfaff system

$$0 \to I(\mathcal{F}) \to \Omega^1_X \to \Omega^1_\mathcal{F} \to 0$$

such that $\Omega^1_{\mathcal{F}}$ is torsion-free, it follows that $I(\mathcal{F})$ must be a line bundle. In the case $X = \mathbb{P}^n(\mathbb{C})$, we thus have $I(\mathcal{F}) \cong \mathcal{O}_{\mathbb{P}^n}(-d)$ for some $d \in \mathbb{Z}$. It is then equivalent to give a Pfaff system and to give a morphism $0 \to \mathcal{O}_{\mathbb{P}^n}(-d) \to \Omega^1_{\mathbb{P}^n}$ which is in turn equivalent to $0 \to \mathcal{O}_{\mathbb{P}^n} \to \Omega^1_{\mathbb{P}^n}(d)$, that is, to give a global section ω of the sheaf $\Omega^1_{\mathbb{P}^n}(d)$.

We can explicitly write such a global section as

$$\omega = \sum_{i=0}^{n} f_i(x_0, \dots, x_n) dx_i,$$

with f_i a homogeneous polynomial of degree d-1 and such that $\sum_i x_i f_i = 0$.

Such an expression gives rise to a foliation with split tangent sheaf if and only if there are n-1 polynomial vector fields

$$X_{1} = g_{1}^{0} \frac{\partial}{\partial x_{0}} + \dots + g_{1}^{n} \frac{\partial}{\partial x_{n}},$$

$$\vdots$$
$$X_{n-1} = g_{n-1}^{0} \frac{\partial}{\partial x_{0}} + \dots + g_{n-1}^{n} \frac{\partial}{\partial x_{n}}$$

such that $\omega(X_i) = 0$, for all $1 \le i \le n - 1$; moreover, on a generic point, the vector fields must be linearly independent.

The singular set of this foliation is given by the ideal $I = (f_0, \ldots, f_n)$. The condition $\omega(X_i) = 0$ means that the ring $\mathbb{C}[x_0, \ldots, x_n]/I$ admits a syzygy of the form

$$0 \to \mathbb{C}[x_0, \dots, x_n]^n \xrightarrow{\begin{pmatrix} x_0 & \cdots & x_n \\ g_1^0 & \cdots & g_1^n \\ \vdots & \ddots & \vdots \\ g_{n-1}^0 & \cdots & g_{n-1}^n \end{pmatrix}} \mathbb{C}[x_0, \dots, x_n]^{n+1} \xrightarrow{(f_0, \dots, f_n)} \mathbb{C}[x_0, \dots, x_n] \to \mathbb{C}[x_0, \dots, x_n]/I \to 0.$$

For such rings a theorem of Hilbert and Schaps tells us the following:

Theorem 9.1 (Hilbert, Schaps). Let $A = k[x_0, ..., x_n]/I$ be such that there is a 3-step resolution of A as above by free modules. Then the ring A is Cohen-Macaulay; in particular, it is equidimensional.

Proof. This is theorem 5.1 in [2]

FEDERICO QUALLBRUNN

We thus recover the theorem of Cukierman and Pereira ([5, Theorem 1]).

Theorem 9.2 ([5]). Let

$$0 \to I(\mathcal{F}) \to \Omega^1_{\mathbb{P}^n \times S \mid S} \to \Omega^1_{\mathcal{F}} \to 0$$

be a flat family of codimension 1 integrable Pfaff systems, and suppose

 $0 \to I(\mathcal{F})_s \to \Omega^1_{\mathbb{P}^n} \to \Omega^1_{\mathcal{F}} \to 0$

defines a foliation with split tangent sheaf. If $\operatorname{sing}(\mathcal{F}_s) \setminus \overline{\mathcal{K}(\mathcal{F}_s)}$ has codimension greater than 2, then every member of the family defines a split tangent sheaf foliation.

Proof. By the above theorem, $\operatorname{sing}(\mathcal{F}_s)$ is equidimensional. The singular locus of a foliation on \mathbb{P}^n always has an irreducible component of codimension 2 (see [14, Teorema 1.13]). If $\operatorname{sing}(\mathcal{F}_s) \setminus \overline{\mathcal{K}(\mathcal{F}_s)}$ has codimension greater than 2, then it must be empty. So $\operatorname{sing}(\mathcal{F}_s) = \overline{\mathcal{K}(\mathcal{F}_s)}$ and we can then apply Theorem 8.13. So the flat family

$$0 \to I(\mathcal{F}) \to \Omega^1_{\mathbb{P}^n \times S|S} \to \Omega^1_{\mathcal{F}} \to 0$$

gives rise to a flat family

$$0 \to T\mathcal{F} \to T_S X \to N_{\mathcal{F}} \to 0,$$

and so $T\mathcal{F}$ must be flat over S, and then $T\mathcal{F}_s$ splits for every $s \in S$.

Remark 9.3. In [5], Theorem 2, a similar statement is proved for an arbitrary codimensional distribution (not necessarily involutive). More concretely, they prove:

Let D be a singular holomorphic distribution on \mathbb{P}^n . If $\operatorname{codim} \operatorname{sing}(D) \geq 3$ and

$$TD \cong \bigoplus_{i=1}^{d} \mathcal{O}(e_i), \qquad e_i \in \mathbb{Z}$$

then there exists a Zariski-open neighborhood \mathcal{U} of the space of Pfaff systems such that, for every Pfaff system $\mathcal{I}' \subset \Omega^1_{\mathbb{P}^n}$ in \mathcal{U} , its annihilator TD' splits as $TD' \cong \bigoplus_{i=1}^d \mathcal{O}(e_i)$.

When we try to arrive at the above statement as a particular case of the the theory hereby exposed, we run into some difficulties, which we will explain now.

Note that generic (non-involutive) singular distributions have isolated singularities. So, in order to apply an analogue of Theorem 8.13 to non-involutive distributions of arbitrary codimension, we should be able to conclude, from the hypothesis codim $sing(D) \ge 3$, that D has isolated singularities. The author was unable to do so. It seems that a keener knowledge of the singular set of arbitrary codimensional distributions is necessary to correctly understand and generalize Theorem 2 in [5].

References

- [1] Allen Altman and Steven Kleiman, Compactifying the Picard Scheme, Advances on Mathematics 35.
- [2] Michael Artin, Lectures on Deformation of Singularities, Tata institute of Fundamental Research, 1976.
- [3] Vladimir Baranovsky, Bundles on non-proper schemes: representability., Sel. Math., New Ser. 16 (2010), no. 2, 297-313 (English).
- [4] George Bergman and Arthur Ogus, Nakayama's Lemma for Half-Exact Functors, Proc. of the AMS 31 (1972).
- [5] Fernando Cukierman and Jorge Vitório Pereira, Stability of Holomorphic Foliations with Split Tangent Sheaf, American Journal of Mathematics 130 (2008).
- [6] Fernando Cukierman, Jorge Vitório Pereira, and Israel Vainsencher, Stability of foliations induced by rational maps, Annales de la faculté des sciences de Toulouse Mathématiques 18 (2009), no. 4, 685-715.
- Michel Demazure, Bifurcations and Catastrophes: Geometry of Solutions to Nonlinear Problems, Universitext Series, Springer, 2000. DOI: 10.1007/978-3-642-57134-3.

- [8] Barbara Fantechi, Lothar Gottsche, Luc Illusie, Steven L. Kleiman, Nitin Nitsure, and AngeloVistoli, Fundamental Algebraic Geometry: Grothendieck's Fga Explained, Mathematical Surveys and Monographs, American Mathematical Society, 2005.
- Xavier Gomez-Mont, The Transverse Dynamics of a Holomorphic Flow, Annals of Mathematics 127 (1988), no. 1, 49-92.
- [10] Alexandre Grothendieck, Éléments de géométrie algébrique. IV: Étude locale des schemas et des morphismes de schemas. (Seconde partie.), Publ. Math., Inst. Hautes tud. Sci. 24 (1965).
- [11] Alexandre Grothendieck and Michéle Raynaud, Séminaire de Géométrie Algébrique du Bois Marie -1960/61, SGA 1. Revetements étales et groupe fondamental., Documents Mathématiques, vol. 3, Societé Mathématique de France, 2003.
- [12] Robin Hartshorne, Generalized divisors on Gorenstein schemes., K-Theory 8 (1994), no. 3, 287-339.
- [13] J.P. Jouanolou, Équations de Pfaff algébriques, Lecture notes in mathematics, Springer, 1979.
- [14] Alcides Lins-Neto, Componentes irredutíveis dos espaços de folheaçoes, Publicações Matemáticas do IMPA, 2007.
- [15] A. de Medeiros, Structural stability of integrable differential forms, Geometry and Topology (1977), 395–428. DOI: 10.1007/bfb0085369.
- [16] Genevive Pourcin, Deformations of coherent foliations on a compact normal space, Annales de l'institut Fourier 37 (1987), 33-48. DOI: 10.5802/aif.1085.
- [17] _____, Deformations of singular holomorphic foliations on reduced compact -analytic spaces (Xavier Gomez-Mont, Jos Seade, and Alberto Verjovski, eds.), Lecture Notes in Mathematics, vol. 1345, Springer Berlin / Heidelberg, 1988.
- [18] T. Suwa, Structure of the singular set of a complex analytic foliation, Preprint series in mathematics. Hokkaido University 33 (1988).
- [19] F.W. Warner, Foundations of Differentiable Manifolds and Lie Groups, 2nd ed., Graduate Texts in Mathematics, Springer, 1983. DOI: 10.1007/978-1-4757-1799-0.

DEPARTAMENTO DE MATEMÁTICA, FCEYN, UNIVERSIDAD DE BUENOS AIRES, CIUDAD UNIVERSITARIA, PA-BELLÓN 1, BUENOS AIRES (ARGENTINA)

E-mail address: fquallb@dm.uba.ar

FREE DIVISORS IN A PENCIL OF CURVES

JEAN VALLÈS

ABSTRACT. A plane curve $D \subset \mathbb{P}^2(\mathbf{k})$, where \mathbf{k} is a field of characteristic zero, is free if its associated sheaf \mathcal{T}_D of vector fields tangent to D is a free $\mathscr{O}_{\mathbb{P}^2(\mathbf{k})}$ -module (see [6] or [5] for a definition in a more general context). Relatively few free curves are known. Here we prove that the union of all singular members of a pencil of plane projective curves with the same degree and with a smooth base locus is a free divisor.

1. INTRODUCTION

Let \boldsymbol{k} be a field of characteristic zero and let $S = \boldsymbol{k}[x, y, z]$ be the graded ring such that $\mathbb{P}^2 = \operatorname{Proj}(S)$. We write $\partial_x := \frac{\partial}{\partial x}$, $\partial_y := \frac{\partial}{\partial y}$, $\partial_z := \frac{\partial}{\partial z}$ and $\nabla F = (\partial_x F, \partial_y F, \partial_z F)$ for a homogenous polynomial $F \in S$.

Let $D = \{F = 0\}$ be a reduced curve of degree n. The kernel \mathcal{T}_D of the map ∇F is a rank two reflexive sheaf, hence a vector bundle on \mathbb{P}^2 . It is the rank two vector bundle of vector fields tangent along D, defined by the following exact sequence:

$$0 \longrightarrow \mathcal{T}_D \longrightarrow \mathscr{O}_{\mathbb{P}^2}^3 \xrightarrow{\nabla F} \mathcal{J}_{\nabla F}(n-1) \longrightarrow 0,$$

where the sheaf $\mathcal{J}_{\nabla F}$ (also denoted $\mathcal{J}_{\nabla D}$ in this text) is the Jacobian ideal of F. Set theoretically $\mathcal{J}_{\nabla F}$ defines the singular points of the divisor D. For instance if D consists of s generic lines then $\mathcal{J}_{\nabla D}$ defines the set of $\binom{s}{2}$ vertices of D.

Remark 1.1. A non zero section $s \in \mathrm{H}^{0}(\mathcal{T}_{D}(a))$, for some shift $a \in \mathbb{N}$, corresponds to a derivation $\delta = P_{a}\partial_{x} + Q_{a}\partial_{y} + R_{a}\partial_{z}$ verifying $\delta(F) = 0$, where $(P_{a}, Q_{a}, R_{a}) \in \mathrm{H}^{0}(\mathscr{O}_{\mathbb{P}^{2}}(a))^{3}$.

In some particular cases that can be found in [5], \mathcal{T}_D is a free $\mathscr{O}_{\mathbb{P}^2}$ -module; it means that there are two vector fields of degrees a and b that form a basis of $\bigoplus_n \mathrm{H}^0(\mathcal{T}_D(n))$ (D is said to be free with exponents (a, b)); it arises, for instance, when D is the union of the nine inflection lines of a smooth cubic curve. The notion of free divisor was introduced by Saito [6] for reduced divisors and studied by Terao [9] for hyperplane arrangements. Here we recall a definition of freeness for projective curves. For a more general definition we refer to Saito [6].

Definition 1.2. A reduced curve $D \subset \mathbb{P}^2$ is free with exponents $(a, b) \in \mathbb{N}^2$ if

$$\mathcal{T}_D \simeq \mathscr{O}_{\mathbb{P}^2}(-a) \oplus \mathscr{O}_{\mathbb{P}^2}(-b).$$

A smooth curve of degree ≥ 2 is not free, an irreducible curve of degree ≥ 3 with only nodes and cusps as singularities is not free (see [1, Example 4.5]). Actually few examples of free curves are known and of course very few families of free curves are known. One such family can be found in [8, Prop. 2.2].

²⁰¹⁰ Mathematics Subject Classification. 14C21, 14N20, 32S22, 14H50.

Key words and phrases. Arrangements of curves, Pencil of curves, Freeness of arrangements, Logarithmic sheaves.

Author partially supported by ANR GEOLMI ANR-11-BS03-0011 and PHC-SAKURA 31944VE.

In a personal communication it was conjectured by E. Artal and J.I. Cogolludo that the union of all the singular members of a pencil of plane curves (assuming that the general one is smooth) should be free. Three different cases occur:

- the base locus is smooth (for instance the union of six lines in a pencil of conics passing through four distinct points);
- the base locus is not smooth but every curve in the pencil is reduced (for instance the four lines in a pencil of conics where two of the base points are infinitely near points);
- the base locus is not smooth and there exists exactly one non reduced curve in the pencil (for instance three lines in a pencil of bitangent conics).

In the third case the divisor of singular members is not reduced but its reduced structure is expected to be free.

We point out that if two distinct curves of the pencil are not reduced then all curves will be singular. Even in this case, we believe that a free divisor can be obtained by chosing a finite number of reduced components through all the singular points.

In this paper we prove that the union of all the singular members of a pencil of degree n plane curves with a smooth base locus (i.e. the base locus consists of n^2 distinct points) is a free divisor and we give its exponents (see theorem 2.7). More generally, we describe the vector bundle of logarithmic vector fields tangent to any union of curves of the pencil (see theorem 2.8) by studying one particular vector field "canonically tangent" to the pencil, that is introduced in the key lemma 2.1.

This gives already a new and easy method to produce free divisors.

I thank J. I. Cogolludo for his useful comments.

2. Pencil of plane curves

2.1. Generalities and notations. Let $\{f = 0\}$ and $\{g = 0\}$ be two reduced curves of degree $n \ge 1$ with no common component. For any $(\alpha, \beta) \in \mathbb{P}^1$ the curve $C_{\alpha,\beta}$ is defined by the equation $\{\alpha f + \beta g = 0\}$ and $\mathcal{C}(f,g) = \{C_{\alpha,\beta} | (\alpha, \beta) \in \mathbb{P}^1\}$ is the pencil of all these curves.

In section 2 we will assume that the general member of the pencil $\mathcal{C}(f,g)$ is a smooth curve and that $C_{\alpha,\beta}$ is reduced for every $(\alpha,\beta) \in \mathbb{P}^1$.

Under these assumptions there are finitely many singular curves in C(f,g) but also finitely many singular points. We recall that the degree of the discriminant variety of degree n curves is $3(n-1)^2$ (it is a particular case of the Boole formula; see [10, Example 6.4]). Since the general curve in the pencil is smooth, the line defined by the pencil C(f,g) in the space of degree ncurves meets the discriminant variety along a finite scheme of length $3(n-1)^2$ (not empty for $n \geq 2$). The number of singular points is of course related to the multiplicity of the singular curves in the pencil as we will see below.

Let us fix some notation. The scheme defined by the ideal sheaf $\mathcal{J}_{\nabla C_{\alpha_i,\beta_i}}$ is denoted by Z_{α_i,β_i} . It is well known that this scheme is locally a complete intersection (for instance, generalize to curves the lemma 2.4 in [7]). The union of all the singular members of the pencil $\mathcal{C}(f,g)$ form a divisor D^{sg} . A union of $k \geq 2$ distinct members of $\mathcal{C}(f,g)$ is denoted by D_k .

2.2. Derivation tangent to a smooth pencil. Let us consider the following derivation, associated "canonically" to the pencil:

Lemma 2.1. For any union D_k of $k \ge 1$ members of the pencil there exists a non zero section $s_{\delta,k} \in \mathrm{H}^0(\mathcal{T}_{D_k}(2n-2))$ induced by the derivation

$$\delta = (\nabla f \wedge \nabla g) \cdot \nabla = (\partial_y f \partial_z g - \partial_z f \partial_y g) \partial_x + (\partial_z f \partial_x g - \partial_x f \partial_z g) \partial_y + (\partial_x f \partial_y g - \partial_y f \partial_x g) \partial_z \cdot \partial_y f \partial_z g - \partial_y f \partial_y g - \partial_y f \partial_z g - \partial_$$

Proof. Since $\delta(\alpha f + \beta g) = \det(\nabla f, \nabla g, \nabla(\alpha f + \beta g)) = 0$ we have for any $k \ge 1$,

$$\delta(f) = \delta(g) = \delta(\alpha f + \beta g) = \delta(\prod_{i=1}^{k} (\alpha_i f + \beta_i g)) = 0.$$

According to the remark 1.1 it gives the desired section.

Let us introduce a rank two sheaf \mathcal{F} defined by the following exact sequence:

$$0 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(2-2n) \xrightarrow{\nabla f \wedge \nabla g} \mathscr{O}_{\mathbb{P}^2}^3 \longrightarrow \mathcal{F} \longrightarrow 0.$$

If we denote by $sg(\mathcal{F}) := \{p \in \mathbb{P}^2 | rank(\mathcal{F} \otimes \mathcal{O}_p) > 2\}$ the set of singular points of \mathcal{F} , we have:

Lemma 2.2. A point $p \in \mathbb{P}^2$ belongs to $sg(\mathcal{F})$ if and only if two smooth members of the pencil share the same tangent line at p or one curve of the pencil is singular at p. Moreover $sg(\mathcal{F})$ is a finite closed scheme with length $l(sg(\mathcal{F})) = 3(n-1)^2$.

Remark 2.3. Let us precise that if two smooth members intersect then all the smooth members of the pencil intersect with the same tangency.

Remark 2.4. If the base locus of C(f,g) consists of n^2 distinct points then two curves of the pencil meet transversaly at the base points and $p \in \operatorname{sg}(\mathcal{F})$ if and only if p is a singular point for a unique curve $C_{\alpha,\beta}$ in the pencil and does not belong to the base locus. One can assume that $p \in \operatorname{sg}(\mathcal{F})$ is singular for $\{f = 0\}$. Then, locally at p, the curve $\{g = 0\}$ can be assumed to be smooth and the local ideals $(\nabla f \wedge \nabla g)_p$ and $(\nabla f)_p$ coincide. In other words, when the base locus is smooth, we have

$$\operatorname{sg}(\mathcal{F}) = \sqcup_{i=1,\ldots,s} Z_{\alpha_i,\beta_i}.$$

Proof. The singular locus of \mathcal{F} is also defined by $\operatorname{sg}(\mathcal{F}) := \{p \in \mathbb{P}^2 | (\nabla f \wedge \nabla g)(p) = 0\}$. The zero scheme defined by $\nabla f \wedge \nabla g$ and $\nabla(\alpha f + \beta g) \wedge \nabla g$ are clearly the same; it means that the singular points of any member in the pencil is a singular point for \mathcal{F} . One can also obtain $(\nabla f \wedge \nabla g)(p) = 0$ at a smooth point when the vectors $(\nabla f)(p)$ and $(\nabla g)(p)$ are proportional i.e. when two smooth curves of the pencil share the same tangent line at p.

Since every curve of the pencil is reduced sg(\mathcal{F}) is finite, its length can be computed by writing the resolution of the ideal $\mathcal{J}_{sg(\mathcal{F})}$ (for a sheaf of ideal \mathcal{J}_Z defining a finite scheme Z of length l(Z), we have $c_2(\mathcal{J}_Z) = l(Z)$). Indeed, if we dualize the following exact sequence

$$0 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(2-2n) \xrightarrow{\nabla f \wedge \nabla g} \mathscr{O}^3_{\mathbb{P}^2} \longrightarrow \mathcal{F} \longrightarrow 0$$

we find, according to Hilbert-Burch theorem,

$$0 \longrightarrow \mathcal{F}^{\vee} \xrightarrow{(\nabla f, \nabla g)} \mathscr{O}_{\mathbb{P}^2}^3 \xrightarrow{\nabla f \wedge \nabla g} \mathscr{O}_{\mathbb{P}^2}(2n-2).$$

It proves that $\mathcal{F}^{\vee} = \mathscr{O}_{\mathbb{P}^2}(1-n)^2$ and that the image of the last map is $\mathcal{J}_{\mathrm{sg}(\mathcal{F})}(2n-2)$.

Then $l(sg(\mathcal{F})) = 3(n-1)^2$. We point out that this number is the degree of the discriminant variety of degree n curves.

Now let us call D_k the divisor defined by $k \ge 2$ members of the pencil and let us consider the section $s_{\delta,k} \in \mathrm{H}^0(\mathcal{T}_{D_k}(2n-2))$ corresponding (see remark 1.1) to the derivation δ . Let $Z_k := Z(s_{\delta,k})$ be the zero locus of $s_{\delta,k}$.

Lemma 2.5. The section $s_{\delta,k}$ vanishes in codimension at least two.

192

where $\mathcal{Q} = \operatorname{coker}(s_{\delta,k})$. Assume that Z_k contains a divisor H. Tensor now the last vertical exact sequence of the above diagram by \mathscr{O}_p for a general point $p \in H$. Since p does not belong to the Jacobian scheme defined by $\mathcal{J}_{\nabla D_k}$ we have $\mathcal{J}_{\nabla D_k} \otimes \mathscr{O}_p = \mathscr{O}_p$ and $\operatorname{Tor}_1(\mathcal{J}_{\nabla D_k}, \mathscr{O}_p) = 0$. Since $p \in H \subset Z_k$ we have $\operatorname{rank}(\mathcal{Q} \otimes \mathscr{O}_p) \geq 2$; it implies $\operatorname{rank}(\mathcal{F} \otimes \mathscr{O}_p) \geq 3$ in other words that $p \in \operatorname{sg}(\mathcal{F})$; this contradicts $\operatorname{codim}(\operatorname{sg}(\mathcal{F}), \mathbb{P}^2) = 2$, proved in lemma 2.2.

Then Q is the ideal sheaf of the codimension two scheme Z_k , i.e. $Q = \mathcal{J}_{Z_k}(n(2-k)-1)$ and we have an exact sequence

$$0 \longrightarrow \mathcal{J}_{Z_k}(n(2-k)-1) \longrightarrow \mathcal{F} \longrightarrow \mathcal{J}_{\nabla D_k}(nk-1) \longrightarrow 0.$$

From this commutative diagram we obtain the following lemma.

Lemma 2.6. Let D_k be a union of $k \ge 2$ members of $\mathcal{C}(f,g)$. Then

$$c_2(\mathcal{J}_{\nabla D_k}) + c_2(\mathcal{J}_{Z_k}) = 3(n-1)^2 + n^2(k-1)^2.$$

Proof. According to the above commutative diagram we compute $c_2(\mathcal{F})$ in two different ways. The horizontal exact sequence gives $c_2(\mathcal{F}) = 4(n-1)^2$ when the vertical one gives

$$c_2(\mathcal{F}) = c_2(\mathcal{J}_{\nabla D_k}) + c_2(\mathcal{J}_{Z_k}) + (n-1)^2 - n^2(k-1)^2$$

The lemma is proved by eliminating $c_2(\mathcal{F})$.

2.3. Free divisors in the pencil. When D_k contains the divisor D^{sg} of all the singular members of the pencil we show now that it is free with exponents (2n - 2, n(k - 2) + 1).

Theorem 2.7. Assume that the base locus of the pencil C(f,g) is smooth. Then,

$$D_k \supseteq D^{\mathrm{sg}} \Leftrightarrow \mathcal{T}_{D_k} = \mathscr{O}_{\mathbb{P}^2}(2-2n) \oplus \mathscr{O}_{\mathbb{P}^2}(n(2-k)-1).$$

Proof. Assume first that $D_k \supseteq D^{\text{sg}}$. Then the singular locus of D_k defined by the Jacobian ideal consists of the base points of the k curves in the pencil and, since it contains all the singular members, of the whole set of singularities of the curves in the pencil. This last set has length $3(n-1)^2$ by lemma 2.2 and remark 2.4. Moreover the subscheme supported by the base points in the scheme defined by $\mathcal{J}_{\nabla D_k}$ has length $n^2(k-1)^2$ since at each point among the n^2 points

JEAN VALLÈS

of the base locus of the pencil, the k curves meet transversaly and define k different directions (i.e. the local ring at the point (0, 0, 1) is isomorphic to $k[x, y]/(x^{k-1}, y^{k-1})$). Then

$$c_2(\mathcal{J}_{\nabla D_k}) = 3(n-1)^2 + n^2(k-1)^2,$$

which implies that $Z_k = \emptyset$. In other words there is an exact sequence

$$0 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(2-2n) \xrightarrow{s_{\delta,k}} \mathscr{T}_{D_k} \longrightarrow \mathscr{O}_{\mathbb{P}^2}(n(2-k)-1) \longrightarrow 0.$$

And such an exact sequence splits.

Conversely, assume that there is a singular member C that does not belong to D_k . Let $p \in C$ be one of its singular point. Since $p \notin D_k$ we have $\mathcal{J}_{\nabla D_k} \otimes \mathcal{O}_p = \mathcal{O}_p$ and $\operatorname{Tor}_1(\mathcal{J}_{\nabla D_k}, \mathcal{O}_p) = 0$. Consider the following exact sequence that comes from the commutative diagram above (in the proof of lemma 2.5) when $Z_k = \emptyset$:

$$0 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(n(2-k)-1) \longrightarrow \mathcal{F} \longrightarrow \mathcal{J}_{\nabla D_k}(nk-1) \longrightarrow 0.$$

If we tensor this exact sequence by \mathscr{O}_p we find $\operatorname{rank}(\mathcal{F} \otimes \mathscr{O}_p) = 2$. This contradicts $p \in \operatorname{sg}(\mathcal{F})$ that was proved in lemma 2.2.

2.4. Singular members ommitted. When $D_k \supset D^{sg}$ we have seen in theorem 2.7 that $Z_k = \emptyset$ by computing the length of the scheme defined by the Jacobian ideal of D_k . More generally we can describe, at least when the base locus is smooth, the scheme Z_k for any union of curves of the pencil.

Theorem 2.8. Assume that the base locus of the pencil C(f,g) is smooth. Assume also that $D_k \supset D^{sg} \setminus \bigcup_{i=1,\ldots,r} C_{\alpha_i,\beta_i}, C_{\alpha_i,\beta_i} \notin D_k$ and C_{α_i,β_i} is a singular curve for $i = 1,\ldots,r$. Then,

$$\mathcal{J}_{Z_k} = \mathcal{J}_{\nabla C_{\alpha_1,\beta_1}} \otimes \cdots \otimes \mathcal{J}_{\nabla C_{\alpha_r,\beta_r}}$$

Remark 2.9. When r = 0 we obtain the freeness again.

Proof. Since the set of singular points of two disctinct curves are disjoint, it is enough to prove it for r = 1 (i.e. $D_k \supset D^{\text{sg}} \setminus C_{\alpha_1,\beta_1}$ and $C_{\alpha_1,\beta_1} \notin D_k$). Recall that $\mathcal{E}xt^1(\mathcal{J}_Z, \mathscr{O}_{\mathbb{P}^2}) = \omega_Z$ where Z is a finite scheme and ω_Z its dualizing sheaf (see [2, Chapter III, section 7]); it is well known that, since the finite scheme Z is locally complete intersection, $\omega_Z = \mathscr{O}_Z$.

Then the dual exact sequence of

$$0 \longrightarrow \mathcal{J}_{Z_k}(n(2-k)-1) \longrightarrow \mathcal{F} \longrightarrow \mathcal{J}_{\nabla D_k}(nk-1) \longrightarrow 0$$

is the long exact sequence

$$0 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(1-nk) \longrightarrow \mathscr{O}_{\mathbb{P}^2}(1-n)^2 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(n(k-2)+1) \longrightarrow \omega_{\nabla D_k} \longrightarrow \mathscr{O}_{\mathrm{sg}(\mathcal{F})} \longrightarrow \omega_{Z_k} \longrightarrow 0.$$

The map $\mathcal{F} \longrightarrow \mathcal{J}_{\nabla D_k}(nk-1)$ can be described by composition; indeed it is given by two polynomials (U, V) such that

$$(U, V).(\nabla f, \nabla g) = \nabla(\prod_{i} (\alpha_i f + \beta_i g)).$$

We find, $U = \sum_{i} \alpha_i \prod_{j \neq i} (\alpha_j f + \beta_j g)$ and $V = \sum_{i} \beta_i \prod_{j \neq i} (\alpha_j f + \beta_j g)$. If a point p belongs to one curve C_{α_1,β_1} in D_k and does not belong to the base locus, then $U(p) \neq 0$ and $V(p) \neq 0$. It shows that these two polynomials vanish simultaneously and precisely along the base locus. Then the complete intersection $T = \{U = 0\} \cap \{V = 0\}$ of length $n^2(k-1)^2$ is supported exactly by the base points. We have

$$0 \longrightarrow \omega_{\nabla D_k} / \mathscr{O}_T \longrightarrow \mathscr{O}_{\mathrm{sg}(\mathcal{F})} \longrightarrow \omega_{Z_k} \longrightarrow 0.$$

We have already seen that the subscheme supported by the base points in the scheme defined by $\mathcal{J}_{\nabla D_k}$ has length $n^2(k-1)^2$. It implies that $\omega_{\nabla D_k}/\mathcal{O}_T = \bigoplus_{i=2,...,s} \mathcal{O}_{Z_{\alpha_i,\beta_i}}$. According to remark 2.4, $\mathcal{O}_{\mathrm{sg}(\mathcal{F})} = \bigoplus_{i=1,...,s} \mathcal{O}_{Z_{\alpha_i,\beta_i}}$. This proves $\omega_{Z_k} = \mathcal{O}_{Z_{\alpha_1,\beta_1}}$.

There are exact sequences relating the vector bundles \mathcal{T}_{D_k} and $\mathcal{T}_{D_k\setminus C}$ when $C \subset D_k$.

Proposition 2.10. We assume that the base locus of the pencil C(f,g) is smooth and that D_k contains D^{sg} . Let C be a singular member in C(f,g) and Z its scheme of singular points. Then there is an exact sequence

$$0 \longrightarrow \mathcal{T}_{D_k} \longrightarrow \mathcal{T}_{D_k \setminus C} \longrightarrow \mathcal{J}_{Z/C}(n(3-k)-1) \longrightarrow 0,$$

where $\mathcal{J}_{Z/C} \subset \mathscr{O}_C$ defines Z into C.

Proof. The derivation $(\nabla f \wedge \nabla g)$. ∇ is tangent to D_k then also to $D_k \setminus C$. It induces the following commutative diagram which proves the proposition:



3. The pencil contains a non-reduced curve

When the pencil C(f,g) contains a non-reduced curve, the arguments used in the previous sections are not valid since the scheme defined by the jacobian ideal contains a divisor. We have to remove this divisor somehow. Remember that if two curves of the pencil are multiple then the general curve is singular. So let us consider that there is only one curve that is not reduced. Let $hh_1^{r_1} \cdots h_s^{r_s} = 0$ be the equation of this unique non-reduced curve where h = 0 is reduced, $\deg(h_i) = m_i \ge 1$ and $r_i \ge 2$. Since the derivation $\frac{1}{\prod_i h_i^{r_i}} (\nabla f \wedge \nabla g) \cdot \nabla$ is still tangent to all curves of the pencil, we believe that the following statement is true:

Conjecture. Let $hh_1^{r_1} \cdots h_s^{r_s} = 0$ be the equation of the unique non-reduced curve where $\{h = 0\}$ is reduced, $\deg(h_i) = m_i \ge 1$ and $r_i \ge 2$. Then,

$$D_k \supseteq D^{\mathrm{sg}} \Leftrightarrow \mathcal{T}_{D_k} = \mathscr{O}_{\mathbb{P}^2}(2 - 2n + \sum_{i=1}^{i=s} (r_i - 1)m_i) \oplus \mathscr{O}_{\mathbb{P}^2}(n(2-k) - 1).$$

JEAN VALLÈS

4. Examples

Let us call $\Sigma_3 \subset \mathbb{P}^9 = \mathbb{P}(\mathrm{H}^0(\mathscr{O}_{\mathbb{P}^2}(3)))$ the hypersurface of singular cubics. It is well known that its degree is 12 (see [3], for instance).

• Pappus arrangement freed by nodal cubics: Let us consider the divisor of the nine lines appearing in the Pappus arrangement; this divisor is the union of three triangles T_1, T_2, T_3 with nine base points. The pencil generated by T_1 and T_2 contains 3 triangles (each one represents a triple point in Σ_3); since 9 < 12, singular cubics are missing in the pencil. There is no other triangle and no smoth conic+line in the Pappus pencil, when it is general enough. We can conclude that the missing cubics are, in general, nodal cubics C_1, C_2, C_3 .

Let $D = T_1 \cup T_2 \cup T_3 \cup C_1 \cup C_2 \cup C_3$ be the union of all singular fibers in the pencil generated by T_1 and T_2 . Then, according to theorem 2.7 we have

$$\mathcal{T}_D = \mathscr{O}_{\mathbb{P}^2}(-4) \oplus \mathscr{O}_{\mathbb{P}^2}(-13).$$

• **Pappus arrangement**: Let $T_1 \cup T_2 \cup T_3$ be the divisor consisting of the nine lines of the projective Pappus arrangement and $D = T_1 \cup T_2 \cup T_3 \cup C_1 \cup C_2 \cup C_3$ be the union of all singular fibers in the pencil generated by two triangles among the T_i 's. Let us call $K := C_1 \cup C_2 \cup C_3$ the union of the nodal cubics and z_1, z_2, z_3 their nodes. Then we have, according to theorem 2.8, $T_D = \mathscr{O}_{\mathbb{P}^2}(-4) \oplus \mathscr{O}_{\mathbb{P}^2}(-13)$ and an exact sequence

$$0 \longrightarrow \mathscr{O}_{\mathbb{P}^2}(-4) \longrightarrow \mathcal{T}_{D\setminus K} \longrightarrow \mathcal{I}_{z_1, z_2, z_3}(-4) \longrightarrow 0.$$

The logarithmic bundle $\mathcal{T}_{D\setminus K}$ associated to the Pappus configuration is semi-stable and its divisor of jumping lines is the triangle $z_1^{\vee} \cup z_2^{\vee} \cup z_3^{\vee}$ as it is proved by retricting the above exact sequence to any line through one of the zeroes.



FIGURE 1. Pappus arrangement

• Hesse arrangement: Let us consider the pencil generated by a smooth cubic C and its hessian Hess(C). The pencil contains 4 triangles T_1, T_2, T_3, T_4 and since the degree of Σ_3 is 12, no other singular cubic can be present. Let us call D the union of these four triangles. Then, according to theorem 2.7 we have

$$\mathcal{T}_D = \mathscr{O}_{\mathbb{P}^2}(-4) \oplus \mathscr{O}_{\mathbb{P}^2}(-7).$$

References

- Dimca, A. and Sernesi, E. Syzygies and logarithmic vector fields along plane curves, J. Éc. polytech. Math. 1 (2014), 247–267. DOI: 10.5802/jep.10
- [2] Hartshorne, R. Algebraic geometry. Graduate Texts in Mathematics, 52. Springer-Verlag, 1977.
- [3] Gelfand, I. M., Kapranov, M. M. and Zelevinsky, A. V. Discriminants, resultants, and multidimensional determinants. Mathematics: Theory and Applications. Birkhäuser Boston, Inc., Boston, MA, 1994.
- [4] Okonek, C., Schneider, M. and Spindler, H. Vector bundles on complex projective spaces. Progress in Mathematics, 3. Birkhäuser, Boston, Mass., 1980.
- [5] Orlik, P. and Terao, H. Arrangements of hyperplanes. Grundlehren der Mathematischen Wissenschaften, 300. Springer-Verlag, Berlin, 1992.
- Saito, K. Theory of logarithmic differential forms and logarithmic vector fields, J. Fac. Sci. Univ. Tokyo, Sect. IA Math. 27 (1980), 265–291.
- Schenck, H. A rank two vector bundle associated to a three arrangement and its Chern polynomial, Adv. Math., 149, (2000), no. 2, 214–229.
- [8] Simis, A. and Tohăneanu, Ş. O. Homology of homogeneous divisors. Israel J. Math. 200 (2014), no. 1, 449–487
- [9] Terao, H. Arrangements of hyperplanes and their freeness. I.II. J. Fac. Sci. Univ. Tokyo Sect. IA Math. 27 (1980), no. 2, 293–312.
- [10] Tevelev, E. A. Projective duality and homogeneous spaces. Encyclopaedia of Mathematical Sciences, 133, Invariant Theory and Algebraic Transformation Groups, IV, Springer-Verlag, Berlin, 2005.

E-mail address: jean.valles@univ-pau.fr

Université de Pau et des Pays de l'Adour, Avenue de l'Université - BP 576 - 64012 PAU Cedex - France

URL: http://web.univ-pau.fr/~jvalles

ISSN #1949-2006



www.journalofsing.org Worldwide Center of Mathematics, LLC